

## Topological Levinson's theorem via index pairings and spectral flow

Angus Stuart Alexander

Supervisor:
Assoc. Prof. Adam Rennie
Co-supervisors:
Prof. Alan Carey & Dr. Alexander Mundey

This thesis is presented as part of the requirements for the conferral of the degree:

Doctor of Philosophy (Mathematics)

The University of Wollongong School of Mathematics and Applied Statistics

#### Declaration

I, Angus Stuart Alexander, declare that this thesis submitted in partial fulfilment of the requirements for the conferral of the degree Doctor of Philosophy (Mathematics), from the University of Wollongong, is wholly my own work unless otherwise referenced or acknowledged. This document has not been submitted for qualifications at any other academic institution.

 $Angus\ Stuart\ Alexander$ 

April 29, 2024

## Abstract

Using techniques from noncommutative geometry, we explore how Levinson's theorem from scattering theory can be interpreted in a topological manner. We use low-energy resolvent expansions to deduce that the wave operator for short range scattering has a particular universal form. The wave operator does not have this form when certain obstructions occur in the resolvent expansions in even dimensions. Using the form of the wave operator, we apply index theoretic techniques to interpret Levinson's theorem as an index pairing between the K-theory class of the unitary scattering operator and the K-homology class of the generator of dilations on the half-line. A careful analysis of the trace class properties of the scattering operator allows us to provide new proofs of Levinson's theorem in all dimensions. We also compute the spectral flow for Euclidean Schrödinger operators, giving another new proof of Levinson's theorem in all dimensions.

## Acknowledgments

Firstly, I am incredibly greatful to my supervisor Adam Rennie for his constant guidance and insight. Thank you for your patience, the countless hours spent teaching and discussing mathematics with me and for editing many drafts of this thesis.

I am also grateful to my co-supervisors Alan Carey and Alex Mundey, for their many helpful discussions and careful reading of draft versions of this thesis.

I would also like to thank Johannes Kellendonk for his hospitality and enlightening discussions during my visit to Lyon, and Serge Richard for providing many insights during his visit to Wollongong. Thank you to Galina Levitina for many helpful discussions during my visits to Canberra.

A big thanks go to my parents for their constant support and my brothers and my friends for making sure that this journey was an enjoyable one.

Finally, to Alessia. Thank you for your endless love, support and encouragement throughout this past year. This thesis would not have been possible without all of you.

## Contents

Abstract							
1	Introduction						
	1.1	Motiv	ation and history	1			
	1.2	Thesis	s structure and main results	4			
<b>2</b>	Scattering theory for Schrödinger operators						
	2.1	Abstra	act scattering theory	8			
		2.1.1	Stationary scattering theory	13			
	2.2	Hamil	tonians and dilation	16			
		2.2.1	The free Hamiltonian	16			
		2.2.2	The perturbed Hamiltonian	23			
		2.2.3	The dilation operator	24			
	2.3	Existe	ence and completeness for Schrödinger operators	30			
	2.4	.4 Spectral properties of the perturbed Hamiltonian					
		2.4.1	The limiting absorption principle	35			
		2.4.2	Generalised eigenfunctions and stationary wave operators	42			
		2.4.3	The scattering matrix	52			
	2.5	Levinson's theorem					
		2.5.1	Time delay	55			
		2.5.2	The spectral shift function	57			
		2.5.3	Trace class properties of the scattering matrix	63			
		2.5.4	Levinson's theorem	75			
		2.5.5	Examples	79			
3	Resolvent expansions and resonances						
	3.1	Algeb	raic and analytic preliminaries	84			
	3.2	Expan	sions of the operator $M(k)$	90			
		3.2.1	Dimension $n \geq 5$ odd	91			
		3.2.2	Dimension $n \geq 5$ even	97			
		3.2.3	Dimension $n = 4$	101			

CONTENTS vi

		3.2.4	Dimension $n=3$	115			
		3.2.5	Dimension $n=2$	129			
		3.2.6	Dimension $n = 1 \dots \dots \dots \dots \dots$	143			
4	The	The structure of the wave operator					
	4.1	The se	cattering matrix	151			
		4.1.1	Low energy expansions of the trace operator	151			
		4.1.2	Low energy expansions of the scattering matrix	153			
	4.2	Intuiti	ion for the form of the wave operator	156			
	4.3	The fo	orm of the wave operator	166			
		4.3.1	Dimension $n = 1 \dots \dots \dots \dots \dots$	166			
		4.3.2	Dimension $n=2$	167			
		4.3.3	Dimension $n \geq 3$	167			
			4.3.3.1 Setting up the proof	167			
			4.3.3.2 Analysing the diagonalisation maps	168			
			4.3.3.3 Interchanging limits	173			
			4.3.3.4 The function of dilation and completion of proof	176			
5	Levinson's theorem as an index pairing 1						
	5.1	Levins	son's theorem as an index theorem	181			
	5.2	2 Constructing an index pairing					
		5.2.1	The spectral triple	185			
		5.2.2	Generalised wave operators	186			
		5.2.3	The index pairing	190			
	5.3	3 The index pairing for scattering operators					
		5.3.1	Low energy corrections in dimension $n = 1 \dots \dots \dots$ .	195			
		5.3.2	Low and high energy corrections in dimension $n=3$	200			
		5.3.3	High energy corrections in dimension $n=2$	205			
		5.3.4	High energy corrections in dimension $n \ge 4$	207			
6	Spe	Spectral flow for Schrödinger operators 2					
	6.1	Spectral flow					
	6.2	Making the operators Fredholm					
	6.3	Pseudodifferential operator calculus					
	6.4	Specti	ral flow for Schrödinger operators	220			
		6.4.1	The Birman-Kreĭn term	222			
		6.4.2	The "integral of one form" term	230			
		6.4.3	The spectral flow formulae	232			
	6.5	Levins	son's theorem	242			

CONTENTS	vii
A Computations of coefficients  A.1 Proof of Lemma 6.4.20	<b>247</b> 247
Bibliography	256

## Chapter 1

## Introduction

## 1.1 Motivation and history

The overall aim of this thesis is to investigate the topological nature of various aspects of scattering theory for Euclidean Schrödinger operators using techniques from noncommutative geometry and operator algebras. The results of this thesis build in particular on those of Kellendonk, Richard and Tiedra de Aldecoa [96, 97, 98, 140, 141, 142] who initiated the topological study of scattering systems with the observation that the wave operator is a Fredholm operator belonging to a particular  $C^*$ -algebra from which its index can be computed directly as a topological invariant.

Mathematical scattering theory has its origins in the physics experiments of the early 20th century, and is now a branch of operator theory devoted to the study of the spectrum of perturbations of self-adjoint linear operators acting on a Hilbert space. The development of scattering theory began with the introduction of the wave operators by Friedrichs [66, 67] and Møller [117], which sparked intense development of the rigorous theory of scattering in the next decades through the works of Kato, Kuroda, Cook, Jauch, Simon, Enss, Amrein, Birman and Kreĭn, to name a few.

Throughout this thesis, we will consider the pair of operators  $H_0 = -\sum_{j=1}^n \frac{\partial^2}{\partial x_j^2}$  and  $H = H_0 + V$  acting on (some dense domain in) the Hilbert space  $\mathcal{H} = L^2(\mathbb{R}^n)$ . Here V is the operator of multiplication by a real function which decays sufficiently fast at infinity (see Assumption 2.2.14) and in some later results is required to be smooth as well. Given our assumptions the operator  $H_0$  has no eigenvalues and the operator H has a finite number of eigenvalues. The absolutely continuous spectrum of H and  $H_0$  given by  $\sigma_{ac}(H) = \sigma_{ac}(H_0) = [0, \infty) =: \mathbb{R}^+$ . When they exist the wave operators are partial isometries defined by

$$W_{\pm} = \underset{t \to \pm \infty}{\text{s-lim}} e^{itH} e^{-itH_0}$$

and the unitary scattering operator is given by  $S = W_+^*W_-$ . Since the scattering operator

commutes with  $H_0$ , there exists a family  $(S(\lambda))_{\lambda \in \mathbb{R}^+}$  of unitary operators on  $L^2(\mathbb{S}^{n-1})$  (here  $\mathbb{S}^{n-1}$  is the unit sphere in  $\mathbb{R}^n$ ). For each  $\lambda \in \mathbb{R}^+$ ,  $S(\lambda)$  is called the scattering matrix at energy  $\lambda$ .

The story for this thesis begins with the work of Levinson [108], who showed that in one dimension the number of eigenvalues N of H (counted with multiplicity) was related to the 'scattering phase' of the pair  $(H, H_0)$  by the formula

$$-N = \frac{1}{\pi} (\delta(\infty) - \delta(0)) + \frac{1}{2} M_R(0),$$

where  $\delta$  is the scattering phase (given by the argument of the determinant of S) and  $M_R(0)$  takes either the value zero or one. There are several interesting features to note here. The result is perhaps initially surprising, since scattering data is related to the continuous spectrum of H and eigenvalues to the discrete spectrum of H. This interplay between the discrete and continuous spectrum of H is indicative of the topological nature of Levinson's theorem. Secondly, there is a correction term  $M_R(0)$ , which can be shown to correspond to the existence of a 'resonance', namely a solution to the equation  $H\psi = 0$  which is not square integrable. Such a correction term is only a feature in low dimensions.

Levinson's theorem was generalised several times in the intervening years. Firstly, Newton [123] showed in dimension n=3 that the same relation holds for a spherically symmetric potential V, essentially by changing to polar coordinates and reducing to a one dimensional problem. For a general non-spherically symmetric potential in dimension n=3, Levinson's theorem was proved by Bollé and Osborn [31], however their analysis excludes the exceptional case of the existence of a 'resonance'. To complete their analysis one needs knowledge of the spectral shift function at zero, as described by Newton [124], which contributes an extra  $\frac{1}{2}$  to Levinson's theorem in the exceptional case. The analysis of Bollé and Osborn demonstrated that Levinson's theorem is sensitive to the high energy behaviour of the scattering matrix also, a feature which is present for all  $n \geq 2$ .

Jensen and Kato [88] initiated a study of the low-energy behaviour of the resolvent  $R(z) = (H - z)^{-1}$  and in dimension n = 3 demonstrated that 'resonances' occur as obstructions to the invertibility of the operator  $T(z) = \operatorname{Id} + R_0(z)V$  near z = 0, providing an explicit characterisation of resonances as distributional solutions  $\psi$  to the equation  $H\psi = 0$  which are not square integrable but live in a weighted Sobolev space. Subsequently, Jensen showed [85] that no such obstructions can occur in dimension  $n \geq 5$  and that in dimension n = 4 [87] the analysis is more involved and that resonances can occur, with an extra logarithmic singularity.

Bollé, Danneels, Gesztesy and Wilk [27, 26] showed that in two dimensions a form of Levinson's theorem holds, with an extra striking feature. In dimension n = 2, two kinds of 'resonances' can occur. In the spherically symmetric case, a resonance which behaves like an s-wave can occur and will not contribute to Levinson's theorem, and a resonance which

behaves like a p-wave can also occur and contributes an integer to Levinson's theorem in the same way as an eigenvalue. The p-resonances behave in much the same way as resonances in dimension n=4 due to the nature of the singularity at zero.

Jensen and Nenciu provide in [89] a systematic procedure for inverting an operator related to T(z). The procedure of Jensen and Nenciu allows one to invert T(z) by isolating the obstructions to invertibility into the range of a finite sequence of decreasing projections. The range of these projections can be characterised in terms of zero energy eigenvalues and zero energy resonances. In particular [89] provides a detailed expansion in the case n=2 in a more tractable way than [26]. In Chapter 3 we describe this procedure in extensive detail in all dimensions.

In [96] Kellendonk and Richard observe that for scattering with a point interaction in dimensions n = 1, 2, 3, the wave operators are of a particular form which is amenable to index theoretic calculations. Kellendonk, Richard and Tiedra de Aldecoa [97, 98, 141] demonstrate that in dimension n = 1, 3 the wave operator is given by

$$W_{-} = \operatorname{Id} + \varphi(D_n)(S - \operatorname{Id}) + K, \tag{1.1}$$

with K a compact operator. Here  $\varphi \in C_0(\mathbb{R}) \cap L^2(\mathbb{R})$  is a universal function (in the sense that it does not depend on the potential). For  $x \in \mathbb{R}$ ,  $\varphi$  is given by

$$\varphi(x) = \frac{1}{2} (1 + \tanh(\pi x) - i \cosh(\pi x)^{-1})$$

and  $D_n$  is the self-adjoint generator of the dilation group on  $L^2(\mathbb{R}^n)$ . In fact the generator  $D_n$  of the dilation group plays a central role in scattering theory: it is used to provide a geometric proof of the existence and completeness of the wave operators in [58, 60, 128] and in proofs of the continuity of the scattering matrix [54]. For the index pairing we prove in Chapter 5, the fact that  $D_n$  and  $\frac{1}{2} \ln (H_0)$  are canonically conjugate variables is essential.

Kellendonk and Richard further show that  $W_{-}$  lives in a  $C^*$ -algebra E generated by continuous functions of  $H_0$  and  $D_n$  with limits at their endpoints. By considering the quotient of E by the ideal J of functions which vanish at their endpoints, the index of  $W_{-}$  can be computed as

$$\operatorname{Index}(W_{-}) = \operatorname{Wind}(q(W_{-})), \tag{1.2}$$

where Wind $(q(W_{-}))$  is the winding number of the image of  $W_{-}$  in the quotient E/J. In dimension n=1 this winding number can be determined explicitly and one recovers the original Levinson's theorem [97]. In dimension n=3, the high energy behaviour of the scattering matrix prevents the map  $\text{Det}(S(\cdot))$  from defining a loop. As a consequence, Equation (1.2) is not immediately comparable to the work of Bollé, Osborn and Newton,

except that in the presence of a resonance an additional factor of  $\frac{1}{2}$  is present in agreement with the work of Newton.

In dimension n=2, Richard and Tiedra de Aldecoa show that a similar formula to Equation (1.1) holds with no resonances and in the presence of s-resonances in [142]. Again due to the high-energy behaviour of the scattering operator one cannot deduce immediately the statement of Levinson's theorem of Bollé et al in [26] from the index theoretic picture.

The original work in the present thesis begins at this point and is described in detail in the next section, where we outline the structure of the thesis and the main results.

## 1.2 Thesis structure and main results

This thesis is organised in the following way.

Chapter 2: The history of scattering theory is intimately linked with that of spectral and perturbation theory for linear operators and so the literature is extensive. Chapter 2 is thus a guided tour of selected topics from scattering theory necessary for understanding the rest of the thesis. In particular, this chapter contains no original results. We begin with an introduction to abstract scattering theory including the wave and scattering operators in Section 2.1 before specialising to perturbations of the Laplacian for the remainder of the thesis, beginning in Section 2.2. We also introduce the generator  $D_n$  of the dilation group on  $L^2(\mathbb{R}^n)$ , which defines the canonically conjugate variable to  $\frac{1}{2} \ln (H_0)$ .

After establishing the existence and completeness of the wave operators in Section 2.3, we discuss the limiting absorption principle in Section 2.4 and how it can be used to construct a stationary approach to scattering theory in terms of generalised eigenfunctions and Fourier transforms. Of particular interest is the scattering matrix, the stationary representation of the scattering operator. In Section 2.5 we introduce the time delay operator, the spectral shift function and their relationship to the well-known Levinson's theorem. In particular, we describe using heat kernel methods how the high-energy behaviour of the time delay operator and the spectral shift function give rise to Levinson's theorem.

Chapter 3: We demonstrate, using the methods of Jensen and Nenciu [89], a key technical tool in our later analysis of the wave operator. The results we present in this chapter are not new and can be found in various places throughout the literature which we cite as necessary. Decomposing the potential V into V = vUv with  $v(x) = |V(x)|^{\frac{1}{2}}$  and U = sign(V) we obtain an expansion of the inverse of the operator  $M(k) = U + vR_0(-k^2)v$  for small k (with  $R_0(-k^2) = (H_0 + k^2)^{-1}$ ). Such an expansion is highly sensitive to the fine structure of the spectrum of H near zero and is dependent on the dimension also. In dimension  $n \leq 4$  there are obstructions to the invertibility of M(k), which lead to a definition of zero energy resonances in each dimension. In dimension n = 2 the procedure

is most difficult and we find two kinds of resonances. For dimension  $n \geq 5$ , general bounds on the resolvent show that there are no obstructions to the invertibility of the operator M(k). The analysis is generally more complicated in even dimensions due to a logarithmic singularity of the integral kernel of  $R_0(-k^2)$  at k=0.

Chapter 4: We demonstrate under certain short-range decay assumptions on the potential V (see Assumption 4.3.1) that the wave operators  $W_{-}$  in dimension  $n \geq 2$  have the same universal form as Equation (1.1). The original results in this chapter have appeared in [5] as joint work with Adam Rennie. In Section 4.1 we begin by determining the behaviour at zero of the scattering matrix in all dimensions. These results are well-known in dimension  $n \leq 3$ , however for  $n \geq 4$  we have not found any statements in the literature. We find in Theorem 4.1.7 the new result that for all dimensions  $n \geq 4$  the scattering matrix satisfies  $S(0) = \operatorname{Id}$  always, with the case n = 2 having been established in [142, Theorem 1.1]. This contrasts the behaviour in dimension n = 1, 3 [29, 88], where the behaviour of the scattering matrix at zero is known to be sensitive to the presence of resonances.

In Section 4.2 we use the integral representation of the wave operator of Theorem 2.4.29 in terms of generalised eigenfunctions in Lemma 2.4.31 to provide intuition as to why the wave operators have the form of Equation (1.1). Unfortunately, the intuitive methods are not sensitive to the fine structure of the spectrum near zero demonstrated in Chapter 3 and thus are unable to be used to demonstrate the compactness of the remainder term except in dimension n = 1, as was the method of proof in [97].

The remainder of the chapter is devoted to a different approach to obtaining the form of the wave operator in the spectral representation. The approach we present has been used to determine the form of the wave operator in dimension n=3 in [141]. In [140] and [142] in two dimensions in the case of no p-resonances the wave operator has been shown to be of a form similar to Equation (1.1) and we demonstrate in Section 4.3.2 the new result that Equation (1.1) holds in dimension n=2 also, in the absence of p-resonances. In Section 4.3.3 we use the resolvent expansions of Chapter 3 to prove the new result that Equation (1.1) holds in all dimensions  $n \geq 4$  also, with the exception of dimension n=4 in the presence of resonances, following the technique outlined in [141].

Chapter 5: We show how the form of the wave operator established in Chapter 4 can be used to prove that Levinson's theorem is the result of an index pairing between the class  $[D_+] \in K^1(C_0(\mathbb{R}^+) \otimes \mathcal{K}(L^2(\mathbb{S}^{n-1})))$  of the dilation operator and the class  $[S] \in K_1(C_0(\mathbb{R}^+) \otimes \mathcal{K}(L^2(\mathbb{S}^{n-1})))$  of the unitary scattering operator. The index pairing in the generic case S(0) = Id has appeared in [5] as joint work with Adam Rennie. That Levinson's theorem is a topological result has been established in dimension n = 1 [97], n = 2 [140, 142] in the absence of p-resonances and n = 3 [98, 141].

The technique used in these references is to use the form of the wave operator to show that  $W_{-}$  is an element of a particular  $C^*$ -algebra generated by functions of the free

Hamiltonian  $H_0$  and the generator of dilations  $D_n$  and to compute the index of the wave operator in the Calkin algebra obtained by taking the quotient by the functions which vanish at the endpoints of the spectrum of  $H_0$  and  $D_n$ . In Section 5.1 we present these constructions in detail, as they play a key role later in Chapter 5.

In Section 5.2 we construct a family of Fredholm operators  $W_U$  indexed by a class of unitary operators U on  $\mathcal{B}(\mathcal{H})$  which vanishes at 0 and  $\infty$  in the spectral representation. The mapping  $U \mapsto W_U$  is a new construction. In Theorem 5.3.2 we show that we can write the pairing

$$\langle [D_+], [U] \rangle = -\text{Index}(W_U). \tag{1.3}$$

In Section 5.3 we show that if S(0) = Id then the pairing of Equation (1.3) holds for  $W_{-} = W_{S}$ , a result which has appeared in [5] as joint work with Adam Rennie.

In dimension n = 1 generically and in the presence of resonances in dimension n = 3 the condition S(0) = Id is not satisfied and we require additional effort to correct this behaviour. In Section 5.3.1 we construct a unitary which corrects the low-energy behaviour of the scattering matrix and as a result obtain in Theorem 5.3.16 a new proof of the well-known Levinson's theorem in one dimension.

**Theorem 1.2.1.** Let n = 1 and S be the scattering operator for  $H = H_0 + V$ . Then the number of bound states of H is given by

$$N = \begin{cases} -\frac{1}{2\pi i} \int_0^\infty \text{Tr}(S(\lambda)^* S'(\lambda)) \, d\lambda, & \text{if there exists a resonance,} \\ -\frac{1}{2\pi i} \int_0^\infty \text{Tr}(S(\lambda)^* S'(\lambda)) \, d\lambda + \frac{1}{2} & \text{otherwise.} \end{cases}$$
(1.4)

In Section 5.3.2 we demonstrate in dimension n=3 that in the presence of resonances we can correct the value of S(0) in the same manner to obtain an index pairing in Theorem 5.3.20. We find also that to obtain a formula for  $\operatorname{Index}(W_{-})$  requires a careful analysis of large  $\lambda$  behaviour of  $S(\lambda)$  in trace norm, which leads again to a new proof of Levinson's theorem in dimension n=3 in Theorem 5.3.24. The statement is as follows.

**Theorem 1.2.2.** Let n = 3 and let S be the scattering operator for  $H = H_0 + V$ . Then the number of bound states of H is given by

$$-N = \frac{1}{2\pi i} \int_0^\infty \left( \text{Tr}(S(\lambda)^* S'(\lambda)) + \frac{i}{4\pi \lambda^{\frac{1}{2}}} \int_{\mathbb{R}^3} V(x) \, \mathrm{d}x \right) \mathrm{d}\lambda + \frac{1}{2} M_R(0), \tag{1.5}$$

where  $M_R(0) = 1$  if there is a zero energy-resonance and zero otherwise.

In Section 5.3.3 we show that as in dimension n=3 the high energy behaviour of  $S(\cdot)$  in trace norm requires some correcting in dimension n=2. After doing so we obtain, in the absence of p-resonances, Theorem 5.3.29, a new proof of Levinson's theorem in dimension n=2 which confirms [26, Theorem 6.3]. The result is as follows.

**Theorem 1.2.3.** Let n = 2 and let S be the scattering operator for  $H = H_0 + V$ . Suppose further that there are no p-resonances. Then the number of bound states of H is given by

$$-N = \frac{1}{2\pi i} \int_{\mathbb{R}^+} \text{Tr}(S(\lambda)^* S'(\lambda)) \, d\lambda + \frac{1}{4\pi} \int_{\mathbb{R}^2} V(x) \, dx.$$

We conclude Chapter 5 in Section 5.3.4 by applying similar techniques to correct the high energy behaviour of the scattering matrix and obtain in Theorem 5.3.34 a new proof of Levinson's theorem in higher dimensions.

Chapter 6: We apply the spectral flow formula of [47, Theorem 9] to the path of operators  $H_t(\alpha) = H_0 + tV + \alpha$ ,  $t \in [0,1]$ , where  $\alpha > 0$  is chosen so that  $H_t(\alpha)$  defines a path of Fredholm operators. The stationary scattering theory for the pair  $(H_0(\alpha), H_1(\alpha))$  is described in Section 6.2. In Section 6.3 we describe the necessary pseudodifferential operator calculus in order to estimate remainder terms in our spectral flow formula. In Section 6.4 we apply the Birman-Kreın trace formula and the pseudodifferential operator calculus of [46] to derive a formula for the spectral flow of the path  $H_t(\alpha)$  in Theorem 6.4.22 for n odd and Theorem 6.4.23 for n even. We give a new approach to the relationship between the spectral shift function and spectral flow, extending work of Azamov, Carey, Dodds and Sukochev [13, 14].

The even dimensional case is more complicated than the odd dimensional case due to the non-vanishing of certain remainder terms. We conclude Chapter 6 in Section 6.5 where we show how the spectral flow gives rise to Levinson's theorem in all dimensions. Our proof shows that with knowledge of the spectral shift function at zero, Theorem 1.2.3 is applicable to the case of p-resonances also. This is work in progress.

**Appendix A**: In this appendix we provide some computations of the high-energy behaviour of the spectral shift function appearing in Levinson's theorem using the spectral flow methods of Chapter 6.

## Chapter 2

# Scattering theory for Schrödinger operators

This chapter contains a brief introduction to some concepts from mathematical scattering theory which are necessary for the understanding of the rest of the thesis, in particular scattering theory for perturbations of the Laplace operator on Euclidean space. Proofs of most results have been included, since the tools and techniques used are often illuminating for later use. The results in this chapter are not new and we do not claim that this chapter is in any way a complete description of scattering theory, even for the Schrödinger operators we restrict ourselves to. The history of scattering theory is intertwined with the history of spectral theory, and so the literature is vast. For a more complete discussion of mathematical scattering theory we refer to [18, 137, 165, 164] and to the extensive references compiled within. Some more physically motivated discussion can be found in [7, 127, 146].

We begin with the abstract theory of scattering, before specialising to Schrödinger operators acting on  $L^2(\mathbb{R}^n)$  for the remainder of the thesis.

## 2.1 Abstract scattering theory

The abstract scattering theory presented here has its origins in the works of Møller [117] and Friedrichs [66, 67], who constructed wave operators explicitly by studying particular integral equations. The abstract scattering theory was first rigorously developed in the works of Kato [92], Cook [53] and Jauch [80, 81]. In this section we follow the presentation of Reed and Simon [137].

Consider two self-adjoint operators A and B on a Hilbert space  $\mathcal{H}$ . By Stone's theorem [134, Theorem VII.7] we have strongly-continuous one parameter unitary groups  $(e^{-itA})_{t\in\mathbb{R}}$  and  $(e^{-itB})_{t\in\mathbb{R}}$  of operators on  $\mathcal{H}$ , which we think of as an interacting dynamics A and a free or comparison dynamics B. For a vector  $f \in \mathcal{H}$  we say that  $e^{-itA}$  is asymptotically

free if there exists  $f_{\pm} \in \mathcal{H}$  (here the subscript ' $\pm$ ' refers to two independent vectors, one for '+' and one for '-') such that

$$0 = \lim_{t \to \pm \infty} \left| \left| e^{-itB} f_{\pm} - e^{-itA} f \right| \right| = \lim_{t \to \pm \infty} \left| \left| e^{itA} e^{-itB} f_{\pm} - f \right| \right|. \tag{2.1}$$

The motivation here is that in the limit as  $t \to \pm \infty$  (long before or long after an interaction has occurred) one expects for physical reasons that for any  $f \in \mathcal{H}$  the interacting evolution  $e^{-itA}f$  can be approximated by a freely evolving state  $e^{-itB}f_{\pm}$  for some  $f_{\pm} \in \mathcal{H}$ . In most applications (in particular the Schrödinger operators we specialise to) the operator B has no discrete spectrum. In cases where B does have discrete spectrum, we need to choose  $f_{\pm} \in \mathcal{H}_{ac}(B)$  (the absolutely continuous subspace for B). To see this, we suppose that  $f_{\pm}$  is an eigenvector of B and note that the strong limit in Equation (2.1) would exist only if  $f_{\pm}$  is an eigenvector for A with the same eigenvalue. Taking these considerations into account leads to the following definition.

**Definition 2.1.1.** Let A and B be self-adjoint operators on the Hilbert space  $\mathcal{H}$  and let  $P_{ac}(B)$  be the projection onto  $\mathcal{H}_{ac}(B)$ , the absolutely continuous subspace of  $\mathcal{H}$  with respect to B. We define the wave operators  $W_{\pm}(A, B)$  by the strong limits

$$W_{\pm}(A, B) := \underset{t \to \pm \infty}{\text{s-lim}} e^{itA} e^{-itB} P_{ac}(B),$$

if they exist. When  $W_{\pm}(A, B)$  exist we define  $\mathcal{H}_{\pm} = \text{Range}(W_{\pm}(A, B))$ .

Remark 2.1.2. The strong limit is the right one to take here [137, p. 17]. On the one hand the norm limit exists if and only if A = B, whilst on the other hand if  $\sigma(A)$  is purely discrete, the weak limit exists (and is 0) even though the operators A and B are not necessarily similar in any way.

**Proposition 2.1.3.** Let A and B be self-adjoint operators in a Hilbert space  $\mathcal{H}$  and suppose that  $W_{\pm}(A, B)$  exist. Then:

- 1.  $W_{\pm}(A, B)$  are partial isometries with initial subspace  $\mathcal{H}_{ac}(B)$  and final subspace  $\mathcal{H}_{\pm}$ ;
- 2. the subspaces  $\mathcal{H}_{\pm}$  satisfy  $A\mathcal{H}_{\pm} \subseteq \mathcal{H}_{\pm}$  (that is they are invariant for A), there is an inclusion  $W_{\pm}(A,B)\mathrm{Dom}(B) \subset \mathrm{Dom}(A)$  and we have the intertwining relation  $AW_{\pm}(A,B) = W_{\pm}(A,B)B$ ; and
- 3.  $\mathcal{H}_{+} \subseteq \mathcal{H}_{ac}(A)$ .

*Proof.* For  $f \in \mathcal{H}_{ac}(B)^{\perp}$  we have  $W_{\pm}f = 0$  by definition. For  $f \in \mathcal{H}_{ac}(B)$  we find that

$$\left|\left|e^{itA}e^{-itB}P_{ac}(B)f\right|\right| = \left|\left|f\right|\right|$$

for all  $t \in \mathbb{R}$  and thus if the wave operators exist we find  $||W_{\pm}(A, B)f|| = ||f||$  and so  $W_{\pm}(A, B)$  are partial isometries.

For the second claim, we note that for fixed  $s \in \mathbb{R}$  we have  $[e^{-isB}, P_{ac}(B)] = 0$ . So we compute that

$$W_{\pm}(A, B) = \underset{t \to +\infty}{\text{s-lim}} e^{itA} e^{-itB} P_{ac}(B) = \underset{t \to +\infty}{\text{s-lim}} e^{i(s+t)A} e^{-i(s+t)B} P_{ac}(B) = e^{isA} W_{\pm}(A, B) e^{-isB},$$

or

$$e^{-isA}W_{\pm}(A,B) = W_{\pm}(A,B)e^{-isB}.$$
 (2.2)

Then for  $f \in Dom(B)$  we have

$$\lim_{s \to 0} W_{\pm}(A, B) \left( \frac{e^{-isB} - Id}{s} \right) f = W_{\pm}(A, B) \left[ \lim_{s \to 0} \left( \frac{e^{-isB} - Id}{s} \right) f \right]$$
$$= -iW_{\pm}(A, B)Bf.$$

Applying Equation (2.2) we find

$$\left(\frac{e^{-isA} - \operatorname{Id}}{s}\right) W_{\pm}(A, B) f = W_{\pm}(A, B) \left(\frac{e^{-isB} - \operatorname{Id}}{s}\right) f.$$

As the limit on the right hand side exists, so does the limit on the left and they are equal. So  $W_{\pm}(A, B)f \in \text{Dom}(A)$  for all  $f \in \text{Dom}(B)$  and thus

$$-iAW_{\pm}(A,B)f = \lim_{s \to 0} \left(\frac{e^{-isB} - \mathrm{Id}}{s}\right) W_{\pm}(A,B)f = \lim_{s \to 0} W_{\pm}(A,B) \left(\frac{e^{-isB} - \mathrm{Id}}{s}\right) f$$
$$= -iW_{\pm}(A,B)B,$$

which proves the intertwining relation and that  $A\mathcal{H}_{\pm} \subseteq \mathcal{H}_{\pm}$ . By Equation (2.2) we have that  $\mathcal{H}_{\pm}$  are invariant under  $e^{-isA}$  for all  $s \in \mathbb{R}$ , which is enough to prove the claim.

For the third claim, we note that  $A|_{\mathcal{H}_{\pm}}$  is a self-adjoint operator on the Hilbert space  $\mathcal{H}_{\pm}$  ( $\mathcal{H}_{\pm}$  are closed since  $W_{\pm}$  are partial isometries). Furthermore, we have that  $W_{\pm}(A,B):\mathcal{H}_{ac}(B)\to\mathcal{H}_{\pm}$  is a unitary and the intertwining relation shows us that  $B|_{\mathcal{H}_{ac}(B)}$  is unitarily equivalent to  $A|_{\mathcal{H}_{\pm}}$  with the unitary equivalence given by  $W_{\pm}(A,B)$ . Hence  $A|_{\mathcal{H}_{\pm}}$  is purely absolutely continuous and  $\mathcal{H}_{\pm}\subseteq\mathcal{H}_{ac}(A)$ , which completes the proof.  $\Box$ 

The following result shows that the wave operators have a partially defined composition rule, often referred to as the chain rule, due to Kato [92].

**Proposition 2.1.4.** Let A, B and C be self-adjoint operators on a Hilbert space  $\mathcal{H}$ . If

 $W_{\pm}(A,B)$  and  $W_{\pm}(B,C)$  exist, then  $W_{\pm}(A,C)$  exist and

$$W_{\pm}(A,C) = W_{\pm}(A,B)W_{\pm}(B,C).$$

*Proof.* By part 3 of Proposition 2.1.3 we have Range $(W_{\pm}(B,C)) \subseteq \mathcal{H}_{ac}(B)$  and thus

$$\lim_{t \to +\infty} \left| \left| (\operatorname{Id} - P_{ac}(B)) e^{itB} e^{-itC} P_{ac}(C) f \right| \right| = 0$$

for any  $f \in \mathcal{H}$ . Hence we may write

$$e^{itA}e^{-itC}P_{ac}(C)f = e^{itA}e^{-itB}P_{ac}(B)e^{itB}e^{-itC}P_{ac}(C)f$$
$$+ e^{itA}e^{-itB}(\operatorname{Id} - P_{ac}(B))e^{itB}e^{-itC}P_{ac}(C),$$

which converges as  $t \to \pm \infty$  to  $W_{\pm}(A, B)W_{\pm}(B, C)f$ , since a product of strongly convergent families of uniformly bounded operators converges strongly (a consequence of the uniform boundedness principle).

**Definition 2.1.5.** Let A and B be self-adjoint operators on a Hilbert space  $\mathcal{H}$  such that the wave operators  $W_{\pm}(A, B)$  exist. Let  $\mathcal{H}_p(A)$  denote the point spectrum subspace of  $\mathcal{H}$  associated to A. We say that the wave operators are asymptotically complete if  $\mathcal{H}_- = \mathcal{H}_+ = (\mathcal{H}_p(A))^{\perp}$ . We say that the wave operators are complete if  $\mathcal{H}_- = \mathcal{H}_+ = \mathcal{H}_{ac}(A)$ .

Remark 2.1.6. Asymptotic completeness is a stronger statement than completeness. In fact, asymptotic completeness is equivalent to the pair of statements  $W_{\pm}(A, B)$  are complete and  $\sigma_{sc}(A) = \emptyset$  [149, Theorem IV.2] (here  $\sigma_{sc}(A)$  denotes the singular continuous spectrum of A).

The completeness of  $W_{\pm}(A, B)$  can actually be reduced to an existence problem.

**Proposition 2.1.7.** Let A and B be self-adjoint operators on the Hilbert space  $\mathcal{H}$  and suppose the wave operators  $W_{\pm}(A, B)$  exist. Then  $W_{\pm}(A, B)$  are complete if and only if  $W_{\pm}(B, A)$  exist.

*Proof.* First suppose that both  $W_{\pm}(A, B)$  and  $W_{\pm}(B, A)$  exist. Then by the chain rule for wave operators (Proposition 2.1.4) we have

$$P_{ac}(A) = W_{\pm}(A, A) = W_{\pm}(A, B)W_{\pm}(B, A)$$

and thus  $\mathcal{H}_{ac}(A) = \text{Range}(W_{\pm}(A, A)) \subseteq \text{Range}(W_{\pm}(A, B))$ . By Proposition 2.1.3 we have  $\text{Range}(W_{\pm}(A, B)) \subset \mathcal{H}_{ac}(A)$  and thus we have equality.

Now suppose that  $W_{\pm}(A, B)$  exist and are complete. Let  $f \in \mathcal{H}_{ac}(A) = \operatorname{Range}(P_{ac}(A))$ . As  $W_{\pm}(A, B)$  are complete there exist  $f_{\pm} \in \mathcal{H}$  such that  $f = W_{\pm}(A, B)f_{\pm}$ . Hence we have

$$\lim_{t \to +\infty} \left| \left| f - e^{itA} e^{-itB} P_{ac}(B) f_{\pm} \right| \right| = 0.$$

Using unitarity we thus obtain

$$\lim_{t \to \pm \infty} \left| \left| e^{itB} e^{-itA} f - P_{ac}(B) f_{\pm} \right| \right| = 0.$$

So 
$$\lim_{t\to\pm\infty} e^{itB} e^{-itA} f = P_{ac}(B) f_{\pm}.$$

Proposition 2.1.7 seems to imply that the problem of completeness is no harder than existence, however this is not the case. The following common existence result, known as Cook's criterion [53], uses the observation that the operator B is often much simpler in some way than the operator A. Since this is not symmetric in A and B, completeness is often a much more difficult question.

**Theorem 2.1.8** (Cook's criterion). Let A and B be self-adjoint operators on the Hilbert space  $\mathcal{H}$  with Dom(A) = Dom(B). Suppose there is a set  $\tilde{D} \subset Dom(B) \cap \mathcal{H}_{ac}(B)$  which is dense in  $\mathcal{H}_{ac}(B)$  such that for any  $f \in \tilde{D}$  there is a  $T_0 \in \mathbb{R}$  satisfying

$$\int_{T_0}^{\infty} \left| \left| (B - A) e^{\pm itB} f \right| \right| dt < \infty.$$
 (2.3)

Then the wave operators  $W_{\pm}(A, B)$  exist.

*Proof.* Fix  $f \in \tilde{D}$  and let  $g(t) = e^{itA}e^{-itB}f$ . Then g is differentiable and

$$g'(t) = -ie^{itA}(B - A)e^{-itB}f.$$

By assumption there exists  $T_0 > 0$  such that the inequality (2.3) holds. Then for  $T_0 < s < t$  we can compute

$$||g(t) - g(s)|| \le \int_s^t ||g'(u)|| du = \int_s^t ||(B - A)e^{-iuB}f|| du.$$

For fixed  $\psi \in \mathcal{H}$  the function  $h : \mathbb{R} \to \mathbb{C}$  defined by  $h(t) = \langle g(t), \psi \rangle$  is differentiable with

$$d_{\psi}(t) := h'(t) = \langle (B - A)e^{-itB}f, ie^{-itA}\psi \rangle,$$

which is seen to be continuous in t. The fundamental theorem of calculus then tells us that

$$\langle (g(t) - g(s))f, \psi \rangle = \int_{s}^{t} \langle (B - A)e^{-iuB}f, ie^{-iuA}\psi \rangle du.$$

Since  $Dom(A) \subset \mathcal{H}$  is dense we find

$$\begin{aligned} ||(g(t) - g(s))f|| &= \sup_{\|\psi\| \le 1, \psi \in \text{Dom}(A)} |\langle (g(t) - g(s))f, \psi \rangle| \\ &\le \sup_{\|\psi\| \le 1, \psi \in \text{Dom}(A)} \int_s^t |\langle (B - A)e^{-iuB}f, ie^{-iuA}\psi \rangle| \, du \\ &\le \int_s^t ||(B - A)e^{-iuB}f|| \, du. \end{aligned}$$

So we see that

$$\lim_{s \to \infty} ||(g(t) - g(s))f|| = 0$$

and thus for all  $f \in \tilde{D}$  we have that the limits

$$\lim_{t \to \infty} e^{itA} e^{-itB} P_{ac}(B) f = \lim_{t \to \infty} g(t)$$
(2.4)

exist. If  $f \in \mathcal{H}_{ac}(B)^{\perp} = \tilde{D}^{\perp}$  then the limit (2.4) exists and is zero. By the density of  $\tilde{D} \oplus \tilde{D}^{\perp}$  we have that the wave operator  $W_{+}(A,B)$  exists.

We can use the wave operators to construct a unitary operator which maps incoming states to outgoing states.

**Definition 2.1.9.** Let A and B be self-adjoint operators such that the wave operators  $W_{\pm}(A,B)$  exist. Then we define the *scattering operator* to be

$$S = S(A, B) = (W_{+}(A, B))^*W_{-}(A, B).$$

**Lemma 2.1.10.** Let A and B be self-adjoint operators such that the wave operators  $W_{\pm}(A,B)$  exist. Then the scattering operator S commutes with A. If in addition the wave operators  $W_{\pm}(A,B)$  are complete then the scattering operator is unitary.

*Proof.* The relation SA = AS follows immediately from the intertwining relation for the wave operators. To see that S is unitary, we check that for all  $f, g \in \mathcal{H}$  we have

$$\langle Sf, Sg \rangle = \langle (W_+(A,B))^*W_-(A,B)f, (W_+(A,B))^*W_-(A,B)g \rangle = \langle f,g \rangle,$$

since by completeness we have  $Range(W_+) = Range(W_-)$ .

#### 2.1.1 Stationary scattering theory

The descriptions of the wave operator in Section 2.1 have been time-dependent in the sense that we directly use  $e^{\pm itA}e^{\mp itB}$  to describe the wave operators  $W_{\pm}(A, B)$ . It is sometimes

more useful to consider a time-independent (stationary) approach to scattering theory. In this section we give an alternative stationary description of the wave operators. We will make a refinement of this description specific to Schrödinger operators in Section 2.4.2.

**Theorem 2.1.11.** Let A and B be self-adjoint operators on a Hilbert space  $\mathcal{H}$  with resolvents, defined for  $z \in \rho(A) \cap \rho(B)$ , given by  $R_A(z) = (A-z)^{-1}$  and  $R_B(z) = (B-z)^{-1}$ . Then for any  $f \in \text{Dom}(B)$  and  $g \in \text{Dom}(A)$  the wave operators  $W_{\pm}(A, B)$  satisfy

$$\langle W_{\pm}(A,B)f,g\rangle = \lim_{\varepsilon \to 0^+} \frac{\varepsilon}{\pi} \int_{\mathbb{R}} \langle R_B(\lambda \mp i\varepsilon)f, R_A(\lambda \mp i\varepsilon)g\rangle \,d\lambda. \tag{2.5}$$

*Proof.* We prove the relation for  $W_+$ , the proof for  $W_-$  being similar. For Im(z) > 0 we have the identity

$$-iz^{-1} = \int_0^\infty e^{izt} dt.$$

For  $f \in \mathcal{H}$ ,  $\varepsilon > 0$ ,  $\lambda \in \mathbb{R}$  and  $z = \lambda + i\varepsilon$  we define  $h_{\varepsilon,A}(t) = e^{-i(A-i\varepsilon)t}f$ . Then we find

$$R_A(z)f = -i\int_0^\infty e^{-i(A-z)t} f dt = -i\int_0^\infty e^{i\lambda t} e^{-i(A-i\varepsilon)t} f dt = -i(2\pi)^{\frac{1}{2}} [\mathcal{F}^*(\chi_{[0,\infty)} h_{\varepsilon,A})](\lambda),$$

where we are using the notation  $\chi_I$  for the characteristic function of the set  $I \subset \mathbb{R}$  and  $\mathcal{F}$  for the one dimensional Fourier transform, which we describe in greater detail in Section 2.2.1. Similarly, we can define for  $g \in \mathcal{H}$  the function  $h_{\varepsilon,B}(t) = e^{-i(B-i\varepsilon)t}g$  and check that

$$R_B(z)g = -i(2\pi)^{\frac{1}{2}} [\mathcal{F}^*(\chi_{[0,\infty)} h_{\varepsilon,B})](\lambda).$$

We can now use the Parseval identity to obtain

$$\int_{\mathbb{R}} \langle R_B(\lambda + i\varepsilon) f, R_A(\lambda + i\varepsilon) g \rangle \, d\lambda = (2\pi) \int_{\mathbb{R}} \langle [\mathcal{F}^*(\chi_{[0,\infty)} h_{\varepsilon,B})](\lambda), [\mathcal{F}^*(\chi_{[0,\infty)} h_{\varepsilon,A})](\lambda) \rangle \, d\lambda$$

$$= (2\pi) \int_0^\infty \langle \chi_{[0,\infty)}(t) h_{\varepsilon,B}(t), \chi_{[0,\infty)}(t) h_{\varepsilon,A}(t) \rangle \, dt$$

$$= (2\pi) \int_0^\infty e^{-2\varepsilon t} \langle e^{-itB} g, e^{-itA} f \rangle \, dt$$

$$= (2\pi) \int_0^\infty e^{-2\varepsilon t} \langle e^{itA} e^{-itB} g, f \rangle \, dt. \tag{2.6}$$

So if  $g \in Dom(W_+)$  we have

$$\lim_{\varepsilon \to 0^+} 2\varepsilon \int_0^\infty e^{-2\varepsilon t} \langle e^{itA} e^{-itB} g, f \rangle dt = \langle W_+ g, f \rangle.$$
 (2.7)

Thus, by combining Equations (2.6) and (2.7) we obtain

$$\langle W_{+}g, f \rangle = \lim_{\varepsilon \to 0^{+}} 2\varepsilon \int_{0}^{\infty} e^{-2\varepsilon t} \langle e^{itA} e^{-itB} g, f \rangle dt$$
$$= \lim_{\varepsilon \to 0^{+}} \frac{\varepsilon}{\pi} \int_{\mathbb{R}} \langle R_{B}(\lambda + i\varepsilon) f, R_{A}(\lambda + i\varepsilon) g \rangle d\lambda,$$

which is the statement of the theorem.

The following statement is known as Stone's formula (see [134, Theorem VII.13] for example) and characterises how the resolvent of a self-adjoint operator can be used to describe its spectral projections.

**Lemma 2.1.12.** Let A be a self-adjoint operator on a Hilbert space  $\mathcal{H}$  and for  $z \in \rho(A)$  define  $R_A(z) = (A-z)^{-1}$ . Then for any  $f, g \in \mathcal{H}$  and any open interval  $I \subseteq \mathbb{R}$  we have

$$\lim_{\varepsilon \to 0} \frac{\varepsilon}{\pi} \int_{I} \langle R_A(\lambda \pm i\varepsilon) f, R_A(\lambda \pm i\varepsilon) g \rangle \, d\lambda = \langle \chi_I(A) f, g \rangle.$$
 (2.8)

*Proof.* Let  $f, g \in \mathcal{H}$ . Functional calculus shows that

$$\frac{\varepsilon}{\pi} \int_{I} \langle R_A(\lambda \pm i\varepsilon) f, R_A(\lambda \pm i\varepsilon) g \rangle \, d\lambda = \int_{I} \left( \int_{\mathbb{R}} \frac{\varepsilon}{\pi(\varepsilon^2 + (\lambda - \rho)^2)} \, d\langle P_\lambda f, g \rangle \right) \, d\rho.$$

Next we define the function

$$f_{\varepsilon}(\lambda) = \int_{I} \frac{\varepsilon}{\pi(\varepsilon^2 + (\lambda - \rho)^2)} d\rho.$$

Noting that  $|f_{\varepsilon}(\lambda)| \leq 1$  for all  $\lambda$  we may apply Fubini's theorem to interchange the order of integration and obtain the relation

$$\frac{\varepsilon}{\pi} \int_{I} \langle R_A(\lambda \pm i\varepsilon) f, R_A(\lambda \pm i\varepsilon) g \rangle \, \mathrm{d}\lambda = \int_{\mathbb{R}} f_{\varepsilon}(\lambda) \, \mathrm{d}\langle P_{\lambda} f, g \rangle.$$

We can compute the integral defining  $f_{\varepsilon}$  explicitly to obtain

$$\lim_{\varepsilon \to 0} f_{\varepsilon}(\lambda) = \begin{cases} 1, & \text{if } \lambda \in I, \\ \frac{1}{2} & \text{if } \lambda \in \partial \overline{I}, \\ 0, & \text{otherwise.} \end{cases}$$

The result now follows immediately.

Corollary 2.1.13. Let A be a self-adjoint operator on a Hilbert space and for  $z \in \rho(A)$ 

define  $R_A(z) = (A-z)^{-1}$ . Then for any  $f, g \in \mathcal{H}$  we have

$$\lim_{\varepsilon \to 0} \frac{\varepsilon}{\pi} \int_{\mathbb{R}} \langle R_A(\lambda \pm i\varepsilon) f, R_A(\lambda \pm i\varepsilon) g \rangle \, d\lambda = \langle f, g \rangle.$$
 (2.9)

*Proof.* Take  $I = \mathbb{R}$  in Lemma 2.1.12

#### 2.2 Hamiltonians and dilation

We are interested in a particular choice of operators A and B in the previous section. We work on the Hilbert space  $\mathcal{H} = L^2(\mathbb{R}^n)$  for some  $n \geq 1$ . We will often have need of the Hilbert space

$$\mathcal{P} = L^2(\mathbb{S}^{n-1}),$$

where  $\mathbb{S}^{n-1} = \{x \in \mathbb{R}^n : |x| = 1\}$  and the space  $L^2(\mathbb{R}^+)$  where  $\mathbb{R}^+ = [0, \infty)$  is the non-negative half-line. We will often make the identification

$$\mathcal{H}_{spec} := L^2(\mathbb{R}^+, \mathcal{P}) = L^2(\mathbb{R}^+) \otimes \mathcal{P}.$$

We will also frequently use the Schwarz space  $\mathcal{S}(\mathbb{R}^n)$  defined by

$$\mathcal{S}(\mathbb{R}^n) = \left\{ f \in C^{\infty}(\mathbb{R}^n) : \sup_{x \in \mathbb{R}^n} |x^{\alpha}[D_{\beta}f](x)| < \infty \text{ for all multi-indices } \alpha, \beta \right\}.$$

Since  $C_c^{\infty}(\mathbb{R}^n) \subset \mathcal{S}(\mathbb{R}^n)$  we have that  $\mathcal{S}(\mathbb{R}^n)$  is dense in  $L^p(\mathbb{R}^n)$  for  $p \in [1, \infty)$ .

#### 2.2.1 The free Hamiltonian

Our reference operator of interest is the Laplacian

$$\Delta = -\sum_{j=0}^{n} \frac{\partial^2}{\partial x_j^2}$$

acting initially on  $C_c^{\infty}(\mathbb{R}^n) \subset \mathcal{H}$ . The Laplacian and its spectral theory have been studied extensively in the literature, for a detailed discussion we refer the reader to [154, 149, 135, 55, 95]. To describe the spectral properties of the Laplacian, we recall the Fourier transform  $\mathcal{F}: \mathcal{H} \to \mathcal{H}$  defined for  $f \in \mathcal{S}(\mathbb{R}^n)$  by

$$[\mathcal{F}f](\xi) = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} e^{-i\langle x,\xi\rangle} f(x) \,\mathrm{d}x$$
 (2.10)

and extended by continuity to all of  $\mathcal{H}$ . We recall that the Fourier transform is a unitary operator with inverse defined by

$$[\mathcal{F}^* f](x) = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} e^{i\langle x,\xi\rangle} f(\xi) \,\mathrm{d}\xi. \tag{2.11}$$

The Fourier transform is an isomorphism from  $\mathcal{S}(\mathbb{R}^n)$  to itself (in fact a homeomorphism when equipped with the Schwarz topology, see [135, Theorem IX.1]). Using the Fourier transform we are able to characterise the maximal domain of definition of  $\Delta$  and its spectrum. The following result is well-known (see for example [135, Theorem IX.27]).

**Theorem 2.2.1.** The Laplacian  $\Delta$  is essentially self-adjoint on  $C_c^{\infty}(\mathbb{R}^n)$  and thus has a unique self-adjoint extension with domain  $Dom(\Delta) = \{ f \in \mathcal{H} : (\xi \mapsto |\xi|^2 [\mathcal{F}f](\xi)) \in \mathcal{H} \}.$ 

**Definition 2.2.2.** We denote by  $H_0$  the unique self-adjoint extension of  $\Delta$  defined by Theorem 2.2.1 and sometimes refer to  $H_0$  as the *free Hamiltonian*.

We know that the Fourier transform 'diagonalises' the operator  $H_0$  in the sense that

$$[\mathcal{F}H_0\mathcal{F}^*f](\xi) = |\xi|^2 f(x) \tag{2.12}$$

for  $f \in \text{Dom}(H_0)$ .

**Definition 2.2.3.** We define  $X^2 = \mathcal{F}H_0\mathcal{F}$  to be the multiplication operator, given for  $f \in \mathcal{S}(\mathbb{R}^n)$  and  $x \in \mathbb{R}^n$  by

$$[X^2 f](x) = |x|^2 f(x).$$

We can use the operator  $H_0$  to define a class of weighted Sobolev spaces which we will use throughout this thesis. For a more detailed discussion of Sobolev spaces we refer the reader to [2, 8].

**Definition 2.2.4.** For any  $s, t \in \mathbb{R}$  the weighted Sobolev space  $H^{s,t} = H^{s,t}(\mathbb{R}^n)$  is given by

$$H^{s,t} = \left\{ f \in \mathcal{S}'(\mathbb{R}^n) : ||f||_{H^{s,t}} = \left| \left| (1 + X^2)^{\frac{t}{2}} (1 + H_0)^{\frac{s}{2}} f \right| \right|_{\mathcal{H}} < \infty \right\}. \tag{2.13}$$

The main properties of the Sobolev spaces we rely on are summarised in the following (see [8, Chapter 4]).

Theorem 2.2.5. Fix  $s, t \in \mathbb{R}$ .

(a) The Fourier transform  $\mathcal{F}$  gives an isomorphism of  $H^{s,t}$  with  $H^{t,s}$  and the sesquilinear  $map \langle \cdot, \cdot \rangle : H^{s,t} \times H^{-s,-t} \to \mathbb{C}$  defined for  $f \in H^{s,t}$  and  $g \in H^{-s,-t}$  by

$$\langle f, g \rangle = \int_{\mathbb{R}^n} \overline{f(x)} g(x) \, \mathrm{d}x$$

gives a natural duality between  $H^{s,t}$  and  $H^{-s,-t}$ .

- (b) For any p < s and m < t the inclusion  $H^{s,t} \hookrightarrow H^{p,m}$  is compact.
- (c) For any m > 0 and  $q : \mathbb{R}^n \to \mathbb{C}$  satisfying  $|q(x)| \leq C(1 + |x|)^{-m}$  for almost all  $x \in \mathbb{R}^n$  the operator of multiplication by q maps  $H^{s,t}$  into  $H^{s,t+m}$ .
- (d) The domain of  $H_0$  is the Sobolev space  $H^{2,0}$ .

Theorem 2.2.5 (b) is often referred to as Rellich's compactness theorem (see for example [8, Theorem 4.1.5]), which we shall make frequent use of along with the mapping properties of multiplication operators in Theorem 2.2.5 (c).

The diagonalisation of Equation (2.12) allows us to determine directly the spectrum of  $H_0$ .

**Lemma 2.2.6.** The operator  $H_0$  has spectrum  $\sigma(H_0) = \sigma_{ac}(H_0) = [0, \infty)$ .

*Proof.* This follows from the unitarity of the Fourier transform and the spectrum of multiplication operators.  $\Box$ 

Equation (2.12) is not the 'correct' diagonalisation, since it is not multiplication by the spectral variable. To find the correct diagonalisation, we introduce a second unitary to rescale the spectral variable appropriately.

Recall that we have defined the spaces  $\mathcal{P} = L^2(\mathbb{S}^{n-1})$  and  $\mathcal{H}_{spec} = L^2(\mathbb{R}^+) \otimes \mathcal{P}$ .

**Definition 2.2.7.** For  $n \geq 2$ ,  $\lambda \in \mathbb{R}^+$  and  $\omega \in \mathbb{S}^{n-1}$  we define the operator  $\mathcal{U} : \mathcal{H} \to \mathcal{H}_{spec}$  by

$$[\mathcal{U}f](\lambda,\omega) = 2^{-\frac{1}{2}}\lambda^{\frac{n-2}{4}}f(\lambda^{\frac{1}{2}}\omega)$$

and  $F_0 = \mathcal{UF}$ . For n = 1,  $\mathcal{P} = \mathbb{C}^2$  and  $\lambda \in \mathbb{R}^+$  we define  $\mathcal{U} : \mathcal{H} \to L^2(\mathbb{R}^+) \otimes \mathcal{P}$  by

$$[\mathcal{U}f](\lambda) = 2^{-\frac{1}{2}} \lambda^{-\frac{1}{2}} \begin{pmatrix} f(\lambda^{\frac{1}{2}}) \\ f(-\lambda^{\frac{1}{2}}) \end{pmatrix}.$$

We define the operator  $F_0: \mathcal{H} \to L^2(\mathbb{R}^+) \otimes \mathcal{P}$  by  $F_0 = \mathcal{UF}$  for all dimensions  $n \geq 1$ .

The operators  $F_0$  and  $\mathcal{U}$  are well-known throughout the spectral theory literature: see [86] for a good discussion of their properties. In dimension n=1 the unitary  $\mathcal{U}$  is not unique, depending on a choice of basis for  $\mathbb{C}^2$ . The diagonalisation of  $H_0$  provided by  $F_0$  provides a more physical interpretation of the spectrum of  $H_0$  as the energy parameter and we will use  $F_0$  to construct a diagonalisation for a perturbed Hamiltonian operator in Section 2.4.2

**Lemma 2.2.8.** For  $n \geq 2$  the operators  $\mathcal{U}$  and  $F_0$  are unitary and for  $f \in F_0(\text{Dom}(H_0))$ ,  $\lambda \in \mathbb{R}^+$  and  $\omega \in \mathbb{S}^{n-1}$  we have

$$[F_0 H_0 F_0^* f](\lambda, \omega) = \lambda f(\lambda, \omega). \tag{2.14}$$

*Proof.* It suffices to check that  $\mathcal{U}$  is unitary, since the Fourier transform  $\mathcal{F}$  is unitary. For  $f, g \in L^2(\mathbb{R}^n)$  we can check directly that

$$\langle \mathcal{U}f, \mathcal{U}g \rangle = \int_{\mathbb{R}^+} \int_{\mathbb{S}^{n-1}} \overline{[\mathcal{U}f](\lambda, \omega)} [\mathcal{U}g](\lambda, \omega) \, d\omega \, d\lambda = \int_{\mathbb{R}^+} \int_{\mathbb{S}^{n-1}} 2^{-1} \lambda^{\frac{n-2}{2}} \overline{f(\lambda^{\frac{1}{2}}\omega)} g(\lambda^{\frac{1}{2}}\omega) \, d\omega \, d\lambda$$
$$= \int_{\mathbb{R}^+} \int_{\mathbb{S}^{n-1}} r^{n-1} \overline{f(r\omega)} g(r\omega) \, d\omega \, dr = \int_{\mathbb{R}^n} \overline{f(x)} g(x) \, dx = \langle f, g \rangle,$$

where we have made the substitutions  $\lambda = r^2$  and  $x = r\omega$ . Thus the operator  $\mathcal{U}$  is unitary as claimed. For the diagonalisation relation of Equation (2.14) it suffices to show that for  $f \in \text{Dom}(H_0)$  we have the relation

$$[F_0 H_0 f](\lambda, \omega) = \lambda [F_0 f](\lambda, \omega). \tag{2.15}$$

Define the operator  $M: \text{Dom}(H_0) \to \mathcal{H}$  by  $[Mf](\xi) = |\xi|^2 f(\xi)$ . Then by conjugating both sides of Equation (2.15) with  $\mathcal{F}$  and  $\mathcal{F}^*$  it suffices to check for  $f \in \text{Dom}(H_0)$  that

$$[\mathcal{U}M\mathcal{U}^*f](\lambda,\omega) = \lambda f(\lambda,\omega).$$

So we compute that

$$[\mathcal{U}M\mathcal{U}^*f](\lambda,\omega) = 2^{-\frac{1}{2}}\lambda^{\frac{n-2}{4}}[M\mathcal{U}^*f](\lambda^{\frac{1}{2}}\omega) = 2^{-\frac{1}{2}}\lambda^{\frac{n+2}{4}}[\mathcal{U}^*f](\lambda^{\frac{1}{2}}\omega)$$
$$= \lambda^{\frac{n+2}{4}}\lambda^{-\frac{n-2}{4}}f(|\lambda^{\frac{1}{2}}\omega|^2,\omega) = \lambda f(\lambda,\omega),$$

which completes the proof.

Lemma 2.2.8 shows that the operator  $F_0$  provides a direct integral decomposition of the free Hamiltonian  $H_0$ . For a detailed discussion of direct integrals we refer the reader to [23, Section 7.1] and [74, Section 7.3]

Even though we have stated that the Fourier transform is not the 'correct' diagonalisation of  $H_0$  for scattering purposes, there are still many applications of use to us. We now turn our attention to developing the functional calculus for  $H_0$  explicitly. Since the Fourier transform provides a diagonalisation of  $H_0$ , the spectral theorem tells us exactly how to do this. For a function  $f: \mathbb{R}^+ \to \mathbb{C}$  with rapidly decaying Fourier transform, we can define the operator  $f(H_0)$  acting on a fixed  $\psi \in \mathcal{S}(\mathbb{R}^n)$  by

$$[f(H_0)\psi](x) = [\mathcal{F}^*\mathcal{F}f(H_0)\psi](x) = [\mathcal{F}^*M_{\tilde{f}}\mathcal{F}\psi](x), \quad x \in \mathbb{R}^n,$$
 (2.16)

where  $\tilde{f}(\xi) = f(|\xi|^2)$ . By writing out the Fourier transforms explicitly, we can write the operator  $f(H_0)$  as an integral operator. For  $x \in \mathbb{R}^n$  we have

$$[f(H_0)\psi](x) = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} e^{i\langle x,\xi\rangle} f(|\xi|^2) [\mathcal{F}\psi](\xi) \,d\xi$$
$$= (2\pi)^{-n} \int_{\mathbb{R}^n} \int_{\mathbb{R}^n} e^{i\langle x,\xi\rangle} e^{-i\langle \xi,y\rangle} f(|\xi|^2) \psi(y) \,dy \,d\xi$$
$$= \int_{\mathbb{R}^n} K(x,y)\psi(y) \,dy,$$

where the integral kernel  $K: \mathbb{R}^n \times \mathbb{R}^n \to \mathbb{C}$  is defined by

$$K(x,y) = (2\pi)^{-n} \int_{\mathbb{R}^n} e^{-i\langle \xi, y - x \rangle} f(|\xi|^2) d\xi, \quad x, y \in \mathbb{R}^n.$$

Defining the function  $g: \mathbb{R}^n \to \mathbb{C}$  by  $g(\xi) = f(|\xi|^2)$  we have

$$K(x,y) = (2\pi)^{-\frac{n}{2}} [\mathcal{F}g](y-x). \tag{2.17}$$

Thus we see that to describe a function f of the free Hamiltonian  $H_0$  as an integral operator, we need f to be such that we can take the Fourier transform of  $g(\xi) = f(|\xi|^2)$ . Some preparatory work is required to compute such integrals.

**Lemma 2.2.9.** For any  $r \in (0, \infty)$  and  $0 \neq y \in \mathbb{R}^n$  we have

$$\int_{\mathbb{S}^{n-1}} e^{-ir\langle \omega, y \rangle} d\omega = (2\pi)^{\frac{n}{2}} (r|y|)^{-\frac{n-2}{2}} J_{\frac{n-2}{2}}(r|y|), \tag{2.18}$$

where  $J_{\nu}$  denotes the Bessel function of the first kind of order  $\nu$ .

*Proof.* First we define the constant  $C_n$  via

$$C_n = \frac{\operatorname{Vol}(\mathbb{S}^{n-1})}{\int_0^{\pi} \sin^{n-2}(\theta_1) d\theta_1}.$$

By using spherical coordinates, the volume of the n-sphere can be computed as

$$\operatorname{Vol}(\mathbb{S}^{n-1}) = \int_0^{\pi} \int_0^{\pi} \cdots \int_0^{\pi} \int_0^{2\pi} \sin^{n-2}(\theta_1) \sin^{n-3}(\theta_2) \cdots \sin(\theta_{n-2}) d\theta_{n-1} d\theta_{n-2} \cdots d\theta_2 d\theta_1$$
$$= \frac{2\pi^{\frac{n}{2}}}{\Gamma(\frac{n}{2})}.$$

This can be checked in many ways, for example by reducing each integral to a Beta function. We can evaluate the  $\theta_1$  integral, for example, via the substitution  $u = \sin(\theta_1)$ 

to obtain

$$\int_0^{\pi} \sin^{n-2}(\theta_1) d\theta_1 = 2 \int_0^{\frac{\pi}{2}} \sin^{n-2}(\theta_1) d\theta_1 = 2 \int_0^1 u^{n-2} (1 - u^2)^{-\frac{1}{2}} du$$
$$= \int_0^1 v^{\frac{n-3}{2}} (1 - v)^{-\frac{1}{2}} dv = \beta \left(\frac{n-1}{2}, \frac{1}{2}\right) = \frac{\Gamma\left(\frac{n-1}{2}\right) \Gamma\left(\frac{1}{2}\right)}{\Gamma\left(\frac{n}{2}\right)},$$

where  $\beta$  denotes the Beta function and  $\Gamma$  the Gamma function. Thus we can write

$$C_n = \frac{2\pi^{\frac{n-1}{2}}}{\Gamma\left(\frac{n-1}{2}\right)} = \operatorname{Vol}\left(\mathbb{S}^{n-2}\right).$$

We are now able to evaluate the integral in Equation (2.18). Choose spherical coordinates so that  $\theta_1$  is the angle between  $\omega$  and y. Thus  $\langle \omega, y \rangle = |y| \cos(\theta_1)$ . Then we find

$$\int_{\mathbb{S}^{n-1}} e^{-ir\langle \omega, y \rangle} d\omega = C_n \int_0^{\pi} e^{-ir|y|\cos(\theta_1)} \sin^{n-2}(\theta_1) d\theta_1 = C_n \int_{-1}^1 e^{-ir|y|u} (1 - u^2)^{\frac{n-3}{2}} du.$$

We now recall a Poisson integral from [61, Equation 7.12.7] for  $z \in \mathbb{C}$  and  $\text{Re}(\nu) > -1$ 

$$\int_{-1}^{1} e^{-izu} (1 - u^2)^{\nu - \frac{1}{2}} du = \Gamma \left(\nu + \frac{1}{2}\right) \pi^{\frac{1}{2}} z^{-\nu} 2^{\nu} J_{\nu}(z).$$

Thus with z = r|y| and  $\nu = \frac{n-2}{2}$  we find

$$\int_{\mathbb{S}^{n-1}} e^{-ir\langle \omega, y \rangle} d\omega = C_n \Gamma\left(\frac{n-1}{2}\right) \pi^{\frac{1}{2}} (r|y|)^{-\frac{n-2}{2}} 2^{\frac{n-2}{2}} J_{\frac{n-2}{2}}(r|y|)$$
$$= (2\pi)^{\frac{n}{2}} (r|y|)^{-\frac{n-2}{2}} J_{\frac{n-2}{2}}(r|y|),$$

as required.  $\Box$ 

The right hand side of Equation (2.18) is the integral kernel of the Hankel transform, which when combined with the statement of Lemma 2.2.9 give a description of the Fourier transform of functions which are invariant under rotations.

**Definition 2.2.10.** Let  $\mathcal{S}(\mathbb{R}^+)$  be the restriction of odd Schwarz functions on all of  $\mathbb{R}$  to the half-line. We define the *Hankel transform* of order  $\nu \in \mathbb{C}$  with  $\text{Re}(\nu) > -1$ , denoted  $H_{\nu}: \mathcal{S}(\mathbb{R}^+) \to \mathcal{S}(\mathbb{R}^+)$  by

$$[H_{\nu}\psi](\mu) = \int_{\mathbb{R}^+} \psi(\lambda)(\mu\lambda)^{\frac{1}{2}} J_{\nu}(\mu\lambda) \,d\lambda.$$

Remark 2.2.11. The Hankel transform is actually the integral transform diagonalising the differential operator associated to the Bessel differential equation obtained by considering only the radial component of  $H_0$  in polar coordinates. We will make use of this transform

in Chapter 4 when developing intuition for the form of the wave operators. Although we have only defined the Hankel transform for  $\psi \in \mathcal{S}(\mathbb{R}^+)$ , one can check that the Hankel transform extends to a unitary operator from  $L^2(\mathbb{R}^+, r^{\frac{1}{2}} dr)$  to itself, see [113]. A comprehensive collection of Hankel transforms can be found in [126].

**Lemma 2.2.12.** Suppose  $\psi \in \mathcal{S}(\mathbb{R}^n)$  is invariant under rotations. Let  $\tilde{\psi}$  be defined by  $\tilde{\psi}(r) = r^{\frac{n-1}{2}}\psi(r)$ . Then the Fourier transform of  $\psi$  is given by

$$[\mathcal{F}\psi](\xi) = |\xi|^{-\frac{n-1}{2}} [H_{\frac{n-2}{2}}\tilde{\psi}](|\xi|), \quad \xi \in \mathbb{R}^n,$$

where  $H_{\nu}$  denotes the Hankel transform of order  $\nu$ .

*Proof.* We take the Fourier transform of  $\psi$  and change to polar coordinates to obtain

$$[\mathcal{F}\psi](\xi) = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} e^{-i\langle \xi, y \rangle} \psi(y) \, \mathrm{d}y = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^+} r^{n-1} \psi(r) \left( \int_{\mathbb{S}^{n-1}} e^{-ir\langle \xi, \omega \rangle} \, \mathrm{d}\omega \right) \, \mathrm{d}r.$$

Applying Lemma 2.2.9 we then have

$$[\mathcal{F}\psi](\xi) = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^{+}} r^{n-1}\psi(r) \left( (2\pi)^{\frac{n}{2}} (r|\xi|)^{-\frac{n-2}{2}} J_{\frac{n-2}{2}}(r|\xi|) \right) dr$$

$$= |\xi|^{-\frac{n-1}{2}} \int_{\mathbb{R}^{+}} r^{\frac{n-1}{2}} \psi(r) (r|\xi|)^{\frac{1}{2}} J_{\frac{n-2}{2}}(r|\xi|) dr$$

$$= |\xi|^{-\frac{n-1}{2}} [H_{\frac{n-2}{2}} \tilde{\psi}](|\xi|).$$

An essential operator in our spectral analysis of  $H_0$  is the resolvent operator. For  $z \in \rho(H_0)$  with  $\operatorname{Arg}(z) \in (0, 2\pi)$ , we can define (using the holomorphic functional calculus) the resolvent operator (often called the free resolvent) by

$$R_0(z) = (H_0 - z)^{-1}. (2.19)$$

We can apply the techniques just described to obtain an explicit formula for the free resolvent in any dimension.

Corollary 2.2.13. For  $z \in \rho(H_0)$  with  $\operatorname{Arg}(z) \in (0, 2\pi)$  the integral kernel of the resolvent  $R_0(z) = (H_0 - z)^{-1}$  is given for  $x, y \in \mathbb{R}^n$ ,  $x \neq y$ , by

$$R_0(x,y,z) = \frac{i}{4} (2\pi)^{-\frac{n-2}{2}} z^{\frac{n-2}{4}} |x-y|^{-\frac{n-2}{2}} H_{\frac{n-2}{2}}^{(1)} (z^{\frac{1}{2}} |x-y|), \tag{2.20}$$

where  $H_{\nu}^{(1)} := J_{\nu} + iY_{\nu}$  denotes the Hankel function of the first kind of order  $\nu$ , described in [1, Chapters 9 and 10].

*Proof.* For r > 0 define  $\psi(r) = (r-z)^{-1}$ , so that  $R_0(z) = \psi(H_0)$ . By [125, Equation

1.4.13] the Hankel transform of  $\tilde{\psi}(r) = r^{\frac{n-1}{2}} (r^2 - z)^{-1}$  is given by

$$\[H_{\frac{n-2}{2}}\tilde{\psi}\](\mu) = \frac{i}{4}(2\pi)^{-1}z^{\frac{n-2}{4}}\mu^{\frac{1}{2}}H_{\frac{n-2}{2}}^{(1)}(z^{\frac{1}{2}}\mu). \tag{2.21}$$

Hence using Equation (2.17) the integral kernel of the free resolvent is given explicitly for  $x, y \in \mathbb{R}^n$  by

$$R_0(x,y,z) = (2\pi)^{-\frac{n}{2}} [\mathcal{F}\psi](y-x) = \frac{i}{4} (2\pi)^{-\frac{n-2}{2}} z^{\frac{n-2}{4}} |x-y|^{-\frac{n-2}{2}} H_{\frac{n-2}{2}}^{(1)}(z^{\frac{1}{2}} |x-y|), \quad (2.22)$$

as claimed. 
$$\Box$$

In odd dimensions the formula is particularly nice, for example if n=3 then

$$R_0(x, y, z) = \frac{e^{iz^{\frac{1}{2}}|x-y|}}{4\pi|x-y|}.$$
 (2.23)

One can explicitly compute the integral kernel of the resolvent in higher odd dimensions in terms of polynomials in |x - y| also, see for example [56, Theorem 3.3]. The even dimensional cases are complicated by a logarithmic singularity of the Hankel function near the origin, as we shall see in Chapter 3 where we discuss expansions of the resolvent for small z.

#### 2.2.2 The perturbed Hamiltonian

In this section we describe a class of perturbations of our free Hamiltonian  $H_0$  by real valued potentials V satisfying certain decay assumptions. Such operators  $H = H_0 + V$  are known as Schrödinger operators and have been studied extensively since the introduction of the Schrödinger equation in [147]. The study of such perturbations motivated much of the development of perturbation theory and thus the literature is vast, we refer the reader to [21, 23, 95, 135, 136, 137] and the references therein for a more comprehensive discussion.

The most basic and minimal assumption we make on our potential V is the following.

**Assumption 2.2.14.** The function  $V : \mathbb{R}^n \to \mathbb{R}$  is real-valued and for almost all  $x \in \mathbb{R}^n$  satisfies  $|V(x)| \leq C(1+|x|)^{-\rho}$  for some  $\rho > 1$ .

Throughout this thesis we will need to strengthen Assumption 2.2.14 in various places by requiring greater decay.

**Lemma 2.2.15.** Suppose that V satisfies Assumption 2.2.14. Then the operator  $H = H_0 + V$  is essentially self-adjoint with  $\sigma_{ess}(H) = [0, \infty)$ . Moreover,  $C_c^{\infty}(\mathbb{R}^n)$  is a core for H.

Proof. We use the Kato-Rellich theorem (see [135, Theorem X.12] for example). It is immediate that  $\text{Dom}(H_0) \subset \text{Dom}(V) := \mathcal{H}$ . For  $z \in \rho(H_0)$  we have that  $R_0(z)V$  is compact by [158, Lemma 7.21] and thus V is  $H_0$ -relatively compact. The Kato-Rellich theorem [135, Theorem X.12] then tells us that H is self-adjoint on  $\text{Dom}(H_0)$ , bounded from below, and  $\sigma_{ess}(H) = \sigma_{ess}(H_0) = [0, \infty)$ . That  $C_c^{\infty}(\mathbb{R}^n)$  is a core for H follows from the fact that it is a core for  $H_0$ .

**Definition 2.2.16.** For a function V satisfying Assumption 2.2.14 we call the operator  $H = H_0 + V$  a (short-range) perturbed Hamiltonian.

The following fact, due to Kato [93], guarantees that there are no eigenvalues of H embedded in the continuous spectrum. A simple proof in the case that V is compactly supported can be found as [164, Theorem 6.1.1].

**Theorem 2.2.17** (Kato's theorem). Suppose that V satisfies Assumption 2.2.14. Then the operator  $H = H_0 + V$  has no positive eigenvalues.

The number of eigenvalues is known to be finite for  $\rho > 2$ , although proofs are rather involved so we provide only references. The statement we use here is [164, Section 7.4.2]. Many explicit bounds on the number of eigenvalues are discussed in [151, Chapter 7].

**Theorem 2.2.18.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 2$ . Then the operator  $H = H_0 + V$  has finitely many eigenvalues.

#### 2.2.3 The dilation operator

We first consider the (one-dimensional) Dirac operator  $D = \frac{1}{i} \frac{d}{dx}$  with  $Dom(D) = C_c^{\infty}(\mathbb{R}) \subset L^2(\mathbb{R})$ . It is well-known that D is essentially self-adjoint and so has a unique self-adjoint extension which we also denote by D. The one-dimensional Fourier transform shows that D is unitarily equivalent to the multiplication operator M defined by  $[Mf](\xi) = \xi f(\xi)$  and thus  $\sigma(D) = \sigma_{ac}(D) = \mathbb{R}$ .

By Stone's theorem the operator D defines a strongly continuous one-parameter unitary group  $(V(t))_{t\in\mathbb{R}}$  acting on  $L^2(\mathbb{R})$ . As is well-known, V(t) acts on  $f\in L^2(\mathbb{R})$  by

$$[V(t)f](x) = [e^{-itD}f](x) = f(x-t)$$
(2.24)

so that D is the generator of the translation group on  $\mathbb{R}$ . Since the operator  $H_0$  has absolutely continuous spectrum  $\sigma(H_0) = [0, \infty)$  we are interested in the Hilbert space  $L^2(\mathbb{R}^+)$ . We can naturally identify  $L^2(\mathbb{R}^+)$  with  $L^2(\mathbb{R})$  via the following.

**Lemma 2.2.19.** The map  $L: L^2(\mathbb{R}) \to L^2(\mathbb{R}^+)$  defined for  $\lambda \in \mathbb{R}^+$  by  $[Lf](\lambda) = \lambda^{-\frac{1}{2}} f(\ln(\lambda))$  is unitary.

*Proof.* We check that for  $f, g \in L^2(\mathbb{R})$  we have

$$\langle Lf, Lg \rangle = \int_{\mathbb{R}^+} \overline{[Lf](\lambda)} [Lg](\lambda) \, \mathrm{d}\lambda = \int_{\mathbb{R}^+} \lambda^{-1} \overline{f(\ln{(\lambda)})} g(\ln{(\lambda)}) \, \mathrm{d}\lambda = \int_{\mathbb{R}} \overline{f(y)} g(y) \, \mathrm{d}y = \langle f, g \rangle,$$

where we have made the substitution  $y = \ln(\lambda)$ . So the map L is an isometry. The inverse map  $L^{-1}: L^2(\mathbb{R}^+) \to L^2(\mathbb{R})$  is given for  $y \in \mathbb{R}$  by  $[L^{-1}g](y) = e^{\frac{y}{2}}g(e^y)$ . It is readily checked that  $L^{-1}$  is an isometry also and thus L is unitary as claimed.

The following result shows how the translation group of Equation (2.24) and the generator of translations act when mapped to  $L^2(\mathbb{R}^+)$  by the unitary L.

**Lemma 2.2.20.** The group  $(U_+(t))_{t\in\mathbb{R}}$  defined by  $U_+(t) = LV(-t)L^*$  is given for  $\lambda \in \mathbb{R}^+$  by

$$[U_{+}(t)f](\lambda) = e^{\frac{t}{2}}f(e^{t}\lambda)$$

and the action of the self-adjoint generator  $D_+$  of  $U_+$  on  $f \in C_c^{\infty}(\mathbb{R}^+)$  is given, for  $\lambda \in \mathbb{R}^+$ , by

$$[D_+ f](\lambda) = \left(\frac{\lambda}{i} \frac{\mathrm{d}f}{\mathrm{d}\lambda} + \frac{1}{2i} f(\lambda)\right).$$

*Proof.* Direct computation shows that for  $f \in L^2(\mathbb{R}^+)$  and  $\lambda \in \mathbb{R}^+$  we have

$$[U_{+}(t)f](\lambda) = [LV(-t)L^{*}f](\lambda) = \lambda^{-\frac{1}{2}}[V(-t)L^{*}f](\ln(\lambda))$$
$$= \lambda^{-\frac{1}{2}}[L^{*}f](\ln(\lambda) + t) = \lambda^{-\frac{1}{2}}e^{\frac{\ln(\lambda) + t}{2}}f(e^{\ln(\lambda) + t}) = e^{\frac{t}{2}}f(e^{t}\lambda),$$

as claimed. We can then use Stone's theorem to compute the self-adjoint generator  $D_+$  of the unitary group  $(U_+(t))$  via the computation

$$[D_{+}f](\lambda) = -i\left(\frac{\mathrm{d}}{\mathrm{d}t}[U_{+}(t)f](\lambda)\right)\Big|_{t=0} = -i\left(\frac{\mathrm{d}}{\mathrm{d}t}\mathrm{e}^{\frac{t}{2}}f(\mathrm{e}^{t}\lambda)\right)\Big|_{t=0}$$
$$= -i\left(\frac{\mathrm{d}}{\mathrm{d}t}\frac{1}{2}\mathrm{e}^{\frac{t}{2}}f(\mathrm{e}^{t}\lambda) + \lambda\mathrm{e}^{\frac{3t}{2}}f'(\mathrm{e}^{t}\lambda)\right)\Big|_{t=0} = \left(\frac{1}{2i}f(\lambda) + \frac{\lambda}{i}f'(\lambda)\right),$$

which shows that  $D_+$  takes the form claimed.

We call  $D_+$  the generator of dilations on  $L^2(\mathbb{R}^+)$  and use the group structure to describe a simple and useful functional calculus for  $D_+$ .

**Lemma 2.2.21.** Suppose that  $\varphi \in \mathcal{S}(\mathbb{R}^+)$  (with  $\mathcal{S}(\mathbb{R}^+)$  defined in Definition 2.2.10). Then for  $f \in \text{Dom}(D_+)$  we have

$$[\varphi(D_+)f](\lambda) = (2\pi)^{-\frac{1}{2}} \int_{\mathbb{R}} [\mathcal{F}^*\varphi](t)[U_+(t)f](\lambda) \, \mathrm{d}t, \quad \lambda \in \mathbb{R}^+.$$

*Proof.* We let  $\varphi \in \mathcal{S}(\mathbb{R}^+)$  and define for  $\xi \in \mathbb{R}$  the multiplication operator  $[M_{\varphi}f](\xi) = \varphi(-\xi)f(\xi)$ . Then for  $\lambda \in \mathbb{R}^+$  we have

$$[\varphi(D_+)f](\lambda) = [LL^*\varphi(D_+)f](\lambda) = [L\varphi(-D)L^*f](\lambda)$$
$$= [L\mathcal{F}\mathcal{F}^*\varphi(-D)L^*f](\lambda) = [L\mathcal{F}M_\varphi\mathcal{F}^*L^*f](\lambda).$$

For  $\xi \in \mathbb{R}$  we can explicitly write

$$[\mathcal{F}M_{\varphi}\mathcal{F}^*g](\xi) = (2\pi)^{-\frac{1}{2}} \int_{\mathbb{R}} e^{-iy\xi} \varphi(y) [\mathcal{F}^*g](y) \, dy$$

$$= (2\pi)^{-1} \int_{\mathbb{R}} \int_{\mathbb{R}} e^{-iy(\xi-u)} \varphi(-y) g(u) \, du \, dy$$

$$= (2\pi)^{-\frac{1}{2}} \int_{\mathbb{R}} \left( (2\pi)^{-\frac{1}{2}} \int_{\mathbb{R}} e^{-iy(\xi-u)} \varphi(-y) \, dy \right) \, du$$

$$= (2\pi)^{-\frac{1}{2}} \int_{\mathbb{R}} [\mathcal{F}\varphi](u-\xi) g(u) \, du.$$

So for  $\lambda \in \mathbb{R}^+$  we find

$$[\varphi(D_{+})f](\lambda) = [L\mathcal{F}M_{\varphi}\mathcal{F}^{*}L^{*}f](\lambda) = \lambda^{-\frac{1}{2}}[\mathcal{F}M_{\varphi}\mathcal{F}^{*}L^{*}f](\ln(\lambda))$$

$$= (2\pi)^{-\frac{1}{2}}\lambda^{-\frac{1}{2}} \int_{\mathbb{R}} [\mathcal{F}\varphi](u - \ln(\lambda))[L^{*}f](u) du$$

$$= (2\pi)^{-\frac{1}{2}}\lambda^{-\frac{1}{2}} \int_{\mathbb{R}} [\mathcal{F}\varphi](u - \ln(\lambda))e^{\frac{u}{2}}f(e^{u}) du.$$

Now make the substitution  $t = \ln(\lambda) - u$  to obtain

$$[\varphi(D_{+})f](\lambda) = (2\pi)^{-\frac{1}{2}}\lambda^{-\frac{1}{2}} \int_{\mathbb{R}} [\mathcal{F}^{*}\varphi](\ln(\lambda) - u)e^{\frac{u}{2}}f(e^{u}) du$$
$$= (2\pi)^{-\frac{1}{2}} \int_{\mathbb{R}} [\mathcal{F}^{*}\varphi](t)e^{-\frac{t}{2}}f(e^{-t}\lambda) dt$$
$$= (2\pi)^{-\frac{1}{2}} \int_{\mathbb{R}} [\mathcal{F}^{*}\varphi](t)[U_{+}(t)f](\lambda) dt,$$

which completes the proof.

We can now construct an operator which diagonalises the generator of dilations by using our unitaries.

**Definition 2.2.22.** We define the *Mellin transform*  $\mathcal{M}: L^2(\mathbb{R}^+) \to L^2(\mathbb{R})$  by  $\mathcal{M} = \mathcal{F}L^*$ , where L is defined in Lemma 2.2.19.

**Lemma 2.2.23.** The Mellin transform is given explicitly for  $f \in C_c^{\infty}(\mathbb{R}^+)$  and  $x \in \mathbb{R}$  by

$$[\mathcal{M}f](x) = (2\pi)^{-\frac{1}{2}} \int_{\mathbb{R}^+} y^{-\frac{1}{2} - ix} f(y) \, dy$$
 (2.25)

and satisfies  $\mathcal{M}D_+\mathcal{M}^*=M$ , where M is the multiplication operator by the variable in  $L^2(\mathbb{R})$ .

*Proof.* Equation (2.25) follows from the definitions of  $\mathcal{F}$  and L. The diagonalisation relation is a consequence of the relation  $D_+ = L^*DL$  and the fact that the Fourier transform diagonalises the differentiation operator D.

We can easily generalise the group of dilations to  $L^2(\mathbb{R}^n)$  also.

**Definition 2.2.24.** We define the dilation group  $(U_n(t))_{t\in\mathbb{R}}$  on  $L^2(\mathbb{R}^n)$  by the formula

$$[U_n(t)f](x) = e^{\frac{nt}{2}}f(e^tx), \quad x \in \mathbb{R}^n.$$

**Lemma 2.2.25.** We have that  $(U_n(t))$  is a strongly continuous one-parameter unitary group and the self-adjoint generator  $D_n$  of  $U_n$  is given on  $f \in C_c^{\infty}(\mathbb{R}^n)$  by

$$[D_n f](x) = \frac{1}{i} \sum_{j=1}^n x_j \frac{\partial f}{\partial x_j} + \frac{n}{2i} f(x), \quad x \in \mathbb{R}^n.$$

*Proof.* That  $(U_n(t))$  defines a strongly continuous one-parameter unitary group can be readily checked. Fix  $f \in C_c^{\infty}(\mathbb{R}^n)$ . To determine the self-adjoint generator  $D_n$ , we again use Stone's theorem to find for  $x \in \mathbb{R}^n$  that

$$[D_n f](x) = -i \left( \frac{\mathrm{d}}{\mathrm{d}t} [U_n(t)f](x) \right) \Big|_{t=0} = -i \left( \frac{\mathrm{d}}{\mathrm{d}t} \mathrm{e}^{\frac{nt}{2}} f(\mathrm{e}^t x) \right) \Big|_{t=0}$$
$$= -i \left( \frac{n}{2} \mathrm{e}^{\frac{nt}{2}} f(\mathrm{e}^t x) + \mathrm{e}^{\frac{nt}{2}} \sum_{j=1}^n \frac{\mathrm{d}}{\mathrm{d}t} (\mathrm{e}^t x_j) \frac{\partial f}{\partial x_j} (\mathrm{e}^t x) \right) \Big|_{t=0}$$
$$= -i \left( \frac{n}{2} f(x) + \sum_{j=1}^n x_j \frac{\partial f}{\partial x_j} \right),$$

so that  $D_n$  is given by the claimed formula.

**Lemma 2.2.26.** The operator  $D_n$  anticommutes with the Fourier transform. The operator  $\mathcal{U}$  of Definition 2.2.7 satisfies  $\mathcal{U}D_n\mathcal{U}^* = 2D_+ \otimes \operatorname{Id}$ , and  $F_0D_nF_0^* = -2D_+ \otimes \operatorname{Id}$ .

*Proof.* For  $f \in C_c^{\infty}(\mathbb{R}^n)$  define the operators  $M_j$  by  $[M_j f](x) = x_j \frac{\partial f}{\partial x_j}$ . We compute for  $\xi \in \mathbb{R}^n$  that

$$[\mathcal{F}M_j f](\xi) = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} e^{-i\langle x,\xi\rangle} [M_j f](x) dx = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} e^{-i\langle x,\xi\rangle} x_j \frac{\partial f}{\partial x_j} dx.$$

Integrating by parts in the  $x_i$  variable and using the compact support of f to eliminate

boundary conditions we obtain for  $\xi \in \mathbb{R}^n$  that

$$[\mathcal{F}M_{j}f](\xi) = -(2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^{n}} f(x) \frac{\partial}{\partial x_{j}} \left( x_{j} e^{-i\langle x, \xi \rangle} \right) dx$$

$$= -(2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^{n}} f(x) e^{-i\langle x, \xi \rangle} dx - \xi_{j} (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^{n}} f(x) (-ix_{j}) e^{-i\langle x, \xi \rangle} dx$$

$$= -[\mathcal{F}f](\xi) - \xi_{j} \frac{\partial}{\partial \xi_{j}} [\mathcal{F}f](\xi)$$

$$= -[(\mathrm{Id} + M_{j})\mathcal{F}f](\xi).$$

Using Lemma 2.2.25 the operator  $D_n$  is given by

$$D_n = \frac{n}{2i} \operatorname{Id} + \frac{1}{i} \sum_{j=1}^n M_j.$$

So for  $f \in C_c^{\infty}(\mathbb{R}^n)$  and  $\xi \in \mathbb{R}^n$  we find

$$[\mathcal{F}D_n f](\xi) = \frac{n}{2i} [\mathcal{F}f](\xi) + \frac{1}{i} \sum_{j=1}^n [\mathcal{F}M_j f](\xi) = \frac{n}{2i} [\mathcal{F}f](\xi) - \frac{1}{i} \sum_{j=1}^n [(\mathrm{Id} + M_j)\mathcal{F}f](\xi)$$
$$= -\frac{n}{2i} [\mathcal{F}f](\xi) - \frac{1}{i} \sum_{j=1}^n [M_j \mathcal{F}f](\xi) = -[D_n \mathcal{F}f](\xi),$$

so that  $D_n$  anticommutes with  $\mathcal{F}$ . Fix  $\lambda \in \mathbb{R}^+$  and  $\omega \in \mathbb{S}^{n-1}$ . We next check that for  $f \in C_c^{\infty}(\mathbb{R}^n)$  that

$$[\mathcal{U}M_j f](\lambda, \omega) = 2^{-\frac{1}{2}} \lambda^{\frac{n-2}{4}} [M_j f](\lambda^{\frac{1}{2}} \omega) = 2^{-\frac{1}{2}} \lambda^{\frac{n-2}{4}} \lambda^{\frac{1}{2}} \omega_j \frac{\partial f}{\partial x_j}(\lambda^{\frac{1}{2}} \omega).$$

So for  $f \in C_c^{\infty}(\mathbb{R}^n)$  we use the chain rule to see that

$$\sum_{j=1}^{n} [\mathcal{U}M_{j}f](\lambda,\omega) = \sum_{j=1}^{n} \left(2^{-\frac{1}{2}}\lambda^{\frac{n-2}{4}}\lambda^{\frac{1}{2}}\omega_{j}\frac{\partial f}{\partial x_{j}}(\lambda^{\frac{1}{2}}\omega)\right) = 2^{-\frac{1}{2}}\lambda^{\frac{n-2}{4}}2\lambda\frac{\mathrm{d}}{\mathrm{d}\lambda}f(\lambda^{\frac{1}{2}}\omega)$$
$$= 2\lambda\left(\frac{\mathrm{d}}{\mathrm{d}\lambda}\otimes\mathrm{Id}\right)([\mathcal{U}f](\lambda,\omega)) - \frac{n-2}{2}[\mathcal{U}f](\lambda,\omega).$$

Thus for  $f \in C_c^{\infty}(\mathbb{R}^n)$  we find

$$[\mathcal{U}D_n f](\lambda, \omega) = \frac{n}{2i} [\mathcal{U}f](\lambda, \omega) + \frac{1}{i} \sum_{j=1}^n [\mathcal{U}M_j f](\lambda, \omega)$$

$$= \frac{n}{2i} [\mathcal{U}f](\lambda, \omega) + \left( \left[ 2 \left( \frac{\mathrm{d}}{\mathrm{d}\lambda} \otimes \mathrm{Id} \right) \mathcal{U}f \right] (\lambda, \omega) - \frac{n-2}{2i} [\mathcal{U}f](\lambda, \omega) \right)$$

$$= 2 \left[ (D_+ \otimes \mathrm{Id}) \mathcal{U}f \right] (\lambda, \omega)$$

and so  $\mathcal{U}D_n\mathcal{U}^* = 2(D_+ \otimes \mathrm{Id})$ . The final claim now follows from the definition  $F_0 = \mathcal{U}\mathcal{F}$ .

Corollary 2.2.27. We have  $\sigma(D_n) = \sigma_{ac}(D_n) = \mathbb{R}$ .

Proof. Lemma 2.2.26 shows that  $D_n$  is unitarily equivalent to  $D_+ \otimes \operatorname{Id}_{\mathcal{P}}$  (where  $\mathcal{P} = L^2(\mathbb{S}^{n-1})$ ), which is in turn unitarily equivalent to  $D \otimes \operatorname{Id}_{\mathcal{P}}$  by Lemma 2.2.20. Thus we find that  $D_n$  has purely absolutely continuous spectrum and  $\sigma_{ac}(D_n) = \sigma_{ac}(D \otimes \operatorname{Id}_{\mathcal{P}}) = \mathbb{R}$ .  $\square$ 

Corollary 2.2.28. Let  $B = \frac{1}{2} \ln (H_0)$  acting on  $\mathcal{H} = L^2(\mathbb{R}^n)$ . Then we have  $[D_n, B] = i \mathrm{Id}$ .

Proof. It is well-known that the operator  $M: \mathrm{Dom}(D) \to L^2(\mathbb{R})$  defined for  $f \in C_c^{\infty}(\mathbb{R})$  and  $\xi \in \mathbb{R}$  by  $[Mf](\xi) = \xi f(\xi)$  satisfies the relation  $[D, M] = -i\mathrm{Id}$  on  $\mathrm{Dom}(D)$ . Define for  $g \in C_c^{\infty}(\mathbb{R}^+)$  and  $\lambda \in \mathbb{R}^+$  the operator  $M_{\mathrm{ln}}$  on  $C_c^{\infty}(\mathbb{R}^+)$  by  $[M_{\mathrm{ln}}g](\lambda) = \ln(\lambda)g(\lambda)$ . We can compute that

$$[D_n, B] = F_0^* F_0[D_n, B] F_0^* F_0 = F_0^* [F_0 D_n F_0^*, F_0 B F_0^*] F_0$$
  
=  $F_0^* [-2(D_+ \otimes \mathrm{Id}), \frac{1}{2} (M_{\mathrm{ln}} \otimes \mathrm{Id})] F_0 = -F_0^* ([D_+, M_{\mathrm{ln}}] \otimes \mathrm{Id}) F_0.$ 

Conjugating by the unitary L of Lemma 2.2.19 we find

$$[D_n, B] = -F_0^*([D_+, M_{\ln}] \otimes \operatorname{Id})F_0 = -F_0^*(L \otimes \operatorname{Id})[L^*D_+L, L^*M_{\ln}L](L^* \otimes \operatorname{Id})F_0$$
$$= -F_0^*(L \otimes \operatorname{Id})[D, M](L^* \otimes \operatorname{Id})F_0 = i\operatorname{Id},$$

as claimed.  $\Box$ 

Corollary 2.2.28 shows that the operators B and  $D_n$  are canonically conjugate and thus we can use Stone-von Neumann theorem methods of [151, Chapter 4] (with some modifications, see Lemma 4.2.6) to deduce the compactness of operators of the form  $f(D_n)g(B)$ . The basic result is the following.

**Theorem 2.2.29** ([151, Theorem 4.1]). If  $f, g \in L^p(\mathbb{R}^n)$  with  $p \in [2, \infty)$  then we have the inclusion  $f(X)g(D) \in \mathcal{L}^p(\mathcal{H})$  (the p-th Schatten class)

$$||f(X)g(D)||_p \le (2\pi)^{-\frac{n}{p}} ||f||_p ||g||_p$$
.

We can use the structure of the dilation group to obtain a useful form of the functional calculus for  $D_n$  in an analogous manner to the functional calculus for  $D_+$  in Lemma 2.2.21

**Lemma 2.2.30.** Suppose that  $\varphi \in \mathcal{S}(\mathbb{R})$ . Then for  $f \in \text{Dom}(D_n)$  we have

$$[\varphi(D_n)f](x) = (2\pi)^{-\frac{1}{2}} \int_{\mathbb{D}} [\mathcal{F}^*\varphi](t)[U_n(t)f](x) dt.$$

**Definition 2.2.31.** We denote by  $P_{\pm}$  the positive and negative spectral projections of  $D_n$ , so that  $P_+ = \chi_{[0,\infty)}(D_n)$ ,  $P_- = \chi_{(-\infty,0)}(D_n)$  and  $P_+ + P_- = \text{Id}$ .

The spectral projections  $P_{\pm}$  will be used in the proof of the asymptotic completeness of the wave operators for the pair  $(H, H_0)$  and the absence of singular continuous spectrum of H in Section 2.3.

# 2.3 Existence and completeness for Schrödinger operators

In this section we demonstrate both the existence and asymptotic completeness of the wave operators for Schrödinger operators with potentials satisfying Assumption 2.2.14. The generator of dilations is used in a rather remarkable way to demonstrate the asymptotic completeness of the wave operators for a short range potential. This geometric approach to asymptotic completeness is due to Enss [58, 59, 60]. There are other methods of obtaining asymptotic completeness, for example the method of  $H_0$ -smoothness discussed in [164, Chapter 1.6]. The technique we present here follows closely that of Perry [128] (see also [127, Chapter 12] for more details in dimension n = 3).

We first apply Cook's criterion in the manner described in [106, Section 3.4] to obtain the existence of the wave operators.

**Theorem 2.3.1.** Let  $H = H_0 + V$  with V satisfying Assumption 2.2.14. Then the wave operators  $W_{\pm}(H, H_0)$  exist.

*Proof.* By Cook's criterion it suffices to show that

$$\int_{\mathbb{R}} \left| \left| V e^{-itH_0} f \right| \right|_2 \, \mathrm{d}t < \infty \tag{2.26}$$

for  $\mathcal{F}f \in C_c^{\infty}(\mathbb{R}^n \setminus \{0\})$ . Suppose that  $[\mathcal{F}f](\xi) = 0$  for  $|\xi| \leq r$  and let  $P_r$  denote the projection onto functions supported on the ball of radius r in  $\mathbb{R}^n$ . Then we have

$$||Ve^{-itH_0}f||_2 \le ||V||_{\infty} ||P_{r|t|}e^{-iH_0t}f||_2 + ||V(Id - P_{r|t|})||_{\infty} ||f||_2.$$

The method of stationary phase (see [78, Lemma A.1]) and [164, Lemma 1.2.5] show that the first term on the right hand side is bounded by any positive integer power of  $|t|^{-1}$ . Since V satisfies Assumption 2.2.14 for some  $\rho > 1$ , the second term on the right hand side is  $O(|t|^{-\rho})$  and thus the integral in Equation (2.26) is finite.

To deduce the compactness of certain operators, we need the following result, which can be found as [128, Lemma 4].

**Theorem 2.3.2.** Suppose that  $f \in C_0(\mathbb{R})$  and that V satisfies Assumption 2.2.14 for some  $\rho > \frac{n}{2}$ . Then the operator  $f(H) - f(H_0)$  is compact.

Proof. The assumption  $\rho > \frac{n}{2}$  guarantees that  $R(z) - R_0(z) \in \mathcal{L}^p(\mathcal{H})$  for all p > n and  $z \in \mathbb{C} \setminus \mathbb{R}$  and is thus compact. By analogous arguments to [150, Lemma 4] one can show that  $R(-i)^j R(i)^\ell - R_0(-i)^j R_0(i)^\ell$  is compact for all  $j, \ell \in \mathbb{N}$  and thus by Stone-Weierstrass the conclusion follows for arbitrary  $f \in C_0(\mathbb{R})$ .

We now proceed to the proof of asymptotic completeness of the wave operators, following closely the method of Perry [128]. The key tool in the following argument is the dilation operator  $D_n$  and the Mellin transform diagonalising  $D_n$ .

**Definition 2.3.3.** The Mellin transform  $\mathcal{M}: C_c^{\infty}(\mathbb{R}^n \setminus \{0\}) \to L^2(\mathbb{R}) \otimes \mathcal{P}$  is defined for  $f \in C_c^{\infty}(\mathbb{R}^n \setminus \{0\}), s \in \mathbb{R}$  and  $\omega \in \mathbb{S}^{n-1}$  by

$$[\mathcal{M}f](s,\omega) = (2\pi)^{-\frac{1}{2}} \int_0^\infty \lambda^{\frac{n-2}{2} - is} f(\lambda\omega) \,d\lambda.$$

The Mellin transform extends to a unitary operator  $\mathcal{M}: L^2(\mathbb{R}^n) \to L^2(\mathbb{R}) \otimes \mathcal{P}$  which satisfies  $\mathcal{M}D_n\mathcal{M}^* = M_s$ , the operator of multiplication by the first variable on  $L^2(\mathbb{R}) \otimes \mathcal{P}$ . Recall the non-negative and negative spectral projections  $P_{\pm}$  for  $D_n$  of Definition 2.2.31. As usual for  $z \in \mathbb{C} \setminus \mathbb{R}$  we write  $R(z) = (H - z)^{-1}$  and  $R_0(z) = (H_0 - z)^{-1}$ .

**Lemma 2.3.4.** Let  $g \in C_c^{\infty}(\mathbb{R}^+)$  have support in an interval [a,b] and let  $\delta \in (0,a)$ . Let  $\tilde{\chi}_A$  denote the characteristic function of a set  $A \subset \mathbb{R}^n$ . Let  $k \in \mathbb{N}$  and  $t \in (0,\pm\infty)$ . For  $|x| < \delta |t|$  define  $K_{x,t} \in C_c^{\infty}(\mathbb{R}^n \setminus \{0\})$  by

$$K_{x,t}(\xi) = e^{i|\xi|^2 t - i\langle\xi, x\rangle} \overline{g(|\xi|^2)}.$$
(2.27)

Then we have for  $\pm t > 0$  the estimate

$$||P_{\mp}K_{x,t}|| \le C_k(1+|t|)^{-k+\frac{1}{2}}$$

for some constant  $C_k$ .

*Proof.* We show only the case t > 0, since the case t < 0 is similar. Using the unitarity of the Mellin transform we have  $||P_{-}K_{x,t}|| = ||\chi_{(-\infty,0)}\mathcal{M}K_{x,t}||$ . Since  $K_{x,t} \in C_c^{\infty}(\mathbb{R}^n \setminus \{0\})$ , we have for  $s \in \mathbb{R}$  and  $\omega \in \mathbb{S}^{n-1}$  that

$$[\mathcal{M}K_{x,t}](s,\omega) = (2\pi)^{-\frac{1}{2}} \int_0^\infty \lambda^{\frac{n-2}{2}} e^{it\lambda^2 - i\langle\lambda\omega,x\rangle - is\log(\lambda)} \overline{g(\lambda^2)} \,d\lambda.$$

The function  $f:(0,\infty)\to\mathbb{C}$  defined by  $f(\lambda)=\lambda^{\frac{n-2}{2}}\overline{g(\lambda^2)}$  is smooth and compactly supported. The phase function  $\phi_{x,s,t}=\phi:\mathbb{R}^+\times\mathbb{S}^{n-1}\to\mathbb{R}$ , defined by  $\phi(\lambda,\omega)=t\lambda^2-i\langle\lambda x,\omega\rangle-is\log(\lambda)$ , has derivative  $\frac{\partial\phi}{\partial\lambda}=2\lambda t-\langle x,\omega\rangle-\frac{s}{\lambda}$  has no stationary points for the allowed range of x,t,s. The method of stationary phase [78, Lemma A.3] then

gives the pointwise estimate

$$|[\mathcal{M}K_{x,t}](s,\omega)| < C_{N,g}(1+|t|+|s|)^{-k}.$$

Integrating this estimate in s and  $\omega$  gives directly the statement of the lemma.

**Lemma 2.3.5.** Let  $g \in C_c^{\infty}(\mathbb{R}^+)$  have support in an interval  $[a^2, b^2]$  and let  $\delta \in (0, a)$ . Let  $\chi_A$  denote the characteristic function of a set  $A \subset \mathbb{R}^n$ . Then for any  $k \in \mathbb{N}$  and  $t \in (0, \pm \infty)$  we have

$$||\chi_{B(\delta|t|)}e^{-itH_0}g(H_0)P_{\pm}|| \le C_{N,g}(1+|t|)^{-k+n+1},$$

where  $B(\delta|t|) = \{y \in \mathbb{R}^n : |y| < \delta|t|\}$ , the open ball of radius  $\delta|t|$ .

*Proof.* For  $x \in \mathbb{R}^n$ ,  $g \in C_c^{\infty}(\mathbb{R}^+)$  and  $\psi \in C_c^{\infty}(\mathbb{R}^n)$  define

$$\psi_{t\pm}(x) = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} e^{i\langle \xi, x \rangle - i|\xi|^2 t} g(|\xi|^2) [P_{\pm} \mathcal{F} \psi](\xi) \,d\xi.$$

We consider the equality  $\psi_{t\pm}(x) = \langle K_{x,t}, P_{\mp} \mathcal{F} \psi \rangle$ . The Cauchy-Schwarz inequality then gives us

$$|\psi_{t\pm}(x)| = |\langle K_{x,t}, P_{\mp}\mathcal{F}\psi\rangle| = |\langle P_{\mp}K_{x,t}, \mathcal{F}\psi\rangle| \le ||P_{\mp}K_{x,t}|| \, ||\psi||.$$

By Lemma 2.3.4 we find  $||P_{\mp}K_{x,t}|| \leq C_{k,g}(1+|t|)^{-k+\frac{1}{2}}$  and thus

$$|\psi_{t\pm}(x)| \le C_{k,g} (1+|t|)^{-k+\frac{1}{2}} ||\psi||.$$

We thus find

$$\begin{aligned} \left| \left| \chi_{B(\delta|t|)} e^{-itH_0} g(H_0) P_{\pm} \right| \right|^2 &= \sup_{\|\psi\|=1} \int_{|x| \le \delta|t|} \left| \left[ e^{-itH_0} g(H_0) P_{\pm} \psi \right] (x) \right|^2 dx \\ &= \sup_{\|\psi\|=1} \int_{|x| < \delta|t|} \left| \psi_{t\pm}(x) \right|^2 dx \\ &\le C_{k,g}^2 \int_{|x| < \delta|t|} (1+|t|)^{-2k+1} dx \\ &= \tilde{C} \delta^n |t|^n (1+|t|)^{-2k+1} \\ &\le \tilde{C} \delta^n (1+|t|)^{-2k+n+1}, \end{aligned}$$

from which the claim follows.

**Lemma 2.3.6.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > n$ . Let  $\alpha \ge 1$  and  $g \in C_c^{\infty}(\mathbb{R}^+)$ . Then the operator  $B = (H+i)^{-\alpha}(W_{\mp} - \operatorname{Id})g(H_0)P_{\pm}$  is compact.

*Proof.* We show that B is the norm limit of compact operators. Consider for  $\pm t > 0$  the equality

$$R(-i)^{\alpha} (e^{itH} e^{-itH_0} - Id) g(H_0) P_{\pm} = \int_0^t R(-i)^{\alpha} e^{isH} V e^{-isH_0} g(H_0) P_{\pm} ds.$$
 (2.28)

Since the family  $s \mapsto e^{isH}$  is strongly continuous,  $s \mapsto e^{-isH_0}g(H_0)$  is norm continuous and the operator  $R(-i) - R_0(-i)$  is compact (see the proof of Theorem 2.3.2), we find that the right-hand side of Equation (2.28) defines a compact operator for finite t. It therefore suffices to show that the map  $s \mapsto ||R(-i)^{\alpha}e^{isH}Ve^{-isH_0}g(H_0)P_{\pm}||$  defines an integrable function on  $(0, \pm \infty)$ . So we make for  $\beta \geq 1$  the estimate

$$\begin{aligned} \left| \left| R(-i)^{\alpha} V e^{-itH_0} g(H_0) P_{\pm} \right| \right| &\leq \left| \left| R(-i)^{\alpha} V R_0(-i)^{\beta} \right| \left| \left| \left| \chi_{B(\delta|t|)} e^{-itH_0} R_0(-i)^{-\beta} g(H_0) P_{\pm} \right| \right| \\ &+ C \left| \left| R(-i)^{\alpha} V R_0(-i)^{\beta} \chi_{B(\delta|t|)^c} \right| \right|. \end{aligned}$$

By Lemma 2.3.5 the first term decays rapidly in t and is thus integrable. The second term is integrable by our assumption on the potential. Thus we have shown that the operator B is the norm limit of compact operators and is hence compact.

**Theorem 2.3.7.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > \frac{n+1}{2}$ . Then the wave operators  $W_{\pm}$  are asymptotically complete.

Proof. We first show that H has no singular continuous spectrum. Let  $P_A(H)$  be the proejection onto the spectral subspace of H associated to the Borel set  $A \subset \mathbb{R}$ . For an interval I, let  $P_{sc,I} = P_{sc}(H)P_I(H)$ , where  $P_{sc}(H)$  denotes the projection onto the singular continuous spectrum of H. Let  $\alpha, \beta \geq 1$  and  $g \in C_c^{\infty}(\mathbb{R}^+)$ . By Proposition 2.1.3 we have  $W_{\pm}W_{\pm}^* = P_{ac}(H)$  and  $W_{\pm}^*W_{\pm} = \mathrm{Id}$ . So we find

$$P_{sc}(H)W_{\pm} = P_{sc}(H)W_{\pm}W_{\pm}^*W_{\pm} = P_{sc}(H)P_{ac}(H)W_{\pm} = 0.$$

By an adjoint calculation we have  $W_{\pm}^* P_{sc}(H) = 0$ . The operators  $P_{\pm} g(H_0) (W_{\pm}^* - \mathrm{Id}) R(-i)^{\alpha}$  are compact. Thus we find

$$g(H_0)(H+i)^{-\alpha}P_{sc,I} = g(H_0)R(-i)^{\alpha}P_{sc}(H)P_I(H)$$

$$= (P_+ + P_-)g(H_0)R(-i)^{\alpha}P_{sc}(H)P_I(H)$$

$$= -(P_+ + P_-)g(H_0)(W_{\pm}^* - \operatorname{Id})R(-i)^{\alpha}P_{sc}(H)P_I(H),$$

so that  $g(H_0)R(-i)^{\alpha}P_{sc,I}$  is compact. Define the function  $\tilde{g}$  by  $\tilde{g}(\lambda) = g(\lambda)(\lambda + i)^{-1}$  and choose g such that  $\tilde{g} = 1$  on I. Then  $P_{sc,I}$  is a compact projection, so is finite rank and hence is zero.

We next show that  $W_{-}$  is complete, with the proof for  $W_{+}$  being similar. Suppose that  $f \in \mathcal{H}_{ac}(H)$  with  $f \in \text{Range}(W_{-})^{\perp}$ . Without loss of generality we may suppose that

there exists  $g \in C_c^{\infty}(\mathbb{R}^+)$  with g(H)f = f. Then the vectors  $f_t = e^{-itH}f$  converge weakly to 0 as  $t \to \infty$ . For  $\alpha \ge 1$  we thus have

$$||f||^2 = \langle f_t, f_t \rangle = \langle R(-i)^{\alpha} f_t, R(-i)^{-\alpha} f_t \rangle,$$

since  $f \in \text{Dom}(R(-i)^{-\alpha})$ . The compactness of  $R(-i)^{\alpha}(W_{\mp} - \text{Id})g(H_0)P_{\pm}$  and Theorem 2.3.2 then give

$$R(-i)^{\alpha} f_t - [R(-i)^{\alpha} W_+ g(H_0) P_- f_t + R(-i)^{\alpha} W_- g(H_0) P_+ f_t] \to 0$$

strongly as  $t \to \infty$ . So

$$||f||^2 = \lim_{t \to \infty} [\langle f_t, W_+ g(H_0) P_- f_t \rangle + \langle f_t, W_- g(H_0) P_+ f_t \rangle].$$

Since  $e^{-itH}$  leaves Range $(W_{-})$  invariant and  $f \in \text{Range}(W_{-})^{\perp}$  we have that the second term vanishes. As  $t \to \infty$ , the first vanishes since

$$\langle f_t, W_+ g(H_0) P_- f_t \rangle = \langle P_- g(H_0) e^{-itH_0} W_+^* f, f_t \rangle$$

and  $P_{-}g(H_0)e^{-itH_0} \to 0$  strongly as  $t \to \infty$  by an adjoint estimate to Lemma 2.3.5.

Remark 2.3.8. Theorem 2.3.7 can be proved under the weaker assumption that V satisfies Assumption 2.2.14 for some  $\rho > 1$  (see [164, Theorem 1.6.2]), however the proof method we have provided emphasises the key role of the generator of dilations in scattering theory.

**Proposition 2.3.9.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 1$ . The wave operators  $W_{\pm}(H, H_0)$  are Fredholm and the number of bound states (eigenvalues counted with multiplicity) of  $H = H_0 + V$ , denoted by N, is given by

$$\operatorname{Index}(W_+) = -N.$$

Proof. Note first that by Theorem 2.2.18 the projection  $P_p$  onto the eigenspace of H is finite rank. If  $f \in \text{Ker}(W_{\pm})$  then the relation  $W_{\pm}^*W_{\pm} = \text{Id}$  shows f = 0. Suppose  $f \in \text{Ker}(W_{\pm}^*)$ . By Theorem 2.3.7 we have that H has no singular continuous spectrum and thus  $P_{ac}(H) = \text{Id} - P_p(H)$ . The relation  $W_{\pm}W_{\pm}^* = P_{ac}(H) = \text{Id} - P_p(H)$  then shows  $P_p f = f$  and thus f is an eigenfunction for H. Conversely, if f is an eigenfunction for H we have  $W_{\pm}^*f = 0$  and thus  $f \in \text{Ker}(W_{\pm}^*)$ , so that  $\text{Ker}(W_{\pm}^*) = P_p \mathcal{H}$ . Hence  $W_{\pm}$  are Fredholm operators and we can compute

$$\operatorname{Index}(W_{\pm}) = \operatorname{Dim}(\operatorname{Ker}(W_{\pm})) - \operatorname{Dim}(\operatorname{Ker}(W_{+}^{*})) = 0 - N = -N.$$

# 2.4 Spectral properties of the perturbed Hamiltonian

The existence and asymptotic completeness of the wave operators proved in the previous section shows that  $\sigma_{ac}(H) = \sigma_{ac}(H_0) = [0, \infty)$ . In this section we first describe the point spectrum of the perturbed Hamiltonian. Then we describe the limiting absorption principle for boundary values of the resolvent and use this to define a family of generalised eigenfunctions for H. The limiting absorption principle also allows us to obtain explicit stationary formulas for the wave and scattering operators in terms of generalised eigenfunctions.

We shall regularly use the well known fact [110, p. 44] that a function in  $H^{s,0}$  with  $s > \frac{1}{2}$  has a trace (in the sense of restriction) on any smooth (n-1)-dimensional manifold embedded in  $\mathbb{R}^n$ .

**Theorem 2.4.1.** Let  $\Omega$  be a smooth compact (n-1)-dimensional manifold embedded in  $\mathbb{R}^n$  and let  $d\omega$  be the measure induced on  $\Omega$  by Lebesgue measure. Then for any  $s > \frac{1}{2}$  there exists a bounded linear map  $\tau: H^{s,0} \to L^2(\Omega)$  such that  $\tau f = f|_{\Omega}$  for  $u \in H^{s,0} \cap C(\mathbb{R}^n)$ .

In particular, for  $\lambda > 0$  and  $s > \frac{1}{2}$  we shall often let  $\Omega = \mathbb{S}^{n-1}$  and denote for  $\lambda > 0$  the trace operator by  $\gamma(\lambda^{\frac{1}{2}}): H^{s,0} \to L^2(\mathbb{S}^{n-1})$ , whose action on  $f \in H^{s,0}$  is given by

$$[\gamma(\lambda^{\frac{1}{2}})f](\omega) = f(\lambda^{\frac{1}{2}}\omega). \tag{2.29}$$

## 2.4.1 The limiting absorption principle

For  $z \notin \sigma(H)$  we let  $R(z) = (H-z)^{-1}$ . In this section we show in what sense R(z) assumes boundary values along the positive real axis, as limits of values from the upper and lower half plane. Since  $(0, \infty) \subset \sigma(H)$  we see that such limits cannot exist in  $\mathcal{B}(\mathcal{H})$ , however we will see that when considered in weaker topologies (that of  $\mathcal{B}(H^{0,t}, H^{0,-t})$ ) for  $t > \frac{1}{2}$ ) such limits do exist.

This fact is known as the limiting absorption principle and has its origins in the work of Ignatowski [79], who considered the propagation of electromagnetic waves in a wire, and Friedrichs [67] who established the existence of related limits to construct some explicit wave operators. The terminology limiting absorption principle was introduced by Svešnikov [156, 157]. The limiting absorption principle was used by Kato [92] and Kuroda [102, 103] and established more generally in the work of Vaĭnberg [159]. Proofs of the limiting absorption principle can be found in many texts on scattering theory, however the proof presented here follows closely those of Agmon [3, Section 4], Kuroda [106, Chapter 5] and Yafaev[164, Chapter 6].

The proof is rather long and so will be broken into a number of intermediate steps, the first of which is to prove the limiting absorption principle for  $H_0$ . We let  $R_0(z) = (H_0 - z)^{-1}$ 

and consider  $R_0(z)$  as an analytic operator valued function on  $\mathbb{C} \setminus [0, \infty)$  with values in  $\mathcal{B}(H^{0,t}, H^{2,-t})$  for any  $t > \frac{1}{2}$ .

**Lemma 2.4.2.** Let  $t > \frac{1}{2}$  and  $f, g \in H^{0,t}$ . Then for any  $\lambda \in (0, \infty)$  the limits

$$\langle R_0(\lambda \pm i0)f, g \rangle := \lim_{\varepsilon \to 0} \langle R_0(\lambda \pm i\varepsilon)f, g \rangle$$
 (2.30)

exist. Furthermore, for any  $\lambda \in (0, \infty)$  and  $f \in H^{0,t}$  the limit

$$R_0(\lambda \pm i0)f := \lim_{\varepsilon \to 0} R_0(\lambda \pm i\varepsilon)f, \tag{2.31}$$

taken in the weak topology of  $H^{2,-t}$ , exists in  $H^{2,-t}$ .

*Proof.* Let  $t > \frac{1}{2}$  and  $f, g \in H^{0,t}$ . We show that the function  $F : \mathbb{C} \setminus [0, \infty) \to \mathbb{C}$  given by  $F(z) = \langle R_0(z)f, g \rangle$  has continuous boundary values on both sides of  $(0, \infty)$  (that is for  $\lambda \in (0, \infty)$  the limits  $\lim_{\varepsilon \to 0} F(\lambda \pm i\varepsilon)$  exist and are continuous in  $\lambda$ ). By a version of the elliptic estimate [3, Theorem A.1] we have

$$||R_0(z)f||_{H^{2,-t}} \le C ||f||_{H^{0,t}} \tag{2.32}$$

for all  $f \in H^{0,t}$  and  $z \in \mathbb{C} \setminus \mathbb{R}^+$  such that  $K^{-1} \leq |z| \leq K$  for fixed K > 1. For  $g \in H^{2,-t}$  we obtain

$$|F(z)| = |\langle R_0(z)f, g \rangle| \le ||R_0(z)f||_{H^{2,-t}} ||g||_{H^{-2,t}} \le C ||f||_{H^{0,t}} ||g||_{H^{0,t}}$$
(2.33)

for another constant C and  $K^{-1} \leq |z| \leq K$ . The estimate (2.33) shows that it suffices to prove that F assumes continuous boundary values on both sides of  $(0, \infty)$  on a dense subset of  $H^{0,t}$ . So for  $f, g \in C_c^{\infty}(\mathbb{R}^n)$  we have by Equation (2.16) that

$$F(z) = \langle R_0(z)f, g \rangle = \langle \mathcal{F}R_0(z)f, \mathcal{F}g \rangle = \int_{\mathbb{R}^n} \frac{\overline{[\mathcal{F}f](\xi)}[\mathcal{F}g](\xi)}{|\xi|^2 - z} \,\mathrm{d}\xi$$
$$= \frac{1}{2} \int_0^\infty \int_{\mathbb{S}^{n-1}} \frac{k^{\frac{n-2}{2}}}{k - z} \overline{[\mathcal{F}f](k^{\frac{1}{2}}\omega)} [\mathcal{F}g](k^{\frac{1}{2}}\omega) \,\mathrm{d}\omega \,\mathrm{d}k.$$

This is a Cauchy type integral, which in the limit can be evaluated using the Plemelj formula [131, Chapter 14] (see also [161, Theorem 9.8]) to obtain

$$\lim_{\varepsilon \to 0} \langle R_0(\lambda \pm i\varepsilon) f, g \rangle \qquad (2.34)$$

$$= \pm \frac{\pi i \lambda^{\frac{n-2}{2}}}{2} \int_{\mathbb{S}^{n-1}} \overline{[\mathcal{F}f](\lambda^{\frac{1}{2}}\omega)} [\mathcal{F}g](\lambda^{\frac{1}{2}}\omega) d\omega + \text{P.V.} \left( \int_{\mathbb{R}^n} \overline{\frac{[\mathcal{F}f](\xi)}{|\xi|^2 - \lambda}} d\xi \right),$$

where the last integral is singular and taken in the principal value sense. Applying [121, Theorem 20.1] (see also [17, Theorem 2]) to Equation (2.34) shows that F admits continu-

ous boundary values on both sides of  $(0, \infty)$ . The existence of the limits in Equation (2.34) shows that for  $f \in H^{0,t}$  the limit

$$\lim_{\varepsilon \to 0} R_0(\lambda \pm i\varepsilon) f, \tag{2.35}$$

taken in the weak topology of  $H^{0,-t}$ , exists in  $H^{0,-t}$ . Then  $R_0(z)f$  is bounded near  $\lambda$  when considered as a function with values in  $H^{2,-t}$  by the estimate (2.32). The weak compactness of the unit ball in  $H^{2,-t}$  then shows that the limit (2.35) exists also in  $H^{2,-t}$  and is by definition given by the function  $R_0(\lambda \pm i0)f$  of Equation (2.31).

Corollary 2.4.3. For any  $t > \frac{1}{2}$  and  $f \in H^{0,t}$  we have the identity

$$\operatorname{Im}\left(\langle R_0(\lambda \pm i0)f, f \rangle\right) = \pm \frac{\pi \lambda^{\frac{1}{2}}}{2} \int_{\mathbb{S}^{n-1}} |[\gamma(\lambda^{\frac{1}{2}})\mathcal{F}f](\omega)|^2 d\omega, \tag{2.36}$$

where  $\gamma$  is defined by Equation (2.29).

*Proof.* The result follows immediately by taking f = g in Equation (2.34) and then taking the imaginary part.

Corollary 2.4.4. For any  $t > \frac{1}{2}$ ,  $f \in H^{0,t}$  and  $\lambda \in (0, \infty)$  the function  $\psi = R_0(\lambda \pm i0)f$  satisfies the differential equation

$$(H_0 - \lambda)\psi = f \tag{2.37}$$

in the sense of distributions.

*Proof.* For any  $\varphi \in C_c^{\infty}(\mathbb{R}^n)$  and  $z \notin \mathbb{R}$  we have the relation

$$\langle R_0(z) f, (H_0 - \overline{z}) \varphi \rangle = \langle f, \varphi \rangle.$$

By Lemma 2.4.2 we thus have for  $\lambda \in (0, \infty)$  the equalities

$$\langle f, \varphi \rangle = \lim_{\varepsilon \to 0} \langle R_0(\lambda \pm i\varepsilon) f, (H_0 - \lambda) \varphi \rangle = \langle \psi, (H_0 - \lambda) \varphi \rangle = \langle (H_0 - \lambda) \psi, \varphi \rangle.$$

Since this holds for all  $\varphi \in C_c^{\infty}(\mathbb{R}^n)$ , we find  $(H_0 - \lambda)\psi = f$  in the sense of distributions.  $\square$ 

We will now show that the weak limits (2.31) exist in a stronger sense.

**Theorem 2.4.5** (Limiting absorption principle for  $H_0$ ). Consider the resolvent operator  $R_0(z) = (H_0 - z)^{-1}$ . Then for any  $\lambda \in (0, \infty)$  and  $t > \frac{1}{2}$  the limits

$$\lim_{\varepsilon \to 0} R_0(\lambda \pm i\varepsilon) =: R_0(\lambda \pm i0) \tag{2.38}$$

exist in the uniform operator topology of  $\mathcal{B}(H^{0,t},H^{2,-t})$ 

*Proof.* Let  $\mathbb{C}_{\pm} = \{z \in \mathbb{C} : \pm \operatorname{Im}(z) > 0\}$  and  $\overline{\mathbb{C}_{\pm}} = \mathbb{C}_{\pm} \cup (0, \infty)$ . Define the operator-valued functions  $R_0^{\pm} : \overline{\mathbb{C}_{\pm}} \to \mathcal{B}(H^{0,t}, H^{2,-t})$  by

$$R_0^{\pm}(z) = \begin{cases} R_0(z), & \text{if } \pm \text{Im}(z) > 0, \\ R_0(\lambda \pm i0), & \text{if } z = \lambda \in (0, \infty). \end{cases}$$
 (2.39)

From Lemma 2.4.2 we have that the  $R_0^{\pm}$  define weak operator topology continuous operatorvalued functions on  $\overline{\mathbb{C}_{\pm}}$ . We now show that  $R_0^{\pm}$  are actually continuous on  $\overline{\mathbb{C}_{\pm}}$  in the uniform operator topology of  $\mathcal{B}(H^{0,t},H^{2,-t})$ . Since for  $u \in H^{2-,t}$  we have that  $||u||_{H^{2,-t}}$ is equivalent, via the elliptic estimate [76, p. 283], to  $||\cdot||_{eq} = ||\cdot||_{H^{0,-t}} + ||H_0\cdot||_{H^{0,-t}}$  it suffices to show that both  $R_0^{\pm}$  and  $H_0R_0^{\pm}$  are continuous in the uniform operator topology of  $\mathcal{B}(H^{0,t},H^{0,-t})$ . Since  $H_0R_0^{\pm}(z) = \mathrm{Id} + zR_0^{\pm}(z)$  it further suffices to show that the operators  $R_0^{\pm}$  are continuous in the uniform operator topology of  $\mathcal{B}(H^{0,t},H^{0,-t})$ . We will prove this in several steps.

First we show that  $R_0^{\pm}$  are continuous on  $\overline{\mathbb{C}_{\pm}}$  in the strong topology of  $\mathcal{B}(H^{0,t},H^{0,-t})$ . Note that the inclusion  $H^{2,-m} \hookrightarrow H^{0,-t}$  is compact for any m < t (a consequence of Rellich's compactness theorem, see Theorem 2.2.5 (b)), and that for  $f \in H^{0,t}$  we have  $R_0(z)f \in H^{2,-m}$  for any  $m > \frac{1}{2}$ . Then for  $z_0 \in \mathbb{C}_{\pm}$  we have

$$R_0^{\pm}(z_0)f := \lim_{z \to z_0} R_0^{\pm}(z)f \tag{2.40}$$

exists in  $H^{0,-t}$  by the continuity of  $R_0(z) \in \mathcal{B}(H^{0,t}, H^{2,-t})$  and so in  $\mathcal{B}(H^{0,t}, H^{0,-t})$  by compactness of the inclusion map  $H^{2,-t} \hookrightarrow H^{0,-t}$ . For  $z_0 \in (0,\infty)$  we use that the limit (2.31) exists to see that the limit (2.40) holds in  $H^{2,-t}$  and thus  $R_0(z_0)$  defines an operator in  $\mathcal{B}(H^{0,t}, H^{2,-t})$  and so also in  $\mathcal{B}(H^{0,t}, H^{0,-t})$  by compactness of the inclusion  $H^{2,-t} \hookrightarrow H^{0,-t}$ . Thus we have shown that  $R_0^{\pm}$  are continuous on  $\overline{\mathbb{C}_{\pm}}$  in the strong topology of  $\mathcal{B}(H^{0,t}, H^{0,-t})$ .

Let  $(z_k) \subset \mathbb{C}_{\pm}$  and  $(f_j) \subset H^{0,t}$  be sequences with  $z_k \to z_0 \in \mathbb{C}_{\pm}$  and  $\lim f_j = f \in H^{0,t}$  in the weak topology of  $H^{0,t}$ . Then we can compute for any  $g \in H^{0,t}$  that

$$\lim_{i \to \infty} \langle R_0^{\mp}(z_j) f_j, g \rangle = \lim_{i \to \infty} \langle f_j, R_0^{\pm}(\overline{z_j}) g \rangle = \langle f, R_0^{\pm}(\overline{z_0}) g \rangle = \langle R_0^{\mp}(z_0) f, g \rangle.$$

Hence we find the existence of the limit

$$\lim_{j \to \infty} R_0^{\pm}(z_j) f_j = R_0^{\pm}(z_0) f \tag{2.41}$$

in the weak topology of  $H^{0,-t}$ , considered as the dual of  $H^{0,t}$ . The estimate (2.32) shows that the sequence  $(R_0^{\pm}(z_j)f_j)$  is bounded in  $H^{2,-m}$  for any  $m > \frac{1}{2}$ . The compactness of the inclusion  $H^{2,-m} \hookrightarrow H^{0,-t}$  for m < t then shows that the existence of the limit (2.41)

gives the existence of the limit

$$\lim_{j \to \infty} R_0^{\pm}(z_j) f_j = R_0^{\pm}(z_0) f \tag{2.42}$$

in the strong topology of  $H^{0,-t}$  also. We now show that it follows from Equation (2.42) that  $R_0^{\pm}$  is continuous on  $\overline{\mathbb{C}_{\pm}}$  in the uniform operator topology of  $\mathcal{B}(H^{0,t},H^{0,-t})$ . Suppose for contradiction that  $R_0^{\pm}$  is not continuous on  $\overline{\mathbb{C}_{\pm}}$  in the uniform operator topology of  $\mathcal{B}(H^{0,t},H^{0,-t})$ . Then there exists a sequence  $(z_j)\subset\overline{\mathbb{C}_{\pm}}$  with  $z_j\to z_0\in\overline{\mathbb{C}_{\pm}}$  and a sequence  $(f_j)\subset H^{0,t}$  with  $||f_j||_{H^{0,t}}=1$  such that

$$\lim_{j \to \infty} \left| \left| \left( R_0^{\pm}(z_j) - R_0^{\pm}(z_0) \right) f_j \right| \right|_{H^{0,-t}} > 0. \tag{2.43}$$

We can assume (after passing to a subsequence if necessary) the existence of the weak limit  $\lim_{j\to\infty} f_j =: f \in H^{0,t}$  in the weak topology of  $H^{0,t}$ . Then by Equation (2.42) we have

$$\lim_{j \to \infty} R_0^{\pm}(z_j) f_j = R_0^{\pm}(z_0) f = \lim_{j \to \infty} R_0^{\pm}(z_0) f_j,$$

in  $H^{0,-t}$  (with the limits in the strong topology), which contradicts the inequality (2.43).

**Definition 2.4.6.** A function  $\psi \in H^{2,0}_{loc}$  will be called k-outgoing (for '+') or k-incoming (for '-') if for k > 0 the relation

$$\psi = R_0(k^2 \pm i0)f \tag{2.44}$$

holds for some  $f \in H^{0,t}$  and  $t > \frac{1}{2}$ . Here the subscript loc denotes that  $\psi$  is locally integrable, that is  $\psi \in H^{2,0}(K)$  for any compact  $K \subset \mathbb{R}^n$ ,

**Lemma 2.4.7.** Let  $\psi \in H^{2,0}_{loc}$  be a k-outgoing (or k-incoming) function satisfying the differential equation

$$(H_0 + V)\psi = k^2\psi \tag{2.45}$$

for some V satisfying Assumption 2.2.14. Then  $\psi \in H^{0,t}$  for all  $t \in \mathbb{R}$ .

*Proof.* We shall provide a proof for  $\psi$  outgoing, the proof for  $\psi$  incoming being similar. Since  $\psi$  is outgoing there exists  $t_0 > \frac{1}{2}$  and  $f \in H^{0,t_0}$  such that  $\psi = R_0(k^2 + i0)f$ . Then we compute

$$(H_0 - k^2)\psi = (H_0 - k^2) \left( \lim_{\varepsilon \to 0} R_0(k^2 + i\varepsilon) f \right) = \lim_{\varepsilon \to 0} (H_0 - k^2) R_0(k^2 + i\varepsilon) f$$
$$= \lim_{\varepsilon \to 0} \left( f + i\varepsilon R_0(k^2 + i\varepsilon) f \right) = f,$$

by Theorem 2.4.5. By Equation (2.45) we have  $f = -V\psi$ . By Corollary 2.4.3 and the fact that V is real-valued, evaluating  $\mathcal{F}f$  on the sphere of radius k we find

$$k^{n-1} \int_{\mathbb{S}^{n-1}} |[\mathcal{F}f](k\omega)|^2 d\omega = \frac{2k}{\pi} \operatorname{Im}\left(\langle R_0(k^2 + i0)f, f \rangle\right) = -\frac{2k}{\pi} \operatorname{Im}\left(\langle \psi, V\psi \rangle\right) = 0. \quad (2.46)$$

This shows that the restriction of  $\mathcal{F}f$  to the sphere of radius k vanishes and so by [3, Theorem 3.2] we have the function  $h: \mathbb{R}^n \to \mathbb{C}$  defined by  $h(\xi) = [\mathcal{F}f](\xi)(|\xi|^2 - k^2)^{-1}$  is such that  $h \in L^1_{loc}(\mathbb{R}^n)$  (that is  $h \in L^1(K)$  for all  $K \subset \mathbb{R}^n$  compact). Define  $\psi \in H^{0,-t_0}$  by  $\psi = \lim_{\varepsilon \to 0} R_0(k^2 + i\varepsilon)f$  and let  $g \in \mathcal{S}(\mathbb{R}^n)$ . We compute that

$$\langle \psi, g \rangle = \lim_{\varepsilon \to 0} \langle R_0(k^2 + i\varepsilon) f, g \rangle = \lim_{\varepsilon \to 0} \int_{\mathbb{R}^n} \frac{\overline{[\mathcal{F}f](\xi)}[\mathcal{F}g](\xi)}{|\xi|^2 - k^2} \frac{|\xi|^2 - k^2}{|\xi|^2 - k^2 - i\varepsilon} \, \mathrm{d}\xi$$
$$= \int_{\mathbb{R}^n} \frac{\overline{[\mathcal{F}f](\xi)}}{|\xi|^2 - k^2} [\mathcal{F}g](\xi) \, \mathrm{d}\xi, \tag{2.47}$$

where we have used that  $\psi \in H^{0,-t_0}$ , the unitarity of  $\mathcal{F}$  and Lebesgue's dominated convergence theorem. Thus  $\mathcal{F}\psi \in L^1_{loc}(\mathbb{R}^n)$  and since Equation (2.47) holds for all  $g \in \mathcal{S}(\mathbb{R}^n)$ , we find  $\mathcal{F}\psi = h$  in the sense of distributions. Then by [3, Theorem 3.3] we have  $\psi \in H^{2,t} \subset H^{0,t}$  for all  $t \in \mathbb{R}$ , which proves the result.

We can now state and prove the limiting absorption principle for general Schrödinger operators. For  $z \in \mathbb{C} \setminus \sigma(H)$  we let  $R(z) = (H - z)^{-1}$  considered as an analytic operator valued function on  $\mathbb{C} \setminus \sigma(H)$  with values in  $\mathcal{B}(H^{0,t}, H^{2,-t})$  for any  $t > \frac{1}{2}$ .

**Theorem 2.4.8** (Limiting absorption principle for H). Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 2$  and consider the operator  $R(z) = (H - z)^{-1}$ . Then for any  $\lambda \in (0, \infty)$  the limits

$$R(\lambda \pm i0) := \lim_{\varepsilon \to 0} R(\lambda \pm i\varepsilon) \tag{2.48}$$

exist in the uniform operator topology of  $\mathcal{B}(H^{0,t},H^{2,-t})$  for any  $t>\frac{1}{2}$ .

Proof. We prove the theorem for  $R(\lambda+i0)$  since the proof for  $R(\lambda-i0)$  is similar. Without loss of generality we suppose that  $t \in \left(\frac{1}{2}, \frac{1}{2} + \varepsilon\right)$  for some  $\varepsilon > 0$ . We choose  $\varepsilon$  sufficiently small so that V defines a compact operator from  $H^{2,-\frac{1}{2}-\varepsilon}$  to  $H^{0,\frac{1}{2}+\varepsilon}$  (possible by Rellich's compactness theorem since  $|V(x)| \leq C(1+|x|)^{-\rho}$  for some  $\rho > 1$  and almost all  $x \in \mathbb{R}^n$ ).

For any  $z \in \overline{\mathbb{C}_+}$  we define  $Q(z) \in \mathcal{B}(H^{2,-t},H^{2,-t})$  by

$$Q(z)\psi = R_0(z+i0)V\psi. \tag{2.49}$$

From Theorem 2.4.5 and the compactness of V as an operator from  $H^{2,-t}$  to  $H^{0,t}$  we have that Q(z) defines a compact operator for every  $z \in \overline{\mathbb{C}_+}$ , and that the operator-valued function Q is continuous on  $\overline{\mathbb{C}_+}$  in the uniform operator topology of  $\mathcal{B}(H^{2,-t},H^{2,-t})$ .

For Im(z) > 0 we can use the resolvent equation

$$R(z) = R_0(z) - R_0(z)VR(z)$$
(2.50)

and the definition of Q(z) to see that for any  $f \in \mathcal{H}$  and  $\psi = R(z)f \in H^{2,0}$  we have

$$(\mathrm{Id} + Q(z))\psi = R_0(z)f.$$
 (2.51)

We see, by varying f, that  $H^{2,0} \subset \text{Range}(\text{Id} + Q(z)) \subset H^{2,-t}$ . Since the closure of the domain of the Laplacian with respect to the weight  $\mathbb{R}^n \ni x \mapsto (1+|x|^2)^{-\frac{t}{2}}$  is  $H^{2,-t}$  we find  $\overline{\text{Range}(\text{Id} + Q(z))} = H^{2,-t}$ . Standard Fredholm theory (see [106, Proposition 5.2.1]) then shows that the inverse  $(\text{Id} + Q(z))^{-1}$  exists in  $\mathcal{B}(H^{2,-t}, H^{2,-t})$ .

Now we let  $z = \lambda \in (0, \infty)$ . Then the analytic Fredholm alternative says that  $\mathrm{Id} + Q(\lambda)$  is invertible if and only if -1 is not an eigenvalue of  $Q(\lambda)$ . We now show that for  $\lambda > 0$ , -1 is an eigenvalue of  $Q(\lambda)$  if and only if  $\lambda$  is an eigenvalue of H.

So we suppose that -1 is an eigenvalue of  $Q(\lambda)$  and that  $\psi \in H^{2,-t}$  is a corresponding eigenfunction. The definition of Q shows that  $\psi = -R_0(\lambda + i0)V\psi$  which implies that  $\psi$  is a  $\lambda^{\frac{1}{2}}$ -outgoing solution of the equation  $(H_0 + V)\psi = \lambda\psi$ . By Lemma 2.4.7 we see that  $\psi \in H^{0,t_0}$  for all  $t_0 \in \mathbb{R}$ . So  $\mathcal{F}f \in C^{\infty}(\mathbb{R}^n)$  and hence f has rapid decay. Thus  $f \in \text{Dom}(H_0)$  and  $\lambda$  is an eigenvalue of H, contradicting the fact that H has no positive eigenvalues.

Conversely, we suppose that  $\lambda > 0$  is an eigenvalue of H with corresponding eigenfunction  $\psi \in \text{Dom}(H)$ . Using Equation (2.50) we find that  $\psi + R_0(z)V\psi = (\lambda - z)R_0(z)\psi$  for  $z \in \mathbb{C}_+$ . Taking the limit as  $z \to \lambda$  we find  $\psi + R_0^+(\lambda)V\psi = 0$  and thus -1 is an eigenvalue of  $Q(\lambda)$ . Since H has no positive eigenvalues we find that  $(\text{Id} + Q(z))^{-1}$  exists for all  $z \in \overline{\mathbb{C}_+}$ .

Since Q is continuous on  $\overline{\mathbb{C}_+}$  in the uniform operator topology of  $\mathcal{B}(H^{2,-t},H^{2,-t})$ , it follows that the operator-valued function  $(\mathrm{Id}+Q(z))^{-1}$  is also continuous on  $\overline{\mathbb{C}_+}$  in the uniform operator topology of  $\mathcal{B}(H^{2,-t},H^{2,-t})$ . The resolvent equation (2.50) and the definition of Q give us that

$$R(z) = (\mathrm{Id} + Q(z))^{-1} R_0(z)$$
(2.52)

for Im(z) > 0. Using the continuity properties of  $(\text{Id} + Q(z))^{-1}$  and  $R_0(z)$  it follows that the limits

$$\lim_{\varepsilon \to 0} R(\lambda + i\varepsilon) = (\mathrm{Id} + Q(\lambda))^{-1} R_0(\lambda + i0)$$
(2.53)

exist in the uniform operator topology of  $\mathcal{B}(H^{0,t},H^{2,-t})$ .

Corollary 2.4.9. For any  $t > \frac{1}{2}$  and  $f \in H^{0,t}$  we have

$$R(\lambda \pm i0)f = R_0(\lambda \pm i0)f - R_0(\lambda \pm i0)VR(\lambda \pm i0)f. \tag{2.54}$$

In particular,  $\psi_+ = R(\lambda + i0)f$  is a  $\lambda^{\frac{1}{2}}$ -outgoing solution and  $\psi_- = R(\lambda - i0)f$  is a  $\lambda^{\frac{1}{2}}$ -incoming solution of the differential equation

$$(H_0 + V)\psi = f. \tag{2.55}$$

*Proof.* Using the continuity properties of R(z) established in Theorem 2.4.8 we obtain Equation (2.54) immediately from Equation (2.50). The fact that  $\psi_+ = R(\lambda + i0)f$  is a  $\lambda^{\frac{1}{2}}$ -outgoing solution was established and used in the proof of Theorem 2.4.8.

We conclude this section by using similar methods to determine the high energy behaviour of the resolvent. The result is the following.

**Lemma 2.4.10.** Suppose that  $\rho > \frac{n+1}{2}$  in Assumption (2.2.14) and define the complex domain  $C_+ = \{z \in \mathbb{C} : \operatorname{Re}(z) \geq 1 \text{ and } \operatorname{Im}(z) > 0\}$  and  $t \in (\frac{1}{2}, \rho - \frac{1}{2})$ . Then  $VR_0(z)$  and  $(\operatorname{Id} + VR_0(z))^{-1}$  can be extended to continuous and uniformly bounded functions from  $C_+$  to  $\mathcal{B}(H^{0,t}, H^{0,t})$ . Similarly,  $R_0(z)V$  and  $(\operatorname{Id} + R_0(z)V)^{-1}$  can be extended to continuous and uniformly bounded functions from  $C_+$  to  $\mathcal{B}(H^{0,-t}, H^{0,-t})$ .

*Proof.* Since for any  $\rho > \frac{n+1}{2}$  and  $t \in (\frac{1}{2}, \rho - \frac{1}{2})$  we have  $R_0(z) \in \mathcal{B}(H^{0,t}, H^{0,t-\rho})$  is continuous in  $z \in C_+$  by [85, proof of Lemma 3.1], and since  $V \in \mathcal{B}(H^{0,t-\rho}, H^{0,t})$  by assumption, we find that  $VR_0(z) \in \mathcal{B}(H^{0,t}, H^{0,t})$  for any  $t \in (\frac{1}{2}, \rho - \frac{1}{2})$ .

Since  $VR_0(z)$  is compact and has no eigenvalue -1 for  $z \in C_+$  (see the discussion in the proof of Theorem 2.4.8), the operator  $(\mathrm{Id} + VR_0(z))^{-1} \in \mathcal{B}(H^{0,t}, H^{0,t})$  exists and is continuous in  $z \in C_+$ . By [120, Theorem 1] we have that  $VR_0(z) \to 0$  as  $|z| \to \infty$  in  $C_+$  in the norm of  $\mathcal{B}(H^{0,t}, H^{0,t})$  for any  $t > \frac{1}{2}$ , so that the operator  $(\mathrm{Id} + VR_0(z))^{-1}$  is uniformly bounded. A similar computation (or duality) can be used to prove the second claim.

### 2.4.2 Generalised eigenfunctions and stationary wave operators

In this section we use the spectral properties of  $H_0$  in conjunction with the limiting absorption principle to define a family of 'generalised eigenfunctions' for the operator H.

**Definition 2.4.11.** For each  $\lambda > 0$  and  $\omega \in \mathbb{S}^{n-1}$  the function  $\psi_0(\cdot, \omega, \lambda) : \mathbb{R}^n \to \mathbb{C}$  defined by

$$\psi_0(x,\omega,\lambda) = e^{i\lambda^{\frac{1}{2}\langle x,\omega\rangle}}$$

satisfies the relation  $[H_0\psi_0(\cdot,\omega,\lambda)](x) = \lambda\psi_0(x,\omega,\lambda)$  and thus defines a 'generalised eigenfunction' for the continuous spectrum of  $H_0$ .

We now demonstrate how the generalised eigenfunctions give rise to Stone's formula, relating the spectral measure for  $H_0$  to the limiting absorption principle.

**Theorem 2.4.12.** For  $\lambda > 0$  we have for  $t > \frac{1}{2}$ , the operator  $B(\lambda) \in \mathcal{B}(H^{0,t}, H^{0,-t})$  by  $B(\lambda) = R_0(\lambda + i0) - R_0(\lambda - i0)$  is an integral operator with kernel

$$R_0(x, y, \lambda + i0) - R_0(x, y, \lambda - i0) = \frac{i}{2} \lambda^{\frac{n-2}{2}} (2\pi)^{-(n-1)} \int_{\mathbb{S}^{n-1}} e^{i\lambda^{\frac{1}{2}} \langle \omega, x - y \rangle} d\omega, \qquad (2.56)$$

for  $x, y \in \mathbb{R}^n$ .

*Proof.* Fix  $\lambda > 0$ . From Corollary 2.4.3 we have for any  $f \in H^{0,t}$  with  $t > \frac{1}{2}$  the relation

$$\frac{1}{2i}\langle (R_0(\lambda+i0)-R_0(\lambda-i0))f,f\rangle = \frac{\pi\lambda^{\frac{n-2}{2}}}{4}\int_{\mathbb{S}^{n-1}}|[\gamma(\lambda^{\frac{1}{2}})\mathcal{F}f](\omega)|^2 d\omega.$$

So we compute that

$$\int_{\mathbb{S}^{n-1}} |[\gamma(\lambda^{\frac{1}{2}})\mathcal{F}f](\omega)|^2 d\omega$$

$$= (2\pi)^{-n} \int_{\mathbb{S}^{n-1}} \left( \int_{\mathbb{R}^n} \overline{f(x)} e^{i\lambda^{\frac{1}{2}}\langle x,\omega\rangle} dx \right) \left( \int_{\mathbb{R}^n} f(y) e^{-i\lambda^{\frac{1}{2}}\langle y,\omega\rangle} dy \right) d\omega$$

$$= (2\pi)^{-n} \int_{\mathbb{R}^n} \int_{\mathbb{R}^n} \overline{f(x)} f(y) \left( \int_{\mathbb{S}^{n-1}} e^{-i\lambda^{\frac{1}{2}}\langle y-x,\omega\rangle} d\omega \right) dy dx.$$

We next note that

$$\langle (R_0(\lambda + i0) - R_0(\lambda - i0)) f, f \rangle = \int_{\mathbb{R}^n} \overline{[(R_0(\lambda + i0) - R_0(\lambda - i0)) f](x)} f(y) dy$$
$$= \int_{\mathbb{R}^n} \int_{\mathbb{R}^n} (R_0(x, y, \lambda - i0) - R_0(x, y, \lambda + i0)) \overline{f(x)} f(y) dy dx.$$

Since  $f \in H^{0,t}$  was arbitrary, the claim follows.

Using the limiting absorption principle, one can define analogous generalised eigenfunctions for the operator H.

**Proposition 2.4.13** (Lippmann-Schwinger). Suppose that V satisfies Assumption 2.2.14 for some  $\rho > \frac{n+1}{2}$ . Then for all  $\lambda > 0$ ,  $\omega \in \mathbb{S}^{n-1}$  and  $m \in \left(\frac{n}{2}, \rho - \frac{n}{2}\right)$  there is a unique solution  $\psi_{\pm} \in H^{0,-m}$  to

$$\psi_{\pm}(x,\omega,\lambda) = \psi_0(x,\omega,\lambda) - [R_0(\lambda \pm i0)V\psi_{\pm}(\cdot,\omega,\lambda)](x), \quad x \in \mathbb{R}^n.$$
 (2.57)

This solution is given by the equation

$$\psi_{\pm}(x,\omega,\lambda) = \psi_0(x,\omega,\lambda) - [R(\lambda \pm i0)V\psi_0(\cdot,\omega,\lambda)](x). \tag{2.58}$$

Furthermore, for each fixed  $\lambda > 0$  and  $\omega \in \mathbb{S}^{n-1}$  we have

$$[H\psi_{\pm}(\cdot,\omega,\lambda)](x) = \lambda\psi_{\pm}(x,\omega,\lambda). \tag{2.59}$$

*Proof.* First note that for all  $s \in (\frac{1}{2}, \rho - \frac{n}{2})$  we have  $V\psi_0(\cdot, \omega, \lambda) \in H^{0,s}$ . So defining  $\psi_{\pm}$  by Equation (2.58) we find that  $\psi_{\pm}(\cdot, \omega, \lambda) \in H^{0,-m}$  for all  $m > \frac{n}{2}$ . Apply the operator  $\mathrm{Id} + R_0(\lambda \mp i0)V$  to both sides of Equation (2.58) in conjunction with the resolvent identity

$$(\operatorname{Id} + R_0(z)V)(\operatorname{Id} - R(z)V) = \operatorname{Id}$$
(2.60)

gives that  $\psi_{\pm}(\cdot, \omega, \lambda)$  satisfies Equation (2.57). If  $\varphi_{\pm}(\cdot, \omega, \lambda) \in H^{0,-m}$  were another solution to Equation (2.57), we apply the operator  $\mathrm{Id} - R(\lambda \mp i0)V$  to both sides of Equation (2.57) in conjunction with (the adjoint of) Equation (2.60) to see that  $\varphi_{\pm} = \psi_{\pm}$ . Equation (2.59) then follows from Corollary 2.4.9.

Equations (2.57) and (2.58) are known as the Lippmann-Schwinger equations and were first introduced in a different form in [111, 112]. Equation (2.59) shows that the  $\psi_{\pm}$  define generalised eigenfunctions of the continuous spectrum of H. We now consider the asymptotics of the generalised eigenfunctions  $\psi_{\pm}$  as  $|x| \to \infty$ . To do so requires the introduction of the following operator.

**Definition 2.4.14.** Define for  $\lambda > 0$  and  $t > \frac{1}{2}$  the operator  $\Gamma_0(\lambda) : H^{0,t} \to \mathcal{P}$  by

$$[\Gamma_0(\lambda)f](\omega) = [F_0f](\lambda,\omega) = 2^{-\frac{1}{2}}\lambda^{\frac{n-2}{4}}[\mathcal{F}f](\lambda^{\frac{1}{2}}\omega), \quad \omega \in \mathbb{S}^{n-1},$$

with  $F_0$  from Definition 2.2.7.

**Proposition 2.4.15.** For  $\lambda > 0$  and  $t > \frac{1}{2}$  the operator  $\Gamma_0(\lambda) : H^{0,t} \to \mathcal{P}$  and its formal adjoint  $\Gamma_0(\lambda)^* : \mathcal{P} \to H^{0,-t}$  are uniformly bounded on compact intervals. Moreover, the operator-valued function  $\Gamma_0(\lambda)$  is strongly continuous in  $\lambda > 0$ .

*Proof.* For  $f \in H^{0,t}$  and  $\lambda > 0$  we use the estimate (2.32) to obtain for  $\lambda$  in some compact interval the bound

$$||\Gamma_0(\lambda)f||^2 = \lim_{\varepsilon \to 0} \varepsilon ||R_0(\lambda \pm i\varepsilon)f||^2 \le C ||f||_{H^{0,t}}^2,$$

and thus  $\Gamma_0(\lambda)$  is uniformly bounded. To see that  $\Gamma_0(\lambda)$  is strongly continuous in  $\lambda > 0$  we fix  $\mu, \lambda > 0$  and compute that

$$|[(\Gamma_0(\mu) - \Gamma_0(\lambda))f](\omega)|^2 = |[\Gamma_0(\mu)f](\omega)|^2 + |[\Gamma_0(\lambda)f](\omega)|^2 - 2\operatorname{Re}([\Gamma_0(\mu)f](\omega)\overline{[\Gamma_0(\lambda)f](\omega)}).$$

So we write

$$[\Gamma_0(\mu)f](\omega)\overline{[\Gamma_0(\lambda)f](\omega)} = \frac{(\lambda\mu)^{\frac{n-2}{4}}}{2(2\pi)^n} \left( \int_{\mathbb{R}^n} e^{-i\mu^{\frac{1}{2}}\langle x,\omega\rangle} f(x) dx \right) \left( \int_{\mathbb{R}^n} e^{i\lambda^{\frac{1}{2}}\langle x,\omega\rangle} f(x) dx \right),$$

which converges to  $|[\Gamma_0(\lambda)f](\omega)|^2$  as  $\mu \to \lambda$ . Thus we find

$$||(\Gamma_0(\mu) - \Gamma_0(\lambda))f||^2 = \int_{\mathbb{S}^{n-1}} |[(\Gamma_0(\mu) - \Gamma_0(\lambda))f](\omega)|^2 d\omega \to 0$$

as  $\mu \to \lambda$ , so that  $\Gamma_0(\lambda)$  is strongly continuous in  $\lambda$ .

We can use the operator  $\Gamma_0(\lambda)$  to obtain the asymptotics of  $\psi_{\pm}$  in terms of scattering data. To do so requires the following definition.

**Definition 2.4.16.** For  $r, \lambda > 0$  we define the outgoing and incoming spherical waves by

$$w_{\pm}(r,\lambda) = r^{-\frac{n-1}{2}} e^{\pm i\lambda^{\frac{1}{2}} r \mp i\pi^{\frac{n-3}{2}}}.$$
 (2.61)

**Theorem 2.4.17.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > \frac{n+1}{2}$ . Then for fixed  $\omega \in \mathbb{S}^{n-1}$  and  $\lambda > 0$ , with  $\psi_{\pm}(\cdot, \omega, \lambda)$  as in Proposition 2.4.13, we have as  $|x| \to \infty$  that

$$\psi_{+}(x,\omega,\lambda) = \psi_{0}(x,\omega,\lambda) + a_{+}(\hat{x},\omega,\lambda)w_{\pm}(|x|,\lambda) + o_{av}(|x|^{-\frac{n+1}{2}}), \tag{2.62}$$

where for  $x \neq 0$  we have defined  $\hat{x} = |x|^{-1}x$ . The coefficient  $a_{\pm}$  can be recovered by the formula

$$a_{\pm}(\theta,\omega,\lambda) = -\pi^{\frac{1}{2}}\lambda^{-\frac{1}{4}}[\Gamma_0(\lambda)V\psi_{\pm}(\cdot,\omega,\lambda)](\mp\theta), \qquad (2.63)$$

with  $\theta, \omega \in \mathbb{S}^{n-1}$  and  $\lambda > 0$ .

Here  $o_{av}$  denotes a Cesaro averaged asymptotics. We refer to 2.4.17 [164, Theorem 6.7.4] for the proof, which relies on the method of stationary phase. Our only use of Theorem 2.4.17 is in Section 4.2 to develop intuition for the form of the wave operator and thus the details are not necessary here.

**Definition 2.4.18.** For fixed  $\omega \in \mathbb{S}^{n-1}$  and  $\lambda > 0$  the coefficient  $a_{-}(\cdot, \omega, \lambda) : \mathbb{S}^{n-1} \to \mathbb{C}$  of the outgoing spherical wave in the asymptotics (2.62) is called the *scattering amplitude* for the energy  $\lambda$  and direction  $\omega$ .

We will justify the name scattering amplitude in Lemma 2.4.33.

For  $\lambda > 0$ , the operator  $\Gamma_0(\lambda)$  is related to the limiting absorption principle via Stone's formula 2.4.12.

**Lemma 2.4.19.** For  $\lambda > 0$  and  $t > \frac{1}{2}$  the operator  $\Gamma_0(\lambda)^*\Gamma_0(\lambda) : H^{0,t} \to H^{0,-t}$  satisfies

$$2\pi i \Gamma_0(\lambda)^* \Gamma_0(\lambda) = R_0(\lambda + i0) - R_0(\lambda - i0).$$

In particular,  $2\pi i\Gamma_0(\lambda)^*\Gamma_0(\lambda)$  is an integral operator with kernel given by Equation (2.56). Proof. For  $t > \frac{1}{2}$ ,  $\lambda > 0$ ,  $f \in H^{0,t}$  and  $x \in \mathbb{R}^n$  we check that

$$[\Gamma_{0}(\lambda)^{*}\Gamma_{0}(\lambda)f](x) = 2^{-\frac{1}{2}}(2\pi)^{-\frac{n}{2}}\lambda^{\frac{n-2}{4}} \int_{\mathbb{S}^{n-1}} e^{i\lambda^{\frac{1}{2}}\langle\omega,x\rangle} [\Gamma_{0}(\lambda)f](\omega) d\omega$$

$$= 2^{-1}(2\pi)^{-n}\lambda^{\frac{n-2}{2}} \int_{\mathbb{S}^{n-1}} e^{i\lambda^{\frac{1}{2}}\langle x,\omega\rangle} \left(\int_{\mathbb{R}^{n}} e^{-i\lambda^{\frac{1}{2}}\langle y,\omega\rangle} f(y) dy\right) d\omega$$

$$= \frac{\lambda^{\frac{n-2}{2}}}{2(2\pi)^{n}} \int_{\mathbb{R}^{n}} \left(\int_{\mathbb{S}^{n-1}} e^{-i\lambda^{\frac{1}{2}}\langle y-x,\omega\rangle} d\omega\right) f(y) dy.$$

Since  $f \in H^{0,t}$  was arbitrary, a comparison with Equation (2.56) gives the result.

We now factorise the potential V in order to use the integral kernels defined by Theorem 2.4.12 and Lemma 2.4.19 to define trace class operators, which we shall use extensively in Chapter 5. For  $x \in \mathbb{R}^n$  we introduce the notation

$$v(x) = |V(x)|^{\frac{1}{2}}, \quad U(x) = \begin{cases} 1, & \text{if } V(x) \ge 0, \\ -1, & \text{if } V(x) < 0, \end{cases}$$
 (2.64)

so that V = vUv.

**Lemma 2.4.20.** Suppose that  $g_1, g_2 : \mathbb{R}^n \to \mathbb{C}$  are compactly supported with  $g = g_1g_2$ . For  $\lambda > 0$  define the operator  $B(\lambda) \in \mathcal{B}(\mathcal{H})$  by  $B(\lambda) = g_1(R_0(\lambda + i0) - R_0(\lambda - i0)) g_2$ . Then  $B(\lambda) \in \mathcal{L}^1(\mathcal{H})$  and

$$\operatorname{Tr}(B(\lambda)) = \frac{(2\pi i)\lambda^{\frac{n-2}{2}}\operatorname{Vol}(\mathbb{S}^{n-1})}{2(2\pi)^n} \int_{\mathbb{R}^n} g(x) \, \mathrm{d}x.$$

Furthermore, for  $t > \frac{1}{2}$  and  $\lambda > 0$  we have the equality  $R_0(\lambda + i0) - R_0(\lambda - i0) = 2\pi i \Gamma_0(\lambda)^* \Gamma_0(\lambda)$  as operators in  $\mathcal{B}(H^{0,t}, H^{0,-t})$ .

*Proof.* Fix  $t > \frac{1}{2}$  and  $\lambda > 0$ . As an operator in  $\mathcal{B}(H^{0,t}, H^{0,-t})$  we have (see [3, Equation 4.3] and [6, Equation 15]) that  $A(\lambda) = R_0(\lambda + i0) - R_0(\lambda - i0)$  is an integral operator with integral kernel

$$A(\lambda, x, y) = \frac{(2\pi i)}{2(2\pi)^n} \lambda^{\frac{n-2}{2}} \int_{\mathbb{S}^{n-1}} e^{i\lambda^{\frac{1}{2}} \langle \omega, x-y \rangle} d\omega,$$

for  $x, y \in \mathbb{R}^n$ . Direct computation shows that the integral kernel of  $(2\pi i)\Gamma_0(\lambda)^*\Gamma_0(\lambda)$  is the same as that of  $A(\lambda)$ .

That  $B(\lambda)$  is trace class is proved in [73, Lemma III.2] (see also the comments on the bottom of page 32), however we provide an alternative proof. For any  $t > \frac{1}{2}$  direct computation shows that  $g_1\Gamma_0(\lambda)^* \in \mathcal{L}^2(\mathcal{P},\mathcal{H})$  and  $\Gamma_0(\lambda)g_2 \in \mathcal{L}^2(\mathcal{H},\mathcal{P})$  and thus  $B(\lambda) \in \mathcal{L}^1(\mathcal{H})$ .

We can then compute the trace of  $B(\lambda)$  by integrating along the diagonal, which completes the proof.

Remark 2.4.21. The assumption of compact support on the potential V is stronger than necessary to guarantee that the operator  $B(\lambda)$  is trace class. It is sufficient that the multiplication operators corresponding to  $g_1, g_2$  map  $\mathcal{H}$  into  $H^{0,t}$  for some  $t > \frac{1}{2}$ . The computation of the trace of  $B(\lambda)$  was first done by Buslaev [36] (see also Newton [124] and Bollé and Osborn [31]).

In fact we can differentiate  $B(\lambda)$  arbitrarily in  $\lambda$  in the trace norm.

**Lemma 2.4.22.** Suppose that  $g_1, g_2$  are compactly supported and  $g = g_1g_2$ . For  $\lambda > 0$  define the operator  $B(\lambda) \in \mathcal{L}^1(\mathcal{H})$  by  $B(\lambda) = g_1 (R_0(\lambda + i0) - R_0(\lambda - i0)) g_2$ . Then  $B(\lambda)$  is differentiable in  $\lambda$  in the norm of  $\mathcal{L}^1(\mathcal{H})$  and for all  $\ell \in \mathbb{N}$  we have

$$\frac{\mathrm{d}^{\ell-1}}{\mathrm{d}\lambda^{\ell-1}}B(\lambda) = (\ell-1)!g_1 \left( R_0(\lambda+i0)^{\ell} - R_0(\lambda-i0)^{\ell} \right) g_2. \tag{2.65}$$

In particular for  $\lambda > 0$  and  $\ell \in \mathbb{N}$  we have that  $g_1(R_0(\lambda + i0)^{\ell} - R_0(\lambda - i0)^{\ell}) g_2$  has for  $x, y \in \mathbb{R}^n$  the integral kernel

$$\frac{(2\pi i)\Gamma\left(\frac{n}{2}\right)\lambda^{\frac{n}{2}-\ell}}{2(\ell-1)!\Gamma\left(\frac{n}{2}+1-\ell\right)(2\pi)^n}g_1(x)g_2(y)\int_{\mathbb{S}^{n-1}}e^{-i\lambda^{\frac{1}{2}}\langle\omega,y-x\rangle}\,\mathrm{d}\omega+\tilde{A}(\lambda,x,y)$$

where  $\tilde{A}(\lambda,\cdot,\cdot)$  vanishes on the diagonal and thus

$$\operatorname{Tr}\left(g_1\left(R_0(\lambda+i0)^{\ell}-R_0(\lambda-i0)^{\ell}\right)g_2\right) = \frac{(2\pi i)\Gamma\left(\frac{n}{2}\right)\lambda^{\frac{n}{2}-\ell}\operatorname{Vol}(\mathbb{S}^{n-1})}{2(\ell-1)!\Gamma\left(\frac{n}{2}+1-\ell\right)(2\pi)^n}\int_{\mathbb{R}^n}g(x)\,\mathrm{d}x.$$
(2.66)

*Proof.* That  $B(\lambda)$  is differentiable is proved in [164, Lemma 8.1.8] and so we obtain Equation (2.65). Differentiating  $(\ell - 1)$  times the integral kernel for  $B(\lambda)$  we obtain

$$\frac{\mathrm{d}^{\ell-1}}{\mathrm{d}\lambda^{\ell-1}}B(\lambda,x,y) = \frac{(2\pi i)}{2(2\pi)^n}g_1(x)g_2(y)\sum_{j=0}^{\ell-1} \binom{\ell-1}{j} \left(\frac{\mathrm{d}^j}{\mathrm{d}\lambda^j}\lambda^{\frac{n-2}{2}}\right) \frac{\mathrm{d}^{\ell-1-j}}{\mathrm{d}\lambda^{\ell-1-j}} \int_{\mathbb{S}^{n-1}} \mathrm{e}^{i\lambda^{\frac{1}{2}}\langle\omega,x-y\rangle} \,\mathrm{d}\omega.$$

By factoring as the product of two Hilbert-Schmidt operators as in Lemma 2.4.20, each term in the sum is individually trace class. All terms except the  $j = \ell - 1$  term vanish on the diagonal since they contain an  $\langle \omega, x - y \rangle^j$  term, so integrating over the diagonal gives Equation (2.66).

Remark 2.4.23. The differentiability of  $B(\lambda)$  and Equation (2.65) can be checked directly at the level of kernels. Doing so demonstrates that again the assumption of compact support is stronger than necessary, it is enough that the multiplication operators  $(1 + |\cdot|)^{\ell}g_1$ ,  $(1 + |\cdot|)^{\ell}g_2$  map  $\mathcal{H}$  into  $H^{0,t}$  for some  $t > \frac{1}{2}$ .

In fact we can easily generalise the above result to all positive integer powers of  $A(\lambda)$  and  $B(\lambda)$  as follows.

**Lemma 2.4.24.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 2n$ . For  $\lambda > 0$  define the operators  $A(\lambda) \in \mathcal{B}(\mathcal{H})$  and  $B(\lambda) \in \mathcal{B}(\mathcal{P})$  by  $A(\lambda) = (\Gamma_0(\lambda)V\Gamma_0(\lambda)^*$  and  $B(\lambda) = v(R_0(\lambda + i0) - R_0(\lambda - i0))Uv$ . Then we have

$$\operatorname{Tr}(A(\lambda)^{\ell}) = (2\pi i)^{-\ell} \operatorname{Tr}(B(\lambda)^{\ell})$$

for all  $\ell \in \mathbb{N}$ .

*Proof.* That  $A(\lambda)^{\ell}$  and  $B(\lambda)^{\ell}$  are trace class follows from the fact that  $A(\lambda)$  and  $B(\lambda)$  are. We first write down the integral kernels explicitly and then integrate to show the traces are the same. Fix  $\lambda > 0$  and compute for any  $f \in H^{s,t}$  that

$$[\Gamma_0(\lambda)^* \Gamma_0(\lambda) f](x) = 2^{-\frac{1}{2}} \lambda^{\frac{n-2}{4}} (2\pi)^{-\frac{n}{2}} \int_{\mathbb{S}^{n-1}} e^{i\lambda^{\frac{1}{2}} \langle x, \omega \rangle} [\Gamma_0(\lambda) f](\omega) d\omega$$

$$= 2^{-1} \lambda^{\frac{n-2}{4}} (2\pi)^{-n} \int_{\mathbb{S}^{n-1}} \int_{\mathbb{R}^n} e^{-i\lambda^{\frac{1}{2}} \langle \omega, y - x \rangle} f(y) dy d\omega$$

$$= \frac{1}{2\pi i} [(R_0(\lambda + i0) - R_0(\lambda - i0)) f](x),$$

where we have applied Theorem 2.4.12 for the final equality. We can thus determine the expression

$$\begin{split} A(\lambda)^{\ell} &= (\Gamma_{0}(\lambda)V\Gamma_{0}(\lambda)^{*})^{\ell} \\ &= \Gamma_{0}(\lambda)V(\Gamma_{0}(\lambda)^{*}\Gamma_{0}(\lambda)V)^{\ell-1}\Gamma_{0}(\lambda)^{*} \\ &= (2\pi i)^{-(\ell-1)}\Gamma_{0}(\lambda)V((R_{0}(\lambda+i0)-R_{0}(\lambda-i0))V)^{\ell-1}\Gamma_{0}(\lambda)^{*} \\ &= (2\pi i)^{-(\ell-1)}\Gamma_{0}(\lambda)VB(\lambda)^{\ell-1}\Gamma_{0}(\lambda)^{*}. \end{split}$$

Noting the identity  $\Gamma_0(\lambda)^*\Gamma_0(\lambda) = (2\pi i)^{-1}(R_0(\lambda+i0)-R_0(\lambda-i0))$  we find using cyclicity of the trace that

$$\operatorname{Tr}(A(\lambda)^{\ell}) = (2\pi i)^{-(\ell-1)} \operatorname{Tr}(\Gamma_0(\lambda) V B(\lambda)^{\ell-1} \Gamma_0(\lambda)^*) = (2\pi i)^{-\ell} \operatorname{Tr}(B(\lambda)^{\ell}).$$

We can also show the equality of traces at the level of integral kernels, however this is not necessary.  $\Box$ 

We now use the operator  $\Gamma_0(\lambda)$  to construct a generalised Fourier transform which

diagonalises the perturbed Hamiltonian H.

**Definition 2.4.25.** Define for  $\lambda > 0$  and  $t > \frac{1}{2}$  the operators  $\Gamma_{\pm}(\lambda) : H^{0,t} \to \mathcal{P}$  by

$$\Gamma_{\pm}(\lambda) = \Gamma_0(\lambda)(\mathrm{Id} - VR(\lambda \pm i0)).$$

**Proposition 2.4.26.** For  $\lambda > 0$  and  $t > \frac{1}{2}$  the operators  $\Gamma_{\pm}(\lambda) : H^{0,t} \to \mathcal{P}$  and their formal adjoints  $\Gamma_{\pm}(\lambda)^* : \mathcal{P} \to H^{0,-t}$  are uniformly bounded on compact intervals. Moreover, the operator-valued functions  $\Gamma_{\pm}(\lambda)$  are strongly continuous in  $\lambda > 0$ .

*Proof.* This follows from Lemma 2.4.15 and the limiting absorption principle for H (Theorem 2.4.8).

**Definition 2.4.27.** For  $s \in \mathbb{R}$  and  $t > \frac{1}{2}$  we define the generalised Fourier transforms  $F_{\pm}: H^{s,t} \to \mathcal{H}_{spec}$  by

$$[F_{\pm}f](\lambda,\omega) = [\Gamma_{\pm}(\lambda)f](\omega).$$

**Theorem 2.4.28.** Suppose that V satisfies Assymption 2.2.14. Then the operators  $F_{\pm}$  extend by continuity to bounded operators  $F_{\pm}: \mathcal{H} \to \mathcal{H}_{ac}$  which satisfy  $F_{\pm}F_{\pm}^* = \operatorname{Id}$  and  $F_{\pm}^*F_{\pm} = P_{ac}(H)$ . The operators  $F_{\pm}$  diagonalise H in the sense that for  $f \in \operatorname{Dom}(H)$  and  $\varphi: \mathbb{R}^+ \to \mathbb{C}$  measurable we have

$$[F_{\pm}\varphi(H)F_{\pm}^*f](\lambda,\omega) = \varphi(\lambda)f(\lambda,\omega), \quad \lambda \in \mathbb{R}^+, \ \omega \in \mathbb{S}^{n-1}.$$
 (2.67)

*Proof.* First note that for  $f \in H^{0,t}$  and  $\lambda > 0$  we have the limits

$$\pi^{-1} \lim_{\varepsilon \to 0} \varepsilon \left| \left| R_0(\lambda \pm i\varepsilon) f \right| \right|^2 = \lim_{\varepsilon \to 0} \left\langle \left( R_0(\lambda + i\varepsilon) - R_0(\lambda - i\varepsilon) \right) f, f \right\rangle = |\Gamma_0(\lambda) f|^2.$$

The limiting absorption principle also gives us for  $\lambda > 0$  the limit

$$\lim_{\varepsilon \to 0} ||VR(\lambda + i\varepsilon)f - VR(\lambda + i0)f||_{H^{0,t}} = 0.$$

Combining these, we find for any  $f \in H^{0,t}$  and  $\lambda > 0$  that

$$\begin{split} |\Gamma_{\pm}(\lambda)f|^2 &= |\Gamma_0(\lambda)(\operatorname{Id} - VR(\lambda \pm i0))f|^2 \\ &= \pi^{-1} \lim_{\varepsilon \to 0} \varepsilon \, ||R_0(\lambda \pm i\varepsilon)(\operatorname{Id} - VR(\lambda \pm i\varepsilon))f||^2 \\ &= \pi^{-1} \lim_{\varepsilon \to 0} \varepsilon \, ||R(\lambda \pm i\varepsilon)f||^2 \\ &= \frac{\mathrm{d}\langle P_{\lambda}f, f\rangle}{\mathrm{d}\lambda}, \end{split}$$

by Stone's theorem (see [164, Equation (0.1.10)]. Integrating over  $\mathbb{R}^+$  we find

$$||F_{\pm}f||^2 = \int_{\mathbb{R}^+} |\Gamma_{\pm}(\lambda)f|^2 d\lambda = \int_{\mathbb{R}^+} \frac{d\langle P_{\lambda}f, f\rangle}{d\lambda} d\lambda = ||P_{ac}(H)f||^2,$$

which implies  $F_{\pm}^*F_{\pm} = P_{ac}(H)$ . For the second partial isometry relation, we proceed by contradiction. Supposing that  $F_{\pm}F_{\pm}^* \neq \text{Id}$ , we find there exists  $g \in \mathcal{H}_{spec}$  such that all  $f \in H^{0,t}$  and intervals  $I \subset \mathbb{R}^+$  we have  $\langle F_{\pm}P_If, g \rangle = 0$ , which further implies

$$\int_{I} \langle \Gamma_{\pm}(\lambda) f, g \rangle \, \mathrm{d}\lambda = 0.$$

Since I is an arbitrary interval we find  $\langle \Gamma_{\pm}(\lambda)f, g(\lambda, \cdot) \rangle = \langle f, \Gamma_{\pm}(\lambda)^*g(\lambda, \cdot) \rangle = 0$  for almost every  $\lambda > 0$ . Since  $f \in H^{0,t}$  is arbitrary we have  $\Gamma_{\pm}(\lambda)^*g(\lambda, \cdot) = 0$  and thus by [164, Lemma 6.6.4] we find  $g(\lambda, \cdot) = 0$  for almost all  $\lambda > 0$ .

To prove the intertwining relation of Equation (2.67), it suffices to check that for all  $f \in C_c^{\infty}(\mathbb{R}^n)$  and  $g \in C_c(\mathbb{R}^+) \otimes L^2(\mathbb{S}^{n-1})$  that

$$0 = \int_{\mathbb{R}^+} \langle \Gamma_{\pm}(\lambda)(H - \lambda)f, g(\lambda, \cdot) \rangle \, d\lambda = \int_{\mathbb{R}^+} \langle f, (H - \lambda)\Gamma_{\pm}(\lambda)^* g(\lambda, \cdot) \rangle \, d\lambda. \tag{2.68}$$

We readily check that  $[H_0\Gamma_0(\lambda)^*g(\lambda,\cdot)](x) = \lambda g(\lambda,x)$  and so

$$[(H - \lambda)\Gamma_{\pm}(\lambda)^*g(\lambda, \cdot)](x) = [(H - \lambda)(\operatorname{Id} - R(\lambda \mp i0)V)\Gamma_0(\lambda)^*g(\lambda, \cdot)](x)$$

$$= [(H_0 + V - \lambda)\Gamma_0(\lambda)^*g(\lambda, \cdot)](x)$$

$$- [(H - \lambda)R(\lambda \mp i0)V\Gamma_0(\lambda)^*g(\lambda, \cdot)](x)$$

$$= 0,$$

which shows that the final integrand of Equation (2.68) is zero.

**Theorem 2.4.29.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 1$ . Then for all  $t > \frac{1}{2}$  and  $f, g \in H^{0,t}$  we have

$$\langle W_{\pm}f, g \rangle = \int_0^\infty \langle \Gamma_0(\lambda)f, \Gamma_{\pm}(\lambda)f \rangle \, \mathrm{d}\lambda = \langle F_0f, F_{\pm}g \rangle.$$

That is the wave operators are given by  $W_{\pm} = F_{\pm}^* F_0$ .

*Proof.* We first compute for  $f, g \in H^{0,t}$  with  $t > \frac{1}{2}$  that

$$\langle \Gamma_0(\lambda) f, \Gamma_{\pm}(\lambda) g \rangle = \langle \Gamma_0(\lambda) f, \Gamma_0(\lambda) (\operatorname{Id} - VR(\lambda \pm i\varepsilon)) g \rangle$$

$$= (2\pi i)^{-1} \lim_{\varepsilon \to 0} \langle (R(\lambda + i\varepsilon) - R(\lambda - i\varepsilon)) f, (\operatorname{Id} - VR(\lambda \pm \varepsilon)) g \rangle$$

$$= \pi^{-1} \lim_{\varepsilon \to 0} \varepsilon \langle R_0(\lambda \pm i\varepsilon) f, R(\lambda \pm i\varepsilon) g \rangle.$$

Then by [164, Lemma 6.6.8] we can compute by bringing the limit inside the integral that

$$\pi^{-1} \lim_{\varepsilon \to 0} \int_{\mathbb{R}} \langle R_0(\lambda \pm i\varepsilon) f, R(\lambda \pm \varepsilon) g \rangle \, d\lambda = \int_0^\infty \langle \Gamma_0(\lambda) f, \Gamma_{\pm}(\lambda) g \rangle \, d\lambda.$$
 (2.69)

One readily checks using Theorem 2.1.11 that

$$\int_0^\infty \langle \Gamma_0(\lambda) f, \Gamma_{\pm}(\lambda) g \rangle \, \mathrm{d}\lambda = \lim_{\varepsilon \to 0} 2\varepsilon \int_0^\infty \mathrm{e}^{-2\varepsilon t} \langle \mathrm{e}^{itH_0} f, \mathrm{e}^{\pm itH} g \rangle \, \mathrm{d}t$$

which gives the result.

**Definition 2.4.30.** Suppose V satisfies Assumption 2.2.14 for some  $\rho > \frac{n+1}{2}$ . We define the operators  $\mathcal{F}_{\pm} : \mathcal{H} \to \mathcal{H}$  by  $\mathcal{F}_{\pm} = F_{\pm}\mathcal{U}^*$ , where  $\mathcal{U}$  is as in Definition 2.2.7.

For  $0 \neq \xi \in \mathbb{R}^n$  we use the notation

$$\mathbb{S}^{n-1} \ni \hat{\xi} = |\xi|^{-1} \xi \tag{2.70}$$

for the unit vector in the direction of  $\xi$ .

**Lemma 2.4.31.** Suppose V satisfies Assumption 2.2.14 for some  $\rho > \frac{n+1}{2}$ . Let  $\psi_{\pm}$  be the generalised eigenfunctions for H of Proposition 2.4.13. For  $f, g \in C_c^{\infty}(\mathbb{R}^n)$  and  $x, \xi \in \mathbb{R}^n$  we have

$$[\mathcal{F}_{\pm}f](\xi) = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} \overline{\psi_{\pm}(x,\hat{\xi},|\xi|^2)} f(x) \, \mathrm{d}x, \quad \xi \in \mathbb{R}^n,$$
 (2.71)

and

$$[\mathcal{F}_{\pm}^* g](x) = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} \psi_{\pm}(x, \hat{\xi}, |\xi|^2) g(\xi) \, d\xi, \quad x \in \mathbb{R}^n.$$
 (2.72)

*Proof.* For  $t > \frac{n+1}{2}$ ,  $f \in H^{0,t}$ ,  $\omega \in \mathbb{S}^{n-1}$  and  $\lambda > 0$  we note that

$$2^{\frac{1}{2}}\lambda^{-\frac{n-2}{4}}(2\pi)^{\frac{n}{2}}[\Gamma_{\pm}(\lambda)f](\omega) = \langle (\mathrm{Id} - VR(\lambda \pm i0))f, \psi_0(\cdot, \omega, \lambda) \rangle$$
$$= \langle f, (\mathrm{Id} - R(\lambda \mp i0)V)\psi_0(\cdot, \omega, \lambda) \rangle$$
$$= \langle f, \psi_{\pm}(\cdot, \omega, \lambda) \rangle.$$

Since  $\psi_{\pm} \in H^{0,t}$ , this immediately implies the formula for  $\mathcal{F}_{\pm}$ . To determine the adjoint, we let  $t > \frac{n+1}{2}$  and fix  $f \in H^{0,t}$  and  $g \in \mathcal{H}$  with compact support in  $\mathbb{R}^n \setminus \{0\}$ . Then we have

$$\langle \mathcal{F}_{\pm}^* f, g \rangle = \langle g, \mathcal{F}_{\pm} f \rangle = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} g(\xi) \left( \int_{\mathbb{R}^n} \overline{f(x)} \psi_{\pm}(x, \hat{\xi}, |\xi|^2) \, \mathrm{d}x \right) \, \mathrm{d}\xi.$$

Since the integral is absolutely convergent, Fubini's theorem allows us to interchange the

order of integration and obtain the formula for  $\mathcal{F}_{\pm}^*$  as claimed. For arbitrary  $g \in \mathcal{H}$  and  $x \in \mathbb{R}^n$  one should use the formula

$$[\mathcal{F}_{\pm}^* g](x) = (2\pi)^{-\frac{n}{2}} \lim_{r \to \infty} \int_{r^{-1} < |\xi| < r} g(\xi) \psi_{\pm}(x, \hat{\xi}, |\xi|^2) d\xi,$$

where the limit is taken in  $\mathcal{H}$ .

The operators  $\mathcal{F}_{\pm}$  can be considered as 'generalised Fourier transforms' which diagonalise the operator  $H = H_0 + V$  in the sense that  $\mathcal{F}_{\pm}H\mathcal{F}_{\pm}^* = M$ , and share many interesting properties with the Fourier transform. In particular we note that if V = 0 we recover the standard Fourier transform and we have the for  $f \in C_c^{\infty}(\mathbb{R}^n)$  and  $\xi \in \mathbb{R}^n$  the relation

$$f(\xi) = [\mathcal{F}_{\pm}\mathcal{F}_{\pm}^*f](\xi) = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} \overline{\psi_{\pm}(x,\hat{\xi},|\xi|^2)} [\mathcal{F}_{\pm}^*f](x) \, dx$$

$$= (2\pi)^{-n} \int_{\mathbb{R}^n} \int_{\mathbb{R}^n} \overline{\psi_{\pm}(x,\hat{\xi},|\xi|^2)} \psi_{\pm}(x,\hat{\rho},|\rho|^2) f(\rho) \, d\rho \, dx$$

$$= \int_{\mathbb{R}^n} \left( (2\pi)^{-n} \int_{\mathbb{R}^n} \overline{\psi_{\pm}(x,\hat{\xi},|\xi|^2)} \psi_{\pm}(x,\hat{\rho},|\rho|^2) \, dx \right) f(\rho) \, d\rho,$$

so that the generalised eigenfunctions  $\psi_{\pm}$  gives rise to a delta function, as in the case of the plane waves  $\psi_0$  and the standard Fourier transform.

## 2.4.3 The scattering matrix

Let S be the scattering operator for the pair  $(H, H_0)$  as defined in Definition 2.1.9. Since the scattering operator S satisfies  $[S, H_0] = 0$  the spectral theorem asserts that there exists a family  $\{S(\lambda)\}_{\lambda \in \mathbb{R}^+}$  of unitary operators on  $\mathcal{P} = L^2(\mathbb{S}^{n-1})$  such that for all  $\lambda \in \mathbb{R}^+$ we have  $[F_0SF_0^*f](\lambda,\omega) = S(\lambda)f(\lambda,\omega)$ . For  $\lambda \in \mathbb{R}^+$  the operator  $S(\lambda)$  is referred to as the scattering matrix at energy  $\lambda$  and was first studied by Wheeler [162] and Heisenberg [75]. The term 'scattering matrix' is historical and due to the fact that in dimension n=1we have  $L^2(\mathbb{S}^0) = \mathbb{C}^2$ , so that the scattering matrix is a bona fide matrix-valued function  $S: \mathbb{R}^+ \to M_2(\mathbb{C})$ .

For potentials decaying sufficiently fast, Kuroda [104, 105] used stationary scattering theory to establish an expression for the scattering matrix in terms of the potential and resolvent using the limiting absorption principal.

**Theorem 2.4.32** ([164, Theorem 6.10]). Suppose that V satisfies Assumption 2.2.14. Then the scattering matrix is given for all  $\lambda \in (0, \infty)$  by the equation

$$S(\lambda) = \operatorname{Id} - 2\pi i \Gamma_0(\lambda) (V - VR(\lambda + i0)V) \Gamma_0(\lambda)^*. \tag{2.73}$$

The operator  $S(\lambda)$  is unitary in  $L^2(\mathbb{S}^{n-1})$  and depends norm continuously on  $\lambda \in (0, \infty)$ .

*Proof.* By Theorem 2.4.29 we can write the wave operators in terms of generalised Fourier transforms and thus  $S = W_+^*W_- = (F_+^*F_0)^*(F_-^*F_0) = F_0^*(F_+F_-^*)F_0$ . Thus the scattering matrix is given by  $S = F_+F_-^*$ . So to prove that Equation (2.73) defines the scattering matrix it suffices to check for any  $t > \frac{1}{2}$  and  $f \in H^{0,t}$  that  $S(\lambda)$  defined by Equation (2.73) satisfies

$$[F_+f](\lambda,\omega) = S(\lambda)[F_-f](\lambda,\omega), \quad \lambda \in \mathbb{R}^+, \omega \in \mathbb{S}^{n-1},$$

which by Definition 2.4.27 is equivalent to the relation

$$\Gamma_{+}(\lambda)f = S(\lambda)\Gamma_{-}(\lambda)f. \tag{2.74}$$

So we compute the right hand side of Equation (2.74) as

$$S(\lambda)\Gamma_{-}(\lambda)f = \Gamma_{-}(\lambda)f - 2\pi i\Gamma_{0}(\lambda)(V - VR(\lambda + i0)V)\Gamma_{0}(\lambda)^{*}\Gamma_{-}(\lambda)f$$

$$= \Gamma_{0}(\lambda)(\operatorname{Id} - VR(\lambda - i0))f$$

$$- 2\pi i\Gamma_{0}(\lambda)(V - VR(\lambda + i0)V)\Gamma_{0}(\lambda)^{*}\Gamma_{0}(\lambda)(\operatorname{Id} - VR(\lambda - i0))f$$

$$= \Gamma_{0}(\lambda)f - \Gamma_{0}(\lambda)VR(\lambda - i0)f$$

$$- 2\pi i\Gamma_{0}(\lambda)(V - VR(\lambda + i0)V)\Gamma_{0}(\lambda)^{*}\Gamma_{0}(\lambda)(\operatorname{Id} - VR(\lambda - i0))f$$

The left hand side of Equation (2.74) can be written as

$$\Gamma_{+}(\lambda)f = \Gamma_{0}(\lambda)f - \Gamma_{0}(\lambda)VR(\lambda + i0)f.$$

Taking the difference of the two expressions shows that Equation (2.74) is equivalent to

$$\Gamma_0(\lambda)(R(\lambda - i0) - R(\lambda + i0))f$$

$$= -2\pi i \Gamma_0(\lambda)(V - VR(\lambda + i0)V)\Gamma_0(\lambda)^* \Gamma_0(\lambda)(\mathrm{Id} - VR(\lambda - i0))f.$$

We recall the identity  $2\pi i \Gamma_0(\lambda) \Gamma_0(\lambda)^* = R_0(\lambda + i0) - R_0(\lambda - i0)$  (see Lemma 2.4.19) and thus it suffices to check

$$R(\lambda + i0) - R(\lambda - i0) = V(\text{Id} - R(\lambda + i0)V)(R_0(\lambda + i0) - R_0(\lambda - i0))(\text{Id} - VR(\lambda - i0)),$$
(2.75)

where both sides are considered as mappings from  $H^{0,t}$  to  $H^{0,-t}$ . For  $z \in \rho(H)$ , Equation (2.75) follows from resolvent identities (see Equation (2.50)) and for  $z = \lambda + i0$  the result follows by the limiting absorption principle (Theorems 2.4.5 and 2.4.8).

We can use the Lippmann-Schwinger equation to explicitly determine the integral kernel of the scattering matrix in terms of the scattering amplitude of Definition 2.4.18.

**Lemma 2.4.33.** [164, Theorem 6.7.8] Suppose that V satisfies Assumption 2.2.14 for some  $\rho > \frac{n+1}{2}$ . Then for  $\lambda > 0$  the operator  $S(\lambda) - \mathrm{Id} \in \mathcal{B}(\mathcal{P})$  is an integral operator with integral kernel

$$\tilde{S}(\theta,\omega,\lambda) = (2\pi i)(2\pi)^{-\frac{n+1}{2}}\lambda^{\frac{n-1}{4}}a_{-}(\theta,\omega,\lambda),$$

where  $a_{-}$  is the scattering amplitude of Definition 2.4.18.

*Proof.* Note that for  $t > \frac{n}{2}$  and  $f \in H^{0,t}$  we can write

$$[\Gamma_0(\lambda)f](\omega) = 2^{-\frac{1}{2}}\lambda^{\frac{n-2}{4}}(2\pi)^{-\frac{n}{2}}\langle f, \psi_0(\cdot, \omega, \lambda)\rangle,$$

where here we are considering  $\langle \cdot, \cdot \rangle$  as the duality between  $H^{0,t}$  and  $H^{0,-t}$ . Thus by Equation (2.63) we have the scattering amplitude  $a_{\pm}$  given by the absolutely convergent integral

$$a_{-}(\theta,\omega,\lambda) = -2^{-1}(2\pi)^{-\frac{n-1}{2}}\lambda^{\frac{n-3}{4}} \int_{\mathbb{R}^n} e^{-i\lambda^{\frac{1}{2}}\langle\theta,x\rangle} V(x)\psi_{\pm}(x,\omega,\lambda) dx$$
$$= -2^{-1}(2\pi)^{-\frac{n-1}{2}}\lambda^{\frac{n-3}{4}} \langle (V - VR(\lambda + i0)V)\psi_{0}(\cdot,\omega,\lambda), \psi_{0}(\cdot,\theta,\lambda) \rangle.$$

Comparing with Equation (2.73) completes the proof.

Birman and Kreĭn [25] note that for trace class perturbations V the operator  $S(\lambda)$ —Id defines a compact operator on  $\mathcal{P}$  for all  $\lambda \in \mathbb{R}^+$ . In fact much stronger summability conditions for the scattering operator are known. The following can be found as [164, Proposition 8.1.5], we omit the proof since it requires substantial techniques from the method of  $H_0$ -smoothness.

**Theorem 2.4.34.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 1$ . Then for all  $\lambda > 0$  and  $p > \frac{n-1}{\rho-1}$  we have  $S(\lambda) - \mathrm{Id} \in \mathcal{L}^p(\mathcal{P})$ , the p-th Schatten class. In particular, if  $\rho > n$  then  $S(\lambda)$  – Id is trace class.

In fact the proof of Theorem 2.4.34 can be used to establish differentiability of the scattering matrix in the Schatten norms also. The following is [164, Proposition 8.1.9].

**Proposition 2.4.35.** Let  $\ell \in \mathbb{N} \cup \{0\}$  and  $p \geq 1$  and suppose that V satisfies Assumption 2.2.14 for some  $\rho > \ell + 1 + \frac{n-1}{p}$ . Then the scattering matrix  $S : \mathbb{R}^+ \to \mathcal{L}^p(\mathcal{P})$  is  $\ell$  times continuously differentiable in  $\mathcal{L}^p(\mathcal{H})$ . In particular, if  $\rho > \ell + n$  then S is  $\ell$  times differentiable in trace norm.

### 2.5 Levinson's theorem

Levinson's theorem [108] is a fundamental result in scattering theory relating the number of bound states of a Hamiltonian  $H = H_0 + V$  to a quantity related to the 'total phase

shift'. The original proof of Levinson's theorem uses classical ODE theory, complex analysis and the subtleties of a scattering object known as the Jost function. As such the proof is specific to one dimension. In this section we describe the time-delay operator and the spectral shift function, important quantities in scattering theory which can be used to prove and generalise Levinson's theorem to higher dimensions. We discuss the high energy behaviour of the time-delay operator and spectral shift function and how they relate to Levinson's theorem. We conclude with several simple examples.

#### 2.5.1 Time delay

In this section we describe intuitively the time delay of a quantum mechanical system and construct an operator, known as the Eisenbud-Wigner time delay operator which is defined to be the time delay in the spectral representation and can be explicitly described in terms of the scattering matrix. The time delay of particles was first considered by Eisenbud [57] and Wigner [163] in a stationary manner.

To define the time delay, we consider for r > 0 the orthogonal projection  $P_r$  onto the functions with support in the ball of radius r in  $\mathcal{H}$ . For  $f \in \mathcal{H}$  a measure of the total time spent in  $P_r\mathcal{H}$  is given by

$$\int_{\mathbb{D}} \left| \left| P_r e^{-itH} f \right| \right|^2 dt. \tag{2.76}$$

For any  $f \in \mathcal{H}$ , the quantities  $e^{-itH_0}f$  and  $e^{-itH}W_-f$  agree in the limit as  $t \to -\infty$ . The difference of times spent in  $P_r\mathcal{H}$  of these states is given by

$$\Delta T_r(f) = \int_{\mathbb{R}} \left( \left| \left| P_r e^{-itH} W_- f \right| \right|^2 - \left| \left| P_r e^{-itH_0} f \right| \right|^2 \right) dt.$$
 (2.77)

As  $r \to \infty$  one expects a finite limit (at least for a dense set of  $f \in \mathcal{H}$ ). We will show that there exists a self-adjoint operator T, called the Eisenbud-Wigner time delay operator, which satisfies

$$\langle f, Tf \rangle := \lim_{r \to \infty} \Delta T_r(f).$$
 (2.78)

This time-dependent definition of the time-delay operator was introduced in [82] and further discussed in [83]. Proofs of Equations (2.77) and (2.78) were first given in [114] in the case that the scattering matrix is scalar and later extended in [86]. An alternative definition of time delay is provided in [122]. A thorough and physically motivated discussion of the time delay operator is given in [115], see also [48]. The operator T is given explicitly in terms of the scattering matrix. To prove such a result requires some intermediary results on the differentiability of the scattering matrix.

Recall from Theorem 2.4.1 for  $\mu > 0$  the trace operator  $\gamma(\mu) : C_c^{\infty}(\mathbb{R}^n) \to \mathcal{P}$  defined

by  $[\gamma(\mu)f](\omega) = f(\mu\omega)$ . It is well-known that for any  $s > \frac{1}{2}$  and  $m \in \mathbb{R}$  the operator  $\gamma(\mu)$  extends to an element of  $\mathcal{B}(H^{s,m},\mathcal{P})$ . The following result, which demonstrates the differentiability of the trace map, is due to Jensen [86, Lemma 3.3].

**Lemma 2.5.1.** Let  $m \in \mathbb{R}$ ,  $k \geq 0$  an integer and  $s > k + \frac{1}{2}$ . Then  $\mu \mapsto \gamma(\mu)$  is k times continuously differentiable in norm on  $\mathcal{B}(H^{s,m},\mathcal{P})$ .

To construct the time delay operator we also need differentiability properties of the boundary values of the free resolvent along the positive real axis. The following result is a combination of [86, Lemma 3.4].

**Lemma 2.5.2.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 2$ . Suppose that  $s, m > \frac{3}{2}$ . Then  $\lambda \mapsto R(\lambda + i0)$  is continuously differentiable as a map from  $(0, \infty)$  to  $\mathcal{B}(H^{-1,s}, H^{1,-m})$ .

With the differentiability of the trace and free resolvent established, we can use the stationary formula for the scattering matrix of Theorem 2.4.32 to determine the differentiability of the scattering matrix.

We are now ready to construct the time delay operator as in [86, Theorem 3.7].

**Theorem 2.5.3.** Let V satisfy Assumption 2.2.14 for some  $\varepsilon > 0$  and  $\rho > 2 + \varepsilon$  and let T be defined by

$$\operatorname{Dom}(T) = \{ f \in \mathcal{H} : P_{[a,b]}(H_0)f = f \text{ for some } [a,b] \subset (0,\infty) \} \subset \operatorname{Dom}(H_0) \}$$

with  $T = -iF_0^*S(\cdot)^*S'(\cdot)F_0$ . The operator T is essentially self-adjoint on Dom(T) and T commutes with  $H_0$ .

*Proof.* By Proposition 2.4.35 we have that the scattering matrix is differentiable in the pth Schatten class  $\mathcal{L}^p(\mathcal{P}) \subset \mathcal{B}(\mathcal{P})$  for all  $p > \frac{n-1}{\varepsilon}$ . Unitarity of the scattering matrix then implies the relation

$$0 = \frac{\mathrm{d}}{\mathrm{d}\lambda}(S(\lambda)^*S(\lambda)) = S(\lambda)^*S'(\lambda) + (S'(\lambda))^*S(\lambda)$$

and thus for each  $\lambda \in (0, \infty)$  we find  $-iS(\lambda)^*S'(\lambda)$  is a bounded self-adjoint operator in  $\mathcal{B}(\mathcal{P})$ . So T is densely-defined, symmetric and for each  $[a,b] \subset (0,\infty)$  we have  $TP_{[a,b]}$  is a bounded self-adjoint operator. So T is essentially self-adjoint on Dom(T). That T commutes with  $H_0$  follows from the definition in terms of S.

We refer to T as the Eisenbud-Wigner time delay operator, or simply the time delay. The explicit representation of T in Theorem 2.5.3 was first given in [152]. We note that here we have only discussed time delay for Euclidean Schrödinger operators, however time delay is a more fundamental concept of a scattering system and its definition generalises to a more abstract setting. We refer to [139] for a discussion of such topics.

**Lemma 2.5.4.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > n + 1$ . Then for all  $\lambda \in \mathbb{R}^+$  the operator  $S(\lambda)^*S'(\lambda)$  is trace class on  $\mathcal{P}$ .

*Proof.* By Theorem 2.4.34 the operator  $S'(\lambda)$  is trace class. Since the trace class operators form an ideal, the product  $S(\lambda)^*S'(\lambda)$  is trace class also.

### 2.5.2 The spectral shift function

The spectral shift function was first introduced by Lifshits to study the size of the eigenvalue shift for finite rank perturbations B of a lattice model operator  $A_0$  in quantum mechanics and later extended in [109] to computing the trace of  $f(A_0 + B) - f(A_0)$  for a large class of functions. Lifshits suggested the formula

$$\operatorname{Tr}(f(A_0 + B) - f(A_0)) = \int_{\mathbb{R}} f'(\lambda)\xi(\lambda, A_0 + B, A_0) \,d\lambda$$
 (2.79)

where the function  $\xi(\cdot, A_0 + B, A_0)$  depends only on the self-adjoint operators  $A_0$  and  $A_0 + B$ . The extension from discrete models to continuous models was first studied by Kreĭn [100], who provided a suitable class of perturbations for which the spectral shift function  $\xi$  exists and Equation (2.79) holds. In [101] Kreĭn showed that the spectral shift function exists and satisfies Equation (2.79) if the operator  $(A_0 + B - z)^{-1} - (A_0 - z)^{-1}$  is trace class for some (and hence all)  $z \in \mathbb{C} \setminus \mathbb{R}$ . The approach of Kreĭn is based on the theory of perturbation determinants as in Definition 2.5.5 (see [33], [99], [70, Section IV.3] and [151, Chapter 5] for more details).

**Definition 2.5.5.** Suppose that  $\mathcal{H}$  is a Hilbert space and A is a trace class operator on  $\mathcal{H}$ . Then we define

$$Det(Id + A) = \prod_{j=1}^{\infty} (1 + \mu_j(A)),$$

where the  $\mu_j$  denote the eigenvalues of A. The absolute convergence of the product is guaranteed by the assumption that A is trace class. If  $A \in \mathcal{L}^p(\mathcal{H})$  (the p-th Schatten class) for some  $p \in \mathbb{N}$  then we can define the regularised determinant by

$$Det_p(Id + A) = \prod_{j=1}^{\infty} \left( (1 + \mu_j(A)) \exp\left( \sum_{\ell=1}^{p-1} \frac{(-1)^{\ell}}{\ell} \mu_j(A)^{\ell} \right) \right).$$

**Definition 2.5.6.** Let A and  $B = A + \tilde{V}$  be self-adjoint operators on  $\mathcal{H}$  such that  $\tilde{V}(A-z)^{-1}$  defines a trace class operator for some (and hence all)  $z \in \rho(A)$ . Then we define for  $z \in \rho(A)$  the function

$$D(z) = \text{Det}(\text{Id} + \tilde{V}(A - z)^{-1}).$$

The analytic Fredholm alternative tells us that D(z) is defined and holomorphic on the set  $\rho(A)$  and that  $D(z) \neq 0$  for all  $z \in \rho(A) \cap \rho(B)$ . Furthermore for  $z \in \rho(A) \cap \rho(B)$ we have [164, Equation 0.9.3] the equality

$$D^{-1}(z)D'(z) = \text{Tr}((A-z)^{-1} - (B-z)^{-1}).$$

The first theorem of Kreĭn demonstrates the existence of the spectral shift function for a trace class perturbation  $\tilde{V}$ .

**Theorem 2.5.7.** [164, Theorem 0.9.1] Let A and B be self-adjoint operators on  $\mathcal{H}$  such that  $\tilde{V} := B - A$  is trace class. Then for almost all  $\lambda \in \mathbb{R}$  the limit

$$\xi(\lambda, A, B) := \pi^{-1} \lim_{\varepsilon \to 0} \operatorname{Arg}(D(\lambda + i\varepsilon))$$
 (2.80)

exists and satisfies

$$\int_{\mathbb{R}} |\xi(\lambda, A, B)| \, \mathrm{d}\lambda \le \left| \left| \tilde{V} \right| \right|_{1}, \quad \text{ and } \quad \int_{\mathbb{R}} \xi(\lambda, A, B) \, \mathrm{d}\lambda = \mathrm{Tr}(\tilde{V}).$$

**Proposition 2.5.8.** [164, Proposition 0.9.1] Let A and B be self-adjoint operators on  $\mathcal{H}$  such that  $\tilde{V} := B - A$  is trace class. On component intervals of the set  $\rho(A) \cap \rho(B)$  of common regular points of the operators A and B the spectral shift function  $\xi(\lambda, A, B)$  assumes constant integer values. If  $\lambda$  is an isolated eigenvalue of finite multiplicity  $M_A(\lambda)$  of the operator A and  $M_B(\lambda)$  of the operator B, then

$$\xi(\lambda + 0, A, B) - \xi(\lambda - 0, A, B) = M_A(\lambda) - M_B(\lambda).$$

The second theorem of Kreĭn establishes Equation (2.79) for trace class perturbations  $\tilde{V}$ .

**Theorem 2.5.9.** [164, Theorem 0.9.3] Let A and B be self-adjoint operators on  $\mathcal{H}$  such that  $\tilde{V} := B - A$  is trace class and suppose that  $f \in C^1(\mathbb{R})$  and its derivative admits the representative

$$f'(\lambda) = \int_{\mathbb{R}} e^{-i\lambda t} dm(t)$$

for a finite (complex) measure m and almost all  $\lambda \in \mathbb{R}$ . Then  $f(B) - f(A) \in \mathcal{L}^1(\mathcal{H})$  and we have the trace formula

$$Tr(f(B) - f(A)) = \int_{\mathbb{R}} f'(\lambda)\xi(\lambda, A, B) d\lambda.$$
 (2.81)

We now specialise again to the operators  $H_0$  and  $H = H_0 + V$ . The following statement gives a relationship between the spectral shift function, the scattering matrix and the time

delay operator.

**Theorem 2.5.10.** Suppose  $V \in \mathcal{L}^1(\mathcal{H})$ . Then we have

$$Det(S(\lambda)) = e^{-2\pi i \xi(\lambda)}$$
(2.82)

for almost all  $\lambda \in \sigma_{ac}(H_0)$ . Furthermore, we have

$$Tr(S(\lambda)^*S'(\lambda)) = -2\pi i \xi'(\lambda). \tag{2.83}$$

Equation (2.82) dates back to an observation of Beth and Uhlenbeck [22] relating a phase shift to the second virial coefficient, and was first proved in the context of the spectral shift function by Birman and Kreĭn [25]. Unfortunately the above definition needs some modification since our multiplication operator V is not usually compact. To fix this, we consider relatively trace class perturbations.

**Lemma 2.5.11.** [73, Proposition III.1] Suppose that V satisfies Assumption 2.2.14 for some  $\rho > \frac{n+1}{2}$  and fix  $c > |\inf \sigma_p(H)|$ . Then the operators  $H_0 + c$  and H + c are positive definite and we have the difference  $R(-c)^m - R_0(-c)^m$  is trace class for any integer  $m \ge \rho$ .

Define the operators  $G = R(-c)^m$  and  $G_0 = R_0(-c)^m$ . Then we can use Theorem 2.5.7 to define the spectral shift function for the pair  $(G, G_0)$  and lead us to the following definition.

**Definition 2.5.12.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > n$ . We define the *spectral shift function* for the pair  $(H, H_0)$  to be the function

$$\xi(\lambda, H, H_0) = \begin{cases} -\xi((\lambda + c)^{-m}, G, G_0), & \text{if } \lambda > -c, \\ 0 & \text{if } \lambda - \le c. \end{cases}$$

The spectral shift function is determined uniquely by the constraint  $\xi(\lambda, H, H_0) = 0$  for  $\lambda \leq -c$ . We will often just write  $\xi(\lambda)$  for the spectral shift function of the pair  $(H, H_0)$  and drop the reference to these operators.

The spectral shift function satisfies the following properties.

**Theorem 2.5.13.** [164, Theorem 0.9.7] Suppose that V satisfies Assumption 2.2.14 for some  $\rho > n$ . Then the spectral shift function  $\xi$  of Definition 2.5.12 satisfies the estimate

$$m \int_{\mathbb{D}} |\xi(\lambda)| (1+|\lambda|)^{-m-1} d\lambda \le ||R(-c)^m - R_0(-c)^m||_1 < \infty$$

and is related to the scattering matrix by Equation (2.82). Furthermore, if  $f: \mathbb{R} \to \mathbb{C}$  has

two locally bounded derivatives and for some  $\varepsilon > 0$  we have

$$(\lambda^{m+1} f'(\lambda))' = O(\lambda^{-1-\varepsilon}) \tag{2.84}$$

as  $\lambda \to \infty$  then the trace formula of Equation (2.81) holds. The conclusion of Proposition 2.5.8 holds also.

In Chapter 5 we will use Equation (2.82) to determine the value of the spectral shift function as  $\lambda \to 0$  from above.

We have the following statement regarding the properties of  $\text{Det}(S(\lambda))$  due to Guillopé [73, Theorem III.1].

**Lemma 2.5.14.** Suppose that  $V = q_1q_2$  with  $q_1, q_2 \in C_c^{\infty}(\mathbb{R}^n)$ . Then for all  $\lambda > 0$  and  $p \geq \lfloor \frac{n}{2} \rfloor$  we have

 $\operatorname{Det}\left(S(\lambda)\right)$ 

$$= \frac{\operatorname{Det}_{p}\left(\operatorname{Id} + q_{1}R_{0}(\lambda - i0)q_{2}\right)}{\operatorname{Det}_{p}\left(\operatorname{Id} + q_{1}R_{0}(\lambda + i0)q_{2}\right)} \exp\left(\sum_{\ell=1}^{p-1} \frac{(-1)^{\ell}}{\ell} \operatorname{Tr}\left(\left(q_{1}R_{0}(\lambda + i0)q_{2}\right)^{\ell} - \left(q_{1}R_{0}(\lambda - i0)q_{2}\right)^{\ell}\right)\right).$$

Remark 2.5.15. If n = 1 and p = 1 or n = 2, 3 and p = 2 we only require that V satisfies Assumption 4.3.1 to obtain the statement of Lemma 2.5.14 (see [164, Section 9.1]).

We also recall the following [164, Proposition 9.1.3].

**Lemma 2.5.16.** Suppose that  $\rho$  satisfies Assumption 4.3.1. If n = 1, let  $p \ge 1$ , if n = 2, 3 let  $p \ge 2$  and if  $n \ge 4$  let  $p \ge n$ . Define for  $z \in \mathbb{C} \setminus \mathbb{R}$  the function

$$D_p(z) = \mathrm{Det}_p(\mathrm{Id} + q_1 R_0(z) q_2).$$

Then we have

$$\lim_{|z| \to \infty} D_p(z) = 1$$

uniformly in Arg(z). By the limiting absorption principle this limit extends to the positive real axis also.

Combining the results of Lemmas 2.5.14 and 2.5.16 with Theorem 2.5.13 we can determine the high-energy behaviour of the spectral shift function.

**Lemma 2.5.17.** Suppose that  $V = q_1q_2$  with  $q_1, q_2 \in C_c^{\infty}(\mathbb{R}^n)$ . If n = 1, let  $p \geq 1$ , if n = 2, 3 let  $p \geq 2$  and if  $n \geq 4$  let  $p \geq n$ . Then we have the limit

$$\lim_{\lambda \to \infty} \left( -2\pi i \xi(\lambda) + \sum_{\ell=1}^{p-1} \frac{(-1)^{\ell}}{\ell} \operatorname{Tr} \left( (q_1 R_0(\lambda + i0) q_2)^{\ell} - (q_1 R_0(\lambda - i0) q_2)^{\ell} \right) \right) = 2\pi m$$

for some  $m \in \mathbb{Z}$ .

*Proof.* Using Lemma 2.5.14 and Theorem 2.5.13 we have

$$\frac{\text{Det}_{p} \left( \text{Id} + q_{1} R_{0}(\lambda - i0) q_{2} \right)}{\text{Det}_{p} \left( \text{Id} + q_{1} R_{0}(\lambda + i0) q_{2} \right)} 
= \text{Det}(S(\lambda)) \exp \left( \sum_{\ell=1}^{p-1} \frac{(-1)^{\ell}}{\ell} \text{Tr} \left( (q_{1} R_{0}(\lambda + i0) q_{2})^{\ell} - (q_{1} R_{0}(\lambda - i0) q_{2})^{\ell} \right) \right) 
= \exp \left( -2\pi i \xi(\lambda) + \sum_{\ell=1}^{p-1} \frac{(-1)^{\ell}}{\ell} \text{Tr} \left( (q_{1} R_{0}(\lambda + i0) q_{2})^{\ell} - (q_{1} R_{0}(\lambda - i0) q_{2})^{\ell} \right) \right).$$

By Lemma 2.5.16 we have

$$\lim_{\lambda \to \infty} \frac{\operatorname{Det}_p\left(\operatorname{Id} + q_1 R_0(\lambda - i0)q_2\right)}{\operatorname{Det}_p\left(\operatorname{Id} + q_1 R_0(\lambda + i0)q_2\right)} = 1,$$

from which the result follows.

Remark 2.5.18. We can fix a branch of the meromorphic function  $\ln \text{Det}_p(\text{Id} + q_1 R_0(z) q_2)$  by the condition

$$\lim_{|z|\to\infty} \arg \operatorname{Det}_p(\operatorname{Id} + q_1 R_0(z) q_2) = 0.$$

Fixing this branch gives m=0 in Lemma 2.5.17 and we take this convention for the results of Chapter 5.

We now demonstrate some further special properties of the spectral shift function for Schrödinger operators. We enumerate the eigenvalues of H in increasing order as  $\lambda_K < \cdots < \lambda_2 < \lambda_1 \le 0$  and write  $M(\lambda_k)$  for the multiplicity of the eigenvalue  $\lambda_k$ . We also write  $N_0 = M(0)$ .

**Theorem 2.5.19** (Birman-Kreĭn). Suppose that V satisfies Assumption 2.2.14 for some  $\rho > n$ . Then for any differentiable f such that Equation (2.81) holds we have

$$\operatorname{Tr}(f(H) - f(H_0)) = \sum_{k=1}^{K} f(\lambda_k) M(\lambda_k) + f(0) (\xi(0-) - \xi(0+) - N_0)$$

$$+ \frac{1}{2\pi i} \int_0^\infty f(\lambda) \operatorname{Tr}(S(\lambda)^* S'(\lambda)) \, d\lambda.$$
(2.85)

*Proof.* Suppose that 0 is an eigenvalue of H, so that when enumerated the distinct eigenvalues of H are  $\lambda_K < \cdots < \lambda_2 < \lambda_1 = 0$ . The case when 0 is not an eigenvalue is similar.

We begin by applying Proposition 2.5.8 and the trace formula (2.81) to note that

$$\operatorname{Tr}(f(H) - f(H_0)) = \int_{\mathbb{R}} f'(\lambda)\xi(\lambda) \,d\lambda = \int_{-\infty}^{0} f'(\lambda)\xi(\lambda) \,d\lambda + \int_{0}^{\infty} f'(\lambda)\xi(\lambda) \,d\lambda$$
$$= \int_{-\infty}^{\lambda_K} f'(\lambda)\xi(\lambda) \,d\lambda + \sum_{k=2}^{K-1} \int_{\lambda_{k+1}}^{\lambda_k} f'(\lambda)\xi(\lambda) \,d\lambda + \int_{\lambda_2}^{0} f'(\lambda)\xi(\lambda) \,d\lambda$$
$$+ \int_{0}^{\infty} f'(\lambda)\xi(\lambda) \,d\lambda.$$

The spectral shift function is constant in intervals between eigenvalues by Proposition 2.5.8 and so we obtain

$$\operatorname{Tr}(f(H) - f(H_0)) = \xi(\lambda_K - )(f(\lambda_K) - f(-\infty)) + \sum_{k=2}^{K-1} (f(\lambda_k)\xi(\lambda_k - ) - f(\lambda_{k+1})\xi(\lambda_{k+1} + ))$$

$$+ (f(0)\xi(0 - ) - f(\lambda_2)\xi(\lambda_2 + )) + \int_0^\infty f'(\lambda)\xi(\lambda) \,d\lambda$$

$$= \sum_{k=2}^K f(\lambda_k)(\xi(\lambda_k - ) - \xi(\lambda_k + )) + f(0)\xi(0 - ) + \int_0^\infty f'(\lambda)\xi(\lambda) \,d\lambda$$

$$= \sum_{k=1}^K f(\lambda_k)M(\lambda_k) - f(0)N_0 + f(0)\xi(0 - ) + \int_0^\infty f'(\lambda)\xi(\lambda) \,d\lambda.$$

We note that by [164, Lemma 9.1.20] we have  $\xi|_{(0,\infty)} \in C^1((0,\infty))$  provided  $\rho > n$ . We integrate by parts and apply Equation (2.83) (valid by Theorem 2.5.13) to obtain

$$Tr(f(H) - f(H_0)) = \sum_{k=1}^{K} f(\lambda_k) M(\lambda_k) - f(0) N_0 + f(0) \xi(0-) + \int_0^{\infty} f'(\lambda) \xi(\lambda) d\lambda$$

$$= \sum_{k=1}^{K} f(\lambda_k) M(\lambda_k) - f(0) N_0 + f(0) \xi(0-)$$

$$+ (f(\infty) \xi(\infty) - f(0) \xi(0+)) + \frac{1}{2\pi i} \int_0^{\infty} f(\lambda) Tr(S(\lambda)^* S'(\lambda)) d\lambda.$$

Noting that  $f(\infty) = 0$  completes the proof.

We will discuss further in Chapter 5 how the value of  $\xi(0+)$  depends on the existence of resonances in each dimension after developing the relevant low energy expansions of the resolvent. Such results have been discussed in dimension  $n \neq 2, 4$  in [73, Theorem III.2] using perturbation determinant methods and the low energy behaviour of the resolvent discussed in Chapter 3. The Birman-Kreĭn formula can also be obtained (in odd dimensions) using the Helffer-Sjöstrand functional calculus, see [56, Theorem 3.51] for example. The Birman-Kreĭn formula can also be shown to hold in other scattering contexts, such as for massive Dirac operators in  $\mathbb{R}^3$  [51, Theorem 4.2] and for obstacle scattering [155,

Theorem 6.1]. A discussion of the spectral shift function and Birman-Kreĭn formula for scattering on asymptotically cylindrical manifolds can be found in [119]. A discussion of the spectral shift function in supersymmetric scattering can be found in [32] and [118].

#### 2.5.3 Trace class properties of the scattering matrix

In this section we describe the high-energy behaviour of the scattering operator in the trace norm using the expression for  $\text{Det}(S(\lambda))$  of Lemma 2.5.14. As a result we determine, using pseudodifferential expansions of the resolvent and the limiting absorption principle, the leading order high-energy asymptotics of the spectral shift function  $\xi$  and its derivative. These leading order asymptotics are related to the heat kernel expansion as  $t \to 0^+$  of the trace of the difference  $e^{-tH} - e^{-tH_0}$ .

To evaluate some further traces, we need to be able to integrate polynomials over  $\mathbb{S}^{n-1}$ . We use the following result [65].

**Lemma 2.5.20.** Let  $\alpha$  be a multi-index of length n and let  $P_{\alpha} : \mathbb{R}^n \to \mathbb{C}$  be given by  $P_{\alpha}(x) = x^{\alpha} = x_1^{\alpha_1} \cdots x_n^{\alpha_n}$ . Then

$$\int_{\mathbb{S}^{n-1}} P_{\alpha}(\omega) d\omega = \begin{cases} 0, & \text{if some } \alpha_j \text{ is odd,} \\ \frac{2\Gamma\left(\frac{\alpha_1+1}{2}\right)\cdots\Gamma\left(\frac{\alpha_n+1}{2}\right)}{\Gamma\left(\frac{n+|\alpha|}{2}\right)}, & \text{if all } \alpha_j \text{ are even.} \end{cases}$$

**Lemma 2.5.21.** Suppose that  $\alpha$  is a multi-index of length n,  $g \in C_c^{\infty}(\mathbb{R}^n)$  and let  $X = g\partial^{\alpha}$ , a differential operator of order  $|\alpha|$ . Then for t > 0 we have

$$\operatorname{Tr}\left(Xe^{-tH_0}\right) = \frac{(-i)^{|\alpha|}\Gamma\left(\frac{\alpha_1+1}{2}\right)\cdots\Gamma\left(\frac{\alpha_n+1}{2}\right)t^{-\frac{n+|\alpha|}{2}}}{(2\pi)^n}\left(\int_{\mathbb{R}^n}g(x)\,\mathrm{d}x\right)$$

if all  $\alpha_j$  are even and  $\operatorname{Tr}\left(Xe^{-tH_0}\right) = 0$  otherwise.

*Proof.* We compute the trace as

$$\operatorname{Tr}\left(Xe^{-tH_0}\right) = \frac{(-i)^{|\alpha|}}{(2\pi)^n} \left(\int_{\mathbb{R}^n} g(x) \, \mathrm{d}x\right) \left(\int_{\mathbb{R}^n} y^{\alpha} e^{-t|y|^2} \, \mathrm{d}y\right).$$

If any of the components  $\alpha_j$  is odd, then by changing to polar coordinates we find that  $\text{Tr}(Xe^{-tH_0}) = 0$  by Lemma 2.5.20. If all of the  $\alpha_j$  are even we obtain

$$\operatorname{Tr}\left(Xe^{-tH_0}\right) = \frac{2(-i)^{|\alpha|}\Gamma\left(\frac{\alpha_1+1}{2}\right)\cdots\Gamma\left(\frac{\alpha_n+1}{2}\right)}{(2\pi)^n\Gamma\left(\frac{n+|\alpha|}{2}\right)} \left(\int_{\mathbb{R}^n} g(x) \,\mathrm{d}x\right) \left(\int_0^\infty r^{|\alpha|+n-1}e^{-tr^2} \,\mathrm{d}r\right)$$
$$= \frac{(-i)^{|\alpha|}\Gamma\left(\frac{\alpha_1+1}{2}\right)\cdots\Gamma\left(\frac{\alpha_n+1}{2}\right)t^{-\frac{n+|\alpha|}{2}}}{(2\pi)^n} \left(\int_{\mathbb{R}^n} g(x) \,\mathrm{d}x\right),$$

where we have used polar coordinates and Lemma 2.5.20 to compute the spherical integral.

**Lemma 2.5.22.** Suppose that  $\alpha$  is a multi-index of length n,  $g_1, g_2 \in C_c^{\infty}(\mathbb{R}^n)$  and let  $g = g_1g_2$ . Let  $X = g_2\partial^{\alpha}$  be a differential operator of order  $|\alpha|$ . Fix  $\ell \in \mathbb{N}$  with  $\ell \leq \frac{n+|\alpha|}{2}$  if n is even. Then for  $\lambda > 0$  we have

$$\operatorname{Tr}\left(X\left(R_0(\lambda+i0)^{\ell}-R_0(\lambda-i0)^{\ell}\right)g_1\right)$$

$$=\frac{(-i)^{|\alpha|}(2\pi i)\Gamma\left(\frac{\alpha_1+1}{2}\right)\cdots\Gamma\left(\frac{\alpha_n+1}{2}\right)\lambda^{\frac{n+|\alpha|}{2}-\ell}}{(\ell-1)!\Gamma\left(\frac{n}{2}+1-\ell+\frac{|\alpha|}{2}\right)(2\pi)^n}\left(\int_{\mathbb{R}^n}g(x)\,\mathrm{d}x\right)$$

if all  $\alpha_j$  are even and  $\operatorname{Tr}\left(X\left(R_0(\lambda+i0)^\ell-R_0(\lambda-i0)^\ell\right)g_1\right)=0$  otherwise. If n is even and  $\ell>\frac{n+|\alpha|}{2}$  then  $\operatorname{Tr}\left(X\left(R_0(\lambda+i0)^\ell-R_0(\lambda-i0)^\ell\right)g_1\right)=0$  also.

*Proof.* Fix  $\lambda > 0$ . The integral kernel of  $B(\lambda) = (R_0(\lambda + i0) - R_0(\lambda - i0)) g_1$  is given by

$$B(\lambda, x, y) = \frac{(2\pi i)\lambda^{\frac{n}{2}-1}}{2(2\pi)^n} g_1(y) \int_{\mathbb{S}^{n-1}} e^{-i\lambda^{\frac{1}{2}}\langle \omega, y-x \rangle} d\omega.$$

Letting  $A(\lambda) = X (R_0(\lambda + i0) - R_0(\lambda - i0)) g_1$  we see by Lemma 2.4.20 that  $A(\lambda)$  has integral kernel

$$A(\lambda, x, y) = \frac{(-i)^{|\alpha|} (2\pi i) \lambda^{\frac{n+|\alpha|}{2}-1}}{2(2\pi)^n} g_2(x) g_1(y) \int_{\mathbb{S}^{n-1}} \omega_1^{\alpha_1} \cdots \omega_n^{\alpha_n} e^{-i\lambda^{\frac{1}{2}} \langle \omega, y - x \rangle} d\omega.$$

Applying now the techniques of Lemma 2.4.22 we differentiate  $(\ell - 1)$  times to find that the operator  $A_{\ell}(\lambda) = X(R_0(\lambda + i0)^{\ell} - R_0(\lambda - i0)^{\ell})g_1$  has integral kernel

$$A_{\ell}(\lambda, x, y) = \frac{(-i)^{|\alpha|} (2\pi i) \Gamma\left(\frac{n+|\alpha|}{2}\right) \lambda^{\frac{n+|\alpha|}{2}-1}}{2\Gamma\left(\frac{n}{2}+1-\ell+\frac{|\alpha|}{2}\right) (2\pi)^n} g_2(x) g_1(y) \int_{\mathbb{S}^{n-1}} \omega_1^{\alpha_1} \cdots \omega_n^{\alpha_n} e^{-i\lambda^{\frac{1}{2}} \langle \omega, y-x \rangle} d\omega + \tilde{A}(\lambda, x, y),$$

where  $\tilde{A}(\lambda,\cdot,\cdot)$  has vanishing trace. Integrating over the diagonal gives that

$$\operatorname{Tr}\left(X\left(R_0(\lambda+i0)^{\ell}-R_0(\lambda-i0)^{\ell}\right)g_1\right)$$

$$=\frac{(-i)^{|\alpha|}(2\pi i)\Gamma\left(\frac{n+|\alpha|}{2}\right)\lambda^{\frac{n+|\alpha|}{2}-\ell}}{2(\ell-1)!\Gamma\left(\frac{n}{2}+1-\ell+\frac{|\alpha|}{2}\right)(2\pi)^n}\left(\int_{\mathbb{R}^n}g(x)\,\mathrm{d}x\right)\left(\int_{\mathbb{S}^{n-1}}\omega_1^{\alpha_1}\cdots\omega_n^{\alpha_n}\,\mathrm{d}\omega\right).$$

If some  $\alpha_j$  is odd we find  $\operatorname{Tr}\left(X\left(R_0(\lambda+i0)^\ell-R_0(\lambda-i0)^\ell\right)g_1\right)=0$  by Lemma 2.5.20. If

all  $\alpha_i$  are even we have

$$\operatorname{Tr}\left(X\left(R_0(\lambda+i0)^{\ell}-R_0(\lambda-i0)^{\ell}\right)g_1\right)$$

$$=\frac{(-i)^{|\alpha|}(2\pi i)\Gamma\left(\frac{\alpha_1+1}{2}\right)\cdots\Gamma\left(\frac{\alpha_n+1}{2}\right)\lambda^{\frac{n+|\alpha|}{2}-\ell}}{(\ell-1)!\Gamma\left(\frac{n}{2}+1-\ell+\frac{|\alpha|}{2}\right)(2\pi)^n}\left(\int_{\mathbb{R}^n}g(x)\,\mathrm{d}x\right),$$

again by Lemma 2.5.20.

For the next statement, we introduce for  $m \in \mathbb{N} \cup \{0\}$  and  $f \in C_c^{\infty}(\mathbb{R}^n)$  the notation  $f^{(m)} = [H_0, [H_0, [\cdots, [H_0, f] \cdots]]]$  (where the expression has m commutators). Note that  $f^{(m)}$  is a differential operator of order m.

**Lemma 2.5.23.** Suppose that  $q_1, q_2 \in C_c^{\infty}(\mathbb{R}^n)$  with  $V = q_1q_2$ . Then for all  $z \in \mathbb{C} \setminus \mathbb{R}$ ,  $\ell \in \mathbb{N}$  and  $K \in \mathbb{N} \cup \{0\}$  we have

$$(q_1 R_0(z) q_2)^{\ell} = q_1 \left( \sum_{|k|=0}^K (-1)^{|k|+1} C_{\ell-1}(k) V^{(k_1)} \cdots V^{(k_{\ell-1})} R_0(z)^{\ell+|k|} \right) q_2 + q_1 P_{K,\ell}(z) q_2,$$
(2.86)

where  $P_{K,\ell}(z)$  is of order at most  $-2\ell - K - 1$ , k is a multi-index of length  $(\ell - 1)$  and

$$C_{\ell}(k) = \frac{(|k| + \ell)!}{k_1! k_2! \cdots k_{\ell}! (k_1 + 1)(k_1 + k_2 + 2) \cdots (|k| + \ell)}.$$

When  $\ell = 1$  we have no remainder term. For all  $z \in \mathbb{C} \setminus \mathbb{R}$ ,  $M \in \mathbb{N}$  and  $K \in \mathbb{N} \cup \{0\}$  we have

$$R(z) - R_0(z) = \sum_{m=1}^{M} \left( \sum_{|k|=0}^{K} (-1)^{m+|k|} C_m(k) V^{(k_1)} \cdots V^{(k_m)} R_0(z)^{m+|k|+1} + P_{K,m}(z) \right)$$

$$+ (-1)^{M+1} (R_0(z)V)^{M+1} R(z),$$
(2.87)

where  $P_{K,m}(z)$  has order (at most) -2m - K - 3 and k is a multi-index of length m.

*Proof.* Equation (2.86) follows from the pseudodifferential calculus of [46, Lemma 6.11]. For Equation (2.87), we write

$$R(z) - R_0(z) = \sum_{m=1}^{M} (-1)^m (R_0(z)V)^m R_0(z) + (-1)^{M+1} (R_0(z)V)^{M+1} R(z).$$

Applying again [46, Lemma 6.11] we have

$$R(z) - R_0(z) = \sum_{m=1}^M \left( \sum_{|k|=0}^K (-1)^{m+|k|} C_m(k) V^{(k_1)} \cdots V^{(k_m)} R_0(z)^{m+|k|+1} + P_{K,m}(z) \right) + (-1)^{M+1} (R_0(z)V)^{M+1} R(z),$$

where  $P_{K,m}(z)$  has order (at most) -2m - K - 3.

Since  $V \in C_c^{\infty}(\mathbb{R}^n)$  and for  $\ell \in \mathbb{N}$  and k a multi-index of length  $\ell$  we have  $V^{(k_1)} \cdots V^{(k_\ell)}$  is a differential operator of order |k| with smooth compactly supported coefficients, we can write

$$V^{(k_1)} \cdots V^{(k_\ell)} = \sum_{|r|=0}^{|k|} g_{k,r} \partial^r, \qquad (2.88)$$

with r a multi-index with n components and  $g_{r,k} \in C_c^{\infty}(\mathbb{R}^n)$ . With this notation, we obtain the following heat kernel expansion due to Colin de Verdière [160] in odd dimensions (see also [56, Theorem 3.64] for a general proof and [15] for explicit computations of some coefficients). We provide a detailed proof using the pseudodifferential expansion of Lemma 2.5.23 and the trace formulae of Lemma 2.5.21. As a result we obtain an expression for the coefficients which differs from [56, Theorem 3.64] or [15], although is equivalent. This new expression is necessary for a comparison with the high-energy behaviour of the scattering operator, as we show in Theorem 2.5.29.

**Proposition 2.5.24.** Suppose that  $V \in C_c^{\infty}(\mathbb{R}^n)$ . Then for any  $J \in \mathbb{N}$  we have the expansion

Tr 
$$\left(e^{-tH} - e^{-tH_0}\right) = \sum_{j=1}^{J} a_j(n, V) t^{j-\frac{n}{2}} + E_J(t),$$

where  $E_J(t) = O\left(t^{J+1-\frac{n}{2}}\right)$  as  $t \to 0^+$ . For  $j \in \mathbb{N}$  we define the set

$$Q_{M,K}(j) = \left\{ (m, k, r) \in \{0, 1, \dots, M\} \times \{0, 1, \dots, K\}^m \times \{0, 1, \dots, K\}^n : |r| \le |k|, \right.$$
all  $r_j$  are even and  $m + |k| + 1 - \frac{|r|}{2} = j \right\}.$ 

The constants  $a_i(n, V)$  are given by

$$a_{j}(n,V) = \sum_{(m,k,r)\in Q_{M,K}(j)} \frac{(-i)^{|r|}C_{m}(k)(-1)^{m+|k|+1}\Gamma\left(\frac{r_{1}+1}{2}\right)\cdots\Gamma\left(\frac{r_{n}+1}{2}\right)}{(2\pi)^{n}(m+1)(m+|k|)!} \left(\int_{\mathbb{R}^{n}} V(x)g_{k,r}(x) dx\right),$$

where we have used the notation of Equation (2.88).

*Proof.* Fix t > 0. By [56, Theorem 3.64] we have  $e^{-tH} - e^{-tH_0} \in \mathcal{L}^1(\mathcal{H})$ . Choose a > 0 such that  $a > |\lambda|$  for all  $\lambda \in \sigma_p(H)$  and define the vertical line  $\gamma_t = \{-a - \frac{1}{t} + iv : v \in \mathbb{R}\}$ . We write

$$e^{-tH} - e^{-tH_0} = \int_0^1 \frac{d}{ds} e^{-t(H_0 + sV)} ds = -t \int_0^1 V e^{-t(H_0 + sV)} ds.$$

Cauchy's integral theorem tells us that for all  $s \in [0,1]$  we have

$$e^{-t(H_0+sV)} = \frac{1}{2\pi i} \int_{\gamma_t} e^{-tz} (z - H_0 - sV)^{-1} dz = -\frac{1}{2\pi i} \int_{\gamma_t} e^{-tz} R_s(z) dz,$$

where we have introduced the notation  $R_s(z) = (H_0 + sV - z)^{-1}$ . An application of Lemma 2.5.23 gives us for all  $M, K \in \mathbb{N} \cup \{0\}$  the expansion

$$R_s(z) = \sum_{m=0}^{M} \left( \sum_{|k|=0}^{K} (-1)^{m+|k|} s^m C_m(k) V^{(k_1)} \cdots V^{(k_m)} R_0(z)^{m+|k|+1} + s^m P_{K,m}(z) \right) + (-1)^{M+1} s^{M+1} (R_0(z)V)^{M+1} R_s(z).$$

Thus we write

$$\operatorname{Tr}\left(Ve^{-t(H_0+sV)}\right) = -\operatorname{Tr}\left(\frac{1}{2\pi i} \int_{\gamma_t} e^{-tz} R_s(z) dz\right)$$

$$= -\operatorname{Tr}\left(\frac{1}{2\pi i} \int_{\gamma_t} e^{-tz} \sum_{m=0}^M \sum_{|k|=0}^K (-1)^{m+|k|} s^m C_m(k) V V^{(k_1)} \cdots V^{(k_m)} R_0(z)^{m+|k|+1} dz\right)$$

$$-\operatorname{Tr}\left(\frac{1}{2\pi i} \int_{\gamma_t} e^{-tz} \sum_{m=0}^M s^m V P_{K,m}(z) dz - \frac{1}{2\pi i} \int_{\gamma_t} e^{-tz} (-1)^{M+1} s^{M+1} V (R_0(z)V)^{M+1} R_s(z) dz\right)$$

$$:= -\operatorname{Tr}\left(\frac{1}{2\pi i} \int_{\gamma_t} e^{-tz} \sum_{m=0}^M \sum_{|k|=0}^K (-1)^{m+|k|} s^m C_m(k) V V^{(k_1)} \cdots V^{(k_m)} R_0(z)^{m+|k|+1} dz\right)$$

$$+ e_{M,K}(t,s).$$

We now apply Cauchy's integral formula again to obtain

$$\frac{1}{2\pi i} \int_{\gamma_t} e^{-tz} R_0(z)^{m+|k|+1} dz = (-1)^{m+|k|+1} \frac{1}{2\pi i} \int_{\gamma_t} e^{-tz} (z - H_0)^{-m-|k|-1} dz 
= \frac{(-1)^{m+|k|+1}}{(m+|k|)!} \frac{d^{m+|k|}}{dz^{m+|k|}} \left(e^{-tz}\right)|_{z=H_0} = -\frac{t^{m+|k|}}{(m+|k|)!} e^{-tH_0}.$$

Hence we find

$$\operatorname{Tr}\left(Ve^{-t(H_0+sV)}\right) = \sum_{m=0}^{M} \sum_{|k|=0}^{K} (-1)^{m+|k|} s^m \frac{C_m(k)}{(m+|k|)!} t^{m+|k|} \operatorname{Tr}\left(VV^{(k_1)} \cdots V^{(k_m)} e^{-tH_0}\right) + e_{M,K}(t,s).$$

Integrating out the s variable we have

$$\operatorname{Tr}\left(e^{-tH} - e^{-tH_0}\right) = -t \int_0^1 \operatorname{Tr}\left(Ve^{-t(H_0 + sV)}\right) ds$$

$$= \sum_{m=0}^M \sum_{|k|=0}^K \frac{(-1)^{m+|k|+1} C_m(k)}{(m+1)(m+|k|)!} t^{m+|k|+1} \operatorname{Tr}\left(VV^{(k_1)} \cdots V^{(k_m)} e^{-tH_0}\right)$$

$$- t \int_0^1 e_{M,K}(t,s) ds.$$

We now use Lemma 2.5.21 to write

$$\sum_{m=0}^{M} \sum_{|k|=0}^{K} \frac{(-1)^{m+|k|+1} C_m(k)}{(m+1)(m+|k|)!} t^{m+|k|+1} \operatorname{Tr} \left( V V^{(k_1)} \cdots V^{(k_m)} e^{-tH_0} \right) 
= \sum_{m=0}^{M} \sum_{|k|=0}^{K} \sum_{|r|=0}^{|k|} \frac{(-i)^{|r|} (-1)^{\ell+m+|k|} C_m(k) \Gamma\left(\frac{r_1+1}{2}\right) \cdots \Gamma\left(\frac{r_n+1}{2}\right) t^{m+|k|+1-\frac{n+|r|}{2}}}{(m+1)(m+|k|)! (2\pi)^n} 
\times \left( \int_{\mathbb{R}^n} V(x) g_{k,r}(x) \, \mathrm{d}x \right),$$

where the sum is over all multi-indices r of length n such that all  $r_j$  are even. We now collect together powers of t. For  $j \in \mathbb{N}$  we define the set

$$Q_{M,K}(j) = \left\{ (m, k, r) \in \{0, 1, \dots, M\} \times \{0, 1, \dots, K\}^m \times \{0, 1, \dots, K\}^n : |r| \le |k|, \right.$$
 all  $r_j$  are even and  $m + |k| + 1 - \frac{|r|}{2} = j \right\}$ 

and the coefficients

$$a_{j}(n,V) = \sum_{\substack{(m,k,r) \in Q_{M,K}(j)}} \frac{(-i)^{|r|}(-1)^{m+|k|+1}C_{m}(k)\Gamma\left(\frac{r_{1}+1}{2}\right)\cdots\Gamma\left(\frac{r_{n}+1}{2}\right)}{(m+1)(m+|k|)!(2\pi)^{n}} \left(\int_{\mathbb{R}^{n}} V(x)g_{k,r}(x) dx\right).$$

The  $a_i(n, V)$  allow us to write

$$\sum_{m=0}^{M} \sum_{|k|=0}^{K} \frac{(-1)^{m+|k|+1} C_m(k)}{(m+1)(m+|k|)!} t^{m+|k|+1} \operatorname{Tr} \left( V V^{\{k_1\}} \cdots V^{\{k_m\}} e^{-tH_0} \right) = \sum_{j=1}^{M+1} a_j(n,V) t^{-\frac{n}{2}+j} + G(t),$$

where  $G(t) = O(t^{M+2-\frac{n}{2}})$  as  $t \to 0^+$ . Thus we have the expansion

$$\operatorname{Tr}\left(e^{-tH} - e^{-tH_0}\right) = \sum_{j=1}^{M+1} a_j(n, V) t^{-\frac{n}{2}+j} + G(t) - t \int_0^1 e_{M,K}(t, s).$$

It remains to check the behaviour of the final remainder term, which consists of two types of terms. Recall that the operator  $P_{K,m}(z)$  has order (at most) -2m - K - 3. By [46, Lemma 6.12] there exists a constant C (depending on V, m and K but not z) such that

$$\left| \left| R_0(z)^{-m-\frac{K}{2}-\frac{1}{2}} q_2 P_{K,m}(z) \right| \right| \le C.$$

Choose K = n - m, large enough so that  $q_1 R_0(z)^{m + \frac{K}{2} + \frac{1}{2}}$  defines a trace-class operator. So we use Hölder's inequality for Schatten classes to find

$$||VP_{K,m}(z)||_1 \le C \left| \left| q_1 R_0(z)^{m + \frac{K}{2} + \frac{1}{2}} \right| \right|_1.$$

We can now estimate the remainder term via

$$\left\| \int_{0}^{1} s^{m} \int_{\gamma_{t}} e^{-tz} P_{K,m}(z) dz ds \right\|_{1} \leq \frac{C}{m+1} \int_{\mathbb{R}} e^{at+1} \left\| q_{1} R_{0}(z)^{m+\frac{K}{2}+\frac{1}{2}} \right\|_{1} dv$$

$$\leq \tilde{C} e^{at+1} \int_{\mathbb{R}} \left( \left( a + \frac{1}{t} \right)^{2} + v^{2} \right)^{-\frac{m}{2} - \frac{K}{4} - \frac{1}{4} + \frac{n}{4}} dv.$$

Make the substitution  $v = (a + \frac{1}{t}) w$  to find

$$\left\| \int_{0}^{1} s^{m} \int_{\gamma_{t}} e^{-tz} P_{K,m}(z) dz ds \right\|_{1} \leq \tilde{C} e^{at+1} \int_{\mathbb{R}} \left( \left( a + \frac{1}{t} \right)^{2} + v^{2} \right)^{-\frac{m}{2} - \frac{K}{4} - \frac{1}{4} + \frac{n}{4}} dv$$

$$= \tilde{C} e^{at} \left( a + \frac{1}{t} \right)^{-m - \frac{K}{2} - \frac{1}{2} + \frac{n}{2}} \int_{\mathbb{R}} (1 + w^{2})^{-\frac{m}{2} - \frac{K}{4} - \frac{1}{4} + \frac{n}{4}} dw$$

$$\leq C e^{at} t^{m + \frac{K}{2} + \frac{1}{2} - \frac{n}{2}} \int_{\mathbb{R}} (1 + w^{2})^{-\frac{m}{2} - \frac{K}{4} - \frac{1}{4} + \frac{n}{4}} dw$$

$$= O\left( t^{m + \frac{K+1}{2} - \frac{n}{2}} \right)$$

as  $t \to 0+$ . A similar estimate shows that

$$\left\| \int_0^1 s^{M+1} \int_{\gamma_t} V(R_0(z)V)^{M+1} R_s(z) e^{-tz} dz ds \right\|_1 = O\left(t^{M+2-\frac{n}{2}}\right)$$

as  $t \to 0^+$ . Taking J = M + 1 completes the proof.

Remark 2.5.25. The heat kernel coefficients  $a_j(n, V)$  are well-known and have been computed for small j by many authors, see for example [15] and [160] in this context. The

first few are

$$a_1(n, V) = -\frac{\Gamma\left(\frac{n}{2}\right) \operatorname{Vol}(\mathbb{S}^{n-1})}{2(2\pi)^n} \int_{\mathbb{R}^n} V(x) \, \mathrm{d}x,$$

$$a_2(n, V) = \frac{\Gamma\left(\frac{n}{2}\right) \operatorname{Vol}(\mathbb{S}^{n-1})}{4(2\pi)^n} \int_{\mathbb{R}^n} V(x)^2 \, \mathrm{d}x,$$

$$a_3(n, V) = -\frac{\Gamma\left(\frac{n}{2}\right) \operatorname{Vol}(\mathbb{S}^{n-1})}{6(2\pi)^n} \int_{\mathbb{R}^n} \left(V(x)^3 + \frac{1}{2} |[\nabla V](x)|^2\right) \, \mathrm{d}x.$$

The expression for the  $a_j(n, V)$  provided by Proposition 2.5.24 provides a systematic way of computing these coefficients.

Truncating the heat kernel expansion of Proposition 2.5.24 at  $J = \lfloor \frac{n}{2} \rfloor$  we can write for t > 0 the expression

$$\operatorname{Tr}\left(e^{-tH} - e^{-tH_0}\right) = \sum_{j=1}^{\lfloor \frac{n}{2} \rfloor} a_j(n, V) t^{j-\frac{n}{2}} + E_{\lfloor \frac{n}{2} \rfloor}(t)$$

$$= \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} \frac{a_j(n, V)}{\Gamma\left(\frac{n}{2} - j\right)} \int_0^\infty \lambda^{\frac{n}{2} - j - 1} e^{-t\lambda} d\lambda + \frac{(-1)^n + 1}{2} a_{\lfloor \frac{n}{2} \rfloor}(n, V) + E_{\lfloor \frac{n}{2} \rfloor}(t),$$
(2.89)

where we have separated out the constant term in even dimensions as

$$\beta_n(V) = \frac{(-1)^n + 1}{2} a_{\lfloor \frac{n}{2} \rfloor}(n, V). \tag{2.90}$$

**Definition 2.5.26.** We define the high-energy polynomial for  $\xi'$  to be  $p_n:(0,\infty)\to\mathbb{C}$  given for  $\lambda\in(0,\infty)$  by

$$p_n(\lambda) = \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_j(n,V) \lambda^{\frac{n}{2}-j-1} := \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} \frac{(2\pi i) a_j(n,V)}{\Gamma\left(\frac{n}{2}-j\right)} \lambda^{\frac{n}{2}-j-1}.$$

By [144, Theorem 1.2] the high-energy polynomial  $p_n$  is related to the spectral shift function by  $\lim_{\lambda \to \infty} (-2\pi i \xi'(\lambda) - p_n(\lambda)) = 0$ .

We can explicitly determine  $p_n$  for small n as  $p_1 = p_2 = 0$ ,

$$p_3(\lambda) = -\frac{(2\pi i)\lambda^{-\frac{1}{2}} \text{Vol}(\mathbb{S}^2)}{4(2\pi)^3} \int_{\mathbb{R}^3} V(x) \, \mathrm{d}x = \frac{(2\pi i)\lambda^{-\frac{1}{2}} a_1(3, V)}{\Gamma\left(\frac{1}{2}\right)},$$
$$p_4(\lambda) = -\frac{(2\pi i) \text{Vol}(\mathbb{S}^3)}{2(2\pi)^4} \int_{\mathbb{R}^4} V(x) \, \mathrm{d}x = (2\pi i)a_1(4, V).$$

We can use Proposition 2.5.24 and the Birman-Kreĭn trace formula to analyse the integrability of the (derivative of) the spectral shift function on  $\mathbb{R}^+$ .

**Lemma 2.5.27.** If n=1,2,3 suppose that V satisfies Assumption 4.3.1. If  $n\geq 4$  suppose that  $V\in C_c^\infty(\mathbb{R}^n)$ . Then the function  $\mathrm{Tr}\,(S(\cdot)^*S'(\cdot))-p_n$  is integrable on  $\mathbb{R}^+$ . In particular, if n=1,2 we have  $\mathrm{Tr}\,(S(\cdot)^*S'(\cdot))\in L^1(\mathbb{R}^+)$ .

Proof. Since  $\xi|_{(0,\infty)} \in C^1((0,\infty))$  by [164, Lemma 9.1.20] it suffices to check integrability in a neighbourhood of zero and a neighbourhood of infinity. The lowest power in the high-energy polynomial  $p_n$  is  $\lambda^{-\frac{1}{2}}$  and thus  $p_n$  is integrable in a neighbourhood of zero. That  $\xi'$  is integrable in a neighbourhood of zero is the statement of [90, Theorem 5.2]. This can also be proved directly by using the resolvent expansions of [88, 85, 87, 89] and Equation (2.73) to analyse the small  $\lambda$  behaviour of  $\text{Tr}(S(\lambda)^*S'(\lambda))$  as in [52, Lemma 5.1].

For the claim regarding the integrability in a neighbourhood of infinity, we use the high-energy asymptotics of [132, Theorem 1]. The result is that as  $\lambda \to \infty$  we have the expansion

$$\operatorname{Tr}\left(S(\lambda)^* S'(\lambda)\right) - p_n(\lambda) \sim \sum_{j=\lfloor \frac{n}{2} \rfloor + 1}^{\infty} c_j(n, V) \lambda^{\frac{n}{2} - j - 1}. \tag{2.91}$$

We note also that there is no coefficient of  $\lambda^{-1}$  in even dimensions by [90, Theorem 5.3]. The expansion (2.91) implies that for sufficiently large  $\lambda$  we have the estimate

$$|\operatorname{Tr}(S(\lambda)^*S'(\lambda)) - p_n(\lambda)| \le C\lambda^{\frac{n}{2} - \lfloor \frac{n}{2} \rfloor - 2},$$

from which the integrability of  $\operatorname{Tr}(S(\cdot)^*S'(\cdot)) - p_n$  in a neighbourhood of infinity follows.

We now analyse the high-energy behaviour of the (determinant of the) scattering matrix using the limiting absorption principle and the pseudodifferential expansion of Lemma 2.5.23. To do so we define the following high-energy polynomial.

**Definition 2.5.28.** We define the high-energy polynomial for  $\xi$  to be  $P_n:(0,\infty)\to\mathbb{C}$  given for  $\lambda\in(0,\infty)$  by

$$P_n(\lambda) = 2\pi i \beta_n(V) + \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} \frac{c_j(n,V)}{\frac{n}{2} - j} \lambda^{\frac{n}{2} - j} = \sum_{j=1}^{\lfloor \frac{n}{2} \rfloor} \frac{(2\pi i)a_j(n,V)}{\Gamma\left(\frac{n}{2} - j + 1\right)} \lambda^{\frac{n}{2} - j}.$$

We note also that  $P'_n = p_n$ , with  $p_n$  the high-energy polynomial for  $\xi'$  of Definition 2.5.26.

We can explicitly compute the lowest order polynomials, finding  $P_1 = 0$ ,

$$P_{2}(\lambda) = -\frac{(2\pi i)\operatorname{Vol}(\mathbb{S}^{1})}{2(2\pi)^{2}} \int_{\mathbb{R}^{2}} V(x) \, \mathrm{d}x = -\frac{2\pi i}{4\pi} \int_{\mathbb{R}^{2}} V(x) \, \mathrm{d}x = (2\pi i)\beta_{2}(V),$$

$$P_{3}(\lambda) = -\frac{(2\pi i)\lambda^{\frac{1}{2}}\operatorname{Vol}(\mathbb{S}^{2})}{2(2\pi)^{3}} \int_{\mathbb{R}^{3}} V(x) \, \mathrm{d}x = -\frac{(2\pi i)\lambda^{\frac{1}{2}}}{4\pi^{2}} \int_{\mathbb{R}^{3}} V(x) \, \mathrm{d}x = \frac{(2\pi i)a_{1}(3,V)\lambda^{\frac{1}{2}}}{\Gamma\left(\frac{3}{2}\right)},$$

$$P_{4}(\lambda) = -\frac{(2\pi i)\lambda\operatorname{Vol}(\mathbb{S}^{3})}{2(2\pi)^{4}} \int_{\mathbb{R}^{4}} V(x) \, \mathrm{d}x + \frac{(2\pi i)\operatorname{Vol}(\mathbb{S}^{3})}{4(2\pi)^{4}} \int_{\mathbb{R}^{4}} V(x)^{2} \, \mathrm{d}x.$$

**Theorem 2.5.29.** Suppose that  $q_1, q_2 \in C_c^{\infty}(\mathbb{R}^n)$  with  $V = q_1q_2$ . Then for all  $\lambda > 0$  and  $J \in \mathbb{N}$  we have

$$\sum_{\ell=1}^{J} \left( \frac{(-1)^{\ell}}{\ell} \operatorname{Tr} \left( (q_1 R_0 (\lambda + i0) q_2)^{\ell} - (q_1 R_0 (\lambda - i0))^{\ell} q_2 \right) \right) = -\sum_{j=1}^{J} C_j(n, V) \lambda^{\frac{n}{2} - j} + E_J(\lambda),$$
(2.92)

where  $E_J(\lambda) = O\left(\lambda^{\frac{n}{2}-J-3}\right)$  as  $\lambda \to \infty$  and  $E_J$  is differentiable. If n is even we have  $C_j(n,V) = 0$  for all  $j > \frac{n}{2}$ . If n is even and  $j \leq \frac{n}{2}$  or n is odd the coefficients  $C_j(n,V)$  are given by

$$C_j(n,V) = \frac{(2\pi i)a_j(n,V)}{\Gamma\left(\frac{n}{2} - j + 1\right)},\tag{2.93}$$

with the  $a_j(n, V)$  the heat kernel coefficients of Proposition 2.5.24. Note also that for n even we have  $C_{\frac{n}{2}}(n, V) = 2\pi i \beta_n(V)$ .

*Proof.* For  $\lambda > 0$ , we have

$$\operatorname{Tr}\left(q_{1}\left(R_{0}(\lambda+i0)-R_{0}(\lambda-i0)\right)q_{2}\right) = \frac{(2\pi i)\lambda^{\frac{n-2}{2}}\operatorname{Vol}(\mathbb{S}^{n-1})}{2(2\pi)^{n}}\int_{\mathbb{R}^{n}}V(x)\,\mathrm{d}x$$
$$= -\frac{(2\pi i)a_{1}(n,V)\lambda^{\frac{n}{2}-1}}{\Gamma\left(\frac{n}{2}\right)}.$$

by Lemma 2.4.20, with no need for any kind of expansion. We now consider the  $\ell \geq 2$  terms in the sum. For  $z \in \mathbb{C} \setminus \mathbb{R}$  we use Lemma 2.5.23 to obtain the expansion

$$\frac{(-1)^{\ell}}{\ell} (q_1 R_0(z) q_2)^{\ell} 
= \left( \frac{(-1)^{\ell}}{\ell} q_1 \left( \sum_{|k|=0}^K (-1)^{|k|+1} C_{\ell-1}(k) V^{(k_1)} \cdots V^{(k_{\ell-1})} R_0(z)^{\ell+|k|} \right) q_2 + \frac{(-1)^{\ell}}{\ell} q_1 P_{K,\ell}(z) q_2 \right),$$

where  $P_{K,\ell}(z)$  is of order (at most)  $-2\ell - K - 1$ . By the limiting absorption principle,

this equality extends to  $z = \lambda \pm i0$  for  $\lambda \in (0, \infty)$  and thus we find

$$\frac{(-1)^{\ell}}{\ell} \operatorname{Tr} \left( (q_{1}(R_{0}(\lambda + i0)q_{2})^{\ell} - (q_{1}R_{0}(\lambda - i0))q_{2})^{\ell} \right) 
= \operatorname{Tr} \left( q_{1} \left( \sum_{|k|=0}^{K} \frac{(-1)^{\ell+|k|+1}}{\ell} C_{\ell-1}(k) V^{(k_{1})} \cdots V^{(k_{\ell-1})} \left( R_{0}(\lambda + i0)^{\ell+|k|} - R_{0}(\lambda - i0)^{\ell+|k|} \right) \right) q_{2} 
+ \frac{(-1)^{\ell}}{\ell} q_{1} \left( P_{K,\ell}(\lambda + i0) - P_{L,\ell}(\lambda - i0) \right) q_{2} \right).$$

We now use the expansion of Equation (2.88) to write

$$\frac{(-1)^{\ell}}{\ell} \operatorname{Tr} \left( q_{1} (R_{0}(\lambda + i0)q_{2})^{\ell} - (q_{1}R_{0}(\lambda - i0)q_{2})^{\ell} \right) 
= \operatorname{Tr} \left( q_{1} \left( \sum_{|k|=0}^{K} \sum_{|r|=0}^{|k|} \frac{(-1)^{\ell+|k|+1}}{\ell} C_{\ell-1}(k) q_{1} g_{k,r} \partial^{r} \left( R_{0}(\lambda + i0)^{\ell+|k|} - R_{0}(\lambda - i0)^{\ell+|k|} \right) \right) q_{2} 
+ \frac{(-1)^{\ell}}{\ell} q_{1} \left( P_{L,\ell}(\lambda + i0) - P_{L,\ell}(\lambda - i0) \right) q_{2} \right).$$

Ignoring for a moment the remainder term we find using Lemma 2.4.22 that

$$\operatorname{Tr}\left(\frac{(-1)^{\ell}}{\ell}q_{1}\left(\sum_{|k|=0}^{K}\sum_{|r|=0}^{|k|}(-1)^{|k|+1}C_{\ell-1}(k)g_{k,r}\partial^{r}\left(R_{0}(\lambda+i0)^{\ell+|k|}-R_{0}(\lambda-i0)^{\ell+|k|}\right)\right)q_{2}\right)$$

$$=\sum_{|k|=0}^{K}\sum_{|r|=0}^{|k|}\frac{(-1)^{\ell+|k|}(2\pi i)C_{\ell-1}(k)(-i)^{|r|}\Gamma\left(\frac{r_{1}+1}{2}\right)\cdots\Gamma\left(\frac{r_{n}+1}{2}\right)\lambda^{\frac{n+|r|}{2}-\ell-|k|}}{\ell(\ell+|k|-1)!\Gamma\left(\frac{n}{2}+1-\ell-|k|+\frac{|r|}{2}\right)(2\pi)^{n}}\int_{\mathbb{R}^{n}}V(x)g_{k,r}(x)\,\mathrm{d}x.$$

Lemma 2.4.22 also gives that for n even all terms with  $\ell + |k| - \frac{|r|}{2} > \frac{n}{2}$  vanish. We now collect together the powers of  $\lambda$ . For  $j \in \mathbb{N}$  we define the set

$$Q_{L,K}(j) = \left\{ (\ell, k, r) \in \{1, \dots, L\} \times \{0, \dots, K\}^{\ell} \times \{0, \dots, K\}^n : |r| \le |k| \le K, \right\}$$
 all  $r_j$  are even, and  $\ell + |k| - \frac{|r|}{2} = j \right\}$ 

So we write

$$\sum_{\ell=2}^{L} \operatorname{Tr} \left( \frac{(-1)^{\ell}}{\ell} q_1 \left( \sum_{|k|=0}^{K} \sum_{|r|=0}^{|k|} (-1)^{|k|+1} C_{\ell-1}(k) q_1 g_{r,k} \partial^r \left( R_0 (\lambda + i0)^{\ell+|k|} - R_0 (\lambda - i0)^{\ell+|k|} \right) \right) q_2 \right) \\
= -\sum_{j=2}^{J} C_j(n, V) \lambda^{\frac{n}{2} - j},$$

where the coefficients  $C_i(n, V)$  are given by

$$C_{j}(n,V) = \sum_{(\ell,k,r)\in Q_{L,K}(j)} \frac{(-1)^{\ell+|k|}(2\pi i)C_{\ell-1}(k)(-i)^{|r|}\Gamma\left(\frac{r_{1}+1}{2}\right)\cdots\Gamma\left(\frac{r_{n}+1}{2}\right)}{\ell(\ell+|k|-1)!\Gamma\left(\frac{n}{2}+1-\ell-|k|+\frac{|r|}{2}\right)(2\pi)^{n}} \int_{\mathbb{R}^{n}} V(x)g_{r,k}(x) dx.$$

Direct comparison with Proposition 2.5.24 shows that we have the relation

$$C_j(n,V) = \frac{(2\pi i)a_j(n,V)}{\Gamma\left(\frac{n}{2} + 1 - j\right)}.$$

We now return to the remainder term. Due to the form of the remainder terms  $P_{K,\ell}(\lambda \pm i0)$  the difference  $q_1\left(P_{K,\ell}(\lambda + i0) - P_{K,\ell}(\lambda - i0)\right)q_2$  is always trace-class, as we now show. The proof of [46, Lemma 6.12] shows that  $P_{K,\ell}(\lambda \pm i0)$  is a linear combination of terms of the form

$$\prod_{m=1}^{M} A_m R_0 (\lambda \pm i0)^{\alpha_m},$$

for some  $M, \alpha_m \in \mathbb{N}$  and differential operators  $A_m$  of order  $a_m < 2\alpha_m$  with smooth compactly supported coefficients and

$$\sum_{m=1}^{M} \left( \alpha_m - \frac{a_m}{2} \right) \ge -2\ell - L - 1.$$

Each  $A_m$  can be factored as  $f_m \tilde{A}_m g_m$  for  $f_m, g_m \in C_c^{\infty}(\mathbb{R}^n)$  and  $\tilde{A}_m$  also of order  $a_m$ . Taking the difference and factorising we find

$$\left(\prod_{m=1}^{M} A_m R_0 (\lambda + i0)^{\alpha_m} - \prod_{m=1}^{M} A_m R_0 (\lambda - i0)^{\alpha_m}\right) \\
= \sum_{p=1}^{M} \left(\prod_{m < p} A_m R_0 (\lambda - i0)^{\alpha_m}\right) (A_p (R_0 (\lambda + i0)^{\alpha_p} - R_0 (\lambda - i0)^{\alpha_p})) \left(\prod_{m > p} A_m R_0 (\lambda + i0)^{\alpha_m}\right),$$

each term of which is trace-class by Lemma 2.4.22. Hölder's inequality for the trace norm then shows that

$$\left\| q_{1} \left( \prod_{m=1}^{M} A_{m} R_{0} (\lambda + i0)^{\alpha_{m}} - \prod_{m=1}^{M} A_{m} R_{0} (\lambda - i0)^{\alpha_{m}} \right) q_{2} \right\|_{1}$$

$$\leq \sum_{p=1}^{M} \left\| q_{1} f_{1} \left( \prod_{m < p} \tilde{A}_{m} g_{m} R_{0} (\lambda - i0)^{\alpha_{m}} f_{m+1} \right) \right\| \left\| \tilde{A}_{p} g_{p} \left( R_{0} (\lambda + i0)^{\alpha_{p}} - R_{0} (\lambda - i0)^{\alpha_{p}} \right) f_{p+1} \right\|_{1}$$

$$\times \left\| \tilde{A}_{p+1} g_{p+1} \left( \prod_{p < m \leq M} R_{0} (\lambda + i0)^{\alpha_{m}} f_{m+1} \right) q_{2} \right\|,$$

where we use the convention  $f_{M+1} = 1$ . By [3, Theorem A.1] (see also [120, Theorem 1]) we have for any  $\alpha > 0$  and differential operator A of order  $a < 2\alpha$  the estimate

$$||q_1 A R_0(\lambda \pm i0)^{\alpha} q_2|| = O\left(\lambda^{-\frac{1}{2}\alpha + \frac{a}{4}}\right)$$

as  $\lambda \to \infty$ . By Lemma 2.5.22 we have the estimate

$$||q_1 A (R_0(\lambda + i0)^{\alpha} - R_0(\lambda - i0)^{\alpha}) q_2||_1 \le C \lambda^{\frac{n}{2} - \alpha + \frac{a}{2}}.$$

Combining these we obtain

$$\left\| q_1 \left( \prod_{m=1}^{M} q_1 A_m q_2 R_0 (\lambda + i0)^{\alpha_m} - \prod_{m=1}^{M} q_1 A_m q_2 R_0 (\lambda - i0)^{\alpha_m} \right) q_2 \right\|_1$$

$$= C \sum_{p=1}^{M} \lambda^{\frac{n}{2} - \alpha_p + \frac{a_p}{2}} \left( \prod_{m \neq p} O\left(\lambda^{-\frac{1}{2}\alpha_m + \frac{a_m}{4}}\right) \right) \le O\left(\lambda^{\frac{n}{2} - \ell - \frac{L+1}{2}}\right)$$

as  $\lambda \to \infty$ . By varying K we obtain Equation (2.92). That  $E_L$  is differentiable follows from the observation that the left-hand side of Equation (2.92) is differentiable, as is each power of  $\lambda$  on the right-hand side.

Remark 2.5.30. Choosing  $J \ge \lfloor \frac{n}{2} \rfloor$  in Theorem 2.5.29 gives

$$\sum_{\ell=1}^{J} \left( \frac{(-1)^{\ell}}{\ell} \operatorname{Tr} \left( (q_1 R_0(\lambda + i0) q_2)^{\ell} - (q_1 R_0(\lambda - i0) q_2)^{\ell} \right) \right) = -P_n(\lambda) - E(\lambda),$$

with  $E(\lambda) = O(\lambda^{-\frac{1}{2}})$  as  $\lambda \to \infty$  with E differentiable. A comparison with Lemma 2.5.17 and using the convention of Remark 2.5.18 gives the relation

$$\lim_{\lambda \to \infty} \left( -2\pi i \xi(\lambda) - P_n(\lambda) \right) = 0,$$

justifying the terminology of  $P_n$  being the high-energy polynomial for  $\xi$ .

#### 2.5.4 Levinson's theorem

Levinson's original theorem [108] gives, in one dimension, a relation between the number of bound states of a system and a quantity related to the scattering data of the system. The original proof makes use of the Jost functions [91] and relies heavily on complex analysis. Such techniques are specific to one dimension and not easily generalisable to higher dimensions, except in the case of spherically symmetric potentials which can essentially be reduced to a one dimensional problem, see [123, Section 5] and [137, Theorem XI.59]. We will present in Chapter 5 a proof based based on the index of the wave operator, which will

allow us to obtain analogous statements in higher dimensions. Statements of Levinson's theorem require slight modification when certain obstructions, called resonances, exist at zero energy. The nature of these obstructions depends on the parity of the dimension, and they cannot occur for dimension  $n \geq 5$ . We will develop the required machinery to motivate definitions of resonances in Chapter 3, however for now we make reference to them without providing a precise definition.

**Theorem 2.5.31** (Levinson's theorem in one dimension). Let n = 1 and suppose that V satisfies Assumption 2.2.14 for some  $\rho > 2$ . Then the number of bound states (eigenvalues counted with multiplicity) N of the operator  $H = H_0 + V$  is given by

$$-N = \frac{1}{2\pi i} \int_0^\infty \text{Tr}(S(\lambda)^* S'(\lambda)) \,d\lambda + \frac{1}{2} \left( M_R(0) - 1 \right), \tag{2.94}$$

where  $M_R(0) = 1$  if there exists a resonance or  $H = H_0$ , and  $M_R(0) = 0$  otherwise.

Levinson's theorem is more complicated in even dimensions due to the nature of the logarithmic singularity of the perturbed resolvent  $R(z) = (H - z)^{-1}$  near z = 0.

**Theorem 2.5.32** (Levinson's theorem in two dimensions). Let n=2 and suppose that V satisfies Assumption 2.2.14 for some  $\rho > 11$ . Then the number of bound states (eigenvalues counted with multiplicity) of  $H = H_0 + V$  is given by

$$-N = \frac{1}{2\pi i} \int_0^\infty \text{Tr}(S(\lambda)^* S'(\lambda)) \,d\lambda + M_p(0) + \frac{1}{4\pi} \int_{\mathbb{R}^2} V(x) \,dx,$$
 (2.95)

where  $M_p(0) \in \{0, 1, 2\}$  is the number of p-resonances.

Theorem 2.5.32 was originally proved in [26, Theorem 6.3], although the proof is rather complicated and requires a good deal of supplementary arguments from the reader. In Chapter 5 we provide a new proof based on the index of the wave operator in the case of no p-resonances. The extra term involving the integral of the potential in Equation (2.95) is related to the high energy behaviour of the scattering matrix via

$$\frac{1}{2\pi i} \operatorname{Log}(\operatorname{Det}(S(\infty))) = \frac{1}{4\pi} \int_{\mathbb{R}^2} V(x) \, \mathrm{d}x.$$
 (2.96)

The intricate behaviour of the scattering matrix at infinity plays a role in Levinson's theorem in all dimensions and is rather subtle. The fact that the term on the right hand side of Equation (2.96) does not appear under the  $\lambda$  integral with  $\text{Tr}(S(\lambda)^*S'(\lambda))$  is peculiar to even dimensions.

In three dimensions (and all higher dimensions), the large  $\lambda$  behaviour of the scattering matrix means that the function  $\text{Tr}(S(\cdot)^*S'(\cdot))$  is no longer integrable (see Theorem 2.5.34) and thus we require additional correction terms to Levinson's theorem. Such correction

terms are a feature for all  $n \geq 3$ , however quickly become very difficult to compute (see for example [73, Section V], [143, Theorem 1.2] and [56, Section 3.11]). The following statement is due to [31] and can be deduced also from [124] under the assumption that the spectral shift function belongs to  $C^1(\mathbb{R}^+)$ , although as in the dimension n = 2 case a good deal of additional argument from the reader is required to make the proofs work.

**Theorem 2.5.33** (Levinson's theorem in three dimensions). Let n=3 and suppose that V satisfies Assumption 2.2.14 for some  $\rho > 5$ . Then the number of bound states (eigenvalues counted with multiplicity) of  $H = H_0 + V$  is given by

$$-N = \frac{1}{2\pi i} \int_0^\infty \left( \text{Tr}(S(\lambda)^* S'(\lambda))) - c_1(3, V) \lambda^{-\frac{1}{2}} \right) d\lambda + \frac{1}{2} M_R(0),$$

where  $M_R(0) = 1$  if there is a zero-energy resonance and  $M_R(0) = 0$  otherwise. The constant  $c_1$  is given by

$$c_1(3, V) = -\frac{i}{4\pi} \int_{\mathbb{R}^3} V(x) \, \mathrm{d}x.$$

A higher dimensional version of Levinson's theorem can be obtained from the Birman-Kreĭn trace formula with knowledge of heat asymptotics, although as in lower dimensions requires knowledge of the spectral shift function at zero. We have the following, due in odd dimensions to Guillopé [73, Theorem IV.5] (see also [56, Theorem 3.66]). We provide only a sketch of the proof as there are many subleties.

**Theorem 2.5.34.** [Levinson's theorem in higher dimensions] Suppose that  $V \in C_c^{\infty}(\mathbb{R}^n)$ . Then the number of bound states of  $H = H_0 + V$  is given by

$$-N = \frac{1}{2\pi i} \int_0^\infty \left( \operatorname{Tr}(S(\lambda)^* S'(\lambda)) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_j(n, V) \lambda^{\frac{n-2}{2} - j} \right) d\lambda - N_0$$
$$- (\xi(0+) - \xi(0-)) - \beta_n(V),$$

where the  $c_j(n, V)$  and  $\beta_n(V)$  are constants depending on V and its derivatives with  $b_n = 0$  for n odd.

*Proof sketch.* From Proposition 2.5.24 we have the limit

$$0 = \lim_{t \to 0^+} \left( \text{Tr} \left( e^{-tH} - e^{-tH_0} \right) - \sum_{j=1}^{\lfloor \frac{n}{2} \rfloor} a_j(n, V) t^{k - \frac{n}{2}} \right).$$

Using Equation (2.89) and the Birman-Krein trace formula in the form of Theorem 2.5.19

we obtain

$$0 = \lim_{t \to 0^{+}} \left( \frac{1}{2\pi i} \int_{0}^{\infty} e^{-t\lambda} \operatorname{Tr}\left(S(\lambda)^{*}S'(\lambda)\right) d\lambda + \sum_{k=1}^{K} e^{-t\lambda_{k}} M(\lambda_{k}) \right)$$

$$- N - \xi(0+) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} \frac{a_{j}(n,V)}{\Gamma\left(\frac{n}{2}-k\right)} \int_{0}^{\infty} \lambda^{\frac{n}{2}-k-1} e^{-t\lambda} d\lambda - \beta_{n}(V) d\lambda \right)$$

$$= \lim_{t \to 0^{+}} \left( \frac{1}{2\pi i} \int_{0}^{\infty} e^{-t\lambda} \left( \operatorname{Tr}\left(S(\lambda)^{*}S'(\lambda)\right) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} \frac{(2\pi i)a_{j}(n,V)\lambda^{\frac{n}{2}-k-1}}{\Gamma\left(\frac{n}{2}-k\right)} d\lambda \right) d\lambda + \sum_{k=1}^{K} e^{-t\lambda_{k}} M(\lambda_{k}) - N - \xi(0+) - \beta_{n}(V) d\lambda - \xi(0+) - \beta_{n}(V).$$

Here the constant  $\beta_n(V)$  is as in Equation (2.90). An application of Lemma 2.5.27 and the dominated convergence theorem then allows us to bring the limit as  $t \to 0^+$  inside the integral to obtain

$$\xi(0+) = \frac{1}{2\pi i} \int_0^\infty \left( \text{Tr} \left( S(\lambda)^* S'(\lambda) \right) - p_n(\lambda) \right) d\lambda - \beta_n(V).$$

The key element missing in Theorem 2.5.34 is the difference  $\xi(0+) - \xi(0-)$  of the spectral shift function from the left and right at zero. The value  $\xi(0-) = -N + N_0$  is determined by Theorem 2.5.13 and thus the only unknown element in statements of Levinson's theorem is the value  $\xi(0+)$ , which we determine in Chapter 5 to be dependent on the existence of resonances. In fact we can rewrite Levinson's theorem as

$$\xi(0+) = \frac{1}{2\pi i} \int_0^\infty \left( \operatorname{Tr}(S(\lambda)^* S'(\lambda)) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_j(n, V) \lambda^{\frac{n}{2} - j - 1} \right) d\lambda - \beta_n(V). \tag{2.97}$$

We shall not have occasion to use any higher order terms, although they can be computed if necessary (see [56, Theorem 3.64] for more details). These coefficients will return in Chapter 5 when we study high energy corrections to the scattering operator and in Chapter 6 when we study spectral flow.

An interesting point to note is that in all even dimensions, Levinson's theorem has an additional term outside of the integral expression. This additional term is given in low dimensions by

$$\beta_2(V) = -\frac{\text{Vol}(\mathbb{S}^1)}{2(2\pi)^2} \int_{\mathbb{R}^2} V(x) \, dx,$$

$$\beta_4(V) = \frac{\text{Vol}(\mathbb{S}^3)}{4(2\pi)^4} \int_{\mathbb{R}^4} V(x)^2 \, dx,$$

$$\beta_6(V) = -\frac{\text{Vol}(\mathbb{S}^5)}{6(2\pi)^6} \int_{\mathbb{R}^6} \left( V(x)^3 + \frac{1}{2} |[\nabla V](x)|^2 \right) \, dx.$$

Here we have used the coefficients  $c_i(n, V)$  to determine the  $\beta_n(V)$ .

### 2.5.5 Examples

In this section we demonstrate some explicit low dimensional examples, explicitly computing the scattering operator and describing the bound states and any anomalous behaviour.

In dimension n=1, we see that for fixed energy  $\lambda \in \mathbb{R}^+$  we have the unitary scattering matrix  $S(\lambda) \in \mathcal{B}(L^2(\mathbb{S}^0)) = M_2(\mathbb{C})$ . In our definition of  $\mathcal{U}$  we have made a choice of basis of  $\mathbb{C}^2$ . Some authors use a different convention which leads to different forms of the scattering matrix. Recall that  $\mathcal{U}: L^2(\mathbb{R}) \to L^2(\mathbb{R}^+, \mathbb{C}^2)$  is unitary and defined by

$$[\mathcal{U}f](r) = 2^{-\frac{1}{2}} \lambda^{-\frac{1}{2}} \begin{pmatrix} f(\lambda^{\frac{1}{2}}) \\ f(-\lambda^{\frac{1}{2}}) \end{pmatrix}.$$

Some authors prefer to use the map  $\mathcal{V}: L^2(\mathbb{R}) \to L^2(\mathbb{R}^+, \mathbb{C}^2)$  defined by

$$[\mathcal{V}f](r) = 2^{-\frac{1}{2}} \lambda^{-\frac{1}{2}} \begin{pmatrix} \frac{f(\lambda^{\frac{1}{2}}) + f(-\lambda^{\frac{1}{2}})}{2} \\ \frac{f(\lambda^{\frac{1}{2}}) - f(-\lambda^{\frac{1}{2}})}{2} \end{pmatrix}.$$

The operator  $\mathcal{V}$  is unitary and related to  $\mathcal{U}$  by the change of basis,

$$B = 2^{-\frac{1}{2}} \begin{pmatrix} 1 & 1 \\ 1 & -1 \end{pmatrix}.$$

which satisfies  $B^{-1}\mathcal{U}(\lambda,\cdot)B = \mathcal{V}(\lambda,\cdot)$ .

The first example we consider is the finite square well in dimension n=1.

Example 2.5.35. We fix  $E_0, L > 0$  and define the function  $V : \mathbb{R} \to \mathbb{R}$  by

$$V(x) = \begin{cases} 0, & \text{if } |x| > L, \\ -E_0, & \text{if } |x| \le L. \end{cases}$$

The potential V is a step function which is only non-zero inside the region [-L, L] with depth  $-E_0$ . We note that V satisfies Assumption 2.2.14 for any  $\rho > 1$  and thus all of

our results apply. In particular, the operator  $H = H_0 + V$  has finitely many non-positive eigenvalues.

Recall the change of basis matrix  $B = 2^{-\frac{1}{2}} \begin{pmatrix} 1 & 1 \\ 1 & -1 \end{pmatrix}$  from the standard polar decomposition of  $L^2(\mathbb{R}^+, \mathbb{C}^2)$  into the basis provided by the operator  $\mathcal{V}$ . The scattering matrix is given explicitly by

$$B^{-1}S(\lambda)B = -e^{-2i\lambda^{\frac{1}{2}}L} \begin{pmatrix} \frac{\left(\frac{\lambda+E_{0}}{\lambda}\right)^{\frac{1}{2}}\tan\left((\lambda+E_{0})^{\frac{1}{2}}L\right) - i}{\left(\frac{\lambda+E_{0}}{\lambda}\right)^{\frac{1}{2}}\tan\left((\lambda+E_{0})^{\frac{1}{2}}L\right) + i} & 0\\ 0 & \frac{-\left(\frac{\lambda}{\lambda+E_{0}}\right)^{\frac{1}{2}}\tan\left((\lambda+E_{0})^{\frac{1}{2}}L\right) + i}{\left(\frac{\lambda}{\lambda+E_{0}}\right)^{\frac{1}{2}}\tan\left((\lambda+E_{0})^{\frac{1}{2}}L\right) + i} \end{pmatrix}.$$

Due to the symmetry of the potential V, the expression for S is much nicer when decomposed according to  $\mathcal{V}$  rather than  $\mathcal{U}$ . An interesting point to note is that the poles of  $S(\cdot)$  lie on the imaginary axis and correspond precisely to the bound states of H, see [19] for a numerical discussion of the behaviour of such poles.

We note that if  $E_0^{\frac{1}{2}}L \neq m\pi$  for any  $m \in \mathbb{Z}$  and  $E_0^{\frac{1}{2}}L \neq (m+\frac{1}{2})\pi$  for any  $m \in \mathbb{Z}$  then we find Det(S(0)) = -1 and we call this the generic case, which corresponds to the non-existence of resonances. See Theorem 4.1.3 for more details.

If there exists  $m \in \mathbb{Z}$  such that  $E_0^{\frac{1}{2}}L = m\pi$  or  $E_0^{\frac{1}{2}}L = (m + \frac{1}{2})\pi$  then Det(S(0)) = 1, which is atypical and corresponds to the existence of resonances.

Remark 2.5.36. Example 2.5.35 can be extended to higher dimensions in a straightforward manner, decomposing the scattering matrix into spherical harmonics. The explicit terms are however rather difficult to describe, involving ratios of Bessel functions.

We next consider another one dimensional example in which scattering quantities can be computed explicitly, known as the Pöschl-Teller potential.

Example 2.5.37 (Pöschl-Teller potential). For fixed  $\alpha \in \mathbb{R}$  and  $\beta > 0$  we define the potential

$$V(x) = -\frac{\alpha^2 \beta(\beta - 1)}{\cosh^2(\alpha x)}.$$

For  $k \geq 0$  solutions to the equation  $(H_0 + V)\psi = -k^2\psi$  can be determined explicitly in terms of hypergeometric functions (see [49, Equation 9]). Note that if  $\beta = 0, 1$  then we have V = 0. For  $\beta \neq 0, 1$  the scattering matrix can be determined explicitly (see [49, Equations (14) and (15)]) as

$$B^{-1}S(k^2)B = \begin{pmatrix} -\frac{\Gamma(1+ik)\Gamma(\beta-ik)\Gamma(1-\beta-ik)}{\Gamma(1-ik)\Gamma(\beta)} & \frac{\Gamma(\beta-ik)\Gamma(1-\beta-ik)}{\Gamma(1-ik)\Gamma(-ik)} \\ \frac{\Gamma(1+ik)\Gamma(ik)}{\Gamma(1+ik-\beta)\Gamma(ik+\beta)} - \frac{\Gamma(1+ik)\Gamma(ik)\Gamma(\beta-ik)\Gamma(1-\beta-ik)}{\Gamma(\beta)^2\Gamma(1-\beta)^2} & \frac{\Gamma(ik)\Gamma(\beta-ik)\Gamma(1-\beta-ik)}{\Gamma(-ik)\Gamma(1-\beta)\Gamma(\beta)} \end{pmatrix}$$

One can compute directly that

$$B^{-1}S(0)B = \begin{pmatrix} -1 & 0\\ 0 & 1, \end{pmatrix}$$

provided that  $\beta \notin \mathbb{Z}$ .

 $\triangle$ 

The above examples demonstrate that even simple potentials V can result in extremely complicated scattering behaviour and bound state solutions. In fact for most potentials the Schrödinger equation cannot be solved analytically, the list of exactly solveable potentials is short (see [4] and [35] for some limited examples).

Example 2.5.38. Suppose that  $\varphi \in C^{\infty}(\mathbb{R})$  is such that  $\varphi(\pm \infty) = c_{\pm}$  for some  $c_{\pm} \in \mathbb{R} \setminus \{0\}$  with  $c_{+}^{2} + c_{-}^{2} = 1$  and  $\varphi(x) \neq 0$  for all  $x \in \mathbb{R}$ . Then we define  $V = \frac{\varphi''}{\varphi}$  and compute that

$$(H_0 + V)\varphi = 0,$$

whilst  $\varphi \notin L^2(\mathbb{R})$ . Such a function will be called a zero energy resonance in Chapter 3. In this case one can show directly (see [56, Remark on p. 46]) that

$$S(0) = \begin{pmatrix} 2c_{+}c_{-} & c_{+}^{2} - c_{-}^{2} \\ c_{-}^{2} - c_{+}^{2} & 2c_{1}c_{2} \end{pmatrix}.$$

Direct computation shows that Det(S(0)) = 1.

 $\triangle$ 

# Chapter 3

# Resolvent expansions and resonances

In this chapter we develop low energy expansions of operators related to the resolvents  $R_0(z)$  and R(z). The expansions we consider here follow the technique of Jensen and Nenciu [89] and can be found throughout the literature. In particular, there are no new results in this chapter. In dimension  $n \leq 4$ , such expansions are complicated by analytic obstructions to invertibility at zero, which give rise to definitions of zero-energy resonances. These resolvent expansions can be used to analyse the behaviour of the scattering matrix at zero-energy and the form of the wave operator in 4.0.1, which allows us to proveide new proofs of Levinson's theorem.

We decompose the potential V into V = vUv where  $v(x) = |V(x)|^{\frac{1}{2}}$  and U(x) = sign(V). Such a factorisation is well-known and was first used in the work of Birman [24] and Schwinger [148]. For  $z \in \mathbb{C} \setminus \mathbb{R}$  with  $\text{Im}(z^{\frac{1}{2}}) > 0$ , we let  $k = -iz^{\frac{1}{2}}$ . We have the symmetrised resolvent identity

$$R(-k^{2}) - R_{0}(-k^{2}) = -R_{0}(-k^{2})v(U + vR_{0}(-k^{2})v)^{-1}vR_{0}(-k^{2})$$
  
=:  $-R_{0}(-k^{2})vM(k)^{-1}vR_{0}(-k^{2}).$  (3.1)

Thus we see to obtain expansions of the resolvent  $R(-k^2)$  it suffices to consider the known expansions of the operators  $R_0(-k^2)$  and (the inverse of)

$$M(k) := U + vR_0(-k^2)v. (3.2)$$

The analysis of the inverse of M(k) as  $k \to 0$  is sensitive to the fine structure of the spectrum of H near zero. In particular the presence of zero energy resonances will force multiple uses of the Feshbach inversion method of Lemma 3.1.1. We shall find that such obstructions cannot occur for  $n \geq 5$ . Also the inversion of M(k) is sensitive to the parity of the dimension, due to the fact that the integral kernel of the free resolvent has a logarithmic singularity near k = 0 in even dimensions. As such, our analysis is split into six different cases. We begin with the easiest cases of  $n \geq 5$  odd to demonstrate the use

of Lemma 3.1.1 and end with the most involved cases of dimension n=2 and n=1.

Throughout this chapter we will have to strengthen the decay assumptions on the potential to guarantee that the operators we consider are bounded. The most general assumption we shall need is the following.

**Assumption 3.0.1.** We suppose that the potential satisfies Assumption 2.2.14 for

- 1.  $\rho > \frac{n+1}{2}$  if  $n \ge 11$ ;
- 2.  $\rho > 7$  if n = 7, 8, 9, 10;
- 3.  $\rho > 7$  if n = 5, 6;
- 4.  $\rho > 12$  if n = 4;
- 5.  $\rho > 5$  if n = 3;
- 6.  $\rho > 11$  if n = 2; and
- 7.  $\rho > 7$  if n = 1.

We do not claim that Assumption 3.0.1 gives the optimal decay required for the existence of the expansions in each case. For ease of exposition, we prefer to give in each dimension only a sufficient value of  $\rho$  required for all expansions to exist. More optimal results can be found in the cited references in each section.

The expansions we obtain for the operator M(k) are summarised in the following result.

**Theorem 3.0.2.** Suppose that V satisfies Assumption 3.0.1. Then for sufficiently small k and with  $\eta = \frac{1}{\ln(k)}$  we have the expansion

$$M(k)^{-1} = -k^{-2}A + k^{-2}g(k)A_n + k^{-1}h(k)B_n + \tilde{R}(k),$$

where  $A_n, B_n = 0$  for  $n \geq 5$ ,  $A_n = 0$  for n = 1, 3,  $g(k), h(k) = O(\eta^{-1})$ ,  $A, A_n, B_n$  are bounded operators and  $\tilde{R}(k)$  is uniformly bounded in k.

For ease of exposition, we have used the phrase 'for sufficiently small k ...' as shorthand for the statement 'there exists  $k_0 > 0$  such that for all  $0 < |k| < k_0 ...$ '. The coefficient operators  $A_n$  and  $B_n$  are related to the presence of analytic obstructions to invertibility, called resonances. Their behaviour is dimension dependent, with a single type of resonance occurring in dimensions n = 1, 3, 4 and two distinct types occurring in dimension n = 2. Despite the fact that there is only one type, resonances occurring in dimension n = 4 behave more similarly to those in dimension n = 2 due to the nature of the obstruction and the logarithmic singularity of the free resolvent.

## 3.1 Algebraic and analytic preliminaries

The starting point for our analysis is the following inversion formulae for particular operator-valued matrices (see [89, Section 2]).

**Lemma 3.1.1.** Let A be a closed operator on  $\mathcal{H}$ , Q a projection and suppose A + Q has a bounded inverse. Then A has a bounded inverse if and only if

$$B = Q - Q(A+Q)^{-1}Q$$

has a bounded inverse in QH. In this case  $A^{-1}$  is given by

$$C = (A+Q)^{-1} + (A+Q)^{-1}QB^{-1}Q(A+Q)^{-1}.$$
(3.3)

*Proof.* Suppose that B has a bounded inverse and let C be as in Equation (3.3). Then we can check the relations

$$AC = (A + Q - Q)C$$

$$= \operatorname{Id} + QB^{-1}Q(A + Q)^{-1} - Q((A + Q)^{-1} + (A + Q)^{-1}QB^{-1}Q(A + Q)^{-1})$$

$$= \operatorname{Id} + (Q - B - Q(A + Q)^{-1}Q)B^{-1}Q(A + Q)^{-1}$$

$$= \operatorname{Id}.$$

A similar calculation shows that CA = Id so that A has a bounded inverse given by  $A^{-1} = C$ . Next suppose that A has a bounded inverse. Then the operator

$$D = (A+Q)A^{-1}(A+Q) - (A+Q) = Q(\mathrm{Id} + A^{-1})Q$$

defines a bounded operator on  $Q\mathcal{H}$ . We then compute that

$$BD = (Q - Q(A + Q)^{-1}Q) Q(\operatorname{Id} + A^{-1})Q$$

$$= Q (\operatorname{Id} - (A + Q)^{-1} + A^{-1} - (A + Q)^{-1}QA^{-1}) Q$$

$$= Q (\operatorname{Id} - (A + Q)^{-1} + A^{-1} - A^{-1} + (A + Q)^{-1}) Q$$

$$= Q,$$

the identity on  $Q\mathcal{H}$ . A similar calculation shows that DB = Q and thus the operator B has a bounded inverse on  $Q\mathcal{H}$  given by  $B^{-1} = D$ .

**Corollary 3.1.2.** Let  $F \subset \mathbb{C}$  have zero as an accumulation point. Let A(z),  $z \in F$ , be a family of bounded operators of the form

$$A(z) = A_0 + zA_1(z) (3.4)$$

with  $A_1(z)$  uniformly bounded as  $z \to 0$  and  $A_0$  self-adjoint. Suppose 0 is an isolated point of the spectrum of  $A_0$ , and let Q be the orthogonal projection onto  $Ker(A_0)$ . Then for sufficiently small  $z \in F$  the operator  $B(z) : Q\mathcal{H} \to Q\mathcal{H}$  defined by

$$B(z) = z^{-1}(Q - Q(A(z) + Q)^{-1}Q) = \sum_{j=0}^{\infty} (-1)^j z^j Q[A_1(z)(A_0 + Q)^{-1}]^{j+1}Q$$

is uniformly bounded as  $z \to 0$ . The operator A(z) has a bounded inverse in  $\mathcal{H}$  if and only if B(z) has a bounded inverse in  $Q\mathcal{H}$ . In this case

$$A(z)^{-1} = (A(z) + Q)^{-1} + z^{-1}(A(z) + Q)^{-1}QB(z)^{-1}Q(A(z) + Q)^{-1}$$

*Proof.* By construction the operator  $A_0 + Q$  is invertible. Define  $D_0 = (A_0 + Q)^{-1}$ . For sufficiently small z the operator  $A_1(z)$  satisfies  $||zA_1(z)D_0|| < 1$  and so we can compute the Neumann expansion

$$(A(z) + Q)^{-1} = (A_0 + Q + zA_1(z))^{-1} = D_0(\operatorname{Id} + zA_1(z)D_0)^{-1}$$
$$= D_0 \sum_{j=0}^{\infty} (-1)^j z^j (A_1(z)D_0)^j.$$

Since Q is the projection onto  $Ker(A_0)$  and  $A_0$  is self-adjoint, we find  $Q = Q(A_0 + Q)D_0 = QD_0$  and so by taking adjoints  $D_0Q = Q$  also. We can thus compute

$$B(z) = z^{-1}(Q - Q(A(z) + Q)^{-1}Q) = z^{-1} \left( Q - Q \left( D_0 \sum_{j=0}^{\infty} (-1)^j z^j (A_1(z)D_0)^j \right) Q \right)$$

$$= z^{-1} \left( Q - QD_0 Q - QD_0 \left( \sum_{j=1}^{\infty} (-1)^j z^j (A_1(z)D_0)^j \right) Q \right)$$

$$= \sum_{j=0}^{\infty} (-1)^j z^j Q[A_1(z)(A_0 + Q)^{-1}]^{j+1} Q.$$

Since  $A_1(z)$  is uniformly bounded as  $z \to 0$  we see that B(z) is also. The fact that A(z) has a bounded inverse if and only if B(z) has a bounded inverse given by

$$A(z)^{-1} = (A(z) + Q)^{-1} + z^{-1}(A(z) + Q)^{-1}QB(z)^{-1}Q(A(z) + Q)^{-1}$$

now follows from Lemma 3.1.1.

The next lemma is a form of what is known as the Feshbach formula.

**Lemma 3.1.3.** Let A be an operator-valued matrix on the direct sum of Hilbert spaces

 $\mathcal{H} = \mathcal{H}_1 \oplus \mathcal{H}_2$ ,

$$A = \begin{pmatrix} a_{11} & a_{12} \\ a_{21} & a_{22} \end{pmatrix}$$

with  $a_{ij}: \mathcal{H}_j \to \mathcal{H}_i$ . Suppose that  $a_{11}$  and  $a_{22}$  are closed and  $a_{12}$  and  $a_{21}$  are bounded. Suppose  $a_{22}$  has a bounded inverse. Then A has a bounded inverse if and only if

$$B = (a_{11} - a_{12}a_{22}^{-1}a_{21})^{-1}$$

exists and is bounded. Furthermore we have

$$A^{-1} = \begin{pmatrix} B & -Ba_{12}a_{22}^{-1} \\ -a_{22}^{-1}a_{21}B & a_{22}^{-1}a_{21}Ba_{12}a_{22}^{-1} + a_{22}^{-1} \end{pmatrix}.$$

*Proof.* Suppose that B exists and is bounded and let

$$C = \begin{pmatrix} B & -Ba_{12}a_{22}^{-1} \\ -a_{22}^{-1}a_{21}B & a_{22}^{-1}a_{21}Ba_{12}a_{22}^{-1} + a_{22}^{-1} \end{pmatrix}.$$

Then we can compute directly that

$$AC = \begin{pmatrix} a_{11} & a_{12} \\ a_{21} & a_{22} \end{pmatrix} \begin{pmatrix} B & -Ba_{12}a_{22}^{-1} \\ -a_{22}^{-1}a_{21}B & a_{22}^{-1}a_{21}Ba_{12}a_{22}^{-1} + a_{22}^{-1} \end{pmatrix}$$

$$= \begin{pmatrix} a_{11}B - a_{12}a_{22}^{-1}a_{21}B & -a_{11}Ba_{12}a_{22}^{-1} + a_{12}a_{22}^{-1}a_{21}Ba_{12}a_{22}^{-1} + a_{12}a_{22}^{-1} \\ a_{21}B - a_{21}B & -a_{21}Ba_{12}a_{22}^{-1} + a_{21}Ba_{12}a_{22}^{-1} + \operatorname{Id} \end{pmatrix}.$$

We can then calculate each term individually, finding first

$$(AC)_{11} = a_{11}B - a_{12}a_{22}^{-1}a_{21}B = (a_{11} - a_{12}a_{22}^{-1}a_{21})B = \mathrm{Id}_{\mathcal{H}_1}.$$

Next we have

$$(AC)_{12} = -a_{11}Ba_{12}a_{22}^{-1} + a_{12}a_{22}^{-1}a_{21}Ba_{12}a_{22}^{-1} + a_{12}a_{22}^{-1}$$

$$= -a_{11}Ba_{12}a_{22}^{-1} + a_{11}Ba_{12} - (a_{11} - a_{12}a_{22}^{-1}a_{21})Ba_{12}a_{22}^{-1} + a_{12}a_{22}^{-1}$$

$$= -a_{11}Ba_{12}a_{22}^{-1} + a_{11}Ba_{12} - a_{12}a_{22}^{-1} + a_{12}a_{22}^{-1}$$

$$= 0$$

We see immediately that  $(AC)_{21} = 0$  and  $(AC)_{22} = \mathrm{Id}_{\mathcal{H}_2}$ . A similar calculation shows that  $CA = \mathrm{Id}_{\mathcal{H}}$  and thus A has a bounded inverse given by C. For the converse, we

suppose that A has a bounded inverse given by

$$C = \begin{pmatrix} c_{11} & c_{12} \\ c_{21} & c_{22} \end{pmatrix}.$$

We now use the relations AC = CA = Id to demonstrate the existence of the operator B. The relation AC = Id gives us the four equations

$$a_{11}c_{11} + a_{12}c_{21} = \operatorname{Id}_{\mathcal{H}_1}, \quad a_{11}c_{12} + a_{12}c_{22} = 0,$$
 (3.5)

$$a_{21}c_{11} + a_{22}c_{21} = 0, , \quad a_{21}c_{12} + a_{22}c_{22} = \mathrm{Id}_{\mathcal{H}_2}.$$
 (3.6)

The invertibility of  $a_{22}$  and Equation (3.6) then immediately gives us  $c_{21} = -a_{22}^{-1}a_{21}c_{11}$ . Combining with Equation (3.5) we find

$$\operatorname{Id}_{\mathcal{H}_1} = a_{11}c_{11} + a_{12}c_{21} = a_{11}c_{11} - a_{12}a_{22}^{-1}a_{21}c_{11} = (a_{11} - a_{12}a_{22}^{-1}a_{21})c_{11}$$

so that  $a_{11} - a_{12}a_{22}^{-1}a_{21}$  is a left inverse for  $c_{11}$ . Similarly the relation  $CA = \operatorname{Id}_{\mathcal{H}}$  gives us the four equations

$$c_{11}a_{11} + c_{12}a_{21} = \operatorname{Id}_{\mathcal{H}_1}, \quad c_{21}a_{11} + c_{22}a_{21} = 0,$$
 (3.7)

$$c_{11}a_{12} + c_{12}a_{22} = 0, \quad c_{21}a_{12} + c_{22}a_{22} = \mathrm{Id}_{\mathcal{H}_2}.$$
 (3.8)

The invertibility of  $a_{22}$  and Equation (3.8) then immediately gives us  $c_{12} = -c_{11}a_{12}a_{22}^{-1}$ . Combining with Equation (3.7) we find

$$\operatorname{Id}_{\mathcal{H}_1} = c_{11}a_{11} + c_{12}a_{21} = c_{11}a_{11} - c_{11}a_{12}a_{22}^{-1}a_{21} = c_{11}(a_{11} - a_{12}a_{22}^{-1}a_{21}),$$

so that  $a_{11} - a_{12}a_{22}^{-1}a_{21}$  is a right inverse of  $c_{11}$  also. Thus we have shown  $c_{11} = (a_{11} - a_{12}a_{22}^{-1}a_{21})^{-1} = B$  exists and defines a bounded operator.

We shall now give a broad overview of how the above algebraic results will be used. If the operator M(k) is of the form  $M(k) = A_0 + kA_1(k)$  with  $A_0$  self-adjoint and not invertible and  $A_1(k)$  uniformly bounded as  $k \to 0$ , define the orthogonal projection  $Q_1$  onto the kernel of  $A_0$ . Then for sufficiently small k we have  $||kA_1(k)|| < 1$  and thus we can compute the inverse of  $M(k) + Q_1 = A_0 + Q_1 + kA_1(k)$  via a Neumann expansion. Then by Lemma 3.1.1 the operator M(k) is invertible if and only if

$$B_1(k) = Q_1 - Q_1(M(k) + Q_1)^{-1}Q_1$$

is invertible. If  $B_1(k)$  can be written as  $B_1(k) = A_2 + kA_3(k)$  with  $A_2$  invertible, then for sufficiently small k we can invert  $B_1(k)$  via a Neumann expansion and find the inverse of

M(k) as

$$M(k)^{-1} = (M(k) + Q_1)^{-1} + (M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}.$$

If  $B_1(k)$  has self-adjoint leading term  $A_2$  which is not invertible, we repeat the procedure by defining  $Q_2$  to be the orthogonal projection onto the kernel of  $A_2$  so that the operator  $B_1(k)$  is invertible if and only if

$$B_2(k) = Q_2 - Q_2(B_1(k) + Q_2)^{-1}Q_2$$

is invertible. Write  $B_2(k) = A_4 + kA_5(k)$ . If the leading term  $A_4$  of  $B_2(k)$  is invertible, we can again use (for sufficiently small k, taking a new  $k_0$  if necessary) a Neumann expansion to obtain

$$B_1(k)^{-1} = (B_1(k) + Q_2)^{-1} + (B_1(k) + Q_2)^{-1}Q_2B_2(k)^{-1}Q_2(B_1(k) + Q_2)^{-1}.$$

Substituting back into the formula for  $M(k)^{-1}$  we obtain

$$M(k)^{-1} = (M(k) + Q_1)^{-1} + (M(k) + Q_1)^{-1}Q_1(B_1(k) + Q_2)^{-1}Q_1(M(k) + Q_1)^{-1} + (M(k) + Q_1)^{-1}Q_1(B_1(k) + Q_2)^{-1}Q_2B_2(k)^{-1}Q_2(B_1(k) + Q_2)^{-1}Q_1(M(k) + Q_1)^{-1}.$$

If the self-adjoint leading term  $A_4$  of  $B_2(k)$  is not invertible, we repeat the procedure by defining the orthogonal projection  $Q_3$  onto the kernel of  $A_4$ . Due to general resolvent estimates (see [89]), the procedure terminates after a finite number of steps. In fact, we shall show that at most three projections are required (this occurs in the case n = 2) and the final inversion formula takes the form

$$\begin{split} M(k)^{-1} &= (M(k) + Q_1)^{-1} + (M(k) + Q_1)^{-1}Q_1(B_1(k) + Q_2)^{-1}Q_1(M(k) + Q_1)^{-1} \\ &+ (M(k) + Q_1)^{-1}Q_1(B_1(k) + Q_2)^{-1}Q_2B_2(k)^{-1}Q_2(B_1(k) + Q_2)^{-1}Q_1(M(k) + Q_1)^{-1} \\ &+ (M(k) + Q_1)^{-1}Q_1(B_1(k) + Q_2)^{-1}Q_2(B_2(k) + Q_2)^{-1}Q_2(B_1(k) + Q_2)^{-1}Q_1(M(k) + Q_1)^{-1} \\ &+ (M(k) + Q_1)^{-1}Q_1(B_1(k) + Q_2)^{-1}Q_2(B_2(k) + Q_2)^{-1}Q_3B_3(k)^{-1}Q_3 \\ &\times (B_2(k) + Q_2)^{-1}Q_2(B_1(k) + Q_2)^{-1}Q_1(M(k) + Q_1)^{-1}. \end{split}$$

Another key feature to note is that the above is only a heuristic, in some of our applications the leading term is multiplied by a function of k and this is the scenario in which an application of Corollary 3.1.2 is more appropriate. In some cases the leading term is of the form  $A + \eta^{-1}B$  and thus it is easier to decompose the Hilbert space into the span of complementary projections. In this case an application of Lemma 3.1.3 is more appropriate. We note also that by construction the projections satisfy  $Q_3 \leq Q_2 \leq Q_1$ .

The advantage of the Feshbach method just described is that it allows us to system-

atically isolate the obstructions to invertibility of the operator M(k) into the range of a finite number of decreasing projections, which can be related to the spectrum of H using analytic methods.

We now note several algebraic identities which will be frequently used in our expansions.

**Lemma 3.1.4.** Suppose that  $A_0$  is a bounded self-adjoint operator on a Hilbert space  $\mathcal{H}$  and that  $Q_1$  is the orthogonal projection onto  $\operatorname{Ker}(A_0)$  with zero an isolated point of the spectrum. Then the operator  $A_0 + Q_1$  is invertible with inverse  $D_0 = (A_0 + Q_1)^{-1}$  satisfying the relations

$$Q_1 = D_0 Q_1 = Q_1 D_0.$$

*Proof.* Since  $Q_1$  is the projection onto  $Ker(A_0)$  we have the relation  $Q_1A_0=0$  and thus we find

$$Q_1 = Q_1(A_0 + Q_1)D = Q_1D.$$

Taking adjoints gives us  $DQ_1 = Q_1$  also, since  $D_0$  is a self-adjoint operator.

Remark 3.1.5. If  $Q \geq Q_1$  is an orthogonal projection and  $QA_0 = A_0Q = A_0$  then we have the additional relations  $QD_0 = D_0Q = D_0$ , which follow immediately from the fact that Q is the identity on  $Q\mathcal{H}$ .

We end this section by describing a result which will allow us to show that the inversion procedure terminates in each dimension. The termination of the Feshbach procedure of Lemma 3.1.1 is guaranteed by more abstract bounds on the resolvent, however the next result also allows us to deduce useful relationships between certain integral operators.

**Lemma 3.1.6.** Suppose that  $f \in L^1(\mathbb{R}^n) \cap C^1(\mathbb{R}^n)$  satisfies f(0) = 0. Then for  $n \geq 3$  we have

$$\lim_{k \to 0} \int_{\mathbb{R}^n} \frac{1}{|\xi|^2 (|\xi|^2 + k^2)} |f(\xi)|^2 d\xi = \int_{\mathbb{R}^n} \frac{1}{|\xi|^4} |f(\xi)|^2 d\xi.$$
 (3.9)

If in addition we have  $(x \mapsto |x|^{-2} f(x)) \in L^1(\mathbb{R}^n)$  then Equation (3.9) holds in dimension n = 1, 2 also.

*Proof.* Since  $f \in L^1(\mathbb{R}^n)$  and f(0) = 0 we know that |f(r)| = O(r) as  $r \to 0$ . We compute for any R > 0 that

$$\int_{\mathbb{R}^{n}} \frac{1}{|\xi|^{4}} |f(\xi)|^{2} d\xi = \int_{\mathbb{S}^{n-1}} \int_{0}^{\infty} r^{n-5} |f(r\omega)|^{2} dr d\omega 
= \int_{\mathbb{S}^{n-1}} \int_{0}^{R} r^{n-5} |f(r\omega)|^{2} dr d\omega + \int_{\mathbb{S}^{n-1}} \int_{R}^{\infty} r^{n-5} |f(r\omega)|^{2} dr d\omega.$$

The integral from R to infinity always converges given the assumptions on f, so we must consider only the integral from 0 to  $R_0$ . Since |f(r)| = O(r) as  $r \to 0$  we see that the integral over [0, R] converges for all  $n \ge 3$ . Applying the dominated convergence theorem now gives the equality of Equation (3.9).

If in addition we have that  $|x|^{-2}f(x) \in L^1(\mathbb{R}^n)$  then we obtain  $|f(r)| = O(r^3)$  as  $r \to 0$  and thus the integral over [0, R] converges in dimension n = 1, 2 also. An application of the dominated convergence theorem completes the proof.

## **3.2** Expansions of the operator M(k)

We follow the method outlined in [89]. The expansions we present have been computed in various forms throughout the literature, in particular in dimension n = 1 in [89, 116], in dimension n = 2 in [63, 89, 166] in dimension n = 3 in [64, 88], in dimension n = 4 in [62, 87] and in dimension  $n \ge 5$  in [71, 72, 85]. The older papers [85, 87, 88, 116] compute a non-symmetrised resolvent expansion whilst the more recent papers use the symmetrised technique outlined in [89].

Due to the dependence of the integral kernel of  $R_0(z)$  on the parity of the dimension n our analysis is divided into odd and even dimensions. There are no resonances for  $n \geq 5$ , the only poles of the resolvent at z = 0 arising from the presence of zero energy eigenvalues. As a result the analysis is much simpler in dimension  $n \geq 5$  and so we consider this first. As a general rule, the logarithmic behaviour of the integral kernel of  $R_0(z)$  makes the analysis more difficult in even dimensions.

As discussed in Corollary 2.2.13 for  $z \in \mathbb{C} \setminus [0, \infty)$  with  $\operatorname{Im}(z^{\frac{1}{2}}) > 0$  (here we are using the principal branch of the logarithm) the integral kernel of the resolvent  $R_0(z)$  is given by

$$R_0(x,y,z) = \frac{i}{4} z^{\frac{n-2}{4}} (2\pi)^{-\frac{n-2}{2}} |x-y|^{-\frac{n-2}{2}} H_{\frac{n-2}{2}}^{(1)}(z^{\frac{1}{2}} |x-y|), \tag{3.10}$$

where  $H_{\nu}^{(1)}$  is the Hankel function of the first kind of order  $\nu$ . We will use the variable  $k=-iz^{\frac{1}{2}}$  so that  $z=-k^2$ . For n=1 we thus have the expansion

$$R_0(x, y, -k^2) = \frac{1}{2} \sum_{p=0}^{\infty} \frac{(-1)^p}{p!} k^{p-1} |x - y|^p.$$
 (3.11)

For  $n \geq 3$  odd we have the expansion

$$R_0(x, y, -k^2) = |x - y|^{-(n-2)} \sum_{p=0}^{\infty} c_{n,p} k^p |x - y|^p.$$
(3.12)

Here the  $c_{n,p}$  are numerical coefficients whose value is not needed and can be found in [89,

Section 3], although we note that  $c_{n,1} = 0$  and in fact  $c_{n,p} = 0$  for p = 1, 3, ..., n - 4 by [85, Lemma 3.3] (the argument here is purely combinatorial and uses an expansion of the heat kernel). If n is even we obtain

$$R_0(x, y, -k^2) = k^{n-2} \ln(k|x - y|) \sum_{p=0}^{\infty} c_{n,p} k^{2p} |x - y|^{2p} + |x - y|^{-(n-2)} \sum_{p=0}^{\infty} d_{n,p} k^{2p} |x - y|^{2p}$$
(3.13)

where the  $c_{n,p}$  and  $d_{n,p}$  are numerical coefficients whose value will be recalled as needed and can be found in [89, Section 3].

The above expansions are only formal and one needs to be more precise in what sense they are valid, truncating after finitely many terms. The expansions of Equation (3.11), (3.12) and (3.13) will be used in conjunction with the definition

$$M(k) = U + vR_0(-k^2)v$$

to determine expansions of M(k) in each dimension.

## 3.2.1 Dimension $n \ge 5$ odd

In this section we employ the methods of [89] to compute an expansion of  $M(k)^{-1}$  for  $n \geq 5$  odd. For our purposes only the lowest order term in our expansions will be needed and thus we only compute the first term in the expansion, although we remark that more terms could be computed if necessary. We will refer to [85] for some boundedness properties of the operators involved in our expansions. The symmetrised expansions we present have appeared in [71].

The assumptions we make on the potential depend heavily on the dimension.

**Assumption 3.2.1.** If  $n \geq 5$  is odd we assume that V satisfies Assumption 2.2.14 for

- 1.  $\rho > 7$  if n = 5,
- 2.  $\rho > 6$  if n = 7, 9 and
- 3.  $\rho > \frac{n+1}{2}$  if  $n \ge 11$ .

We make the following definition, based the asymptotic expansion of  $R_0(-k^2)$  in Equation (3.12).

**Definition 3.2.2.** Suppose that  $n \geq 5$  is odd and V satisfies Assumption 3.2.1. We define for  $p \in \mathbb{N}$  the integral operators  $G_p$  with integral kernels

$$G_p(x,y) = c_{n,p}|x-y|^{2-n+p}$$

and for  $p \geq 1$  define the operators  $M_p$  by the integral kernels

$$M_p(x,y) = v(x)G_p(x,y)v(y).$$

For p = 0 we define the operator  $M_0$  by the integral kernel

$$(M_0 - U)(x, y) = v(x)G_0(x, y)v(y).$$

Note that we have  $M_1 = 0$  by [85, Lemma 3.3]. Thus by the definition of M(k) we have the expansion

$$M(k) = U + vR_0(-k^2)v = M_0 + k^2M_2 + k^3\tilde{R}_0(k),$$

where  $\tilde{R}_0(k)$  is uniformly bounded in  $F = \{k \in \mathbb{C} : \text{Re}(k) \ge 0 \text{ and } |k| \le 1\}.$ 

**Lemma 3.2.3.** Suppose that  $n \geq 5$  is odd and V satisfies Assumption 3.2.1. Then the operators  $M_0$  and  $M_2$  are bounded and  $\tilde{R}_0$  is uniformly bounded in norm for  $k \in F$ .

Proof. We have by [85, Lemma 3.4] that  $G_0, G_2 \in \mathcal{B}(H^{-1,t}, H^{1,-t})$  for all  $t \geq 1$  and  $G_2 \in \mathcal{B}(H^{-1,t}, H^{1,-t})$  for  $t \geq 2$  and  $\rho$  satisfying Assumption 3.2.1. Thus the operators  $M_0$  and  $M_2$  are bounded also.

Note that  $M_0$  is a compact self-adjoint perturbation of U and so has essential spectrum contained in  $\{-1,1\}$ . Thus 0 is an isolated point of the spectrum of  $M_0$  and  $\operatorname{DimKer}(M_0) < \infty$ . If 0 is an eigenvalue of H then  $M_0$  is not invertible, however we can use the Feshbach inversion method to overcome this and make the following definition.

**Definition 3.2.4.** If the operator  $M_0$  is invertible then we say that zero is a regular point of the spectrum of H. If  $M_0$  is not invertible, we say that zero is an exceptional point of the spectrum of H. For an exceptional point we can define the orthogonal projection  $Q_1$  onto the kernel of  $M_0$  and note that  $M_0 + Q_1$  is invertible on  $\mathcal{H}$ , so that we may define the operator

$$D_0 = (M_0 + Q_1)^{-1}.$$

Since  $M_0$  is a compact perturbation of U (an invertible operator), the Fredholm alternative guarantees that  $Q_1$  is a finite rank operator.

We will now apply Lemma 3.1.1 with A = M(k) and the projection  $Q_1$ . Thus we need to show that  $M(k) + Q_1$  has a bounded inverse on  $\mathcal{H}$ .

**Lemma 3.2.5.** Suppose that  $n \ge 5$  is odd and V satisfies Assumption 3.2.1 and that zero is an exceptional point of the spectrum of H. Define the operator  $D_0 = (M_0 + Q_1)^{-1}$ .

Then for sufficiently small k we have that  $M(k) + Q_1$  is invertible on  $\mathcal{H}$ . Furthermore, we have the expansion

$$(M(k) + Q_1)^{-1} = D_0 - k^2 D_0 M_2 D_0 + k^3 \tilde{R}_1(k),$$
(3.14)

where  $\tilde{R}_1(k)$  is uniformly bounded in k.

*Proof.* For sufficiently small k we find that

$$\left| \left| k^2 M_2 D_0 + k^3 \tilde{R}_0(k) D_0 \right| \right| < 1.$$

We can thus use a Neumann expansion to compute that

$$(M(k) + Q_1)^{-1} = \left(M_0 + Q_1 + k^2 M_2 + k^3 \tilde{R}_0(k)\right)^{-1} = D_0 \left(\operatorname{Id} + (k^2 M_2 + k^3 \tilde{R}_0(k))D_0\right)^{-1}$$

$$= D_0 \left(\operatorname{Id} + \sum_{j=1}^{\infty} (-1)^j \left((k^2 M_2 + k^3 \tilde{R}_0(k))D_0\right)^j\right)$$

$$= D_0 - k^2 D_0 M_2 D_0 - k^3 D_0 \tilde{R}_0(k)D_0$$

$$+ \sum_{j=2}^{\infty} (-1)^j D_0 \left((k^2 M_2 + k^3 \tilde{R}_0(k))D_0\right)^j D_0.$$

Defining the operator

$$\tilde{R}_1(k) = -D_0 \tilde{R}_0(k) D_0 + \sum_{j=2}^{\infty} (-1)^j k^{-3} D_0 \left( (k^2 M_2 + k^3 \tilde{R}_0(k)) D_0 \right)^j D_0$$

completes the proof.

Lemma 3.1.1 states that M(k) is invertible if and only if the operator

$$B_1(k) := Q_1 - Q_1(M(k) + Q_1)^{-1}Q_1,$$

is invertible on  $Q_1\mathcal{H}$  and in this case

$$M(k)^{-1} = (M(k) + Q_1)^{-1} + (M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}.$$

We thus need to determine an inverse for  $B_1(k)$ . Using Lemma 3.2.5 and the fact that  $Q_1 = Q_1 D_0 Q_1$  (see Lemma 3.1.4) we can first determine an expansion for  $B_1(k)$ .

**Lemma 3.2.6.** Suppose that  $n \geq 5$  is odd and V satisfies Assumption 3.2.1 and that zero is an exceptional point of the spectrum of H. Then for sufficiently small k we have the

expansion

$$B_1(k) = k^2 Q_1 M_2 Q_1 - k^3 Q_1 \tilde{R}_1(k). \tag{3.15}$$

*Proof.* We compute that

$$B_1(k) = Q_1 - Q_1(M(k) + Q_1)^{-1}Q_1 = Q_1 - Q_1\left(D_0 - k^2D_0M_2D_0 + k^3\tilde{R}_1(k)\right)Q_1$$
  
=  $Q_1 - Q_1D_0Q_1 + k^2Q_1D_0M_2D_0Q_1 - k^3Q_1\tilde{R}_1(k)Q_1$   
=  $k^2Q_1M_2Q_1 - k^3Q_1\tilde{R}_1(k)$ .

where we have used the relations  $Q_1D_0=D_0Q_1=Q_1$  of Lemma 3.1.4 several times.  $\square$ 

Hence the invertibility of  $B_1(k)$  depends on the invertibility of the operator  $Q_1M_2Q_1$  on  $Q_1\mathcal{H}$ .

**Lemma 3.2.7.** [71, Lemma 5.3] Suppose that  $n \geq 5$  is odd and V satisfies Assumption 3.2.1 and that zero is an exceptional point of the spectrum of H. Then the operator  $Q_1M_2Q_1$  is invertible on  $Q_1\mathcal{H}$ . Furthermore, for  $f \in Q_1\mathcal{H}$  we have the relation  $\langle G_2vf, vf \rangle = \langle G_0vf, G_0vf \rangle$ .

*Proof.* Take  $f \in Q_1\mathcal{H}$ , so that  $(U + vG_0v)f = 0$ , and suppose  $Q_1M_2Q_1f = 0$ . We note that the resolvent expansion (3.12) implies the weak operator topology limit

$$\lim_{k \to 0} k^{-2} (R_0(k^2) - G_0) = G_2.$$

Note also that  $vf \in L^1(\mathbb{R}^n)$  and  $x \mapsto |x|vf \in L^1(\mathbb{R}^n)$ , so that  $\mathcal{F}(vf) \in L^1(\mathbb{R}^n) \cap C^1(\mathbb{R}^n)$ . So we find

$$\langle Q_1 v G_2 v Q_1 f, f \rangle = \langle G_2 v f, v f \rangle = \lim_{k \to 0} \langle k^{-2} (R_0(k^2) - G_0) v f, v f \rangle$$

$$= \lim_{k \to 0} k^{-2} \int_{\mathbb{R}^n} \left( |\xi|^2 - k^2 \right)^{-1} - |\xi|^{-2} \right) [\mathcal{F}(vf)](\xi) \overline{[\mathcal{F}(vf)](\xi)} \, \mathrm{d}\xi$$

$$= \lim_{k \to 0} \int_{\mathbb{R}^n} \frac{1}{|\xi|^2 (|\xi|^2 - \lambda^2)} |[\mathcal{F}(vf)](\xi)|^2 \, \mathrm{d}\xi$$

$$= \int_{\mathbb{R}^n} |\xi|^{-4} |[\mathcal{F}(vf)](\xi)|^2 \, \mathrm{d}\xi$$

$$= \langle G_0 v f, G_0 v f \rangle,$$

where we have used Lemma 3.1.6 to bring the limit inside the integral. The relation  $\langle G_2vf, vf \rangle = \langle G_0vf, G_0vf \rangle$  follows from the above equalities. Since  $Q_1M_2Q_1f = 0$  the final equality implies that  $\mathcal{F}(vf) = 0$  which in turn gives vf = 0. So we see that  $\operatorname{Ker}(Q_1M_2Q_1) = \{0\}$  and the claim is proved.

We now have a well-defined bounded operator  $D_1 = (Q_1 M_2 Q_1)^{-1}$  on  $Q_1 \mathcal{H}$ . Noting that  $D_1 = Q_1 D_1 Q_1$  we see that  $D_1$  is finite rank.

**Lemma 3.2.8.** Suppose that  $n \geq 5$  is odd and V satisfies Assumption 3.2.1 and that zero is an exceptional point of the spectrum of H. Then for sufficiently small k the operator  $B_1(k)$  is invertible and we have the expansion

$$B_1(k)^{-1} = k^{-2}D_1 + k^{-1}D_1Q_1\tilde{R}_1(k)Q_1D_1 + k^2\tilde{R}_2(k),$$

where  $\tilde{R}_2(k)$  is uniformly bounded in k and  $D_1 = (Q_1 M_2 Q_1)^{-1}$ .

*Proof.* For sufficiently small k we have the estimate

$$\left| \left| kQ_1 \tilde{R}_1(k) Q_1 D_1 \right| \right| < 1.$$

We thus begin with Equation (3.15) to compute the Neumann expansion

$$B_1(k)^{-1} = \left(k^2 Q_1 M_2 Q_1 - k^3 Q_1 \tilde{R}_1(k)\right)^{-1} = k^{-2} D_1 \left(\operatorname{Id} - k Q_1 \tilde{R}_1(k) Q_1 D_1\right)^{-1}$$
$$= k^{-2} D_1 \left(\operatorname{Id} + k Q_1 \tilde{R}_1(k) Q_1 D_1 + \sum_{j=2}^{\infty} k^j (\tilde{R}_1(k)^j Q_1 D_1)^j\right).$$

Defining 
$$\tilde{R}_2(k) = \sum_{j=2}^{\infty} k^{j-2} (\tilde{R}_1(k)^j Q D_1)^j$$
 completes the proof.

We finally have the required tools for our expansion of M(k).

**Theorem 3.2.9.** Suppose that  $n \geq 5$  is odd and V satisfies Assumption 3.2.1 and that zero is an exceptional point for the spectrum of H. Then for sufficiently small k we have the expansion

$$M(k)^{-1} = k^{-2}Q_1D_1Q_1 + \tilde{R}_3(k), \tag{3.16}$$

where the operator  $\tilde{R}_3(k)$  is uniformly bounded in k.

*Proof.* We apply Lemma 3.1.1 to A = M(k) and use Lemmas 3.2.5 and 3.2.8 to obtain

$$M(k)^{-1} = (M(k) + Q_1)^{-1} + (M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}$$

$$= D_0 - k^2D_0M_2D_0 + k^3\tilde{R}_1(k) + (D_0 - k^2D_0M_2D_0 + k^3\tilde{R}_1(k))$$

$$\times Q_1(k^{-2}D_1 + k^{-1}D_1Q_1\tilde{R}_1(k)Q_1D_1 + k^2\tilde{R}_2(k))Q_1(D_0 - k^2D_0M_2D_0 + k^3\tilde{R}_1(k)).$$

Expanding out the product shows that

$$(M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1} = k^{-2}Q_1D_1Q_1 - Q_1D_1Q_1M_2D_0 + k^{-1}Q_1D_1Q_1\tilde{R}_1(k)Q_1D_1Q_1 + k\tilde{R}_4(k),$$

where  $\tilde{R}_4(k)$  is uniformly bounded for  $k \in F$ . Thus we find  $M(k)^{-1} = k^{-2}Q_1D_1Q_1 + \tilde{R}_3(k)$ , where we have defined the operator

$$\tilde{R}_3(k) = D_0 - k^2 D_0 M_2 D_0 + k^3 \tilde{R}_1(k) - Q_1 D_1 Q_1 M_2 D_0 + k^{-1} Q_1 D_1 Q_1 \tilde{R}_1(k) Q_1 D_1 Q_1 + k \tilde{R}_4(k),$$

which is uniformly bounded in k.

Finally we characterise the zero eigenspace of H in terms of the operators just described. The following result is implicit in [85] and can be found as [71, Lemma 5.5].

**Lemma 3.2.10.** [71, Lemma 5.5] Suppose that  $n \geq 5$  is odd and V satisfies Assumption 3.2.1 and that zero is an exceptional point of the spectrum of H. Then the projection onto the eigenspace at zero is  $P_0 = G_0 v Q_1 D_1 Q_1 v G_0$ .

*Proof.* Suppose that  $Dim(Q_1\mathcal{H}) = N_0$  and let  $(\varphi_j)_{j=1}^{N_0}$  be an orthonormal basis for  $Q_1\mathcal{H}$ . Then we have

$$0 = M_0 \varphi_j = U M_0 \varphi_j = U (U \varphi_j + v G_0 v \varphi_j) = \varphi_j + U v G_0 v \varphi_j.$$

Write  $\varphi_j = Uv\psi_j$  for  $(\psi_j)_{j=1}^{N_0} \subset \mathcal{H}$  a linearly independent set (see [71, Lemma 5.1]), so that

$$0 = Uv\psi_j + UvG_0vUv\psi_j = Uv(\psi_j + G_0V\psi_j),$$

which implies  $\psi_j + G_0 V \psi_j = 0$ . Thus for any  $f \in \mathcal{H}$  we may write

$$Q_1 v G_0 f = \sum_{j=1}^{N_0} \langle v G_0 f, \varphi_j \rangle \varphi_j = \sum_{j=1}^{N_0} \langle f, G_0 v \varphi_j \rangle \varphi_j$$
$$= \sum_{j=1}^{N_0} \langle f, G_0 V \psi_j \rangle \varphi_j = -\sum_{j=1}^{N_0} \langle f, \psi_j \rangle \varphi_j.$$

We let  $(A_{ij})$  be the matrix representation of  $Q_1M_2Q_1$  with respect to  $(\varphi_j)_{j=1}^{N_0}$ , so that

$$A_{ij} = \langle \varphi_i, Q_1 M_2 Q_1 \varphi_j \rangle = \langle v \varphi_i, G_2 v \varphi_j \rangle = \langle G_0 v \varphi_i, G_0 v \varphi_j \rangle = \langle \varphi_i, \varphi_j \rangle,$$

where we have used the relation  $\langle G_2vf, vg \rangle = \langle G_0vf, G_0vg \rangle$  valid for all  $f, g \in Q_1\mathcal{H}$  (see

Lemma 3.2.7). Then for  $f \in \mathcal{H}$  we may calculate that

$$P_{0}f = G_{0}vQD_{1}QvG_{0}f = G_{0}vQ_{1}D_{1}\left(-\sum_{j=1}^{N_{0}}\langle f, \psi_{j}f\rangle\varphi_{j}\right) = -\sum_{j=1}^{N_{0}}G_{0}vQD_{1}\varphi_{j}\langle f, \psi_{j}\rangle$$

$$= -\sum_{j=1}^{N_{0}}\sum_{p=1}^{N_{0}}G_{0}vQ_{1}(A^{-1})_{pj}\varphi_{p}\langle f, \psi_{j}\rangle = -\sum_{j=1}^{N_{0}}\sum_{p=1}^{N_{0}}G_{0}v\varphi_{p}(A^{-1})_{pj}\langle f, \psi_{j}\rangle$$

$$= \sum_{j=1}^{N_{0}}\sum_{p=1}^{N_{0}}(A^{-1})_{pj}\psi_{p}\langle f, \psi_{j}\rangle.$$

In particular for  $f = \psi_m$  we find  $P_0 \psi_m = \psi_m$  and so the range of  $P_0$  is the span of  $(\psi_m)_{m=1}^{N_0}$  and since  $P_0$  is self-adjoint it is the desired projection onto the kernel of H.

The operator  $P_0$  arises when considering the lowest order term in the expansion of

$$R(-k^2) = R_0(-k^2) - R_0(-k^2)vM(k)^{-1}vR_0(-k^2),$$

as we now demonstrate.

**Theorem 3.2.11.** Suppose that  $n \geq 5$  is odd and V satisfies Assumption 3.2.1. Then for sufficiently small k we have the expansion

$$R(-k^2) - R_0(-k^2) = -k^{-2}P_0 + \tilde{R}(k), \tag{3.17}$$

where  $\tilde{R}(k)$  is uniformly bounded.

*Proof.* We note that

$$R(-k^{2}) - R_{0}(-k^{2}) = -R_{0}(-k^{2})vM(k)^{-1}vR_{0}(-k^{2})$$

$$= -(G_{0}v + \tilde{R}_{1}(k))(k^{-2}Q_{1}D_{1}Q_{1} + \tilde{R}_{3}(k))(vG_{0} + \tilde{R}_{2}(k))$$

$$= -k^{-2}G_{0}vQ_{1}D_{1}Q_{1}vG_{0} + \tilde{R}(k).$$

An application of Lemma 3.2.10 completes the proof.

### 3.2.2 Dimension $n \geq 5$ even

In this section we compute the low energy behaviour of the operator M(k) in even dimensions  $n \geq 5$ . We use the Feshbach inversion formula as described in [89]. Such expansions have been considered in a different context in [72] and first appeared in a non-symmetrised form in [85], to which we will refer for some boundedness properties of the operators involved in our expansions. The statements and proofs in this section are rather similar to the case  $n \geq 5$  odd, however are made more cumbersome by the presence

of the logarithmic behaviour of the Hankel function appearing in the integral kernel of  $R_0(-k^2)$ . We thus introduce the variable  $\eta = \ln(k)^{-1}$  (we are using the principal branch of the logarithm).

We require stronger assumptions on the potential than Assumption 2.2.14.

**Assumption 3.2.12.** If  $n \geq 5$  is even we assume that V satisfies Assumption 2.2.14 for

- 1.  $\rho > 7$  if n = 6,
- 2.  $\rho > 6$  if n = 8, 10 and
- 3.  $\rho > \frac{n+1}{2}$  if  $n \ge 12$ .

We recall that if n is even we have the expansion

$$R_0(x,y,z) = k^{n-2} \ln(k|x-y|) \sum_{p=0}^{\infty} c_{n,p} k^{2p} |x-y|^{2p} + |x-y|^{-(n-2)} \sum_{p=0}^{\infty} d_{n,p} k^{2p} |x-y|^{2p}$$
(3.18)

Considering the terms in the expansion of  $R_0(x, y, k^2)$  of Equation (3.18) with logarithmic coefficients leads to the following operator definitions.

**Definition 3.2.13.** For  $0 \le p \le \frac{n}{2} - 2$  we define the operators  $G_{2p,-1} = 0$  and the integral operators  $G_{2p,0}$  with the integral kernels

$$G_{2p,0}(x,y) = d_{n,|x-y|^{2p+2-n}}.$$

For  $p \geq \frac{n}{2} - 1$  we define the operators  $G_{2p,0}$  and  $G_{2p,-1}$  by the integral kernels

$$G_{2p,0}(x,y) = |x-y|^{2p} \left( c_{n,2p} \ln \left( |x-y| \right) + d_{n,2p+n-2} \right),$$
  

$$G_{2p,-1}(x,y) = c_{n,p} |x-y|^{2p}.$$

For  $(0,0) \neq (p,q) \in \mathbb{N} \times \{0,-1\}$  we can further define the operators  $M_{0,0} = U + vG_{0,0}v$  and  $M_{p,q} = vG_{p,q}v$ .

We will only need the first few terms in this expansion, although it is helpful to see how each is defined. We may then write using the definition of M(k) the expansion

$$M(k) = U + vR_0(-k^2)v = M_{0,0} + k^2M_{2,0} + k^4\eta^{-1}\tilde{R}_0(k)$$

with  $\tilde{R}_0(k)$  uniformly bounded in k. The labelling convention we use is that a pair (i,j) of subscripts refers to an operator with coefficient  $k^i\eta^j$ . We make the following definition of exceptional points.

**Definition 3.2.14.** We say that zero is a regular point for H if the operator  $M_{0,0}$  is invertible. If  $M_{0,0}$  is not invertible, we say zero is an exceptional point for H and define the operator  $Q_1$  to be the orthogonal projection onto the kernel of  $M_{0,0}$ . The operator  $M_{0,0} + Q_1$  is then invertible and we let  $D_0 = (M_{0,0} + Q_1)^{-1}$ .

We now proceed as in the odd case to obtain an expansion for  $M(k)^{-1}$ .

**Lemma 3.2.15.** Suppose that  $n \geq 5$  is even and V satisfies Assumption 3.2.12 and that zero is an exceptional point for H. Then for sufficiently small k we have the expansion

$$(M(k) + Q_1)^{-1} = D_0 - k^2 D_0 M_{2,0} D_0 + k^4 \eta^{-1} \tilde{R}_2(k),$$

where  $\tilde{R}_2(k)$  is uniformly bounded in k and  $D_0 = (M_{0,0} + Q_1)^{-1}$ .

*Proof.* For sufficiently small k we have the estimate

$$\left| \left| k^2 M_{2,0} D_0 + k^4 \eta^{-1} \tilde{R}_0(k) D_0 \right| \right| < 1.$$

We are thus able to compute using a Neumann expansion the expansion

$$(M(k) + Q_1)^{-1} = (M_{0,0} + k^2 M_{2,0} + k^4 \eta^{-1} \tilde{R}_0(k) + Q_1)^{-1}$$
  
=  $D_0 (\text{Id} + k^2 M_{2,0} D_0 + k^4 \eta^{-1} \tilde{R}_0(k) D_0)^{-1}$   
=  $D_0 (\text{Id} - k^2 M_{2,0} D_0 - k^4 \eta^{-1} \tilde{R}_0(k) D_0 + k^4 \eta^{-1} \tilde{R}_1(k)),$ 

where  $\tilde{R}_1(k)$  is uniformly bounded in k. The result follows by defining the remainder  $\tilde{R}_2(k) = \tilde{R}_0(k) + \tilde{R}_1(k)$ .

We next define the operator  $B_1(k)$  by

$$B_1(k) = Q_1 - Q_1(M(k) + Q_1)^{-1}Q_1. (3.19)$$

**Lemma 3.2.16.** Suppose that  $n \geq 5$  is even and V satisfies Assumption 3.2.12 and that zero is an exceptional point for H. Then for sufficiently small k we have the expansion

$$B_1(k) = k^2 Q_1 M_{2,0} Q_1 - Q_1 k^4 \tilde{R}_2(k).$$

*Proof.* We compute using the definition of  $B_1(k)$  and Lemma 3.2.15 that

$$B_1(k) = Q_1 - Q_1(M(k) + Q_1)^{-1}Q_1 = Q_1 - Q_1(D_0 - k^2D_0M_{2,0}D_0 + k^4\eta^{-1}\tilde{R}_2(k))Q_1$$
  
=  $k^2Q_1M_{2,0}Q_1 - k^4Q_1\eta^{-1}\tilde{R}_2(k)Q_1$ ,

where we have used the relations  $Q_1D_0=D_0Q_1=Q$  (see Lemma 3.1.4).

We see that the invertibility of  $B_1(k)$  depends on the invertibility of the operator  $Q_1M_{2,0}Q_1$ . The following result shows that the Feshbach procedure of Lemma 3.1.1 terminates.

**Lemma 3.2.17.** [72, Lemma 5.2] Suppose that  $n \geq 5$  is even and V satisfies Assumption 3.2.12. Then the operator  $Q_1M_{2,0}Q_1$  is invertible on  $Q_1\mathcal{H}$ .

*Proof.* The proof is identical to that of Lemma 3.2.7 with the operator  $M_2$  replaced by  $M_{2,0}$ , since the integral kernels are scalar multiples of each other.

We can now determine an expansion for the inverse of the operator  $B_1(k)$ . We write  $D_1 = (Q_1 M_{2,0} Q_1)^{-1}$  on  $Q_1 \mathcal{H}$ .

**Lemma 3.2.18.** Suppose that  $n \geq 5$  is even and V satisfies Assumption 3.2.12 and that zero is an exceptional point for H. Then for sufficiently small k we have the expansion

$$B_1(k)^{-1} = k^{-2}D_1 - k^2D_1Q_1\eta^{-1}\tilde{R}_2(k)Q_1D_1 + k^2\eta^{-1}\tilde{R}_3(k),$$

where  $\tilde{R}_3(k)$  is uniformly bounded in k and  $D_1 = (Q_1 M_{2,0} Q_1)^{-1}$ .

*Proof.* For sufficiently small k we have the estimate

$$\left| \left| k^4 Q_1 \eta^{-1} \tilde{R}_2(k) Q_1 D_1 \right| \right| < 1$$

We are thus able to compute using the definition of  $B_1(k)$  and Lemma 3.2.17 the Neumann expansion

$$B_1(k)^{-1} = (k^2 Q_1 M_{2,0} Q_1 - k^4 Q_1 \eta^{-1} \tilde{R}_2(k) Q_1)^{-1} = k^{-2} D_1 (\operatorname{Id} - k^4 Q_1 \eta^{-1} \tilde{R}_2(k) Q_1 D_1)^{-1}$$
  
=  $k^{-2} D_1 - k^2 D_1 Q_1 \eta^{-1} \tilde{R}_2(k) Q_1 D_1 + k^2 \eta^{-1} \tilde{R}_3(k),$ 

where  $\tilde{R}_3(k)$  is uniformly bounded in k.

We can now use Lemma 3.1.1 to invert M(k).

**Theorem 3.2.19.** Suppose that  $n \geq 5$  is even and V satisfies Assumption 3.2.12 and that zero is an exceptional point for H. Then for sufficiently small k we have the expansion

$$M(k)^{-1} = k^{-2}Q_1D_1Q_1 + \eta^{-1}\tilde{R}_6(k),$$

where  $\tilde{R}_6(k)$  is uniformly bounded in k.

*Proof.* We use Lemma 3.2.15 to find

$$(M(k) + Q_1)^{-1}Q_1 = Q_1 - k^2 D_0 M_{2,0} Q_1 + k^4 \eta^{-1} \tilde{R}_2(k) Q_1,$$
  

$$Q_1 (M(k) + Q_1)^{-1} = Q_1 - k^2 Q_1 M_{2,0} D_0 + k^4 Q_1 \eta^{-1} \tilde{R}_2(k).$$

We can thus use Lemma 3.2.18 to find

$$(M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}$$

$$= (Q_1 - k^2D_0M_{2,0}Q_1 + k^4\eta^{-1}\tilde{R}_2(k)Q_1)(k^{-2}D_1 - k^2D_1Q_1\eta^{-1}\tilde{R}_2(k)Q_1D_0 + k^2\eta^{-1}\tilde{R}_3(k))$$

$$\times (Q_1 - k^2Q_1M_{2,0}D_0 + k^4Q_1\eta^{-1}\tilde{R}_2(k))$$

$$= k^{-2}Q_1D_1Q_1 + \eta^{-1}\tilde{R}_5(k),$$

where  $\tilde{R}_5(k)$  is uniformly bounded in k. Lemma 3.1.1 then gives us the expansion

$$M(k)^{-1} = (M(k) + Q_1)^{-1} + (M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}$$
  

$$= D_0 - k^2D_0M_{2,0}D_0 + k^4\eta^{-1}\tilde{R}_2(k) + k^{-2}Q_1D_1Q_1 + \eta^{-1}\tilde{R}_5(k)$$
  

$$= k^{-2}Q_1D_1Q_1 + \eta^{-1}\tilde{R}_6(k),$$

where  $\tilde{R}_6(k)$  is uniformly bounded in k.

We observe that despite the logarithmic singularity in the integral kernel of  $R_0(-k^2)$ , the leading behaviour is identical to that of the case  $n \geq 5$  odd. We will demonstrate shortly that this is no longer the case in dimension n = 2, 4 and must be handled with care.

**Lemma 3.2.20.** [72, Lemma 5.3] Suppose that  $n \geq 5$  is even and V satisfies Assumption 3.2.12. Then the projection  $P_0$  onto the zero eigenspace of H is the operator  $G_{0,0}vQ_1D_1Q_1vG_{0,0}$ .

*Proof.* The proof is identical to that of Lemma 3.2.10, so we omit the details.  $\Box$ 

**Theorem 3.2.21.** Suppose that  $n \geq 5$  is even and V satisfies Assumption 3.2.12. Then for sufficiently small k we have

$$R(-k^2) - R(-k^2) = -k^{-2}P_0 + \tilde{R}(k), \tag{3.20}$$

where  $\tilde{R}(k)$  is uniformly bounded.

The proof is identical to the odd dimensional case.

#### 3.2.3 Dimension n=4

In this section we demonstrate an expansion for the symmetrised resolvent in dimension n=4. Such results have been demonstrated in the unsymmetrised case in [87] and have been used in various contexts, see [62]. We will again employ the Feshbach inversion method of [89] discussed at the beginning of this chapter. The resolvent expansions are similar to those found in the case  $n \geq 5$  even, however are sensitive to extra obstructions

to the invertibility of the operator M(k) near zero arising from the presence of zero eigenvalues and of another phenomena, called a zero energy resonance. Our inversion procedure will lead to a natural definition of zero energy resonance in dimension n = 4.

Throughout this section we make the following assumption on the potential.

**Assumption 3.2.22.** We assume the potential V satisfies Assumption 2.2.14 for some  $\rho > 12$ .

Equation (3.13) in dimension n = 4 reads

$$R_0(x,y,z) = k^2 \ln(k|x-y|) \sum_{p=0}^{\infty} c_{4,p} k^{2p} |x-y|^{2p} + |x-y|^{-2} \sum_{p=0}^{\infty} d_{4,p} k^{2p} |x-y|^{2p}$$
 (3.21)

and leads to the following operator definitions.

**Definition 3.2.23.** Suppose that n = 4 and V satisfies Assumption 3.2.22. Define the integral operator  $G_{0,0}$  by the kernel

$$G_{0,0}(x,y) = d_{4,0}|x-y|^{-2}$$

and for integers  $p \geq 1$  the integral operators  $G_{2p,0}$  and  $G_{2p,-1}$  by the kernels

$$G_{2p,0}(x,y) = (c_{4,2p} \ln(|x-y|) + d_{4,2p+2})|x-y|^{2p-2},$$
  

$$G_{2p,-1}(x,y) = c_{4,2p}|x-y|^{2p-2}.$$

Further define the operators  $M_{0,0} = U + vG_{0,0}v$  and  $M_{2p,j} = vG_{2p,j}v$  for j = 0, -1 and  $p \ge 1$  an integer.

The operator  $G_{2p,j}$  is the operator coefficient of the term  $k^{2p} \ln(k)^j$  in Equation (3.21). As in the previous section we introduce the variable  $\eta = \ln(k)^{-1}$ . Then using the definition of M(k) we have the expansions

$$M(k) = U + vR_0(-k^2)v = M_{0,0} + k^2M_{2,0} + k^2\eta^{-1}M_{2,-1} + k^4\eta^{-1}\tilde{R}_0(k)$$
(3.22)

and

$$M(k) = M_{0,0} + k^2 M_{2,0} + k^2 \eta^{-1} M_{2,-1} + k^4 M_{4,0} + k^4 \eta^{-1} M_{4,-1} + k^6 \eta^{-1} \tilde{R}_1(k),$$
 (3.23)

where  $\tilde{R}_0(k)$  and  $\tilde{R}_1(k)$  are uniformly bounded for  $k \in F$ . The expansion of Equation (3.22) does not have sufficiently many terms for some of the more involved calculations and in these situations we will consider the expansion of Equation (3.23). We also define the projection  $P = ||v||^{-2} \langle \cdot, v \rangle v$  which is by definition a scalar multiple of  $M_{2,-1}$ .

**Definition 3.2.24.** Suppose that n = 4 and V satisfies Assumption 3.2.22. We say that zero is a regular point of the spectrum of H if the operator  $M_{0,0}$  is invertible. If zero is not a regular point of the spectrum of H, we define  $Q_1$  to be the orthogonal projection onto the kernel of  $M_{0,0}$ . Then  $M_{0,0} + Q_1$  is invertible and we can define  $D_0 = (M_{0,0} + Q_1)^{-1}$ . We say there is an exceptional point of the first kind if  $Q_1M_{2,-1}Q_1$  is invertible on  $Q_1H$ . If  $Q_1M_{2,-1}Q_1$  is not invertible, we define  $Q_2$  to be the orthogonal projection onto the kernel of  $Q_1M_{2,-1}Q_1$  (as an operator on  $Q_1H$ ). Then  $Q_1M_{2,-1}Q_1 + Q_2$  is invertible on  $Q_1H$ . If  $Q_1 = Q_2$  we say there is an exceptional point of the second kind. Otherwise, we say there is an exceptional point of the third kind.

In our definition,  $Q_2 \neq 0$  (an exceptional point of the second kind) will be shown to correspond to the existence of an eigenvalue whilst  $Q_1 \neq 0$  will lead to a definition of resonance in dimension n = 4. Before proceeding to our expansion of  $M(k)^{-1}$ , we characterise the range of the projections  $Q_1$  and  $Q_2$  in terms of solutions to the equation  $H\psi = 0$ . We can characterise the image of the projections  $Q_1$  and  $Q_2$  in terms of the spectrum of H.

**Lemma 3.2.25.** [62, Lemma 7.1] Suppose that n = 4 and V satisfies Assumption 3.2.22. Then  $0 \neq f \in Q_1\mathcal{H}$  if and only if f = Uvg for some  $g \in H^{0,-t}$  for some t > 0 such that Hg = 0 in the sense of distributions.

*Proof.* Composing with  $H_0$  shows that Hg = 0 if and only if  $(\mathrm{Id} + G_{0,0}V)g = 0$ . So suppose that  $f \in Q_1\mathcal{H}$  which gives us  $(U + vG_{0,0}v)f = 0$  and upon multiplying by U we obtain

$$f(x) = -U(x)v(x)[G_{0,0}vf](x) = d_{4,0}U(x)v(x)\int_{\mathbb{R}^4} |x-y|^{-2}v(y)f(y) \,dy.$$

Thus we define the function q by

$$g(x) = d_{4,0} \int_{\mathbb{R}^4} |x - y|^{-2} v(y) f(y) \, dy = -[G_{0,0} v f](x).$$
 (3.24)

Since  $vf \in H^{0,2}$  we find by [85, Lemma 2.3] that  $g \in H^{0,-t}$  for some t > 0. By construction we have f = Uvg and we have the relation

$$0 = q + G_{0,0}vf = q + G_{0,0}Vq = (\mathrm{Id} + G_{0,0}V)q,$$

so that g satisfies Hg = 0 in the sense of distributions. For the converse, suppose f = Uvg for some non-zero g satisfying Hg = 0. Then we have

$$(U + vG_{0,0}v)f = vq + vG_{0,0}Vq = v(\mathrm{Id} + G_{0,0}V)q = 0,$$

so that  $f \in Q_1 \mathcal{H}$ .

**Lemma 3.2.26.** [62, Lemma 7.2] Suppose n=4 and V satisfies Assumption 3.2.22. Then  $0 \neq f \in Q_2\mathcal{H}$  if and only if f=Uvg for some  $0 \neq g \in \mathcal{H}$  such that Hg=0 in the sense of distributions.

*Proof.* Suppose that  $f \in Q_2\mathcal{H}$ . Since  $Q_2 \leq Q_1$  we have by Lemma 3.2.25 that f = Uvg and it suffices to show  $g \in \mathcal{H}$ . We note that  $PQ_2 = 0$  implies

$$\int_{\mathbb{R}^4} v(y)f(y) \, \mathrm{d}y = 0. \tag{3.25}$$

By Lemma 3.2.25 we have that g satisfies Equation (3.24). Equation (3.25) gives

$$(1+|x|^2)^{-1} \int_{\mathbb{R}^4} v(y)f(y) \, \mathrm{d}y = 0$$

and thus

$$g(x) = d_{4,0} \int_{\mathbb{R}^4} (|x - y|^{-2} - (1 + |x|^2)^{-1}) v(y) f(y) dy.$$

For  $x \neq y \in \mathbb{R}^n$  we have the estimate (see [87, Equation (2.8)])

$$\left| |x - y|^{-2} - (1 + |x|^2)^{-1} \right| \le 5 \left( \frac{(1 + |y|^2)^{\frac{1}{2}}}{(1 + |x|^2)^{\frac{1}{2}} |x - y|^2} + \frac{(1 + |y|^2)^{\frac{1}{2}}}{(1 + |x|^2)|x - y|^2} \right). \tag{3.26}$$

Both terms on the right hand side of the estimate (3.26) are the integral kernels of Riesz operators (see [153, Chapter V.1]). Define the operators  $L_1, L_2 : C_c^{\infty}(\mathbb{R}^4) \to C_c^{\infty}(\mathbb{R}^4)$  by the integral kernels

$$L_1(x,y) = \frac{1}{|x-y|^2}$$
 and  $L_2(x,y) = \frac{1}{|x-y|^2}$ .

By Theorem 2.2.5 we have that for  $\alpha \in \mathbb{R}$  the operator  $q_{\alpha}$  of multiplication by the function  $q_{\alpha} = (1+|\cdot|^2)^{\frac{\alpha}{2}}$  maps  $H^{s,t} \to H^{s,t-\alpha}$  for any  $s,t \in \mathbb{R}$ . For t > 0 we have by [153, Chapter V.1]  $L_1 : H^{0,1+t} \to H^{0,-1}$  and  $L_2 : H^{0,1+t} \to H^{0,-2}$ . Since  $q_1 v f \in H^{0,1+t}$  we find for j = 1, 2 that  $q_{-j}L_jq_1v : \mathcal{H} \to \mathcal{H}$  and thus  $g \in \mathcal{H}$ .

For the converse, suppose f = Uvg and use Equations (3.24) and (3.25) to show

$$g(x) = d_{4,0} \int_{\mathbb{R}^4} (|x - y|^{-2} - (1 + |x|^2)^{-1}) v(y) f(y) dy + d_{4,0} (1 + |x|^2)^{-1} \int_{\mathbb{R}^4} v(y) f(y) dy.$$
(3.27)

Then by assumption  $g \in \mathcal{H}$  and the first term in Equation (3.27) is in  $\mathcal{H}$  by the estimate

(3.26), which forces the second term in Equation (3.27) to be in  $\mathcal{H}$  also. That is

$$\left(\mathbb{R}^4 \ni x \mapsto (1+|x|^2)^{-1} \int_{\mathbb{R}^4} v(y) f(y) \, \mathrm{d}y\right) \in \mathcal{H}.$$

which forces  $0 = Pf = Q_1PQ_1f$  and so  $f \in \mathcal{H}$ .

Corollary 3.2.27. Suppose that n = 4 and V satisfies Assumption 3.2.22. Then

$$\operatorname{Rank}(Q_1) \leq \operatorname{Rank}(Q_2) + 1.$$

*Proof.* Suppose that  $0 \neq f_1, f_2 \in Q_1\mathcal{H}$  and let  $g_1, g_2$  be the corresponding distributional solutions of Hg = 0. Then we just need to show there exists a constant  $c \in \mathbb{C}$  and a function  $h \in \mathcal{H}$  such that  $g_2 - cg_1 = h$ , which follows from Equation (3.27).

The invertibility of the operator M(k) depends on the nature of the singularity at the point zero, however the following result will be used in each exceptional case.

**Lemma 3.2.28.** Suppose that n = 4 and V satisfies Assumption 3.2.22, and that zero is not a regular point of the spectrum of H. Then for sufficiently small k we have the expansion

$$(M(k) + Q_1)^{-1} = D_0 - k^2 D_0 M_{2.0} D_0 - k^2 \eta^{-1} D_0 M_{2.-1} D_0 - k^2 \eta^{-1} \tilde{R}_2(k), \tag{3.28}$$

where  $\tilde{R}_2(k)$  is uniformly bounded in k and  $D_0 = (M_{0,0} + Q_1)^{-1}$ .

*Proof.* For sufficiently small k we have the estimate

$$||M(k) - M_{0,0} + Q_1||_2 < 1.$$

We are thus able to compute the Neumann expansion

$$(M(k) + Q_1)^{-1} = (M_{0,0} + Q_1 + k^2 M_{2,0} + k^2 \eta^{-1} M_{2,-1} + k^4 M_{4,0} + k^4 \eta^{-1} M_{4,-1} + k^4 \eta^{-1} \tilde{R}_1(k))^{-1}$$

$$= D_0 (\operatorname{Id} + k^2 M_{2,0} D_0 + k^2 \eta^{-1} M_{2,-1} D_0 + k^4 M_{4,0} D_0 + k^4 \eta^{-1} M_{4,-1} D_0 + k^4 \eta^{-1} D_0 \tilde{R}_1(k))^{-1}$$

$$= D_0 (\operatorname{Id} - k^2 M_{2,0} D_0 - k^2 \eta^{-1} M_{2,-1} D_0 - k^4 M_{4,0} D_0 - k^4 \eta^{-1} M_{4,-1} D_0 - k^4 \eta^{-1} D_0 \tilde{R}_1(k) + k^4 \eta^{-1} \tilde{R}_4(k))$$

$$= D_0 - k^2 D_0 M_{2,0} D_0 - k^2 \eta^{-1} D_0 M_{2,-1} D_0 + k^2 \eta^{-1} \tilde{R}_2(k),$$

as claimed.  $\Box$ 

Remark 3.2.29. If we use the extra terms in Equation (3.23) we find

$$(M(k) + Q_1)^{-1} = D_0 - k^2 D_0 M_{2,0} D_0 - k^2 \eta^{-1} D_0 M_{2,-1} D_0 - k^4 D_0 M_{4,0} D_0$$
$$- k^4 \eta^{-1} D_0 M_{4,-1} D_0 + k^4 D_0 M_{2,0} D_0 M_{2,0} D_0$$
$$+ k^4 \eta^{-1} (D_0 M_{2,0} D_0 M_{2,-1} D_0 + D_0 M_{2,-1} D_0 M_{2,0} D_0)$$
$$+ k^4 \eta^{-2} D_0 M_{2,-1} D_0 M_{2,-1} D_0 - k^6 \eta^{-1} \tilde{R}'_2(k),$$

where  $\tilde{R}'_2(k)$  is uniformly bounded in k.

We can now handle the relatively simple case of exceptional points of the first kind, which corresponds to resonances with no eigenvalues. Lemma 3.1.1 tells us that M(k) is invertible if and only if  $B_1(k)$  defined by

$$B_1(k) := Q_1 - Q_1 (M(k) + Q_1)^{-1} Q_1$$

is invertible.

**Lemma 3.2.30.** Suppose that n=4 and V satisfies Assumption 3.2.22, and that zero is an exceptional point of the first kind. Then for sufficiently small k we have  $B_1(k)$  is invertible and there exist  $c_1, c_2 \in \mathbb{C}$  and a uniformly bounded function  $\tilde{c}_3 : F \to \mathbb{C}$  such that

$$B_1(k)^{-1} = \left(k^2 c_1 + k^2 \eta^{-1} c_2 + k^4 \eta^{-1} \tilde{c}_3(k)\right)^{-1} Q_1.$$

*Proof.* We have (see Lemma 3.1.4) the relations  $Q_1D_0 = D_0Q_1 = Q_1$  and thus using Lemma 3.2.28 we find

$$Q_1 (M(k) + Q_1)^{-1} Q_1 = Q_1 \left( D_0 - k^2 D_0 M_{2,0} D_0 - k^2 \eta^{-1} D_0 M_{2,-1} D_0 - k^4 \eta^{-1} \tilde{R}_2(k) \right) Q_1$$
  
=  $Q_1 - k^2 Q_1 M_{2,0} Q_1 - k^2 \eta^{-1} Q_1 M_{2,-1} Q_1 - k^4 \eta^{-1} Q_1 \tilde{R}_2(k) Q_1.$ 

Since  $Q_1$  is a rank one operator in the case of an exceptional point of the first kind by Corollary 3.2.27, we find that  $Q_1M_{2,0}Q_1 = c_1Q_1$  for some constant  $c_1$ , and similarly we have  $Q_1M_{2,-1}Q_1 = c_2Q_1$  for some constant  $c_2$ . In the same manner we find there exists  $\tilde{c}_3: F \to \mathbb{C}$  such that  $Q_1\tilde{R}_2(k)Q_1 = \tilde{c}_3(k)Q_1$ . Then we have

$$B_{1}(k) = Q_{1} - Q_{1} (M(k) + Q_{1})^{-1} Q_{1}$$

$$= Q_{1} - Q_{1} \left( D_{0} - k^{2} D_{0} M_{2,0} D_{0} - k^{2} \eta^{-1} D_{0} M_{2,-1} D_{0} - k^{2} \eta^{-1} \tilde{R}_{2}(k) \right)$$

$$= k^{2} Q_{1} M_{2,0} Q_{1} + k^{2} \eta^{-1} Q_{1} M_{2,-1} Q_{1} + k^{2} \eta^{-1} Q_{1} \tilde{R}_{2}(k) Q_{1}$$

$$= (k^{2} c_{1} + k^{2} \eta^{-1} c_{2} + k^{4} \eta^{-1} \tilde{c}_{3}(k)) Q_{1}.$$

We compute the inverse as  $B_1(k)^{-1} = (k^2c_1 + k^2\eta^{-1}c_2 + k^4\eta^{-1}\tilde{c}_3(k))^{-1}Q_1$ , as claimed.  $\Box$ 

We can now use Lemma 3.1.1 to compute the inverse of M(k). We define a recurring function of k to ease the rather involved calculations,

$$g_1(k) = (c_1 + \eta^{-1}c_2 + k^2\eta^{-1}\tilde{c}_3(k))^{-1}.$$

We will use the formula

$$M(k)^{-1} = (M(k) + Q_1)^{-1} + (M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}.$$

Note that the first term in this expansion has already been obtained in Lemma 3.2.28.

**Theorem 3.2.31.** Suppose that n = 4 and V satisfies Assumption 3.2.22, and that zero is an exceptional point of the first kind. Then for sufficiently small k we have the expansion

$$M(k)^{-1} = k^{-2}g_1(k)Q_1 + D_0 - g_1(k)(D_0M_{2,0}Q_1 + Q_1M_{2,0}D_0) - \eta^{-1}g_1(k)(D_0M_{2,-1}Q_1 + Q_1M_{2,-1}D_0) + k^2\tilde{R}_4(k),$$

where  $\tilde{R}_4(k)$  is uniformly bounded in k and  $D_0 = (M_{0,0} + Q_1)^{-1}$ .

Proof. Lemma 3.1.1 gives us that

$$M(k)^{-1} = (M(k) + Q_1)^{-1} + (M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}$$
  
=  $(M(k) + Q_1)^{-1} + g(k)(M(k) + Q_1)^{-1}Q_1(M(k) + Q_1)^{-1}$ .

To handle the second term we make, using Lemma 3.2.28 the computations

$$(M(k) + Q_1)^{-1}Q_1 = \left(D_0 - k^2 D_0 M_{2,0} D_0 - k^2 \eta^{-1} D_0 M_{2,-1} D_0 - k^2 \eta^{-1} \tilde{R}_2(k)\right) Q_1$$
  
=  $Q_1 - k^2 D_0 M_{2,0} Q_1 - k^2 \eta^{-1} D_0 M_{2,-1} Q_1 - k^2 \eta^{-1} \tilde{R}_2(k) Q_1$ 

and

$$Q_1(M(k) + Q_1)^{-1} = Q_1 \left( D_0 - k^2 D_0 M_{2,0} D_0 - k^2 \eta^{-1} D_0 M_{2,-1} D_0 - k^2 \eta^{-1} \tilde{R}_2(k) \right)$$
  
=  $Q_1 - k^2 Q_1 M_{2,0} D_0 - k^2 \eta^{-1} Q_1 M_{2,-1} D_0 - k^2 \eta^{-1} Q_1 \tilde{R}_2(k).$ 

We thus make the computation

$$\begin{split} g_1(k)(M(k)+Q_1)^{-1}Q_1(M(k)+Q_1)^{-1} \\ &= g_1(k)\left(Q_1-k^2D_0M_{2,0}Q_1-k^2\eta^{-1}D_0M_{2,-1}Q_1-k^2\eta^{-1}\tilde{R}_2(k)Q_1\right) \\ &\quad \times \left(Q_1-k^2Q_1M_{2,0}D_0-k^2\eta^{-1}Q_1M_{2,-1}D_0-k^2\eta^{-1}Q_1\tilde{R}_2(k)\right) \\ &= g_1(k)\left(Q_1-k^2Q_1M_{2,0}D_0-k^2D_0M_{2,0}Q_1-k^2\eta^{-1}(D_0M_{2,-1}Q_1+Q_1M_{2,-1}D_0)+\tilde{R}_3(k),\right) \end{split}$$

where  $\tilde{R}_3(k)$  is uniformly bounded in k. So we find the inverse of M(k) as

$$\begin{split} M(k)^{-1} &= (M(k) + Q_1)^{-1} + k^{-2}g_1(k)(M(k) + Q_1)^{-1}Q_1(M(k) + Q_1)^{-1} \\ &= D_0 - k^2D_0M_{2,0}D_0 - k^2\eta^{-1}D_0M_{2,-1}D_0 - k^2\eta^{-1}\tilde{R}_2(k) \\ &+ g_1(k)\big(Q_1 - k^2Q_1M_{2,0}D_0 - k^2D_0M_{2,0}Q_1 - k^2\eta^{-1}\tilde{R}_3(k)\big) \\ &= g_1(k)Q_1 + D_0 - k^2g_1(k)(Q_1M_{2,0}D_0 + D_0M_{2,0}Q_1) + k^2\eta^{-1}\tilde{R}_4(k) \\ &= g_1(k)Q_1 + D_0 - c_2^{-1}(Q_1M_{2,0}D_0 + D_0M_{2,0}Q_1) \\ &+ k^2\eta^{-1}(D_0M_{2,-1}Q_1 + Q_1M_{2,-1}D_0) + k^4\tilde{R}_4(k) \end{split}$$

where  $\tilde{R}_4(k)$  is uniformly bounded in k.

The nature of the singularity in the case of exceptional points of the second and third kind requires a little more care to handle, although is not significantly worse. We will need to use the extra terms in the expansion of  $(M(k)+Q_1)^{-1}$  in Remark 3.2.29. If there is an exceptional point of the second kind for H then the operator  $Q_1M_{2,-1}Q_1$  is not invertible on  $Q_1\mathcal{H}$  and we define  $Q_2$  to be the orthogonal projection onto the kernel of  $Q_1M_{2,-1}Q_1$ . Then the operator  $Q_1M_{2,-1}Q_1+Q_2$  is invertible on  $Q_2\mathcal{H}$ .

**Lemma 3.2.32.** [62, Lemma 7.4] Suppose that n = 4 and V satisfies Assumption 3.2.22. Then the operator  $Q_2M_{2,0}Q_2$  is invertible on  $Q_2\mathcal{H}$ .

Proof. Suppose that  $f \in Q_2\mathcal{H}$  is such that  $Q_2M_{2,0}Q_2f = 0$ . Then  $\langle G_{2,0}vf, vf \rangle = 0$ . Note also that  $vf \in L^1(\mathbb{R}^n)$  and  $x \mapsto |x|vf \in L^1(\mathbb{R}^n)$ , so that  $\mathcal{F}(vf) \in L^1(\mathbb{R}^n) \cap C^1(\mathbb{R}^n)$ . Using that Pf = 0 for  $f \in Q_2\mathcal{H}$  we have the limits

$$\langle G_{2,0}vf, vf \rangle = \lim_{k \to 0} \langle k^{-2}(R_0(k^2) - G_0)vf, vf \rangle = \lim_{k \to 0} \langle k^{-2}(R_0(k^2) - G_0)vf, vf \rangle$$

$$= \lim_{k \to 0} \int_{\mathbb{R}^4} \left( |\xi|^{-2} - (|\xi|^2 + k^2)^{-1} \right) [\mathcal{F}(vf)](\xi) \overline{[\mathcal{F}(vf)](\xi)} \, d\xi$$

$$= \lim_{k \to 0} \int_{\mathbb{R}^4} \frac{|[\mathcal{F}(vf)](\xi)|^2}{|\xi|^2(|\xi|^2 + k^2)} \, d\xi = \int_{\mathbb{R}^4} |\xi|^{-4} |[\mathcal{F}(vf)](\xi)|^2 \, d\xi$$

$$= \langle G_{0,0}vf, G_{0,0}vf \rangle,$$

where we have used Lemma 3.1.6 to bring the limit inside the integral. Since f satisfies  $Q_2M_{2,0}Q_2f=0$ , we find  $\langle G_{2,0}vf,vf=0$ . Thus  $\mathcal{F}(vf)=0$  and so vf=0, which implies that  $\operatorname{Ker}(Q_2vG_{2,0}vQ_2)=\{0\}$ .

If zero is an exceptional point of the second kind, we have  $Q_1 = Q_2$  and so we can define  $D_2 = (Q_1 M_{2,0} Q_1)^{-1}$  as an operator on  $Q_1 \mathcal{H} = Q_2 \mathcal{H}$ . The operator  $D_2$  satisfies the relation  $D_2 = Q_1 D_2 Q_1$  and is thus Hilbert-Schmidt.

**Theorem 3.2.33.** Suppose that n = 4 and V satisfies Assumption 3.2.22 and that zero is an exceptional point of the second kind. Then for sufficiently small k we have the

expansion

$$M(k)^{-1} = k^{-2}Q_1D_2Q_1 - \eta^{-1}Q_1D_2Q_1M_{4,-1}Q_1D_2Q_1 + \tilde{R}_7(k),$$

where  $\tilde{R}_7(k)$  is uniformly bounded in k and K is bounded and  $D_2 = (Q_1 M_{2,0} Q_1)^{-1}$ .

*Proof.* We have the identity  $Q_2M_{2,-1} = Q_2P = PQ_2 = M_{2,-1}Q_2 = 0$ , since for any  $f \in Q_2\mathcal{H}$  we have

$$0 = \langle Q_1 P Q_1 f, f \rangle = \langle P f, P f \rangle = ||P f||^2.$$

Then we can compute using the expansion in Remark 3.2.29 the relation

$$\begin{split} Q_1(M(k)+Q_1)^{-1}Q_1 \\ &= Q_1 - k^2Q_1M_{2,0}Q_1 - k^2\eta^{-1}Q_1M_{2,-1}Q_1 - k^4Q_1M_{4,0}Q_1 \\ &- k^4\eta^{-1}Q_1M_{4,-1}Q_1 + k^4Q_1M_{2,0}D_0M_{2,0}Q_1 \\ &+ k^4\eta^{-1}(Q_1M_{2,0}D_0M_{2,-1}Q_1 + Q_1M_{2,-1}D_0M_{2,0}Q_1) + k^4\eta^{-2}Q_1M_{2,-1}Q_1M_{2,-1}Q_1 \\ &- k^6\eta^{-1}Q_1\tilde{R}_2'(k)Q_1 \\ &= Q_1 - k^2Q_1M_{2,0}Q_1 - k^4Q_1M_{4,0}Q_1 - k^4\eta^{-1}Q_1M_{4,-1}Q_1 + k^4Q_1M_{2,0}D_0M_{2,0}Q_1 \\ &- k^6\eta^{-1}Q_1\tilde{R}_2'(k)Q_1. \end{split}$$

Note that we have used several times the relations  $Q_1 = Q_2$  and  $Q_2P = PQ_2 = 0$ . Thus we can write  $B_1(k)$  as

$$B_1(k) = Q_1 - Q_1(M(k) + Q_1)^{-1}Q_1$$
  
=  $k^2 Q_1 M_{2,0} Q_1 + k^4 Q_1 M_{4,0} Q_1 + k^4 \eta^{-1} Q_1 M_{4,-1} Q_1 - k^4 Q_1 M_{2,0} D_0 M_{2,0} Q_1$   
+  $k^6 \eta^{-1} Q_1 \tilde{R}'_2(k) Q_1$ .

Since each of the operators involved is bounded, we have for sufficiently small k the estimate

$$\left| \left| k^2 Q_1 \left( M_{4,0} Q_1 D_2 + \eta^{-1} M_{4,-1} Q_1 D_2 - M_{2,0} D_0 M_{2,0} Q_1 D_2 + k^2 \eta^{-1} \tilde{R}_2'(k) Q_1 D_2 \right) \right| \right| < 1.$$

Thus we can use a Neumann expansion to determine the inverse of B(k) as

$$\begin{split} B_{1}(k)^{-1} &= k^{-2} \left( Q_{1} M_{2,0} Q_{1} + k^{2} Q_{1} M_{4,0} Q_{1} + k^{2} \eta^{-1} Q_{1} M_{4,-1} Q_{1} - k^{2} Q_{1} M_{2,0} D_{0} M_{2,0} Q_{1} \right. \\ &+ k^{4} \eta^{-1} Q_{1} \tilde{R}_{2}'(k) Q_{1} \right)^{-1} \\ &= k^{-2} D_{2} \left( \operatorname{Id} + k^{2} Q_{1} M_{4,0} Q_{1} D_{2} + k^{2} \eta^{-1} Q_{1} M_{4,-1} Q_{1} D_{2} - k^{2} Q_{1} M_{2,0} D_{0} M_{2,0} Q_{1} D_{2} \right. \\ &+ k^{4} \eta^{-1} Q_{1} \tilde{R}_{2}'(k) Q_{1} D_{2} \right)^{-1} \\ &= k^{-2} D_{2} \left( \operatorname{Id} - k^{2} Q_{1} M_{4,0} Q_{1} D_{2} - k^{2} \eta^{-1} Q_{1} M_{4,-1} Q_{1} D_{2} + k^{2} Q_{1} M_{2,0} D_{0} M_{2,0} Q_{1} D_{2} \right. \\ &- k^{4} \eta^{-1} Q_{1} \tilde{R}_{2}'(k) Q_{1} D_{2} + k^{4} \tilde{R}_{4}(k) \right)^{-1} \\ &= k^{-2} D_{2} - D_{2} Q_{1} M_{4,0} Q_{1} D_{2} - \eta^{-1} D_{2} Q_{2} M_{4,-1} Q_{1} D_{2} + Q_{1} M_{2,0} D_{0} M_{2,0} Q_{1} D_{2} \\ &- k^{2} \eta^{-1} \tilde{R}_{5}(k), \end{split}$$

where  $\tilde{R}_4(k)$  and  $\tilde{R}_5(k)$  are uniformly bounded in k. We can compute using Remark 3.2.29 that

$$(M(k) + Q_1)^{-1}Q_1 = Q_1 - k^2 D_0 M_{2,0} Q_1 - k^2 \eta^{-1} D_0 M_{2,-1} Q_1 - k^4 D_0 M_{4,0} Q_1$$
$$- k^4 \eta^{-1} D_0 M_{4,-1} Q_1 + k^4 D_0 M_{2,0} D_0 M_{2,0} Q_1 + k^4 \eta^{-1} (D_0 M_{2,0} D_0 M_{2,-1} Q_1$$
$$+ D_0 M_{2,-1} D_0 M_{2,0} Q_1) + k^4 \eta^{-2} D_0 M_{2,-1} D_0 M_{2,-1} Q_1 - k^6 \eta^{-1} \tilde{R}'_2(k) Q_1$$
$$= Q_1 - k^2 D_0 M_{2,0} Q_1 - k^4 D_0 M_{4,0} Q_1 - k^4 \eta^{-1} D_0 M_{4,-1} Q_1$$
$$+ k^4 D_0 M_{2,0} D_0 M_{2,0} Q_1 + k^4 \eta^{-1} D_0 M_{2,-1} D_0 M_{2,0} Q_1 - k^6 \eta^{-1} \tilde{R}'_2(k) Q_1$$

and similarly

$$\begin{split} Q_1(M(k)+Q_1)^{-1} &= Q_1 - k^2 Q_1 M_{2,0} D_0 - k^2 \eta^{-1} Q_1 M_{2,-1} D_0 - k^4 Q_1 M_{4,0} D_0 \\ &- k^4 \eta^{-1} Q_1 M_{4,-1} D_0 + k^4 Q_1 M_{2,0} D_0 M_{2,0} D_0 + k^4 \eta^{-1} (Q_1 M_{2,0} D_0 M_{2,-1} D_0 \\ &+ Q_1 M_{2,-1} D_0 M_{2,0} D_0) + k^4 \eta^{-2} Q_1 M_{2,-1} D_0 M_{2,-1} D_0 - k^6 \eta^{-1} Q_1 \tilde{R}_2'(k) \\ &= Q_1 - k^2 Q_1 M_{2,0} D_0 - k^4 Q_1 M_{4,0} D_0 - k^4 \eta^{-1} Q_1 M_{4,-1} D_0 \\ &+ k^4 Q_1 M_{2,0} D_0 M_{2,0} D_0 + k^4 \eta^{-1} Q_1 M_{2,0} D_0 M_{2,-1} D_0 - k^6 \eta^{-1} Q_1 \tilde{R}_2'(k). \end{split}$$

We can then make the computation

$$(M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1} = k^{-2}Q_1D_2Q_1 - \eta^{-1}Q_1D_2Q_1M_{4,-1}Q_1D_2Q_1$$
$$-Q_1D_2Q_1M_{4,0}Q_1D_2Q_1 - D_0M_{2,0}Q_1D_2Q_1 - Q_1D_2Q_1M_{2,0}D_1 + k^2\tilde{R}_6(k)$$

where  $\tilde{R}_6(k)$  is uniformly bounded in k. Applying Lemma 3.1.1 then gives

$$M(k)^{-1} = D_0 + k^{-2}Q_1D_2Q_1 - \eta^{-1}Q_1D_2Q_1M_{4,-1}Q_1D_2Q_1 + \tilde{R}_7(k),$$

where  $\tilde{R}_7(k)$  is uniformly bounded in k.

We now proceed to the case of an exceptional point of the third kind, which covers the simultaneous existence of a resonance and a zero eigenvalue for H.

**Definition 3.2.34.** Suppose that n=4 and V satisfies Assumption 3.2.22. We define the operator  $T_1=Q_1-Q_2$  acting on  $Q_1\mathcal{H}$  and decompose  $Q_1\mathcal{H}=T_1\mathcal{H}\oplus Q_2\mathcal{H}$ . Let  $D_2=(Q_2M_{2,0}Q_2)^{-1}$ . We further define on  $Q_1\mathcal{H}$  the related operator

$$\tilde{T}_1 = \begin{pmatrix} T_1 & -T_1 M_{2,0} D_2 \\ -D_2 M_{2,0} T_1 & D_2 M_{2,0} T_1 M_{2,0} D_2 \end{pmatrix}. \tag{3.29}$$

**Lemma 3.2.35.** Suppose that V satisfies Assumption 3.2.22 and that zero is an exceptional point of the third kind for H. Then the operator  $\tilde{T}_1$  defined by Equation (3.29) is a finite rank operator.

*Proof.* Since  $Q_1 \geq Q_2$  by Corollary 3.2.27 we have that  $Q_1 - Q_2 = T_1$  is a rank one operator, which implies  $\tilde{T}_1$  is a finite rank operator.

Since the operator  $T_1$  is rank one, there exists constants  $d_1, d_2, d_3 \in \mathbb{C}$  such that  $T_1M_{4,0}T_1 = d_1T_1$ ,  $T_1M_{2,-1}T_1 = d_2T_1$  and  $T_1M_{2,0}D_2M_{2,0}T_1 = d_3T_1$ . Then for sufficiently small k we may define the function  $h: F \to \mathbb{C}$  by

$$h(k) = (d_1 + \eta^{-1}d_2 - d_3)^{-1}.$$

**Lemma 3.2.36.** Suppose that n = 4 and V satisfies Assumption 3.2.22 and that zero is an exceptional point of the third kind for H. Then there exist bounded operators A, B, C such that for sufficiently small k and with  $\tilde{T}_1$  is as in Definition 3.2.34 we have the expansion

$$B_1(k)^{-1} = k^{-2}h(k)\tilde{T}_1 + k^{-2}D_2 - h(k)\tilde{T}_1A\tilde{T}_1 - D_2AD_2$$
$$-\eta^{-1}h(k)\tilde{T}_1B\tilde{T}_1 - \eta^{-2}h(k)\tilde{T}_1CT_1 + k^6\eta^{-1}\tilde{R}_9(k)$$

where  $D_2 = (Q_2 M_{2,0} Q_2)^{-1}$  and  $\tilde{R}_9(k)$  is uniformly bounded in k.

*Proof.* Note that  $Q_2P = PQ_2 = 0$  so that we can write in the direct sum decomposition the equality

$$\begin{split} B_1(k) &= k^2 Q_1 M_{2,0} Q_1 + k^2 \eta^{-1} Q_1 M_{2,-1} Q_1 + k^4 Q_1 M_{4,0} Q_1 + k^4 \eta^{-1} Q_1 M_{4,-1} Q_1 \\ &+ k^4 \eta^{-1} Q_1 \tilde{R}_2'(k) Q_1 \\ &= k^2 \begin{pmatrix} T_1 M_{2,0} T_1 + \eta^{-1} T_1 M_{2,-1} T_1 & T_1 M_{2,0} Q_2 \\ Q_2 M_{2,0} T_1 & Q_2 M_{2,0} Q_2 \end{pmatrix} + k^4 A + k^4 \eta^{-1} B + k^4 \eta^{-2} C \\ &+ k^6 \eta^{-1} \tilde{R}_8(k) \\ &:= B_0(k) + k^4 A + k^4 \eta^{-1} B + k^4 \eta^{-2} C + k^6 \eta^{-1} \tilde{R}_8(k). \end{split}$$

Here we have defined the coefficient operators

$$A = Q_1 M_{4,0} Q_1 - Q_1 M_{2,0} D_0 M_{2,0} Q_1,$$
  

$$B = Q_1 M_{2,-1} Q_1 - Q_1 M_{2,0} D_0 M_{2,-1} Q_1 - Q_1 M_{2,-1} D_0 M_{2,0} Q_1,$$
  

$$C = -Q_1 M_{2,-1} D_0 M_{2,-1} Q_1.$$

Lemma 3.1.3 tells us that the invertibility of  $B_0(k)$  depends on the invertibility of the operator  $Q_2M_{2,0}Q_2$  (which is guaranteed by Lemma 3.2.32) and the invertibility of

$$a(k) = T_1 M_{2,0} T_1 + \eta^{-1} T_1 M_{2,-1} T_1 - T_1 M_{2,0} D_2 M_{2,0} T_1 = h(k)^{-1} T_1.$$

Thus we find  $a(k)^{-1} = h(k)T_1$  and the inverse of  $B_0(k)$  is given by

$$B_0(k)^{-1} = k^{-2}h(k) \begin{pmatrix} T_1 & -T_1M_{2,0}D_2 \\ -D_2M_{2,0}T_1 & D_2M_{2,0}T_1M_{2,0}D_2 \end{pmatrix} + k^{-2}D_2.$$

Then for sufficiently small k we can thus invert  $B_1(k)$  via a Neumann expansion to obtain

$$\begin{split} B_{1}(k)^{-1} &= (B_{0}(k) + k^{4}A + k^{4}\eta^{-1}B + k^{4}\eta^{-2}C + k^{6}\eta^{-1}\tilde{R}_{8}(k))^{-1} \\ &= B_{0}(k)^{-1}(\operatorname{Id} + k^{4}AB_{0}(k)^{-1} + k^{4}\eta^{-1}BB_{0}(k)^{-1} + k^{4}\eta^{-2}CB_{0}(k)^{-1} \\ &+ k^{6}\eta^{-1}\tilde{R}_{8}(k)B_{0}(k)^{-1})^{-1} \\ &= B_{0}(k)^{-1} - k^{4}B_{0}(k)^{-1}AB_{0}(k)^{-1} - k^{4}\eta^{-1}B_{0}(k)^{-1}BB_{0}(k)^{-1} \\ &- k^{4}\eta^{-2}B_{0}(k)^{-1}CB_{0}(k)^{-1} + k^{6}\eta^{-1}\tilde{R}_{9}(k) \\ &= k^{-2}h(k)\begin{pmatrix} T_{1} & -T_{1}M_{2,0}D_{2} \\ -D_{2}M_{2,0}T_{1} & D_{2}M_{2,0}T_{1}M_{2,0}D_{2} \end{pmatrix} + k^{-2}D_{2} - k^{4}B_{0}(k)^{-1}AB_{0}(k)^{-1} \\ &- k^{4}\eta^{-1}B_{0}(k)^{-1}BB_{0}(k)^{-1} - k^{4}\eta^{-2}B_{0}(k)^{-1}CB_{0}(k)^{-1} + k^{6}\eta^{-1}\tilde{R}_{9}(k) \\ &= k^{-2}h(k)\tilde{T}_{1} + k^{-2}D_{2} - k^{4}B_{0}(k)^{-1}AB_{0}(k)^{-1} \\ &- k^{4}\eta^{-1}B_{0}(k)^{-1}BB_{0}(k)^{-1} - k^{4}\eta^{-2}B_{0}(k)^{-1}CB_{0}(k)^{-1}) + k^{6}\eta^{-1}\tilde{R}_{9}(k), \end{split}$$

where  $\tilde{R}_9(k)$  is uniformly bounded in k. Expanding out the  $B_0(k)^{-1}$  terms we obtain

$$B_1(k)^{-1} = k^{-2}h(k)\tilde{T}_1 + k^{-2}D_2 - h(k)\tilde{T}_1A\tilde{T}_1 - D_2AD_2$$
$$-\eta^{-1}h(k)\tilde{T}_1B\tilde{T}_1 - \eta^{-1}D_2BD_2 - \eta^{-2}h(k)\tilde{T}_1C\tilde{T}_1 - \eta^{-2}D_2CD_2 + k^6\eta^{-1}\tilde{R}_9(k).$$

Noting that  $D_2Q_2 = Q_2D_2 = D_2$  and  $Q_2M_{2,-1} = M_{2,-1}Q_2 = 0$  we can further simplify to the expression

$$B_1(k)^{-1} = k^{-2}h(k)\tilde{T}_1 + k^{-2}D_2 - h(k)\tilde{T}_1A\tilde{T}_1 - D_2AD_2 - \eta^{-1}h(k)\tilde{T}_1B\tilde{T}_1 - \eta^{-2}h(k)\tilde{T}_1C\tilde{T}_1 + k^6\eta^{-1}\tilde{R}_9(k),$$

as claimed.  $\Box$ 

We are now finally ready to invert M(k) in the case of an exceptional point of the third kind.

**Theorem 3.2.37.** Suppose that n = 4 and V satisfies Assumption 3.2.22 and that zero is an exceptional point of the third kind for H. Then for sufficiently small k we have

$$M(k)^{-1} = k^{-2}Q_1D_2Q_1 + k^{-2}h(k)Q_1\tilde{T}_1Q_1 + h(k)(Q_1D_2Q_1M_{2,0}D_0 + D_0M_{2,0}Q_1D_2Q_1)$$
$$+ \eta^{-1}h(k)(D_0M_{2,-1}Q_1\tilde{T}_1Q_1 + Q_1\tilde{T}_1Q_1M_{2,-1}D_0 + Q_1\tilde{T}_1BT_1Q_1)$$
$$+ -\eta^{-1}Q_1D_2Q_1M_{4,-1}Q_1D_2Q_1 + \tilde{R}_{10}(k),$$

where  $\tilde{R}_{10}(k)$  is uniformly bounded in k.

*Proof.* Lemma 3.1.1 shows that

$$M(k)^{-1} = (M(k) + Q_1)^{-1} + (M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}.$$

Remark 3.2.29 handles the first term  $(M(k) + Q_1)^{-1}$ . Using Lemma 3.1.4 we find the relations  $Q_1D_0 = D_0Q_1 = Q_1$  and so Remark 3.2.29 allows us to show that

$$(M(k) + Q_1)^{-1}Q_1 = Q_1 - k^2 D_0 M_{2,0} Q_1 - k^2 \eta^{-1} D_0 M_{2,-1} Q_1 - k^4 D_0 M_{4,0} Q_1$$
$$- k^4 \eta^{-1} D_0 M_{4,-1} Q_1 + k^4 D_0 M_{2,0} D_0 M_{2,0} Q_1 + k^4 \eta^{-1} (D_0 M_{2,0} D_0 M_{2,-1} Q_1$$
$$+ D_0 M_{2,-1} D_0 M_{2,0} Q_1) + k^4 \eta^{-2} D_0 M_{2,-1} D_0 M_{2,-1} Q_1 - k^6 \eta^{-1} \tilde{R}'_2(k) Q_1$$

and similarly

$$Q_{1}(M(k) + Q_{1})^{-1} = Q_{1} - k^{2}Q_{1}M_{2,0}D_{0} - k^{2}\eta^{-1}Q_{1}M_{2,-1}D_{0} - k^{4}Q_{1}M_{4,0}D_{0}$$
$$- k^{4}\eta^{-1}Q_{1}M_{4,-1}D_{0} + k^{4}Q_{1}M_{2,0}D_{0}M_{2,0}D_{0} + k^{4}\eta^{-1}(Q_{1}M_{2,0}D_{0}M_{2,-1}D_{0}$$
$$+ Q_{1}M_{2,-1}D_{0}M_{2,0}D_{0}) + k^{4}\eta^{-2}Q_{1}M_{2,-1}D_{0}M_{2,-1}D_{0} - k^{6}\eta^{-1}Q_{1}\tilde{R}'_{2}(k).$$

Note that unlike in Theorem 3.2.33 many terms do not vanish. We use Lemma 3.2.36 to expand out

$$(M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}$$

$$= k^{-2}Q_1D_2Q_1 + k^{-2}h(k)Q_1\tilde{T}_1Q_1 + h(k)(Q_1D_2Q_1M_{2,0}D_0 + D_0M_{2,0}Q_1D_2Q_1)$$

$$+ \eta^{-1}(D_0M_{2,-1}Q_1D_2Q_1 + Q_1D_2Q_1M_{2,-1}D_0) + \eta^{-1}h(k)(D_0M_{2,-1}Q_1\tilde{T}_1Q_1$$

$$+ Q_1\tilde{T}_1Q_1M_{2,-1}D_0 + Q_1\tilde{T}_1B\tilde{T}_1Q_1) + \tilde{R}_{10}(k),$$

where  $\tilde{R}_{10}(k)$  is uniformly bounded in k. The relations  $D_2Q_1P = D_2Q_2P = 0$  combined with the observation that all terms in the expansion of  $(M(k) + Q_1)^{-1}$  are uniformly

bounded in k gives the statement of the theorem.

The above calculations lead us to a natural definition of resonances.

**Definition 3.2.38.** Suppose that n=4 and V satisfies Assumption 3.2.22. Then if  $Q_2 \neq Q_1 \neq 0$  we fix  $\varphi \in Q_1 \mathcal{H} \ominus Q_2 \mathcal{H}$  and note that by Lemma 3.2.25 there exists  $\psi \in H^{0,-t}$  (for some t>0) such that  $\varphi = Uv\psi$ . We say that  $\psi$  is a zero energy resonance for H. We define a normalised resonance  $\psi$  by the condition

$$||v\psi||_1 = (4\pi)^{-1}.$$

We note that the definition of resonance here differs slightly from that of [87, p. 405], although the two are equivalent. The following characterisation of the zero eigenspace is implicit in [87] and can be found as [62, Lemma 7.5].

**Lemma 3.2.39.** [62, Lemma 7.5] Suppose that n = 4 and V satisfies Assumption 3.2.22. Then the projection  $P_0$  onto the eigenspace at zero is given by  $G_{0,0}vQ_2D_2Q_2vG_{0,0}$ .

*Proof.* Let  $(\varphi_j)_{j=1}^{N_0}$  be an orthonormal basis for  $Q_2\mathcal{H}$ . Then by the definition of  $Q_1$  and the fact  $Q_2 \leq Q_1$  we have

$$0 = U(U + vG_{0,0}v)\varphi_i = (\text{Id} + UvG_{0,0}v)\varphi_i = \psi_i + UvG_{0,0}v\varphi_i.$$

Letting  $\psi_j = -G_{0,0}v\varphi_j$  we see that the  $\psi_j$  are linearly independent and that  $\varphi_j = Uv\psi_j$ . Thus for any  $f \in \mathcal{H}$  we have

$$Q_2vG_{0,0}f = \sum_{j=1}^{N_0} \langle Q_2vG_{0,0}f, \varphi_j \rangle \varphi_j = \sum_{j=1}^{N_0} \langle f, G_{0,0}v\varphi_j \rangle \varphi_j = -\sum_{j=1}^{N_0} \langle f, \psi_j \rangle \varphi_j.$$

Let  $(A_{ij})$  be the matrix representation of the operator  $Q_2M_{4,-1}Q_2$ . Then we find

$$A_{ij} = \langle \varphi_i, Q_2 M_{4,-1} Q_2 \varphi_j \rangle = \langle G_{0,0} v \varphi_i, G_{0,0} v \varphi_j \rangle = \langle G_{0,0} V \psi_i, G_{0,0} V \psi_j \rangle = \langle \psi_i, \psi_j \rangle.$$

Then for  $f \in \mathcal{H}$  we have

$$G_{0,0}vQ_{2}D_{2}Q_{2}vG_{0,0}f = -\left(\sum_{j=1}^{N_{0}} \langle f, \psi_{j} \rangle G_{0,0}vQ_{2}D_{2}Q_{2}\varphi_{j}\right) = -\sum_{i,j=1}^{N_{0}} G_{0,0}vQ_{2}(A_{ij}^{-1})\varphi_{i}\langle f, \psi_{j} \rangle$$

$$= -\sum_{i,j=1}^{N_{0}} G_{0,0}v(A_{ij}^{-1})\varphi_{i}\langle f, \psi_{j} \rangle = \sum_{i,j=1}^{N_{0}} (A_{ij}^{-1})\psi_{i}\langle f, \psi_{j} \rangle.$$

In particular for  $f = \psi_m$  we find  $G_{0,0}vQ_2D_2Q_2vG_{0,0}\psi_m = \psi_m$ . So  $G_{0,0}vQ_2D_2Q_2vG_{0,0}$  is the identity on span $(\psi_j)_{j=1}^{N_0}$  and is the identity on  $Q_2\mathcal{H}$  and is thus the projection onto the zero eigenspace of H.

We conclude this section by determining a low energy expansion of the resolvent  $R(-k^2)$ .

**Theorem 3.2.40.** Suppose n = 4 and V satisfies Assumption 3.2.22. Then for sufficiently small k we have

$$R(-k^{2}) - R_{0}(-k^{2}) = -k^{-2}P_{0} - k^{-2}h(k)G_{0,0}vQ_{1}\tilde{T}_{1}Q_{1}vG_{0,0} + \eta^{-1}(G_{0,0}v\eta^{-1}Q_{1}D_{2}Q_{1}M_{4,-1}Q_{1}D_{2}Q_{1}vG_{0,0}) + \tilde{R}(k),$$

where  $\tilde{R}(k)$  is uniformly bounded in k.

*Proof.* We begin with the relation

$$R(-k^2) - R_0(-k^2) = -R_0(-k^2)vM(k)^{-1}vR_0(-k^2).$$

Multiplying the expansion of  $R_0(-k^2)$  in Equation (3.21) with the expansion of  $M(k)^{-1}$  found in Theorem 3.2.37 yields the result immediately.

## **3.2.4** Dimension n = 3

In this section we demonstrate the expansion of the operator  $M(k)^{-1}$  near k=0 in dimension n=3 and use this to define a zero-energy resonance. The results of this section are a straightforward application of the results of [89]. Such expansions have been considered in a different context in [64] and first appeared in a non-symmetrised form in [88].

The assumption on the potential required in this section is the following.

**Assumption 3.2.41.** Suppose that n = 3. We assume that V satisfies Assumption 2.2.14 for some  $\rho > 5$ .

In dimension n = 3 Equation (3.12) reads

$$R_0(x, y, -k^2) = |x - y|^{-1} \sum_{p=0}^{\infty} c_{3,p} k^p |x - y|^p.$$
(3.30)

The above expansion leads to the following operator definitions.

**Definition 3.2.42.** Suppose that n = 3 and V satisfies Assumption 3.2.41. We define for  $p \in \mathbb{N}$  the integral operators  $G_i$  by the kernels

$$G_p(x,y) = (4\pi p!)^{-1}|x-y|^{p-1}$$

and for  $p \geq 1$  the integral operators  $M_p$  by the kernels

$$M_p(x,y) = i^p v(x) G_p(x,y) v(y).$$

We also define the operator  $M_0$  by the kernel

$$(M_0 - U)(x, y) = v(x)G_0(x, y)v(y).$$

Then from the definition of M(k) we have

$$M(k) = U + vR_0(-k^2)v = M_0 + kM_1 + k^2M_2 + k^3\tilde{R}_0(k),$$

where  $\tilde{R}_0(k)$  is uniformly bounded for  $k \in F = \{k \in \mathbb{C} : \text{Re}(k) \ge 0 \text{ and } |k| \le 1\}.$ 

**Lemma 3.2.43.** Suppose that n = 3 and V satisfies Assumption 3.2.41. Then for  $k \in F$  the operator  $M(k) - M_0$  is Hilbert-Schmidt.

*Proof.* The operator  $M(k) - M_0$  has the kernel

$$(M(k) - M_0)(x, y) = k^{-1}v(x)(4\pi)^{-1}|x - y|^{-1}(e^{ik|x - y|} - 1)v(y)$$

and thus we have the estimate

$$|(M(k) - M_0)(x, y)| \le (4\pi)^{-1} |v(x)| |v(y)|.$$

Hence we see that  $M(k) - M_0 \in \mathcal{L}^2(\mathcal{H})$  provided  $|v(x)| \leq C(1+|x|)^{-\frac{\rho}{2}}$  for some  $\rho > 5$ , which is our assumption on V.

We can further define the projection  $P: \mathcal{H} \to \mathcal{H}$  by

$$[Pf](x) = ||v||_2^{-2} v(x) \int_{\mathbb{R}^3} v(y) f(y) \, dy = ||v||_2^{-2} \langle v, f \rangle v(x).$$

The projection P is related to  $M_1$  by the formula  $M_1 = i(4\pi)^{-1} ||v||_2^2 P$ .

Note that  $M_0$  is a compact self-adjoint perturbation of U and thus has essential spectrum  $\sigma_{ess}(M_0) \subset \{-1,1\}$ . So 0 is an isolated point of the spectrum of  $M_0$  and  $\text{Dim}(\text{Ker}(M_0)) < \infty$ . These considerations lead us to the following definition.

**Definition 3.2.44.** Suppose that n = 3 and V satisfies Assumption 3.2.41. We say that zero is a regular point of the spectrum of H if the operator  $M_0$  is invertible. If zero is not a regular point of the spectrum of H, we define  $Q_1$  to be the orthogonal projection onto the kernel of  $M_0$ . Then  $M_0 + Q_1$  is invertible and we can define  $D_0 = (M_0 + Q)^{-1}$ . We say there is an exceptional point of the first kind if  $Q_1M_1Q_1$  is invertible on  $Q_1H$ . If  $Q_1M_1Q_1$  is not invertible, we define  $Q_2$  to be the orthogonal projection onto the kernel of  $Q_1M_1Q_1$  (as an operator on  $Q_1H$ ). Then  $Q_1M_1Q_1 + Q_2$  is invertible (with inverse defined to be  $D_1$ ) on  $Q_1H$ . If  $Q_1 = Q_2$  we say there is an exceptional point of the second kind. Otherwise, we say there is an exceptional point of the third kind.

We will show  $Q_2 \neq 0$  (an exceptional point of the second kind) corresponds to the existence of an eigenvalue at zero whilst  $Q_1 \neq 0$  corresponds to the existence of a resonance in dimension n = 3. By definition, the projections  $Q_1$  and  $Q_2$  are finite rank operators and  $Q_2 \leq Q_1$ . The following expansion will be used in each exceptional case.

**Lemma 3.2.45.** Suppose that n = 3 and V satisfies Assumption 3.2.41 and that zero is not a regular point of the spectrum of H. Then for sufficiently small k we have the expansion

$$(M(k) + Q_1)^{-1} = D_0 - kD_0M_1D_0 - k^2D_0M_2D_0 + k^2D_0M_1D_0M_1D_0 + k^3\tilde{R}_1(k)$$

where  $\tilde{R}_1(k)$  is uniformly bounded in k and  $D_0 = (M_0 + Q_1)^{-1}$ .

*Proof.* For sufficiently small k we have the estimate

$$\left| \left| k M_1 D_0 + k^2 M_2 D_0 + k^3 \tilde{R}_0(k) D_0 \right| \right| < 1,$$

since each of the operators is bounded. We are thus able to factor and use a Neumann expansion to find

$$(M(k) + Q_1)^{-1} = \left(M_0 + kM_1 + k^2M_2 + k^3\tilde{R}_0(k) + Q_1\right)^{-1}$$

$$= D_0 \left(\operatorname{Id} + kM_1D_0 + k^2M_2D_0 + k^3\tilde{R}_0(k)D_0\right)^{-1}$$

$$= D_0 \left(\operatorname{Id} - kM_1D_0 - k^2M_2D_0 - k^3\tilde{R}_0(k)D_0 + \sum_{j=2}^{\infty} (-1)^j ((M(k) - M_0)D_0)^j\right)$$

$$= D_0 - kD_0M_1D_0 - k^2D_0M_2D_0 - k^3D_0\tilde{R}_0(k)D_0 + \sum_{j=2}^{\infty} (-1)^j D_0((M(k) - M_0)D_0)^j$$

Computing the lowest order term in the sum as  $k^2D_0M_1D_0M_1D_0$  and defining

$$\tilde{R}_1(k) = -D_0 M_1 D_0 M_1 D_0 - k D_0 \tilde{R}_0(k) D_0 + k^{-2} \sum_{j=2}^{\infty} (-1)^j D_0 ((M(k) - M_0) D_0)^j$$

completes the proof.  $\Box$ 

Remark 3.2.46. We can further compute the lowest order terms in  $\tilde{R}_1(k)$  to obtain

$$\tilde{R}_1(k) = -D_0 M_1 D_0 M_1 D_0 - k(D_0 M_3 D_0 + D_0 M_1 D_0 M_1 D_0 M_1 D_0) + k^2 \tilde{R}(k).$$

Corollary 3.1.2 shows that M(k) is invertible if and only if the operator

$$B_1(k) = k^{-1} \left( Q_1 - Q_1 (M(k) + Q_1)^{-1} Q_1 \right)$$
(3.31)

is invertible on  $Q_1\mathcal{H}$  and in this case

$$M(k)^{-1} = (M(k) + Q_1)^{-1} + k^{-1}(M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}.$$

So we now investigate the invertibility of the operator  $B_1(k)$  for sufficiently small k.

**Lemma 3.2.47.** Suppose that n = 3 and V satisfies Assumption 3.2.41 and that zero is an exceptional point of the first kind for H. Then for sufficiently small k we have the expansion

$$B_1(k) = Q_1 M_1 Q_1 + k Q_1 M_2 Q_1 - k Q_1 M_1 D_0 M_1 Q_1 - k^2 Q_1 \tilde{R}_1(k) Q_1.$$

*Proof.* We substitute the result of Lemma 3.2.45 into the definition of  $B_1(k)$  in Equation (3.31) to obtain

$$kB_{1}(k) = Q_{1} - Q_{1}(M(k) + Q_{1})^{-1}Q_{1}$$

$$= Q_{1} - Q_{1}\left(D_{0} - kD_{0}M_{1}D_{0} - k^{2}D_{0}M_{2}D_{0} + k^{2}D_{0}M_{1}D_{0}M_{1}D_{0} + k^{3}\tilde{R}_{1}(k)\right)Q_{1}$$

$$= Q_{1} - Q_{1}D_{0}Q_{1} + kQ_{1}D_{0}M_{1}D_{0}Q_{1} + k^{2}Q_{1}D_{0}M_{2}D_{0}Q_{1}$$

$$- k^{2}Q_{1}M_{1}D_{0}M_{1}Q_{1} - k^{3}Q_{1}\tilde{R}_{1}(k)Q_{1}$$

$$= kQ_{1}M_{1}Q_{1} + k^{2}Q_{1}M_{2}Q_{1} - k^{2}Q_{1}M_{1}D_{0}M_{1}Q_{1} - k^{3}Q_{1}\tilde{R}_{1}(k)Q_{1},$$

where we have used Lemma 3.1.4 several times.

If the term  $Q_1M_1Q_1$  is invertible (there is an exceptional point of the first kind) then we can invert  $B_1(k)$  for small k using a Neumann series and thus obtain an expansion for  $M(k)^{-1}$  also. Since the operator  $Q_1M_1Q_1$  has rank one, this can only occur if  $\operatorname{Rank}(Q_1) = 1$ . In this case we can define the operator  $D_1 = (Q_1M_1Q_1)^{-1}$ . The result is the following.

**Lemma 3.2.48.** Suppose that V satisfies Assumption 3.2.41 and that zero is an exceptional point of the first kind for H. Then for sufficiently small k we have the expansion

$$B_1(k)^{-1} = D_1 - kD_1Q_1M_2Q_1D_1 + kD_1Q_1M_1D_0M_1Q_1D_1 + k^2\tilde{R}_2(k),$$

where  $\tilde{R}_2(k)$  is uniformly bounded in k and  $D_1 = (Q_1 M_1 Q_1)^{-1}$ .

*Proof.* We compute that

$$\begin{split} B_{1}(k)^{-1} &= \left(Q_{1}M_{1}Q_{1} + kQ_{1}M_{2}Q_{1} - kQ_{1}M_{1}D_{0}M_{1}Q_{1} - k^{2}Q_{1}\tilde{R}_{1}(k)Q_{1}\right)^{-1} \\ &= D_{1}\left(\operatorname{Id} + kQ_{1}M_{2}Q_{1}D_{1} - kQ_{1}M_{1}D_{0}M_{1}Q_{1}D_{1} - k^{2}Q_{1}\tilde{R}_{1}(k)Q_{1}D_{1}\right)^{-1} \\ &= D_{1}\left(\operatorname{Id} - kQ_{1}M_{2}Q_{1}D_{1} + kQ_{1}M_{1}D_{0}M_{1}Q_{1}D_{1} + k^{2}Q_{1}\tilde{R}_{1}(k)Q_{1}D_{1}\right) \\ &+ \sum_{j=2}^{\infty} (-1)^{j}\left(kQ_{1}M_{2}Q_{1}D_{1} - kQ_{1}M_{1}D_{0}M_{1}Q_{1}D_{1} - k^{2}Q_{1}\tilde{R}_{1}(k)Q_{1}D_{1}\right)^{j}\right) \\ &= D_{1} - kD_{1}Q_{1}M_{2}Q_{1}D_{1} + kQ_{1}M_{1}D_{0}M_{1}Q_{1}D_{1} + k^{2}D_{1}Q_{1}\tilde{R}_{1}(k)Q_{1}D_{1} \\ &+ \sum_{j=2}^{\infty} (-1)^{j}D_{1}\left(kQ_{1}M_{2}Q_{1}D_{1} + kQ_{1}M_{1}D_{0}M_{1}Q_{1}D_{1} - k^{2}Q_{1}\tilde{R}_{1}(k)Q_{1}D_{1}\right)^{j}. \end{split}$$

Defining

$$\tilde{R}_2(k) = D_1 Q_1 \tilde{R}_1(k) Q_1 D_1 + k^{-2} \sum_{j=2}^{\infty} (-1)^j D_1 \left( k Q_1 M_2 Q_1 D_1 - k Q_1 \tilde{R}_1(k) Q_1 D_1 \right)^j$$

completes the proof.

We now have the following expansion for  $M(k)^{-1}$ .

**Theorem 3.2.49.** Suppose that n = 3 and V satisfies Assumption 3.2.41 and that zero is an exceptional point of the first kind for H. Then for sufficiently small k we have the expansion

$$M(k)^{-1} = k^{-1}Q_1D_1Q_1 - Q_1D_1Q_1M_2Q_1D_1Q_1 + Q_1D_1Q_1M_1D_0M_1Q_1D_1Q_1$$
$$- D_0M_1Q_1D_1Q_1 + D_0 + k\tilde{R}_3(k),$$

where  $\tilde{R}_3(k)$  is uniformly bounded in k and  $D_1 = (Q_1 M_1 Q_1)^{-1}$ .

*Proof.* Lemma 3.2.45 shows that  $M(k) + Q_1$  is invertible and we can compute that

$$(M(k) + Q_1)^{-1}Q_1 = Q_1 - kD_0M_1Q_1 + k^2D_0M_2Q_1 - k^2D_0M_1D_0M_1Q_1 + k^3\tilde{R}_1(k)Q_1,$$

and

$$Q_1(M(k) + Q_1)^{-1} = Q_1 - kQ_1M_1D_0 + k^2Q_1M_2D_0 - k^2Q_1M_1D_0M_1D_0 + k^3Q_1\tilde{R}_1(k).$$

Lemma 3.2.48 shows that  $B_1(k)$  is invertible on  $Q_1\mathcal{H}$  and thus we have by Lemmas 3.1.1

and 3.2.45 the expansion

$$M(k)^{-1} = (M(k) + Q_1)^{-1} + k^{-1}(M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}$$

$$= D_0 - kD_0M_1D_0 + k^2D_0M_2D_0 - k^2D_0M_1D_0M_1D_0 + k^3\tilde{R}_1(k)$$

$$+ k^{-1}\left(Q_1 - kD_0M_1Q_1 + k^2D_0M_2Q_1 - k^2D_0M_1D_0M_1Q_1 + k^3\tilde{R}_1(k)Q_1\right)$$

$$\times (D_1 - kD_1Q_1M_2Q_1D_1 + kD_1Q_1M_1D_0M_1Q_1D_1 + k^2\tilde{R}_2(k))$$

$$\times \left(Q_1 - kQ_1M_1D_0 + k^2Q_1M_2D_0 - k^2Q_1M_1D_0M_1D_0 + k^3Q_1\tilde{R}_1(k)\right).$$

Explicitly computing the lowest order terms in the product we obtain the statement of the theorem.  $\Box$ 

If  $Q_1M_1Q_1$  is not invertible, we define  $Q_2: Q_1\mathcal{H} \to Q_1\mathcal{H}$  to be the orthogonal projection onto the kernel of  $Q_1M_1Q_1$ . For sufficiently small k we have  $B_1(k) + Q_2$  is invertible in  $Q_1\mathcal{H}$ . We define the operator  $D_1 = (Q_1M_1Q_1 + Q_2)^{-1}$ .

Note also that the fact  $Q_2 \leq Q_1$  and the definition of  $Q_2$  imply  $0 = Q_1 M_1 Q_2 = Q_1 P Q_2$ . Lemma 3.1.1 tells us that  $B_1(k)$  is invertible if and only if

$$B_2(k) = (Q_2 - Q_2(B_1(k) + Q_2)^{-1}Q_2)$$
(3.32)

is invertible on  $Q_2\mathcal{H}$  and in this case we have

$$B_1(k)^{-1} = (B_1(k) + Q_2)^{-1} + k^{-1}(B_1(k) + Q_2)^{-1}Q_2B_2(k)^{-1}Q_2(B_1(k) + Q_2)^{-1}.$$

**Lemma 3.2.50.** Suppose that n = 3 and V satisfies Assumption 3.2.41 and that zero is an exceptional point of the third kind for H. Then for sufficiently small k we have the expansion

$$(B_1(k) + Q_2)^{-1} = D_1 - kD_1Q_1M_2Q_1D_1 - kD_1Q_1M_1D_0M_1Q_1D_1 + k^2\tilde{R}_4(k),$$

where  $\tilde{R}_4(k)$  is uniformly bounded in k and  $D_1 = (Q_1M_1Q_1 + Q_2)^{-1}$ .

*Proof.* Since each of the coefficient operators is bounded, we have for sufficiently small k the estimate

$$\left| \left| kQ_1 M_2 Q_1 D_1 - kQ_1 M_1 D_0 M_1 Q_1 D_1 - k^2 Q_1 \tilde{R}_1(k) D_1 \right| \right| < 1.$$

Thus we make the computation

$$(B_{1}(k) + Q_{2})^{-1} = \left(Q_{1}M_{1}Q_{1} + kQ_{1}M_{2}Q_{1} - kQ_{1}M_{1}D_{0}M_{1}Q_{1} - k^{2}Q_{1}\tilde{R}_{1}(k)Q_{1} + Q_{2}\right)^{-1}$$

$$= D_{1}\left(\operatorname{Id} + kQ_{1}M_{2}Q_{1}D_{1} - kQ_{1}M_{1}D_{0}M_{1}Q_{1}D_{1} - k^{2}Q_{1}\tilde{R}_{1}(k)D_{1}\right)^{-1}$$

$$= D_{1}\left(\operatorname{Id} - kQ_{1}M_{2}Q_{1}D_{1} + kQ_{1}M_{1}D_{0}M_{1}Q_{1}D_{1} + k^{2}Q_{1}\tilde{R}_{1}(k)D_{1}\right)$$

$$+ \sum_{j=2}^{\infty} (-1)^{j}\left(kQ_{1}M_{2}Q_{1}D_{1} - kQ_{1}M_{1}D_{0}M_{1}Q_{1}D_{1} - k^{2}Q_{1}\tilde{R}_{1}(k)D_{1}\right)^{j}\right)$$

$$= D_{1} - kD_{1}Q_{1}M_{2}Q_{1}D_{1} + kD_{1}Q_{1}M_{1}D_{0}M_{1}Q_{1}D_{1} + k^{2}D_{1}Q_{1}\tilde{R}_{1}(k)D_{1}$$

$$+ \sum_{j=2}^{\infty} (-1)^{j}D_{1}\left(kQ_{1}M_{2}Q_{1}D_{1} - kQ_{1}M_{1}D_{0}M_{1}Q_{1}D_{1} - k^{2}Q_{1}\tilde{R}_{1}(k)D_{1}\right)^{j}$$

where we have used Lemma 3.2.48 and a Neumann expansion. Defining

 $\tilde{R}_4(k)$ 

$$=D_1Q_1\tilde{R}_1(k)D_1+k^{-2}\sum_{j=2}^{\infty}(-1)^jD_1\left(kQ_1M_2Q_1D_1-kQ_1M_1D_0M_1Q_1-k^2Q_1\tilde{R}_1(k)D_1\right)^j$$

completes the proof.

**Lemma 3.2.51.** Suppose that n = 3 and V satisfies Assumption 3.2.41 and that zero is an exceptional point of the third kind for H. Then for sufficiently small k we have the expansion

$$B_2(k) = Q_2 M_2 Q_2 - Q_2 k \tilde{R}_4(k) Q_2.$$

*Proof.* We use Lemmas 3.2.50 and 3.1.4 and the definition of  $B_2(k)$  to obtain

$$\begin{split} B_2(k) &= k^{-1}(Q_2 - Q_2(B_1(k) + Q_2)^{-1}Q_2) \\ &= k^{-1}(Q_2 - Q_2(D_1 - kQ_1M_2Q_1D_1 + kD_1Q_1M_1D_0M_1Q_1D_1 + k^2\tilde{R}_4(k))Q_2) \\ &= k^{-1}(Q_2 - Q_2D_1Q_2 + kQ_2Q_1M_2Q_1Q_2 - kQ_2M_1D_0M_1Q_2 - k^2Q_2\tilde{R}_4(k)Q_2) \\ &= Q_2M_2Q_2 - Q_2M_1D_0M_1Q_2 - Q_2k\tilde{R}_4(k)Q_2, \end{split}$$

where we have used the relations  $Q_1Q_1=Q_1Q_2=Q_2$  and  $Q_2D_1Q_2=Q_2$ . Note also that  $Q_2M_1=0$ . To complete the proof we note by [88, Lemma 2.2] that we have  $G_2 \in \mathcal{L}^2(H^{0,t},H^{0,-t})$  for  $t>\frac{5}{2}$  and thus we have  $Q_2M_2Q_2 \in \mathcal{L}^2(\mathcal{H})$  provided  $\rho>5$  in Assumption 2.2.14.

We now characterise the range of the projections  $Q_1$  and  $Q_2$ . The next two results can be inferred from related results in [88] and can also be found in [64, Lemmas 5 and 6].

**Lemma 3.2.52.** [64, Lemma 5] Suppose that n = 3 and V satisfies Assumption 3.2.41. Then  $0 \neq f \in Q_1\mathcal{H}$  if and only if there exists  $t > \frac{1}{2}$  and  $g \in H^{0,-t}$  such that f = Uvg and Hg = 0 in the sense of distributions.

*Proof.* By [88, Lemma 2.4] we find that  $g \in H^{0,-t}$  satisfies Hg = 0 if and only if

$$(\mathrm{Id} + G_0 V)g = 0.$$

Fix  $0 \neq f \in Q_1 \mathcal{H}$  so that

$$0 = [M_0 f](x) = U(x)f(x) + (4\pi)^{-1}v(x) \int_{\mathbb{R}^3} |x - y|^{-1}v(y)f(y) \, dy.$$

Multiplying by U(x) we obtain

$$f(x) = -(4\pi)^{-1}U(x)v(x) \int_{\mathbb{R}^3} |x - y|^{-1}v(y)f(y) \, dy.$$

Defining  $g \in H^{0,-t}$  by

$$g(x) = -(4\pi)^{-1} \int_{\mathbb{R}^3} |x - y|^{-1} v(y) f(y) \, dy$$
 (3.33)

we have f = Uvg. To see that Hg = 0 holds we compute that

$$g(x) = -(4\pi)^{-1} \int_{\mathbb{R}^3} |x - y|^{-1} v(y) f(y) \, dy = -(4\pi)^{-1} \int_{\mathbb{R}^3} |x - y|^{-1} V(x) g(y) \, dy$$
$$= -[G_0 V g](x).$$

Conversely, we suppose that f = Uvg for some  $0 \neq g \in H^{0,-t}$  satisfying Equation Hg = 0. Then  $f \in H^{0,2t}$  and applying  $M_0$  we find

$$[M_0 f](x) = U(x)f(x) + (4\pi)^{-1}v(x) \int_{\mathbb{R}^3} |x - y|^{-1}v(y)f(y) \, dy$$
$$= v(x) \left( g(x) + (4\pi)^{-1} \int_{\mathbb{R}^3} |x - y|^{-1}V(y)g(y) \, dy \right)$$
$$= v(x)[(\mathrm{Id} + G_0 V)g](x)$$
$$= 0.$$

Thus we have  $0 \neq f \in Q_1 \mathcal{H}$ .

**Lemma 3.2.53.** [64, Lemma 6] Suppose that n = 3 and V satisfies Assumption 3.2.41. Then  $0 \neq f \in Q_2\mathcal{H}$  if and only if there exists  $g \in \mathcal{H}$  such that f = Uvg and Hg = 0 in the sense of distributions.

*Proof.* Suppose  $0 \neq f \in Q_2\mathcal{H}$ . Since  $Q_2 \leq Q_1$  we have that by Lemma 3.2.52 that

f = Uvg for some  $g \in H^{0,-t}$  with  $t > \frac{1}{2}$  and Hg = 0 in the sense of distributions. By the definition of  $Q_2$  we find  $Q_1Pf = 0$ , which holds if and only if either  $Q_1v = 0$  or Pf = 0. In either case we have

$$0 = \int_{\mathbb{R}^3} v(y)f(y) \, \mathrm{d}y. \tag{3.34}$$

Using Equation (3.34) in Equation (3.33) we find that g satisfies

$$g(x) = -(4\pi)^{-1} \int_{\mathbb{R}^3} (|x - y|^{-1} - (1 + |x|)^{-1}) v(y) f(y) dy.$$

Define the operator  $L: C_c^{\infty}(\mathbb{R}^3) \to C_c^{\infty}(\mathbb{R}^3)$  by the integral kernel  $L(x,y) = \frac{1+|y|}{|x-y|(1+|x|)}$ . By [88, Lemma 2.5] we have  $L: H^{0,t} \to H^{2,t-2}$ . Then for  $t > \frac{3}{2}$  we have that

$$\left| |x - y|^{-1} - (1 + |x|)^{-1} \right| \le \frac{1 + |y|}{|x - y|(1 + |x|)} \tag{3.35}$$

which when combined with the fact that  $f = Uvg \in H^{0,\frac{\rho}{2}+t}$  shows that  $g \in H^{0,t}$  and thus  $g \in H^{2,t-2} \subset \mathcal{H}$ . For the converse, we assume f = wg for some  $0 \neq g \in \mathcal{H}$  satisfying Hg = 0. Then we may write

$$g(x)$$

$$= -(4\pi)^{-1} \int_{\mathbb{R}^3} (|x-y|^{-1} - (1+|x|)^{-1}) v(y) f(y) dy + (4\pi)^{-1} (1+|x|)^{-1} \int_{\mathbb{R}^3} v(y) f(y) dy.$$

The estimate (3.35) shows that the first term is square integrable and thus we find the function  $h: \mathbb{R}^3 \to \mathbb{C}$  defined by

$$h(x) = (4\pi)^{-1}(1+|x|)^{-1} \int_{\mathbb{R}^3} v(y)f(y) \, \mathrm{d}y$$

is square integrable also since  $(x \mapsto (1+|x|)^{-1} \notin L^2(\mathbb{R}^3)$  we have

$$0 = \int_{\mathbb{R}^3} v(y)f(y) \,\mathrm{d}y. \tag{3.36}$$

As discussed at the beginning of the proof, Equation (3.36) is equivalent to  $f \in Q_2\mathcal{H}$ .  $\square$ 

The proof of Lemma 3.2.53 allows us to show that the Feshbach expansion terminates after only two applications and thus invert the operator  $B_2(k)$ .

**Lemma 3.2.54.** [64, Lemma 7] Suppose that n = 3 and V satisfies Assumption 3.2.41 and that zero is an exceptional point of the third kind for H. Then the operator  $Q_2M_2Q_2$  is invertible in  $Q_2\mathcal{H}$ .

*Proof.* Suppose that  $f \in Q_2\mathcal{H}$  and  $Q_2M_2Q_2f = 0$ . Then we find

$$0 = \langle Q_2 M_2 Q_2 f, f \rangle = \langle M_2 f, f \rangle = \langle G_2 v f, v f \rangle.$$

The proof of Lemma 3.2.53 shows that we also have

$$0 = \int_{\mathbb{R}^3} v(y) f(y) \, \mathrm{d}y.$$

Note also that  $vf \in L^1(\mathbb{R}^n)$  and  $x \mapsto |x|vf \in L^1(\mathbb{R}^n)$ , so that  $\mathcal{F}(vf) \in L^1(\mathbb{R}^n) \cap C^1(\mathbb{R}^n)$ . Thus we may compute the limits

$$\langle G_{2}vf, vf \rangle = \lim_{k \to 0} k^{-2} \langle (R_{0}(k^{2}) - G_{0})vf, vf \rangle$$

$$= \lim_{k \to 0} k^{-2} \int_{\mathbb{R}^{3}} ((|\xi|^{2} + k^{2})^{-1} - |\xi|^{-2}) \overline{[\mathcal{F}(vf)](\xi)} [\mathcal{F}(vf)](\xi) \, d\xi$$

$$= \lim_{k \to 0} \int_{\mathbb{R}^{3}} (|\xi|^{2} (|\xi|^{2} + k^{2}))^{-1} |[\mathcal{F}(vf)](\xi)|^{2} \, d\xi$$

$$= \int_{\mathbb{R}^{3}} |\xi|^{-4} |[\mathcal{F}(vf)](\xi)|^{2} \, d\xi = \langle G_{0}vf, G_{0}vf \rangle,$$

where we have used Lemma 3.1.6 to bring the limit inside the integral. Since  $Q_2M_2Q_2f = 0$  we thus find  $\mathcal{F}(vf) = 0$  and so vf = 0, which further implies f = 0 and so  $Q_2M_2Q_2$  is invertible in  $Q_2\mathcal{H}$ .

Since the operator  $Q_2M_2Q_2$  is invertible in  $Q_2\mathcal{H}$  we can define  $D_2=(Q_2M_2Q_2)^{-1}$ .

**Lemma 3.2.55.** Suppose that n = 3 and V satisfies Assumption 3.2.41 and that zero is an exceptional point of the third kind for H. Then for sufficiently small k we have the expansion

$$B_2(k)^{-1} = D_2 + kB + k^2 \tilde{R}_5(k)$$

where  $\tilde{R}_5(k)$  is uniformly bounded and

$$B = -D_2 Q_2 M_3 Q_2 D_2 + D_2 Q_2 M_2 Q_1 D_1 Q_1 M_2 Q_2 D_2.$$

*Proof.* For sufficiently small k we have the estimate

$$\left| \left| kQ_2 \tilde{R}_4(k) Q_2 D_2 \right| \right| < 1.$$

We thus compute the Neumann expansion

$$B_{2}(k)^{-1} = \left(Q_{2}M_{2}Q_{2} - kQ_{2}\tilde{R}_{4}(k)Q_{2}\right)^{-1} = D_{2}\left(\operatorname{Id} - kQ_{2}\tilde{R}_{4}(k)Q_{2}D_{2}\right)^{-1}$$

$$= D_{2}\left(\operatorname{Id} + \sum_{j=1}^{\infty} k^{j}(Q_{2}\tilde{R}_{4}(k)Q_{2}D_{2})^{j}\right)$$

$$= D_{2} + kD_{2}Q_{2}\tilde{R}_{4}(k)Q_{2}D_{2} + \sum_{j=2}^{\infty} k^{j}D_{2}(Q_{2}\tilde{R}_{4}(k)Q_{2}D_{2})^{j}.$$

The lowest order term in  $\tilde{R}_4(k)$  can be computed explicitly (see Remark 3.2.46) as

$$A = -D_0 M_3 D_0 - D_1 Q_1 D_0 M_1 D_0 M_1 D_0 M_1 D_0 D_1 + D_1 (Q_1 M_2 Q_1 D_1 - Q_1 M_1 D_0 M_1 Q_1)^2.$$

The relations  $Q_2D_1 = D_1Q_2 = Q_2$ ,  $Q_2M_1 = M_1Q_2 = 0$  and  $Q_2 \leq Q_1$  then give us

$$D_2Q_2AQ_2D_2 = -D_2Q_2M_3Q_2D_2 + D_2Q_2M_2Q_1D_1Q_1M_2Q_2D_2.$$

Defining 
$$\tilde{R}_5(k) = k^{-2} \sum_{i=2}^{\infty} D_2(Q_2 \tilde{R}_4(k) Q_2 D_2)^j$$
 completes the proof.

**Lemma 3.2.56.** Suppose that n = 3 and V satisfies Assumption 3.2.41 and that zero is an exceptional point of the third kind for H. Then for sufficiently small k we have the expansion

$$B_1(k)^{-1} = -k^{-1}Q_2D_2Q_2 + D_1 + Q_2BQ_2 - Q_2D_2Q_2M_2Q_1D_1 - D_1Q_1M_2Q_2D_2Q_2 + k\tilde{R}_6(k),$$

where  $\tilde{R}_6(k)$  is uniformly bounded.

Proof. Combining Lemmas 3.2.50 and 3.2.55 gives

$$B_{1}(k)^{-1} = (B_{1}(k) + Q_{2})^{-1} + k^{-1}(B_{1}(k) + Q_{2})^{-1}Q_{2}B_{2}(k)^{-1}Q_{2}(B_{1}(k) + Q_{2})^{-1}$$

$$= D_{1} - kD_{1}Q_{1}M_{2}Q_{1}D_{1} + kD_{1}Q_{1}M_{1}D_{0}M_{1}Q_{1}D_{1} + k^{2}\tilde{R}_{4}(k)$$

$$+ k^{-1}(D_{1} - kD_{1}Q_{1}M_{2}Q_{1}D_{1} + kD_{1}Q_{1}M_{1}D_{0}M_{1}Q_{1}D_{1} + k^{2}\tilde{R}_{4}(k))$$

$$\times Q_{2}(D_{2} + kB + k^{2}\tilde{R}_{5}(k))$$

$$\times Q_{2}(D_{1} - kD_{1}Q_{1}M_{2}Q_{1}D_{1} + kD_{1}Q_{1}M_{1}D_{0}M_{1}Q_{1}D_{1} + k^{2}\tilde{R}_{4}(k))$$

Collecting all lower order terms gives the result.

We can now compute an expansion for  $M(k)^{-1}$ .

**Theorem 3.2.57.** Suppose that n = 3 and V satisfies Assumption 3.2.41 and that zero is an exceptional point of the third kind for H. Then for sufficiently small k we have the

expansion

$$M(k)^{-1} = k^{-2}Q_2D_2Q_2 + k^{-1}C_{-\frac{1}{2}} + \tilde{R}_8(k),$$

where  $\tilde{R}_7(k)$  is uniformly bounded and

$$C_{-\frac{1}{2}} = D_1 + D_2 M_2 D_1 M_2 D_2 + D_2 M_3 D_2 - Q_1 M_2 D_2 - D_2 M_2 Q_1.$$

*Proof.* We begin with the fact that

$$M(k)^{-1} = (M(k) + Q_1)^{-1} + k^{-1}(M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}$$
(3.37)

So we compute the products

$$(M(k) + Q_1)^{-1}Q_1 = Q_1 - kD_0M_1Q_1 + k^2D_0M_2Q_1 - k^2D_0M_1D_0M_1Q_1 + k^3\tilde{R}_1(k)Q_1, \text{ and}$$

$$Q_1(M(k) + Q_1)^{-1} = Q_1 - kQ_1M_1D_0 + k^2Q_1M_2D_0 - k^2Q_1M_1D_0M_1D_0 + k^3Q_1\tilde{R}_1(k).$$

We are interested only in terms in the product with coefficient  $k^{-2}$  or  $k^{-1}$ . Expanding out the product we obtain

$$k^{-1}(M(k) + Q_{1})^{-1}Q_{1}B_{1}(k)^{-1}Q_{1}(M(k) + Q_{1})^{-1}$$

$$= k^{-1}(Q_{1} - kD_{0}M_{1}Q_{1} + k^{2}D_{0}M_{2}Q_{1} - k^{2}D_{0}M_{1}D_{0}M_{1}Q_{1} + k^{3}\tilde{R}_{1}(k)Q_{1})$$

$$\times (k^{-1}Q_{2}D_{2}Q_{2} + Q_{2}BQ_{2} - Q_{2}D_{2}Q_{2}M_{2}Q_{1}D_{1} - D_{1}Q_{1}M_{2}Q_{2}D_{2}Q_{2} + k\tilde{R}_{6}(k))$$

$$\times (Q_{1} - kQ_{1}M_{1}D_{0} + k^{2}Q_{1}M_{2}D_{0} - k^{2}Q_{1}M_{1}D_{0}M_{1}D_{0} + k^{3}Q_{1}\tilde{R}_{1}(k))$$

$$= k^{-2}Q_{2}D_{2}Q_{2} + k^{-1}(D_{2}M_{2}D_{1}M_{2}D_{2} + D_{2}M_{3}D_{2} - Q_{1}M_{2}D_{2} - D_{2}M_{2}Q_{1}) + \tilde{R}_{7}(k).$$

We thus find

$$k^{-1}(M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}$$

$$= k^{-1}D_1 + k^{-2}Q_2D_2Q_2 + k^{-1}(D_2M_2D_1M_2D_2 + D_2M_3D_2 - Q_1M_2D_2 - D_2M_2Q_1) + \tilde{R}_7(k)$$

Using Equation (3.37) and the fact that  $(M(k) + Q_1)^{-1}$  is uniformly bounded in k completes the proof.

We note that in the case  $Q_2 = 0$  this reduces to the result of Theorem 3.2.49.

**Definition 3.2.58.** If  $Q_1 \neq 0$  (an exceptional point of the first or third kind) we say there exists a zero energy resonance for H. In this case Lemmas 3.2.52 and 3.2.53 show that the set  $Q_1\mathcal{H} \ominus Q_2\mathcal{H}$  is one dimensional and for each  $\varphi \in Q_1\mathcal{H}/Q_2\mathcal{H}$  there exists a  $t > \frac{1}{2}$  and  $\psi \in H^{0,-t}$  such that  $\varphi = Uv\psi$ . We define a normalised zero energy resonance

by the condition

$$||V\psi||_1 = ||v\varphi||_1 = (4\pi)^{\frac{1}{2}}.$$

We also use the notation  $T_1 = Q_1 - Q_2$  for the projection onto the span of  $\varphi$ .

We now justify the choice of normalisation for our resonance  $\psi$ .

**Lemma 3.2.59.** Suppose that n=3 and V satisfies Assumption 3.2.41 and that there exists a zero energy resonance for H. Let  $\psi$  be the normalised zero energy resonance of Definition 3.2.58. Then for any  $t_1, t_2 > \frac{3}{2}$  we have in  $\mathcal{B}(H^{-1,t_1}, H^{1,-t_2})$  the equality

$$G_0vT_1D_1T_1vG_0 = \langle \psi, \cdot \rangle \psi,$$

where  $\psi$  is the normalised zero energy resonance of Definition 3.2.58.

Proof. Fix  $t_1, t_2 > \frac{3}{2}$ . Since  $T_1\mathcal{H}$  is one dimensional, we choose a non-zero  $\varphi \in T_1\mathcal{H}$  (a basis). Since  $T_1$  is a rank one operator, we find there exists  $\mu \in \mathbb{C}$  such that  $T_1D_1T_1\varphi = \mu\varphi$ . By Lemma 3.2.52 there exists  $t_3 > \frac{1}{2}$  and  $\tilde{\psi} \in H^{0,-t_3}$  such that  $\varphi = Uv\tilde{\psi}$  and  $\tilde{\psi} = -G_0V\tilde{\psi}$ . By rescaling  $\varphi$  if necessary, we may choose  $\tilde{\psi} = \psi$ . So we compute that for any  $f \in H^{1,-t_1}$  we have

$$T_1 v G_0 f = \langle v G_0 f, \varphi \rangle \varphi = \langle f, G_0 v \varphi \rangle \varphi = \langle f, G_0 V \psi \rangle \varphi = -\langle f, \psi \rangle \varphi.$$

Then we find

$$G_0 v T_1 D_1 T_1 v G_0 f = -\langle f, \psi \rangle G_0 v T_1 D_1 T_1 \varphi = -\mu \langle f, \psi \rangle G_0 v \varphi$$
$$= -\mu \langle f, \psi \rangle G_0 V \psi = \mu \langle f, \psi \rangle \psi.$$

We determine  $\mu$  as  $\mu = \langle \varphi, M_1 \varphi \rangle = V \langle \psi, G_1 V \psi \rangle = (4\pi)^{-1} \langle V \psi, 1 \rangle^2 = 1$ , from which the result follows.

We can also recognise the projection onto the zero eigenspace for H in terms of the operators we have seen. The following characterisation is implicit in [88] and can be found as [64, Lemma 10].

**Lemma 3.2.60.** [64, Lemma 10] Suppose that n = 3 and V satisfies Assumption 3.2.41 and let  $P_0$  be the kernel projection of H. Then we have the equality  $P_0 = G_0 v Q_2 D_2 Q_2 v G_0$ .

*Proof.* Let  $(\varphi_j)_{j=1}^{N_0}$  be an orthonormal basis for  $Q_2\mathcal{H}$ . Then by Lemmas 3.2.52 and 3.2.53 we have

$$0 = \varphi_j + UvG_0v\varphi_j$$

for  $1 \leq j \leq N_0$  and we can write  $\varphi_j = Uv\psi_j$  for some  $\psi_j \in \mathcal{H}$ . Furthermore, we have

$$0 = \int_{\mathbb{R}^3} V(x)\psi_j(x) \, \mathrm{d}x = \int_{\mathbb{R}^3} v(x)\varphi_j(x) \, \mathrm{d}x,$$

the  $\psi_j$  are linearly independent and satisfy the relation  $0 = \psi_j + G_0 V \psi_j$ . We thus compute for  $f \in \mathcal{H}$  that

$$Q_2 v G_0 f = \sum_{j=1}^{N_0} \langle v G_0 f, \varphi_j \rangle \varphi_j = \sum_{j=1}^{N_0} \langle f, G_0 v \varphi_j \rangle \varphi_j = -\sum_{j=1}^{N_0} \langle f, \psi_j \rangle \varphi_j.$$

Let  $(A_{ij})$  be the matrix representation of  $Q_2vG_2vQ_2$  (the inverse of  $D_2$ ) relative to the basis  $(\varphi_j)_{j=1}^{N_0}$ . Since we have  $\int_{\mathbb{R}^3} v(x)\varphi_j(x) dx = 0$  we find

$$A_{ij} = \langle \varphi_i, Q_2 v G_2 v Q_2 \varphi_j \rangle = \langle G_0 v \varphi_i, G_0 v \varphi_j \rangle = \langle G_0 V \psi_i, G_0 V \psi_j \rangle = \langle \psi_i, \psi_j \rangle.$$

Then we find for  $f \in \mathcal{H}$  that

$$G_0 v Q_2 D_2 Q_2 v G_0 f = -\sum_{j=1}^{N_0} G_0 v Q_2 D_2 \varphi_j \langle f, \psi_j \rangle = -\sum_{i,j=1}^{N_0} G_0 v Q_2 \varphi_i (A^{-1})_{ij} \langle f, \psi_j \rangle$$
$$= \sum_{i,j=1}^{N_0} (A^{-1})_{ij} \psi_i \langle f, \varphi_j \rangle.$$

In particular for  $f = \psi_m$  we find  $G_0 v Q_2 D_2 Q_2 v G_0 \psi_m = \psi_m$ . Thus the range of the operator  $G_0 v Q_2 D_2 Q_2 v G_0$  is  $\operatorname{span}(\psi_j)_{j=1}^{N_0}$  and  $G_0 v Q_2 D_2 Q_2 v G_0 = \operatorname{Id}$  on  $\operatorname{Range}(G_0 v Q_2 D_2 Q_2 v G_0)$ . Since  $G_0 v Q_2 D_2 Q_2 v G_0$  is self-adjoint, we find  $P_0$  is the projection onto the zero eigenspace  $\operatorname{span}(\psi_j)_{j=1}^{N_0}$  as claimed.

We thus obtain the following expansion for the difference of resolvents.

**Theorem 3.2.61.** Suppose that n = 3 and V satisfies Assumption 3.2.41. Then for any  $s, t > \frac{3}{2}$  and sufficiently small k we have in  $\mathcal{B}(H^{-1,t_1}, H^{1,-t_2})$  the expansion

$$R(-k^2) - R_0(-k^2) = k^{-1}(-\langle \cdot, \psi \rangle \psi + D_{-\frac{1}{2}}) - k^{-2}P_0 + \tilde{R}(k),$$

where  $\tilde{R}(k)$  is uniformly bounded in k and

$$D_{-\frac{1}{2}} = -G_0 v (D_2 M_2 D_1 M_2 D_2 + D_2 M_3 D_2 - Q_1 M_2 D_2 - D_2 M_2 Q_1) v G_0.$$

*Proof.* Use the relation

$$R(-k^2) - R_0(-k^2) = -R_0(-k^2)vM(k)^{-1}vR_0(-k^2)$$

in conjunction with Theorem 3.2.57 and Lemmas 3.2.59 and 3.2.60.

## **3.2.5** Dimension n = 2

In this section we determine a low energy expansion of the resolvent in two dimensions using the symmetrised technique introduced in [89]. Such an expansion has been determined explicitly in [89, Theorem 6.2], however we present some additional details for clarity. Similar expansions have been introduced in different contexts in [26], [63] and [166].

We make the following assumption on the potential.

**Assumption 3.2.62.** Suppose that n = 2. We assume the potential V satisfies Assumption 2.2.14 for some  $\rho > 11$ .

In dimension n = 2 Equation (3.13) reads

$$R_0(x,y,z) = \ln(k|x-y|) \sum_{p=0}^{\infty} c_{2,p} k^{2p} |x-y|^{2p} + \sum_{p=0}^{\infty} d_{2,p} k^{2p} |x-y|^{2p}$$
(3.38)

Equation (3.38) leads to the following operator definitions.

**Definition 3.2.63.** Suppose that n=2 and V satisfies Assumption 3.2.62. Then we define the integral kernels

$$G_{2p,0}(x,y) = (c_{2,p} \ln(|x-y|) + d_{2,p}) |x-y|^{2p},$$
 and  $G_{2p,-1}(x,y) = c_{2,p} |x-y|^{2p}.$ 

Then for  $(2p, j) \neq (0, 0)$  we define the operators  $M_{0,0} = U + vG_{0,0}v$  and  $M_{2p,j} = vG_{2p,j}v$ .

The operator  $G_{2p,j}$  is the operator coefficient of the term  $k^{2p} \ln(k)^j$  in Equation (3.38). As in higher even dimensions, we make the definition  $\eta = \ln(k)^{-1}$ . Then we can use the definition of M(k) to write

$$M(k) = U + vR_0(-k^2)v = \sum_{p=0}^{\infty} (M_{2p,0} + \eta^{-1}M_{2p,-1})k^{2p}.$$

We immediately obtain the following result.

**Lemma 3.2.64.** [89, Lemma 6.1] Suppose that n=2 and V satisfies Assumption 3.2.62 and let  $F = \{k \in \mathbb{C} : \text{Re}(k) \geq 0, |k| \leq 1\}$ . Then  $M(k) - \eta^{-1}M_{0,-1} - U$  is a uniformly bounded compact operator-valued function on F and for sufficiently small k we have

$$M(k) = M_{0,0} + \eta^{-1} M_{0,-1} + k^2 M_{2,0} + k^2 \eta^{-1} M_{2,-1} + k^4 \eta^{-1} \tilde{R}_0(k),$$

where  $\tilde{R}_0(k)$  is uniformly bounded. For j=0,1 the operators  $M_{2j,0}$  are compact and self-adjoint and the operators  $M_{2j,-1}$  are finite rank.

In particular, we note that  $M_{0,-1} = c_{2,0} ||v||_2^2 P$ , where P is the rank one projection  $P = ||v||_2^{-2} \langle v, \cdot \rangle v$ . We denote  $Q = \operatorname{Id} - P$  the complementary projection.

We now aim to use the Feshbach formula of Lemma 3.1.1 to invert M(k). To do so we need the following definition.

**Definition 3.2.65.** Suppose that n = 2 and V satisfies Assumption 3.2.62. We say that zero is a regular point of the spectrum of H if  $QM_{0,0}Q$  is invertible on  $Q\mathcal{H}$ . If zero is not a regular point of the spectrum of H, we denote by  $Q_1$  the orthogonal projection onto  $\text{Ker}(QM_{0,0}Q)$ . In this case the operator  $QM_{0,0}Q + Q_1$  is invertible and we denote the inverse by  $D_0$ .

**Lemma 3.2.66.** Suppose that n = 2 and V satisfies Assumption 3.2.62 and that zero is not a regular point for the spectrum of H. Denote by  $M_0(k) = M(k) - \eta^{-1}M_{0,-1}$ . For sufficiently small k, the operator  $QM_0(k)Q + Q_1$  is invertible and there exists a function  $g(k) = \eta^{-1}\tilde{g}(k)$  with  $\tilde{g}$  uniformly bounded such that

$$(M(k) + Q_1)^{-1} = g(k)^{-1}A + QD_0Q - k^2D(k) + k^4\tilde{R}_2(k),$$

with D(k) and  $\tilde{R}_2(k)$  uniformly bounded in k and

$$A := P - PM_{0,0}QD_0Q - QD_0QM_{0,0}P + QD_0QM_{0,0}PM_{0,0}QD_0Q.$$

*Proof.* We decompose  $\mathcal{H} = P\mathcal{H} \oplus Q\mathcal{H}$  so that we may write

$$\begin{split} M(k) + Q_1 &= \begin{pmatrix} P(M(k) + Q_1)P & P(M(k) + Q_1)Q \\ Q(M(k) + Q_1)P & Q(M(k) + Q_1)Q \end{pmatrix} \\ &= \begin{pmatrix} P(M(k) - \eta^{-1}M_{0,-1} + \eta^{-1}M_{0,-1})P & P(M(k) - \eta^{-1}M_{0,-1} + \eta^{-1}M_{0,-1})Q \\ Q(M(k) - \eta^{-1}M_{0,-1} + \eta^{-1}M_{0,-1})P & Q(M(k) - \eta^{-1}M_{0,-1} + \eta^{-1}M_{0,-1})Q + Q_1 \end{pmatrix}, \end{split}$$

where we have used that QP = PQ = 0 and  $QQ_1 = Q_1Q = Q_1$ . Since we have defined  $M_0(k) = M(k) - \eta^{-1}M_{0,-1}$  we can use the observation  $PM_{0,-1} = M_{0,-1}P = c_{2,0}||v||_2^2 P$  to obtain

$$M(k) + Q_{1} = \begin{pmatrix} PM_{0}(k)P + \eta^{-1}c_{2,0} ||v||_{2}^{2} P & PM_{0}(k)Q \\ QM_{0}(k)P & QM_{0}(k)Q + Q_{1} \end{pmatrix}$$

$$= \begin{pmatrix} PM_{0,0}P + \eta^{-1}c_{2,0} ||v||_{2}^{2} P & PM_{0,0}Q \\ QM_{0,0}P & QM_{0,0}Q + Q_{1} \end{pmatrix}$$

$$+ k^{2}(M_{2,0} + \eta^{-1}M_{2,-1}) + k^{4}\eta^{-1}\tilde{R}_{0}(k),$$

where  $\tilde{R}_0(k)$  is uniformly bounded. Define the operator

$$A(k) = \begin{pmatrix} PM_{0,0}P + \eta^{-1}c_{2,0} ||v||_2^2 P & PM_{0,0}Q \\ QM_{0,0}P & QM_{0,0}Q + Q_1 \end{pmatrix}.$$
(3.39)

Then by Lemma 3.1.1 the operator A(k) is invertible if and only if the operator

$$B(k) = (PM_{0,0}P + \eta^{-1}c_{2,0}||v||_2^2 P - PM_{0,0}QD_0QM_{0,0}P)^{-1}$$

exists and is bounded, where  $D_0 = (QM_{0,0}Q + Q_1)^{-1}$  as in Definition 3.2.65. Since P is a rank one projection there exists a constant c such that

$$B(k) = (c + \eta^{-1}c_{2,0}||v||_2^2)^{-1}P =: g(k)^{-1}P.$$

Thus we use Lemma 3.1.3 to write

$$A(k)^{-1} = \begin{pmatrix} B(k) & -B(k)PM_{0,0}QD_0 \\ -D_0QM_{0,0}PB(k) & D_0QM_{0,0}PB(k)PM_{0,0}QD_0 + D_0 \end{pmatrix}$$

$$= g(k)^{-1} \begin{pmatrix} P & -PM_{0,0}QD_0 \\ -D_0QM_{0,0}P & D_0QM_{0,0}PM_{0,0}QD_0 + g(k)D_0 \end{pmatrix}$$

$$= g(k)^{-1} (P - PM_{0,0}QD_0Q - QD_0QM_{0,0}P + QD_0QM_{0,0}PM_{0,0}QD_0Q) + QD_0Q$$

$$= g(k)^{-1}A + QD_0Q.$$

Note that for sufficiently small k we have the estimate

$$\left| \left| k^2 (M_{2,0} + \eta^{-1} M_{2,-1}) A(k)^{-1} + k^4 \eta^{-1} \tilde{R}_0(k) A(k)^{-1} \right| \right| < 1.$$

Then we can compute the inverse of  $M(k) + Q_1$  using a Neumann expansion as

$$(M(k) + Q_{1})^{-1} = (A(k) + k^{2}(M_{2,0} + \eta^{-1}M_{2,-1}) + k^{4}\eta^{-1}\tilde{R}_{0}(k))^{-1}$$

$$= A(k)^{-1}(\operatorname{Id} + k^{2}(M_{2,0} + \eta^{-1}M_{2,-1})A(k)^{-1} + k^{4}\eta^{-1}\tilde{R}_{0}(k)A(k)^{-1})^{-1}$$

$$= A(k)^{-1}(\operatorname{Id} - k^{2}(M_{2,0} + \eta^{-1}M_{2,-1})A(k)^{-1} - k^{4}\eta^{-1}\tilde{R}_{0}(k)A(k)^{-1}$$

$$+ k^{4}\tilde{R}_{1}(k))$$

$$= g(k)^{-1}(P - PM_{0,0}QD_{0}Q - QD_{0}QM_{0,0}P$$

$$+ QD_{0}QM_{0,0}PM_{0,0}QD_{0}Q) + QD_{0}Q$$

$$- A(k)^{-1}(k^{2}(M_{2,0} + \eta^{-1}M_{2,-1})A(k)^{-1} - k^{4}\eta^{-1}\tilde{R}_{0}(k)A(k)^{-1} + k^{4}\tilde{R}_{1}(k))$$

$$= g(k)^{-1}A + QD_{0}Q - k^{2}A(k)^{-1}(M_{2,0} + \eta^{-1}M_{2,-1})A(k)^{-1} + k^{4}\tilde{R}_{2}(k)$$

$$= g(k)^{-1}A + QD_{0}Q - k^{2}D(k) + k^{4}\tilde{R}_{2}(k).$$

Here we have defined the term  $D(k) = A(k)^{-1}(M_{2,0} + \eta^{-1}M_{2,-1})A(k)^{-1}$ .

Remark 3.2.67. We can determine more terms in  $\tilde{R}_2(k)$  as

$$\tilde{R}_2(k) = (A(k)^{-1}(M_{2,0} + \eta^{-1}M_{2,-1}))^2 A(k)^{-1} - A(k)^{-1}(M_{4,0} + \eta^{-1}M_{4,-1})A(k)^{-1} + k^2 \tilde{R}_3(k),$$

where  $\tilde{R}_3(k)$  is uniformly bounded in k

We next need to determine the inverse of the operator

$$B_1(k) = Q_1 - Q_1(M(k) + Q_1)^{-1}Q_1.$$

**Definition 3.2.68.** Let n=2 and suppose that V satisfies Assumption 3.2.62 and suppose that zero is not a regular point for H. We say that zero is an exceptional point of the first kind if the operator  $Q_1M_{0,0}PM_{0,0}Q_1$  is invertible on  $Q_1\mathcal{H}$ . In this case we write  $D_1 = (Q_1M_{0,0}PM_{0,0}Q_1)^{-1}$ .

**Lemma 3.2.69.** Suppose that n = 2 and V satisfies Assumption 3.2.62 and that zero is an exceptional point of the first kind for H. Then for sufficiently small k we have

$$B_1(k)^{-1} = -g(k)D_1 + k^2 \tilde{R}_5(k)$$

where  $\tilde{R}_5(k)$  is uniformly bounded.

*Proof.* We use only the lowest order terms in Lemma 3.2.66, denoting the remainder  $\tilde{R}_3(k) = -D(k) + k^2 \tilde{R}_2(k)$ . We use the relations  $Q_1 Q = Q_1 D_0 = Q Q_1 = D_0 Q_1 = Q_1$  and  $Q_1 P = P Q_1 = 0$  to write

$$B_1(k) = Q_1 - g(k)^{-1}Q_1AQ_1 - Q_1 - k^2Q_1\tilde{R}_3(k)Q_1$$
  
=  $-g(k)^{-1}Q_1M_{0,0}PM_{0,0}Q_1 - k^2Q_1\tilde{R}_3(k)Q_1$ 

By assumption zero is an exceptional point of the first kind and so the operator  $QM_{0,0}PM_{0,0}Q$  is invertible with inverse  $D_1$  and using a Neumann expansion we find

$$B_{1}(k)^{-1} = (-g(k)^{-1}QM_{0,0}PM_{0,0}Q - k^{2}Q_{1}\tilde{R}_{3}(k)Q_{1})^{-1}$$

$$= -g(k)D_{1}(\operatorname{Id} + k^{2}g(k)Q_{1}\tilde{R}_{3}(k)Q_{1}D_{1})^{-1}$$

$$= -g(k)D_{1}(\operatorname{Id} + k^{2}g(k)Q_{1}\tilde{R}_{3}(k)Q_{1}D_{1} + k^{4}\tilde{R}_{4}(k)) = -g(k)D_{1} + k^{2}\tilde{R}_{5}(k)). \square$$

Remark 3.2.70. We note that if more terms are required in the expansion, we may write

$$\tilde{R}_5(k) = g(k)^2 D_1 Q_1 A(k)^{-1} (M_{2,0} + \eta^{-1} M_{2,-1}) A(k)^{-1} Q_1 D_1 - k^2 g(k)^{-1} D_1 Q_1 R_2(k) Q_1 D_1 - g(k)^3 k^2 D_1 Q_1 A(k)^{-1} ((M_{2,0} + \eta^{-1} M_{2,-1}) A(k)^{-1} Q_1 D_1)^2 + k^4 \eta^{-1} \tilde{R}_6(k)$$

where  $\tilde{R}_6(k)$  is uniformly bounded and we have defined the term

$$R_2(k) = A(k)^{-1}((M_{2,0} + \eta^{-1}M_{2,-1})A(k)^{-1})^2 0A(k)^{-1}(M_{4,0} + \eta^{-1}M_{4,-1}).$$

Lemma 3.1.1 shows that in the case of an exceptional point of the first kind we can compute  $M(k)^{-1}$  via the formula

$$M(k)^{-1} = (M(k) + Q_1)^{-1} + (M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}.$$
 (3.40)

**Theorem 3.2.71.** Suppose that n = 2 and V satisfies Assumption 3.2.62 and that zero is an exceptional point of the first kind for H. Then for sufficiently small k we have

$$M(k)^{-1} = g(k)^{-1}A + QD_0Q - g(k)^{-1}AQ_1D_1Q_1A - AQ_1D_1Q_1 - Q_1D_1Q_1A - g(k)Q_1D_1Q_1 + \tilde{R}_8(k)$$

where  $\tilde{R}_7(k)$  is uniformly bounded.

*Proof.* By Lemma 3.1.1 we have that M(k) has inverse

$$M(k)^{-1} = (M(k) + Q_1)^{-1} + (M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}.$$

Combining Lemmas 3.2.66 and 3.2.69 we can compute the expression

$$(M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1} = -(g(k)^{-1}A + QD_0Q + k^2\tilde{R}_3(k))Q_1(g(k)D_1 + k^2\tilde{R}_5(k)) \times Q_1(g(k)^{-1}A + QD_0Q + k^2\tilde{R}_3(k))$$
  
=  $-g(k)^{-1}AQ_1D_1Q_1A - AQ_1D_1Q_1 - Q_1D_1Q_1A - g(k)Q_1D_1Q_1 + \tilde{R}_7(k),$ 

where we have used the relations  $Q_1QD_0Q = QD_0QQ_1 = Q_1$ . Thus we find

$$M(k)^{-1} = (M(k) + Q_1)^{-1} + (M(k) + Q_1)^{-1}Q_1B_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}$$
  
=  $g(k)^{-1}A + QD_0Q - g(k)^{-1}AQ_1D_1Q_1A - AQ_1D_1Q_1 - Q_1D_1Q_1A$   
-  $g(k)Q_1D_1Q_1 + \tilde{R}_8(k)$ 

where  $\tilde{R}_8(k)$  is uniformly bounded.

If the leading term of  $B_1(k)$  is not invertible we need to apply Lemma 3.1.1 again.

**Definition 3.2.72.** Suppose that n = 2 and V satisfies Assumption 3.2.62 and that the operator  $Q_1M_{0,0}PM_{0,0}Q_1$  is not invertible. Let  $Q_2$  be the orthogonal projection onto the kernel of  $Q_1M_{0,0}PM_{0,0}Q_1$ . Then the operator  $Q_1M_{0,0}PM_{0,0}Q_1 + Q_2$  is invertible with inverse  $D_1$ . We say that zero is an exceptional point of the second kind if the operator  $Q_2M_{2,-1}Q_2$  is invertible and we denote its inverse by  $D_2$ .

By repeating the arguments of Lemma 3.2.69 and keeping track of more terms, we can determine an expansion for  $B_1(k)^{-1}$  in the case of an exceptional point of the second kind. First we define the operator  $\tilde{B}_1(k) = -g(k)B_1(k)$ .

**Lemma 3.2.73.** Suppose that n = 2 and V satisfies Assumption 3.2.62 and that zero is an exceptional point of the second kind for H. Then for sufficiently small k we have

$$(\tilde{B}_1(k) + Q_2)^{-1} = D_1 + k^2 \tilde{R}_{11}(k),$$

where  $\tilde{R}_{11}(k)$  is uniformly bounded.

*Proof.* We compute the Neumann expansion

$$\begin{split} (\tilde{B}_{1}(k) + Q_{2})^{-1} &= (-g(k)B_{1}(k) + Q_{2})^{-1} = (Q_{1}M_{0,0}PM_{0,0}Q_{1} + Q_{2} - k^{2}g(k)Q_{1}\tilde{R}_{3}(k)Q_{1})^{-1} \\ &= D_{1}(\operatorname{Id} - k^{2}g(k)Q_{1}\tilde{R}_{3}(k)Q_{1}D_{1})^{-1} \\ &= D_{1} - k^{2}g(k)D_{1}Q_{1}\tilde{R}_{3}(k)Q_{1}D_{1} + k^{4}\tilde{R}_{10}(k) \\ &= D_{1} + k^{2}\tilde{R}_{11}(k), \end{split}$$

as claimed.  $\Box$ 

Remark 3.2.74. As before we can compute more terms in this expansion, finding

$$\begin{split} \tilde{R}_{11}(k) &= -g(k)D_1Q_1\tilde{R}_3(k)Q_1D_1 + k^2\tilde{R}_{10}(k) \\ &= -g(k)D_1Q_1(-D(k) + k^2\tilde{R}_2(k)Q_1D_1 + k^2\tilde{R}_{10}(k) \\ &= g(k)D_1Q_1D(k)Q_1D_1 + k^2g(k)Q_1D_1\tilde{R}_2(k)D_1Q_1 + k^2\tilde{R}_{10}(k) \\ &= -g(k)D_1Q_1D(k)Q_1D_1 + k^2\tilde{R}_{12}(k), \end{split}$$

where  $\tilde{R}_{12}(k)$  is uniformly bounded.

To compute inverses, we need to use further terms in the expansion of  $(M(k) + Q_1)^{-1}$  and so we first prove the following to simplify some later expressions.

**Lemma 3.2.75.** Suppose that n = 2 and V satisfies Assumption 3.2.62 and that zero is not a regular point or an exceptional point of the second kind for H. Then

$$A(k)^{-1}Q_2 = Q_2 A(k)^{-1} = Q_2,$$

where A(k) is defined in Equation (3.39). Furthermore we have the relations

$$0 = M_{0.0}Q_2 = Q_2M_{0.0}$$
.

*Proof.* Noting that  $Q_2$  is the projection onto the kernel of  $Q_1M_{0,0}Q_1$  we find that for any

 $f \in \mathcal{H}$  we have the relation

$$\langle PM_{0,0}Q_2f, PM_{0,0}Q_2f \rangle = \langle Q_2M_{0,0}PM_{0,0}Q_2f, f \rangle = \langle Q_2(Q_1M_{0,0}PM_{0,0}Q_1)Q_2f, f \rangle$$
  
=  $\langle Q_2M_{0,0}PM_{0,0}Q_2f, f \rangle = 0$ 

and thus  $PM_{0,0}Q_2 = 0$ . We then compute using  $Q_2D_0 = D_0Q_2 = Q_2$ ,  $Q_2P = 0 = PQ_2$  and the fact that  $Q_2 \leq Q$  to find

$$Q_2 A(k)^{-1} = g(k)^{-1} Q_2 (P - PM_{0,0} D_0 Q - QD_0 QM_{0,0} P + QD_0 QM_{0,0} PQD_0 Q) + Q_2 D_0 Q$$
  
=  $g(k)^{-1} (Q_2 M_{0,0} P + Q_2 M_{0,0} PQD_0 Q) + Q_2$   
=  $Q_2$ ,

where we have used the relation  $Q_2M_{0,0}P = 0$ . To prove the final relation, we note that by definition we have  $QM_{0,0}QQ_1 = 0$  and thus  $PM_{0,0}QQ_1 = M_{0,0}Q_1$ . Thus we may compute

$$Q_1 M_{0.0} P M_{0.0} Q_1 = (Q_1 M_{0.0} P)(P M_{0.0} Q_1) = (P M_{0.0} Q_1)^* (P M_{0.0} Q_1) = (M_{0.0} Q_1)^* (M_{0.0} Q_1).$$

Thus we see that  $Ker(Q_1M_{0,0}PM_{0,0}Q_1) = Ker(M_{0,0}Q_1)$  and so the definition of  $Q_2$  gives  $M_{0,0}Q_1Q_2 = M_{0,0}Q_2 = 0$ .

Keeping track of the many terms appearing in the following resolvent expansions can be rather cumbersome, so the relations in Lemma 3.2.75 will allow us to simplify some terms without too much computational effort.

**Lemma 3.2.76.** Suppose that n = 2 and V satisfies Assumption 3.2.62 and that zero is an exceptional point of the second kind for H. Then we have the expansion

$$\tilde{B}_1(k)^{-1} = D_1 + k^{-2}g(k)^{-1}Q_2D_2(k)Q_2 - (D_1Q_1D(k)Q_2D_2(k)Q_2 + Q_2D_2(k)Q_2D(k)Q_1D_1) + k^2\tilde{R}_{15}(k),$$

where  $\tilde{R}_{15}(k)$  is uniformly bounded.

*Proof.* We know by Lemma 3.1.1 that the operator  $\tilde{B}_1(k)$  is invertible if and only if the operator

$$B_2(k) = Q_2 - Q_2(\tilde{B}_1(k) + Q_2)^{-1}Q_2$$

is invertible on  $Q_2\mathcal{H}$ . Noting the identity  $Q_2D_1=D_1Q_2=Q_2$  we find

$$B_2(k) = Q_2 - Q_2 D_1 Q_2 + k^2 g(k) Q_2 D(k) Q_2 - k^4 Q_2 \tilde{R}_{12}(k) Q_2$$
  
=  $k^2 g(k) Q_2 D(k) Q_2 - k^4 Q_2 \tilde{R}_{12}(k) Q_2$ .

We also recall that  $D(k) = A(k)^{-1}(M_{2,0} + \eta^{-1}M_{2,-1})A(k)^{-1}$ , so that an application of Lemma 3.2.75 gives

$$Q_2D(k)Q_2 = Q_2(M_{2,0} + \eta^{-1}M_{2,-1})Q_2.$$

Since zero is an exceptional point of the second kind for H, the operator  $Q_2M_{2,-1}Q_2$  is invertible with inverse  $D_2$  and thus for sufficiently small k the operator  $Q_2D(k)Q_2$  is invertible with inverse  $D_2(k)$  and we can then form the Neumann expansion

$$\begin{split} B_2(k)^{-1} &= (g(k)k^2Q_2M_{2,0}Q_2 + k^2\eta^{-1}Q_2M_{2,-1}Q_2 - k^4Q_2\tilde{R}_{12}(k)Q_2)^{-1} \\ &= k^{-2}g(k)^{-1}D_2(k)(\operatorname{Id} - k^2g(k)^{-1}Q_2\tilde{R}_{10}(k)Q_2D_2(k))^{-1} \\ &= k^{-2}g(k)^{-1}D_2(k)(\operatorname{Id} + k^2g(k)^{-1}Q_2\tilde{R}_{10}(k)Q_2D_2(k) + k^4\tilde{R}_{13}(k)) \\ &= k^{-2}g(k)^{-1}D_2(k) + \tilde{R}_{14}(k), \end{split}$$

where  $\tilde{R}_{14}(k)$  is uniformly bounded. We thus find

$$\begin{split} \tilde{B}_{1}(k)^{-1} &= (\tilde{B}_{1}(k) + Q_{2})^{-1} + (\tilde{B}_{1}(k) + Q_{2})^{-1}Q_{2}\tilde{B}_{2}(k)^{-1}Q_{2}(\tilde{B}_{1}(k) + Q_{2})^{-1} \\ &= D_{1} - k^{2}g(k)D_{1}Q_{1}D(k)Q_{1}D_{1} + k^{4}\tilde{R}_{12}(k) + (D_{1} - k^{2}g(k)D_{1}Q_{1}D(k)Q_{1}D_{1} \\ &+ k^{4}\tilde{R}_{12}(k))Q_{2}(k^{-2}g(k)^{-1}D_{2}(k) + \tilde{R}_{14}(k))Q_{2} \\ &\times (D_{1} - k^{2}g(k)D_{1}Q_{1}D(k)Q_{1}D_{1} + k^{4}\tilde{R}_{12}(k)) \\ &= D_{1} + k^{-2}g(k)^{-1}Q_{2}D_{2}(k)Q_{2} - (D_{1}Q_{1}D(k)Q_{2}D_{2}(k)Q_{2} + Q_{2}D_{2}(k)Q_{2}D(k)Q_{1}D_{1}) \\ &+ k^{2}\tilde{R}_{15}(k), \end{split}$$

where  $\tilde{R}_{15}(k)$  is uniformly bounded.

We are now ready to compute  $M(k)^{-1}$  in the case of an exceptional point of the second kind for H. Two applications of Lemma 3.1.1 show that we can use the formula

$$M(k)^{-1} = (M(k) + Q_1)^{-1} - g(k)^{-1}(M(k) + Q_1)^{-1}Q_1\tilde{B}_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}.$$
 (3.41)

**Theorem 3.2.77.** Suppose that n = 2 and V satisfies Assumption 3.2.62 and that zero is an exceptional point of the second kind for H. Then for sufficiently small k we have

$$M(k)^{-1} = k^{-2}g(k)^{-1}Q_2D_2(k)Q_2 + Q_2 + g(k)^{-1}A - g(k)^{-1}D(k)Q_2D_2(k)Q_2 + k^2\tilde{R}_{18}(k) - g(k)^{-1}Q_2D_2(k)Q_2D(k) - (Q_2D(k)Q_2D_2(k)Q_2 + Q_2D_2(k)Q_2D(k)Q_2),$$

where  $\tilde{R}_{18}(k)$  is uniformly bounded.

*Proof.* We begin by noting that

$$M(k)^{-1} = (M(k) + Q_1)^{-1} - g(k)^{-1}(M(k) + Q_1)^{-1}Q_1\tilde{B}_1(k)^{-1}Q_1(M(k) + Q_1)^{-1}.$$

By Lemma 3.2.75 we have  $Q_2A = AQ_2 = 0$ . So we make the computations

$$Q_2(M(k) + Q_1)^{-1} = Q_2(g(k)^{-1}A + QD_0Q - k^2D(k) + k^4\tilde{R}_2(k))$$
$$= Q_2 - k^2Q_2D(k) + k^4Q_2\tilde{R}_2(k)$$

and

$$(M(k) + Q_1)^{-1}Q_2 = (g(k)^{-1}A + QD_0Q - k^2D(k) + k^4\tilde{R}_2(k))Q_2$$
$$= Q_2 - k^2D(k)Q_2 + k^4\tilde{R}_2(k)Q_2.$$

So noting that  $\tilde{B}_1(k)^{-1} = Q_2\tilde{B}_1(k)^{-1}Q_2$ , we can expand out the product in the next expression to obtain

$$\begin{split} &(M(k)+Q_1)^{-1}Q_1\tilde{B}_1(k)^{-1}Q_1(M(k)+Q_1)^{-1} = (Q_2-k^2D(k)Q_2+k^4\tilde{R}_2(k)Q_2)\\ &\times (D_1+k^{-2}g(k)^{-1}Q_2D_2(k)Q_2 - (D_1Q_1D(k)Q_2D_2(k)Q_2+Q_2D_2(k)Q_2D(k)Q_1D_1)\\ &+k^2\tilde{R}_{15}(k))\\ &=k^{-2}g(k)^{-1}Q_2D_2(k)Q_2+Q_2-g(k)^{-1}D(k)Q_2D_2(k)Q_2-g(k)^{-1}Q_2D_2(k)Q_2D(k)-\\ &-g(k)(Q_2D(k)Q_2D_2(k)Q_2+Q_2D_2(k)Q_2D(k)Q_2)+k^2\tilde{R}_{17}(k), \end{split}$$

where  $\tilde{R}_{17}(k)$  is uniformly bounded. So we find

$$\begin{split} M(k)^{-1} &= (M(k) + Q_1)^{-1} - g(k)^{-1} (M(k) + Q_1)^{-1} Q_1 \tilde{B}_1(k)^{-1} Q_1 (M(k) + Q_1)^{-1} \\ &= g(k)^{-1} A + Q D_0 Q - k^2 D(k) + k^4 \tilde{R}_2(k) + k^{-2} g(k)^{-1} Q_2 D_2(k) Q_2 \\ &\quad + Q_2 - g(k)^{-1} D(k) Q_2 D_2(k) Q_2 - g(k)^{-1} Q_2 D_2(k) Q_2 D(k) \\ &\quad - g(k) (Q_2 D(k) Q_2 D_2(k) Q_2 + Q_2 D_2(k) Q_2 D(k) Q_2) + k^2 \tilde{R}_{17}(k) \\ &= k^{-2} g(k)^{-1} Q_2 D_2(k) Q_2 + Q_2 + g(k)^{-1} A - g(k)^{-1} D(k) Q_2 D_2(k) Q_2 \\ &\quad - g(k)^{-1} Q_2 D_2(k) Q_2 D(k) - (Q_2 D(k) Q_2 D_2(k) Q_2 \\ &\quad + Q_2 D_2(k) Q_2 D(k) Q_2) + k^2 \tilde{R}_{18}(k), \end{split}$$

where  $\tilde{R}_{18}(k)$  is uniformly bounded.

**Definition 3.2.78.** Suppose that n = 2 and V satisfies Assumption 3.2.62 and that the operator  $Q_1M_{0,0}PM_{0,0}Q_1$  is not invertible. We say that zero is an exceptional point of the third kind if the operator  $Q_2M_{2,-1}Q_2$  is not invertible. In this case we let  $Q_3$  be the orthogonal projection onto the kernel of  $Q_2M_{2,-1}Q_2$ . Then the operator  $Q_2M_{2,-1}Q_2 + Q_3$ 

is invertible and we write its inverse as  $D_2$ . We also define the projection  $T_3 = Q_2 - Q_3$ .

We now confirm that the Feshbach expansion procedure of Lemma 3.1.1 terminates.

**Lemma 3.2.79.** [63, Lemma 5.4] Suppose that n = 2 and V satisfies Assumption 3.2.62. Then the operator  $Q_3M_{2.0}Q_3$  is invertible on  $Q_3\mathcal{H}$ .

*Proof.* Suppose that  $f \in Q_3\mathcal{H}$  satisfies  $Q_3M_{2,0}Q_3f = 0$ . Then we have

$$\int_{\mathbb{R}^2} v(y) f(y) \, \mathrm{d}y = 0$$

and  $Q_2M_{2,-1}Q_2f=0$ . Note also that  $vf\in L^1(\mathbb{R}^n)$  and  $x\mapsto |x|vf\in L^1(\mathbb{R}^n)$ , so that  $\mathcal{F}(vf)\in L^1(\mathbb{R}^n)\cap C^1(\mathbb{R}^n)$  Thus we find

$$\langle Q_{2}M_{2,0}Q_{2}f, f \rangle = \langle M_{2,0}f, f \rangle = \lim_{k \to 0} \langle k^{-2}(R_{0}(k^{2}) - G_{0,0})vf, vf \rangle$$

$$= \frac{1}{4\pi^{2}} \lim_{k \to 0} k^{-2} \int_{\mathbb{R}^{2}} \left( \frac{1}{|\xi|^{2} + k^{2}} - \frac{1}{|\xi|^{2}} \right) |[\mathcal{F}(vf)](\xi)|^{2} d\xi$$

$$= \frac{1}{4\pi^{2}} \lim_{k \to 0} \int_{\mathbb{R}^{2}} \frac{1}{|\xi|^{2} (|\xi|^{2} + k^{2})} |[\mathcal{F}(vf)](\xi)|^{2} d\xi$$

$$= \frac{1}{4\pi^{2}} \int_{\mathbb{R}^{2}} \frac{|[\mathcal{F}(vf)](\xi)|^{2}}{|\xi|^{4}} d\xi,$$

where we have used Lemma 3.1.6 to bring the limit inside the integral. Our assumptions on v and f guarantee that  $(x \mapsto |x|^{-2}v(x)f(x)) \in L^1(\mathbb{R}^2)$  and thus  $\mathcal{F}(vf)$  satisfies the extra decay hypothesis of Lemma 3.1.6. Since  $Q_2M_{2,-1}Q_2f = 0$ , our assumptions on v and f guarantee that  $vf \in L^1(\mathbb{R}^2)$  and so we find vf = 0. Since  $f \in Q_1\mathcal{H}$  gives  $f = Uv\psi$  for some  $\psi \in \mathcal{H}$  and thus f = 0.

We shall write  $D_3$  for the inverse of  $Q_3M_{2,0}Q_3$ .

Remark 3.2.80. We note that the above proof also implies the useful relation

$$\langle G_{0,0}vf, G_{0,0}vf \rangle = \langle M_{2,0}f, f \rangle. \tag{3.42}$$

We can further characterise the obstruction subspaces in terms of resonances and zero energy eigenvalues.

**Lemma 3.2.81.** [89, Lemma 6.4] Suppose that n=2 and V satisfies Assumption 3.2.62 and  $Q_1 \neq 0$  and fix  $\varphi \in Q_1\mathcal{H}$  with  $||\varphi|| = 1$ . Then there exists  $\psi \in L^{\infty}(\mathbb{R}^2)$  such that  $\varphi = Uv\psi$  and  $H\psi = 0$  in the sense of distributions. Furthermore, there exists  $\tilde{\psi} \in \mathcal{H}$  and  $c_0, c_1, c_2 \in \mathbb{R}$  such that

$$\psi(x) = c_0 + \sum_{j=1}^{2} c_j \frac{x_j}{(1+|x|^2)} + \tilde{\psi}(x). \tag{3.43}$$

**Lemma 3.2.82.** [89, Lemma 6.4] Suppose that n = 2 and V satisfies Assumption 3.2.62. Suppose  $\psi = c + \psi_1 + \psi_2$  with  $c \in \mathbb{C}$ ,  $\psi_1 \in L^q(\mathbb{R}^2)$  for some  $q \in (2, \infty)$  and  $\psi_2 \in \mathcal{H}$ . If  $H\psi = 0$  in the sense of distributions then  $\varphi = Uv\psi \in Q_1\mathcal{H}$ .

The above characterisation leads naturally to a definition of resonances in two dimensions.

**Definition 3.2.83.** If  $Q_3 \neq 0$  there exists a zero energy bound state, a solution to  $H\varphi = 0$  for which  $\varphi \in \text{Dom}(H)$ . If  $T_3 \neq 0$  we say there exists a p-resonance, which corresponds to the existence of a distributional solution to  $H\varphi = 0$  which satisfies  $\varphi \in L^q(\mathbb{R}^2)$  for some  $q \in (2, \infty)$ . If  $T_2 \neq 0$  we say there exists an s-resonance, which corresponds to the existence of a distributional solution of  $H\varphi = 0$  with  $\varphi \in L^{\infty}(\mathbb{R}^2)$ .

We now repeat the proof of Theorem 3.2.77 in the case of an exceptional point of the third kind. We use the notation

$$D_2(k) = (Q_2 M_{2,-1} Q_2 + \eta Q_2 M_{2,0} Q_2 + Q_3)^{-1} = D_2 + \eta C_2(k),$$

so that  $D_2(0) = D_2$ . Note that the second equality is obtained via a Neumann expansion.

**Lemma 3.2.84.** Suppose that n = 2 and V satisfies Assumption 3.2.62. Then for sufficiently small k the operator  $\tilde{B}_2(k) = k^{-2}g(k)^{-1}B_2(k)$  has inverse

$$\tilde{B}_2(k)^{-1} = \tilde{T}_3(k) - k^2 g(k)^{-1} \tilde{T}_3(k) Q_2(M_{4,0} + \eta^{-1} M_{4,-1}) Q_2 \tilde{T}_3(k) + k^4 \tilde{R}_{13}(k),$$

where  $\tilde{R}_{13}(k)$  is uniformly bounded and

$$\tilde{T}_3(k) := (Q_2 D(k) Q_2)^{-1} = \begin{pmatrix} a & -a T_3 M_{2,0} Q_3 D_3 \\ -D_3 Q_3 M_{2,0} T_3 a & D_3 Q_3 M_{2,0} T_3 a T_3 M_{2,0} Q_3 D_3 + D_3 \end{pmatrix}$$

with 
$$a = (T_3 M_{2.0} T_3 + \eta^{-1} T_3 M_{2.-1} T_3 - T_3 M_{2.0} Q_3 (Q_3 M_{2.0} Q_3)^{-1} Q_3 M_{2.0} T_3)^{-1}$$
.

*Proof.* We begin by defining  $\tilde{B}_2(k) = k^{-2}g(k)^{-1}B_2(k)$  and consider the decomposition  $Q_2\mathcal{H} = T_3\mathcal{H} \oplus Q_3\mathcal{H}$ . Then we can determine the expression

$$Q_2D(k)Q_2 = Q_2(M_{2,0} + \eta^{-1}M_{2,-1})Q_2 = \begin{pmatrix} T_3M_{2,0}T_3 + \eta^{-1}T_3M_{2,-1}T_3 & T_3M_{2,0}Q_3 \\ Q_3M_{2,0}T_3 & Q_3M_{2,0}Q_3 \end{pmatrix}.$$

Lemma 3.1.3 shows that  $Q_2D_2(k)Q_2$  is invertible provided  $Q_3M_{2,0}Q_3$  is (which is guaranteed by Lemma 3.2.79) and the operator

$$a = (T_3 M_{2.0} T_3 + \eta^{-1} T_3 M_{2.-1} T_3 - T_3 M_{2.0} Q_3 (Q_3 M_{2.0} Q_3)^{-1} Q_3 M_{2.0} T_3)^{-1}$$

exists as a bounded operator (note that the explicit k dependence of a comes from the  $\eta$  term). Letting  $D_3 = (Q_3 M_{2,0} Q_3)^{-1}$  we find

$$\tilde{T}_3(k) := (Q_2 D(k) Q_2)^{-1} = \begin{pmatrix} a & -a T_3 M_{2,0} Q_3 D_3 \\ -D_3 Q_3 M_{2,0} T_3 a & D_3 Q_3 M_{2,0} T_3 a T_3 M_{2,0} Q_3 D_3 + D_3 \end{pmatrix}.$$

For sufficiently small k we have the estimate

$$\left| \left| k^2 g(k)^{-1} Q_2 \tilde{R}_{12}(k) Q_2 \tilde{T}_3(k) \right| \right| < 1$$

and thus we can determine the Neumann expansion

$$\tilde{B}_{2}(k)^{-1} = \left(Q_{2}D(k)Q_{2} - k^{2}g(k)^{-1}Q_{2}\tilde{R}_{12}(k)Q_{2}\right)^{-1}$$

$$= \tilde{T}_{3}(k) \left(\operatorname{Id} - k^{2}g(k)^{-1}Q_{2}\tilde{R}_{12}(k)Q_{2}\tilde{T}_{3}(k)\right)^{-1}$$

$$= \tilde{T}_{3}(k) - k^{2}g(k)^{-1}\tilde{T}_{3}(k)Q_{2}\tilde{R}_{12}(k)Q_{2}\tilde{T}_{3}(k) + k^{4}\tilde{R}_{13}(k)$$

where  $\tilde{R}_{13}(k)$  is uniformly bounded. We can determine further terms in  $\tilde{R}_{12}(k)$  as

$$\tilde{R}_{12}(k) = g(k)Q_1D_1\tilde{R}_2(k) + k^2\tilde{R}_{10}(k)$$

$$= g(k)Q_1D_1A(k)^{-1}(M_{4,0} + \eta^{-1}M_{4,-1})A(k)^{-1} + k^2\tilde{R}_{19}(k).$$

Thus we can determine  $Q_2\tilde{R}_{12}(k)Q_2 = g(k)Q_2(M_{4,0} + \eta^{-1}M_{4,-1})Q_2 + k^2Q_2\tilde{R}_{19}(k)$ .

We now apply Lemma 3.1.3 to the operator  $B_2(k)$  to obtain the expansion for  $M(k)^{-1}$ .

**Theorem 3.2.85.** Suppose that n = 2 and V satisfies Assumption 3.2.62. Then for sufficiently small k we have the expansion

$$M(k)^{-1} = -k^{-2}Q_2\tilde{T}_3(k)Q_2 + \eta^{-1}\tilde{R}(k),$$

where  $\tilde{R}(k)$  is uniformly bounded.

*Proof.* The most singular term which we have not considered yet is given by

$$-k^{-2}(M(k)+Q_1)^{-1}Q_1(\tilde{B}_1(k)+Q_2)^{-1}Q_2\tilde{B}_2(k)^{-1}Q_2(\tilde{B}_1(k)+Q_2)^{-1}Q_1(M(k)+Q_1)^{-1}$$

$$=-k^{-2}Q_2\tilde{T}_3(k)Q_2.$$

Remark 3.2.86. Note that we can write explicitly the term

$$\tilde{T}_3(k) = D_3 + (a - aT_3M_{2,0}D_3 - D_3M_{2,0}T_3 + D_3M_{2,0}T_3aT_3M_{2,0}D_3.$$

We now characterise the image of the operators  $T_2$  and  $T_3$  to analyse resonant behaviour.

**Lemma 3.2.87.** [89, Theorem 6.2] Suppose that n = 2 and V satisfies Assumption 3.2.62. The space Range $(T_2)$  is spanned by the vector  $\Theta_0 := Q_1 M_{0,0} v$ . The space Range $(T_3)$  is spanned by the vectors  $\Theta_j := Q_2 X_j v$ , where  $X_j$  denotes the multiplication operator by the coordinate  $x_j$ .

*Proof.* By definition we have

$$Range(T_2) = Q_1 \mathcal{H} \cap Range(Q_1 M_{0,0} P M_{0,0} Q_1),$$

which by Lemma 3.2.81 and the definition of  $M_{0,0}$  is spanned by  $\Theta_0$ . Similarly we have

$$\operatorname{Range}(T_3) = Q_2 \mathcal{H} \cap \operatorname{Range}(Q_2 M_{2,-1} Q_2).$$

Since  $PQ_1 = PQ_2 = 0$  we find

$$Q_2M_{2-1}Q_2 = c_{2-1}Q_2TQ_2 = 2c_{2-1}Q_2WQ_2$$

where T and W are the integral operators with kernels

$$T(x,y) = v(x)|x-y|^2v(y)$$
, and  $W(x,y) = v(x)\langle x,y\rangle v(y)$ .

Letting  $X_j$  be the operator of multiplication by  $x_j$  (for  $x = (x_1, x_2)$ ) we find

$$Q_2 M_{2,-1} Q_2 = 2c_{2,0} \sum_{j=1}^d \langle \Theta_j, \cdot \rangle \Theta_j,$$

where we have defined  $\Theta_j = Q_2 X_j v$ .

**Lemma 3.2.88.** [89, Theorem 6.2] Suppose that V satisfies Assumption 3.2.62. Then the zero eigenspace of H satisfies  $Ker(H) \cong Range(Q_3)$ .

*Proof.* Suppose that  $\varphi \in \text{Range}(Q_3)$ . Then  $\varphi$  is orthogonal to both  $\text{Range}(T_2)$  and  $\text{Range}(T_3)$ , so that

$$\langle v, M_{0,0}\varphi \rangle = 0 = \langle v, X_j\varphi \rangle$$
 (3.44)

for j=1,2. Then Lemma 3.2.81 shows that  $\psi=Uv\varphi\in\mathcal{N}$ . Conversely, suppose that  $\psi\in\mathcal{N}$  and let  $\varphi=Uv\psi$ . Since  $\psi\in\mathcal{H}$ , Lemma 3.2.82 shows that  $\varphi\in Q_1\mathcal{H}$  and Equation (3.44) holds for  $\varphi$ . Thus  $\varphi\in\mathrm{Range}(Q_3)$ . Hence we find

$$Dim(Range(Q_3)) = Dim(Ker(H))$$

and the map  $\varphi \mapsto \psi$  is a bijection.

We can explicitly characterise the projection onto the zero eigenspace in terms of the operators we have seen.

**Lemma 3.2.89.** [63, Lemma 5.6] Suppose that n = 2 and V satisfies Assumption 3.2.62. Then the projection  $P_0$  onto the kernel of H is given by

$$P_0 = G_{0,0}vQ_3D_3Q_3vG_{0,0}.$$

*Proof.* Let  $(\varphi_j)_{j=1}^{N_0}$  be an orthonormal basis for Range $(Q_3)$ . Then for each  $1 \leq j \leq N_0$  we have  $0 = \varphi_j + UvG_{0,0}v\varphi_j$  and  $\varphi_j = Uv\psi_j$  for some  $\psi_j \in \mathcal{H}$ . Since  $PQ_3 = 0$  we also find

$$\int_{\mathbb{R}^2} V(x)\psi_j(x) \, \mathrm{d}x = \int_{\mathbb{R}^2} v(x)\varphi_j(x) \, \mathrm{d}x = 0$$

for all j. Since the  $(\varphi_j)$  are linearly independent, so are the  $(\psi_j)$  and they satisfy

$$\psi_j + G_{0,0}V\psi_j = 0.$$

Using the fact that the  $(\varphi_j)$  are a basis for  $Q_3\mathcal{H}$  we find for any  $f\in\mathcal{H}$  that

$$Q_3 v G_{0,0} f = \sum_{j=1}^{N_0} \langle v G_{0,0} f, \varphi_j \rangle \varphi_j = -\sum_{j=1}^{N_0} \langle f, \psi_j \rangle \varphi_j.$$

Let  $A = (A_{ij})$  be the matrix representation of  $Q_3vG_{2,-1}vQ_3$  with respect to the orthonormal basis  $(\varphi_j)$ . Then using Equation (3.42) we find

$$A_{ij} = \langle \varphi_i, Q_3 v G_{2,-1} v Q_3 \varphi_j \rangle = \langle G_{0,0} v \varphi_i, G_{0,0} v \varphi_j \rangle = \langle G_{0,0} V \psi_i, G_{0,0} V \psi_j \rangle = \langle \psi_i, \psi_j \rangle.$$

Defining  $P_0 = G_{0,0}vQ_3D_3Q_3vG_{0,0}$  we find for any  $f \in \mathcal{H}$  that

$$P_0 f = -\sum_{j=1}^{N_0} G_{0,0} v Q_3 D_3 \varphi_j \langle f, \psi_j \rangle = -\sum_{i,j=1}^{N_0} G_{0,0} v Q_3 (A^{-1})_{ij} \varphi_i \langle f, \psi_j \rangle = \sum_{i,j=1}^{N_0} \psi_i (A^{-1})_{ij} \langle f, \psi_j \rangle.$$

Thus we see that  $P_0\psi_j = \psi_j$ , so that the range of  $P_0$  is span $(\psi_j)$  and  $P_0$  is the identity on its range, which implies  $P_0$  is the projection onto the zero eigenspace by Lemma 3.2.88.  $\square$ 

**Theorem 3.2.90.** Suppose that n = 2 and V satisfies Assumption 3.2.62. Then for sufficiently small k we have

$$R(k^{2}) - R_{0}(k^{2}) = -k^{-2}(P_{0} - X) + k^{-2}\eta^{-1}\tilde{R}_{20}(k),$$

where  $\tilde{R}_{20}(k)$  is uniformly bounded. Here X is given by

$$X = G_{0,0}vYvG_{0,-1} + G_{0,-1}vYvG_{0,0}$$

with  $Y = \lim_{k \to 0} \eta^{-1} \tilde{T}_3(k)$ .

*Proof.* The proof follows from Theorem 3.2.85 and the definition

$$R(-k^2) = R_0(-k^2) - R_0(-k^2)vM(k)^{-1}vR_0(-k^2)$$

and a comparison of the coefficients of  $k^{-2}$ .

#### **3.2.6** Dimension n = 1

In this section we determine the low energy resolvent expansion in dimension n=1 using the Feshbach inversion method of Lemma 3.1.1 until it terminates. The expansions lead to explicit characterisations of the obstruction subspaces in terms of the kernel of H and zero energy resonances. The method we follow has been described in [89], although more detail has been included here regarding the specific coefficient operators as will be required in later chapters. We note that similar expansions have been introduced in various contexts throughout the literature, see for example [116], [29] and [28].

We first write

$$R_0(x, y, k^2) = (2k)^{-1} e^{-k|x-y|} = (2k)^{-1} - \frac{1}{2}|x-y| + \frac{k}{4}|x-y|^2 + O(k^2).$$

**Definition 3.2.91.** We define the operators  $G_j$  for j = -1, 0, 1 by the integral kernels

$$G_{-1}(x,y) = \frac{1}{2}$$
,  $G_0(x,y) = -\frac{1}{2}|x-y|$ , and  $G_1(x,y) = \frac{1}{4}|x-y|^2$ 

and the operators  $M_j$  by  $M_0 = U + vG_0v$  and  $M_j = vG_jv$ . Note that  $M_{-1} = \frac{1}{2}||v||_2^2 P$ , where  $P = ||v||_2^{-2} \langle v, \cdot \rangle v$ .

**Lemma 3.2.92.** [89, Lemma 5.1] Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 7$ . Then  $M - k^{-1}M_{-1} - U$  is a uniformly bounded compact operator valued function on  $F = \{k \in \mathbb{C} : \text{Re}(k) \geq 0 \text{ and } |k| \leq 1\}$ . Furthermore, we have the expansion

$$M(k) = k^{-1}M_{-1} + M_0 + kM_1 + k^2\tilde{R}_0(k),$$

where  $\tilde{R}_0(k)$  is uniformly bounded in k.

We let  $\alpha = ||v||_2^2$  and note that since P is a rank one projection, P is never invertible, Define the projection  $Q = \operatorname{Id} - P$ . We let  $\tilde{M}(k) = 2k\alpha^{-1}M(k)$  and obtain the following.

**Lemma 3.2.93.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 7$ . Then for sufficiently small k the operator  $\tilde{M}(k) + Q$  is invertible and we have the expansion

$$(\tilde{M}(k) + Q)^{-1} = \operatorname{Id} - 2k\alpha^{-1}M_0 - 2k^2\alpha^{-1}M_1 + 4\alpha^{-2}k^2M_0^2 + k^3\tilde{R}_1(k),$$

where  $\tilde{R}_1(k)$  is uniformly bounded.

*Proof.* For sufficiently small k we have the estimate

$$\left| \left| 2k\alpha^{-1}M_0 + 2k^2\alpha^{-1}M_1 + 2k^3\alpha^{-1}\tilde{R}_0(k) \right| \right| < 1.$$

Note that

$$\tilde{M}(k) = P + 2k\alpha^{-1}M_0 + 2k^2\alpha^{-1}M_1 + 2k^3\alpha^{-1}\tilde{R}_0(k).$$

We thus compute the Neumann expansion

$$(\tilde{M}(k) + Q)^{-1} = (\operatorname{Id} + 2k\alpha^{-1}M_0 + 2k^2\alpha^{-1}M_1 + 2k^3\alpha^{-1}\tilde{R}_0(k))^{-1}$$
$$= \operatorname{Id} - 2k\alpha^{-1}M_0 - 2k^2\alpha^{-1}M_1 + 4\alpha^{-2}k^2M_0^2 + k^3\tilde{R}_1(k),$$

where  $\tilde{R}_1(k)$  is uniformly bounded.

We define the operator

$$B_1(k) = Q - Q(\tilde{M}(k) + Q)^{-1}Q$$
  
=  $2k\alpha^{-1}QM_0Q + 2k^2\alpha^{-1}QM_1Q - 4\alpha^{-2}k^2QM_0^2Q - k^3Q\tilde{R}_1(k)Q$ .

We also define the operator  $\tilde{B}_1(k) = \frac{\alpha}{2k}B_1(k)$ , leading to the following definition.

**Definition 3.2.94.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 7$ . We say that zero is a regular point for H if the operator  $QM_0Q$  is invertible on  $Q\mathcal{H}$ . In this case we write  $D_1 = (QM_0Q)^{-1}$ .

In the case of a regular point for H we can determine the inverse of the operator M(k) using Lemma 3.1.1.

**Lemma 3.2.95.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 7$  and that zero is a regular point for H. Then we have the expansion

$$M(k)^{-1} = D_1 + 2\alpha^{-1} \mathrm{Id} + \tilde{R}_3(k),$$

where  $\tilde{R}_3(k)$  is uniformly bounded.

*Proof.* For sufficiently small k we have the estimate

$$\left| \left| kQM_1QD_1 - 2\alpha^{-1}kQM_0^2QD_1 - \frac{\alpha}{2}k^2Q\tilde{R}_1(k)QD_1 \right| \right| < 1.$$

We can thus compute the Neumann expansion

$$\tilde{B}_{1}(k)^{-1} = (QM_{0}Q + kQM_{1}Q - 2\alpha^{-1}kQM_{0}^{2}Q - \frac{\alpha}{2}k^{2}Q\tilde{R}_{1}(k)Q)^{-1}$$

$$= D_{1}(\mathrm{Id} + kQM_{1}QD_{1} - 2\alpha^{-1}kQM_{0}^{2}QD_{1} - \frac{\alpha}{2}k^{2}Q\tilde{R}_{1}(k)QD_{1})^{-1}$$

$$= D_{1} - kD_{1}M_{1}D_{1} + 2\alpha^{-1}kD_{1}M_{0}^{2}D_{1} + k^{2}\tilde{R}_{2}(k),$$

where  $\tilde{R}_2(k)$  is uniformly bounded. We now use Lemma 3.1.1 to obtain

$$M(k)^{-1} = 2\alpha^{-1}k\tilde{M}(k)^{-1} = 2\alpha^{-1}((\tilde{M}(k) + Q)^{-1} + \frac{\alpha}{2k}(\tilde{M}(k) + Q)^{-1}Q\tilde{B}_1(k)^{-1}(\tilde{M}(k) + Q)^{-1})$$
  
=  $D_1 + 2\alpha^{-1}\text{Id} + k\tilde{R}_3(k)$ ,

where  $\tilde{R}_3(k)$  is uniformly bounded.

The definition of the operator  $\tilde{B}_1(k)$  also leads to the following definition.

**Definition 3.2.96.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 7$ . If the operator  $QM_0Q$  is not invertible on  $Q\mathcal{H}$  we say that zero is an *exceptional point* for  $\mathcal{H}$ . In this case we let  $Q_1$  denote the orthogonal projection onto the kernel of  $QM_0Q$ . Then the operator  $QM_0Q + Q_1$  is invertible and we denote the inverse by  $D_1$ .

In the case of an exceptional point, we cannot simply use a Neumann expansion to invert  $\tilde{B}_1(k)$  directly. Instead we require another application of Lemma 3.1.1.

**Lemma 3.2.97.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 7$  and that zero is an exceptional point of the second kind for H. Then for sufficiently small k the operator  $\tilde{B}_1(k) + Q_1$  is invertible with inverse

$$(\tilde{B}_1(k) + Q_1)^{-1} = D_1 - kD_1M_1D_1 + 2\alpha^{-1}kD_1M_0^2D_1 + k^2\tilde{R}_2(k)$$

where  $\tilde{R}_2(k)$  is uniformly bounded.

*Proof.* For sufficiently small k we have the estimate

$$\left| \left| kQM_1QD_1 - 2\alpha^{-1}kQM_0^2QD_1 - \frac{\alpha}{2}k^2Q\tilde{R}_1(k)QD_1 \right| \right| < 1.$$

We can thus compute the Neumann expansion

$$(\tilde{B}_{1}(k) + Q_{1})^{-1} = (QM_{0}Q + Q_{1} + kQM_{1}Q - 2\alpha^{-1}kQM_{0}^{2}Q - \frac{\alpha}{2}k^{2}Q\tilde{R}_{1}(k)Q)^{-1}$$

$$= D_{1}(\mathrm{Id} + kQM_{1}QD_{1} - 2\alpha^{-1}kQM_{0}^{2}QD_{1} - \frac{\alpha}{2}k^{2}Q\tilde{R}_{1}(k)QD_{1})^{-1}$$

$$= D_{1} - kD_{1}M_{1}D_{1} + 2\alpha^{-1}kD_{1}M_{0}^{2}D_{1} + k^{2}\tilde{R}_{2}(k),$$

where  $\tilde{R}_2(k)$  is uniformly bounded.

We now apply Lemma 3.1.1 to the operator  $\tilde{B}_1(k)$ , however we first need to know that the Feshbach expansion terminates. To do so we determine the obstruction subspace  $Q_1$  explicitly and show that  $Q_1$  is a rank one operator.

**Lemma 3.2.98.** [89, Theorem 5.2] Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 7$  and that  $Q_1 \neq 0$ . Then  $\varphi \in Q_1\mathcal{H}$  with  $||\varphi|| = 1$  if and only if there exists  $\psi \in L^{\infty}(\mathbb{R})$ ,  $\psi \notin \mathcal{H}$  such that  $H\psi = 0$  in the sense of distributions and  $Uv\psi = \varphi$ . Furthermore, the space  $Q_1\mathcal{H}$  is at most one dimensional.

*Proof.* Suppose  $\varphi \in Q_1 \mathcal{H}$  with  $||\varphi|| = 1$  and define

$$\psi(x) = c_1 + \frac{1}{2} \int_{\mathbb{R}} |x - y| v(y) \varphi(y) \, \mathrm{d}y,$$

where  $c_1 = ||v||_2^{-2} \langle v, M_0 \varphi \rangle$ . Then we find

$$Uv\psi = c_1Uv + U(U - M_0)\varphi = c_1Uv + \varphi - UM_0\varphi = c_1Uv + \varphi - UPM_0\varphi = \varphi.$$

Differentiating in the sense of distributions shows  $H\psi = 0$ . To see that  $\psi \notin \mathcal{H}$ , we let

$$c_2 = \frac{1}{2} \int_{\mathbb{R}} y v(y) \varphi(y) \, \mathrm{d}y$$

and note that

$$\psi(x) = c_1 - c_2 \operatorname{sign}(x) + \begin{cases} -x \int_x^\infty v(y) \varphi(y) \, dy & \text{if } x \ge 0, \\ x \int_{-\infty}^x v(y) \varphi(y) \, dy & \text{if } x \le 0. \end{cases}$$
(3.45)

Suppose that  $c_1 = c_2 = 0$ . Then Equation (3.45) gives

$$\psi(x) = \int_{x}^{\infty} (y - x)V(y)\psi(y) \,dy,$$

a homogenous Volterra equation. Our decay assumptions on V give that  $\psi(x) = 0$  for large |x|. Uniqueness properties of the Volterra equation then give  $\psi = 0$  for all x and

thus  $c_1 = c_2 = 0$  if and only if  $\psi = 0$ . We can determine

$$c_{+} := \lim_{x \to \infty} \psi(x) = c_{1} + c_{2}, \quad \text{and} \quad c_{-} := \lim_{x \to -\infty} \psi(x) = c_{1} - c_{2}$$
 (3.46)

and thus  $\psi \in L^{\infty}(\mathbb{R})$  and  $\psi \notin \mathcal{H}$ . For the converse, suppose  $\psi \in L^{\infty}(\mathbb{R})$  satisfies  $H\psi = 0$  in the sense of distributions. Defining  $\varphi = Uv\psi$  we have (in the sense of distributions) the equality

$$\frac{\mathrm{d}^2}{\mathrm{d}x^2}\psi(x) = V(x)\psi(x) = v(x)\varphi(x).$$

Let  $\chi \in C_c^{\infty}(\mathbb{R})$  be such that  $\chi(x) = 1$  for  $|x| \le 1$  and  $\chi(x) = 0$  for |x| > 2. Then for any  $\delta > 0$  we have the estimate

$$\left| \int_{\mathbb{R}} v(x) \varphi(x) \chi(\delta x) \, \mathrm{d}x \right| = \left| \int_{\mathbb{R}} \left( \frac{\mathrm{d}^2 \psi}{\mathrm{d}x^2} \right) \chi(\delta x) \, \mathrm{d}x \right| = \left| \int_{\mathbb{R}} \psi(x) \left( \frac{\mathrm{d}^2}{\mathrm{d}x^2} \chi(\delta x) \right) \, \mathrm{d}x \right|$$
$$= \delta^2 \left| \int_{\mathbb{R}} \psi(x) \chi''(\delta x) \, \mathrm{d}x \right| \le \delta ||\psi||_{\infty} \int_{\mathbb{R}} |\chi''(x)| \, \mathrm{d}x.$$

Taking the limit as  $\delta \to 0$  using the dominated convergence theorem gives

$$\int_{\mathbb{R}} v(x)\varphi(x) \, \mathrm{d}x = 0,$$

so that  $\varphi \in Q\mathcal{H}$ . Define

$$\Theta(x) = \frac{1}{2} \int_{\mathbb{R}} |x - y| v(y) \varphi(y) \, \mathrm{d}y = \frac{1}{2} \int_{\mathbb{R}} |x - y| V(y) \psi(y) \, \mathrm{d}y.$$

Differentiating in the sense of distributions gives the equality

$$\frac{\mathrm{d}^2 \psi}{\mathrm{d}x^2} = \frac{\mathrm{d}^2 \Theta}{\mathrm{d}x^2},$$

so that  $\Theta(x) = a + bx + \psi(x)$  for some  $a, b \in \mathbb{C}$ . A similar argument to the one leading to Equation (3.45) shows that  $\Theta \in L^{\infty}(\mathbb{R})$ . Multiplying  $\Theta$  by v we obtain the relation  $(U - M_0)\varphi = U\varphi + av$ , which implies  $M_0\varphi = -av$  and thus  $QM_0\varphi = 0$  and so  $\varphi \in Q_1\mathcal{H}$ . By construction we have  $||\varphi|| = 1$ . To prove that  $Q_1\mathcal{H}$  is one-dimensional we suppose that there exist  $\varphi, \tilde{\varphi} \in Q_1\mathcal{H}$ . Then for  $x \geq 0$  we have the corresponding  $\psi, \tilde{\psi}$  as

$$\psi(x) = c_1 - c_2 - x \int_x^\infty v(y)\varphi(y) \, dy, \quad \tilde{\psi}(x) = \tilde{c}_1 - \tilde{c}_2 + \int_x^\infty (y - x)v(y)\tilde{\varphi}(y) \, dy.$$

There exists  $a \in \mathbb{C}$  such that  $c_1 - c_2 = a(\tilde{c}_1 - \tilde{c}_2)$  and so we find

$$\psi(x) - a\tilde{\psi}(x) = -x \int_{x}^{\infty} v(y)(\varphi(y) - a\tilde{\varphi}(y)) \,dy.$$

Again by uniqueness properties of Volterra type equations, we find  $\psi - a\tilde{\psi} = 0$  and so  $\varphi - a\tilde{\varphi} = 0$  also. Thus we have  $\text{Dim}(Q_1\mathcal{H}) = 1$ .

**Definition 3.2.99.** Suppose that  $Q_1 \neq 0$ . Then for any  $\varphi \in Q_1\mathcal{H}$  we say that the corresponding  $\psi$  with  $\varphi = Uv\psi$  from Lemma 3.2.98 is a zero energy resonance for H. We choose a normalised zero energy resonance by the condition that  $c_+^2 + c_-^2 = 1$ .

**Lemma 3.2.100.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 7$  and that zero is an exceptional point for H. Then for sufficiently small k the operator  $B_2(k)$  is invertible with inverse

$$B_2(k)^{-1} = k^{-1}\tilde{c}^{-1}Q_1 - \tilde{c}^{-2}Q_1\tilde{R}_2(k)Q_2 + k\tilde{R}_3(k),$$

where  $\tilde{R}_3(k)$  is uniformly bounded in k.

*Proof.* Lemma 3.1.1 gives that the operator  $\tilde{B}_1(k)$  is invertible if and only if the operator

$$B_2(k) = Q_1 - Q_1(\tilde{B}_1(k) + Q_1)^{-1}Q_1$$

is invertible. For sufficiently small k we have

$$\left| \left| kQ_1 \tilde{R}_2(k) Q_1 \right| \right| < 1.$$

So we compute that  $B_2(k) = kQ_1M_1Q_1 - \alpha^{-1}kQ_1M_0^2Q_1 - k^2Q_1\tilde{R}_2(k)$ , where we have used the relation  $Q_1D_1 = D_1Q_1 = Q_1$ . Note that there exists a constant  $\tilde{c}$  such that  $Q_1M_1Q_1 - \alpha^{-1}Q_1M_0^2Q_1 = \tilde{c}Q_1$ . We now use a Neumann expansion to obtain

$$B_2(k)^{-1} = k^{-1}\tilde{c}^{-1}Q_1(\operatorname{Id} - \tilde{c}^{-1}kQ_1\tilde{R}_2(k)Q_1)^{-1} = k^{-1}\tilde{c}^{-1}Q_1 - \tilde{c}^{-2}Q_1\tilde{R}_2(k)Q_2 + k\tilde{R}_3(k),$$

where  $\tilde{R}_3(k)$  is uniformly bounded in k.

We are finally ready to determine the inversion formula for M(k).

**Theorem 3.2.101.** [89, Theorem 5.2] Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 7$  and that zero is an exceptional point of the second kind for H. Then for sufficiently small k we have the expansion

$$M(k)^{-1} = \tilde{c}^{-1}k^{-1}Q_1 + \tilde{R}_4(k),$$

where  $\tilde{R}_4(k)$  is uniformly bounded.

Proof. Lemma 3.1.1 gives that

$$M(k)^{-1}$$

$$= 2\alpha^{-1}k \left( (\tilde{M}(k) + Q)^{-1} + \frac{\alpha}{2}k^{-1}(\tilde{M}(k) + Q)^{-1}Q\tilde{B}_{1}(k)^{-1}Q(\tilde{M}(k) + Q)^{-1} \right)$$

$$= 2\alpha^{-1}k \left( (\tilde{M}(k) + Q)^{-1} + \frac{\alpha}{2}k^{-1}(\tilde{M}(k) + Q)^{-1}Q(\tilde{B}_{1}(k) + Q_{1})^{-1}Q(\tilde{M}(k) + Q)^{-1} + \frac{\alpha}{2}k^{-1}(\tilde{M}(k) + Q)^{-1}Q(\tilde{B}_{1}(k) + Q_{1})^{-1}Q(\tilde{M}(k) + Q)^{-1} \right)$$

$$+ \frac{\alpha}{2}k^{-1}(\tilde{M}(k) + Q)^{-1}Q(\tilde{B}_{1}(k) + Q_{1})^{-1}Q_{1}B_{2}(k)^{-1}Q_{1}(\tilde{B}_{1}(k) + Q_{1})^{-1}Q(\tilde{M}(k) + Q)^{-1} \right).$$

The first two terms have already been determined in Lemma 3.2.95. So we expand out the lowest order terms in the expression

$$(\tilde{M}(k) + Q)^{-1}Q(\tilde{B}_{1}(k) + Q_{1})^{-1}Q_{1}B_{2}(k)^{-1}Q_{1}(\tilde{B}_{1}(k) + Q_{1})^{-1}Q(\tilde{M}(k) + Q)^{-1}$$

$$= \tilde{c}^{-1}k^{-1}Q_{1} + \alpha^{-1}\tilde{c}^{-1}(M_{0}Q_{1} + Q_{1}M_{1}) + \tilde{c}^{-1}(D_{1}M_{1}Q_{1} + Q_{1}M_{1}D_{1}) - \tilde{c}^{-3}Q_{1}\tilde{R}_{2}(k)Q_{1}$$

$$+ \tilde{R}_{4}(k),$$

where  $\tilde{R}_4(k)$  is uniformly bounded.

Remark 3.2.102. We can further simplify the coefficients in the above expansion. Since  $Q_1$  is a rank one projection we can obtain (see [89, Lemma 5.4(iii)]) that

$$\tilde{c}^{-1} = -\alpha^{-1}(c_1^2 + c_2^2) = -\frac{1}{2\alpha}(c_+^2 + c_-^2),$$

with  $c_1, c_2, c_{\pm}$  defined in the proof of Lemma 3.2.98.

We conclude this chapter with an expansion for the resolvent difference  $R(-k^2)$  –  $R_0(-k^2)$ . We provide only a reference with no proof, since the details are rather involved (see [116, Section 5]). The important feature to note is that as in higher dimensions, the worst singularity as  $k \to 0$  is related to the existence of zero-energy eigenvalues for H and the next worst singularity is related to the existence of resonances.

**Theorem 3.2.103** ([116, Section 5]). Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 7$ . Then for sufficiently small k we have the expansion

$$R(k-^2) - R_0(-k^2) = -P_0k^{-2} + k^{-1}A + \tilde{R}_{16}(k)$$

where A is a bounded operator depending on the existence of a resonance and  $\tilde{R}_{16}(k)$  is uniformly bounded.

# Chapter 4

# The structure of the wave operator

In this chapter we show that the wave operator  $W_{-}$  satisfies

$$W_{-} = \operatorname{Id} + \frac{1}{2} \left( \operatorname{Id} + \tanh (\pi D_n) - i \cosh (\pi D_n)^{-1} \right) (S - \operatorname{Id}) + K, \tag{4.1}$$

where  $D_n$  is the generator of dilations on  $L^2(\mathbb{R}^n)$ , S the scattering matrix and K is a compact operator. We will frequently use the notation  $\varphi(x) = \frac{1}{2}(1+\tanh(\pi x)-i\cosh(\pi x)^{-1})$ . In Section 4.1 we use the resolvent expansions of Chapter 3 to determine the low energy behaviour of the scattering matrix. In Section 4.2 we give a direct intuitive argument why Equation (4.1) should hold based on the generalised eigenfunctions described in Section 2.4.2. Unfortunately this method is limited by difficulties in showing the remainder is compact and is not sensitive to the fine structure of the spectrum of H exhibited in Chapter 3. In Section 4.3 we give a different approach, which yields the answer less directly, using carefully the resolvent expansions of Chapter 3. The main result of this chapter is the following theorem, whose proof relies on several preparatory results and some additional decay assumptions on the potential, described in Assumption 4.3.1.

**Theorem 4.0.1.** Let  $n \geq 2$  and suppose that  $\rho$  satisfies Assumption 4.3.1, and let V satisfy  $|V(x)| \leq C(1+|x|)^{-\rho}$  for almost every  $x \in \mathbb{R}^n$ . In dimension n=2 we suppose also that there are no p-resonances and in dimension n=4 we suppose there are no resonances. Then in  $\mathcal{B}(\mathcal{H})$  we have the equality

$$W_{-} = \operatorname{Id} + \varphi(D_n)(S - \operatorname{Id}) + K,$$

where 
$$K \in \mathcal{K}(\mathcal{H})$$
 and  $\varphi(x) = \frac{1}{2}(1 + \tanh(\pi x) - i\cosh(\pi x)^{-1}).$ 

Theorem 4.0.1 has appeared in [5] as joint work with Adam Rennie, with the case n=3 having been proved already in [141].

# 4.1 The scattering matrix

In this section we present how the low energy behaviour of the scattering matrix in each dimension. The stationary expression for the scattering matrix of Theorem 2.4.32 shows that such low energy expansions depend on the low energy behaviour of the resolvent discussed in Chapter 3 and the trace operator  $\Gamma_0(\cdot)$  of Definition 2.4.14.

## 4.1.1 Low energy expansions of the trace operator

In this subsection we present the low energy behaviour of the operator  $\Gamma_0(\cdot)$  of Definition 2.4.14. Such an expansion was considered in dimension n=3 in [88, Section 5] and the results we present here are straightforward generalisations.

For  $\lambda > 0$  the trace operator  $\gamma(\lambda) : \mathcal{S}(\mathbb{R}^n) \to \mathcal{P}$  is continuous and extends to a bounded operator in  $\mathcal{B}(H^{s,t},\mathcal{P})$  for each  $s > \frac{1}{2}$  and  $t \in \mathbb{R}$  [106, Theorem 2.4.3]. We need the asymptotic development of  $\gamma(\lambda^{\frac{1}{2}})\mathcal{F}$  as  $\lambda \to 0$ . We can compute for  $f \in C_c^{\infty}(\mathbb{R}^n)$  and  $\omega \in \mathbb{S}^{n-1}$  the expansion

$$[\gamma(\lambda^{\frac{1}{2}})\mathcal{F}f](\omega) = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} e^{-i\lambda^{\frac{1}{2}}\langle x,\omega\rangle} f(x) dx$$

$$= (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} \sum_{j=0}^K \frac{1}{j!} (i\lambda^{\frac{1}{2}})^j (-\langle x,\omega\rangle)^j f(x) dx + O\left(\lambda^{\frac{K+1}{2}}\right)$$

$$:= \sum_{j=0}^K (i\lambda^{\frac{1}{2}})^j [\gamma_j f](\omega) + O\left(\lambda^{\frac{K+1}{2}}\right)$$

as  $\lambda \to 0$  in  $\mathcal{B}(H^{s,t}, \mathcal{P})$  for appropriate s, t. The operators  $\gamma_j$  can be considered as operators in certain weighted Sobolev spaces, with higher terms in the series requiring convergence in Sobolev spaces with higher decay. Jensen [88, Equation 5.4] states the following result in dimension n=3 and the result generalises in a straightforward manner to each dimension.

**Lemma 4.1.1.** Fix  $j \in \mathbb{N}$ . For each  $s \geq 0$  and  $t > j + \frac{n}{2}$  we have  $\gamma_j \in \mathcal{B}(H^{s,t}, \mathcal{P})$ .

*Proof.* For  $t > j + \frac{n}{2}$  and s = 0, we can estimate that

$$||\gamma_{j}f||_{\mathcal{P}}^{2} = \int_{\mathbb{S}^{n-1}} |[\gamma_{j}f](\omega)|^{2} d\omega = \int_{\mathbb{S}^{n-1}} \left| (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^{n}} (-\langle x, \omega \rangle)^{j} f(x) dx \right|^{2} d\omega$$

$$\leq (2\pi)^{-n} \int_{\mathbb{S}^{n-1}} \left( \int_{\mathbb{R}^{n}} |\langle \omega, x \rangle|^{j} |f(x)| dx \right)^{2} d\omega$$

$$= (2\pi)^{-n} \int_{\mathbb{S}^{n-1}} \left( \int_{\mathbb{R}^{n}} |\langle \omega, x \rangle|^{j} (1 + |x|^{2})^{-\frac{t}{2}} (1 + |x|^{2})^{\frac{t}{2}} |f(x)| dx \right)^{2} d\omega.$$

We next use the Cauchy-Schwarz inequality to obtain the estimate

$$(2\pi)^{-n} \int_{\mathbb{S}^{n-1}} \left( \int_{\mathbb{R}^n} |\langle \omega, x \rangle|^j (1 + |x|^2)^{-\frac{t}{2}} (1 + |x|^2)^{\frac{t}{2}} |f(x)| \, \mathrm{d}x \right)^2 \, \mathrm{d}\omega$$

$$\leq (2\pi)^{-n} \int_{\mathbb{S}^{n-1}} \left( \int_{\mathbb{R}^n} |\langle \omega, x \rangle|^{2j} (1 + |x|^2)^{-t} \, \mathrm{d}x \right)^2 \left( \int_{\mathbb{R}^n} (1 + |y|^2)^t |f(y)|^2 \, \mathrm{d}y \right)^2 \, \mathrm{d}\omega$$

$$= C_j ||f||_{H^{0,t}}^2.$$

Thus for  $t > j + \frac{n}{2}$  we find  $\gamma_j \in \mathcal{B}(H^{0,t}, \mathcal{P})$ . For s > 0 we use the inclusion  $H^{s,t} \subset H^{0,t}$ .  $\square$ 

For each  $\lambda > 0$  the operator  $\Gamma_0(\lambda) : \mathcal{S}(\mathbb{R}^n) \to \mathcal{P}$  extends to an element of  $\mathcal{B}(H^{s,t},\mathcal{P})$  for each  $s \in \mathbb{R}$  and  $t > \frac{1}{2}$ , as well as  $\lambda \mapsto \Gamma_0(\lambda) \in \mathcal{B}(H^{s,t},\mathcal{P})$  is continuous. These follow immediately since  $\Gamma_0(\lambda) = 2^{-\frac{1}{2}} \lambda^{\frac{n-2}{4}} \gamma(\lambda^{\frac{1}{2}}) \mathcal{F}$ .

**Lemma 4.1.2.** For  $K \in \mathbb{N}$ ,  $s \geq 0$  and  $t > \frac{n}{2} + K$  we have the expansion

$$\Gamma_0(\lambda) = 2^{-\frac{1}{2}\lambda^{\frac{n-2}{4}}} \sum_{j=0}^K (i\lambda^{\frac{1}{2}})^j \gamma_j + O\left(\lambda^{\frac{n}{4} + \frac{K}{2}}\right)$$

as  $\lambda \to 0$  in  $\mathcal{B}(H^{s,t}, \mathcal{P})$ .

*Proof.* Let s=0 and  $t>\frac{n}{2}+K$ . Let  $Q_K$  be the K-th Taylor polynomial for the exponential function. Note for  $y\in\mathbb{R}$  the estimate

$$\left| e^{iy} - Q_K(iy) \right| \le \frac{|y|^{K+1}}{(K+1)!}.$$

Then we can compute for  $f \in H^{0,t}$  and  $\lambda \geq 0$  that

$$\left\| \left( \Gamma_0(\lambda) - 2^{-\frac{1}{2}} \lambda^{\frac{n-2}{4}} \sum_{j=0}^K (i\lambda^{\frac{1}{2}})^j \gamma_j \right) f \right\|_{\mathcal{P}}^2$$

$$= 2^{-1} \lambda^{\frac{n-2}{2}} \int_{\mathbb{S}^{n-1}} \left( \left[ \mathcal{F} f \right] (\lambda^{\frac{1}{2}} \omega) - \sum_{j=0}^K (i\lambda^{\frac{1}{2}})^j [\gamma_j f] (\omega) \right)^2 d\omega$$

$$\leq 2^{-1} \lambda^{\frac{n-2}{2}} (2\pi)^{-n} \int_{\mathbb{S}^{n-1}} \left( \int_{\mathbb{R}^n} \left| e^{-i\lambda^{\frac{1}{2}} \langle x, \omega \rangle} - Q_K (i\lambda^{\frac{1}{2}} \langle x, \omega \rangle) \right| |f(x)| dx \right)^2 d\omega$$

$$\leq \frac{\lambda^{\frac{n}{2} + K}}{2(K+1)! (2\pi)^n} \int_{\mathbb{S}^{n-1}} \left( \int_{\mathbb{R}^n} |\langle x, \omega \rangle|^{K+1} |f(x)|^2 dx \right)^2 d\omega.$$

Using the same trick as Lemma 4.1.1 we write

$$\left\| \left( \Gamma_0(\lambda) - 2^{-\frac{1}{2}} \lambda^{\frac{n-2}{4}} \sum_{j=0}^K (i\lambda^{\frac{1}{2}})^j \gamma_j \right) f \right\|_{\mathcal{P}}^2$$

$$\leq \frac{\lambda^{\frac{n-2}{2}}}{2(K+1)!(2\pi)^n} \int_{\mathbb{S}^{n-1}} \left( \int_{\mathbb{R}^n} |\langle x, \omega \rangle|^{K+1} (1+|x|^2)^{-\frac{t}{2}} (1+|x|^2)^{\frac{t}{2}} |f(x)|^2 dx \right)^2 d\omega$$

$$\leq \frac{C_K \lambda^{\frac{n}{2}+K}}{2(K+1)!(2\pi)^n} \left\| |f| \right\|_{H^{0,t}}^2.$$

This proves the claim for s=0. The inclusion  $H^{s,t}\subset H^{0,t}$  for s>0 completes the proof.

Combining Lemma 4.1.2 with the results of Chapter 3 we can determine the low energy behaviour of the scattering matrix.

## 4.1.2 Low energy expansions of the scattering matrix

In this section we show how the low energy expansions of the resolvent can be used, in conjunction with Theorem 2.4.32, to determine the low energy behaviour of the scattering matrix. We will show that in odd dimensions the low energy behaviour is sensitive to the presence of resonances, whilst in even dimensions resonances have no effect. In dimensions  $n \geq 2$  it is found that generically, we have S(0) = Id.

The low energy behaviour of the scattering matrix has been determined using resolvent expansions in dimension n = 1 in [29] (see also [56, Theorem 2.15], [116] and [97, Proposition 9]).

**Theorem 4.1.3.** Suppose that n = 1 and V satisfies Assumption 2.2.14 for some  $\rho > \frac{5}{2}$ . Then we have

$$S(0) = \begin{pmatrix} 0 & -1 \\ -1 & 0 \end{pmatrix} \tag{4.2}$$

if there are no resonances and

$$S(0) = \begin{pmatrix} 2c_{+}c_{-} & c_{+}^{2} - c_{-}^{2} \\ c_{-}^{2} - c_{+}^{2} & 2c_{+}c_{-} \end{pmatrix}$$

$$\tag{4.3}$$

where  $c_{\pm} \in \mathbb{R} \setminus \{0\}$  with  $c_{+}^{2} + c_{-}^{2} = 1$ .

Remark 4.1.4. The  $c_{\pm}$  here denote the asymptotic values of the (normalised) resonance wave function at  $\pm \infty$  (see Lemma 3.2.98). Due to the use of a different basis some authors (see [97, Proposition 9]) obtain a slightly different expression for S(0). The relevant bases are described in Section 2.5.5. The essential feature to distinguish resonances is

that if there are no resonances then  $\operatorname{Det}(S(0)) = -1$  and if there are resonances then  $\operatorname{Det}(S(0)) = 1$ . Many more details about such calculations can be found also in [30] and [116]. We also deduce from our description the relation  $\lim_{\lambda \to \infty} S(\lambda) = \operatorname{Id}$ , which we will demonstrate in Corollary 4.3.10 holds in all dimensions in  $\mathcal{B}(\mathcal{P})$ .

We note that the dimension n = 1 result is atypical in the sense that for all dimensions  $n \ge 2$ , if there does not exist a resonance at zero then S(0) = Id. The generic failure of S(0) = Id means that the index pairing we discuss in Chapter 5 requires some modification to be applicable to dimension n = 1.

The low energy behaviour of the scattering matrix in dimension n = 2 has been proved in [142, Theorem 1.1] (see also [26, Theorem 4.3]). We note that in contrast to dimension n = 1 the behaviour is independent of the presence of resonances.

**Theorem 4.1.5.** Let n=2 and suppose V satisfies Assumption 2.2.14 for some  $\rho > 11$ . Then the scattering matrix satisfies  $S(0) = \operatorname{Id}$ .

*Proof.* For  $\lambda > 0$  we begin with Equation (2.73) in the symmetrised form

$$S(\lambda) - \operatorname{Id} = -2\pi i \Gamma_0(\lambda) v (U + v R_0(\lambda + i0) v)^{-1} v \Gamma_0(\lambda)^* = -2\pi i \Gamma_0(\lambda) v M(\lambda^{\frac{1}{2}})^{-1} v \Gamma_0(\lambda)^*$$

$$= -\pi i (\gamma_0 - i\lambda^{\frac{1}{2}} \gamma_1 + E(\lambda)) v M(\lambda^{\frac{1}{2}})^{-1} v (\gamma_0^* + i\lambda^{\frac{1}{2}} \gamma_1^* + E(\lambda)^*)$$

$$= -\pi i (\gamma_0 - i\lambda^{\frac{1}{2}} \gamma_1 + E(\lambda)) v \left( -\lambda^{-1} D_3 - \lambda^{-1} \tilde{T}_3(\lambda^{\frac{1}{2}}) + \frac{\lambda}{\ln(\lambda) - \frac{i\pi}{2}} K(\lambda) \right)$$

$$\times v (\gamma_0^* + i\lambda^{\frac{1}{2}} \gamma_1^* + E(\lambda)^*)$$

Here the operators  $E(\lambda)$  are  $O(\lambda)$  as  $\lambda \to 0$  by Lemma 4.1.2 and the operator  $K(\lambda)$  is bounded by Theorem 3.2.85. Thus any term involving  $E(\lambda)$  or  $E(\lambda)^*$  vanishes as  $\lambda \to 0$ . Using the relations  $PQ_3 = PD_3 = PT_3 = 0$  we obtain also  $\gamma_0 vQ_3 = Q_3 v\gamma_0^* = 0 = \gamma_0 vT_3 = T_3 v\gamma_0^*$ . Observe also that  $Q_3T_3 = 0$  and thus  $Q_3v\gamma_1^* = \gamma_1 vQ_3 = 0$ . Finally we check that

$$\lim_{\lambda \to 0} \frac{\lambda}{\ln(\lambda) - \frac{i\pi}{2}} = 0,$$

so that terms involving  $K(\lambda)$  vanish as  $\lambda \to 0$  and thus we obtain  $S(\lambda) \to \mathrm{Id}$ , as claimed.

As in dimension n = 1, the behaviour of the scattering matrix at zero in dimension n = 3 is dependent on the existence of resonances and has been determined in [88, Theorems 5.1-5.3], which we state below.

**Theorem 4.1.6.** Let n=3 and suppose that V satisfies Assumption 2.2.14 for some  $\rho > 5$ . Then we have

$$S(0) = \text{Id} - 2P_s$$

where  $P_s = 0$  if there are no resonances and  $P_s$  is the projection onto the spherical harmonic subspace of order 0 if there does exist a resonance (irrespective of whether or not V is spherically symmetric).

*Proof.* For  $\lambda > 0$  we begin with Equation (2.73) in the symmetrised form

$$S(\lambda) = \operatorname{Id} - 2\pi i \Gamma_0(\lambda) v (U + v R_0(\lambda + i0) v)^{-1} v \Gamma_0(\lambda)^* = \operatorname{Id} - 2\pi i \Gamma_0(\lambda) v M(\lambda^{\frac{1}{2}})^{-1} v \Gamma_0(\lambda)^*.$$

Now for  $\lambda > 0$  we use the expansion of Theorem 3.2.57 to obtain

$$S(\lambda) - \operatorname{Id} = -2\pi i \Gamma_0(\lambda) v M(\lambda^{\frac{1}{2}})^{-1} v \Gamma_0(\lambda)^*$$

$$= -2\pi i \Gamma_0(\lambda) v (\lambda^{-1} Q_2 D_2 Q_2 - \lambda^{-\frac{1}{2}} C_{-\frac{1}{2}} + \tilde{R}_7(\lambda^{\frac{1}{2}})) v \Gamma_0(\lambda)^*$$

$$= -\pi i (2\pi)^{-3} \lambda^{\frac{1}{2}} (\gamma_0 + E(\lambda)) v (\lambda^{-1} Q_2 D_2 Q_2 - \lambda^{-\frac{1}{2}} i C_{-\frac{1}{2}} + \tilde{R}_7(\lambda^{\frac{1}{2}}))$$

$$\times v (\gamma_0^* + E(\lambda)^*).$$

Here we have  $E(\lambda) = O(\lambda^{\frac{1}{2}})$  as  $\lambda \to 0$  by Lemma 4.1.2. The proof of Lemma 3.2.53 shows that  $\gamma_0 v Q_2 = 0$  and thus by duality  $Q_2 v \gamma_0^* = 0$ . So for  $\lambda > 0$  we find

$$S(\lambda) - \mathrm{Id} = -\pi (2\pi)^{-3} \gamma_0 v C_{-\frac{1}{2}} v \gamma_0^* + O(\lambda^{\frac{1}{2}}) = \pi i \gamma_0 v T_1 D_1 T_1 v \gamma_0^* + O(\lambda^{\frac{1}{2}}),$$

since every term containing a  $\gamma_0 vQ_2$  or  $Q_2 v\gamma_0^*$  vanishes. For  $f \in L^2(\mathbb{S}^2)$  and  $\omega \in \mathbb{S}^2$  we compute the expression

$$[\gamma_0 v T_1 D_1 T_1 v \gamma_0^* f](\omega) = \int_{\mathbb{R}^3} v(x) [T_1 D_1 T_1 v \gamma_0^* f](x) \, \mathrm{d}x = \int_{\mathbb{R}^3} v(x) [T_1 v \gamma_0^* f](x) \, \mathrm{d}x$$

$$= \int_{\mathbb{R}^3} v(x) \varphi(x) \langle v \gamma_0^* f, \varphi \rangle \, \mathrm{d}x$$

$$= \left( \int_{\mathbb{R}^3} v(x) \varphi(x) \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^3} \varphi(y) v(y) \, \mathrm{d}y \right) \left( \int_{\mathbb{S}^2} f(\theta) \, \mathrm{d}\theta \right)$$

$$= ||v\varphi||_1^2 \int_{\mathbb{S}^2} f(\theta) \, \mathrm{d}\theta.$$

We can write

$$[\gamma_0 f](\omega) = \int_{\mathbb{S}^2} f(\omega) d\omega = 4\pi [P_s f](\omega).$$

Recall that  $\varphi$  was chosen so that  $||v\varphi||_1 = (4\pi)^{\frac{1}{2}}$  (see Definition 3.2.58 and the proof of Lemma 3.2.60) and so  $4\pi^2 ||v\varphi||_1^2 (2\pi)^{-3} = 2$ . Any remaining terms containing an  $E(\lambda)$  or  $E(\lambda)^*$  vanish as  $\lambda \to 0$ . Hence we obtain  $S(\lambda) - \mathrm{Id} = -2P_s$  if there exists a zero energy resonance.

The notation  $P_s$  is to indicate that  $P_s$  is the projection onto the zeroth order spherical harmonic subspace, which correspond to s-waves. The next statement was almost

certainly known to Jensen based on the comments in [85, p.1], although we have found no statements in the literature.

**Theorem 4.1.7.** Suppose that  $n \ge 4$  and that V satisfies Assumption 2.2.14 for some  $\rho > 12$ . Then we have  $S(0) = \operatorname{Id}$ .

*Proof.* For  $\lambda > 0$  we begin with Equation (2.73) in the symmetrised form

$$S(\lambda) = \operatorname{Id} - 2\pi i \Gamma_0(\lambda) v (U + v R_0(\lambda + i0) v)^{-1} v \Gamma_0(\lambda)^* = \operatorname{Id} - 2\pi i \Gamma_0(\lambda) v M(\lambda^{\frac{1}{2}})^{-1} v \Gamma_0(\lambda)^*.$$

As  $\lambda \to 0$  we can then apply Lemma 4.1.2 to obtain the expansion for  $\Gamma_0(\lambda)$  and  $\Gamma_0(\lambda)^*$ . For  $n \ge 5$  we apply, for  $\lambda > 0$ , Theorem 3.2.9 if n is odd or Theorem 3.2.19 if n is even, the result is the expansion

$$S(\lambda) - \operatorname{Id} = -2\pi i \Gamma_0 v M(\lambda^{\frac{1}{2}}) v \Gamma_0(\lambda)^*$$
  
=  $-\pi i \lambda^{\frac{n-2}{2}} (\gamma_0 + E(\lambda)) v(\lambda^{-1} D_1 + \tilde{R}(\lambda^{\frac{1}{2}})) v(\gamma_0^* + E(\lambda)^*) \to 0$ 

as  $\lambda \to 0$  since  $\frac{n-2}{2} > 1$ . Here we have  $E(\lambda) = O(\lambda^{\frac{1}{2}})$  as  $\lambda \to 0$  by Lemma 4.1.2. In the case n = 4, we use for  $\lambda > 0$ , Theorem 3.2.37 to obtain the expansion

$$S(\lambda) - \operatorname{Id} = -2\pi i \Gamma_0(\lambda) v(M(\lambda^{\frac{1}{2}}) v \Gamma_0(\lambda)^*$$

$$= -2\pi i \lambda \left(\gamma_0 + E(\lambda)\right) \left(\lambda^{-1} h(\lambda^{\frac{1}{2}}) Q_1 \tilde{T}_1 Q_1 + \lambda^{-1} D_2 + h(\lambda^{\frac{1}{2}}) K_1 + \tilde{R}(\lambda^{\frac{1}{2}})\right)$$

$$\times \left(\gamma_0^* + E(\lambda)^*\right) \to 0$$

as  $\lambda \to 0$ , since  $\tilde{R}(\lambda^{\frac{1}{2}})$  is uniformly bounded,  $h(\lambda^{\frac{1}{2}}) \to 0$  as  $\lambda \to 0$ , the fact that  $\gamma_0 v Q_2 = 0$  by the proof of Lemma 3.2.26, the relation  $Q_2 D_2 Q_2 = D_2$  and  $E(\lambda) = O(\lambda^{\frac{1}{2}})$  as  $\lambda \to 0$ .  $\square$ 

# 4.2 Intuition for the form of the wave operator

Unpacking Equation (2.62) for the generalised plane waves  $\psi_{\pm}$  gives intuition about the form of the wave operator  $W_{-}$ , much like [98] in dimension n=3, although unfortunately does not lead to a complete proof.

Recall from Theorem 2.4.17 that the scattering amplitude  $a_{-}$  is defined to be the sub-leading behaviour of the generalised eigenfunction  $\psi_{-}$  as  $|x| \to \infty$ ; that is,

$$\psi_{-}(x,\omega,\lambda) = \psi_{0}(x,\omega,\lambda) + a_{-}(\hat{x},\omega,\lambda)w_{+}(|x|,\lambda) + k(x,\omega,\lambda) \tag{4.4}$$

where k decays as  $|x| \to \infty$  and we have used the notation  $\hat{x} = |x|^{-1}x$  for  $x \neq 0$ . The spherical wave is given by

$$w_{\pm}(r,|x|) = |x|^{-\frac{n-1}{2}} r^{\frac{n-2}{2}} e^{\pm ir|x|} e^{\mp i\frac{n-3}{4}\pi}.$$
 (4.5)

Define the operator  $T: C_c^{\infty}(\mathbb{R}^n) \subset L^2(\mathbb{R}^n) \to L^2(\mathbb{R}^n)$  by

$$[Tf](x) = c_n |x|^{-\frac{n-1}{2}} \int_{\mathbb{R}^+} \lambda^{\frac{n-3}{4}} e^{i\lambda^{\frac{1}{2}}|x|} [\mathcal{F}f](\lambda^{\frac{1}{2}}\hat{x}) d\lambda = 2c_n |x|^{-\frac{n-1}{2}} \int_{\mathbb{R}^+} \tau^{\frac{n-1}{2}} e^{i\tau|x|} [\mathcal{F}f](\tau \hat{x}) d\tau,$$
(4.6)

where  $f \in C_c^{\infty}(\mathbb{R}^n)$ ,  $x \in \mathbb{R}^n$  and  $c_n = -i2^{-1}(2\pi)^{-\frac{1}{2}}e^{-i\frac{n-3}{4}\pi}$ . Define the integral operator  $K: C_c^{\infty}(\mathbb{R}^n) \subset L^2(\mathbb{R}^n) \to L^2(\mathbb{R}^n)$  by

$$[Kf](x) = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} k(x,\hat{\xi},|\xi|^2) [\mathcal{F}f](\xi) \,d\xi.$$
 (4.7)

Using the approximation to the generalised eigenfunction  $\psi_{-}$  in Equation (4.4) in the stationary expression for the wave operators of Theorem 2.4.29 combined with the generalised Fourier transforms of Lemma 2.4.31 we obtain the following.

**Lemma 4.2.1.** Suppose that  $|V(x)| \le (1+|x|)^{-\rho}$  for some  $\rho > \frac{n+1}{2}$ . Define operators T and K by Equations (4.6) and (4.7). Then the wave operator  $W_-$  satisfies

$$W_{-} = \mathrm{Id} + T(S - \mathrm{Id}) + K.$$
 (4.8)

*Proof.* We recall from Theorem 2.4.29 and Lemma 2.4.31 the action of the wave operator as an integral against generalised eigenfunctions  $\psi_{-}$  for  $f \in C_c^{\infty}(\mathbb{R}^n) \subset L^2(\mathbb{R}^n)$  and  $x \in \mathbb{R}^n$  as

$$[W_{-}f](x) = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^{n}} \psi_{-}(x,\hat{\xi},|\xi|^{2}) [\mathcal{F}f](\xi) \,d\xi$$

$$= (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^{n}} \psi_{0}(x,\hat{\xi},|\xi|^{2}) [\mathcal{F}f](\xi) \,d\xi + (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^{n}} k(x,\xi) [\mathcal{F}f](\xi) \,d\xi$$

$$+ (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^{n}} a_{-}(\hat{x},\hat{\xi},|\xi|^{2}) w_{+}(|x|,|\xi|^{2}) [\mathcal{F}f](\xi) \,d\xi$$

$$= f(x) + (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^{n}} a_{-}(\hat{x},\hat{\xi},|\xi|^{2}) w_{+}(|x|,|\xi|^{2}) [\mathcal{F}f](\xi) \,d\xi + [Kf](x).$$

We have  $[(W_- - \mathrm{Id})f](x) = [Zf](x) + [Kf](x)$ , where for  $x \in \mathbb{R}^n$  we have defined

$$[Zf](x) = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} a_-(\hat{x}, \hat{\xi}, |\xi|^2) w_+(|x|, |\xi|^2) [\mathcal{F}f](\xi) \,d\xi.$$

It remains to show that  $Z = T(S - \operatorname{Id})$ . To see this, we recall that for  $\lambda \in \mathbb{R}^+$  and  $\omega \in \mathbb{S}^{n-1}$  the scattering operator satisfies  $[F_0Sf](\lambda,\omega) = S(\lambda)[F_0f](\lambda,\omega)$ , where  $S(\lambda) \in \mathcal{B}(L^2(\mathbb{S}^{n-1}))$  is the scattering matrix at energy  $\lambda$ , and by Lemma 2.4.33 we have for

 $\theta \in \mathbb{S}^{n-1}$  the expression

$$[(S(\lambda) - \operatorname{Id})g](\theta) = 2\pi i (2\pi)^{-\frac{n+1}{2}} \lambda^{\frac{n-1}{4}} \int_{\mathbb{S}^{n-1}} a_{-}(\theta, \omega, \lambda) g(\omega) d\omega.$$

Thus, for  $\lambda > 0$  and  $\theta \in \mathbb{S}^{n-1}$  we have

$$[F_0(S - \operatorname{Id})f](\lambda, \theta) = (S(\lambda) - \operatorname{Id})[F_0f](\lambda, \theta)$$
$$= 2\pi i (2\pi)^{-\frac{n+1}{2}} \lambda^{\frac{n-1}{4}} \int_{\mathbb{S}^{n-1}} a_-(\theta, \omega, \lambda)[F_0f](\lambda, \omega) d\omega,$$

so that

$$\int_{\mathbb{S}^{n-1}} a_{-}(\theta,\omega,\lambda) [F_0 f](\lambda,\omega) d\omega = \frac{1}{2\pi i} (2\pi)^{\frac{n+1}{2}} \lambda^{-\frac{n-1}{4}} [F_0 (S - \mathrm{Id}) f](\lambda,\theta). \tag{4.9}$$

Recalling that  $[F_0 f](\lambda, \omega) = 2^{-\frac{1}{2}} \lambda^{\frac{n-2}{4}} [\mathcal{F} f](\lambda^{\frac{1}{2}} \omega)$  we return to the operator Z and decompose into polar coordinates to find for  $x \in \mathbb{R}^n$  that

$$\begin{split} [Zf](x) &= (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^{+}} \int_{\mathbb{S}^{n-1}} r^{n-1} a_{-}(\hat{x}, \omega, r^{2}) w_{+}(|x|, r^{2}) [\mathcal{F}f](r\omega) \, d\omega \, dr \\ &= 2^{-1} (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^{+}} \int_{\mathbb{S}^{n-1}} \lambda^{\frac{n-2}{2}} a_{-}(\hat{x}, \omega, \lambda) w_{+}(|x|, \lambda) [\mathcal{F}f](\lambda^{\frac{1}{2}}\omega) \, d\omega \, d\lambda \\ &= 2^{-\frac{1}{2}} (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^{+}} \int_{\mathbb{S}^{n-1}} \lambda^{\frac{n-2}{4}} a_{-}(\hat{x}, \omega, \lambda) w_{+}(|x|, \lambda) \left( 2^{-\frac{1}{2}} \lambda^{\frac{n-2}{4}} [\mathcal{F}f](\lambda^{\frac{1}{2}}\omega) \right) \, d\omega \, d\lambda \\ &= 2^{-\frac{1}{2}} (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^{+}} \int_{\mathbb{S}^{n-1}} \lambda^{\frac{n-2}{4}} a_{-}(\hat{x}, \omega, \lambda) w_{+}(|x|, \lambda) [F_{0}f](\lambda, \omega) \, d\omega \, d\lambda. \end{split}$$

Using the relation between  $a_{-}$  and S described in Equation (4.9) we have

$$[Zf](x) = 2^{-\frac{1}{2}} (2\pi i)^{-1} (2\pi)^{\frac{n+1}{2}} (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^+} \lambda^{\frac{n-2}{4}} \lambda^{-\frac{n-1}{4}} w_+(|x|, \lambda) [F_0(S - \mathrm{Id})f](\lambda, \hat{x}) \, \mathrm{d}\lambda$$
$$= 2^{-\frac{1}{2}} (2\pi i)^{-1} (2\pi)^{\frac{1}{2}} \int_{\mathbb{R}^+} \lambda^{\frac{1}{4}} w_+(|x|, \lambda) [F_0(S - \mathrm{Id})f](\lambda, \hat{x}) \, \mathrm{d}\lambda.$$

Noting that

$$[Tf](x) = \frac{\sqrt{\pi}}{2\pi i} \int_{\mathbb{R}^+} \lambda^{\frac{1}{4}} w_+(|x|, \lambda) [F_0 f](\lambda, \hat{x}) \, d\lambda, \tag{4.10}$$

we see that Z = T(S - Id) and we are done.

The operator T satisfies a number of useful properties, summarised in the next theorem.

**Theorem 4.2.2.** The operator  $T: C_c^{\infty}(\mathbb{R}^n) \subset L^2(\mathbb{R}^n) \to L^2(\mathbb{R}^n)$  of Equation (4.6) extends to a bounded operator  $T: L^2(\mathbb{R}^n) \to L^2(\mathbb{R}^n)$  with norm  $||T|| \leq 1$ . Furthermore,

the operator T commutes with the dilation group  $(U_n(t))$  of Definition 2.2.24 and with all rotations.

*Proof.* We first check that T is bounded. Note that  $|c_n|^2 = \frac{1}{8\pi}$ . For  $\tau, \rho \in \mathbb{R}$  we have

$$\int_{\mathbb{R}^+} e^{-ir(\tau-\rho)} dr = \frac{i}{\rho-\tau} + \pi\delta(\tau-\rho)$$
 (4.11)

in the sense of distributions. So for  $f \in \mathcal{S}(\mathbb{R}^n)$  we have

$$\langle Tf, Tf \rangle = \int_{\mathbb{R}^n} \overline{[Tf](x)} [Tf](x) dx$$

$$= 4|c_n|^2 \int_{\mathbb{R}^n} \int_{\mathbb{R}^+} \int_{\mathbb{R}^+} |x|^{-(n-1)} (\tau \rho)^{\frac{n-1}{2}} e^{-i|x|(\tau-\rho)} \overline{[\mathcal{F}f](\tau \hat{x})} [\mathcal{F}f](\rho \hat{x}) d\rho d\tau dx$$

$$= \frac{1}{2\pi} \int_{\mathbb{S}^{n-1}} \int_{\mathbb{R}^+} \int_{\mathbb{R}^+} \int_{\mathbb{R}^+} (\tau \rho)^{\frac{n-1}{2}} e^{-ir(\tau-\rho)} \overline{[\mathcal{F}f](\tau \omega)} [\mathcal{F}f](\rho \omega) d\rho d\tau dr d\omega,$$

where in the last line we have changed to polar coordinates. Using Equation (4.11) we can compute the r integral and obtain

$$\langle Tf, Tf \rangle = \frac{1}{2\pi} \int_{\mathbb{S}^{n-1}} \int_{\mathbb{R}^{+}} \int_{\mathbb{R}^{+}} (\tau \rho)^{\frac{n-1}{2}} \left( \frac{i}{\rho - \tau} + \pi \delta(\tau - \rho) \right) \overline{[\mathcal{F}f](\tau \omega)} [\mathcal{F}f](\rho \omega) \, \mathrm{d}\rho \, \mathrm{d}\tau \, \mathrm{d}\omega$$

$$= \frac{1}{2} \int_{\mathbb{S}^{n-1}} \int_{\mathbb{R}^{+}} \int_{\mathbb{R}^{+}} (\tau \rho)^{\frac{n-1}{2}} \delta(\tau - \rho) \overline{[\mathcal{F}f](\tau \omega)} [\mathcal{F}f](\rho \omega) \, \mathrm{d}\rho \, \mathrm{d}\tau \, \mathrm{d}\omega$$

$$+ \frac{i}{2\pi} \int_{\mathbb{S}^{n-1}} \int_{\mathbb{R}^{+}} \int_{\mathbb{R}^{+}} \frac{(\tau \rho)^{\frac{n-1}{2}}}{\rho - \tau} \overline{[\mathcal{F}f](\tau \omega)} [\mathcal{F}f](\rho \omega) \, \mathrm{d}\rho \, \mathrm{d}\tau \, \mathrm{d}\omega$$

$$:= T_{1} + T_{2}.$$

Both of these integrals are computable. The first is simply

$$T_{1} = \frac{1}{2} \int_{\mathbb{S}^{n-1}} \int_{\mathbb{R}^{+}} \int_{\mathbb{R}^{+}} (\tau \rho)^{\frac{n-1}{2}} \delta(\tau - \rho) \overline{[\mathcal{F}f](\tau \omega)} [\mathcal{F}f](\rho \omega) \, d\rho \, d\tau \, d\omega$$
$$= \frac{1}{2} \int_{\mathbb{S}^{n-1}} \int_{\mathbb{R}^{+}} \rho^{n-1} |[\mathcal{F}f](\rho \omega)|^{2} \, d\rho \, d\omega = \frac{1}{2} \int_{\mathbb{R}^{n}} |[\mathcal{F}f](x)|^{2} \, dx$$
$$= \frac{1}{2} \langle \mathcal{F}f, \mathcal{F}f \rangle.$$

The second requires a little more care. We make the substitution  $\rho = e^t \tau$ ,  $d\rho = e^t \tau dt$  to

obtain

$$T_{2} = \frac{i}{2\pi} \int_{\mathbb{S}^{n-1}} \int_{\mathbb{R}^{+}} \int_{\mathbb{R}^{+}} \frac{(\tau \rho)^{\frac{n-1}{2}}}{\rho - \tau} \overline{[\mathcal{F}f](\tau \omega)} [\mathcal{F}f](\rho \omega) \, d\rho \, d\tau \, d\omega$$

$$= \frac{i}{2\pi} \int_{\mathbb{S}^{n-1}} \int_{\mathbb{R}^{+}} \int_{\mathbb{R}} \frac{\tau^{n-1} e^{\frac{n-1}{2}t}}{e^{t}\tau - \tau} \overline{[\mathcal{F}f](\tau \omega)} [\mathcal{F}f](e^{t}\tau \omega) e^{t}\tau \, dt \, d\tau \, d\omega$$

$$= \frac{i(2\pi)^{\frac{1}{2}}}{4\pi} \int_{\mathbb{S}^{n-1}} \int_{\mathbb{R}^{+}} \tau^{n-1} \overline{[\mathcal{F}f](\tau \omega)} \left( (2\pi)^{-\frac{1}{2}} \int_{\mathbb{R}} \frac{2}{e^{\frac{t}{2}} - e^{-\frac{t}{2}}} e^{\frac{nt}{2}} [\mathcal{F}f](e^{t}\tau \omega) \, dt \right) \, d\tau \, d\omega$$

$$= \frac{i(2\pi)^{\frac{1}{2}}}{4\pi} \int_{\mathbb{S}^{n-1}} \int_{\mathbb{R}^{+}} \tau^{n-1} \overline{[\mathcal{F}f](\tau \omega)} \left( -i(2\pi)^{\frac{1}{2}} [\tanh(\pi D_{n})\mathcal{F}f](\tau \omega) \right) \, d\tau \, d\omega$$

$$= \frac{1}{2} \int_{\mathbb{R}^{n}} \overline{[\mathcal{F}f](x)} [\tanh(\pi D_{n})\mathcal{F}f](x) \, dx$$

$$= \frac{1}{2} \langle \mathcal{F}f, \tanh(\pi D_{n})\mathcal{F}f \rangle,$$

where we have used the functional calculus for  $D_n$  of Lemma 2.2.30. Now note that  $0 \le 1 + \tanh(\pi y) \le 2$  for all y and thus we find

$$\langle Tf, Tf \rangle = T_1 + T_2 = \frac{1}{2} \langle \mathcal{F}f, (\operatorname{Id} + \tanh(\pi D_n)) \mathcal{F}f \rangle \leq \langle \mathcal{F}f, \mathcal{F}f \rangle = \langle f, f \rangle.$$

Since  $\mathcal{S}(\mathbb{R}^n)$  is dense in  $L^2(\mathbb{R}^n)$ , the operator T extends by continuity to a bounded operator  $T:L^2(\mathbb{R}^n)\to L^2(\mathbb{R}^n)$  with  $||T||\leq 1$ . That T commutes with dilations and rotations follows by direct computation.

Since T commutes with the dilation group  $U_n$ , we can consider T as a function of the generator  $D_n$  of dilations on  $L^2(\mathbb{R}^n)$ . Since T commutes with rotations we can decompose  $L^2(\mathbb{R}^n)$  into spherical harmonics and compute the action of T on each harmonic component. The spherical harmonic decomposition demonstrated here was proved in dimension three in [98].

**Definition 4.2.3.** A polynomial P in the variables  $x_1, \ldots, x_n$  which is homogenous of degree  $\ell$  (so  $P(x) = |x|^{\ell} P(\hat{x})$  for all  $x \in \mathbb{R}^n$ ) which satisfies  $\Delta P = 0$  is called a *harmonic polynomial of degree*  $\ell$ . Let  $V_{\ell}$  be the vector space of harmonic polynomials of degree  $\ell$  in n variables. Define the subspace  $H_{\ell}$  of *spherical harmonics* by

$$H_{\ell} = \{ f : \mathbb{S}^{n-1} \to \mathbb{C} : f = P|_{\mathbb{S}^{n-1}} \text{ for some } P \in V_{\ell} \}.$$

Clearly  $H_{\ell} \subset L^2(\mathbb{S}^{n-1})$  and inherits the inner product. The Funcke-Hecke theorem [9, Theorem 9.7.1] allows us to decompose functions on the sphere into spherical harmonics. In particular, the following result shows how to decompose a plane wave into spherical harmonics, which will be particularly useful for decomposing Fourier transforms into spherical harmonic subspaces.

**Lemma 4.2.4** ([9, Lemma 9.10.2]). For any spherical harmonic  $S_{\ell}$  of degree  $\ell$  in n variables and  $\theta \in \mathbb{S}^{n-1}$  we have

$$\int_{\mathbb{S}^{n-1}} e^{-it\langle\theta,\omega\rangle} S_{\ell}(\omega) d\omega = (2\pi)^{\frac{n}{2}} i^{-\ell} S_{\ell}(\theta) t^{-\frac{n-2}{2}} J_{\ell+\frac{n-2}{2}}(t). \tag{4.12}$$

Thus, for a Schwarz class function  $g \in \mathcal{S}(\mathbb{R}^+)$  and  $S_{\ell} \in H_{\ell}$  we define  $f : \mathbb{R}^n \to \mathbb{C}$  by  $f(x) = g(|x|)S_{\ell}(\hat{x})$  and compute for  $\xi \in \mathbb{R}^n$  the Fourier transform

$$[\mathcal{F}f](\xi) = (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} e^{-i\langle x,\xi \rangle} g(|x|) S_{\ell}(\hat{x}) dx$$

$$= (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^+} r^{n-1} g(r) \left( \int_{\mathbb{S}^{n-1}} e^{-ir|\xi|\langle \omega, \hat{\xi} \rangle} S_{\ell}(\omega) d\omega \right) dr$$

$$= i^{-\ell} S_{\ell}(\hat{\xi}) \int_{\mathbb{R}^+} (r|\xi|)^{-\frac{n-2}{2}} r^{n-1} g(r) J_{\ell+\frac{n-2}{2}}(r|\xi|) dr.$$

The Fourier transform leaves the space  $B_{\ell} = C_c^{\infty}(\mathbb{R}^+) \otimes H_{\ell}$  (considered as a subspace of  $C_c^{\infty}(\mathbb{R}^n)$ ) invariant. Restricting  $\mathcal{F}$  to the harmonic subspace  $B_{\ell}$  defines the operator  $\mathcal{F}_{\ell}: C_c^{\infty}(\mathbb{R}^+) \subset L^2(\mathbb{R}^+, r^{n-1} dr) \to C_c^{\infty}(\mathbb{R}^+) \subset L^2(\mathbb{R}^+, r^{n-1} dr)$  for  $\eta \in \mathbb{R}^+$  and  $g \in L^2(\mathbb{R}^+, r^{n-1} dr)$  by

$$[\mathcal{F}_{\ell}g](\eta) = i^{-\ell} \int_{\mathbb{R}^+} (r\eta)^{-\frac{n-2}{2}} g(r) J_{\ell+\frac{n-2}{2}}(r\eta) r^{n-1} dr.$$
 (4.13)

The operator  $\mathcal{F}_{\ell}$  is related to the Hankel transform of Definition 2.2.10 and extends to a unitary from  $\mathcal{H}_r = L^2(\mathbb{R}^+, r^{n-1} dr)$  to itself, a fact which follows from the Parseval relation for the Hankel transform [113, Theorem I].

Since T commutes with any rotation, T leaves each subspace  $B_{\ell}$  invariant, so restricts to an operator  $T_{\ell} \otimes \mathrm{Id}$  on each  $B_{\ell}$ . Recall that T is defined for  $f \in C_c^{\infty}(\mathbb{R}^n)$  by

$$[Tf](x) = 2(2\pi)^{-\frac{n}{2}} c_n |x|^{-\frac{n-1}{2}} \int_{\mathbb{R}^+} \tau^{\frac{n-1}{2}} e^{i\tau|x|} [\mathcal{F}f](\tau \hat{x}) d\tau.$$

Thus we can define an operator  $T_{\ell}: C_c^{\infty}(\mathbb{R}^+, r^{n-1} dr) \to C_c^{\infty}(\mathbb{R}^+, r^{n-1} dr)$  for  $\eta \in \mathbb{R}^+$  by

$$[T_{\ell}g](\eta) = 2(2\pi)^{-\frac{n}{2}} c_n \eta^{-\frac{n-1}{2}} \int_{\mathbb{R}^+} \tau^{\frac{n-1}{2}} e^{i\tau\eta} [\mathcal{F}_{\ell}g](\tau) d\tau.$$
 (4.14)

We recall from Lemma 2.2.21 formula for functions of dilation on the half-line,

$$[\varphi(D_+)g](\eta) = (2\pi)^{-\frac{1}{2}} \int_{\mathbb{R}} [\mathcal{F}^*\varphi](t) e^{\frac{t}{2}} g(e^t \eta) dt,$$

for  $g \in C_c^{\infty}(\mathbb{R}^+)$  and  $\eta \in \mathbb{R}^+$ . We shall recognise each  $T_{\ell}$  as a function  $\varphi_{\ell}(D_+)$ . The following is a generalisation of [98, Proposition 2].

**Proposition 4.2.5.** For each  $\ell$  the operator  $T_{\ell}$  extends continuously to the bounded oper-

ator  $\varphi_{\ell}(D_n)$  on  $\mathcal{H}_r = L^2(\mathbb{R}^+, r^{n-1} dr)$ , where  $\varphi_{\ell} \in C(\mathbb{R})$  has limits at  $\pm \infty$  and is defined for  $y \in \mathbb{R}^+$  by by

$$\varphi_{\ell}(y) = i^{-\ell} e^{-i\frac{n-3}{4}\pi} \frac{1}{2} \frac{\Gamma\left(\frac{1}{2}\left(\ell + \frac{n}{2} + iy\right)\right)}{\Gamma\left(\frac{1}{2}\left(\ell + \frac{n}{2} - iy\right)\right)} \frac{\Gamma\left(\frac{1}{2}\left(\frac{3}{2} - iy\right)\right)}{\Gamma\left(\frac{1}{2}\left(\frac{3}{2} + iy\right)\right)} \left(1 + \tanh\left(\pi y\right) - i\cosh\left(\pi y\right)^{-1}\right).$$

$$(4.15)$$

*Proof.* We recall from Equation (4.6) that for  $f \in C_c^{\infty}(\mathbb{R}^n)$  and  $x \in \mathbb{R}^n$  the operator T is given by

$$[Tf](x) = k_n |x|^{-\frac{n-1}{2}} \int_{\mathbb{R}^+} \tau^{\frac{n-1}{2}} e^{i\tau |x|} [\mathcal{F}f](\tau \hat{x}) d\tau,$$

where for convenience we have let  $k_n = 2c_n$ . On the space  $\mathcal{H}_{\ell} = \{f \in C_c^{\infty}(\mathbb{R}^n) : f(x) = g(|x|)Y_{\ell m}(\hat{x}) \text{ for some } g \in C_c^{\infty}(\mathbb{R}^+), Y_{\ell,m} \in P_{\ell}\}$  we have that  $T|_{\mathcal{H}_{\ell}} = T_{\ell} \otimes \operatorname{Id}_{P_{\ell}}$  where  $T_{\ell} : L^2(\mathbb{R}^+, r^{n-1} dr) \to L^2(\mathbb{R}^+, r^{n-1} dr)$  is given for  $g \in C_c^{\infty}(\mathbb{R}^+)$  and  $r \in \mathbb{R}^+$  by

$$[T_{\ell}g](r) = k_n r^{-\frac{n-1}{2}} \int_{\mathbb{R}^+} \tau^{\frac{n-1}{2}} e^{i\tau r} [\mathcal{F}_{\ell}g](\tau) d\tau.$$

We recall the Fourier transform restricted to the space  $\mathcal{H}_{\ell}$  has the form  $\mathcal{F}|_{\mathcal{H}_{\ell}} = \mathcal{F}_{\ell} \otimes \operatorname{Id}_{P_{\ell}}$ , where  $\mathcal{F}_{\ell} : L^{2}(\mathbb{R}^{+}, r^{n-1} dr) \to L^{2}(\mathbb{R}^{+}, r^{n-1} dr)$  is given by Equation (4.13). Hence we can write  $T_{\ell}$  explicitly as

$$[T_{\ell}g](r) = k_n i^{\ell} r^{-\frac{n-1}{2}} \int_{\mathbb{R}^+} \int_{\mathbb{R}^+} \tau^{\frac{n-1}{2}} e^{i\tau r} (\tau \rho)^{-\frac{n-2}{2}} \rho^{n-1} g(\rho) J_{\ell+\frac{n-2}{2}}(\tau \rho) d\rho d\tau$$
$$= k_n i^{\ell} r^{-\frac{n-1}{2}} \int_{\mathbb{R}^+} \int_{\mathbb{R}^+} \tau^{\frac{1}{2}} e^{i\tau r} \rho^{\frac{n}{2}} g(\rho) J_{\ell+\frac{n-2}{2}}(\tau \rho) d\rho d\tau.$$

and so we may use the substitution  $\eta = re^t$  to write

$$[T_{\ell}g](r) = k_n i^{\ell} r^{-\frac{n-1}{2}} \int_{\mathbb{R}^+} \int_{\mathbb{R}^+} \tau^{\frac{1}{2}} e^{i\tau r} \eta^{\frac{n}{2}} g(\eta) J_{\ell+\frac{n-2}{2}}(\tau \eta) d\eta d\tau$$
$$= k_n i^{\ell} r^{\frac{3}{2}} \int_{\mathbb{R}} e^{\frac{nt}{2}} g(e^t r) e^t \left( \int_{\mathbb{R}^+} \tau^{\frac{1}{2}} e^{i\tau r} J_{\ell+\frac{n-2}{2}}(e^t r \tau) d\tau \right) dt.$$

Define the distribution  $\psi_{\ell}$  by

$$\psi_{\ell}(t) = k_n i^{\ell} (2\pi)^{\frac{1}{2}} r^{\frac{3}{2}} e^{-t} \int_{\mathbb{R}^+} \tau^{\frac{1}{2}} e^{i\tau r} J_{\ell + \frac{n-2}{2}} (e^{-t} r \tau) d\tau.$$

We now find the Fourier transform  $\varphi_{\ell} = \mathcal{F}\psi_{\ell}$  and we may write

$$[T_{\ell}g](r) = (2\pi)^{-\frac{1}{2}} \int_{\mathbb{R}} [\mathcal{F}^*\varphi_{\ell}](-t) e^{\frac{nt}{2}} g(e^t r) dt.$$

Thus we compute the Fourier transform as

$$\begin{split} \varphi_{\ell}(y) &= [\mathcal{F}\psi_{\ell}](t) = k_{n}i^{\ell}r^{\frac{3}{2}} \int_{\mathbb{R}} \mathrm{e}^{-iyt}\psi(t)\mathrm{d}t \\ &= k_{n}i^{\ell}r^{\frac{3}{2}} \int_{\mathbb{R}} \int_{\mathbb{R}^{+}} \mathrm{e}^{-iyt}\mathrm{e}^{-t}\tau^{\frac{1}{2}}\mathrm{e}^{i\tau r}J_{\ell+\frac{n-2}{2}}(\mathrm{e}^{-t}r\tau)\,\mathrm{d}\tau\,\mathrm{d}t \\ &= \mathrm{e}^{-i\frac{\pi}{4}}k_{n}i^{\ell}r^{\frac{3}{2}} \int_{\mathbb{R}} \int_{\mathbb{R}^{+}} \mathrm{e}^{-iyt}\mathrm{e}^{-t}\tau^{\frac{1}{2}}(\tau r)^{\frac{1}{2}}K_{\frac{1}{2}}(-i\tau r)J_{\ell+\frac{n-2}{2}}(\mathrm{e}^{-t}r\tau)\,\mathrm{d}\tau\,\mathrm{d}t, \end{split}$$

where we have used the relation  $e^{-w} = \sqrt{\frac{2}{\pi}} w^{\frac{1}{2}} K_{\frac{1}{2}}(w)$  (here  $K_{\nu}$  denotes a modified Bessel function). Making the substitution  $v = e^{-t}r\tau$ ,  $d\tau = e^{t}r^{-1}dv$  we obtain

$$\begin{split} \varphi_{\ell}(y) &= \mathrm{e}^{-i\frac{\pi}{4}} k_n i^{\ell} \sqrt{\frac{2}{\pi}} r^{\frac{3}{2}} \int_{\mathbb{R}} \int_{\mathbb{R}^{+}} \mathrm{e}^{-iyt} \mathrm{e}^{-t} \tau^{\frac{1}{2}} (\tau r)^{\frac{1}{2}} K_{\frac{1}{2}} (-i\tau r) J_{\ell + \frac{n-2}{2}} (\mathrm{e}^{-t} r \tau) \, \mathrm{d}\tau \, \mathrm{d}t \\ &= \mathrm{e}^{-i\frac{\pi}{4}} k_n i^{\ell} \sqrt{\frac{2}{\pi}} r^{\frac{3}{2}} \int_{\mathbb{R}} \int_{\mathbb{R}^{+}} \mathrm{e}^{-iyt} \mathrm{e}^{-t} v^{\frac{1}{2}} \mathrm{e}^{\frac{t}{2}} r^{-\frac{1}{2}} v^{\frac{1}{2}} \mathrm{e}^{\frac{t}{2}} K_{\frac{1}{2}} (-iv\mathrm{e}^{t}) J_{\ell + \frac{n-2}{2}} (v) \mathrm{e}^{t} r^{-1} \, \mathrm{d}v \, \mathrm{d}t \\ &= \mathrm{e}^{-i\frac{\pi}{4}} k_n i^{\ell} \sqrt{\frac{2}{\pi}} \int_{\mathbb{R}} \int_{\mathbb{R}^{+}} \mathrm{e}^{-iyt} \mathrm{e}^{t} v K_{\frac{1}{2}} (-iv\mathrm{e}^{t}) J_{\ell + \frac{n-2}{2}} (v) \, \mathrm{d}v \, \mathrm{d}t. \end{split}$$

We now make the substitution  $s = ve^t$ ,  $dt = s^{-1}ds$  to see that

$$\varphi_{\ell}(y) = e^{-i\frac{\pi}{4}} k_n i^{\ell} \sqrt{\frac{2}{\pi}} \int_{\mathbb{R}} \int_{\mathbb{R}^+} e^{-iyt} e^{t} v K_{\frac{1}{2}}(-ive^{t}) J_{\ell+\frac{n-2}{2}}(v) dv dt 
= e^{-i\frac{\pi}{4}} k_n i^{\ell} \sqrt{\frac{2}{\pi}} \int_{\mathbb{R}^+} \int_{\mathbb{R}^+} s^{-it} v^{iy} s v^{-1} v K_{\frac{1}{2}}(-is) J_{\ell+\frac{n-2}{2}}(v) s^{-1} dv ds 
= e^{-i\frac{\pi}{4}} k_n i^{\ell} \sqrt{\frac{2}{\pi}} \left( \int_{\mathbb{R}^+} s^{-iy} K_{\frac{1}{2}}(-is) ds \right) \left( \int_{\mathbb{R}^+} v^{iy} J_{\ell+\frac{n-2}{2}}(v) dv \right).$$

By [126, p. 10.1] we have

$$\int_{\mathbb{R}^+} v^{iy} J_{\ell+\frac{n-2}{2}}(v) \mathrm{d}v = 2^{iy} \frac{\Gamma\left(\frac{1}{2}\left(\ell+\frac{n}{2}+iy\right)\right)}{\Gamma\left(\frac{1}{2}\left(\ell+\frac{n}{2}-iy\right)\right)}.$$

Using [126, p. 11.1] we find

$$\int_{\mathbb{R}^{+}} s^{-iy} K_{\frac{1}{2}}(-is) ds = 2^{-iy-1} (-i)^{-iy+1} \Gamma\left(\frac{1}{2} \left(\frac{3}{2} - iy\right)\right) \Gamma\left(\frac{1}{2} \left(\frac{1}{2} - iy\right)\right).$$

Then  $\varphi_{\ell}$  is given by

$$\varphi_{\ell}(y) = -i\mathrm{e}^{-i\frac{\pi}{4}} \sqrt{\frac{2}{\pi}} k_n i^{\ell} 2^{-1} \mathrm{e}^{\frac{\pi y}{2}} \frac{\Gamma\left(\frac{1}{2}\left(\ell + \frac{n}{2} + iy\right)\right)}{\Gamma\left(\frac{1}{2}\left(\ell + \frac{n}{2} - iy\right)\right)} \Gamma\left(\frac{1}{2}\left(\frac{3}{2} - iy\right)\right) \Gamma\left(\frac{1}{2}\left(\frac{1}{2} - iy\right)\right).$$

A quick calculation shows that

$$\frac{e^{\frac{i\pi}{4}}e^{\frac{\pi y}{2}}}{e^{\frac{\pi i}{4}}e^{\frac{\pi y}{2}} - e^{-\frac{\pi i}{4}}e^{-\frac{\pi y}{2}}} = \frac{e^{\frac{\pi y}{2}}}{e^{\frac{\pi y}{2}} + ie^{-\frac{\pi y}{2}}} = \frac{1}{2}\frac{\cosh(\pi y) + \sinh(\pi y) - i}{\cosh(\pi y)}$$
$$= \frac{1}{2}\left(1 + \tanh(\pi y) - i\cosh(\pi y)^{-1}\right).$$

Using the relation  $\Gamma(z)\Gamma(1-z) = \frac{\pi}{\sin{(\pi z)}}$  with  $z = \frac{1}{2}\left(\frac{1}{2} - iy\right)$  we find

$$e^{-\frac{i\pi}{4}} e^{\frac{\pi y}{2}} \Gamma\left(\frac{1}{2}\left(\frac{1}{2} - iy\right)\right) = \frac{\pi}{\Gamma\left(1 - \frac{1}{2}\left(\frac{1}{2} - iy\right)\right) \sin\left(\frac{1}{2}\left(\frac{1}{2} - iy\right)\pi\right)}$$

$$= e^{-\frac{i\pi}{4}} e^{\frac{\pi y}{2}} \frac{2i\pi}{\Gamma\left(\frac{1}{2}\left(\frac{3}{2} + iy\right)\right) \left(e^{\frac{i\pi}{4}} e^{\frac{\pi y}{2}} - e^{-i\frac{\pi}{4}} e^{-\frac{\pi y}{2}}\right)}$$

$$= \frac{i\pi}{\Gamma\left(\frac{1}{2}\left(\frac{3}{2} + iy\right)\right)} \left(1 + \tanh(\pi y) - i\cosh(\pi y)^{-1}\right).$$

Thus,

$$\varphi_{\ell}(y) = \pi k_n i^{\ell} \sqrt{\frac{2}{\pi}} \frac{1}{2} \frac{\Gamma\left(\frac{1}{2}\left(\ell + \frac{n}{2} + iy\right)\right)}{\Gamma\left(\frac{1}{2}\left(\ell + \frac{n}{2} - iy\right)\right)} \frac{\Gamma\left(\frac{1}{2}\left(\frac{3}{2} - iy\right)\right)}{\Gamma\left(\frac{1}{2}\left(\frac{3}{2} + iy\right)\right)} \left(1 + \tanh\left(\pi y\right) - i\cosh\left(\pi y\right)^{-1}\right).$$

Recall that the constant  $k_n = 2c_n$  is given by  $k_n = -i(2\pi)^{-\frac{1}{2}} e^{-i\frac{n-3}{4}\pi}$  and so we obtain the expression

$$\varphi_{\ell}(y) = i^{\ell-1} e^{-i\frac{n-3}{4}\pi} \frac{1}{2} \frac{\Gamma\left(\frac{1}{2}\left(\ell + \frac{n}{2} + iy\right)\right)}{\Gamma\left(\frac{1}{2}\left(\ell + \frac{n}{2} - iy\right)\right)} \frac{\Gamma\left(\frac{1}{2}\left(\frac{3}{2} - iy\right)\right)}{\Gamma\left(\frac{1}{2}\left(\frac{3}{2} + iy\right)\right)} \left(1 + \tanh\left(\pi y\right) - i\cosh\left(\pi y\right)^{-1}\right).$$

The limits at  $\pm \infty$  follow from the asymptotics of the Gamma function (see [1]).

The result of Proposition 4.2.5 agrees with the result obtained in [98, Proposition 2] in dimension three and that the operator  $\varphi_0(D_n) = \frac{1}{2}(\operatorname{Id} + \tanh(\pi D_n) - i\cosh(\pi D_n)^{-1})$  is precisely of the form of the operator of Equation (4.1). To arrive at a simpler formula without the ratio of Gamma functions, we provide a straightforward generalisation of [151, Theorem 4.1] to operator-valued functions.

**Lemma 4.2.6.** Suppose that  $f \in L^2(\mathbb{R}) \cap C(\mathbb{R})$  and  $g \in C(\mathbb{R}, \mathcal{K}(\mathcal{P}))$  is such that  $g(\pm \infty) = 0$ . Let L denote the (densely defined) operator of multiplication by the variable in  $L^2(\mathbb{R}^+)$ . Then the operator  $(f(D_+) \otimes \operatorname{Id})g(\ln(L))$  defines a compact operator on  $\mathcal{H}_{spec}$ . Similarly if  $h \in C(\mathbb{R}^+, \mathcal{K}(\mathcal{P}))$  is such that  $h(0) = h(\infty) = 0$ , then  $(f(D_+) \otimes \operatorname{Id})h(L)$  defines a compact operator on  $\mathcal{H}_{spec}$ .

*Proof.* The algebraic tensor product  $C_0(\mathbb{R}) \odot \mathcal{K}(\mathcal{P})$  is dense in  $C_0(\mathbb{R}, \mathcal{K}(\mathcal{P}))$  by [133, Theorem 1.15], when  $C_0(\mathbb{R}, \mathcal{K}(\mathcal{P}))$  is equipped with the uniform topology. Thus it suffices

to prove that for  $a \in C_0(\mathbb{R})$  and  $b \in \mathcal{K}(\mathcal{P})$  we have

$$(f(D_+) \otimes \operatorname{Id})(a(\ln(L)) \otimes b) = f(D_+)a(\ln(L)) \otimes b$$

is compact. This follows from [151, Theorem 4.1] after a few steps. First we let  $u_m \in C_c^{\infty}(\mathbb{R})$  be an approximate unit for  $C_0(\mathbb{R})$ . Then  $u_m f$  and  $u_m a$  are  $L^2$ -functions and so

$$(u_m f)(D_+)(u_m a)(\ln(L)) \otimes b$$

is compact by [151, Theorem 4.1], and since  $D_+$  and  $\frac{1}{2} \ln(L)$  are canonically conjugate. Now because f and a vanish at  $\pm \infty$  we readily see that  $(u_m f)(D_+) \to f(D_+)$  in operator norm, and similarly for  $(u_m a)(\ln(L))$ . As the compacts are norm closed, the limit is compact. The final claim follows by writing  $h = g \circ (\ln)$ .

**Lemma 4.2.7.** Let  $n \geq 4$  and define  $\varphi : \mathbb{R} \to \mathbb{C}$  by  $\varphi(y) = \frac{1}{2}(1 + \tanh(\pi y) - i\cosh(\pi y)^{-1})$ . Then  $(T - \varphi(D_n))(S - \mathrm{Id})$  is compact.

Proof. The operator S satisfies  $S(0) = \operatorname{Id}$  and the functions  $\varphi_{\ell}$  of Proposition 4.2.5 satisfy  $\varphi_{\ell}(\infty) = 1$ ,  $\varphi_{\ell}(-\infty) = 0$ . Thus the difference  $T - \varphi(D_n)$  is a function of  $\ell$  and  $D_n$  vanishing at  $\pm \infty$  of  $\sigma(D_n)$  for each  $\ell$ . We also have the limit  $\lim_{\lambda \to \infty} S(\lambda) = \operatorname{Id}$  in  $\mathcal{B}(\mathcal{P})$ , a fact which will be proved in Corollary 4.3.10. The operators  $D_n$  and  $\frac{1}{2} \ln (H_0)$  are canonically conjugate, so by Lemma 4.2.6 the operator  $(T - \varphi(D_n))(S - \operatorname{Id})$  is compact.

The term  $\cosh (\pi D_n)^{-1} (S - \operatorname{Id})$  will give rise to a compact operator by an identical argument, however we shall keep it in our formulas for reasons which will become apparent in Section 5.2.3. We also note that Lemma 4.2.7 is not true in dimension n = 3 in the presence of resonances, since the scattering operator does not satisfy  $S(0) = \operatorname{Id}$ .

Whilst the above arguments give the correct formula for the wave operator, it is difficult to prove that the operator K defined by Equation (4.7) is compact directly. To do so would require all the subtleties of the behaviour of the perturbed resolvent R(z) near z = 0. Sadly, we have not found a direct proof of the compactness of the operator K defined by Equation (4.7) directly from its definition as the remainder in the construction of generalised plane waves.

In the next section we offer a different proof which starts from the expansion of the resolvent near zero. This method was first used in dimension three in [141], in dimension two in the absence of resonances in [140] and in dimension two with (some) resonances in [142]. By expanding and modifying these arguments we will obtain the formula for the wave function in all cases where S(0) = Id. As a consequence of Theorem 4.0.1, the operator K defined by Equation (4.7) is compact.

Remark 4.2.8. The proof of Proposition 4.2.5 determines the dominant behaviour of T, and one may use the method of stationary phase [56, Theorem 3.38] to obtain a similar expression to Equation (4.1).

# 4.3 The form of the wave operator

In this section we use the resolvent expansions of Chapter 3 to provide a proof of Theorem 4.0.1. To make use of the results of Chapter 3 we require some additional assumptions on the potential, which we summarise below.

**Assumption 4.3.1.** In dimension n we fix  $\rho$  and t such that

- 1. if n = 1 then  $\rho > \frac{5}{2}$ ;
- 2. if n = 2 then  $\rho > 11$ ;
- 3. if n = 3 then  $\rho > 5$  and  $t \in (\frac{5}{2}, \rho \frac{5}{2})$ ;
- 4. if n = 4 then  $\rho > 12$  and  $t \in (6, \rho 6)$ ; and
- 5. if  $n \ge 5$  then  $\rho > \frac{3n+4}{2}$  and  $t \in \left(\frac{n}{2}, \rho \frac{n}{2}\right)$ .

We assume that  $|V(x)| \leq C(1+|x|)^{-\rho}$  for almost all  $x \in \mathbb{R}^n$ .

We separate out dimension n = 1 and n = 2 from the higher dimensions, since their proofs use different techniques and have appeared in the literature [97, 140, 142]. The case n = 3 has also been demonstrated in [141], however the proof technique we use follows closely that of [141] and so this case fits neatly into the narrative with the  $n \ge 4$  cases.

### **4.3.1** Dimension n = 1

As we have seen throughout this thesis, dimension n = 1 produces atypical behaviour. In [97, Proposition 4] it was shown that in dimension n = 1 the operator K defined by Equation (4.7) is compact (Hilbert-Schmidt even) by directly using subtleties related to Jost functions to obtain estimates on the integral kernel, a method which is only available in dimension n = 1. The result is the following [97, Proposition 5].

**Theorem 4.3.2.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > \frac{5}{2}$  and let  $A: L^2(\mathbb{R}) \to L^2(\mathbb{R})$  be defined by [Af](x) = f(-x). The wave operator  $W_-$  is of the form

$$W_{-} = \operatorname{Id} + \frac{1}{2} \left( \operatorname{Id} + \tanh (\pi D_{1}) + i A \cosh (\pi D_{1}) \right) (S - \operatorname{Id}) + K, \tag{4.16}$$

where K is a compact operator.

The presence of the operator A is an artefact of dimension n = 1 only and cannot be removed, however it causes no issues in our later analysis.

### **4.3.2** Dimension n = 2

We first recall the known form of the wave operator in two dimensions in the presence of s-resonances. We then massage the formula from Theorem 4.3.3 to obtain the statement in Theorem 4.0.1. The following is [142, Theorem 1.3].

**Theorem 4.3.3.** Let n=2 and V satisfy Assumption (2.2.14) for some  $\rho > 11$  and suppose that there are no p-resonances. Then there exist two continuous compact operator valued functions  $\eta, \tilde{\eta} : \mathbb{R}^+ \to \mathcal{K}(L^2(\mathbb{S}^1))$  vanishing at 0 and  $\infty$  and satisfying the relation  $\eta(H_0) + \tilde{\eta}(H_0) = S - \operatorname{Id}$ , and such that the equality

$$W_{-} = \operatorname{Id} + \frac{1}{2} \left( \operatorname{Id} + \tanh \left( \frac{\pi D_2}{2} \right) \right) \eta(H_0) + \frac{1}{2} \left( \operatorname{Id} + \tanh \left( \pi D_2 \right) - i \cosh \left( \pi D_2 \right)^{-1} \right) \tilde{\eta}(H_0)$$

holds modulo compacts.

With the help of a technical result, we can show rather simply that Theorem 4.0.1 holds in dimension n = 2.

**Lemma 4.3.4.** Let n = 2 and V satisfy Assumption (2.2.14) for some  $\rho > 11$  and suppose that there are no p-resonances. Then  $W_-$  satisfies Equation (4.1).

*Proof.* Consider the difference

$$X = F_0 \left( \frac{1}{2} \left( \operatorname{Id} + \tanh \left( \frac{\pi D_2}{2} \right) \right) \eta(H_0) - \frac{1}{2} \left( \operatorname{Id} + \tanh \left( \pi D_2 \right) - i \cosh \left( \pi D_2 \right)^{-1} \right) \eta(H_0) \right) F_0^*$$

$$= \frac{1}{2} \left( \operatorname{Id} + \tanh \left( \pi D_+ \right) - \operatorname{Id} - \tanh \left( 2\pi D_+ \right) + i \cosh \left( 2\pi D_+ \right)^{-1} \right) \eta(L)$$

$$=: Y(D_+) \eta(L).$$

The function  $\eta$  vanishes at 0 and  $\infty$  and the function Y is continuous and square integrable. Thus an application of Lemma 4.2.6 shows X is compact. Hence  $W_-$  has the form given in Equation (4.1).

### 4.3.3 Dimension $n \geq 3$

#### 4.3.3.1 Setting up the proof

We will now provide a number of preparatory results before proceeding to the proof of Theorem 4.0.1 for  $n \geq 3$ . We note that the case n = 3 has been proved in [141], however fits neatly into this section.

The proof is broken up into a number of steps, whose ultimate goal is to factorise (in the spectral representation) the wave operator into the composition of several operators, whose mapping properties as operators between various weighted Sobolev spaces can be analysed. The properties of these factor operators depend heavily on their low energy behaviour and we use the resolvent expansions of Chapter 3 to determine this behaviour.

**Definition 4.3.5.** Define, for  $\varepsilon > 0$  and  $\lambda \in \mathbb{R}$ , the operator  $\varphi_{\varepsilon}(H_0 - \lambda)$  on  $\mathcal{H}$  by  $\varphi_{\varepsilon}(H_0 - \lambda) = \frac{\varepsilon}{\pi} R_0(\lambda \mp i\varepsilon) R_0(\lambda \pm i\varepsilon)$ . Similarly for  $M = F_0 H_0 F_0^*$ , the operator of multiplication by the spectral variable, set  $\varphi_{\varepsilon}(M - \lambda) = \frac{\varepsilon}{\pi} (M - (\lambda \mp i\varepsilon))^{-1} (M - (\lambda \pm i\varepsilon))^{-1}$ .

By [165, Section 1.4], the limits  $\lim_{\varepsilon \to 0} \langle \varphi_{\varepsilon}(H_0 - \lambda) f, g \rangle$  exist for almost every  $\lambda \in \mathbb{R}$  and  $f, g \in L^2(\mathbb{R}^n)$ . Moreover the limit satisfies Stone's formula (see Corollary 2.1.13)

$$\langle f, g \rangle = \int_{\mathbb{R}} \left( \lim_{\varepsilon \to 0} \left\langle \varphi_{\varepsilon}(H_0 - \lambda) f, g \right\rangle \right) d\lambda.$$

Using the symmetrised resolvent identity and the stationary formula for the wave operator in Equation (2.69) we can then show that

$$\langle (W_{\pm} - \operatorname{Id}) f, g \rangle = - \int_{\mathbb{R}} \left( \lim_{\varepsilon \to 0} \left\langle \varphi_{\varepsilon}(H_0 - \lambda) f, v(U + vR_0(\lambda \mp i\varepsilon)v)^{-1} vR_0(\lambda \pm i\varepsilon)g \right\rangle \right) d\lambda.$$

We now derive formulas for the wave operators in the spectral representation of  $H_0$ , that is,  $F_0(W_- - \operatorname{Id})F_0^*$ . Let  $M = F_0H_0F_0^*$  be the operator of multiplication by the spectral variable. For suitable  $f, g \in \mathcal{H}_{spec}$  we compute that

$$- \langle F_0(W_{\pm} - \operatorname{Id})F_0^*f, g \rangle_{\mathcal{H}_{spec}}$$

$$= \int_{\mathbb{R}} \left( \lim_{\varepsilon \to 0} \left\langle v(U + vR_0(\lambda \mp i\varepsilon)v)^{-1}vF_0^*\varphi_{\varepsilon}(M - \lambda)f, F_0^*(M - \lambda \mp i\varepsilon)^{-1}g \right\rangle_{\mathcal{H}} \right) d\lambda$$

$$= \int_{\mathbb{R}} \left( \lim_{\varepsilon \to 0} \int_0^{\infty} \left\langle \left[ F_0v(U + vR_0(\lambda \mp i\varepsilon)v)^{-1}vF_0^*\varphi_{\varepsilon}(M - \lambda)f \right](\mu, \cdot), \frac{g(\mu, \cdot)}{\mu - \lambda \mp i\varepsilon} \right\rangle_{\mathcal{P}} d\mu \right) d\lambda.$$

With  $M(k) = U + vR_0(z)v$  for  $z \in \mathbb{C}\backslash\mathbb{R}$  and  $M_{\varepsilon}(\lambda) = U + vR_0(\lambda \mp i\varepsilon)v$  we have

$$\langle F_0(W_{\pm} - \operatorname{Id})F_0^*f, g \rangle_{\mathcal{H}_{spec}}$$

$$= -\int_{\mathbb{R}} \left( \lim_{\varepsilon \to 0} \int_0^{\infty} \langle \left[ F_0 M_{\varepsilon}(\lambda)^{-1} F_0^* \varphi_{\varepsilon}(M - \lambda) f \right] (\mu, \cdot), (\mu - \lambda \mp i\varepsilon)^{-1} g(\mu, \cdot) \rangle_{\mathcal{P}} d\mu \right) d\lambda.$$
(4.17)

To interchange the limit as  $\varepsilon \to 0$  and the integral over  $\mu$  in Equation (4.17), which we do in subsection 4.3.3.3, we first analyse  $F_0 v M_{\varepsilon}(\lambda)^{-1} v F_0^*$ .

### 4.3.3.2 Analysing the diagonalisation maps

Recall that the trace operator (see Section 2.4) extends to an element of  $\mathcal{B}(H^{s,t},\mathcal{P})$  for each  $s > \frac{1}{2}$  and  $t \in \mathbb{R}$ , and that  $\lambda \mapsto \gamma(\lambda) \in \mathcal{B}(H^{s,t},\mathcal{P})$  is continuous [86, Section 3]. As a consequence the operator  $\Gamma_0(\lambda) : \mathcal{S}(\mathbb{R}^n) \to \mathcal{P}$  extends to an element of  $\mathcal{B}(H^{s,t},\mathcal{P})$  for each  $s \in \mathbb{R}$  and  $t > \frac{1}{2}$ , as well as  $\lambda \mapsto \Gamma_0(\lambda) \in \mathcal{B}(H^{s,t},\mathcal{P})$  is continuous.

We need stronger continuity and boundedness results for  $\Gamma_0$ .

**Lemma 4.3.6.** Let  $n \geq 3$ ,  $s \geq 0$  and  $t > \frac{n}{2}$ . Then  $G_{\pm} : (0, \infty) \to \mathcal{B}(H^{s,t}, \mathcal{P})$  defined by  $G_{\pm}(\lambda) = \lambda^{\pm \frac{1}{4}} \Gamma_0(\lambda)$  are continuous and bounded.

*Proof.* Continuity is immediate from the previous paragraph. For boundedness, it is sufficient to check boundedness of  $G_{-}$  in a neighbourhood of 0 and  $G_{+}$  in a neighbourhood of  $\infty$ .

For the first bound, we analyse the asymptotic development of  $\gamma(\lambda^{\frac{1}{2}})$  as  $\lambda \to 0$ . Recall that for  $f \in C_c^{\infty}(\mathbb{R}^n)$  and  $\omega \in \mathbb{S}^{n-1}$  we have

$$[\gamma(\lambda^{\frac{1}{2}})\mathcal{F}f](\omega) = \sum_{j=0}^{K} (i\lambda^{\frac{1}{2}})^{j} [\gamma_{j}f](\omega) + O\left(\lambda^{\frac{K+1}{2}}\right).$$

By Lemma 4.1.1 we have  $\gamma_j \in \mathcal{B}(H^{0,t},\mathcal{P})$  for  $t > j + \frac{n}{2}$ . From this expansion we see that for  $t > j + \frac{n}{2}$  and for any  $\alpha \leq \frac{n-2}{4}$  we have

$$\lambda^{-\alpha} \left| \left| \Gamma_0(\lambda) \right| \right| = \lambda^{\frac{n-2}{4} - \alpha} 2^{-\frac{1}{2}} \left| \left| \gamma(\lambda^{\frac{1}{2}}) \right| \right| \tag{4.18}$$

is bounded in a neighbourhood of zero. Since  $n \geq 3$ , taking  $\alpha = \frac{1}{4}$  gives immediately the statement for  $G_{-}$ .

For the second statement, we refer to [164, Theorem 1.1.4] for the estimate

$$r^{n-1} \int_{\mathbb{S}^{n-1}} |u(r\omega)|^2 d\omega \le C ||u||_{H^{t,0}}^2, \tag{4.19}$$

valid for  $t > \frac{1}{2}$ , from which we obtain (with  $r = \lambda^{\frac{1}{2}}$  and  $u = \mathcal{F}f$ ) that

$$||\Gamma_{0}(\lambda)f||_{\mathcal{P}}^{2} = \int_{\mathbb{S}^{n-1}} 2^{-1} \lambda^{\frac{n-2}{2}} |[\mathcal{F}f](\lambda^{\frac{1}{2}}\omega)|^{2} d\omega = 2^{-1} \lambda^{-\frac{1}{2}} \int_{\mathbb{S}^{n-1}} \lambda^{\frac{n-1}{2}} |[\mathcal{F}f](\lambda^{\frac{1}{2}}\omega)|^{2} d\omega$$

$$\leq 2^{-1} \lambda^{-\frac{1}{2}} C ||\mathcal{F}f||_{H^{1,0}}^{2} = 2^{-1} \lambda^{-\frac{1}{2}} C ||f||_{H^{0,t}}^{2}. \tag{4.20}$$

This gives us the estimate  $\lambda^{\frac{1}{4}} ||\Gamma_0(\lambda)f|| \leq \tilde{C} ||f||_{H^{0,t}}$ , so that  $G_+$  is bounded as  $\lambda \to \infty$ . For  $s \geq 0$ , we use the inclusion  $H^{s,t} \subset H^{0,t}$  for any  $s \geq 0$ .

From the arguments above we immediately obtain that  $\lambda \mapsto ||\Gamma_0(\lambda)||_{\mathcal{B}(H^{s,t},\mathcal{P})}$  is continuous and bounded for  $s \geq 0$  and  $t > \frac{1}{2}$ . We can strengthen Lemma 4.3.6 for  $G_-$ .

**Lemma 4.3.7.** Let  $s \geq 0$  and  $t > \frac{n}{2}$ . Then  $\Gamma_0(\lambda) \in \mathcal{K}(H^{s,t}, \mathcal{P})$  for any  $\lambda \in \mathbb{R}^+$  and the function  $G_-: \mathbb{R}^+ \to \mathcal{K}(H^{s,t}, \mathcal{P})$  is continuous and vanishes as  $\lambda \to \infty$  and as  $\lambda \to 0$ .

*Proof.* The compactness statement follows from the compact embedding  $H^{s,t} \subset H^{q,r}$  for any q < s, r < t. See [8, Proposition 4.1.5].

For continuity, the same argument as the previous lemma works. That  $G_-$  vanishes as  $\lambda \to 0$  follows from Equation (4.18) and the fact that  $\gamma(\lambda^{\frac{1}{2}})\mathcal{F} \to \gamma_0 \mathcal{F}$  in the norm of  $\mathcal{B}(H^{s,t},\mathcal{P})$  as  $\lambda \to 0$ . It remains to check the limit as  $\lambda \to \infty$ , which follows from the inclusion  $H^{s,t} \hookrightarrow H^{0,t}$  for any  $s \geq 0$  and the estimate (4.20).

We now consider the multiplication operator  $G_-: C_c(\mathbb{R}^+, H^{s,t}) \to \mathcal{H}_{spec}$  given by

$$[G_{-}f](\lambda,\omega) := [G_{-}(\lambda)f](\lambda,\omega) = \lambda^{-\frac{1}{4}}[\Gamma_{0}(\lambda)f](\lambda,\omega). \tag{4.21}$$

Lemmas 4.3.6 and 4.3.7 show that the operator  $G_-$  extends, for  $s \geq 0$  and  $t > \frac{1}{2}$ , to an element of  $\mathcal{B}(L^2(\mathbb{R}^+, H^{s,t}), \mathcal{H}_{spec})$ . The next step is to deal with the limit as  $\varepsilon \to 0$  of  $\varphi_{\varepsilon}(M-\lambda)$ . The following statement and its proof are simple generalisations of [141, Lemma 2.3] to all  $n \geq 2$ .

**Lemma 4.3.8.** Take  $s \geq 0$  and  $t > \frac{n}{2}$ . For  $\lambda \in \mathbb{R}^+$  and  $f \in C_c(\mathbb{R}^+, \mathcal{P})$  we have

$$\lim_{\varepsilon \to 0} ||F_0^* \varphi_{\varepsilon}(M - \lambda) f - \Gamma_0(\lambda)^* f(\lambda)||_{H^{-s, -t}} = 0.$$

*Proof.* Fix  $f \in C_c(\mathbb{R}^+, \mathcal{P})$ ,  $\lambda \in \mathbb{R}^+$  and  $\varepsilon > 0$ . We use the duality relation of  $H^{s,t}$  with  $H^{-s,-t}$  to compute that

$$\begin{split} &||F_0^*\varphi_\varepsilon(M-\lambda)f-\Gamma_0(\lambda)^*f(\lambda)||_{H^{-s,-t}}\\ &=\sup_{g\in\mathcal{S}(\mathbb{R}^n),||g||_{H^{s,t}=1}}|\langle g,F_0^*\varphi_\varepsilon(M-\lambda)f-\Gamma_0(\lambda)^*f(\lambda)\rangle_{H^{s,t},H^{-s,-t}}|\\ &=\sup_{g\in\mathcal{S}(\mathbb{R}^n),||g||_{H^{s,t}=1}}\left|\frac{1}{\pi}\int_0^\infty\left\langle \Gamma_0(\mu)g,\frac{\varepsilon}{(\mu-\lambda)^2+\varepsilon^2}f(\mu)\right\rangle_{\mathcal{P}}\mathrm{d}\mu-\langle \Gamma_0(\lambda)g,f(\lambda)\rangle_{\mathcal{P}}\right|. \end{split}$$

For fixed  $g \in \mathbb{S}(\mathbb{R}^n)$  with  $||g||_{H^{s,t}} = 1$  define the continuous and bounded function  $h : \mathbb{R}^+ \to \mathbb{C}$  by

$$h(\mu) = \langle \Gamma_0(\mu)g, f(\mu) \rangle_{\mathcal{P}}.$$

By an application of the Plemelj formula (see also [161, Theorem 9.8]) we have that

$$\lim_{\varepsilon \to 0} \frac{1}{\pi} \int_0^\infty \frac{\varepsilon}{(\mu - \lambda)^2 + \varepsilon^2} h(\mu) \, \mathrm{d}\mu = h(\lambda). \tag{4.22}$$

Since  $||f(\cdot)||_{\mathcal{P}}$  and  $||\Gamma_0(\cdot)||_{\mathcal{B}(H^{s,t},\mathcal{P})}$  are bounded, we have that h is bounded by a constant

independent of q. So an application of Equation (4.22) gives

$$\lim_{\varepsilon \to 0} ||F_0^* \varphi_{\varepsilon}(M - \lambda) f - \Gamma_0(\lambda)^* f(\lambda)||_{H^{-s,-t}} \\
= \lim_{\varepsilon \to 0} \sup_{g \in \mathcal{S}(\mathbb{R}^n), ||g||_{H^{s,t}} = 1} \left| \frac{1}{\pi} \int_0^\infty \left\langle \Gamma_0(\mu) g, \frac{\varepsilon}{(\mu - \lambda)^2 + \varepsilon^2} f(\mu) \right\rangle_{\mathcal{P}} d\mu - \left\langle \Gamma_0(\lambda) g, f(\lambda) \right\rangle_{\mathcal{P}} \right| \\
= 0.$$

as claimed.  $\Box$ 

From now on we only consider the operator  $W_{-}$  since our resolvent results have only been in the upper-half-plane. Similar results could be derived in the lower-half-plane, however the operator  $W_{+}$  can be obtained in a much simpler manner from the relation  $W_{+} = W_{-}S^{*}$  at a later stage.

**Lemma 4.3.9.** Suppose that  $n \geq 3$   $\rho$ , t and V satisfy Assumption 4.3.1, and that there are no resonances for n = 4. Then the function  $B : \mathbb{R}^+ \to \mathcal{B}(\mathcal{P}, H^{0,\rho-t})$  defined by  $B(\lambda) = \lambda^{\frac{1}{4}} v M_0(\lambda)^{-1} v \Gamma_0(\lambda)^*$  is continuous and bounded, and  $B : C_c(\mathbb{R}^+, \mathcal{P}) \to L^2(\mathbb{R}^+, H^{0,\rho-t})$  defined by  $[Bf](\lambda) = B(\lambda)f(\lambda)$  extends to an element of  $\mathcal{B}(\mathcal{H}_{spec}, L^2(\mathbb{R}^+, H^{0,\rho-t}))$ .

Proof. The continuity of B follows from the limiting absorption principle (Theorem 2.4.8). For boundedness, it is sufficient to show that the map  $\lambda \mapsto ||B(\lambda)||_{\mathcal{B}(\mathcal{P},H^{0,\rho-t})}$  is bounded in a neighbourhood of 0 and in a neighbourhood of  $\infty$ . For  $\lambda \geq 1$ , a symmetrised analogue of Lemma 2.4.10 shows that the function  $\lambda \mapsto ||vM_0(\lambda)^{-1}v||_{\mathcal{B}(H^{0,-t},H^{0,\rho-t})}$  is bounded provided  $\rho > \frac{n+1}{2}$ , which is guaranteed by our assumptions. We also know from Lemma 4.3.6 that the function  $\mathbb{R}^+ \ni \lambda \mapsto \lambda^{\frac{1}{4}} ||\Gamma_0(\lambda)^*||_{\mathcal{B}(\mathcal{P},H^{0,-t})}$  is bounded. This shows that  $\lambda \mapsto ||B(\lambda)||_{\mathcal{B}(\mathcal{P},H^{0,\rho-t})}$  is bounded in a neighbourhood of  $\infty$ .

For  $\lambda$  in a neighbourhood of 0, we use asymptotic developments for  $M_0(\lambda)$  in Chapter 3 and  $\Gamma_0(\lambda)^*$  in Section 4.1.1. The asymptotics of  $\Gamma_0(\lambda)$  are discussed in Lemma 4.1.2. By taking adjoints in Lemma 4.1.2, we see that for each  $s \geq 0$  we have the expansion

$$\Gamma_0(\lambda)^* = 2^{-\frac{1}{2}} \lambda^{\frac{n-2}{4}} (\gamma_0^* - i\lambda^{\frac{1}{2}} \gamma_1^* + O(\lambda)^{\frac{1}{2}})$$

in  $\mathcal{B}(\mathcal{P}, H^{s,-t})$  as  $\lambda \to 0$ .

The form of the asymptotic development of  $M_0(\lambda)^{-1}$  as  $\lambda \to 0$  depends heavily on the (non)existence of zero energy eigenvalues and resonances as seen in Chapter 3. For  $n \geq 5$  the behaviour as  $\lambda \to 0$  of  $M_0(\lambda)$  is described explicitly in Theorem 3.2.9 (for nodd) and Theorem 3.2.19 (for n even) as

$$M_0(\lambda)^{-1} = \lambda^{-1} D_1 + \tilde{R}(\lambda^{\frac{1}{2}}),$$

in  $\mathcal{B}(\mathcal{H})$  as  $\lambda \to 0$  where  $\tilde{R}$  is uniformly bounded. So the lowest order power of  $\lambda$  in the expansion of  $\lambda^{\frac{1}{4}}vM_0(\lambda)^{-1}v\Gamma_0(\lambda)^*$  is the term  $\lambda^{\frac{n-5}{4}}vD_1v\gamma_0^*$ . So we see that  $\lambda^{\frac{1}{4}}vM_0(\lambda)^{-1}v\Gamma_0(\lambda)^*$ 

is bounded in  $\mathcal{B}(\mathcal{P}, H^{0,\rho-t})$  as  $\lambda \to 0$  for  $n \ge 5$ .

We now consider n = 4. For the asymptotic development of  $M_0(\lambda)^{-1}$ , we appeal to Theorem 3.2.37. We obtain (with  $\varphi$  a normalised vector in  $T_1\mathcal{H}$ ) the expansion

$$M_0(\lambda)^{-1} = \lambda^{-1}D_2 + \lambda^{-1}h(-i\lambda^{\frac{1}{2}})\langle\varphi,\cdot\rangle\varphi - \ln(\lambda)C_1 + \tilde{R}(\lambda^{\frac{1}{2}})$$

as  $\lambda \to 0$  in  $\mathcal{B}(H^{1,-t}, H^{0,t})$ , where  $\tilde{R}$  is uniformly bounded. Hence, we find as  $\lambda \to 0$  that

$$\lambda^{\frac{1}{4}}vM_{0}(\lambda)^{-1}v\Gamma_{0}(\lambda)^{*} = 2^{-\frac{1}{2}}\lambda^{\frac{3}{4}}v\left(\lambda^{-1}D_{2} + \lambda^{-1}h(-i\lambda^{\frac{1}{2}})\langle v\varphi, \cdot \rangle v\varphi - \ln(\lambda)C_{1} + \tilde{R}(\lambda^{\frac{1}{2}})\right)$$

$$\times v\left(\gamma_{0}^{*} - i\lambda^{\frac{1}{2}}\gamma_{1}^{*} + o(\lambda^{\frac{1}{2}})\right)$$

$$= 2^{-\frac{1}{2}}\left(\lambda^{-\frac{1}{4}}D_{2}\gamma_{0}^{*} + \lambda^{-\frac{1}{4}}h(-i\lambda^{\frac{1}{2}})\langle v\varphi, \cdot \rangle v\varphi + o(\lambda^{\frac{1}{4}}\ln(\lambda))\right).$$

There are two terms of concern here as  $\lambda \to 0$ . The first is the term  $\lambda^{-\frac{1}{4}}vD_2v\gamma_0^*$ , which vanishes by the fact  $Q_2P=0$ . The second is the term containing  $\varphi$ , which blows up as  $\lambda \to 0$ . Thus, we make the assumption  $\varphi=0$ , namely there are no resonances in dimension n=4.

Finally we consider n = 3. For the asymptotic development of  $M_0(\lambda)^{-1}$ , we appeal to Theorem 3.2.57. We obtain the expansion

$$M_0(\lambda)^{-1} = \lambda^{-1}D_2 + \lambda^{-\frac{1}{2}}C_{-\frac{1}{2}} + \tilde{R}(\lambda),$$

where  $\tilde{R}$  is uniformly bounded. Then as  $\lambda \to 0$ ,

$$\lambda^{\frac{1}{4}}vM_{0}(\lambda)^{-1}v\Gamma_{0}(\lambda)^{*} = 2^{-\frac{1}{2}}\lambda^{\frac{1}{2}}v(\lambda^{-1}D_{2} + \lambda^{-\frac{1}{2}}C_{-\frac{1}{2}} + \tilde{R}(\lambda))v\left(\gamma_{0}^{*} - i\lambda^{\frac{1}{2}}\gamma_{1}^{*} + o(\lambda^{\frac{1}{2}})\right)$$
$$= 2^{-\frac{1}{2}}\lambda^{-\frac{1}{2}}vD_{2}v\gamma_{0}^{*} + C_{-\frac{1}{2}} + o(\lambda^{\frac{1}{2}}).$$

The only term of concern here is  $vD_2v\gamma_0^*$  which vanishes since  $Q_2P=0$ .

We note that in dimension n=3 a nonsymmetrised proof of Lemma 4.3.9 can be found as [141, Lemma 2.4]. The case n=2 has been resolved in [140] in the non-resonant case and [142] in the case of s-resonances but not p-resonances. The presence of certain resonances in dimensions n=2 and n=4 is well-known to produce complicated behaviour in scattering theory, see for instance [73, Theorem III.3], [29, Theorem 6.3] and the analysis of Chapter 3, and so it is not surprising that we must exclude them in our analysis for dimension n=4, as is done in [142] for dimension n=2. Despite the differing behaviour of resonances in dimensions n=2 and n=4, the analysis required to handle the resonant contributions is similar and is work in progress. We also note the following essential corollary of Lemmas 4.3.7 and 4.3.9.

Corollary 4.3.10. Suppose that V,  $\rho$  and t satisfy Assumption 4.3.1. Then for  $n \geq 2$ 

we have the equality  $\lim_{\lambda\to\infty} S(\lambda) = \operatorname{Id}$  in the norm of  $\mathcal{B}(\mathcal{P})$ .

*Proof.* For dimension  $n \geq 3$ ,  $G_{-}(\lambda)$  and  $B(\lambda)$  satisfy  $G_{-}(\lambda)B(\lambda) = S(\lambda)$  – Id. Since  $G_{-}$  vanishes as  $\lambda \to \infty$  and B is bounded we have the desired result. In dimension n = 2, the statement is an immediate corollary of [142, Theorem 1.3].

Remark 4.3.11. The convergence of  $S(\lambda) \to \operatorname{Id}$  as  $\lambda \to \infty$  in Corollary 4.3.10 is in  $\mathcal{B}(\mathcal{P})$ . When comparing with the equality  $\operatorname{Det}(S(\lambda)) = \mathrm{e}^{-2\pi i \xi(\lambda)}$  we must be careful to note that when this equality holds it is a priori only valid in the trace norm. In fact as a further corollary we note that in dimensions  $n = 2, 3, S(\lambda)$  cannot converge as  $\lambda \to \infty$  in trace norm, except possibly when  $[\mathcal{F}V](0) = 0$ .

### 4.3.3.3 Interchanging limits

In this subsection we interchange the integral over  $\lambda$  and the limit as  $\varepsilon \to 0$  in Equation (4.17). For fixed  $\varepsilon, \lambda > 0$  and  $s \in \mathbb{R}$  we define for  $f, g \in C_c^{\infty}(\mathbb{R}^+, \mathcal{P})$  the following functions

$$h_{\varepsilon}(\lambda) := \lambda^{\frac{1}{4}} v M_{\varepsilon}(\lambda)^{-1} v F_0^* \varphi_{\varepsilon}(M - \lambda) f \in H^{0,s}, \quad \text{and}$$

$$p(\lambda) := \lambda^{-\frac{1}{4}} \Gamma_0(\lambda)^* g(\lambda) \in H^{0,-s}.$$

$$(4.23)$$

We also denote by

$$q_{\lambda}(\nu) = \begin{cases} \left(\frac{\nu+\lambda}{\lambda}\right)^{\frac{1}{4}} p(\nu+\lambda) & \nu > -\lambda \\ 0 & \nu \leq -\lambda \end{cases}$$
 (4.24)

the extension by zero of the function  $(-\lambda, \infty) \ni \nu \mapsto \left(\frac{\nu+\lambda}{\lambda}\right)^{\frac{1}{4}} p(\nu+\lambda) \in H^{0,-s}$  to all of  $\mathbb{R}$ .

**Lemma 4.3.12.** Let  $n \ge 4$  and suppose that  $\rho$ , t satisfy Assumption 4.3.1, and that there are no resonances for n = 4. If  $|V(x)| \le C(1+|x|)^{-\rho}$  then

$$\langle F_0(W_- - \operatorname{Id}) F_0^* f, g \rangle_{\mathcal{H}_{spec}} = -i \int_{\mathbb{R}^+} \int_0^\infty \left\langle h_0(\lambda), \int_0^\infty e^{i\nu z} q_\lambda(\nu) \, d\nu \right\rangle_{H^{0,s}, H^{0,-s}} dz \, d\lambda.$$

*Proof.* Take  $f \in C_c(\mathbb{R}^+, \mathcal{P})$  and  $g \in C_c^{\infty}(\mathbb{R}^+) \odot C(\mathbb{S}^{n-1})$ , and set  $m = \rho - t$ . We may then write the expression (4.17), using the functions  $h_{\varepsilon}$  and p defined in Equation (4.23), as

$$\langle F_{0}(W_{-} - \operatorname{Id})F_{0}^{*}f, g \rangle$$

$$= -\int_{\mathbb{R}} \left( \lim_{\varepsilon \to 0} \int_{0}^{\infty} \langle v M_{\varepsilon}(\lambda) v F_{0}^{*} \varphi_{\varepsilon}(M - \lambda) f, (\mu - \lambda + i\varepsilon)^{-1} \Gamma_{0}(\mu)^{*} g(\mu) \rangle_{H^{0,m}, H^{0,-m}} d\mu \right) d\lambda$$

$$= -\int_{\mathbb{R}^{+}} \left( \lim_{\varepsilon \to 0} \int_{\mathbb{R}^{+}} \left\langle h_{\varepsilon}(\lambda), \frac{\lambda^{-\frac{1}{4}} \mu^{\frac{1}{4}}}{\mu - \lambda + i\varepsilon} p(\mu) \right\rangle_{H^{0,m}, H^{0,-m}} d\mu \right) d\lambda.$$

Now we make the estimate

$$\int_{\mathbb{R}^{+}} \left| \left\langle h_{\varepsilon}(\lambda), \frac{\lambda^{-\frac{1}{4}} \mu^{\frac{1}{4}}}{\mu - \lambda + i\varepsilon} p(\mu) \right\rangle_{H^{0,m}, H^{0,-m}} \right| d\mu$$

$$\leq ||h_{\varepsilon}(\lambda)||_{H^{0,m}} \lambda^{-\frac{1}{4}} \int_{\mathbb{R}^{+}} \mu^{\frac{1}{4}} ((\mu - \lambda)^{2} + \varepsilon^{2})^{-\frac{1}{2}} ||p(\mu)||_{H^{0,-m}} d\mu < \infty, \tag{4.25}$$

since p is defined in terms of g, which is compactly supported. Use the formula

$$(\mu - \lambda + i\varepsilon)^{-1} = -i \int_0^\infty e^{i(\mu - \lambda)z} e^{-\varepsilon z} dz$$

and Fubini's theorem (valid by the estimate (4.25)) to obtain

$$\lim_{\varepsilon \to 0} \int_{0}^{\infty} \left\langle h_{\varepsilon}(\lambda), \frac{\lambda^{-\frac{1}{4}} \mu^{\frac{1}{4}}}{\mu - \lambda + i\varepsilon} p(\mu) \right\rangle_{H^{0,m}, H^{0,-m}} d\mu$$

$$= -i \lim_{\varepsilon \to 0} \int_{0}^{\infty} e^{-\varepsilon z} \left( \left\langle h_{\varepsilon}(\lambda), \int_{0}^{\infty} e^{i(\mu - \lambda)z} \lambda^{-\frac{1}{4}} \mu^{\frac{1}{4}} p(\mu) d\mu \right\rangle_{H^{0,m}, H^{0,-m}} \right) dz$$

$$= -i \lim_{\varepsilon \to 0} \int_{0}^{\infty} e^{-\varepsilon z} \left( \left\langle h_{\varepsilon}(\lambda), \int_{-\lambda}^{\infty} e^{i\nu z} \left( \frac{\nu + \lambda}{\lambda} \right)^{\frac{1}{4}} p(\nu + \lambda) d\nu \right\rangle_{H^{0,m}, H^{0,-m}} \right) dz, \quad (4.26)$$

where in the last line we have made a change of variables to recognise a Fourier transform. The z-integrand of (4.26) can be bounded independently of  $\varepsilon \in (0,1)$  explicitly with

$$\left| e^{-\varepsilon z} \left\langle h_{\varepsilon}(\lambda), \int_{-\lambda}^{\infty} e^{i\nu z} \left( \frac{\nu + \lambda}{\lambda} \right)^{\frac{1}{4}} p(\nu + \lambda) d\nu \right\rangle_{H^{0,m}, H^{0,-m}} \right| \\
\leq \left| \left| h_{\varepsilon}(\lambda) \right| \right|_{H^{0,m}} \left| \left| \int_{-\lambda}^{\infty} e^{i\nu z} \left( \frac{\nu + \lambda}{\lambda} \right)^{\frac{1}{4}} p(\nu + \lambda) d\nu \right| \right|_{H^{0,-m}} := \left| \left| h_{\varepsilon}(\lambda) \right| \right|_{H^{0,m}} j_{\lambda}(z), \quad (4.27)$$

where we have used  $e^{-\varepsilon z} \leq 1$ . We also know from Lemma 4.3.8 that as  $\varepsilon \to 0$ ,  $h_{\varepsilon}(\lambda)$  converges to  $\lambda^{\frac{1}{4}}vM_0(\lambda)^{-1}v\Gamma_0(\lambda)^*f(\lambda)$  in  $H^{0,m}$ . Therefore the family  $||h_{\varepsilon}(\lambda)||_{H^{0,m}}$  (and thus the entire expression (4.27)) is bounded by a constant independent of  $\varepsilon \in (0,1)$ .

In order to exchange the integral over z and the limit as  $\varepsilon \to 0$  in Equation (4.26), it remains to show that the function  $j_{\lambda}$  of (4.27) is in  $L^{1}(\mathbb{R}^{+}, dz)$ . Recall  $q_{\lambda}$  from Equation (4.24). Note that  $j_{\lambda}$  in (4.27) can be rewritten as  $j_{\lambda}(z) = (2\pi)^{\frac{1}{2}} ||[\mathcal{F}^{*}q_{\lambda}](z)||_{H^{0,-m}}$  (here the Fourier transform is one-dimensional).

To estimate  $j_{\lambda}$ , we denote by  $D = -i\frac{\partial}{\partial\nu}$  the self-adjoint operator acting (densely) on  $L^2(\mathbb{R})$  and by  $Y = (\mathrm{Id} + D^2)$ . Then,

$$||[\mathcal{F}^*q_{\lambda}](z)||_{H^{0,-m}} = (1+z^2)^{-1} ||[\mathcal{F}^*Yq_{\lambda}](z)||_{H^{0,-m}}.$$
 (4.28)

The function  $z \mapsto (1+z^2)^{-1}$  is in  $L^1(\mathbb{R}^+, dz)$  and so it suffices to show  $||[\mathcal{F}^*Yq_{\lambda}](z)||_{H^{0,-m}}$  is bounded independently of z. Suppose that  $g = \eta \otimes \xi \in C_c^{\infty}(\mathbb{R}^+) \otimes C(\mathcal{P})$  is a simple tensor, so that

$$\left(\frac{\nu+\lambda}{\lambda}\right)^{\frac{1}{4}} [p(\nu+\lambda)](x) = 2^{-\frac{1}{2}} (2\pi)^{-\frac{n}{2}} \lambda^{-\frac{1}{4}} (\nu+\lambda)^{\frac{n-2}{4}} \eta(\nu+\lambda) \int_{\mathbb{S}^{n-1}} e^{i(\nu+\lambda)^{\frac{1}{2}} \langle x, \omega \rangle} \xi(\omega) d\omega.$$

Differentiating with respect to  $\nu$  twice shows that there are  $\eta_{j,\lambda} \in C_c^{\infty}(\mathbb{R}^+)$  such that for  $x \in \mathbb{R}^n$  we have

$$[Yq_{\lambda}](\nu)(x) = \eta_{1,\lambda}(\nu) \int_{\mathbb{S}^{n-1}} e^{i(\nu+\lambda)^{\frac{1}{2}}\langle x,\omega\rangle} \xi(\omega) d\omega + \eta_{2,\lambda}(\nu) \int_{\mathbb{S}^{n-1}} \langle x,\omega\rangle e^{i(\nu+\lambda)^{\frac{1}{2}}\langle x,\omega\rangle} \xi(\omega) d\omega + \eta_{3,\lambda}(\nu) \int_{\mathbb{S}^{n-1}} (\langle x,\omega\rangle)^2 e^{i(\nu+\lambda)^{\frac{1}{2}}\langle x,\omega\rangle} \xi(\omega) d\omega.$$

Estimating each term individually we obtain  $C_1, C_2, C_3, C > 0$  such that

$$|[\mathcal{F}^*Yq_{\lambda}](z)(x)| \le C_1 + C_2|x| + C_3|x|^2 \le C(1+|x|^2),$$

which gives

$$|[[\mathcal{F}^*Yq_{\lambda}](z)](x)| \le C(1+|x|^2).$$

For  $\tau(x) = (1 + |x|^2)$ , we compute

$$||\tau||_{H^{0,-m}}^2 = \int_{\mathbb{R}^n} (1+|x|^2)^{-\frac{m}{2}} (1+|x|^2) \, \mathrm{d}x = \operatorname{Vol}(\mathbb{S}^{n-1}) \int_0^\infty r^{n-1} (1+r^2)^{1-\frac{m}{2}} \, \mathrm{d}r,$$

which is finite for n-m-1<-1, or m>n+2. So  $\tau\in H^{0,-m}$  for m>n+2.

We note in passing that the condition on  $m = \rho - t$  requires  $\rho > n + t + 2$ , which is guaranteed by Assumption 4.3.1.

By linearity, for each  $g \in C_c^{\infty}(\mathbb{R}^+) \odot C(\mathbb{S}^{n-1})$  we find that  $||[\mathcal{F}^*Yq_{\lambda}](z)||_{H^{0,-m}}$  is bounded independently of z. So using Equation (4.28) we have  $||[\mathcal{F}^*q_{\lambda}](z)||_{H^{0,-m}} \leq \tilde{C}(1+z^2)^{-1}$ , and so  $||[\mathcal{F}^*q_{\lambda}](z)||_{H^{0,-m}} \in L^1(\mathbb{R}^+)$ .

As a consequence, we can apply Lebesgue's dominated convergence theorem to exchange the limit as  $\varepsilon \to 0$  with the integral over z in Equation (4.26) and obtain

$$-i\lim_{\varepsilon\to 0} \int_0^\infty e^{-\varepsilon z} \left( \left\langle h_{\varepsilon}(\lambda), \int_{-\lambda}^\infty e^{i\nu z} \left( \frac{\nu+\lambda}{\lambda} \right)^{\frac{1}{4}} p(\nu+\lambda) d\nu \right\rangle_{H^{0,m},H^{0,-m}} \right) dz$$
$$= -i \left\langle h_0(\lambda), \int_0^\infty \int_{\mathbb{R}} e^{i\nu z} q_{\lambda}(\nu) d\nu dz \right\rangle_{H^{0,m},H^{0,-m}}.$$

Hence on a dense set of f, g,

$$\langle F_0(W_- - \operatorname{Id}) F_0^* f, g \rangle_{\mathcal{H}_{spec}} = -i \int_{\mathbb{R}^+} \int_0^\infty \left\langle h_0(\lambda), \int_{\mathbb{R}} e^{i\nu z} q_\lambda(\nu) \, d\nu \right\rangle_{H^{0,m} H^{0,-m}} dz \, d\lambda. \quad \Box$$

Remark 4.3.13. The proof of Lemma 4.3.12 shows how the various requirements on  $\rho$  in Assumption 4.3.1 are obtained. In particular, in Lemma 2.4.10 we assumed  $\rho > \frac{n+1}{2}$  and in Lemma 4.1.1 we required  $t > \frac{n}{2}$ . In Lemma 4.3.12 we required  $\rho > n+t+2$  which gives  $\rho > \frac{3n+4}{2}$  for  $n \geq 5$ . The case n = 4 requires  $\rho > 11$  to use Theorem 3.2.37.

#### 4.3.3.4 The function of dilation and completion of proof

We recall from Section 2.2.3  $D_+$ , the self-adjoint generator of dilations on  $L^2(\mathbb{R}^+)$ . Lemma 2.2.21 allows us to take functions of  $D_+$  via

$$[\psi(D_+)f](x) = (2\pi)^{-\frac{1}{2}} \int_{\mathbb{R}} [\mathcal{F}^*\psi](t)[U_+(t)f](x) dt.$$

We introduce the function  $\psi \in C(\mathbb{R}) \cap L^{\infty}(\mathbb{R})$  defined by

$$\psi(x) = \frac{1}{2} \left( 1 - \tanh(2\pi x) - i\cosh(2\pi x)^{-1} \right). \tag{4.29}$$

The Hilbert spaces  $L^2(\mathbb{R}^+, H^{s,t})$  and  $\mathcal{H}_{spec}$  can be naturally identified with the Hilbert spaces  $L^2(\mathbb{R}^+) \otimes H^{s,t}$  and  $L^2(\mathbb{R}^+) \otimes \mathcal{P}$ . We recall the operator  $G_-$  of Equation (4.21) and B of Lemma 4.3.9.

**Theorem 4.3.14.** Let  $n \ge 4$ . Suppose that  $\rho$ , t satisfy Assumption 4.3.1, and that there are no resonances for n = 4. Then

$$F_0(W_- - \mathrm{Id})F_0^* = -2\pi i G_-(\psi(D_+) \otimes \mathrm{Id}_{H^{0,\rho-t}})B$$

in  $\mathcal{B}(\mathcal{H}_{spec})$ .

Proof. Take  $f \in C_c(\mathbb{R}^+, \mathcal{P})$  and  $g \in C_c^{\infty}(\mathbb{R}^+) \odot C(\mathbb{S}^{n-1})$ , and set  $m = \rho - t$ . We show that  $\langle F_0(W_- - \operatorname{Id})F_0^*f, g \rangle_{\mathcal{H}_{spec}}$  is equal to  $\langle -2\pi i G_-(\psi(D_+) \otimes \operatorname{Id}_{H^{0,m}})Bf, g \rangle_{\mathcal{H}_{spec}}$ . Write  $\chi_+$  for the characteristic function of  $\mathbb{R}^+$  and note that  $q_{\lambda}$  has compact support to obtain (in the

sense of distributions with values in  $H^{0,-m}$ ),

$$\int_{0}^{\infty} \int_{\mathbb{R}} e^{i\nu z} q_{\lambda}(\nu) d\nu dz = (2\pi)^{\frac{1}{2}} \int_{\mathbb{R}} [\mathcal{F}^{*}\chi_{+}](\nu) q_{\lambda}(\nu) d\nu$$

$$= (2\pi)^{\frac{1}{2}} \int_{-\lambda}^{\infty} [\mathcal{F}^{*}\chi_{+}](\nu) \left(\frac{\nu+\lambda}{\lambda}\right)^{\frac{1}{4}} p(\nu+\lambda) d\nu$$

$$= (2\pi)^{\frac{1}{2}} \int_{\mathbb{R}} [\mathcal{F}^{*}\chi_{+}](\lambda(e^{\mu}-1)) \lambda e^{\frac{5\mu}{4}} p(e^{\mu}\lambda) d\mu$$

$$= (2\pi)^{\frac{1}{2}} \int_{\mathbb{R}} [\mathcal{F}^{*}\chi_{+}](\lambda(e^{\mu}-1)) \lambda e^{\frac{3\mu}{4}} [(U_{+}(\mu) \otimes \operatorname{Id}_{H^{0,-m}})p](\lambda) d\mu.$$

Next, note that the inverse Fourier transform of the characteristic function  $\chi_+$  is the distribution  $[\mathcal{F}^*\chi_+](y) = \pi^{\frac{1}{2}}2^{-\frac{1}{2}}\delta(y) + i(2\pi)^{-\frac{1}{2}}\operatorname{Pv}_{\frac{1}{y}}$  (here Pv denotes the principal value), from which we obtain

$$\int_0^\infty \int_{\mathbb{R}} e^{i\nu z} q_{\lambda}(\nu) d\nu dz = \int_{\mathbb{R}} \left( \pi \delta(e^{\mu} - 1) + i \operatorname{Pv} \frac{e^{\frac{3\mu}{4}}}{e^{\mu} - 1} \right) \left[ (U_+(\mu) \otimes \operatorname{Id}_{H^{0, -s}}) p \right] (\lambda) d\mu$$

Next, note that

$$\frac{e^{\frac{3\mu}{4}}}{e^{\mu} - 1} = \frac{1}{4} \left( \frac{1}{\sinh\left(\frac{\mu}{4}\right)} + \frac{1}{\cosh\left(\frac{\mu}{4}\right)} \right)$$

and the Fourier transform of the function  $\psi$  of Equation (4.29) is [84, Table 20.1]

$$[\mathcal{F}\psi](\mu) = \pi^{\frac{1}{2}} 2^{-\frac{1}{2}} \delta(e^{\mu} - 1) + \frac{i}{4} (2\pi)^{-\frac{1}{2}} \operatorname{Pv}\left(\frac{1}{\sinh\left(\frac{\mu}{4}\right)} + \frac{1}{\cosh\left(\frac{\mu}{4}\right)}\right).$$

Combining these facts,

$$\langle F_0(W_- - \operatorname{Id} f, g \rangle_{\mathcal{H}_{spec}} = i \int_{\mathbb{R}^+} \left\langle h_0(\lambda), \int_{\mathbb{R}} \left( \pi \delta(e^{\mu} - 1) + \frac{1}{4} \operatorname{Pv} \left( \frac{1}{\sinh\left(\frac{\mu}{4}\right)} + \frac{1}{\cosh\left(\frac{\mu}{4}\right)} \right) \times \left[ (U_+(\mu) \otimes \operatorname{Id}_{H^{0,-m}}) p \right](\lambda) \right) d\mu \right\rangle_{H^{0,m}, H^{0,-m}} d\lambda.$$

Since  $[Bf](\lambda) = h_0(\lambda), p = G_-^*g$  and

$$(\psi(D_+)\otimes \operatorname{Id}_{H^{0,m}})^*p = (2\pi)^{-\frac{1}{2}} \int_{\mathbb{R}} [\mathcal{F}\psi](\mu)(U_+(\mu)\otimes \operatorname{Id}_{H^{0,m}})p \,\mathrm{d}\mu.$$

Combining these facts we obtain

$$\langle F_0(W_- - \operatorname{Id}) F_0^* f, g \rangle_{\mathcal{H}_{spec}} = 2\pi i \int_{\mathbb{R}^+} \langle [Bf](\lambda), [(\psi(D_+)^* \otimes \operatorname{Id}_{H^{0,-s}}) G_-^* g](\lambda) \rangle_{H^{0,m}, H^{0,-m}} d\lambda$$
$$= \langle -2\pi i G_-(\psi(D_+) \otimes \operatorname{Id}_{H^{0,m}}) Bf, g \rangle_{\mathcal{H}_{spec}}.$$

Density of  $C_c(\mathbb{R}^+, \mathcal{P})$  and  $C_c^{\infty}(\mathbb{R}^+) \odot C(\mathbb{S}^{n-1})$  in  $\mathcal{H}_{spec}$  concludes the proof.

The following is a straightforward generalisation of [141, Lemma 2.7].

**Lemma 4.3.15.** Take  $s \ge 0$  and  $t > \frac{n}{2}$ , and suppose that there are no resonances for n = 4. Then the difference

$$(\psi(D_+) \otimes \operatorname{Id}_{\mathcal{P}})G_- - G_-(\psi(D_+) \otimes \operatorname{Id}_{H^{s,t}}) \tag{4.30}$$

is in  $\mathcal{K}(L^2(\mathbb{R}^+, H^{s,t}), \mathcal{H}_{spec})$ .

*Proof.* Define the unitary operator  $J: L^2(\mathbb{R}) \to L^2(\mathbb{R}^+)$  by

$$[Jf](\lambda) = \lambda^{-\frac{1}{2}} f(\ln(\lambda)).$$

Then J satisfies  $[J^*U_+(t)Jf](x) = f(x+t)$  and  $[J^*e^{it\ln(M)}Jf](x) = e^{itx}f(x)$  for each  $x, t \in \mathbb{R}$ , with M the operator of multiplication by the variable in  $L^2(\mathbb{R}^+)$ . It follows that  $J^*D_+J=D$  on Dom(D) and  $J^*\ln(M)J=X$  on Dom(X), where D and X are the operators of momentum and position in  $L^2(\mathbb{R})$ .

Take  $f_1, f_2 : \mathbb{R} \to \mathbb{C}$  with limits at  $\pm \infty$ . A result of Cordes [8, Theorem 4.1.10] implies  $[f_1(D), f_2(X)] \in \mathcal{K}(L^2(\mathbb{R}))$ . Conjugating with the unitary operator J, we find that  $[f_1(D_+), f_3(M)] \in \mathcal{K}(L^2(\mathbb{R}^+))$ , with  $f_3 = f_2 \circ \ln$ .

We know that  $[Q\xi](\lambda) = q(\lambda)\xi(\lambda)$  for  $\xi \in C_c(\mathbb{R}^+, H^{s,t})$  and  $\lambda \in \mathbb{R}^+$ , with  $q \in C([0,\infty], \mathcal{K}(H^{s,t},\mathcal{P}))$ , when  $C([0,\infty], \mathcal{K}(H^{s,t},\mathcal{P}))$  is equipped with the uniform topology [133, Theorem 1.15]. So for each  $\varepsilon > 0$  there exists  $n \in \mathbb{N}$ ,  $a_j \in C([0,\infty])$  and  $b_j \in \mathcal{K}(H^{s,t},\mathcal{P})$  such that

$$\left\| N - \sum_{j=1}^{k} a_j(M) \otimes b_j \right\|_{\mathcal{B}(L^2(\mathbb{R}^+, H^{s,t}), \mathcal{H}_{spec})} < \varepsilon.$$

So in order to prove our claim, it is sufficient to show that the operator

$$(\psi(D_{+}) \otimes \operatorname{Id}_{\mathcal{P}}) \left( \sum_{j=1}^{n} a_{j}(M) \otimes b_{j} \right) - \left( \sum_{j=1}^{n} a_{j}(M) \otimes b_{j} \right) (\psi(D_{+}) \otimes \operatorname{Id}_{H^{s,t}})$$

$$= \sum_{j=1}^{n} \left[ \psi(D_{+}), a_{j}(M) \right] \otimes b_{j}$$

is compact. We know that  $b_j \in \mathcal{K}(H^{s,t}, \mathcal{P})$  and that  $[\psi(D_+), a_j(M)] \in \mathcal{K}(L^2(\mathbb{R}^+))$ . It follows that the operator (4.30) is compact, since finite sums and tensor products of compact operators are compact operators (see [77, Theorem 2]).

Recall from Lemma 2.2.26 that the generator of dilations  $D_n$  satisfies  $F_0D_nF_0^* = -2D_+ \otimes \operatorname{Id}_{\mathcal{P}}$ . Defining  $\varphi(x) = \frac{1}{2}(1 + \tanh(\pi x) - i\cosh(\pi x)^{-1})$ , we have

$$F_0\varphi(D_n)F_0^* = \psi(D_+) \otimes \mathrm{Id}_{\mathcal{P}}.\tag{4.31}$$

We can now prove the main result.

*Proof of Theorem 4.0.1.* Using Equation (4.31) at the last equality, we deduce from Theorem 4.3.14 and Lemma 4.3.15 that

$$W_{-} - \operatorname{Id} = -2\pi i F_{0}^{*} G_{-}(\psi(D_{+}) \otimes \operatorname{Id}_{H^{0,\rho-t}}) B F_{0}$$

$$= -2\pi i F_{0}^{*} \left[ G_{-}(\psi(D_{+}) \otimes \operatorname{Id}_{H^{0,\rho-t}}) - (\psi(D_{+}) \otimes \operatorname{Id}_{\mathcal{P}}) G_{-} + (\psi(D_{+}) \otimes \operatorname{Id}_{\mathcal{P}}) G_{-} \right] B F_{0}$$

$$= -2\pi i F_{0}^{*} (\psi(D_{+}) \otimes \operatorname{Id}_{\mathcal{P}}) G_{-} B F_{0} + K$$

$$= \varphi(D_{n}) F_{0}^{*} (-2\pi i) G_{-} B F_{0} + K,$$

with  $K := -2\pi i F_0^* \left( G_-(\psi(D_+) \otimes \operatorname{Id}_{H^{0,\rho-t}}) - (\psi(D_+) \otimes \operatorname{Id}_{\mathcal{P}}) G_- \right) BF_0 \in \mathcal{K}(\mathcal{H})$ . Note that the last equality follows from Equation (4.31). Using Theorem 2.4.32, we compute

$$-2\pi i [G_{-}Bf](\lambda,\omega) = -2\pi i [\Gamma_0(\lambda)vM_0(\lambda)^{-1}v\Gamma_0(\lambda)^*f(\lambda,\cdot)](\omega) = [(S(\lambda) - \mathrm{Id})f(\lambda,\cdot)](\omega).$$

Thus 
$$F_0^*(-2\pi i)G_-BF_0 = S - \mathrm{Id}$$
, proving the claim.

Corollary 4.3.16. Suppose that V satisfies Assumption 4.3.1, that there are no p resonances in dimension n = 2 and no resonances in dimension n = 4. Then the operator K defined by Equation (4.7) is compact.

In the next chapter we show that as an immediate consequence of Theorem 4.0.1 we can view Levinson's theorem in all dimensions as an index theorem as discussed in [98, Sections 4-6]. In Chapter 5 we discuss this topological interpretation and use the result of Theorem 4.0.1 to give a new topological interpretation of Levinson's theorem.

# Chapter 5

# Levinson's theorem as an index pairing

The form of the wave operator found in Theorem 4.0.1 allows us to view Levinson's theorem in a topological manner in multiple ways. In [97, 98, 140, 141, 142] it is shown in low dimensions that Levinson's theorem is an index theorem by considering the wave operator as an element of a particular  $C^*$ -algebra and computing the winding number of its image in a Toeplitz algebra. In Section 5.1 we describe in detail the Toeplitz algebra approach of Kellendonk and Richard [97, 98] and demonstrate using Theorem 4.0.1 how it applies to the wave operator of higher dimensional scattering. Of particular interest is the nature in which resonances contribute to the winding number as considered in dimension n = 1 [97] and dimension n = 3 [98].

We next demonstrate a new topological approach to Levinson's theorem by reinterpreting the problem as an index pairing. In Section 5.2 we construct a class of Fredholm operators  $W_U$  indexed by a particular family of unitaries  $\{U\} \subset \mathcal{B}(\mathcal{H})$ . Each such unitary U defines a K-theory class [U] and can be paired with the K-homology class  $[D_+]$  of the dilation operator, which we construct in Section 5.2.1. We show in Theorem 5.2.14 that such a pairing computes the quantity  $Index(W_U)$ . Theorem 4.0.1 shows that generically we have the equality  $W_- = W_S$  and as a consequence we reinterpret Levinson's theorem as an index pairing in Theorem 5.3.2. This pairing is joint work with Adam Rennie and appears in the preprint [5] in the non-resonant case.

The existence of zero energy resonances causes obstructions to the existence of such a pairing and as such some care is required to modify the index pairing to handle this information. In Section 5.3.1 we provide the necessary modifications in dimension n = 1 to obtain an index pairing in all cases in Theorem 5.3.15. Using results of [97] we obtain the a new proof of the well-known Levinson's theorem [108] in dimension n = 1 in Theorem 5.3.16. Such a statement also allows us to completely determine the spectral shift function for the pair  $(H, H_0)$  in any generic or resonant case.

In Section 5.3.2 we provide the required modifications to obtain an index pairing in

the resonant case in dimension n=3 in Theorem 5.3.20. Unfortunately, the high energy behaviour of the scattering matrix in trace norm provides an additional obstruction to using the index pairing to obtain computational formulae for the index. In Section 5.3.2 we consider a unitary related to S with better trace norm decay at infinity in order to construct in Lemma 5.3.22 an element of the class  $[W_-]$  which can be used to obtain a new proof of the well-known Levinson's theorem in dimension n=3 [31], which we state and prove in Theorem 5.3.24. Such a statement of Levinson's theorem allows us to completely determine the spectral shift function for the pair  $(H, H_0)$  in both the generic and resonant case.

In Section 5.3.3 in dimension n=2 we proceed as in dimension n=3 by constructing in Lemma 5.3.28 an element of  $[W_{-}]$  with better trace norm properties which can be used in Theorem 5.3.29 to obtain (in the case of no p-resonances) a new proof of the known Levinson theorem of [26, Theorem 6.3] (see also [50, Theorem 3.2]). We can further use Theorem 5.3.29 to completely determine the spectral shift function for the pair  $(H, H_0)$  in the case of no p-resonances.

In Section 5.3.4 we demonstrate the necessary modifications to construct in Lemma 5.3.33 an element of  $[W_{-}]$  which we use in Theorem 5.3.34 a new proof of Levinson's theorem in higher dimensions (with the additional assumption that there are no resonances in dimension n = 4). Unfortunately Theorem 5.3.34 has the drawback that most of the correction terms provided by Lemma 5.3.33 are not easily computable. We demonstrate the easiest cases of dimension n = 4, 5 to illustrate this difficulty. We then use Theorem 5.3.34 to obtain a complete characterisation of the spectral shift function for the pair  $(H, H_0)$ .

#### 5.1 Levinson's theorem as an index theorem

As an immediate consequence of the form of the wave operator in Theorem 4.0.1 we can view Levinson's theorem as an index theorem by considering the wave operator as an element of a particular  $C^*$ -algebra and computing the winding number of its image under a quotient map. Such results have appeared in various forms in [97, 98, 138, 140, 141]. In particular we refer to the review article [138] for greater detail.

We construct a  $C^*$ -algebra which contains the wave operator as an element. Let

$$C(\overline{\mathbb{R}^+}, \mathcal{K}(\mathcal{P})) = \{ f \in C(\mathbb{R}^+, \mathcal{K}(\mathcal{P})) : \lim_{y \to \infty} f(y) \text{ exists} \} \quad \text{and} \quad C(\overline{\mathbb{R}}, \mathcal{K}(\mathcal{P})) = \{ g \in C(\mathbb{R}, \mathcal{K}(\mathcal{P})) : \lim_{x \to +\infty} g(x) \text{ exists} \},$$

each equipped with the sup norm. The ideals of functions vanishing at the endpoints are

$$C_0(\overline{\mathbb{R}^+}, \mathcal{K}(\mathcal{P})) = \{ f \in C_0(\mathbb{R}^+, \mathcal{K}(\mathcal{P})) : \lim_{y \to \infty} f(y) = 0 \} = C_0(\mathbb{R}^+, \mathcal{K}(\mathcal{P})) \quad \text{and} \quad C_0(\overline{\mathbb{R}}, \mathcal{K}(\mathcal{P})) = \{ g \in C(\mathbb{R}, \mathcal{K}(\mathcal{P})) : \lim_{x \to +\infty} g(x) = 0 \} = C_0(\mathbb{R}, \mathcal{K}(\mathcal{P})).$$

Note that the continuity here refers to norm continuity.

**Definition 5.1.1.** Define E to be the  $C^*$ -algebra generated by

$$E := \left\{ g(D_n) f(H_0) : f \in C(\overline{\mathbb{R}^+}, \mathcal{K}(\mathcal{P})), g \in C(\overline{\mathbb{R}}, \mathcal{K}(\mathcal{P})) \right\}.$$

Define J to be the ideal generated by

$$J := \left\{ g(D_n) f(H_0) : f \in C_0(\mathbb{R}, \mathcal{K}(\mathcal{P})), g \in C_0(\mathbb{R}^+, \mathcal{K}(\mathcal{P})) \right\}.$$

Note that the  $C^*$ -algebra E is nonunital, since the space  $\mathcal{P}$  is not finite dimensional (except in dimension n=1).  $C^*$ -algebras isomorphic to E and J have been studied in a different context in [68], where the isomorphism is given by the Mellin transform. The key relation is  $\frac{1}{2}[D_n, \ln(H_0)] = i$ , the canonical commutation relation. In a similar manner, we can consider the pair X of multiplication by the variable and the Dirac operator D acting on  $L^2(\mathbb{R})$  and construct a  $C^*$ -algebra.

**Definition 5.1.2.** Define  $E_{D,X}$  to be the  $C^*$ -algebra generated by all f(D)g(X) such that  $f, g \in C(\overline{\mathbb{R}}, \mathcal{K}(\mathcal{P}))$  (that is f, g are continuous with limits at  $\pm \infty$ ) and  $J_{D,X}$  to be the subalgebra generated by all f(D)g(X) such that  $f, g \in C_0(\mathbb{R}, \mathcal{K}(\mathcal{P}))$  (that is f, g are continuous and vanish at  $\pm \infty$ ).

**Lemma 5.1.3.** The subalgebra  $J_{D,X}$  is a an ideal in  $E_{D,X}$  and  $J_{D,X} \cong \mathcal{K}(L^2(\mathbb{R}))$ .

*Proof.* That  $J_{D,X}$  is an ideal in  $E_{D,X}$  is a simple check. That  $J_{D,X} \cong \mathcal{K}(L^2(\mathbb{R}))$  is the statement of [68, Corollary 2.18], however we provide also a more elementary proof based on the Stone-von Neumann theorem.

Since the pair (D, X) satisfy the canonical commutation relations, any  $f, g \in C_0(\mathbb{R})$  satisfy  $f(X)g(D) \in \mathcal{K}(L^2(\mathbb{R}))$  by [151, Theorem 4.1], so  $J_{D,X} \subseteq \mathcal{K}(L^2(\mathbb{R}))$ . For the reverse inclusion, fix  $K \in \mathcal{K}(L^2(\mathbb{R}))$  and approximate K by a sequence  $(K_N)$  of finite rank operators, so that  $||K - K_N|| \to 0$  as  $N \to \infty$ . Write

$$K_N = \sum_{j=1}^N \langle \xi_j, \cdot \rangle \eta_j$$

for  $(\xi_j), (\eta_j) \subset L^2(\mathbb{R})$ . Approximate each  $\xi_j$  and  $\eta_j$  by  $\tilde{\xi}_j$  and  $\tilde{\eta}_j$  with  $\tilde{\xi}_j, \tilde{\eta}_j \in C_0(\mathbb{R}) \cap L^2(\mathbb{R})$ 

(and thus each  $\tilde{\xi}_i, \tilde{\eta}_i$  vanishes at  $\pm \infty$ ). Then

$$\tilde{K}_N := \sum_{j=1}^N \langle \tilde{\xi}_j, \cdot \rangle \tilde{\eta}_j$$

satisfies  $\tilde{K}_N \in J_{D,X}$  and converges in norm to K. Since  $J_{D,X}$  is a closed two-sided ideal,  $K \in J_{D,X}$  also so  $J_{D,X} = \mathcal{K}(L^2(\mathbb{R}))$ .

We provide a convenient description of the quotient  $E_{D,X}/J_{D,X}$  for explicit computations. Let  $q: E_{D,X} \to E_{D,X}/J_{D,X}$  be the quotient map. Consider the square  $\blacksquare = [-\infty, \infty] \times [-\infty, \infty]$  whose boundary  $\square$  is the union of the four edges  $E_1 = \{-\infty\} \times [-\infty, \infty]$ ,  $E_2 = [-\infty, \infty] \times \{\infty\}$ ,  $E_3 = \{\infty\} \times [-\infty, \infty]$ , and  $E_4 = [-\infty, \infty] \times \{-\infty\}$ . We identify the space  $C(\square)$  with the subalgebra of  $C([-\infty, \infty])^{\oplus 4}$  given by elements  $\Gamma = (\Gamma_1, \Gamma_2, \Gamma_3, \Gamma_4)$  which coincide at the corresponding endpoints. That is,  $\Gamma_1(\infty) = \Gamma_2(-\infty)$ ,  $\Gamma_2(\infty) = \Gamma_3(\infty)$ ,  $\Gamma_3(\infty) = \Gamma_4(\infty)$  and  $\Gamma_4(-\infty) = \Gamma_1(-\infty)$ .

**Lemma 5.1.4.** The algebras  $C(\square)$  and  $E_{D,X}/J_{D,X}$  are isomorphic. In particular, for any  $f,g \in C(\overline{\mathbb{R}})$  we have the relation

$$q(f(D)g(X)) = (f(-\infty)g(\cdot), f(\cdot)g(\infty), f(\infty)g(\cdot), f(\cdot)g(-\infty)). \tag{5.1}$$

*Proof.* This is the statement of [68, Proposition 3.22].

From Lemma 5.1.4 we obtain a short exact sequence

$$0 \to \mathcal{K}(L^2(\mathbb{R})) \to E_{D,X} \xrightarrow{q} C(\square) \to 0.$$
 (5.2)

Let  $\mathcal{M}: L^2(\mathbb{R}^+) \to L^2(\mathbb{R}^+)$  be the Mellin transform of Definition 2.2.22 and let  $B = -M_{\text{ln}}$  be the operator of multiplication by  $-\ln$  on  $L^2(\mathbb{R}^+)$ . Note that  $B = -\ln(L)$  where L is the operator of multiplication by the variable in  $L^2(\mathbb{R}^+)$ ,  $\mathcal{M}D_+\mathcal{M}^* = X$ , and  $\mathcal{M}B\mathcal{M}^* = D$ . For  $f \in C(\overline{\mathbb{R}})$  we have the relation

$$\mathcal{M}^* f(D)\mathcal{M} = f(\mathcal{M}^* D \mathcal{M}) = f(B) = f(-\ln(L)) = g(L), \tag{5.3}$$

where  $g = f \circ (-\ln) \in C(\overline{\mathbb{R}^+})$ . This motivates the following definition.

**Definition 5.1.5.** Let  $E_{L,D_+}$  be the  $C^*$ -algebra generated by  $f(L)g(D_+)$  with  $f \in C(\overline{\mathbb{R}^+}, \mathcal{K}(\mathcal{P}))$  and  $g \in C(\overline{\mathbb{R}}, \mathcal{K}(\mathcal{P}))$ . Let  $J_{L,D_+}$  be the subalgebra generated by  $f(L)g(D_+)$  with  $f \in C_0(\overline{\mathbb{R}^+}, \mathcal{K}(\mathcal{P}))$  and  $g \in C_0(\overline{\mathbb{R}}, \mathcal{K}(\mathcal{P}))$ .

**Lemma 5.1.6.** The  $C^*$ -algebras  $E_{L,D_{\perp}}$  and  $E_{D,X}$  are isomorphic.

*Proof.* This follows from the relations  $\mathcal{M}D_+\mathcal{M}^*=X$  and  $\mathcal{M}B\mathcal{M}^*=D$  combined with Equation (5.3).

One can perform an identical identification procedure for  $E_{L,D_+}/J_{L,D_+}$  with functions on a square. Consider the square  $\blacksquare' = [0,\infty] \times [-\infty,\infty]$  whose boundary  $\square'$  is the union of the four edges  $B_1 = \{0\} \times [-\infty,\infty]$ ,  $B_2 = [0,\infty] \times \{\infty\}$ ,  $B_3 = \{\infty\} \times [-\infty,\infty]$  and  $B_4 = [0,\infty] \times \{-\infty\}$ . Identify the space  $C(\square')$  with the subalgebra of  $C([-\infty,\infty]) \oplus$  $C([0,\infty]) \oplus C([-\infty,\infty]) \oplus C([0,\infty])$  given by elements  $\Gamma = (\Gamma_1, \Gamma_2, \Gamma_3, \Gamma_4)$  which coincide at the corresponding endpoints. That is,  $\Gamma_1(\infty) = \Gamma_2(0)$ ,  $\Gamma_2(\infty) = \Gamma_3(\infty)$ ,  $\Gamma_3(-\infty) =$  $\Gamma_4(\infty)$  and  $\Gamma_4(0) = \Gamma_1(-\infty)$ . After defining the quotient map  $q' : E_{L,D_+} \to E_{L,D_+}/J_{L,D_+}$ one also obtains a short exact sequence

$$0 \to \mathcal{K}(L^2(\mathbb{R}^+)) \to E_{L,D_+} \xrightarrow{q'} C(\square') \to 0. \tag{5.4}$$

**Lemma 5.1.7.** [138] We have the isomorphism  $E_{L,D_+}/J_{L,D_+} \cong C(\square')$ .

The following index theoretic result is the basis of the interpretation of Levinson's theorem as an index theorem.

**Theorem 5.1.8** ([138, Theorem 4.4]). Suppose that  $\mathcal{H}$  is a separable Hilbert space and that  $E \subset \mathcal{B}(\mathcal{H})$  is a unital  $C^*$ -algebra with  $\mathcal{K}(\mathcal{H}) \subset E$  and  $E/\mathcal{K}(\mathcal{H}) \cong C(\mathbb{S}^1)$  (with quotient map q). Let  $U \in E/\mathcal{K}(\mathcal{H})$  be a unitary. Then there exists an integer  $m \in \mathbb{Z}$  such that for any Fredholm lift W of U with W a Fredholm partial isometry, we have the equality

$$Wind(U) = mTr([Id - W^*W] - [Id - WW^*]) = mIndex(W).$$

*Proof.* The existence of  $m \in \mathbb{Z}$  is the statement of [145, Proposition 9.2.4]. In fact by [145, Example 8.3.2] we have  $m = \pm 1$ .

Using the identification of Lemma 5.1.7 we can obtain more information about Fred-holm elements of our  $C^*$ -algebra  $E_{L,D_+}$ . The following is the basic tool for computing the index of our wave operators, first stated in this context as [97, Proposition 7].

**Theorem 5.1.9.** Suppose that  $T \in E_{L,D_+}$  is a Fredholm operator and let the image of T under the quotient map be  $q(T) = (\Gamma_1, \Gamma_2, \Gamma_3, \Gamma_4) \in E_{L,D_+}/J_{L,D_+}$ . Suppose further that the maps  $\Gamma_i$  are piecewise differentiable and the integrals

$$w_j = \frac{1}{2\pi i} \int_{B_j} \text{Tr}(\Gamma_j(y)^* \Gamma_j(y)') \,dy$$
 (5.5)

exist. Then we have the equality

$$\operatorname{Index}(T) = \sum_{j=1}^{4} w_j. \tag{5.6}$$

That the index can be computed as a winding number this way is a consequence of the Gohberg-Kreĭn Theorem (see [69, Theorem 10.1]).

# 5.2 Constructing an index pairing

We now show how the formula for the wave operator implies that the number of bound states can be computed as an index pairing between K-theory and K-homology, see [45, 76]. The index pairing we present is based on the form of the wave operator constructed in Chapter 4, and thus the case of resonances in dimensions n = 2, 4 is ruled out. In dimension n = 1 generically and in dimension n = 3 in the case of resonances some subtleties are required to define an index pairing, since  $S(0) \neq \text{Id}$ . In dimension  $n \geq 2$ , further subtleties are required to use the index pairing to deduce statements of Levinson's theorem due to the behaviour of  $S(\lambda)$  as  $\lambda \to \infty$  in trace norm.

#### 5.2.1 The spectral triple

We first recall the definition of a spectral triple.

**Definition 5.2.1.** An *odd spectral triple*  $(A, \mathcal{H}, \mathcal{D})$  is given by a Hilbert space  $\mathcal{H}$ , a dense \*-subalgebra  $A \subset \mathcal{B}(\mathcal{H})$  acting on  $\mathcal{H}$ , and a densely defined unbounded self-adjoint operator  $\mathcal{D}$  such that:

1.  $a \cdot \text{dom } \mathcal{D} \subset \text{dom } \mathcal{D}$  for all  $a \in \mathcal{A}$ , so that  $da := [\mathcal{D}, a]$  is densely defined. Moreover, da extends to a bounded operator for all  $a \in \mathcal{A}$ ;

2. 
$$a(1+\mathcal{D}^2)^{-1/2} \in \mathcal{K}(\mathcal{H})$$
 for all  $a \in \mathcal{A}$ .

Spectral triples define classes in the K-homology of the norm closure  $\overline{\mathcal{A}}$ , a  $C^*$ -algebra [16, 44]. We will produce a spectral triple for  $C_c^{\infty}((0,\infty),\mathcal{K})$  where  $\mathcal{K}$  is the compact operators on  $L^2(\mathbb{S}^{n-1})$ .

**Lemma 5.2.2.** The spaces  $L^2(\mathbb{R}, dx)$  and  $L^2(\mathbb{R}^+, dy)$  are unitarily equivalent via the map  $W: L^2(\mathbb{R}) \to L^2(\mathbb{R}^+)$  given by

$$[Wf](y) = y^{-\frac{1}{2}} f(\ln(y)) \tag{5.7}$$

with adjoint  $W^*: L^2(\mathbb{R}^+, \mathrm{d}y) \to L^2(\mathbb{R}, \mathrm{d}x)$  given by  $[W^*g](x) = \mathrm{e}^{\frac{x}{2}}g(\mathrm{e}^x)$ .

*Proof.* This is a simple check.

Lemma 5.2.3. The two spectral triples

$$\left(C_c^{\infty}(\mathbb{R}), L^2(\mathbb{R}, \mathrm{d}x), \frac{1}{i} \frac{\mathrm{d}}{\mathrm{d}x}\right) \quad and \quad \left(C_c^{\infty}(\mathbb{R}^+), L^2(\mathbb{R}^+, \mathrm{d}y), \frac{y}{i} \frac{\mathrm{d}}{\mathrm{d}y} + \frac{1}{2i}\right)$$

are unitarily equivalent.

*Proof.* Conjugating 
$$\frac{y}{i}\frac{d}{d} + \frac{1}{2i}$$
 by  $W: L^2(\mathbb{R}) \to L^2(\mathbb{R}^+)$  gives  $\frac{1}{i}\frac{d}{dx}$ .

Since  $-i\frac{d}{dx}$  defines a non-trivial (indeed generating) K-homology class for  $C_0(\mathbb{R})$ , we see that the generator of dilations on the half-line defines a non-trivial K-homology class for  $C_0(\mathbb{R}^+)$ .

Using this identification we immediately get the following in our context.

Corollary 5.2.4. The data  $(C_c^{\infty}(\mathbb{R}^+) \otimes \mathcal{K}(L^2(\mathbb{S}^{n-1})), L^2(\mathbb{R}^+) \otimes L^2(\mathbb{S}^{n-1}), D_+ \otimes \mathrm{Id})$  defines an odd spectral triple, and so a class  $[D_+]$  in odd K-homology.

#### 5.2.2 Generalised wave operators

Motivated by the form of the wave operator demonstrated in Chapter 4 we define a class of Fredholm operators, parametrised by particular unitaries, which allow us to reinterpret Levinson's theorem as an index pairing, as well as accounting for the contribution of resonances in dimension 1 and 3. As usual, we let  $D_n$  be the generator of the dilation group on  $\mathcal{H} = L^2(\mathbb{R}^n)$ . Let  $A \in \mathcal{B}(\mathcal{H})$  be a self-adjoint involution which commutes with  $D_n$  and define the operator

$$\varphi(D_n) = \operatorname{Id} + \frac{1}{2} \left( \operatorname{Id} + \tanh(\pi D_n) - i \cosh(\pi D_n)^{-1} A \right).$$
 (5.8)

If  $T \in \mathcal{B}(\mathcal{H})$  commutes with  $H_0$ , there exists a family  $\{T(\lambda)\}_{\lambda \in \mathbb{R}^+}$  of bounded operators on  $L^2(\mathbb{S}^{n-1})$  such that  $[F_0TF_0^*f](\lambda,\omega) = T(\lambda)f(\lambda,\omega)$  for all  $f \in L^2(\mathbb{R}^+) \otimes L^2(\mathbb{S}^{n-1})$  and  $(\lambda,\omega) \in \mathbb{R}^+ \otimes \mathbb{S}^{n-1}$ . We call  $T(\cdot)$  the matrix of T.

**Definition 5.2.5.** We say that a unitary  $U \in \mathcal{B}(\mathcal{H})$  is admissable if  $[U, H_0] = 0$ , the matrix of U is norm continuous in  $\lambda \in \mathbb{R}^+$  and  $U(\lambda) - \mathrm{Id} \in \mathcal{K}(\mathcal{P})$  for all  $\lambda \in \mathbb{R}^+$ . We say that U is properly admissable if in addition we have  $U(0) = \lim_{\lambda \to \infty} U(\lambda) = \mathrm{Id}$  (where the limit is taken in  $\mathcal{B}(\mathcal{P})$ ).

**Lemma 5.2.6.** Let  $U \in \mathcal{B}(\mathcal{H})$  be a properly admissable unitary. Then U defines a class  $[U] \in K_1((C_0(\mathbb{R}^+) \otimes \mathcal{K}(\mathcal{P}))^{\sim})$ .

*Proof.* By definition  $U \in ((C_0(\mathbb{R}^+) \otimes \mathcal{K}(\mathcal{P}))^{\sim})$  and thus defines a K-theory class  $[U] \in K_1((C_0(\mathbb{R}^+) \otimes \mathcal{K}(\mathcal{P}))^{\sim})$ .

**Lemma 5.2.7.** Let  $U \in \mathcal{B}(\mathcal{H})$  be a properly admissable unitary and  $K \in \mathcal{K}(\mathcal{H})$ . Then the operator  $W_U \in \mathcal{B}(\mathcal{H})$  defined by

$$W_U = \mathrm{Id} + \varphi(D_n)(U - \mathrm{Id}) + K, \tag{5.9}$$

is a Fredholm operator. Furthermore,  $F_0W_UF_0^* \in E_{L,D_+}$ .

*Proof.* We show that  $W_U^*$  is an inverse for  $W_U$  up to compacts. We compute

$$W_{U}W_{U}^{*} = (\mathrm{Id} + \varphi(D_{n}))(U - \mathrm{Id}) + K)(\mathrm{Id} + (U^{*} - \mathrm{Id})\varphi(D_{n})^{*} + K^{*})$$

$$= \mathrm{Id} + (U^{*} - \mathrm{Id})\varphi(D_{n})^{*} + \varphi(D_{n})(U - \mathrm{Id}) + \varphi(D_{n})(U - \mathrm{Id})(U^{*} - \mathrm{Id})\varphi(D_{n})^{*}$$

$$+ K(\mathrm{Id} + (U^{*} - \mathrm{Id})\varphi(D_{n})^{*} + K^{*}) + (\varphi(D_{n})(U - \mathrm{Id}) + K)K^{*}.$$

The term  $K_1 = K (\operatorname{Id} + (U^* - \operatorname{Id})\varphi(D_n)^* + K^*) + (\varphi(D_n)(U - \operatorname{Id}) + K) K^*$  is compact, since the compact operators form an ideal, so

$$W_U W_U^* = \operatorname{Id} + (U^* - \operatorname{Id})\varphi(D_n)^* + \varphi(D_n)(U - \operatorname{Id}) + \varphi(D_n)(U - \operatorname{Id})(U^* - \operatorname{Id})\varphi(D_n)^* + K_1.$$

Note that A is a self-adjoint involution which commutes with all functions of  $D_n$ . Then

$$\varphi(D_n)\varphi(D_n)^* = \frac{1}{4} \left( \text{Id} + 2 \tanh(\pi D_n) + i \cosh(\pi D_n)^{-1} A + i \tanh(\pi D_n) A^* \cosh(\pi D_n)^{-1} + \tanh(\pi D_n)^2 - i \cosh(\pi D_n)^{-1} A + \cosh(\pi D_n)^{-1} A \cosh(\pi D_n)^{-1} A^* - i \cosh(\pi D_n)^{-1} A \tanh(\pi D_n) \right)$$

$$= \frac{1}{4} \left( \text{Id} + 2 \tanh(\pi D_n) + \tanh(\pi D_n)^2 + \cosh(\pi D_n)^{-2} \right)$$

$$= \frac{1}{2} \left( \text{Id} + \tanh(\pi D_n) \right),$$

where in the last line we have used the equality  $\cosh(y)^{-2} + \tanh(y)^2 = 1$ . Since  $\cosh(\cdot)^{-2} \in L^2(\mathbb{R})$  and  $U(0) = \lim_{\lambda \to \infty} U(\lambda) = \operatorname{Id}$  (where the limit is taken in  $\mathcal{B}(\mathcal{P})$ ), Lemma 4.2.6 shows that  $\cosh(D_n)^{-1}(U - \operatorname{Id}) \in \mathcal{K}(L^2(\mathbb{R}^n))$ . Thus, up to compacts, we have

$$\varphi(D_n)(U - \mathrm{Id}) = \varphi(D_n)^*(U - \mathrm{Id}) = \varphi(D_n)\varphi(D_n)^*(U - \mathrm{Id})$$
(5.10)

and an analogous statement for  $U^*$ . We can then use commutators to compute that

$$W_{U}W_{U}^{*} = \operatorname{Id} + (U^{*} - \operatorname{Id})\varphi(D_{n})^{*} + \varphi(D_{n})(U - \operatorname{Id}) + \varphi(D_{n})(U - \operatorname{Id})(U^{*} - \operatorname{Id})\varphi(D_{n})^{*} + K_{1}$$

$$= \operatorname{Id} + \varphi(D_{n})^{*}(U^{*} - \operatorname{Id}) + [U^{*}, \varphi(D_{n})^{*}] + \varphi(D_{n})(U - \operatorname{Id})$$

$$+ \varphi(D_{n})\varphi(D_{n})^{*}(U - \operatorname{Id})(U^{*} - \operatorname{Id}) + \varphi(D_{n})(U - \operatorname{Id})[U^{*}, \varphi(D_{n})^{*}]$$

$$+ \varphi(D_{n})[U, \varphi(D_{n})^{*}](U^{*} - \operatorname{Id}) + K_{1}$$

$$= \operatorname{Id} + \varphi(D_{n})^{*}(U^{*} - \operatorname{Id}) + \varphi(D_{n})(U - \operatorname{Id}) + \varphi(D_{n})\varphi(D_{n})^{*}(U - \operatorname{Id})(U^{*} - \operatorname{Id}) + K_{2}.$$

The term  $K_2 = K_1 + [U^*, \varphi(D_n)^*] + \varphi(D_n)(U - \operatorname{Id})[U^*, \varphi(D_n)^*] + \varphi(D_n)[U, \varphi(D_n)^*](U^* - \operatorname{Id})$  is compact by the arguments of [8, Theorem 4.1.10]. Noting the multiplicative identity

$$(U - \operatorname{Id})(U^* - \operatorname{Id}) = 2\operatorname{Id} - U - U^*$$
 we have

$$W_{U}W_{U}^{*} = \operatorname{Id} + \varphi(D_{n})^{*}(U^{*} - \operatorname{Id}) + \varphi(D_{n})(U - \operatorname{Id}) + \varphi(D_{n})\varphi(D_{n})^{*}(U - \operatorname{Id})(U^{*} - \operatorname{Id}) + K_{2}$$

$$= \operatorname{Id} + \varphi(D_{n})(U^{*} - \operatorname{Id} - U - \operatorname{Id} + (U - \operatorname{Id})(U^{*} - \operatorname{Id})) + K_{2} + K_{3}$$

$$= \operatorname{Id} + K_{2} + K_{3},$$

where the term  $K_3 = 2iA \cosh(\pi D_n)^{-1}(U^* - \mathrm{Id}) + iA \cosh(\pi D_n)^{-1}(U - \mathrm{Id})(U^* - \mathrm{Id})$  is compact by Equation (5.10). Thus,  $W_U^*$  is a right inverse for  $W_U$  up to compacts. A similar calculation shows that  $W_U^*$  is a left inverse, so  $W_U$  is Fredholm.

Since the composition of Fredholm operators is Fredholm, and the composition of two properly admissable unitaries is itself a properly admissable unitary, it is natural to ask whether there is a product rule for operators of the form  $W_U$ . The following result shows that this is the case.

**Lemma 5.2.8.** Let  $U_1, U_2 \in \mathcal{B}(\mathcal{H})$  be properly admissable unitaries. Defining, for j = 1, 2, operators  $W_{U_j}$  by Equation (5.9), we have  $W_{U_1}W_{U_2} = W_{U_1U_2}$ , modulo compacts.

*Proof.* We compute

$$W_{U_1}W_{U_2} = (\mathrm{Id} + \varphi(D_n)(U_1 - \mathrm{Id}) + K_1) (\mathrm{Id} + \varphi(D_n)(U_2 - \mathrm{Id}) + K_2)$$

$$= \mathrm{Id} + \varphi(D_n)(U_1 - \mathrm{Id}) + K_1 + \varphi(D_n)(U_2 - \mathrm{Id}) + \varphi(D_n)(U_1 - \mathrm{Id})\varphi(D_n)(U_2 - \mathrm{Id})$$

$$+ K_1\varphi(D_n)(U_2 - \mathrm{Id}) + (\mathrm{Id} + \varphi(D_n)(U_1 - \mathrm{Id}) + K_1) K_2.$$

Define  $K_3 = K_1 + K_1 \varphi(D_n)(U_2 - \operatorname{Id}) + (\operatorname{Id} + \varphi(D_n)(U_1 - \operatorname{Id}) + K_1) K_2$ , which is compact. Then,

$$W_{U_{1}}W_{U_{2}} = \operatorname{Id} + \varphi(D_{n})(U_{1} - \operatorname{Id}) + \varphi(D_{n})(U_{2} - \operatorname{Id}) + \varphi(D_{n})(U_{1} - \operatorname{Id})\varphi(D_{n})(U_{2} - \operatorname{Id}) + K_{3}$$

$$= \operatorname{Id} + \varphi(D_{n})(U_{1} - \operatorname{Id}) + \varphi(D_{n})(U_{2} - \operatorname{Id}) + \varphi(D_{n})^{2}(U_{1} - \operatorname{Id})(U_{2} - \operatorname{Id}) + K_{3}$$

$$+ \varphi(D_{n})[U_{1}, \varphi(D_{n})](U_{2} - \operatorname{Id})$$

$$= \operatorname{Id} + \varphi(D_{n})(U_{1} - \operatorname{Id}) + \varphi(D_{n})(U_{2} - \operatorname{Id}) + \varphi(D_{n})(U_{1} - \operatorname{Id})(U_{2} - \operatorname{Id}) + K_{3}$$

$$+ \varphi(D_{n})[U_{1}, \varphi(D_{n})](U_{2} - \operatorname{Id}) + (\varphi(D_{n})^{2} - \varphi(D_{n}))(U_{1} - \operatorname{Id})(U_{2} - \operatorname{Id}).$$

Note that by [8, Theorem 4.1.10] the commutator  $[U_1, \varphi(D_n)]$  is compact. Noting that  $\varphi(D_n)^2 - \varphi(D_n) = -\frac{1}{2}(i\cosh(\pi D_n)^{-1}\tanh(\pi D_n)A + 2\cosh(\pi D_n)^{-2})$  we find, by similar arguments to the proof of Lemma 5.2.7, that  $(\varphi(D_n)^2 - \varphi(D_n))(U_1 - \mathrm{Id})(U_2 - \mathrm{Id})$  is compact also. Hence,

$$W_{U_1}W_{U_2} = \text{Id} + \varphi(D_n)(U_1 - \text{Id}) + \varphi(D_n)(U_2 - \text{Id}) + \varphi(D_n)(U_1 - \text{Id})(U_2 - \text{Id}) + K_4$$
  
= Id + \varphi(D\_n)(U\_1U\_2 - \text{Id}) + K\_4,

where  $K_4$  is compact. Thus,  $W_{U_1}W_{U_2} = W_{U_1U_2}$  modulo compacts as claimed.

Lemma 5.2.8 can be extended to show that we have a partially defined product rule also.

**Lemma 5.2.9.** Let  $U_1, U_2 \in \mathcal{B}(\mathcal{H})$  be admissable unitaries with  $U_1(0) = U_2(0)$  and  $\lim_{\lambda \to \infty} U_1(\lambda) = \lim_{\lambda \to \infty} U_2(\lambda)$  (with the limits taken in  $\mathcal{B}(\mathcal{P})$ ). Defining, for j = 1, 2, operators  $W_{U_j}$  by Equation (5.9), we have  $W_{U_1} = W_{U_1U_2^*}W_{U_2}$ , modulo compacts.

*Proof.* Our assumptions show that the composition  $U_1U_2^*$  is a properly admissable unitary. Let

$$W_{U_1U_2^*} = \operatorname{Id} + \varphi(D_n)(U_1U_2^* - \operatorname{Id}) + K_{12}, \quad \text{and} \quad W_{U_2} = \operatorname{Id} + \varphi(D_n)(U_2 - \operatorname{Id}) + K_2.$$

Define  $K_3 = K_2 + \varphi(D_n)(U_1U_2^* - \operatorname{Id})K_2 + K_{12}(\operatorname{Id} + \varphi(D_n)(U_2 - \operatorname{Id}) + K_2)$ , a compact operator. Then,

$$W_{U_1U_2^*}W_{U_2} = \operatorname{Id} + \varphi(D_n)(U_2 - \operatorname{Id}) + \varphi(D_n)(U_1U_2^* - \operatorname{Id}) + \varphi(D_n)(U_1U_2^* - \operatorname{Id})\varphi(D_n)(U_2 - \operatorname{Id}) + K_3$$

$$= \operatorname{Id} + \varphi(D_n)(U_2 - \operatorname{Id}) + \varphi(D_n)(U_1U_2^* - \operatorname{Id}) + \varphi(D_n)^2(U_1U_2^* - \operatorname{Id})(U_2 - \operatorname{Id}) + \varphi(D_n)[U_1U_2^*, \varphi(D_n)](U_2 - \operatorname{Id}) + K_3$$

$$= \operatorname{Id} + \varphi(D_n)(U_2 - \operatorname{Id}) + \varphi(D_n)(U_1U_2^* - \operatorname{Id}) + \varphi(D_n)(U_1U_2^* - \operatorname{Id})(U_2 - \operatorname{Id}) + (\varphi(D_n)^2 - \varphi(D_n))(U_1U_2^* - \operatorname{Id})(U_2 - \operatorname{Id}) + \varphi(D_n)[U_1U_2^*, \varphi(D_n)](U_2 - \operatorname{Id}) + K_3.$$

Since  $U_1U_2^*$  is properly admissable, the proof of Lemma 5.2.8 shows that  $(\varphi(D_n)^2 - \varphi(D_n))(U_1U_2^* - \operatorname{Id})(U_2 - \operatorname{Id})$  and  $\varphi(D_n)[U_1U_2^*, \varphi(D_n)](U_2 - \operatorname{Id})$  define compact operators. Thus,

$$W_{U_1U_2^*}W_{U_2} = \operatorname{Id} + \varphi(D_n)(U_2 - \operatorname{Id}) + \varphi(D_n)(U_1U_2^* - \operatorname{Id}) + \varphi(D_n)(U_1U_2^* - \operatorname{Id})(U_2 - \operatorname{Id}) + K_4$$

$$= \operatorname{Id} + \varphi(D_n)(U_1 - \operatorname{Id}) + K_4$$

$$= W_{U_1},$$

as claimed.  $\Box$ 

The following useful property shows that even if neither of the unitaries  $U_1$  and  $U_2$  are properly admissable, we may still be able to deduce the Fredholm property of  $W_{U_2}$  given that of  $W_{U_1}$  and that both are close enough in some sense.

Corollary 5.2.10. Let  $U_1, U_2 \in \mathcal{B}(\mathcal{H})$  be admissable unitaries with  $U_1(0) = U_2(0)$  and  $\lim_{\lambda \to \infty} U_1(\lambda) = \lim_{\lambda \to \infty} U_2(\lambda)$  (with the limits taken in  $\mathcal{B}(\mathcal{P})$ ). Define, for j = 1, 2, operators  $W_{U_j}$  by Equation (5.9). Then if one of the  $W_{U_j}$  is Fredholm, so is the other.

*Proof.* Suppose, without loss of generality, that  $W_{U_2}$  is Fredholm. Then since the unitary  $U_1U_2^*$  is properly admissable, we have  $W_{U_1U_2^*}$  is Fredholm by Lemma 5.2.7. Since the composition of Fredholm operators is also Fredholm, we deduce from Lemma 5.2.9 that  $W_{U_1}$  is Fredholm also.

There is far more underlying algebraic structure to be analysed here. Let

$$G = \{U \in \mathcal{B}(\mathcal{H}) : U \text{ is properly admissable}\}.$$

Define the Calkin algebra of  $\mathcal{H}$  by  $Q(\mathcal{H}) = \mathcal{B}(\mathcal{H})/\mathcal{K}(\mathcal{H})$ . Define the map  $\eta: G \to Q(\mathcal{H})$  by  $\eta(U) = [W_U]$ . We have the following algebraic relationship.

**Lemma 5.2.11.** The set G is a subgroup of  $\mathcal{U}(\mathcal{H})$ , the set  $\eta(G)$  is a group under composition and  $\eta: G \to \eta(G)$  is a group homomorphism.

*Proof.* The group operation and identity of G are inherited from  $\mathcal{U}(\mathcal{H})$ , so we only need to check that G is closed under composition and inverses. Compute for  $U_1, U_2 \in G$  that

$$U_1U_2 - \text{Id} = (U_1U_2 - U_1 - U_2 + \text{Id}) + U_1 + U_2 - 2\text{Id}$$
  
=  $(U_1 - \text{Id})(U_2 - \text{Id}) + (U_1 - \text{Id}) + (U_2 - \text{Id}),$ 

which is compact since  $U_1, U_2 \in G$  and so  $U_1U_2 \in G$ . The statement for inverses follows from the fact that  $U^{-1} = U^*$ . That the set  $\eta(G)$  is a group follows from Lemma 5.2.8. To see that  $\eta$  is a group homomorphism, we again use Lemma 5.2.8 to see that

$$\eta(U_1U_2) = [W_{U_1U_2}] = [W_{U_1}W_{U_2}] = [W_{U_1}][W_{U_2}] = \eta(U_1)\eta(U_2).$$

That  $\eta$  preserves inverses follows in a similar manner.

The index map Index:  $\operatorname{Fred}(Q(\mathcal{H})) \to \mathbb{Z}$  defines a map from Fredholm lifts of unitaries in  $Q(\mathcal{H})$  to  $\mathbb{Z}$ . When considered on equivalence classes, we obtain a group homomorphism Index:  $\eta(G) \to \mathbb{Z}$  and thus can form the quotient group  $G' = \eta(G)/\operatorname{Ker}(\operatorname{Index})$ .

**Lemma 5.2.12.** We have isomorphisms  $G' \cong \pi_1(\mathcal{U}(\mathcal{P})) \cong K_0(\mathcal{K})$ , where  $\pi_1(\mathcal{U}(\mathcal{P}))$  denotes the connected components of the unitary group on  $\mathcal{P}$ .

#### 5.2.3 The index pairing

Corollary 5.2.4 and Lemma 5.2.6 tell us that we can pair the classes  $[D_+]$  and [U] to obtain an integer. See [76, Section 8.7] for details. Sadly, the fact that our algebra  $C_0(\mathbb{R}^+) \otimes \mathcal{K}$ is nonunital (both  $C_0(\mathbb{R}^+)$  and  $\mathcal{K}!$ ), we need to be careful about applying index pairing formulae. See [43, Section 2.3] for a description of the problems. In [43, Section 2.7] it was shown that for spectral triples  $(\mathcal{A}, \mathcal{H}, \mathcal{D})$  over a nonunital algebra, the index pairing with the class of a unitary  $U \in \mathcal{A}$  can be computed as  $\operatorname{Index}(PUP - (\operatorname{Id} - P))$  where  $P = \chi_{[0,\infty)}(\mathcal{D})$  is the non-negative spectral projection of  $\mathcal{D}$ .

In our setting, however, we will be using an approximation of the non-positive spectral projection of  $D_+ \otimes \text{Id}$  and we need to show that we still get a Fredholm operator.

**Lemma 5.2.13.** If  $B = GG^*$  is invertible modulo compacts then G is Fredholm.

*Proof.* Write B = A + K where A is invertible and K is compact. Then  $GG^*A^{-1} = BA^{-1} = \mathrm{Id} + KA^{-1}$ . So G is Fredholm with approximate right inverse  $G^*A^{-1}$ .

**Theorem 5.2.14.** Let  $U \in \mathcal{B}(\mathcal{H})$  be a properly admissable unitary and define  $W_U$  by Equation (5.9). Let  $[D_+]$  be defined by Corollary 5.2.4. Then

$$\langle [U], [D_+] \rangle = -\text{Index}(W_U). \tag{5.11}$$

*Proof.* We know from [76, Section 8.7] that the pairing can be computed as

$$\langle [U], [D_+] \rangle = \operatorname{Index}(PUP - (\operatorname{Id} - P)) = \operatorname{Index}(PUP + (\operatorname{Id} - P)),$$

where P is the non-negative spectral projection for  $D_+ \otimes \operatorname{Id}$ . The second equality follows from the homotopy  $t \mapsto e^{i\pi(1-t)}(\operatorname{Id} - P)$ . We note that the pairing can also be computed using the non-positive spectral projection  $P_-$  for  $D_+ \otimes \operatorname{Id}$ , at the expense of a minus sign, since  $[2P_+ - 1] = -[2P_- - 1] \in K^1(C_0(\mathbb{R}^+, \mathcal{K}))$ . The result is

$$\langle [U], [D_+] \rangle = -\text{Index}(P_-UP_- - (\text{Id} - P_-)) = -\text{Index}(P_-UP_- + (\text{Id} - P_-)).$$

We will use this second form, and for convenience drop the subscript '-'. To compute the pairing, we consider the wave operator in the spectral representation. Recall that the operator  $D_n$  satisfies  $F_0D_nF_0^* = -2(D_+ \otimes \text{Id})$ .

So we need to be able to approximate P with the operator  $\varphi(-2D_+)\otimes \mathrm{Id}$ . We initially work with  $D_n$  for convenience. Let  $T=\varphi(D_n)$  and  $t=\tanh{(\pi D_n)}$  for simplicity. A quick computation shows that

$$TT^* = T^*T = \frac{1}{2}(\mathrm{Id} + t).$$

Starting from  $W_U$  there is a compact operator K such that

$$W_U = \text{Id} - \frac{1}{2}(\text{Id} + t) + TUT^* + K.$$

We can also find a compact  $K_1$  such that

$$\frac{1}{2}(\operatorname{Id} - t)TU^*T^* = \frac{1}{2}(\operatorname{Id} - t)TT^*U^* + \frac{1}{2}(\operatorname{Id} - t)[T, U^*]T^* 
= \frac{1}{4}(\operatorname{Id} - t^2)U^* + \frac{1}{2}(\operatorname{Id} - t)[T, U^*]T^* 
= \frac{1}{4}(\operatorname{Id} - t^2) + \frac{1}{4}(\operatorname{Id} - t^2)(U^* - \operatorname{Id}) + \frac{1}{2}(\operatorname{Id} - t)[T, U^*]T^* 
:= \frac{1}{4}(\operatorname{Id} - t^2) + K_1.$$

Similarly we find compacts  $K_2$ ,  $K_3$  such that

$$\frac{1}{2}TUT^*(\mathrm{Id} - t) = \frac{1}{4}(\mathrm{Id} - t^2) + K_2, \quad \text{and} \quad TUT^*TU^*T^* = \frac{1}{4}(\mathrm{Id} + t)^2 + K_3.$$

Then (with all equalities modulo compacts)

$$W_U W_U^* = (\operatorname{Id} - \frac{1}{2}(\operatorname{Id} + t) + TUT^*)(\operatorname{Id} - \frac{1}{2}(\operatorname{Id} + t) + TU^*T^*) = \operatorname{Id} - \frac{1}{2}(\operatorname{Id} - t^2)$$

which is invertible and hence  $\operatorname{Id} - \frac{1}{2}(\operatorname{Id} + t) + TUT^*$  is invertible modulo compacts by Lemma 5.2.13. Now we compare  $\operatorname{Id} - \frac{1}{2}(\operatorname{Id} + t) + TUT^*$  and  $PSP + (\operatorname{Id} - P)$  to see that they have the same Fredholm index. Taking commutators we have (modulo compacts)

$$\operatorname{Id} - \frac{1}{2}(\operatorname{Id} + t) + TUT^* = \operatorname{Id} - TT^* + TUT^* = \operatorname{Id} + T(U - \operatorname{Id})T^*$$

and

$$PUP + (\mathrm{Id} - P) = \mathrm{Id} + P(U - \mathrm{Id})P.$$

Define, in the spectral representations, the operators  $\tilde{T}$  and  $\tilde{t}$  by

$$\tilde{T} = F_0 T F_0^* = \frac{1}{2} (\text{Id} - \tanh(2\pi D_+) - i \cosh(2\pi D_+)^{-1}) \otimes \text{Id}$$

and

$$\tilde{t} = F_0 t F_0^* = -\tanh\left(2\pi D_+\right) \otimes \mathrm{Id}.$$

Taking the difference, and using  $P = \frac{1}{2}(\operatorname{Id} - \operatorname{sign}(D_+))$  we have

$$(P \otimes \operatorname{Id})U(P \otimes \operatorname{Id}) + (\operatorname{Id} - (P \otimes \operatorname{Id})) - (\operatorname{Id} - \tilde{T}\tilde{T}^* + \tilde{T}U\tilde{T}^*)$$

$$= P(U - \operatorname{Id})P - \tilde{T}(U - \operatorname{Id})\tilde{T}^*$$

$$= (P - \tilde{T}\tilde{T}^*)(U - \operatorname{Id}) + \mathcal{K}(\mathcal{H})$$

$$= \frac{1}{2}(-\operatorname{sign} + \tanh)(D_+)(U - \operatorname{Id}) + \mathcal{K}(\mathcal{H}). \tag{5.12}$$

We note that the second equality, where we have commuted the exact spectral projection with U up to compacts, is justified by the arguments in [43, Section 2.7].

To complete the argument, observe that  $\frac{1}{2}(-\text{sign} + \text{tanh})$  is an  $L^2$  function, and for any compactly supported function  $\chi \in C_c(\mathbb{R}^+)$ ,  $\lambda \mapsto \chi(\lambda) \|U(\lambda) - \text{Id}\|_{\mathcal{B}(\mathcal{P})}$  is  $L^2$  as well. Hence for any such  $\chi$  we have

$$\frac{1}{2}(-\operatorname{sign} + \tanh)(D_{+})\chi(U(\cdot) - \operatorname{Id})$$

$$= \frac{1}{2}(-\operatorname{sign} + \tanh)(D_{+})\chi \|(U(\cdot) - \operatorname{Id})\|_{\mathcal{B}(\mathcal{P})} \|(U(\cdot) - \operatorname{Id})\|_{\mathcal{B}(\mathcal{P})}^{-1}(U(\cdot) - \operatorname{Id})$$

is a product of  $L^2$  functions, one of  $D_+$  and one of  $H_0$ , composed with a uniformly bounded compact operator-valued function  $\|(U(\cdot) - \operatorname{Id})\|_{\mathcal{B}(\mathcal{P})}^{-1}(U(\cdot) - \operatorname{Id})$ . Applying the standard  $f(x)g(\nabla)$  result [151, Theorem 4.1] we obtain a compact operator times a uniformly bounded operator, which is then compact. As  $\|(U(\cdot) - \operatorname{Id})\|_{\mathcal{B}(\mathcal{P})}$  is continuous and vanishes at  $\lambda = 0, \infty$ , we can take an approximate unit  $\chi_m \in C_c(\mathbb{R}^+)$  and see that

$$\frac{1}{2}(-\mathrm{sign} + \tanh)(D_+)\chi_m \|(U(\cdot) - \mathrm{Id})\|_{\mathcal{B}(\mathcal{P})}$$

converges in norm. Hence, the limit  $\frac{1}{2}(-\text{sign} + \tanh)(D_+)||(U(\cdot) - \text{Id})||$  is compact.

Since the difference (5.12) is compact, the two Fredholm operators  $W_U$  and  $P(U-\mathrm{Id})P$  have the same index.

# 5.3 The index pairing for scattering operators

We now apply the results of Section 5.2 to the wave operator  $W_{-}$  and scattering operator S in order to reinterpret Levinson's theorem as an index pairing.

**Lemma 5.3.1.** When  $S(0) = \operatorname{Id}$  the scattering operator defines an element of the odd K-theory  $[S] \in K_1(C_0(\mathbb{R}^+) \otimes \mathcal{K})$ . For suitably decaying potentials, for example those satisfying Assumption 4.3.1,  $S(0) = \operatorname{Id}$  in all dimensions except dimension 1 (generically) and dimension 3 (in the presence of resonances).

*Proof.* The claim follows from Lemma 5.2.6 and the statement that when  $S(0) = \operatorname{Id}$  we have that S defines a properly admissable unitary, since  $\lim_{\lambda \to \infty} S(\lambda) = \operatorname{Id}$  in  $\mathcal{B}(\mathcal{H})$  by Corollary 4.3.10

**Theorem 5.3.2.** Let  $H = H_0 + V$  be such that the wave operators exist, are complete and are of the form of Equation (4.1) with S(0) = Id. Let S be the corresponding scattering operator. We have the pairing

$$\langle [S], [D_+] \rangle = -\operatorname{Index}(W_-) = N,$$

where N is the number of bound states of H (eigenvalues counted with multiplicity).

*Proof.* The scattering operator is a properly admissable unitary by Lemma 5.3.1, and so the first equality is just the statement of Theorem 5.2.14. The fact that  $Index(W_{-}) = -N$  is the statement of Proposition 2.3.9

Note that the case  $S(0) \neq \text{Id}$  cannot immediately be handled by the simple pairing described in Theorem 5.3.2, requiring some subleties which we demonstrate in the next two sections.

Corollary 5.3.3. Let  $V_0$  and  $V_1$  be such that the wave operators exist, are complete and of the form of Equation (4.1). Suppose further that  $S_0(0) = S_1(0) = \text{Id}$ , where  $S_0$  and  $S_1$  denote the scattering operators for the potentials  $V_0$  and  $V_1$ . If the number of eigenstates (counted with multiplicity) for  $V_0$  differs from that of  $V_1$  then their scattering matrices are not (stably) homotopic.

Corollary 5.3.4. Let  $V_0$  and  $V_1$  be such that the wave operators exist, are complete and of the form of Equation (4.1). Suppose further that  $S_0(0) = S_1(0) = \text{Id}$ , where  $S_0$ and  $S_1$  denote the scattering operators for the potentials  $V_0$  and  $V_1$ . Consider the path  $V_t = (1-t)V_0 + tV_1$  for  $t \in [0,1]$  with corresponding scattering operators  $S_t$ . If the number of eigenstates (counted with multiplicity) for  $V_0$  differs from that of  $V_1$ , then the path  $S_t$ is not norm continuous.

In fact, Corollary 5.3.4 shows that there are a discrete number of points at which the path  $S_t$  fails to be norm continuous, corresponding to 'jumps' in the number of eigenvalues for the potential  $V_t$ . The norm holomorphy (which implies norm continuity) of the scattering operator as a function of t is discussed in [34, Theorem 4.2] where an equivalent condition to holomorphy of  $S_t$  is given. The points of failure of norm continuity in Corollary 5.3.4 are also points of failure of holomorphy in [34, Theorem 4.2]. In the case of a Rollnik class potential on  $\mathbb{R}^3$  holomorphy of the scattering matrix as a function of t has been studied in [94, Theorem 6.1] and [34, Theorem 5.2].

**Lemma 5.3.5.** Suppose that  $V_1, V_2$  satisfy Assumption 2.2.14 and define the operators  $H_j = H_0 + V_j$ . Suppose further that the scattering operators  $S_1$  and  $S_2$  are properly admissable unitaries. Then the wave operator  $W_-(H_2, H_1)$  is given by

$$W_{-}(H_2, H_1) = \mathrm{Id} + \varphi(D_n)(S_2S_1^* - \mathrm{Id}) + K$$

for a compact operator K.

*Proof.* The chain rule for wave operators (Proposition 2.1.4) gives that

$$W_{-}(H_2, H_0) = W_{-}(H_2, H_1)W_{-}(H_1, H_0).$$

Using that  $S_2$  and  $S_1$  are properly admissable unitaries and Lemma 5.2.8 we find that

$$W_{-}(H_{2}, H_{1}) = W_{-}(H_{2}, H_{0})W_{-}(H_{1}, H_{0})^{*}$$

$$= (\operatorname{Id} + \varphi(D_{n})(S_{2} - \operatorname{Id}))(\operatorname{Id} + (S_{1}^{*} - \operatorname{Id})\varphi(D_{n})^{*})$$

$$= \operatorname{Id} + \varphi(D_{n})(S_{2}S_{1}^{*} - \operatorname{Id}),$$

where all equalities are modulo compacts and we have used the inclusions  $[S_1^*, \varphi(D_n)], (S^*-\text{Id})(\varphi(D_n) - \varphi(D_n)^*) \in \mathcal{K}(\mathcal{H}).$ 

Such an equality cannot be obtained when the scattering operators are only assumed to be admissable, since we no longer have the required compactness relations.

For  $t \in [0, 1]$  define the operator  $H_t = H_0 + tV$  and suppose that  $H_1$  does not admit a zero energy resonance. The analytic Fredholm alternative gives that there is a discrete set  $(t_j)_{j \leq J} \subset [0, 1]$  such that each  $H_{t_j}$  admits a zero energy resonance. Choose a sequence  $(\tilde{t}_j)_{j \leq J+1} \subset [0, 1]$  with  $\tilde{t}_0 = 0$ ,  $\tilde{t}_{J+1} = 1$  and  $t_{j-1} < \tilde{t}_j < t_j$  for  $1 \leq j \leq J-1$ . Then each  $H_{\tilde{t}_j}$  does not admit a zero energy resonance, and Lemma 5.3.5 gives

$$W_{-}(H_{\tilde{t}_{j+1}}, H_{\tilde{t}_{j}}) = \mathrm{Id} + \varphi(D_{n})(S_{j+1}S_{j}^{*} - \mathrm{Id}) + K_{j}$$

for some compact  $K_j$ , allowing us to prove the following.

**Lemma 5.3.6.** Let V satisfy Assumption 2.2.14 and suppose that the scattering operator S is a properly admissable unitary. Then the wave operator  $W_{-}$  satisfies

$$W_{-} = \prod_{j=0}^{J} W_{-}(H_{\tilde{t}_{j+1}}, H_{\tilde{t}})$$

**Proposition 5.3.7.** Let V satisfy Assumption 4.3.1 and suppose that  $H_1 = H_0 + V$  admits a resonance. Then there exists a  $t_0 > 0$  such that for all  $t \in (1, 1+t_0)$  the operator  $H_t = H_0 + tV$  does not admit a resonance and the number of bound states N(t) of H(t) is constant for all  $t \in [1, 1+t_0)$ . Furthermore, for any  $t \in (1, 1+t_0)$  the scattering operator  $S_t$  defines a properly admissable unitary and

$$W_{-}(H_t, H_0) = W_{-}(H_t, H_1)W_{-}(H_1, H_0)$$

and  $\operatorname{Index}(W_{-}(H_t, H_0)) = \operatorname{Index}(W_{-}(H_1, H_0))$ . In particular,  $\operatorname{Index}(W_{-}(H_t, H_1)) = 0$ .

# **5.3.1** Low energy corrections in dimension n = 1

Due to the generic failure of S(0) = Id in dimension n = 1, the pairing of Theorem 5.3.2 is not applicable in dimension n = 1. In each generic and resonant case, we construct a unitary  $\sigma$  such that  $S\sigma^*$  is a properly admissable unitary and so defines a pairing by

Theorem 5.2.14. As a corollary, we obtain a new proof of the well-known Levinson's theorem [108].

The main tool in this section is the following result which allows us to compute the index of a generalised wave operator  $W_U$  by computing the winding number of the component functions described in Lemma 5.1.4.

**Lemma 5.3.8.** Let  $U \in \mathcal{B}(L^2(\mathbb{R}))$  be a properly admissable unitary such that the matrix of U is differentiable and  $|\text{Tr}(U(\cdot)^*U'(\cdot))| \in L^1(\mathbb{R}^+)$ . Let  $W_U \in \mathcal{B}(L^2(\mathbb{R}))$  be defined by

$$W_U = \mathrm{Id} + \varphi(D_1)(U - \mathrm{Id}) + K, \tag{5.13}$$

with K compact. Then  $W_U \in E_{L,D_+}$  and

$$\operatorname{Index}(W_U) = \frac{1}{2\pi i} \int_0^\infty \operatorname{Tr}(U(\lambda)^* U'(\lambda)) \, d\lambda. \tag{5.14}$$

*Proof.* This is a consequence of Proposition 5.1.9. Since  $U(0) = \operatorname{Id}$ , the only non-trivial winding number contribution around the square  $\square'$  is from the edge corresponding to  $\infty$  in  $\sigma(D_1)$ , which gives us the winding number of  $\operatorname{Id} + \varphi(\infty)(U - \operatorname{Id}) = U$ .

**Lemma 5.3.9.** Suppose that n=1 and V satisfies Assumption 2.2.14 for some  $\rho > \frac{5}{2}$  and let S be the corresponding scattering operator. Let  $\sigma \in U(L^2(\mathbb{R}))$  be an admissable unitary with  $\sigma(0) = S(0)$  and  $\lim_{\lambda \to \infty} \sigma(\lambda) = \operatorname{Id}$  (with the limit in  $\mathcal{B}(\mathcal{P})$ ). If we have  $\operatorname{Tr}(\sigma(\cdot)\sigma'(\cdot)^*) \in L^1(\mathbb{R}^+)$  then

$$\operatorname{Index}(W_{S\sigma^*}) = \frac{1}{2\pi i} \int_0^\infty \operatorname{Tr}(S(\lambda)^* S'(\lambda)) \, d\lambda + \frac{1}{2\pi i} \int_0^\infty \operatorname{Tr}(\sigma(\lambda) \sigma'(\lambda)^*) \, d\lambda. \tag{5.15}$$

*Proof.* Since  $\mathcal{B}(\mathcal{P}) = \mathcal{L}^1(\mathcal{P})$  we have  $\sigma(\lambda)^* \sigma'(\lambda) \in \mathcal{L}^1(\mathcal{P})$  for all  $\lambda \in \mathbb{R}^+$ . The function  $\lambda \mapsto \text{Tr}(S(\cdot)^* S'(\cdot))$  is integrable by Theorem 2.5.34. The Leibniz rule for matrix valued functions shows that

$$\frac{\mathrm{d}}{\mathrm{d}\lambda}(S(\lambda)\sigma(\lambda)^*) = S'(\lambda)\sigma(\lambda)^* + S(\lambda)\sigma'(\lambda)^*.$$

Multiplying by  $(S(\lambda)\sigma(\lambda)^*)^* = \sigma(\lambda)S(\lambda)^*$  and taking the trace we obtain

$$\operatorname{Tr}\left(S(\lambda)\sigma(\lambda)^*)^*\frac{\mathrm{d}}{\mathrm{d}\lambda}S(\lambda)\sigma(\lambda)^*\right) = \operatorname{Tr}\left(\sigma(\lambda)S(\lambda)^*(S'(\lambda)\sigma(\lambda)^* + S(\lambda)\sigma'(\lambda)^*)\right)$$

$$= \operatorname{Tr}\left(\sigma(\lambda)S(\lambda)^*S'(\lambda)\sigma(\lambda)^*\right) + \operatorname{Tr}\left(\sigma(\lambda)S(\lambda)^*S(\lambda)\sigma'(\lambda)^*\right)$$

$$= \operatorname{Tr}\left(S(\lambda)^*S'(\lambda)\right) + \operatorname{Tr}\left(\sigma(\lambda)\sigma'(\lambda)^*\right),$$

where we have used unitarity and cyclicity of the trace. The result follows upon integrating the traces and applying Lemma 5.3.8.

Remark 5.3.10. Equation (5.15) indicates that, in one dimension, Levinson's theorem is related to the Witten index. In particular, the contribution from the scattering operator is the Witten index of the pair  $(D_+, S^*D_+S)$ . For more details regarding this observation, we direct the reader to [37, 38, 39, 40, 42] and the short note [41].

**Definition 5.3.11.** A map  $\sigma: \mathbb{R}^+ \to M_2(\mathbb{C})$  is a generic correction if there exist differentiable functions  $\theta: \mathbb{R}^+ \to \mathbb{R}$  and  $f, g: \mathbb{R}^+ \to \mathbb{C}$  such that  $\theta$  is increasing,  $\theta(\infty) \in 2\pi\mathbb{Z}$ ,  $\theta(0) \in (2\mathbb{Z} + 1)\pi$ ,  $\theta(\infty) - \theta(0) = \pi$ ,  $g(0) = f(\infty) = 1$  and  $f(0) = g(\infty) = 0$  such that  $|f(\lambda)|^2 + |g(\lambda)|^2 = 1$  for all  $\lambda$  and

$$\sigma(\lambda) = \begin{pmatrix} f(\lambda) & g(\lambda) \\ -e^{i\theta(\lambda)} \overline{g(\lambda)} & e^{i\theta(\lambda)} \overline{f(\lambda)} \end{pmatrix}. \tag{5.16}$$

A map  $\sigma: \mathbb{R}^+ \to M_2(\mathbb{C})$  is a resonant correction if there exist differentiable functions  $\theta: \mathbb{R}^+ \to \mathbb{R}$  and  $f, g: \mathbb{R}^+ \to \mathbb{C}$  such that  $\theta$  is increasing,  $\theta(\infty), \theta(0) \in 2\pi\mathbb{Z}$  and  $\theta(\infty) - \theta(0) = 0$  with  $f(0) = 2c_+c_-$ ,  $g(0) = c_+^2 - c_-^2$ ,  $f(\infty) = 1$  and  $g(\infty) = 0$  for some  $c_+, c_- \in \mathbb{R} \setminus \{0\}$  with  $c_+^2 + c_-^2 = 1$  such that  $|f(\lambda)|^2 + |g(\lambda)|^2 = 1$  for all  $\lambda$  and

$$\sigma(\lambda) = \begin{pmatrix} f(\lambda) & g(\lambda) \\ -e^{i\theta(\lambda)} \overline{g(\lambda)} & e^{i\theta(\lambda)} \overline{f(\lambda)} \end{pmatrix}.$$
 (5.17)

For both generic and resonant corrections,  $\sigma(\lambda)$  is unitary for all  $\lambda \in \mathbb{R}^+$ , and  $\lim_{\lambda \to \infty} \sigma(\lambda) = \text{Id}$ , with the limit taken in  $\mathcal{B}(\mathcal{P})$ . To explicitly evaluate the index appearing in one dimension in terms of scattering data, we need the following result [97, Proposition 9].

**Lemma 5.3.12.** Let  $\sigma$  be a correction (generic or resonant) and define  $\Gamma_1 : \mathbb{R} \to M_2(\mathbb{C})$  by

$$\Gamma_1(y) = \text{Id} + \frac{1}{2}(1 + \tanh(\pi y) - i\cosh(\pi y)^{-1}A)(\sigma(0) - \text{Id}),$$

where [Af](x) = f(-x). Then

$$\frac{1}{2\pi i} \int_{\mathbb{R}} \text{Tr}(\Gamma_1(y)^* \Gamma_1'(y)) \, \mathrm{d}y = \begin{cases} -\frac{1}{2}, & \text{if } \sigma \text{ is a generic correction,} \\ 0, & \text{if } \sigma \text{ is a resonant correction.} \end{cases}$$

*Proof.* The proof consists of changing to a basis in which the operator  $\tanh(\pi D_1) + iA \cosh(\pi D_1)$  is diagonal, from which the computation follows. We refer to [97, Proposition 9] for the details.

We use Lemma 5.3.12 to compute directly the index of  $W_{\sigma}$ .

**Lemma 5.3.13.** Let  $\sigma$  be a generic correction. Then the operator  $W_{\sigma}$  is Fredholm and  $\operatorname{Index}(W_{\sigma}) = 0$ .

Proof. Example 2.5.35 shows that there exists a scattering matrix with  $S(0) = \begin{pmatrix} 0 & -1 \\ -1 & 0 \end{pmatrix}$  and  $\lim_{\lambda \to \infty} S(\lambda) = \operatorname{Id}$ , with the limit taken in  $\mathcal{B}(\mathcal{P})$ . By Proposition 2.3.9,  $W_- = W_S$  is Fredholm. That  $W_{\sigma}$  is Fredholm then follows from Corollary 5.2.10. To compute the index we use Lemma 5.1.9. The image  $q(W_{\sigma}) \in E_{L,D_+}/J_{L,D_+}$  is given by  $(\Gamma_1, \Gamma_2, \Gamma_3, \Gamma_4)$  with

$$\Gamma_1(y) = \text{Id} + \frac{1}{2}(1 + \tanh(\pi y) - i\cosh(\pi y)^{-1}A)(\sigma(0) - \text{Id}),$$

 $\Gamma_2(\lambda) = \sigma(\lambda)$  and  $\Gamma_3(y) = \mathrm{Id} = \Gamma_4(\lambda)$ . By Theorem 5.1.9 we find

Index(
$$W_{\sigma}$$
) =  $\frac{1}{2\pi i} \int_{\mathbb{R}} \text{Tr}(\Gamma_1(y)^* \Gamma_1'(y)) dy + \frac{1}{2\pi i} \int_0^{\infty} \text{Tr}(\sigma(\lambda)^* \sigma'(\lambda)) d\lambda$   
=  $-\frac{1}{2} + \frac{1}{2\pi} \int_0^{\infty} \theta'(\lambda) d\lambda$   
= 0,

where we have used Lemma 5.3.12 to calculate the first integral.

**Lemma 5.3.14.** Let  $\sigma$  be a resonant correction. Then the operator  $W_{\sigma}$  is Fredholm and  $\operatorname{Index}(W_{\sigma}) = 0$ .

Proof. Example 2.5.35 shows that for any  $c_+, c_- \in \mathbb{R} \setminus \{0\}$  with  $c_+^2 + c_-^2 = 1$  there exists a potential V with scattering matrix satisfying  $S(0) = \begin{pmatrix} 2c_1c_2 & c_+^2 - c_-^2 \\ c_-^2 - c_+^2 & 2c_1c_2 \end{pmatrix}$  and  $\lim_{\lambda \to \infty} S(\lambda) = \operatorname{Id}$ , with the limit taken in  $\mathcal{B}(\mathcal{P})$ . By Proposition 2.3.9,  $W_- = W_S$  is Fredholm. That  $W_{\sigma}$  is Fredholm then follows from Corollary 5.2.10. To compute the index we use Lemma 5.1.9. The image  $q(W_{\sigma}) \in E_{L,D_+}/J_{L,D_+}$  is given by  $(\Gamma_1, \Gamma_2, \Gamma_3, \Gamma_4)$  with

$$\Gamma_1(y) = \text{Id} + \frac{1}{2}(1 + \tanh(\pi y) - i\cosh(\pi y)^{-1}A)(\sigma(0) - \text{Id}),$$

 $\Gamma_2(\lambda) = \sigma(\lambda)$  and  $\Gamma_3(y) = \mathrm{Id} = \Gamma_4(\lambda)$ . Thus,

$$\operatorname{Index}(W_{\sigma}) = \frac{1}{2\pi i} \int_{\mathbb{R}} \operatorname{Tr}(\Gamma_{1}(y)^{*}\Gamma'_{1}(y) \, \mathrm{d}y + \frac{1}{2\pi i} \int_{0}^{\infty} \operatorname{Tr}(\sigma(\lambda)^{*}\sigma'(\lambda)) \, \mathrm{d}\lambda$$
$$= -\frac{1}{2} + \frac{1}{2\pi i} \int_{0}^{\infty} \theta'(\lambda) \, \mathrm{d}\lambda$$
$$= 0,$$

where again we have used Lemma 5.3.12 to calculate the first integral.

We can now use a generic or resonant correction to prove a modified index pairing in one dimension.

**Theorem 5.3.15.** Let n = 1 and suppose V satisfies Assumption 2.2.14 for some  $\rho > \frac{5}{2}$ . If  $H = H_0 + V$  has no resonances, then for any generic correction  $\sigma$  we have the pairing

$$\langle [S\sigma^*], [D_+] \rangle = -\operatorname{Index}(W_-) = N, \tag{5.18}$$

where N is the number of bound states of H (eigenvalues counted with multiplicity). If  $H = H_0 + V$  has resonances, then for any resonant correction  $\sigma$  the pairing of Equation (5.18) holds.

*Proof.* Since in both cases (generic and resonant)  $S(0) = \sigma(0)$  and  $\lim_{\lambda \to \infty} S(\lambda) = \lim_{\lambda \to \infty} \sigma(\lambda) = \text{Id}$  (with the limits taken in  $\mathcal{B}(\mathcal{P})$ ) we find that S and  $\sigma$  are admissable unitaries with  $S\sigma^*$  a properly admissable unitary. Theorem 5.3.2 then gives us the equality

$$\langle [S\sigma^*], [D_+] \rangle = -\operatorname{Index}(W_{S\sigma^*}). \tag{5.19}$$

Since both =  $W_S$  and  $W_{\sigma^*}$  are Fredholm we find that

$$\operatorname{Index}(W_{S\sigma^*}) = \operatorname{Index}(W_S) + \operatorname{Index}(W_{\sigma}) = -N + 0,$$

where we have used Proposition 2.3.9 and Lemma 5.3.13 (generic) or Lemma 5.3.14 (resonant) to evaluate the indices. The statement follows by noting that  $W_{-} = W_{S}$  (modulo compacts) by Theorem 4.0.1.

We also obtain an alternative proof of the integral form of Levinson's theorem in one dimension.

**Theorem 5.3.16.** Let n=1 and suppose that V satisfies Assumption 2.2.14 for some  $\rho > \frac{5}{2}$ . Then the number of bound states of  $H = H_0 + V$  is given by

$$N = \begin{cases} -\frac{1}{2\pi i} \int_0^\infty \text{Tr}(S(\lambda)^* S'(\lambda)) \, d\lambda, & \text{if there exists a resonance,} \\ -\frac{1}{2\pi i} \int_0^\infty \text{Tr}(S(\lambda)^* S'(\lambda)) \, d\lambda + \frac{1}{2} & \text{otherwise.} \end{cases}$$
(5.20)

*Proof.* Theorem 5.3.15 tells us that for any generic (or resonant if necessary) correction  $\sigma$  we have  $N = -\text{Index}(W_{S\sigma^*})$ . Combining with Lemma 5.3.9 we find that

$$N = -\operatorname{Index}(W_{S\sigma^*}) = -\frac{1}{2\pi i} \int_0^\infty \operatorname{Tr}(S(\lambda)^* S'(\lambda)) \, \mathrm{d}\lambda + \frac{1}{2\pi i} \int_0^\infty \operatorname{Tr}(\sigma(\lambda)^* \sigma'(\lambda)) \, \mathrm{d}\lambda$$
$$= \begin{cases} -\frac{1}{2\pi i} \int_0^\infty \operatorname{Tr}(S(\lambda)^* S'(\lambda)) \, \mathrm{d}\lambda, & \text{if there exists a resonance,} \\ -\frac{1}{2\pi i} \int_0^\infty \operatorname{Tr}(S(\lambda)^* S'(\lambda)) \, \mathrm{d}\lambda + \frac{1}{2} & \text{otherwise.} \end{cases}$$

Here, the  $\sigma$  integral has been evaluated explicitly to  $\frac{1}{2}$  in the generic case and 0 in the resonant case.

Remark 5.3.17. Theorem 5.3.16 allows us to completely characterise the spectral shift function in dimension n = 1. The result is

$$\xi(\lambda) = -\sum_{k=1}^{K} M(\lambda_k) \chi_{[\lambda_k, \infty)}(\lambda) - \frac{1}{2} (1 - M_R(0)) \chi_{[0, \infty)}(\lambda)$$
$$- \frac{1}{2\pi i} \chi_{[0, \infty)}(\lambda) \int_0^{\lambda} \operatorname{Tr}(S(\mu)^* S'(\mu)) d\mu,$$

where  $M_R(0) = 1$  if there exists a zero-energy resonance and zero otherwise.

Corollary 5.3.18. Suppose that n = 1 and V satisfies Assumption 2.2.14 for some  $\rho > \frac{5}{2}$ . Then we have

$$\xi(0+) = -N - \frac{1}{2}(1 - M_R(0)).$$

# 5.3.2 Low and high energy corrections in dimension n = 3

In dimension n=3 the pairing of Theorem 5.3.2 fails in the presence of resonances. In this section we emulate the ideas of the previous section to construct a resonant correction from which an index pairing can be proved. Due to the failure of  $S(\lambda)$  to converge as  $\lambda \to \infty$  in the trace norm, additional corrections are required to pass from the index pairing to an integral formula for the number of bound states. We construct an explicit high energy correction to S obtain a more useful representative of  $[W_-]$  and as a corollary we obtain a new proof of the well known Levinson's theorem in dimension n=3 [31].

Define the function

$$\varphi(D_3) = \frac{1}{2} \left( \text{Id} + \tanh(\pi D_3) - i \cosh(\pi D_3)^{-1} \right).$$
 (5.21)

We are interested in the resonant case, and so we are interested in finding a simple operator  $\sigma \in U(L^2(\mathbb{R}^3))$  such that in the spectral representation we have  $\sigma(0) = S(0)$  and  $\lim_{\lambda \to \infty} \sigma(\lambda) = \mathrm{Id}$ , with the limit taken in  $\mathcal{B}(\mathcal{P})$ .

**Definition 5.3.19.** Let n=3. Then for any increasing differentiable  $\theta: \mathbb{R}^+ \to \mathbb{R}$  with  $\theta(\infty) = \pi$  and  $\theta(0) = 0$  define

$$\sigma(\lambda) = \operatorname{Id} - (1 - e^{i\theta(\lambda)})P_s, \tag{5.22}$$

where  $P_s$  denotes the projection onto the spherical harmonic subspace of order zero. We call  $\sigma$  a resonant correction. A particular choice of  $\theta$  is given by  $\theta = 2 \tan^{-1}(\cdot)$ .

Computation shows that for all  $\lambda \in \mathbb{R}^+$  we have  $\sigma(\lambda)$  is unitary and  $\sigma(\lambda)^* \sigma'(\lambda) \in \mathcal{L}^1(\mathcal{P})$ , and that

$$\frac{1}{2\pi i} \int_0^\infty \text{Tr}(\sigma(\lambda)^* \sigma'(\lambda)) \, d\lambda = \frac{1}{2}.$$

With such a  $\sigma$  we obtain the following pairing result.

**Theorem 5.3.20.** Suppose that n = 3 and V satisfies Assumption 2.2.14 for some  $\rho > 5$ . Suppose further that the operator  $H = H_0 + V$  admits a zero energy resonance. For any resonant correction  $\sigma$  we have

$$\langle [S\sigma^*], [D_+] \rangle = -\operatorname{Index}(W_-) = N,$$

where N is the number of bound states of H (eigenvalues counted with multiplicity).

*Proof.* The operator  $S\sigma^*$  is a properly admissable unitary by construction, and so the first equality is just the statement of Theorem 5.2.14. That  $Index(W_-) = -N$  follows from Proposition 2.3.9. That  $Index(W_{\sigma}) = 0$  is a consequence of Theorem 5.1.9.

While Theorem 5.3.20 demonstrates that the index of  $W_{-}$  can be obtained as an index pairing, it does not supply a method for obtaining computational formulae. To do so requires more subleties of the scattering operator. It was noted in Remark 4.3.11 that while  $S(\lambda) \to \operatorname{Id}$  as  $\lambda \to \infty$  in  $\mathcal{B}(\mathcal{P})$ , this convergence does not happen in  $\mathcal{L}^{1}(\mathcal{P})$ . To remedy this we must find a representative of the class  $[W_{S\sigma^{*}}]$  which has better trace class properties at infinity. This motivates the following definition.

**Definition 5.3.21.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 6$  and let S be the scattering operator for  $H = H_0 + V$ . Let  $\beta : \mathbb{R}^+ \to \mathcal{B}(\mathcal{P})$  be a once continuously differentiable unitary valued function such that  $\beta(\lambda) - \mathrm{Id} \in \mathcal{L}^1(\mathcal{P})$  for all  $\lambda \in \mathbb{R}^+$  and  $\beta(0) = \mathrm{Id}$  and  $\lim_{\lambda \to \infty} \beta(\lambda) = \mathrm{Id}$  in  $\mathcal{B}(\mathcal{P})$ . Then  $\mathrm{Det}(\beta(\lambda)) \in \mathbb{T}$  exists for all  $\lambda \in \mathbb{R}^+$  and we say that  $\beta$  is a high energy correction for S if

$$\lim_{\lambda \to \infty} \text{Det}(S(\lambda)\beta(\lambda)) = 1, \tag{5.23}$$

and  $\operatorname{Index}(W_{\beta}) = 0$ .

Note that the assumption  $\lim_{\lambda\to\infty} (S(\lambda)\beta(\lambda) - \mathrm{Id}) = 0$  in  $\mathcal{L}^1(\mathcal{H})$  implies Equation (5.23).

The following result is applicable to all dimensions  $n \geq 2$ . As usual we decompose the potential into V = vUv with  $v = |v|^{\frac{1}{2}}$  and U = sign(V).

**Lemma 5.3.22.** Suppose that  $n \geq 2$  and  $q_1, q_2 \in C_c^{\infty}(\mathbb{R}^n)$  with  $q_1q_2 = V$  and corresponding scattering operator S and fix  $0 \neq \tilde{\chi} \in C_c^{\infty}(\mathbb{R}^+)$  with  $\tilde{\chi}(\lambda) = 0$  for  $\lambda \leq 1$ . Define

 $\chi: \mathbb{R}^+ \to \mathbb{R} \text{ for } \lambda \in \mathbb{R}^+ \text{ by }$ 

$$\chi(\lambda) = \frac{\int_0^{\lambda} \tilde{\chi}(u) \, \mathrm{d}u}{\int_0^{\infty} \tilde{\chi}(u) \, \mathrm{d}u}.$$

If n=2,3 let p=2 and if  $n\geq 4$  let  $p\geq n$ . For  $\lambda>0$  define the self-adjoint operator  $\tilde{A}(\lambda)\in\mathcal{B}(\mathcal{P})$  by

$$\tilde{A}(\lambda) = -2\pi\Gamma_0(\lambda)q_2 \left( \sum_{\ell=1}^{p-1} \sum_{j=0}^{\ell-1} \frac{(-1)^{\ell}}{\ell} \left( q_1 R_0(\lambda + i0) q_2 \right)^j \left( q_1 R_0(\lambda - i0) q_2 \right)^{\ell-j-1} \right) q_1 \Gamma_0(\lambda)^*,$$

For  $\lambda \in \mathbb{R}^+$  let  $A(\lambda) = \chi(\lambda^{\frac{1}{2}})\tilde{A}(\lambda)$ . Then the unitary operator  $\beta(\lambda) = e^{iA(\lambda)}$  satisfies  $\operatorname{Index}(W_{\beta}) = 0$  and

$$\lim_{\lambda \to \infty} \text{Det}(S(\lambda)\beta(\lambda)) = 1. \tag{5.24}$$

*Proof.* By construction we have that  $A(\lambda)$  is self-adjoint for all  $\lambda \in \mathbb{R}^+$ . We have A(0) = 0 and the norm limit

$$\lim_{\lambda \to \infty} A(\lambda) = 0$$

by an application of the estimate (4.20) and [120, Theorem 1]. By Lemma 2.4.20 we have that  $A(\lambda)$  is a trace-class operator and by cyclicity of the trace we have

$$\operatorname{Tr}(A(\lambda)) = i\chi(\lambda^{\frac{1}{2}}) \sum_{\ell=1}^{p-1} \frac{(-1)^{\ell}}{\ell} \operatorname{Tr}\left( (q_1 R_0(\lambda + i0) q_2)^{\ell} - (q_1 R_0(\lambda - i0) q_2)^{\ell} \right).$$

As a consequence of Hölder's inequality for Schatten ideals,  $\beta(\lambda) - \mathrm{Id} \in \mathcal{L}^1(\mathcal{P})$  and so  $\beta$  defines a properly admissable unitary. Thus  $\mathrm{Det}(\beta(\lambda)) = \mathrm{e}^{i\mathrm{Tr}(A(\lambda))}$  and so by Lemma 2.5.14 we find that

$$Det(S(\lambda))Det(\beta(\lambda)) \frac{Det_{p}(Id + q_{1}R_{0}(\lambda + i0)q_{2})}{Det_{p}(Id + q_{1}R_{0}(\lambda - i0)q_{2})} 
= \exp\left(\sum_{\ell=1}^{p-1} \frac{(-1)^{\ell}}{\ell} Tr\left((q_{1}R_{0}(\lambda + i0)q_{2})^{\ell} - (q_{1}R_{0}(\lambda - i0)q_{2})^{\ell}\right) + i Tr(A(\lambda))\right) 
= \exp\left((1 - \chi(\lambda^{\frac{1}{2}})) \sum_{\ell=1}^{p-1} \frac{(-1)^{\ell}}{\ell} Tr\left((q_{1}R_{0}(\lambda + i0)q_{2})^{\ell} - (q_{1}R_{0}(\lambda - i0)q_{2})^{\ell}\right)\right).$$

An application of Theorem 2.5.29 and Lemma 2.5.16 gives that

$$\lim_{\lambda \to \infty} \operatorname{Det}(S(\lambda)\beta(\lambda)) = 1.$$

To see that  $\operatorname{Index}(W_{\beta}) = 0$ , we consider for  $t \in [0, 1]$  the homotopy

$$A_t(\lambda) = 2\pi \chi((1-t)\lambda^{\frac{1}{2}})\tilde{A}(\lambda).$$

The path  $A_t(\lambda)$  defines a norm-continuous path in  $\mathcal{B}(\mathcal{P})$  from  $A_0(\lambda) = A(\lambda)$  to  $A_1(\lambda) = 0$ . Defining the path  $A_t = A_t(L) \in \mathcal{B}(\mathcal{H})$  we obtain a norm continuous path in  $\mathcal{B}(\mathcal{H})$  from A to 0. To see this, fix  $t_1, t_2 \in [0, 1)$  and define  $d = (\min\{(1 - t_1)^{-1}, (1 - t_2)^{-1}\})^2 \geq 1$ . Then for  $g \in \mathcal{H}_{spec}$  we find

$$\begin{aligned} &||(A_{t_2}(\cdot) - A_{t_1}(\cdot))g||^2_{\mathcal{H}_{spec}} \\ &= \int_0^\infty \int_{\mathbb{S}^{n-1}} |[(A_{t_2}(\lambda) - A_{t_1}(\lambda))g(\lambda, \cdot)](\omega)|^2 d\omega d\lambda \\ &= \int_d^\infty \int_{\mathbb{S}^{n-1}} |[(A_{t_2}(\lambda) - A_{t_1}(\lambda))g(\lambda, \cdot)](\omega)|^2 d\omega d\lambda \\ &\leq \int_d^\infty |\chi(\lambda^{\frac{1}{2}}(1 - t_2)) - \chi(\lambda^{\frac{1}{2}}(1 - t_1))|^2 \int_{\mathbb{S}^{n-1}} |[\tilde{A}(\lambda)g(\lambda, \cdot)](\omega)|^2 d\omega d\lambda \\ &\leq C|t_2 - t_1|^2 \int_d^\infty \lambda \int_{\mathbb{S}^{n-1}} |[\tilde{A}(\lambda)g(\lambda, \cdot)](\omega)|^2 d\omega d\lambda. \end{aligned}$$

It remains to estimate  $|[\tilde{A}(\lambda)g(\lambda,\cdot)](\omega)|$ . We recall from [120, Theorem 1] that for sufficiently large  $\lambda$  we have

$$\left| \left| q_1 R_0 (\lambda \pm i0)^j q_2 \right| \right| = O(\lambda^{-\frac{j}{2}})$$

and so we may define

$$K = \sup \left\{ 2\pi \sum_{\ell=1}^{p-1} \sum_{j=0}^{\ell-1} \frac{1}{\ell} \left| \left| (q_1 R_0(\lambda + i0) q_2)^j (q_1 R_0(\lambda - i0) q_2)^{\ell-j-1} \right| \right| : \lambda \in [d, \infty) \right\}.$$

Using twice the estimate (4.20) we obtain

$$\int_{\mathbb{S}^{n-1}} |[\tilde{A}(\lambda)g(\lambda,\cdot)](\omega)|^2 d\omega = ||A(\lambda)g(\lambda,\cdot)||_{L^2(\mathbb{S}^{n-1})}^2 \\
\leq 2\pi\lambda^{-\frac{1}{2}} \sum_{\ell=1}^{p-1} \sum_{j=0}^{\ell-1} \frac{1}{\ell} ||(q_1R_0(\lambda+i0)q_2)^j (q_1R_0(\lambda-i0)q_2)^{\ell-j-1} q_1\Gamma_0(\lambda)^* g(\lambda,\cdot)||^2 \\
\leq K\lambda^{-\frac{1}{2}} ||q_1\Gamma_0(\lambda)^* g(\lambda,\cdot)||^2 \leq K\lambda^{-1} ||g(\lambda,\cdot)||_{L^2(\mathbb{S}^{n-1})}^2.$$

Combining these we find

$$||(A_{t_2}(\cdot) - A_{t_1}(\cdot))g||_{\mathcal{H}_{spec}}^2 \leq C|t_2 - t_1|^2 \int_d^\infty \lambda \int_{\mathbb{S}^{n-1}} |[\tilde{A}(\lambda)g(\lambda, \cdot)](\omega)|^2 d\omega d\lambda$$

$$\leq CK|t_2 - t_1|^2 \int_d^\infty ||g(\lambda, \cdot)||_{L^2(\mathbb{S}^{n-1})}^2 d\lambda$$

$$\leq CK|t_2 - t_1|^2 ||g||_{\mathcal{H}_{spec}}^2.$$

The case when either or both of  $t_1, t_2$  is one follows from a similar calculation.

As a result, the path  $\beta_t = e^{iA_t}$  defines a norm continuous path of unitary operators in  $\mathcal{B}(\mathcal{H})$  from  $\beta$  to Id. Hence the path  $W_{\beta_t}$  defines a norm continuous path in  $\mathcal{B}(\mathcal{H})$  from  $W_{\beta}$  to Id, along which the Fredholm index is constant and equal to zero.

Remark 5.3.23. In dimension n=2,3 it is sufficient to take  $\chi=\frac{2}{\pi}\tan^{-1}(\cdot)$  due to the milder growth of  $A(\lambda)$  as  $\lambda\to 0$ .

**Theorem 5.3.24.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 6$  and let S be the scattering operator for  $H = H_0 + V$ . Let  $\beta$  be a high energy correction for S and let  $\sigma$  be a resonant correction if H admits a zero energy resonance. Otherwise, let  $\sigma = \operatorname{Id}$ . Then

$$-N = \operatorname{Index}(W_{-}) = \operatorname{Index}(W_{S\beta\sigma^{*}})$$

$$1 \quad f^{\infty} \quad (5.25)$$

$$= \frac{1}{2\pi i} \int_0^\infty \left( \operatorname{Tr}(S(\lambda)^* S'(\lambda)) + \frac{i}{4\pi} \lambda^{-\frac{1}{2}} \int_{\mathbb{R}^3} V(x) \, \mathrm{d}x \right) \mathrm{d}\lambda + \frac{1}{2} M_R(0), \tag{5.26}$$

where we have defined

$$\frac{1}{2}M_R(0) = -\frac{1}{2\pi i} \int_0^\infty \text{Tr}(\sigma(\lambda)^* \sigma'(\lambda)) \,d\lambda.$$

Proof. Since  $\beta$  defines a properly admissable unitary with  $\operatorname{Index}(W_{\beta}) = 0$  we obtain the equality  $-N = \operatorname{Index}(W_{-}) = \operatorname{Index}(W_{S\beta\sigma^*})$ . By construction we have that the map  $\lambda \mapsto \operatorname{Det}(S(\lambda)\beta(\lambda)\sigma(\lambda)^*)$  defines a loop, so has a well-defined winding number. The index of  $W_{S\beta\sigma^*}$  can be computed via this winding number as

$$\operatorname{Index}(W_{S\beta\sigma^*}) = \frac{1}{2\pi i} \int_0^\infty \frac{\frac{\mathrm{d}}{\mathrm{d}\lambda} (\operatorname{Det}(S(\lambda)\beta(\lambda)\sigma(\lambda)^*))}{\operatorname{Det}(S(\lambda)\beta(\lambda)\sigma(\lambda)^*)} \,\mathrm{d}\lambda$$

$$= \frac{1}{2\pi i} \int_0^\infty \left( \operatorname{Tr}(S(\lambda)^*S'(\lambda)) + \operatorname{Tr}(\beta(\lambda)^*\beta'(\lambda)) - \operatorname{Tr}(\sigma(\lambda)^*\sigma'(\lambda)) \right) \,\mathrm{d}\lambda$$

$$= \frac{1}{2\pi i} \int_0^\infty \left( \operatorname{Tr}(S(\lambda)^*S'(\lambda)) + \operatorname{Tr}(\beta(\lambda)^*\beta'(\lambda)) \right) \,\mathrm{d}\lambda + \frac{1}{2} M_R(0).$$

Then Lemma 2.5.27 shows that  $\text{Tr}(S(\cdot)^*S'(\cdot)) - \frac{d}{d\lambda}\text{Tr}(A(\cdot)) \in L^1(\mathbb{R}^+)$ . Noting the equality

 $\operatorname{Tr}(\beta(\lambda)^*\beta'(\lambda)) = \frac{\mathrm{d}}{\mathrm{d}\lambda}\operatorname{Tr}(A(\lambda)),$  we obtain

$$\operatorname{Index}(W_{S\beta\sigma^*}) = \frac{1}{2\pi i} \int_0^\infty \left( \operatorname{Tr}(S(\lambda)^* S'(\lambda)) + \operatorname{Tr}(\beta(\lambda)^* \beta'(\lambda)) \right) d\lambda + \frac{1}{2} M_R(0)$$

$$= \frac{1}{2\pi i} \int_0^\infty \left( \operatorname{Tr}(S(\lambda)^* S'(\lambda)) + \frac{i}{4\pi} \lambda^{-\frac{1}{2}} \int_{\mathbb{R}^3} V(x) dx \right) d\lambda + \frac{1}{2} M_R(0). \quad \Box$$

Remark 5.3.25. With a statement of Levinson's theorem in hand, one can define the spectral shift function directly. Define the function  $P_3: \mathbb{R}^+ \to \mathbb{C}$  by

$$P_3(\lambda) = -\frac{i}{2\pi} \lambda^{\frac{1}{2}} \int_{\mathbb{R}^3} V(x) \, \mathrm{d}x.$$

Then the spectral shift function is given by the relation

$$\xi(\lambda) = -\sum_{k=0}^{K} M(\lambda_k) \chi_{[\lambda_k, \infty)}(\lambda) - \frac{1}{2} M_R(0) \chi_{[0, \infty)} + \chi_{[0, \infty)} \frac{1}{2\pi i} \left( \int_0^{\lambda} (\text{Tr}(S(\mu)^* S'(\mu)) - P_3'(\mu)) \, d\mu + p(\lambda) \right).$$

This agrees with the normalisation convention of Theorem 2.5.13.

Corollary 5.3.26. Suppose that n = 3 and V satisfies Assumption 2.2.14 for some  $\rho > 5$ . Then

$$\xi(0+) = -N - \frac{1}{2}M_R(0).$$

### 5.3.3 High energy corrections in dimension n=2

As we have seen in Theorem 4.1.5 we have, in dimension n=2, the relation  $S(0)=\operatorname{Id}$  always and so no low energy corrections are required to make the index pairing work. We do not however have  $\lim_{\lambda\to\infty} S(\lambda) = \operatorname{Id}$  in trace norm, so a high energy correction is required to obtain a different representative of the class  $[W_S]$  and use the index pairing to obtain a formula for the number of bound states. We demonstrate in this section how to obtain the 2-dimensional statement of Levinson's theorem in Theorem 2.5.32 in the case of no p-resonances, providing a new proof of [26, Theorem 6.3] and [50, Theorem 3.2].

**Definition 5.3.27.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > 11$  and let S be the scattering operator for  $H = H_0 + V$ . Let  $\beta : \mathbb{R}^+ \to \mathcal{B}(\mathcal{P})$  be a once continuously differentiable unitary valued function such that  $\beta(\lambda) - \mathrm{Id} \in \mathcal{L}^1(\mathcal{P})$  for all  $\lambda \in \mathbb{R}^+$  and  $\beta(0) = \mathrm{Id}$  and  $\lim_{\lambda \to \infty} \beta(\lambda) = \mathrm{Id}$  in  $\mathcal{B}(\mathcal{P})$ . Then  $\mathrm{Det}(\beta(\lambda)) \in \mathbb{T}$  exists for all  $\lambda \in \mathbb{R}^+$  and we say that  $\beta$  is a high energy correction for S if

$$\lim_{\lambda \to \infty} \text{Det}(S(\lambda)\beta(\lambda)) = 1, \tag{5.27}$$

and  $\operatorname{Index}(W_{\beta}) = 0$ .

For a given potential V we construct a high energy correction for the corresponding scattering operator.

**Lemma 5.3.28.** Suppose that n=2 and V satisfies Assumption 2.2.14 for some  $\rho > 11$  and let S be the corresponding scattering operator. For  $\lambda > 0$  define the self-adjoint operator  $A(\lambda) = -4i \tan^{-1}(\lambda)i\Gamma_0(\lambda)V\Gamma_0(\lambda)^*$ . Then the unitary operator  $\beta(\lambda) = e^{iA(\lambda)}$  is a high energy correction for S.

*Proof.* This follows directly from Lemma 5.3.22 and Remark 5.3.23.

We note that the choice of the factor  $\frac{2}{\pi} \tan^{-1}(\cdot)$  is simply for convenience, any increasing differentiable function vanishing at 0 and having limit 1 at infinity suffices.

Using the high energy correction of Lemma 5.3.28 we provide a new proof of Levinson's theorem in dimension n=2 in the absence of p-resonances.

**Theorem 5.3.29.** Suppose that n = 2 and V satisfies Assumption 2.2.14 for some  $\rho > 11$  and let S be the corresponding scattering operator. Suppose further that there are no presonances and let  $\beta$  be a high energy correction for S. Then

$$-N = \operatorname{Index}(W_{-}) = \operatorname{Index}(W_{S\beta}) = \frac{1}{2\pi i} \int_{\mathbb{R}^{+}} \operatorname{Tr}(S(\lambda)^{*}S'(\lambda)) d\lambda + \frac{1}{4\pi} \int_{\mathbb{R}^{2}} V(x) dx.$$

*Proof.* We only need to show the equality

Index
$$(W_{S\beta}) = \frac{1}{2\pi i} \int_{\mathbb{R}^+} \text{Tr}(S(\lambda)^* S'(\lambda)) d\lambda + \frac{1}{4\pi} \int_{\mathbb{R}^2} V(x) dx$$

since the others have been established in Theorem 5.3.2. Compute

$$\operatorname{Index}(W_{S\beta}) = \frac{1}{2\pi i} \int_{\mathbb{R}^{+}} \frac{\frac{d}{d\lambda} \left( \operatorname{Det}(S(\lambda)\beta(\lambda)) \right)}{\operatorname{Det}(S(\lambda)\beta(\lambda))} d\lambda$$

$$= \frac{1}{2\pi i} \int_{\mathbb{R}^{+}} \left( \operatorname{Tr}S(\lambda)^{*}S'(\lambda) \right) + \operatorname{Tr}(\beta(\lambda)^{*}\beta'(\lambda)) d\lambda$$

$$= \frac{1}{2\pi i} \int_{\mathbb{R}^{+}} \operatorname{Tr}(S(\lambda)^{*}S'(\lambda)) d\lambda + \frac{1}{2\pi i} \int_{\mathbb{R}^{+}} \frac{d}{d\lambda} \operatorname{Tr}(A(\lambda)) d\lambda$$

$$= \frac{1}{2\pi i} \int_{\mathbb{R}^{+}} \operatorname{Tr}(S(\lambda)^{*}S'(\lambda)) d\lambda + \frac{1}{4\pi} \int_{\mathbb{R}^{2}} V(x) dx,$$

where we have used Lemma 2.4.20 to compute the final integral.

Thus, we have provided a new proof of Theorem 2.5.32 in the case of no p-resonances.

Remark 5.3.30. The result of Theorem 5.3.29 is enough to completely determine the spectral shift function for the pair  $(H, H_0)$  in the case of no p-resonances. We obtain

$$\xi(\lambda) = -\sum_{k=1}^K M(\lambda_k) \chi_{[\lambda_k, \infty)}(\lambda) + \frac{1}{2\pi i} \chi_{[0, \infty)}(\lambda) \int_0^\lambda \operatorname{Tr}(S(\mu)^* S'(\mu)) d\mu.$$

Corollary 5.3.31. Suppose that n=2 and V satisfies Assumption 2.2.14 for some  $\rho > 11$ . Suppose further that there are no p-resonances. Then

$$\xi(0+) = -N.$$

# 5.3.4 High energy corrections in dimension $n \ge 4$

In dimension  $n \geq 4$  the failure of the convergence of the scattering matrix at infinity in trace norm persists as an obstruction to obtaining an explicit formula. In this section we demonstrate how to construct a high energy correction to the scattering operator which will allow us to obtain a statement of Levinson's theorem in higher dimensions.

**Definition 5.3.32.** Suppose that V satisfies Assumption 4.3.1 and let S be the scattering operator for  $H = H_0 + V$ . Let  $\beta : \mathbb{R}^+ \to \mathcal{B}(\mathcal{P})$  be a once continuously differentiable unitary-valued function such that  $\beta(\lambda) - \mathrm{Id} \in \mathcal{L}^1(\mathcal{P})$  for all  $\lambda \in \mathbb{R}^+$  and  $\beta(0) = \mathrm{Id}$  and  $\lim_{\lambda \to \infty} \beta(\lambda) = \mathrm{Id}$  in  $\mathcal{B}(\mathcal{P})$ . Then  $\mathrm{Det}(\beta(\lambda)) \in \mathbb{T}$  exists for all  $\lambda \in \mathbb{R}^+$  and we say that  $\beta$  is a high energy correction for S if

$$\lim_{\lambda \to \infty} \text{Det}(S(\lambda)\beta(\lambda)) = 1, \tag{5.28}$$

and  $\operatorname{Index}(W_{\beta}) = 0$ .

For a given potential V we construct a high energy correction for the corresponding scattering operator. As usual we decompose the potential into V = vUv with  $v = |V|^{\frac{1}{2}}$  and U = sign(V).

**Lemma 5.3.33.** Suppose that  $V \in C_c^{\infty}(\mathbb{R}^n)$  and let S be the corresponding scattering operator. For  $\lambda > 0$  define for  $p \geq n$  the self-adjoint operator  $\tilde{A}(\lambda) \in \mathcal{B}(\mathcal{P})$  by

$$\tilde{A}(\lambda) = 2\pi\Gamma_0(\lambda)V\left(\sum_{\ell=1}^{p-1}\sum_{j=0}^{\ell-1}\frac{(-1)^{\ell}}{\ell}(vR_0(\lambda+i0)vU)^j(vR_0(\lambda-i0)vU)^{\ell-j-1}\right)\Gamma_0(\lambda)^*$$

Further define the self-adjoint operator  $A(\lambda) \in \mathcal{B}(\mathcal{P})$  by

$$A(\lambda) = \chi(\lambda^{\frac{1}{2}})\tilde{A}(\lambda),$$

where  $\chi$  is as in Lemma 5.3.22. Then the unitary  $\beta(\lambda) = e^{iA(\lambda)}$  is a high energy correction for S.

*Proof.* This follows directly from Lemma 5.3.22.

We now use the high energy correction of Lemma 5.3.28 to provide an alternate proof of Levinson's theorem in dimension  $n \geq 4$  in the absence of resonances. Define the function  $Q_n : \mathbb{R}^+ \to \mathbb{C}$  by

$$Q_n(\lambda) = -\text{Tr}(\tilde{A}(\lambda))$$

and recall the high-energy polynomials  $P_n$ ,  $p_n$  from Definitions 2.5.28 and 2.5.26.

**Theorem 5.3.34.** Suppose that  $n \geq 4$ ,  $V \in C_c^{\infty}(\mathbb{R}^n)$  and there are no resonances. Then the number of bound states of H is given by

$$-N = \operatorname{Index}(W_{-}) = \operatorname{Index}(W_{S\beta}) = \frac{1}{2\pi i} \int_{0}^{\infty} \left( \operatorname{Tr}(S(\lambda)^{*}S'(\lambda)) - p_{n}(\lambda) \right) d\lambda - \frac{1}{2\pi i} P_{n}(0).$$

*Proof.* The first two equalities have been established in Theorem 5.3.2. Since the operator  $\text{Det}(S(\lambda)\beta(\lambda)) \to 1$  as both  $\lambda \to 0$  and  $\lambda \to \infty$ , the map  $\lambda \mapsto \text{Det}(S(\lambda)\beta(\lambda))$  defines a loop and we find

 $\operatorname{Index}(W_{S\beta})$ 

$$= \operatorname{Wind}(\operatorname{Det}(S(\cdot)\beta(\cdot))) = \frac{1}{2\pi i} \int_{0}^{\infty} \frac{\frac{d}{d\lambda} \operatorname{Det}(S(\lambda)\beta(\lambda))}{\operatorname{Det}(S(\lambda)\beta(\lambda))} d\lambda$$

$$= \frac{1}{2\pi i} \int_{0}^{\infty} \left( \operatorname{Tr}(S(\lambda)^{*}S'(\lambda)) + \frac{d}{d\lambda} \operatorname{Tr}(A(\lambda)) \right) d\lambda$$

$$= \frac{1}{2\pi i} \int_{0}^{\infty} \left( \operatorname{Tr}(S(\lambda)^{*}S'(\lambda)) - p_{n}(\lambda) \right) d\lambda + \frac{1}{2\pi i} \int_{0}^{\infty} \frac{d}{d\lambda} \left( \chi(\lambda)^{\frac{1}{2}} Q_{n}(\lambda) + P_{n}(\lambda) \right) d\lambda$$

$$= \frac{1}{2\pi i} \int_{0}^{\infty} \left( \operatorname{Tr}(S(\lambda)^{*}S'(\lambda)) - p_{n}(\lambda) \right) d\lambda - \frac{1}{2\pi i} P_{n}(0),$$

where in the last line we have used the properties of  $\chi$  and the fundamental theorem of calculus.

Remark 5.3.35. We note that the statement of Levinson's theorem given in Theorem 5.3.34 completely determines the spectral shift function for the pair  $(H, H_0)$  as

$$\xi(\lambda) = -\sum_{k=1}^{K} M(\lambda_k) \chi_{[\lambda_k, \infty)}(\lambda)$$
$$+ \frac{1}{2\pi i} \chi_{[0, \infty)} \left( \int_0^{\lambda} \left( \text{Tr}(S(\mu)^* S'(\mu)) - p_n(\mu) \right) d\mu + P_n(\lambda) \right).$$

Corollary 5.3.36. Suppose that  $n \ge 4$ , V satisfies Assumption 4.3.1 and there are no resonances. Then  $\xi(0+) = -N$ .

In higher dimensions we can determine the coefficients by performing more commutators, however these become difficult very quickly in a similar manner to the heat coefficients of Theorem 2.5.34. We describe in Proposition 5.3.37 the coefficients arising in dimensions  $n \leq 7$ . In Chapter 6 we provide an alternative method for determining some of these coefficients.

**Proposition 5.3.37.** Let  $n \in \{4, 5, 6, 7\}$  and suppose that V satisfies Assumption 2.2.14 for some  $\rho > n + 1$  and  $V \in C^{\infty}(\mathbb{R}^n)$ . Then the function  $Q_n$  of Lemma 5.3.33 satisfies  $Q_n(\lambda) = P_n(\lambda) + q(\lambda)$ , for some  $q : \mathbb{R}^+ \to \mathbb{C}$ . Explicitly we obtain

$$Q_n(\lambda) = -\frac{(2\pi i)\lambda^{\frac{n-2}{2}} \operatorname{Vol}(\mathbb{S}^{n-1})}{2(2\pi)^n} \int_{\mathbb{R}^n} V(x) \, dx + \frac{(2\pi i)\operatorname{Vol}(\mathbb{S}^{n-1})(n-2)\lambda^{\frac{n-4}{2}}}{8(2\pi)^n} \int_{\mathbb{R}^n} V(x)^2 \, dx - \frac{(2\pi i)\operatorname{Vol}(\mathbb{S}^{n-1})(n-2)(n-4)\lambda^{\frac{n-6}{2}}}{48(2\pi)^n} \int_{\mathbb{R}^n} \left(V(x)^3 + \frac{1}{2}|[\nabla V](x)|^2\right) \, dx + q(\lambda).$$

If n is even, then q = 0. In addition we have q(0) = 0 for n even.

Proposition 5.3.37 follows immediately from the computations in Theorem 2.5.29.

Remark 5.3.38. For explicitness and a later comparison, we record the statements of Levinson's theorem in dimensions n = 4, 5, 6, 7. We use the notation of Theorem 2.5.34. In dimension n = 4 with no resonances we find

$$-N = \frac{1}{2\pi i} \int_0^\infty \left( \operatorname{Tr}(S(\lambda)^* S'(\lambda)) - c_1(4, V) \right) \, d\lambda - \beta_4(V),$$

where

$$c_1(4, V) - \frac{(2\pi i) \text{Vol}(\mathbb{S}^3)}{2(2\pi)^4} \int_{\mathbb{R}^4} V(x) \, dx \quad \text{and}$$
$$\beta_4(V) = \frac{1}{2\pi i} P_4(0) = \frac{\text{Vol}(\mathbb{S}^3)}{4(2\pi)^2} \int_{\mathbb{R}^4} V(x)^2 \, dx.$$

In dimension n = 5 we find

$$-N = \frac{1}{2\pi i} \int_0^\infty \left( \operatorname{Tr}(S(\lambda)^* S'(\lambda)) - \lambda^{\frac{1}{2}} c_1(5, V) - \lambda^{-\frac{1}{2}} c_2(5, V) \right) d\lambda,$$

where the coefficients are

$$c_1(5, V) = -\frac{3}{2} \frac{(2\pi i) \text{Vol}(\mathbb{S}^4)}{2(2\pi)^5} \int_{\mathbb{R}^5} V(x) \, dx \quad \text{and}$$
$$c_2(5, V) = \frac{1}{2} \frac{3(2\pi i) \text{Vol}(\mathbb{S}^4)}{8(2\pi)^5} \int_{\mathbb{R}^5} V(x)^2 \, dx.$$

In dimension n = 6 we have

$$-N = \frac{1}{2\pi i} \int_0^\infty (\text{Tr}(S(\lambda)^* S'(\lambda)) - c_1(6, V)\lambda - c_2(6, V)) d\lambda - \beta_6(V),$$

where

$$c_{1} = -\frac{(2\pi i)\operatorname{Vol}(\mathbb{S}^{5})}{(2\pi)^{6}} \int_{\mathbb{R}^{6}} V(x) \, dx,$$

$$c_{2} = \frac{(2\pi i)\operatorname{Vol}(\mathbb{S}^{5})}{2(2\pi)^{6}} \int_{\mathbb{R}^{6}} V(x)^{2} \, dx \quad \text{and}$$

$$b_{6} = \frac{1}{2\pi i}P_{6}(0) = -\frac{\operatorname{Vol}(\mathbb{S}^{5})}{6(2\pi)^{6}} \int_{\mathbb{R}^{6}} \left(V(x)^{3} + \frac{1}{2}|[\nabla V](x)|^{2}\right) \, dx.$$

In dimension n = 7 we obtain

$$-N = \frac{1}{2\pi i} \int_0^\infty \left( \text{Tr}(S(\lambda)^* S'(\lambda)) - \lambda^{\frac{3}{2}} c_1(7, V) - \lambda^{\frac{1}{2}} c_2(7, V) - \lambda^{-\frac{1}{2}} c_3(7, V) \right) d\lambda,$$

where

$$c_{1}(7, V) = -\frac{5}{2} \frac{(2\pi i) \text{Vol}(\mathbb{S}^{6})}{2(2\pi)^{7}} \int_{\mathbb{R}^{7}} V(x) \, dx,$$

$$c_{2}(7, V) = \frac{3}{2} \frac{5(2\pi i) \text{Vol}(\mathbb{S}^{6})}{8(2\pi)^{7}} \int_{\mathbb{R}^{7}} V(x)^{2} \, dx \quad \text{and}$$

$$c_{3}(7, V) = \frac{1}{2} \frac{15(2\pi i)}{48(2\pi)^{7}} \int_{\mathbb{R}^{7}} \left( V(x)^{3} + \frac{1}{2} |[\nabla V](x)|^{2} \right) \, dx.$$

## Chapter 6

# Spectral flow for Schrödinger operators

In this chapter we consider the spectral flow along the path  $H = H_0 + tV$  and show how Levinson's theorem can be obtained from the spectral flow along this path. To use the spectral flow formula of [47, Theorem 9] we require some simple adjustments to make our path of operators Fredholm. To analyse the terms appearing in [47, Theorem 9] requires an analogue of the Birman-Kreĭn trace formula of Theorem 2.5.19 and the pseudodifferential operator calculus of [43], modified appropriately to our second order operators.

In Section 6.1 we briefly recall the definition of spectral flow of [129, 130] for paths of unbounded self-adjoint Fredholm operators and in particular Equation 6.1 of [47, Theorem 9]. Since the operators  $H_0$  and H are never Fredholm, we consider a family of modified operators  $(H_0(\alpha), H(\alpha))$  which are Fredholm and describe their stationary scattering theory in Section 6.2. In Section 6.3 we briefly recall a number of results from the pseudodifferential operator calculus of [43, 46] modified to our operators  $(H_0(\alpha), H(\alpha))$ .

In Section 6.4 we apply [47, Theorem 9] to the path  $H_t(\alpha) = H_0 + \alpha + tV$  to prove, using the Birman-Kreĭn trace formula and the pseudodifferential operator calculus of Section 6.3 to provide in Theorems 6.4.22 (odd dimensions) and 6.4.23 (even dimensions) an expression for the spectral flow along the path  $H_t(\alpha)$ . Finally, in Section 6.5 we use the high-energy behaviour of the time delay operator discussed in Section 2.5.1 and Chapter 5 to show that Levinson's theorem is a result of spectral flow. In particular, Section 6.4 onwards contains original results.

## 6.1 Spectral flow

The concept of spectral flow was used by Atiyah, Patodi and Singer in [10, 11] as a tool to develop APS index theory. Spectral flow is intuitively defined as the net number of eigenvalues which change sign along a path of self-adjoint operators, with the convention

that an eigenvalue changing from negative to positive will contribute a 1 to the spectral flow. We use the definition due to Phillips [129, 130]. Phillips' definition of spectral flow is valid in the much broader setting of semifinite von Neumann algebras with faithful normal semifinite traces, although we do not need the full power of such a definition.

Consider  $\mathcal{B}(\mathcal{H})$  with trace Tr, let  $\pi: \mathcal{B}(\mathcal{H}) \to \mathcal{B}(\mathcal{H})/\mathcal{K}(\mathcal{H})$  denote the projection onto the Calkin algebra. Let  $\chi = \chi_{[0,\infty)}$  be the characteristic function of the interval  $[0,\infty)$ . Let  $(T_t)$  be any norm continuous path of bounded self-adjoint Fredholm operators in  $\mathcal{B}(\mathcal{H})$ . Then  $\pi(\chi(T_t)) = \chi(\pi(T_t))$ . Since the spectrum of the  $\pi(T_t)$  are bounded away from zero, the path  $\chi(\pi(T_t))$  is continuous. By compactness (and [20, Lemma 4.1]), we can choose a partition  $0 = t_0 < t_1 < \cdots < t_k = 1$  such that

$$||\pi(\chi(T_t)) - \pi(\chi(T_s))|| < \frac{1}{2}$$

for all  $t, s \in [t_{i-1}, t_i]$  and  $1 \le i \le k$ . Defining the projection  $P_i = \chi(T_{t_i})$  we find that  $P_{i-1}P_i: P_i\mathcal{H} \to P_{i-1}\mathcal{H}$  is Fredholm. We recall the following definition, due to Phillips [129, 130].

**Definition 6.1.1.** Let  $\mathcal{H}$  be a Hilbert space. For  $t \in [0, 1]$  let  $(T_t)$  be any norm continuous path of bounded self-adjoint Fredholm operators in  $\mathcal{B}(\mathcal{H})$ . For a partition  $0 = t_0 < t_1 < \cdots < t_k = 1$  of the interval [0, 1] define the operators  $P_i = \chi(T_{t_i})$ . Then we define the spectral flow of the path  $(T_t)$  by

$$\operatorname{sf}(T_t) := \sum_{i=1}^k \operatorname{Ind}(P_{i-1}P_i).$$

We note that the above definition of spectral flow is independent of the choice of partition [107, 129, 130] and agrees with the topological definition used in [10, 11], when both make sense. For unbounded operators, we make the following definition of spectral flow [47].

**Definition 6.1.2.** Let  $\mathcal{H}$  be a Hilbert space with trace Tr. Let  $(D_t)$  be a graph norm continuous path of unbounded self-adjoint Fredholm operators on  $\mathcal{H}$ . Denote the function  $F: \mathbb{R} \to [-1, 1]$  by  $F(x) = x(1+x^2)^{-\frac{1}{2}}$ . The spectral flow along the path  $(D_t)$  is defined by

$$\operatorname{sf}(D_t) = \operatorname{sf}(F(D_t)).$$

Throughout the rest of this section  $[0,1] \ni t \mapsto D_t$  stands for a path of unbounded self-adjoint linear operators on  $\mathcal{H} = L^2(\mathbb{R}^n)$ . We denote by  $(F_t) = (F(D_t))$  the bounded transform of the path  $(D_t)$ . We must also impose a smoothness assumption on  $D_t$  to use analytic formulae for the spectral flow.

**Definition 6.1.3.** 1. A path  $[0,1] \ni t \mapsto D_t$  is called  $\Gamma$ -differentiable at the point  $t = t_0$  if and only if there exists a bounded linear operator T such that

$$\lim_{t \to t_0} \left| \left| t^{-1} (D_t - D_{t_0}) (\operatorname{Id} + D_{t_0}^2)^{-\frac{1}{2}} - T \right| \right| = 0.$$

In this case we set  $\dot{D}_{t_0} = T(\mathrm{Id} + D_{t_0}^2)^{-\frac{1}{2}}$ . The operator  $\dot{D}_t$  is a symmetric linear operator with domain  $\mathrm{Dom}(D_t)$  [47, Lemma 25].

2. If the mapping  $t \mapsto \dot{D}_t(\mathrm{Id} + D_t^2)^{-\frac{1}{2}}$  is defined and continuous with respect to the operator norm, then we call the path  $t \mapsto D_t$  a continuously  $\Gamma$ -differentiable or a  $C_{\Gamma}^1$  path.

The most general analytic spectral flow formula for the case of unbounded operators on a Hilbert space is given by the following theorem [47, Theorem 9].

**Theorem 6.1.4.** Let  $[0,1] \ni t \mapsto D_t$  be a piecewise  $C^1_{\Gamma}$  path of linear operators and  $F_t \in \mathcal{F}_* = \{F \in \mathcal{B}(\mathcal{H}) : F = F^* \text{ is Fredholm and } ||F|| \le 1\}$ . Let  $g : \mathbb{R} \to \mathbb{R}$  be a positive  $C^2$  function such that

- 1.  $\int_{\mathbb{R}} g(x) \, dx = 1$ ;
- 2.  $\int_0^1 \left| \left| \dot{D}_t g(D_t) \right| \right|_1 dt < \infty$ ; and
- 3.  $G(D_1) \frac{1}{2}B_1 G(D_0) + \frac{1}{2}B_0 \in \mathcal{L}^1(\mathcal{H})$  (the trace class), where  $B_j = 2\chi_{[0,\infty)}(D_j) 1$ , and G is the antiderivative of g such that  $G(\pm \infty) = \pm \frac{1}{2}$ .

Then

$$sf(D_t) = \int_0^1 Tr(\dot{D}_t g(D_t)) dt - Tr\left(G(D_1) - \frac{1}{2}B_1 - G(D_0) + \frac{1}{2}B_0\right).$$
 (6.1)

Remark 6.1.5. There is a sign difference on the second term on the right-hand side of Equation (6.1) when compared to the statement of [47, Theorem 9]. This is due to a sign error in the proof of [47, Theorem 9]. In particular, on p. 1823 there is a diagram illustrating the path of integration for the succeeding argument. The diagram is correct, however in the following formula the signs on the terms  $\gamma_1, \gamma_2$  are incorrect due to the wrong choice of orientation of the path in their definition.

Fix  $s > \frac{n+2}{2}$  and make the explicit choice  $g_s(x) = \frac{1}{C_s}(1+x^2)^{-s}$  with  $C_s$  chosen such that  $||g_s||_1 = 1$ . In this case we can determine the antiderivative  $G_s$  of  $g_s$  which satisfies the hypotheses of Theorem 6.1.4 as

$$G_s(x) = -\frac{1}{2} + \int_{-\infty}^x g_s(u) \, \mathrm{d}u.$$

Since we are using the functional calculus on  $G_s$ , we require a simpler expression which does not have the variable as an integration limit. Since  $G_s$  is an odd function, for x > 0 we find

$$G_s(x) = \frac{1}{C_s} \int_0^x (1+u^2)^{-s} du = \frac{1}{C_s} \left( \int_0^\infty (1+u^2)^{-s} du - \int_x^\infty (1+u^2)^{-s} du \right)$$

$$= \frac{1}{2} - \frac{1}{C_s} \int_x^\infty (1+u^2)^{-s} du = \frac{1}{2} - \frac{1}{2C_s} \int_1^\infty x (1+wx^2)^{-s} w^{-\frac{1}{2}} dw$$

$$=: \frac{1}{2} - \frac{1}{2} \eta_s(x).$$

If x < 0 we use the fact  $G_s$  is odd, so  $G_s(-x) = -\frac{1}{2} + \eta_s(-x)$ . The requirement  $s > \frac{n+2}{2}$  is to ensure  $\eta_s$  satisfies Equation (2.84) so that we are able to use Theorem 2.5.13 and in particular (an analogue of) the Birman-Kreĭn trace formula which we state in Lemma 6.2.8. We remark that the function  $\eta_s$  is defined for  $\text{Re}(s) > \frac{1}{2}$ , holomorphic in s with a simple pole at  $s = \frac{1}{2}$ , a fact to be exploited repeatedly at a later point.

## 6.2 Making the operators Fredholm

As usual we let  $H = H_0 + V$  where we make for the rest of this chapter the following assumption on V.

**Assumption 6.2.1.** We assume that V satisfies Assumption 4.3.1 so that we may use the results of Chapter 5 regarding the behaviour of the spectral shift function. We impose the additional assumption that V is smooth.

The assumption on the differentiability of the potential becomes necessary when we employ the pseudodifferential operator calculus, where we must take derivatives of the potential.

To use the spectral flow formulae of the previous section, we consider pairs of Fredholm operators, which  $H_0$  and H are not since 0 is contained in the essential spectrum of both. We construct a pair of Fredholm operators as follows. Enumerate the distinct eigenvalues of H in increasing order as  $\lambda_1 < \lambda_2 < \cdots < \lambda_K \le 0$ . Let  $\mu \in \{\lambda_{K-1}, \lambda_K\}$  be the closest non-zero eigenvalue of H to zero. Then for any  $\alpha \in (0, -\mu)$  the operators  $H_0(\alpha) = H_0 + \alpha$  and  $H(\alpha) = H_0 + V + \alpha$  are invertible and hence are Fredholm operators.

The operators  $H_0(\alpha)$  and  $H(\alpha)$  depend norm-resolvent continuously on  $\alpha \in (0, -\mu)$ . The operator  $H_0(\alpha)$  has purely absolutely continuous spectrum  $\sigma(H_0(\alpha)) = \sigma_{ac}(H_0(\alpha)) = [\alpha, \infty)$ . The operator  $H(\alpha)$  has finitely many eigenvalues in  $(-\infty, \alpha]$  and has absolutely continuous spectrum  $\sigma_{ac}(H(\alpha)) = \sigma_{ac}(H_0(\alpha)) = [\alpha, \infty)$ . We enumerate the distinct eigenvalues of  $H(\alpha)$  in ascending order as  $\lambda_1(\alpha) < \lambda_2(\alpha) < \cdots < \lambda_K(\alpha)$  and note that for each j we have  $\lambda_j(\alpha) = \lambda_j + \alpha$ , where the  $\lambda_j$  denote the distinct eigenvalues of H. We also note that the number of eigenvalues of H is equal to the number of eigenvalues of  $H(\alpha)$ , independent of  $\alpha$ .

For each  $\alpha \in (0, -\mu)$ ,  $z \in \mathbb{C} \setminus \mathbb{R}$  and any  $p \geq n$  we have by (an analogue of) Lemma 2.5.11 that the difference  $(H(\alpha) - z)^{-p} - (H_0(\alpha) - z)^{-p} \in \mathcal{L}^1(\mathcal{H})$  and the wave operators

$$W_{\pm}(\alpha) = \underset{t \to +\infty}{\text{s-lim}} e^{itH(\alpha)} e^{-itH_0(\alpha)} P_{ac}(H_0(\alpha)) = \underset{t \to +\infty}{\text{s-lim}} e^{itH} e^{-itH_0} = W_{\pm}$$

exist and are asymptotically complete by the invariance principle [137, Theorem XI.11] (or an adaptation of the results of Chapter 2). We also note that the wave operators are independent of  $\alpha$ . The  $\alpha$  dependence is seen in the spectral representation for  $H_0(\alpha)$ , which we now describe.

**Definition 6.2.2.** For  $\lambda \in (\alpha, \infty)$ ,  $s \in \mathbb{R}$  and  $t > \frac{1}{2}$  we define the operator  $\Gamma_{\alpha}(\lambda) : H^{s,t} \to \mathcal{P}$  by

$$[\Gamma_{\alpha}(\lambda)f](\omega) = 2^{-\frac{1}{2}}(\lambda - \alpha)^{\frac{n-2}{4}}[\mathcal{F}f]((\lambda - \alpha)^{\frac{1}{2}}\omega) = [\Gamma_{0}(\lambda - \alpha)f](\omega).$$

The operator  $F_{\alpha}: \mathcal{H} \to L^2([\alpha, \infty)) \otimes \mathcal{P}$ , which diagonalises the Hamiltonian  $H_0(\alpha)$ , is given by

$$[F_{\alpha}f](\lambda,\omega) = [\Gamma_{\alpha}(\lambda)f](\omega) = [F_{0}f](\lambda - \alpha,\omega).$$

Since the operators  $F_{\alpha}$  and  $\Gamma_{\alpha}$  are simply translations of  $F_0$  and  $\Gamma_0$  one can easily check the following.

**Lemma 6.2.3.** The operator  $F_{\alpha}$  is unitary. Moreover for  $\lambda \in (\alpha, \infty)$ ,  $\omega \in \mathbb{S}^{n-1}$  and  $f \in L^2([\alpha, \infty)) \otimes \mathcal{P}$  we have

$$[F_{\alpha}H_0(\alpha)F_{\alpha}^*f](\lambda,\omega) = \lambda f(\lambda,\omega).$$

The scattering operator  $S_{\alpha} = W_{+}(\alpha)^{*}W_{-}(\alpha)$  is unitary and commutes with  $H_{0}(\alpha)$  and so there exists a family  $\{S_{\alpha}(\lambda)\}_{\lambda \in [\alpha,\infty)}$  of unitary operators on  $\mathcal{P}$  such that the relation  $[F_{\alpha}S_{\alpha}f](\lambda,\omega) = S_{\alpha}(\lambda)[F_{\alpha}f](\lambda,\omega)$  holds. We call  $S_{\alpha}(\lambda)$  the (translated) scattering matrix at energy  $\lambda$ .

The next result is proved in an identical fashion to Theorem 2.4.32.

**Theorem 6.2.4.** Suppose that V satisfies Assumption (6.2.1). The scattering matrix is given for all  $\lambda \in [\alpha, \infty)$  by the equation

$$S_{\alpha}(\lambda) = \operatorname{Id} - 2\pi i \Gamma_{\alpha}(\lambda) (V - VR(\lambda - \alpha + i0)V) \Gamma_{\alpha}(\lambda)^* = S(\lambda - \alpha).$$

For each  $\lambda \in [\alpha, \infty)$ , the operator  $S_{\alpha}(\lambda)$  is unitary in  $\mathcal{P} = L^2(\mathbb{S}^{n-1})$  and depends norm continuously on  $\lambda \in [\alpha, \infty)$ .

Since the scattering matrix  $S_{\alpha}(\lambda)$  is simply a translation of the scattering matrix  $S(\lambda)$ , the properties of the time-delay and spectral shift function for the pair  $(H_0(\alpha), H(\alpha))$  also follow immediately.

The time delay satisfies  $T_{\alpha}(\lambda) = -iS_{\alpha}^{*}(\lambda)S_{\alpha}'(\lambda) = T(\lambda - \alpha)$ , its existence is guaranteed by the following result, whose proof is identical to that of Theorem 2.5.3.

**Lemma 6.2.5.** Suppose that V satisfies Assumption 6.2.1 and let  $P_0(\alpha)$  denote the spectral family for  $H_0(\alpha)$ . Then the scattering matrix is differentiable. Define  $T_{\alpha}$  by

$$Dom(T_{\alpha}) = \{ f \in \mathcal{H}_{ac} : P_0(\alpha)([a, b]) f = f \text{ for some } [a, b] \subset (0, \infty) \},$$
$$[F_{\alpha}T_{\alpha}f](\lambda, \omega) = -iS_{\alpha}(\lambda)^*S'_{\alpha}(\lambda)[F_{\alpha}f](\lambda, \omega).$$

Then  $T_{\alpha}$  is essentially self-adjoint on  $\text{Dom}(T_{\alpha})$ ,  $\text{Dom}(T_{\alpha}) \subset \text{Dom}(H_0(\alpha))$ , and  $T_{\alpha}$  commutes with  $H_0(\alpha)$ . Furthermore, the operator  $T_{\alpha}(\lambda)$  is trace class for all  $\lambda \in [\alpha, \infty)$ .

**Theorem 6.2.6.** Suppose that V satisfies Assumption 2.2.14 for some  $\rho > n$ . Then there exists a real-valued function  $\xi_{\alpha}$  such that the trace formula

$$\operatorname{Tr}\left(f(H(\alpha)) - f(H_0(\alpha))\right) = \int_{\mathbb{R}} \xi_{\alpha}(\lambda) f'(\lambda) \,d\lambda \tag{6.2}$$

holds at least for all  $f \in C_c^{\infty}(\mathbb{R})$ . The function  $\xi_{\alpha}$  satisfies  $\xi_{\alpha}(\lambda) = \xi(\lambda - \alpha)$ , where  $\xi$  is the spectral shift function for the pair  $(H, H_0)$ .

The function  $\xi_{\alpha} = \xi(H_0(\alpha), H(\alpha))$  is known as the spectral shift function for the pair  $(H_0(\alpha), H(\alpha))$ . As before the trace formula of Equation (6.2) fixes  $\xi_{\alpha}$  up to an additive constant. This constant is fixed by the condition that  $\xi_{\alpha}(\lambda) = 0$  for  $\lambda < |\inf(\sigma_p(H(\alpha)))|$ .

**Lemma 6.2.7.** Suppose that V satisfies Assumption 6.2.1. Then for  $\lambda \in [\alpha, \infty)$  the spectral shift function and the scattering matrix are related by the formula

$$\operatorname{Det}(S_{\alpha}(\lambda)) = e^{-2\pi i \xi_{\alpha}(\lambda)}.$$

For  $\lambda \in [\alpha, \infty)$  the spectral shift function and time-delay operator are related by the formula

$$-2\pi i \xi_{\alpha}'(\lambda) = \text{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)).$$

Our main use of the spectral shift function is the Birman-Kreĭn formula for the pair  $(H(\alpha), H_0(\alpha))$ , which follows immediately from Theorem 2.5.19 by an identical proof.

**Lemma 6.2.8.** Suppose that V satisfies Assumption 6.2.1 and  $f \in C_c^{\infty}(\mathbb{R})$ . Then

$$\operatorname{Tr}(f(H(\alpha)) - f(H_0(\alpha))) = \frac{1}{2\pi i} \int_{\alpha}^{\infty} f(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)) d\lambda + \sum_{k=1}^{K} M(\lambda_k(\alpha)) f(\lambda_k(\alpha)) + f(\alpha)(\xi_{\alpha}(\alpha -) - \xi(\alpha +) - M(\alpha)).$$

## 6.3 Pseudodifferential operator calculus

In this section we use the pseudodifferential theory of [43, 46] to obtain a number of trace class and resolvent estimates which will be used to eliminate remainder terms in our use of Theorem 6.1.4 in Section 6.1. We are not able to directly use the results of [46] and [43], since they are formulated for first order pseudodifferential operators and our Hamiltonians are second order. We thus provide analogous results to keep track of the resulting numerology, although we omit the proofs since they are essentially identical.

The following definition captures the notion of a pseudodifferential operator associated to the operator  $H_0(\alpha)$ .

**Definition 6.3.1.** Define the space  $\mathcal{H}_{\infty} = \bigcap_{k \geq 0} \text{Dom}(H_0^k)$ . For  $T \in \mathcal{B}(\mathcal{H})$  with  $T : \mathcal{H}_{\infty} \to \mathcal{H}_{\infty}$  we let  $\delta(T) = [H_0(\alpha)^{\frac{1}{2}}, T]$ . We define for  $p \in \mathbb{R}$  the set of regular order p pseudodifferential operators associated with  $H_0(\alpha)$  by

$$OP^p := (1 + H_0(\alpha)^2)^{\frac{p}{4}} \left( \bigcap_{n \in \mathbb{N}} \text{Dom}(\delta^n) \right).$$

The topology of  $OP^p$  is associated with the family of norms

$$\sum_{k=0}^{\ell} \left| \left| \delta^k ((1 + H_0(\alpha)^2)^{-\frac{p}{4}} T) \right| \right|$$

for  $\ell \in \mathbb{N} \cup \{0\}$ .

For the remainder of this chapter we let  $r = \frac{1}{2} \min\{\alpha^2, (\mu + \alpha)^2\}$ , so that the inequality  $0 < r < \inf\{\sigma(H_t(\alpha)^2)\}$  holds.

We first give basic estimates for the operator norm of resolvents of certain operators.

**Lemma 6.3.2.** Let  $\alpha \in (0, -\mu)$  and  $\lambda = a + iv$  with  $a \in (0, r)$  Then for all  $t \in [0, 1]$  we have

$$\left| \left| (\lambda - (\mathrm{Id} + H_t(\alpha)^2))^{-1} \right| \right| \le (v^2 + (r-a)^2)^{-\frac{1}{2}} \le (r-a)^{-1}.$$

We will use the notation  $Q = \mathrm{Id} + (H_0 + \alpha)^2$  and for  $A \in OP^{\ell}$  we use the notation  $A^{(k)} = [H_0(\alpha)^2, \cdots [H_0(\alpha)^2, A] \cdots]$ , where the expression has k commutators. It is useful

to note that the operator  $A^{(k)}$  has order  $\ell + 3k$  (each commutator raises the order by 3, one less than the order of  $H_0(\alpha)^2$ ). For Re(s) > n we can use Cauchy's integral formula to write

$$Q^{-s} = \frac{1}{2\pi i} \int_{\gamma_{\alpha}} \lambda^{-s} (\lambda - Q)^{-1} d\lambda$$

where  $\gamma_{\alpha}$  is a vertical line with real part fixed in (0, r). It can be checked using the spectral theorem that this integral converges in operator norm to  $Q^{-s}$ . We can thus reduce computing commutators of  $T \in OP^p$  with  $Q^{-s}$  to iterated calculations of commutators of T with  $(\lambda - Q)^{-1}$ . This leads to the following, an analogue of [46, Lemma 6.9].

**Lemma 6.3.3.** Let p, j, k be non-negative integers and  $A \in OP^{\ell}$ . Then

$$(\lambda - Q)^{-p} A = \sum_{j=0}^{k} {p+j-1 \choose j} A^{(j)} (\lambda - Q)^{-p+j} + P(\lambda)$$

where the remainder term  $P(\lambda)$  has order  $-(4p+k-\ell+1)$  and is given explicitly by

$$P(\lambda) = \sum_{j=1}^{p} {j+k+1 \choose k} (\lambda - Q)^{j-p-1} A^{(k+1)} (\lambda - Q)^{-j-k}.$$

Corollary 6.3.4. Let p, M be positive integers and  $A \in OP^k$ . Let  $R = (\lambda - Q)^{-1}$ . Then

$$R^{p}AR^{-p} = \sum_{j=0}^{M} \binom{p+j-1}{j} A^{(j)}R^{j} + P,$$

where P is of order k - M - 1 and given explicitly by

$$P = \sum_{j=1}^{p} {j+M-1 \choose M} R^{p+1-j} A^{(M+1)} R^{M+j-p}.$$

The following is an analogue of [46, Lemma 6.10], allowing us to uniformly bound pseudodifferential operators when weighted by enough resolvents.

**Lemma 6.3.5.** Let k, p be non-negative integers and suppose  $\lambda \in \mathbb{C}$  with  $a = \text{Re}(\lambda) \in (0, r)$ . Then for  $A \in OP^k$  and with  $R_{\alpha}(\lambda) = (\lambda - (\text{Id} + (H_0 + \alpha)^2))^{-1}$  we have

$$\left| \left| R_{\alpha}(\lambda)^{\frac{p}{4} + \frac{k}{4}} A R_{\alpha}(\lambda)^{-\frac{p}{4}} \right| \right| \leq C_{p,k}, \quad \text{and} \quad \left| \left| R_{\alpha}(\lambda)^{-\frac{p}{4}} A R_{\alpha}(\lambda)^{\frac{p}{4} + \frac{k}{4}} \right| \right| \leq C_{p,k}$$

where  $C_{p,k}$  is a constant independent of  $\lambda$ .

Remark 6.3.6. We think of the operator X in the following lemmas as  $t\{H_0, V\} + 2t\alpha V + t^2V^2$ , and thus  $H_0(\alpha)^2 + X = (H_t(\alpha))^2 \ge 0$ . With  $a = \text{Re}(\lambda) \in (0, r)$  (and r fixed as

before) we see that the spectrum of  $(\mathrm{Id} + H_0(\alpha)^2 + X)$  is bounded away from the line  $\mathrm{Re}(\lambda) = a$  by at least 1-a independent of  $\lambda$  and t. This fact is crucial in our next results.

**Lemma 6.3.7.** Let  $A_i \in OP^{p_i}$  for  $i = 1, ..., \ell$  and let  $0 < a = \text{Re}(\lambda) < r$  as above. We consider the operator

$$R_{\alpha}(\lambda)A_1R_{\alpha}(\lambda)A_2R_{\alpha}(\lambda)\cdots R_{\alpha}(\lambda)A_{\ell}\tilde{R}_{\alpha}(\lambda),$$

where  $R_{\alpha}(\lambda) = (\lambda - (\operatorname{Id} + H_0(\alpha)^2))^{-1}$  and  $\tilde{R}_{\alpha}(\lambda) = (\lambda - (\operatorname{Id} + H_0(\alpha)^2 + X))^{-1}$ , where X is self-adjoint and  $H_0(\alpha)^2 + X \ge 0$ . Then for all  $M \ge 0$ 

$$R_{\alpha}(\lambda)A_{1}R_{\alpha}(\lambda)A_{2}\cdots A_{\ell}\tilde{R}_{\alpha}(\lambda) = \sum_{|k|=0}^{M} C_{\ell}(k)A_{1}^{(k_{1})}\cdots A_{\ell}^{(k_{\ell})}R_{\alpha}(\lambda)^{\ell+|k|}\tilde{R}_{\alpha}(\lambda) + P_{M,\ell}$$

where  $P_{M,\ell}$  is of order at most  $|p| - M - 4\ell - 5$ , and k and p are multi-indices. The constant  $C_{\ell}(k)$  is given by

$$C_{\ell}(k) = \frac{(|k| + \ell)!}{k_1! k_2! \cdots k_{\ell}! (k_1 + 1)(k_1 + k_2 + 2) \cdots (|k| + \ell)}.$$

We now show that when weighted by sufficiently many resolvents, the remainder term  $P_{M,\ell}$  can be uniformly bounded.

**Lemma 6.3.8.** With the assumptions and notation of the previous lemma including the assumption that  $A_i \in OP^{p_i}$  for each i, there is a positive constant C such that

$$\left| \left| (\lambda - (\mathrm{Id} + H_0(\alpha)^2))^{\ell + \frac{M}{4} + 1 - \frac{|p|}{4}} P_{M,\ell} \right| \right| \le C$$

independent of  $\lambda$  and t (although depending on M and  $\ell$  and the  $A_i$ ). If the final  $\tilde{R}_{\alpha}(\lambda)$  is replaced by  $R_{\alpha}(\lambda)$  then the factor 1 in the exponent can be replaced by  $\frac{5}{4}$ .

We require two more estimates to guarantee that the operators which arise from the resolvent expansion and the Cauchy integral formula are trace class.

**Lemma 6.3.9.** Let  $\ell$  be a non-negative integer, and for  $j = 0, ..., \ell$  let  $A_j \in OP^{p_j}$ . Fix  $a \in (0,r)$  and define the vertical line  $\gamma_{\alpha} = \{a + iv : v \in \mathbb{R}\}$ . For  $\lambda \in \gamma_{\alpha}$  let  $R_{\alpha}(\lambda) = (\lambda - (\mathrm{Id} + H_0(\alpha)^2))^{-1}$  and  $\tilde{R}_{\alpha}(\lambda) = (\lambda - (\mathrm{Id} + H_0(\alpha)^2 + X))^{-1}$ . Then for all  $\varepsilon > 0$  and for  $s \in \mathbb{C}$  with  $\mathrm{Re}(s) > \frac{1}{2}$  the operator

$$B_{\alpha}(t) = \frac{1}{2\pi i} \int_{\gamma_{\alpha}} \lambda^{-s} A_0 R_{\alpha}(\lambda) A_1 R_{\alpha}(\lambda) \cdots R_{\alpha}(\lambda) A_{\ell} \tilde{R}_{\alpha}(\lambda) d\lambda$$

is trace class for  $\operatorname{Re}(s) + \ell > \frac{p_0 + |p| + n + \varepsilon}{4}$ , and the function  $t \mapsto ||B_{\alpha}(t)||_1$  is integrable on [0,1].

In the case where the final resolvent is actually an  $R_{\alpha}(\lambda)$ , we can use Cauchy's integral formula to explicitly evaluate the expression obtained after performing the expansion and get sharper trace class estimates.

**Lemma 6.3.10.** Let  $\ell$  be a non-negative integer, and for  $j = 0, ..., \ell$  let  $A_j \in OP^{p_j}$ ,  $p_j \geq 0$ . Fix  $a \in (0,r)$  and define the vertical line  $\gamma_\alpha = \{a + iv : v \in \mathbb{R}\}$ . For  $\lambda \in \gamma_\alpha$  let  $R_\alpha(\lambda) = (\lambda - (\mathrm{Id} + H_0(\alpha)^2))^{-1}$  and  $\tilde{R}_\alpha(\lambda) = (\lambda - (\mathrm{Id} + H_0(\alpha)^2 + X))^{-1}$ . Then the operator

$$B_{\alpha}(t) = \frac{1}{2\pi i} \int_{\gamma_{\alpha}} \lambda^{-s} A_0 R_{\alpha}(\lambda) A_1 R_{\alpha}(\lambda) A_2 \cdots R_{\alpha}(\lambda) A_{\ell} R_{\alpha}(\lambda) d\lambda$$

is trace class for  $\operatorname{Re}(s) + \ell > \frac{n+p_0+|p|}{4}$ , and  $t \mapsto ||B_{\alpha}(t)||_1$  is integrable on [0,1].

Our application of this result will have  $p_0=0$  and each  $p_i=2$  (since  $A_i=X$  for all i), giving  $|p|=2\ell$  and so, if we enforce  $\text{Re}(s)>\frac{1}{2}$ , then a sufficient condition for  $\ell$  is to take  $\ell>\frac{n}{2}-1$ . If n is even we take  $L=\frac{n}{2}$  and if n is odd we take  $L=\frac{n-1}{2}$ .

## 6.4 Spectral flow for Schrödinger operators

The resolvent expansions of the previous section and the Birman-Kreĭn trace formula of Lemma 6.2.8 will be the crucial elements in computing our spectral flow. Define for  $Re(s) > \frac{1}{2}$  the constants

$$C_s = \int_{\mathbb{R}} (1 + x^2)^{-s} \, \mathrm{d}x.$$

The  $C_s$  have a pole at  $s = \frac{1}{2}$  with residue equal to one, a fact to be exploited later. Recall that for  $\text{Re}(s) > \frac{1}{2}$  we have defined the  $\eta$  function by

$$\eta_s(x) = \frac{1}{C_s} \int_1^\infty x (1 + wx^2)^{-s} w^{-\frac{1}{2}} dw = \frac{2}{C_s} \int_x^\infty (1 + v^2)^{-s} dv,$$

with the second formula only valid for x > 0. Note that  $\eta_s$  is odd and vanishes at 0 and  $\pm \infty$ . The result of Theorem 6.1.4 with  $s > \frac{n+2}{2}$  now reads

$$\operatorname{sf}(D_t) = \frac{1}{C_s} \int_0^1 \operatorname{Tr}(\dot{D}_t(\operatorname{Id} + D_t^2)^{-s}) \, dt + \frac{1}{2} \operatorname{Tr}(\eta_s(D_1) - \eta_s(D_0) + (P_{\operatorname{Ker}(D_1)} - P_{\operatorname{Ker}(D_0)})).$$
(6.3)

Multiplying both sides of Equation (6.3) by  $C_s$  we obtain

$$C_s \operatorname{sf}(D_t) = \int_0^1 \operatorname{Tr}(\dot{D}_t(\operatorname{Id} + D_t^2)^{-s}) dt + \frac{1}{2} C_s \operatorname{Tr}(\eta_s(D_1) - \eta_s(D_0) + (P_{\operatorname{Ker}(D_1)} - P_{\operatorname{Ker}(D_0)})).$$
(6.4)

The term  $C_s$  can be analytically continued for  $\text{Re}(s) > \frac{1}{2}$  to a meromorphic function with simple pole of residue 1 at  $s = \frac{1}{2}$ . Thus the left-hand side of Equation (6.4) is a meromorphic function and so the right hand side has a meromorphic continuation. We take the residue at  $s = \frac{1}{2}$  of both sides to obtain

$$\operatorname{sf}(D_{t}) = \operatorname{Res}_{s=\frac{1}{2}} \left( \int_{0}^{1} \operatorname{Tr}(\dot{D}_{t}(\operatorname{Id} + D_{t}^{2})^{-s}) dt + \frac{1}{2} C_{s} \operatorname{Tr}(\eta_{s}(D_{1}) - \eta_{s}(D_{0}) + (P_{\operatorname{Ker}(D_{1})} - P_{\operatorname{Ker}(D_{0})}) \right).$$

$$(6.5)$$

We apply Equation (6.5) to the path  $H_t(\alpha) = H_0 + \alpha + tV$ , so that  $\dot{H}_t(\alpha) = V$ . The Birman-Kreĭn formula of Lemma 6.2.8 will be fundamental to our evaluation of this spectral flow formula. We denote the total number of bound states (eigenvalues counted with multiplicity) by

$$N(\alpha) = \sum_{k=1}^{K} M(\lambda_k(\alpha)).$$

Recall that by construction,  $N(\alpha) = N$  (the number of bound states for H) independent of  $\alpha$ . We refer to the term

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_0^1 \operatorname{Tr}(V(\operatorname{Id} + H_t(\alpha)^2)^{-s}) dt \right)$$

as the 'integral of one form' contribution to the spectral flow, with this labelling justified by [47]. We refer to the term

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2} C_s \operatorname{Tr}(\eta_s(H(\alpha)) - \eta_s(H_0(\alpha)) + (P_{\operatorname{Ker}(H(\alpha))} - P_{\operatorname{Ker}(H_0(\alpha))}) \right)$$

as the Birman-Krein contribution to the spectral flow and the term

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2} C_s \operatorname{Tr}(\eta_s(D_1) - \eta_s(D_0)) \right)$$

as the  $\eta$  contribution to the spectral flow.

#### 6.4.1 The Birman-Kreĭn term

In this subsection we use the Birman-Kreĭn trace formula to determine the kernel and  $\eta$  contributions to the spectral flow in Equation (6.5).

**Lemma 6.4.1.** By construction, the operators  $P_{\text{Ker}(H(\alpha))}$  and  $P_{\text{Ker}(H_0(\alpha))}$  are both zero.

Since the kernel terms both vanish we are now able to evaluate the  $\eta$  contributions.

**Lemma 6.4.2.** The  $\eta$  contribution to the Hamiltonian spectral flow is given by

$$\operatorname{Res}_{s=\frac{1}{2}}(C_s\operatorname{Tr}(\eta_s(H_1(\alpha)) - \eta_s(H_0(\alpha))))$$

$$= -(N(\alpha) - 2M_{\alpha}(\alpha)) + \operatorname{Res}_{s=\frac{1}{2}}\left(\frac{1}{2\pi i} \int_{\alpha}^{\infty} C_s \eta_s(\lambda) \operatorname{Tr}(S_{\alpha}^*(\lambda) S_{\alpha}'(\lambda)) d\lambda\right)$$

$$+ (\xi_{\alpha}(\alpha -) - \xi_{\alpha}(\alpha +) - M_{\alpha}(\alpha)),$$

with  $M_{\alpha}(\alpha)$  the multiplicity of the eigenvalue at energy  $\alpha$ .

*Proof.* Suppose that  $\lambda_K(\alpha) = \alpha$ , that is H has zero as an eigenvalue. Apply the Birman-Kreĭn formula to obtain

$$\operatorname{Tr}(\eta_s(H(\alpha)) - \eta_s(H_0(\alpha))) = \frac{1}{2\pi i} \int_{\alpha}^{\infty} \eta_s(\lambda) \operatorname{Tr}(S_{\alpha}^*(\lambda) S_{\alpha}'(\lambda)) d\lambda + \sum_{k=1}^{K-1} M(\lambda_k(\alpha)) \eta_s(\lambda_k(\alpha)) + \eta_s(\alpha) \left(\xi_{\alpha}(\alpha -) - \xi_{\alpha}(\alpha +) - M(\alpha)\right).$$

To simplify this expression, for k < K we make the computation

$$\eta_s(\lambda_k(\alpha)) = -\eta_s(|\lambda_k(\alpha)|) = -\frac{2}{C_s} \int_{|\lambda_k(\alpha)|}^{\infty} (1+v^2)^{-s} \, \mathrm{d}v = -1 + \frac{2}{C_s} \int_0^{|\lambda_k(\alpha)|} (1+v^2)^{-s} \, \mathrm{d}v.$$

We further compute that

$$\eta_s(\alpha) = 1 - \frac{2}{C_s} \int_0^\alpha (1 + v^2)^{-s} dv.$$

Multiplying the trace by  $C_s$  and taking the residue we obtain

$$\operatorname{Res}_{s=\frac{1}{2}}(C_{s}\operatorname{Tr}(\eta_{s}(H_{1}) - \eta_{s}(H_{0})))$$

$$= \operatorname{Res}_{s=\frac{1}{2}}\left(\frac{1}{2\pi i}\int_{\alpha}^{\infty}C_{s}\eta_{s}(\lambda)\operatorname{Tr}(S_{\alpha}^{*}(\lambda)S_{\alpha}^{\prime}(\lambda))\,\mathrm{d}\lambda + \sum_{k=1}^{K-1}M(\lambda_{k}(\alpha))C_{s}\eta_{s}(\lambda_{k}(\alpha)) + C_{s}\eta_{s}(\alpha)\left(\xi_{\alpha}(\alpha-) - \xi_{\alpha}(\alpha+)\right)\right)$$

$$= \operatorname{Res}_{s=\frac{1}{2}}\left(\frac{1}{2\pi i}\int_{\alpha}^{\infty}C_{s}\eta_{s}(\lambda)\operatorname{Tr}(S_{\alpha}^{*}(\lambda)S_{\alpha}^{\prime}(\lambda))\,\mathrm{d}\lambda\right) - \sum_{k=1}^{K-1}M(\lambda_{k}(\alpha))$$

$$+ M(\alpha) + (\xi_{\alpha}(\alpha-) - \xi_{\alpha}(\alpha+) - M(\alpha))$$

$$= \operatorname{Res}_{s=\frac{1}{2}}\left(\frac{1}{2\pi i}\int_{\alpha}^{\infty}C_{s}\eta_{s}(\lambda)\operatorname{Tr}(S_{\alpha}^{*}(\lambda)S_{\alpha}^{\prime}(\lambda))\,\mathrm{d}\lambda\right) - (N(\alpha) - 2M(\alpha))$$

$$+ (\xi_{\alpha}(\alpha-) - \xi_{\alpha}(\alpha+) - M(\alpha)).$$

The case when  $\lambda_K(\alpha) \neq \alpha$  follows by considering  $M(\alpha) = 0$  in the above arguments.

It is not surprising that the spectral flow is linked to the behaviour of the spectral shift function, such a relationship has been demonstrated in a number of contexts see, for example [12, 13, 14, 37, 38, 39, 40, 42].

We now show how the Birman-Kreĭn formula of Lemma 6.4.2 can be used in conjunction with the high energy behaviour of the function  $\text{Tr}(S(\cdot)^*S'(\cdot))$  to relate to Levinson's theorem.

The key observation for n odd is the following.

**Lemma 6.4.3.** Fix  $f \in C(\mathbb{R}^+)$  and suppose there exists a polynomial p such that  $g(\lambda) = \lambda^{-\frac{1}{2}}p(\lambda)$  satisfies  $|f-g| \in L^1(\mathbb{R}^+)$ . Then

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\mathbb{R}^+} C_s \eta_s(\lambda) f(\lambda) \, d\lambda \right) = \int_{\mathbb{R}^+} \left( f(\lambda) - g(\lambda) \right) d\lambda.$$

*Proof.* We compute that

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\mathbb{R}^{+}} C_{s} \eta_{s}(\lambda) f(\lambda) \, d\lambda \right) = \operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\mathbb{R}^{+}} C_{s} \eta_{s}(\lambda) (f(\lambda) - g(\lambda) + g(\lambda)) \, d\lambda \right)$$
$$= \operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\mathbb{R}^{+}} C_{s} \eta_{s}(\lambda) (f(\lambda) - g(\lambda)) \, d\lambda \right)$$
$$+ \operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\mathbb{R}^{+}} C_{s} \eta_{s}(\lambda) g(\lambda) \, d\lambda \right).$$

Note that  $(s-\frac{1}{2})C_s\eta_s(\lambda)$  is uniformly bounded and thus the fact that  $|f-g|\in L^1(\mathbb{R}^+)$ 

allows us to use the dominated convergence theorem to obtain

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\mathbb{R}^+} C_s \eta_s(\lambda) (f(\lambda) - g(\lambda)) \, d\lambda \right) = \int_{\mathbb{R}^+} \operatorname{Res}_{s=\frac{1}{2}} (C_s \eta_s(\lambda)) (f(\lambda) - g(\lambda)) \, d\lambda$$
$$= \int_{\mathbb{R}^+} (f(\lambda) - g(\lambda)) \, d\lambda.$$

It remains to show that

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\mathbb{R}^+} C_s \eta_s(\lambda) g(\lambda) \, \mathrm{d}\lambda \right) = 0.$$

For  $r \neq -1$  we use integration by parts to calculate

$$\int_{\mathbb{R}^{+}} \lambda^{r} C_{s} \eta_{s}(\lambda) \, d\lambda = \int_{\mathbb{R}^{+}} \frac{d}{d\lambda} \left( \frac{\lambda^{r+1}}{r+1} \right) C_{s} \eta_{s}(\lambda) \, d\lambda 
= \left[ \frac{\lambda^{r+1}}{r+1} C_{s} \eta_{s}(\lambda) \right]_{0}^{\infty} - \frac{1}{r+1} \int_{\mathbb{R}^{+}} \lambda^{r+1} C_{s} \eta'_{s}(\lambda) \, d\lambda 
= \frac{2}{r+1} \int_{\mathbb{R}^{+}} \lambda^{r+1} (1+\lambda^{2})^{-s} \, d\lambda.$$

The final integral can be reduced to a form of the Beta function with the substitution  $\lambda = u^{\frac{1}{2}}$ , which yields

$$\int_{\mathbb{R}^+} \lambda^{r+1} (1+\lambda^2)^{-s} \, \mathrm{d}\lambda = \frac{1}{2} \int_{\mathbb{R}^+} u^{\frac{r}{2}} (1+u)^{-s} \, \mathrm{d}u = \frac{\Gamma\left(\frac{r}{2}+1\right) \Gamma\left(s-\frac{r}{2}-1\right)}{\Gamma(s)}.$$

Thus,

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\mathbb{R}^+} \lambda^r C_s \eta_s(\lambda) \, \mathrm{d}\lambda \right) = \operatorname{Res}_{s=\frac{1}{2}} \left( \frac{\Gamma\left(\frac{r}{2}+1\right) \Gamma\left(s-\frac{r}{2}-1\right)}{(r+1)\Gamma(s)} \right) = 0,$$

provided  $r \neq -2 + k$  for any  $k \in \mathbb{N}$ . We are interested in the case  $r = -\frac{1}{2} + k$  for  $k \in \mathbb{N}$ , since g is a linear combination of such powers. Hence,

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\mathbb{R}^+} C_s \eta_s(\lambda) g(\lambda) \, \mathrm{d}\lambda \right) = 0,$$

which completes the proof.

We now compute the residue of the  $\eta$  contribution in odd dimensions.

**Proposition 6.4.4.** Suppose that V satisfies Assumption 6.2.1 and n is odd. Then we

have

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) \, d\lambda \right)$$

$$= \frac{1}{2\pi i} \int_{0}^{\infty} \left( \operatorname{Tr}(S(\lambda)^{*} S'(\lambda)) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V) \lambda^{\frac{n-2}{2} - j} \right) \, d\lambda,$$

where the  $c_j$  are as defined in Theorem 2.5.34. Note that this statement is independent of the choice of  $\alpha \in (0, -\mu)$ .

*Proof.* By Theorem 2.5.34 the map

$$[\alpha, \infty) \ni \lambda \mapsto \operatorname{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)) - \sum_{i=1}^{\lfloor \frac{n-1}{2} \rfloor} c_j(n, V) (\lambda - \alpha)^{\frac{n-2}{2} - j}$$

is integrable. Thus we compute

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) \, d\lambda \right) \\
= \operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) \left( \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V)(\lambda - \alpha)^{\frac{n-2}{2} - j} \right) \, d\lambda \right) \\
+ \operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) - \left( \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V)(\lambda - \alpha)^{\frac{n-2}{2} - j} \right) \, d\lambda \right) \\
= \frac{1}{2\pi i} \int_{\alpha}^{\infty} \left( \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V)(\lambda - \alpha)^{\frac{n-2}{2} - j} \right) \, d\lambda \\
+ \operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{0}^{\infty} C_{s} \eta_{s}(u + \alpha) - \left( \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V) u^{\frac{n-2}{2} - j} \right) \, du \right),$$

where we have used the observation  $(s - \frac{1}{2})C_s\eta_s(\lambda)$  is uniformly bounded to apply the dominated convergence theorem. For  $k \in \mathbb{N}$  the estimate

$$\left| \int_{0}^{\infty} \int_{u+\alpha}^{u} u^{k-\frac{1}{2}} (1+v^{2})^{-s} \, dv \, du \right| \leq \int_{0}^{\infty} \int_{u}^{u+\alpha} u^{k-\frac{1}{2}} (1+v^{2})^{-\operatorname{Re}(s)} \, dv \, du$$

$$\leq \alpha \int_{0}^{\infty} u^{k-\frac{1}{2}} (1+u^{2})^{-\operatorname{Re}(s)} \, du$$

$$= \alpha \frac{\Gamma\left(\frac{k}{2} - \frac{1}{4}\right) \Gamma\left(\operatorname{Re}(s) - \frac{k}{2} + \frac{1}{4}\right)}{2\Gamma(\operatorname{Re}(s))}$$

shows that the map

$$s \mapsto \left(\frac{1}{2\pi i} \int_0^\infty C_s(\eta_s(u+\alpha) - \eta_s(u)) \left(\sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_j(n,V) u^{\frac{n-2}{2} - j}\right) du\right)$$

is holomorphic at  $s = \frac{1}{2}$ . So we find

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) d\lambda \right)$$

$$= \frac{1}{2\pi i} \int_{\alpha}^{\infty} \left( \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V) (\lambda - \alpha)^{\frac{n-2}{2} - j} \right) d\lambda$$

$$+ \operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{0}^{\infty} C_{s} \eta_{s}(u) \left( \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V) u^{\frac{n-2}{2} - j} \right) du \right).$$

Let  $f(\lambda) = (\text{Tr}(S(\lambda)^*S'(\lambda)))$  and  $g(\lambda) = p_n(\lambda)$  with  $p_n$  as in Definition 2.5.26. An application of Lemma 6.4.3 to the functions f, g then gives that the second residue in the above vanishes and thus

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) d\lambda \right)$$

$$= \frac{1}{2\pi i} \int_{\alpha}^{\infty} \left( \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V) (\lambda - \alpha)^{\frac{n-2}{2} - j} \right) d\lambda.$$

The statement of the proposition follows from the substitution  $\lambda = u + \alpha$ .

For n even the residue of the  $\eta$  term is non-zero and can be computed using Theorem 2.5.34 as follows.

**Proposition 6.4.5.** Suppose that n is even and V satisfies Assumption 6.2.1. Then

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) d\lambda \right) \\
= \frac{1}{2\pi i} \int_{0}^{\infty} \left( \operatorname{Tr}(S(\lambda)^{*} S'(\lambda)) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V) \lambda^{\frac{n-2}{2} - j} \right) d\lambda \\
+ \frac{1}{2\pi i} \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} \sum_{0 \le \ell \le \frac{n-2}{2} - j \text{ odd}} {\binom{\frac{n-2}{2} - j}{\ell}} c_{j}(n, V) (-1)^{j-\ell} \alpha^{j-\ell} \frac{\Gamma\left(\frac{\ell}{2} + 1\right) (-1)^{\frac{\ell+1}{2}}}{(\ell+1)\left(\frac{\ell+1}{2}\right)! \Gamma\left(\frac{1}{2}\right)} \\
- \frac{1}{2\pi i} \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} \sum_{\ell=0}^{\frac{n-2}{2} - j} c_{j}(n, V) {\binom{\frac{n-2}{2} - j}{\ell}} (-1)^{\frac{n-2}{2} - j-\ell} \alpha^{\frac{n-2}{2} - j-\ell} \frac{\alpha^{\ell+1}}{2(\ell+1)},$$

where the  $c_i(n, V)$  are defined in Definition 2.5.26.

*Proof.* By Theorem 2.5.34 the map

$$[\alpha, \infty) \ni \lambda \mapsto \operatorname{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_j(n, V) (\lambda - \alpha)^{\frac{n-2}{2} - j}$$

is integrable. Thus we compute

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) d\lambda \right) \\
= \operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) \left( \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V)(\lambda - \alpha)^{\frac{n-2}{2} - j} \right) d\lambda \right) \\
+ \operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) \left( \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V)(\lambda - \alpha)^{\frac{n-2}{2} - j} \right) d\lambda \right) \\
= \frac{1}{2\pi i} \int_{\alpha}^{\infty} \left( \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V)(\lambda - \alpha)^{\frac{n-2}{2} - j} \right) d\lambda \\
+ \operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) \left( \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V)(\lambda - \alpha)^{\frac{n-2}{2} - j} \right) d\lambda \right),$$

where we have used the observation  $(s - \frac{1}{2})C_s\eta_s(\lambda)$  is uniformly bounded to apply the dominated convergence theorem. For  $k \in \mathbb{N}$  we use the binomial theorem to obtain

$$\int_{\alpha}^{\infty} C_s \eta_s(\lambda) (\lambda - \alpha)^k \, d\lambda = \sum_{\ell=0}^{k} (-1)^{k-\ell} {k \choose \ell} \alpha^{k-\ell} \int_{\alpha}^{\infty} C_s \eta_s(\lambda) \lambda^{\ell} \, d\lambda.$$
 (6.6)

We integrate by parts to obtain

$$\int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) \lambda^{\ell} d\lambda 
= \frac{1}{\ell+1} \int_{\alpha}^{\infty} \lambda^{\ell+1} (1+\lambda^{2})^{-s} d\lambda - \frac{1}{\ell+1} \alpha^{\ell+1} \int_{\alpha}^{\infty} (1+v^{2})^{-s} dv 
= \frac{1}{\ell+1} \int_{0}^{\infty} \lambda^{\ell+1} (1+\lambda^{2})^{-s} d\lambda - \frac{1}{\ell+1} \alpha^{\ell+1} \int_{0}^{\infty} (1+v^{2})^{-s} dv 
- \frac{1}{\ell+1} \int_{0}^{\alpha} \lambda^{\ell+1} (1+\lambda^{2})^{-s} d\lambda + \frac{1}{\ell+1} \alpha^{\ell+1} \int_{0}^{\alpha} (1+v^{2})^{-s} dv 
= \frac{1}{2(\ell+1)} \frac{\Gamma(\frac{\ell}{2}+1) \Gamma(s-\frac{\ell}{2}-1)}{\Gamma(s)} - \frac{C_{s}}{2(\ell+1)} \alpha^{\ell+1} 
- \frac{1}{\ell+1} \int_{0}^{\alpha} \lambda^{\ell+1} (1+\lambda^{2})^{-s} d\lambda + \frac{1}{\ell+1} \alpha^{\ell+1} \int_{0}^{\alpha} (1+v^{2})^{-s} dv$$

Noting that the integrals from 0 to  $\alpha$  are over a finite region and thus define a holomorphic function of s, we can take the residue at  $s = \frac{1}{2}$  to obtain

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\alpha}^{\infty} C_s \eta_s(\alpha) \lambda^{\ell} d\lambda \right)$$

$$= -\frac{\alpha^{\ell+1}}{2(\ell+1)} + \begin{cases} 0, & \text{if } \ell \text{ is even,} \\ \frac{\Gamma\left(\frac{\ell}{2}+1\right)(-1)^{\frac{\ell+1}{2}}}{(\ell+1)\left(\frac{\ell+1}{2}\right)!\Gamma\left(\frac{1}{2}\right)}, & \text{if } \ell \text{ is odd.} \end{cases}$$

Returning to Equation (6.6) we have

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) (\lambda - \alpha)^{k} \, d\lambda \right)$$

$$= \sum_{0 \leq \ell \leq k \text{ odd}} {k \choose \ell} (-1)^{k-\ell} \alpha^{k-\ell} \frac{\Gamma\left(\frac{\ell}{2} + 1\right) (-1)^{\frac{\ell+1}{2}}}{(\ell+1)\left(\frac{\ell+1}{2}\right)!\Gamma\left(\frac{1}{2}\right)}$$

$$- \sum_{\ell=0}^{k} {k \choose \ell} (-1)^{k-\ell} \alpha^{k-\ell} \frac{\alpha^{\ell+1}}{2(\ell+1)}$$

So we find

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) \left( \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V)(\lambda - \alpha)^{\frac{n-2}{2} - j} \right) d\lambda \right) \\
= \frac{1}{2\pi i} \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} \sum_{0 \le \ell \le \frac{n-2}{2} - j \text{ odd}} {\binom{\frac{n-2}{2} - j}{\ell}} c_{j}(n, V)(-1)^{j-\ell} \alpha^{j-\ell} \frac{\Gamma\left(\frac{\ell}{2} + 1\right)(-1)^{\frac{\ell+1}{2}}}{(\ell+1)\left(\frac{\ell+1}{2}\right)!\Gamma\left(\frac{1}{2}\right)} \\
- \frac{1}{2\pi i} \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} \sum_{\ell=0}^{\frac{n-2}{2} - j} c_{j}(n, V) {\binom{\frac{n-2}{2} - j}{\ell}} (-1)^{\frac{n-2}{2} - j-\ell} \alpha^{\frac{n-2}{2} - j-\ell} \frac{\alpha^{\ell+1}}{2(\ell+1)}.$$

Hence we have

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) d\lambda \right) \\
= \frac{1}{2\pi i} \int_{\alpha}^{\infty} \left( \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V) (\lambda - \alpha)^{\frac{n-2}{2} - j} \right) d\lambda \\
+ \frac{1}{2\pi i} \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} \sum_{0 \le \ell \le \frac{n-2}{2} - j \text{ odd}} \left( \frac{\frac{n-2}{2} - j}{\ell} \right) c_{j}(n, V) (-1)^{j-\ell} \alpha^{j-\ell} \frac{\Gamma\left(\frac{\ell}{2} + 1\right) (-1)^{\frac{\ell+1}{2}}}{(\ell+1)\left(\frac{\ell+1}{2}\right)! \Gamma\left(\frac{1}{2}\right)} \\
- \frac{1}{2\pi i} \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} \sum_{\ell=0}^{\frac{n-2}{2} - j} c_{j}(n, V) \left( \frac{n-2}{2} - j \right) (-1)^{\frac{n-2}{2} - j-\ell} \alpha^{\frac{n-2}{2} - j-\ell} \frac{\alpha^{\ell+1}}{2(\ell+1)}.$$

The statement of the proposition follows from the substitution  $\lambda = u + \alpha$ .

In practice it is easier to take the limit as  $\alpha \to 0^+$  before taking the residue. To do so requires the following technical result.

**Lemma 6.4.6.** Fix  $f \in C(\mathbb{R}^+)$  and suppose there exists a polynomial  $p(\lambda) = \sum_{j=0}^k a_j \lambda^j$  such that  $|f - p| \in L^1(\mathbb{R}^+)$ . Then

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\mathbb{R}^+} C_s \eta_s(\lambda) f(\lambda) \, \mathrm{d}\lambda \right) = \int_{\mathbb{R}^+} \left( f(\lambda) - p(\lambda) \right) \, \mathrm{d}\lambda + \sum_{m=1}^{\lfloor \frac{k}{2} \rfloor} \frac{\Gamma\left(m + \frac{1}{2}\right) (-1)^m}{2m(m!)\Gamma\left(\frac{1}{2}\right)} a_{2m-1}.$$

*Proof.* The same computations as Lemma 6.4.3 apply until the evaluation of the residue. Recall that for  $r \neq -1$  we have

$$\int_{\mathbb{R}^+} \lambda^r C_s \eta_s(\lambda) \, d\lambda = \frac{\Gamma\left(\frac{r}{2} + 1\right) \Gamma\left(s - \frac{r}{2} - 1\right)}{(r+1)\Gamma(s)}.$$

We use the fact that the Gamma function has a pole of order 1 at each non-positive integer -k with residue  $\frac{(-1)^k}{\Gamma(k+1)}$  to find for a positive integer j that

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\mathbb{R}^+} \lambda^j C_s \eta_s(\lambda) \, \mathrm{d}\lambda \right) = \operatorname{Res}_{s=\frac{1}{2}} \left( \frac{\Gamma\left(\frac{j}{2}+1\right) \Gamma\left(s-\frac{j}{2}-1\right)}{(j+1)\Gamma(s)} \right).$$

If j is even, the residue vanishes. If j=2m-1 for some  $m\in\mathbb{N}$  then we find

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\mathbb{R}^+} \lambda^j C_s \eta_s(\lambda) \, \mathrm{d}\lambda \right) = \operatorname{Res}_{s=\frac{1}{2}} \left( \frac{\Gamma\left(m+\frac{1}{2}\right)\Gamma\left(s-m-\frac{1}{2}\right)}{2m\Gamma(s)} \right) = \frac{\Gamma\left(m+\frac{1}{2}\right)(-1)^m}{2m(m!)\Gamma\left(\frac{1}{2}\right)},$$

which completes the proof.

Corollary 6.4.7. Suppose that n is even and V satisfies Assumption 6.2.1. Then

$$\lim_{\alpha \to 0^{+}} \operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{\alpha}^{\infty} C_{s} \eta_{s}(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) \, \mathrm{d}\lambda \right)$$

$$= \frac{1}{2\pi i} \int_{0}^{\infty} \left( \operatorname{Tr}(S(\lambda)^{*} S'(\lambda)) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V) \lambda^{\frac{n-2}{2} - j} \right) \, \mathrm{d}\lambda$$

$$+ \frac{1}{2\pi i} \sum_{1 \le j \le \lfloor \frac{n-1}{2} \rfloor \text{ odd}} \left( \frac{\frac{n-2}{2} - j}{j} \right) c_{j}(n, V) \frac{\Gamma\left(\frac{j}{2} + 1\right) \left(-1\right)^{\frac{j+1}{2}}}{(j+1)\left(\frac{j+1}{2}\right)! \Gamma\left(\frac{1}{2}\right)}$$

$$= \operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{0}^{\infty} C_{s} \eta_{s}(\lambda) \operatorname{Tr}(S(\lambda)^{*} S'(\lambda)) \, \mathrm{d}\lambda \right),$$

where the  $c_j(n, V)$  are defined in Theorem 2.5.34.

*Proof.* By Theorem 2.5.34 we have that there exists  $c_i(n,V) \in \mathbb{C}$  with

$$\left(\mathbb{R}^+ \ni \lambda \mapsto \operatorname{Tr}(S(\lambda)^* S'(\lambda)) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_j(n, V) \lambda^{\frac{n-2}{2} - j} \right) \in L^1(\mathbb{R}^+).$$

An application of Lemma 6.4.6 then gives

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{0}^{\infty} C_{s} \eta_{s}(\lambda) \operatorname{Tr}(S(\lambda)^{*} S'(\lambda)) \, \mathrm{d}\lambda \right)$$

$$= \frac{1}{2\pi i} \int_{0}^{\infty} \left( \operatorname{Tr}(S(\lambda)^{*} S'(\lambda)) - \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_{j}(n, V) \lambda^{\frac{n-2}{2} - j} \right) \, \mathrm{d}\lambda$$

$$+ \frac{1}{2\pi i} \sum_{1 \leq j \leq \lfloor \frac{n-1}{2} \rfloor \text{ odd}} {\binom{\frac{n-2}{2} - j}{j}} c_{j}(n, V) \frac{\Gamma\left(\frac{j}{2} + 1\right) \left(-1\right)^{\frac{j+1}{2}}}{(j+1)\left(\frac{j+1}{2}\right)!\Gamma\left(\frac{1}{2}\right)}.$$

Taking the limit as  $\alpha \to 0^+$  in Proposition 6.4.5 completes the proof.

Since there are far less terms to consider, in practice it is more convenient to use Corollary 6.4.7 than Proposition 6.4.5. We demonstrate the use of Proposition 6.4.5 in dimension n = 4, the lowest dimension in which non-trivial  $\alpha$  dependence is found.

Corollary 6.4.8. Suppose that n = 4 and V satisfies Assumption 6.2.1. Let  $c_1(4, V)$  be as in Theorem 2.5.34. Then

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_{\alpha}^{\infty} C_s \eta_s(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)) \, d\lambda \right) = \frac{1}{2\pi i} \int_{0}^{\infty} \left( \operatorname{Tr}(S(\lambda)^* S'(\lambda)) - c_1(4, V) \right) \, d\lambda - \frac{c_1(4, V)}{2(2\pi i)} \alpha.$$

These computations will be used to relate spectral flow to Levinson's theorem in Section 6.5.

## 6.4.2 The "integral of one form" term

Finally we determine the contribution of the "integral of one form" term to the spectral flow formula with the aid of some technical results. As before we fix  $r = \frac{1}{2} \min\{\alpha^2, (\mu + \alpha)^2\}$ .

**Lemma 6.4.9.** For  $\alpha \in (0, -\mu)$ ,  $\text{Re}(s) > \frac{1}{2}$  and  $a \in (0, r)$  define the vertical line  $\gamma_{\alpha} = \{a + iv : v \in \mathbb{R}\}$ . Define the perturbation  $X = X(t, \alpha) = t\{H_0, V\} + 2t\alpha V + t^2 V^2$ . Then

for each  $L \in \mathbb{N}$ , we have

$$(\operatorname{Id} + H_t(\alpha)^2)^{-s} = \sum_{\ell=0}^{L} \frac{1}{2\pi i} \int_{\gamma_{\alpha}} \lambda^{-s} (X(\lambda - (\operatorname{Id} + H_0(\alpha)^2))^{-1})^{\ell} (\lambda - (\operatorname{Id} + H_0(\alpha)^2))^{-1} d\lambda + \frac{1}{2\pi i} \int_{\gamma_{\alpha}} \lambda^{-s} (X(\lambda - (\operatorname{Id} + H_0(\alpha)^2))^{-1})^{L+1} (\lambda - (\operatorname{Id} + H_t(\alpha)^2))^{-1} d\lambda.$$
(6.7)

*Proof.* An application of Cauchy's integral formula gives

$$(\mathrm{Id} + H_t(\alpha)^2)^{-s} = \frac{1}{2\pi i} \int_{\gamma_\alpha} \lambda^{-s} (\lambda - (\mathrm{Id} + H_t(\alpha)^2))^{-1} \, \mathrm{d}\lambda.$$

For self-adjoint X, Y the resolvent identity

$$(\lambda - (X+Y))^{-1} = \sum_{\ell=0}^{L} (X(\lambda - Y)^{-1})^{\ell} (\lambda - Y)^{-1} + (X(\lambda - Y)^{-1})^{L} (\lambda - (X+Y))^{-1}$$

completes the proof.

**Lemma 6.4.10.** If n is even, let  $L = \frac{n}{2}$  and if n is odd, let  $L = \frac{n-1}{2}$ . Let  $X = X(t, \alpha)$  by given by  $X = t\{H_0, V\} + 2t\alpha V + t^2V^2$  and  $Re(s) > \frac{1}{2}$ . For a multi-index k and  $0 \le \ell \le L$ , define the constant  $Y_{\ell}(k, s)$  by

$$Y_{\ell}(k,s) = (-1)^{\ell+|k|} C_{\ell}(k) \frac{\Gamma(\ell+|k|+s)}{\Gamma(s)(\ell+|k|)!}$$

For 4Re(s) + L > n we have

$$\operatorname{Tr}(V(\operatorname{Id} + (H_t(\alpha))^2)^{-s}) = \sum_{\ell=0}^{L} \sum_{|k|=0}^{M_{\ell}} Y_{\ell}(k, s) \operatorname{Tr}\left(VX^{(k_1)} \cdots X^{(k_{\ell})} (\operatorname{Id} + H_0(\alpha)^2)^{-s-\ell-|k|}\right) + P,$$
(6.8)

where  $M_{\ell} = \frac{n}{2} + 2 + \ell$  and P is holomorphic at  $s = \frac{1}{2}$ .

*Proof.* Apply Lemma 6.4.9 to obtain

$$(\operatorname{Id} + H_t(\alpha)^2)^{-s} = \sum_{\ell=0}^{L} \frac{1}{2\pi i} \int_{\gamma_{\alpha}} \lambda^{-s} (X(\lambda - (\operatorname{Id} + H_0(\alpha)^2))^{-1})^{\ell} (\lambda - (\operatorname{Id} + H_0(\alpha)^2))^{-1} d\lambda + \frac{1}{2\pi i} \int_{\gamma_{\alpha}} \lambda^{-s} (X(\lambda - (\operatorname{Id} + H_0(\alpha)^2))^{-1})^{L+1} (\lambda - (\operatorname{Id} + H_t(\alpha)^2))^{-1} d\lambda.$$

An application of Lemma 6.3.10 shows that the term

$$\frac{1}{2\pi i} \int_{\gamma_{\alpha}} \lambda^{-s} (X(\lambda - (\mathrm{Id} + H_0(\alpha)^2))^{-1})^{L+1} (\lambda - (\mathrm{Id} + H_t(\alpha)^2))^{-1} d\lambda$$

is trace class for  $L+1>\frac{n}{2}$ . An argument similar to the proof of [46, Lemma 7.4] shows that this term is holomorphic in s also. An application of Lemma 6.3.9 shows that the term

$$\frac{1}{2\pi i} \int_{\gamma_{\alpha}} \lambda^{-s} (X(\lambda - (\mathrm{Id} + H_0(\alpha)^2))^{-1})^{\ell} (\lambda - (\mathrm{Id} + H_0(\alpha)^2))^{-1} d\lambda$$

is trace class for  $\text{Re}(s) > \frac{n}{4} - \ell + \frac{|p|+p_0}{4}$ . We have  $0 \le |p| + p_0 \le 2\ell$  and so we require  $\text{Re}(s) > \frac{n}{4} - \frac{\ell}{2}$ . Thus for  $\ell \ge \frac{n}{2} - 1$  we find that each term in the sum will have a non-zero residue and all terms with  $L > \frac{n}{2}$  will have vanishing residue. We now apply Lemma 6.3.7 to obtain for any  $M \ge 0$  the expansion

$$\frac{1}{2\pi i} \int_{\gamma_{\alpha}} \lambda^{-s} (X(\lambda - (\mathrm{Id} + H_{0}(\alpha)^{2}))^{-1})^{\ell} (\lambda - (\mathrm{Id} + H_{0}(\alpha)^{2}))^{-1} d\lambda 
= \sum_{|k|=0}^{M} \frac{1}{2\pi i} \int_{\gamma_{\alpha}} \lambda^{-s} C_{\ell}(k) X^{(k_{1})} \cdots X^{(k_{\ell})} (\lambda - (\mathrm{Id} + H_{0}(\alpha)^{2}))^{-\ell-|k|-1} d\lambda 
+ \frac{1}{2\pi i} \int_{\gamma_{\alpha}} P_{M,\ell}(\lambda) (\lambda - (\mathrm{Id} + H_{0}(\alpha)^{2}))^{-1} d\lambda.$$

An application of Cauchy's integral formula shows that the first term can be evaluated explicitly to obtain

$$\frac{1}{2\pi i} \int_{\gamma_{\alpha}} \lambda^{-s} (X(\lambda - (\mathrm{Id} + H_0(\alpha)^2))^{-1})^{\ell} (\lambda - (\mathrm{Id} + H_0(\alpha)^2))^{-1} d\lambda 
= \sum_{|k|=0}^{M} Y_{\ell}(k, s) X^{(k_1)} \cdots X^{(k_{\ell})} (\mathrm{Id} + H_0(\alpha)^2))^{-\ell - |k| - s} 
+ \frac{1}{2\pi i} \int_{\gamma_{\alpha}} P_{M,\ell}(\lambda) (\lambda - (\mathrm{Id} + H_0(\alpha)^2))^{-1} d\lambda.$$

The above expressions are trace class and for  $M \ge \frac{n}{2} + 1 + \ell$  the remainder is holomorphic at  $s = \frac{1}{2}$ .

## 6.4.3 The spectral flow formulae

In this subsection we combine the results of Sections 6.4.1 and 6.4.2 to determine an expression for the spectral flow along the path  $H_t(\alpha)$ . Such a formula is dependent on the parity of the dimension and the expressions we obtain are rather complicated, with many terms obtained by combinatorial arguments which are difficult to evaluate explicitly. To

do so requires the following integral computations.

**Lemma 6.4.11.** For a, b > 0 with  $b > \frac{a}{4}$  we have

$$\int_0^\infty r^{a-1} (1+r^4)^{-b} dr = \frac{\Gamma\left(\frac{a}{4}\right) \Gamma\left(b-\frac{a}{4}\right)}{4\Gamma(b)}.$$
 (6.9)

For  $b > \frac{n+2}{4}$  and  $1 \le m \le n$  we have

$$\int_{\mathbb{R}^n} \xi_m^2 (1+|\xi|^4)^{-b} \,\mathrm{d}\xi = \frac{\mathrm{Vol}(\mathbb{S}^{n-1})}{n} \frac{\Gamma\left(\frac{n+2}{4}\right) \Gamma\left(b - \frac{n+2}{4}\right)}{4\Gamma(b)},\tag{6.10}$$

independent of the direction  $\xi_m$ .

*Proof.* Making the substitution  $u = r^4$  we find

$$\int_0^\infty r^{a-1} (1+r^4)^{-b} dr = \frac{1}{4} \int_0^\infty u^{\frac{a}{4}-1} (1+u)^{-b} du = \frac{\Gamma\left(\frac{a}{4}\right) \Gamma\left(b-\frac{a}{4}\right)}{4\Gamma\left(b\right)},$$

where the integral was evaluated as a known form of the Beta function. For the second integral, we note that by symmetry we have

$$\int_{\mathbb{R}^n} \xi_m^2 (1+|\xi|^4)^{-\beta} \,\mathrm{d}\xi = \frac{1}{n} \int_{\mathbb{R}^n} |\xi|^2 (1+|\xi|^4)^{-b} \,\mathrm{d}\xi.$$

Changing to polar coordinates and using Equation (6.9) gives Equation (6.10).

**Lemma 6.4.12.** For m = 2k a positive even integer and  $q \in \mathbb{N}_0$  we have

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{0}^{\infty} r^{m-1} (1 + (r^{2} + \alpha)^{2})^{-s} \, dr \right)$$

$$= \operatorname{Res}_{s=\frac{1}{2}} \left( \sum_{\ell=0}^{k-1} {k-1 \choose \ell} (-1)^{k-\ell-1} \alpha^{k-\ell-1} \frac{\Gamma\left(\frac{\ell+1}{2}\right) \Gamma\left(s+q-\frac{\ell+1}{2}\right)}{4\Gamma(s+q)} \right).$$

*Proof.* Let m=2k and compute that

$$\int_0^\infty r^{2k-1} (1 + (r^2 + \alpha)^2)^{-s-q} dr = \frac{1}{2} \int_0^\infty u^{k-1} (1 + (u + \alpha)^2)^{-s-q} du$$
$$= \frac{1}{2} \int_\alpha^\infty (v - \alpha)^{k-1} (1 + v^2)^{-s-q} dv.$$

Applying the binomial expansion we have

$$\int_0^\infty r^{2k-1} (1 + (r^2 + \alpha)^2)^{-s} dr = \frac{1}{2} \int_\alpha^\infty (v - \alpha)^{k-1} (1 + v^2)^{-s-q} dv$$

$$= \frac{1}{2} \sum_{\ell=0}^{k-1} \binom{k}{\ell} (-1)^{k-\ell-1} \alpha^{k-\ell-1} \int_\alpha^\infty v^\ell (1 + v^2)^{-s-q} dv.$$

Note that

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_0^\alpha v^j (1+v^2)^{-s-q} \, \mathrm{d}v \right) = 0.$$

Thus we obtain

$$\begin{aligned} & \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( \int_{0}^{\infty} r^{2k-1} (1 + (r^{2} + \alpha)^{2})^{-s} \, \mathrm{d}r \right) \\ & = \frac{1}{2} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( \sum_{\ell=0}^{k-1} {k-1 \choose \ell} (-1)^{k-\ell-1} \alpha^{k-\ell-1} \int_{\alpha}^{\infty} v^{\ell} (1 + v^{2})^{-s-q} \, \mathrm{d}v \right) \\ & = \frac{1}{2} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( \sum_{\ell=0}^{k-1} {k-1 \choose \ell} (-1)^{k-\ell-1} \alpha^{k-\ell-1} \int_{0}^{\infty} v^{\ell} (1 + v^{2})^{-s-q} \, \mathrm{d}v \right) \\ & = \frac{1}{4} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( \sum_{\ell=0}^{k-1} {k-1 \choose \ell} (-1)^{k-\ell-1} \alpha^{k-\ell-1} \int_{0}^{\infty} w^{\frac{\ell+1}{2}-1} (1 + w)^{-s-q} \, \mathrm{d}v \right) \\ & = \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( \sum_{\ell=0}^{k-1} {k-1 \choose \ell} (-1)^{k-\ell-1} \alpha^{k-\ell-1} \frac{\Gamma\left(\frac{\ell+1}{2}\right) \Gamma\left(s + q - \frac{\ell+1}{2}\right)}{4\Gamma(s+q)} \right), \end{aligned}$$

which is the statement of the lemma.

To avoid confusion, we introduce the notation  $(\Delta V)$  for the action of the Laplacian  $H_0$  on the potential V.

**Lemma 6.4.13.** For n even we define

$$X_0 = 2t\alpha V + t^2 V^2 + t(\Delta V), \quad X_1 = -2t \sum_{j=1}^n \frac{\partial V}{\partial x_j} \frac{\partial}{\partial x_j}$$
 and  $X_2 = 2tV H_0$ .

For  $0 \le \ell \le \frac{n}{2}$  define the set

$$J_{\ell}(n) = \left\{ (k, p) \in \mathbb{N}_{0}^{\ell} \times \{0, 1, 2\}^{\ell} : |p| + n \ge 2 + 4\ell + |k| \right\}.$$

Then we have

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{0}^{1} \operatorname{Tr}(V(\operatorname{Id} + H_{t}(\alpha)^{2})^{-s}) \, dt \right)$$

$$= \operatorname{Res}_{s=\frac{1}{2}} \left( \sum_{\ell=0}^{\frac{n}{2}} \sum_{(k,p)\in J_{\ell}(n)} Y_{\ell}(k,s) \int_{0}^{1} \operatorname{Tr}\left(VX_{p_{1}}^{(k_{1})} \cdots X_{p_{\ell}}^{(k_{\ell})}(\operatorname{Id} + H_{0}(\alpha)^{2})^{-s-\ell-|k|}\right) \, dt \right). \quad (6.11)$$

*Proof.* By Lemma 6.4.10 we can write the expression (modulo holomorphic terms)

$$\operatorname{Tr}(V(\operatorname{Id} + (H(\alpha))^{2})^{-s}) = \sum_{\ell=0}^{L} \sum_{|k|=0}^{M} Y_{\ell}(k,s) \operatorname{Tr}\left(VX^{(k_{1})} \cdots X^{(k_{\ell})} (\operatorname{Id} + H_{0}(\alpha)^{2})^{-s-\ell-|k|}\right).$$

We have written

$$X = \left(2t\alpha V + t^2 V^2 + t(\Delta V)\right) - 2t \sum_{j=1}^{n} \frac{\partial V}{\partial x_j} \frac{\partial}{\partial x_j} + 2tV H_0 = X_0 + X_1 + X_2$$

as the sum of an order zero, an order one and an order two component. The terms in the summand which are not trace class satisfy  $|p| + 3|k| - 4\text{Re}(s) - 4\ell - 4|k| \ge -n$  and with our decomposition of X we have  $p_i \in \{0, 1, 2\}$ . Thus for fixed  $n, \ell$  the terms which are not trace class are precisely those with  $(k, p) \in J_{\ell}(n)$ . Summing over these terms and taking the residue at  $s = \frac{1}{2}$  gives the result.

#### Lemma 6.4.14. For n odd we have

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_0^1 \operatorname{Tr}(V(\operatorname{Id} + H_t(\alpha)^2)^{-s}) \, \mathrm{d}t \right) = 0.$$

*Proof.* For n odd, the same argument as n even in Lemma 6.4.13 shows that Equation (6.11) holds with  $X_1, X_2, X_3$  defined as in Lemma 6.4.13. We show that all residues vanish for one of two reasons. Write  $X_{p_1}^{(k_1)} \cdots X_{p_\ell}^{(k_\ell)} = \sum_{j=0}^{|p|+3|k|} G_j$ , where  $G_j$  is an operator of order j. By performing commutators, we can assume without loss of generality that  $G_j$  is of the form

$$G_j = g \sum_{|\beta|=j} g_{\beta} \partial^{\beta}$$

for some multi-index  $\beta$  of length n and  $g_{\beta} \in C_c^{\infty}(\mathbb{R}^n)$ . Then we find

$$\operatorname{Tr}\left(VG_{j}(\operatorname{Id} + H_{0}(\alpha)^{2})^{-s-\ell-|k|}\right) = (2\pi)^{-n} \sum_{|\beta|=j} \left( \int_{\mathbb{R}^{n}} V(x)g_{\beta}(x) \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^{n}} \xi^{\beta} (1 + (|\xi|^{2} + \alpha)^{2})^{-s-\ell-|k|} \, \mathrm{d}\xi \right).$$

If any of the elements of the multi-index  $\beta$  are odd, the trace vanishes for the same reason. Thus we may assume that j is even and each component  $\beta_i$  of  $\alpha$  is even. Write  $\beta_i = 2b_i$  for a multi-index b of length n. Then we can write

$$G_j = \sum_{\substack{2|b|=j\\2}} g_{2b} \partial_1^{2b_1} \cdots \partial_n^{2b_n}.$$

We find that

$$\operatorname{Tr}\left(VG_{j}(\operatorname{Id} + H_{0}(\alpha)^{2})^{-s-\ell-|k|}\right) \\
= i^{|\beta|}(2\pi)^{-n} \sum_{2|b|=j} \left( \int_{\mathbb{R}^{n}} V(x)g_{2b}(x) \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^{n}} \xi_{1}^{2b_{1}} \cdots \xi_{n}^{2b_{n}} (1 + (|\xi|^{2} + \alpha)^{2})^{-s-\ell-|k|} \, \mathrm{d}\xi \right) \\
= i^{|\beta|}(2\pi)^{-n} \sum_{2|b|=j} \left( \int_{\mathbb{R}^{n}} V(x)g_{2b}(x) \, \mathrm{d}x \right) \left( \int_{\mathbb{S}^{n-1}} \omega_{1}^{2b_{1}} \cdots \omega_{n}^{2b_{n}} \, \mathrm{d}\omega \right) \times \\
\left( \int_{0}^{\infty} r^{j} (1 + (r^{2} + \alpha)^{2})^{-s-\ell-|k|} \, \mathrm{d}r \right) \\
= i^{|\beta|} \frac{2\Gamma\left(\frac{1}{2} + b_{1}\right) \cdots \Gamma\left(\frac{1}{2} + b_{n}\right)}{\Gamma\left(\frac{n}{2} + |b|\right) (2\pi)^{n}} \sum_{2|b|=j} \left( \int_{\mathbb{R}^{n}} V(x)g_{2b}(x) \, \mathrm{d}x \right) \left( \int_{\mathbb{S}^{n-1}} \omega_{1}^{2b_{1}} \cdots \omega_{n}^{2b_{n}} \, \mathrm{d}\omega \right) \times \\
\left( \int_{0}^{\infty} r^{n+j-1} (1 + (r^{2} + \alpha)^{2})^{-s-\ell-|k|} \, \mathrm{d}r \right),$$

where we have used Lemma 2.5.20 to evaluate the integral over  $\mathbb{S}^{n-1}$ . We make the estimate

$$\left| \int_0^\infty |r^{n+j-1} (1 + (r^2 + \alpha)^2)^{-s-\ell-|k|} \, dr \right| \le \int_0^\infty r^{n+j-1} (1 + r^4)^{-\operatorname{Re}(s)-\ell-|k|} \, dr$$

$$= \frac{\Gamma\left(\frac{n+j}{4}\right) \Gamma\left(\operatorname{Re}(s) + \ell + |k| - \frac{n+j}{4}\right)}{4\Gamma\left(\operatorname{Re}(s) + \ell + |k|\right)}.$$

Since n is odd and j is even, we find that  $\frac{1}{2} + \ell + |k| - \frac{n+j}{4}$  is never an integer and so

$$s \mapsto \int_0^\infty r^{n+j-1} (1 + (r^2 + \alpha)^2)^{-s-\ell-|k|} dr$$

is holomorphic at  $s = \frac{1}{2}$ . Thus  $\operatorname{Tr}\left(VG_j(\operatorname{Id} + H_0(\alpha)^2)^{-s-\ell-|k|}\right)$  is holomorphic at  $s = \frac{1}{2}$  and we have shown that for n odd all residues at  $s = \frac{1}{2}$  vanish, independent of  $\alpha > 0$ .  $\square$ 

In higher even dimensions the terms appearing in Lemma 6.4.13 become cumbersome to compute rather quickly, however in dimension n = 2, 4, 6 they can be computed directly, with the aid of Lemmas 6.4.11 and 6.4.12.

**Lemma 6.4.15.** Let n = 2 and suppose V satisfies Assumption 6.2.1. Then

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{0}^{1} \operatorname{Tr}(V(\operatorname{Id} + H_{t}(\alpha)^{2})^{-s}) \, dt \right) = \frac{1}{8\pi} \int_{\mathbb{R}^{2}} V(x) \, dx,$$

independent of  $\alpha > 0$ .

*Proof.* Since n=2 we find that  $J_0(2)=\{(0,0)\}$  and  $J_1(2)=\emptyset$ . Noting that for  $\ell=0$  we

have  $Y_0(0,\frac{1}{2})=1$  we thus have

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_0^1 \operatorname{Tr}(V(\operatorname{Id} + H_t(\alpha)^2)^{-s}) \, \mathrm{d}t \right) = \operatorname{Res}_{s=\frac{1}{2}} \left( \operatorname{Tr}(V(\operatorname{Id} + H_0(\alpha)^2)^{-s}) \right).$$

So we compute that

$$\operatorname{Tr}(V(\operatorname{Id} + H_0(\alpha)^2)^{-s}) = (2\pi)^{-2} \left( \int_{\mathbb{R}^2} V(x) \, dx \right) \left( \int_{\mathbb{R}^2} (1 + (|y|^2 + \alpha)^2)^{-s} \, dy \right)$$
$$= (2\pi)^{-2} \operatorname{Vol}(\mathbb{S}^1) \left( \int_{\mathbb{R}^2} V(x) \, dx \right) \left( \int_0^\infty r(1 + (r^2 + \alpha)^2)^{-s} \, dr \right),$$

where we have used polar coordinates and Lemma 6.4.11 to evaluate the y integral. Taking the residue at  $s = \frac{1}{2}$  using Lemma 6.4.12 we find

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \operatorname{Tr}(V(\operatorname{Id} + H_0(\alpha)^2)^{-s}) \right) = (2\pi)^{-2} \operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\mathbb{R}^2} V(x) \, \mathrm{d}x \right) \left( \operatorname{Vol}(\mathbb{S}^1) \frac{\Gamma\left(\frac{1}{2}\right) \Gamma\left(s - \frac{1}{2}\right)}{4\Gamma(s)} \right)$$
$$= \frac{\operatorname{Vol}(\mathbb{S}^1)}{4(2\pi)^2} \left( \int_{\mathbb{R}^2} V(x) \, \mathrm{d}x \right),$$

independent of  $\alpha$ .

Remark 6.4.16. In the notation of Theorems 2.5.34 and 5.3.29, the statement of Lemma 6.4.15 is

Res<sub>$$s=\frac{1}{2}$$</sub>  $\left( \int_0^1 \text{Tr}(V(\text{Id} + H_t(\alpha)^2)^{-s}) dt \right) = \frac{1}{2}b_2,$ 

independent of  $\alpha > 0$ .

**Lemma 6.4.17.** Let n = 4 and suppose V satisfies Assumption 6.2.1. Then

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_0^1 \operatorname{Tr}(V(\operatorname{Id} + H_t(\alpha)^2)^{-s}) \, \mathrm{d}t \right) = -\alpha \frac{\operatorname{Vol}(\mathbb{S}^3)}{8(2\pi)^4} \int_{\mathbb{R}^4} V(x) \, \mathrm{d}x - \frac{\operatorname{Vol}(\mathbb{S}^3)}{8(2\pi)^6} \int_{\mathbb{R}^4} V(x)^2 \, \mathrm{d}x.$$

Proof. Since n = 4 we have  $J_1(4) = \{(0,2)\}$ ,  $J_0(4) = \{(0,0)\}$  and  $J_{\ell}(4) = \emptyset$  for  $\ell \geq 2$ . Thus only two terms corresponding to  $\ell = 1, k = 0, p = 2$  and  $\ell = 0, k = 0 = p$  will contribute. We first consider  $\ell = 0, k = 0, p = 0$ . We find

$$\operatorname{Res}_{s=\frac{1}{2}} \left( Y_0(0,s) \int_0^1 \operatorname{Tr}(V(\operatorname{Id} + H_0(\alpha)^2)^{-s}) \, \mathrm{d}t \right) \\
= (2\pi)^{-4} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_0(0,s) \left( \int_{\mathbb{R}^4} V(x) \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^4} (1 + (|\xi|^2 + \alpha)^2)^{-s} \, \mathrm{d}\xi \right) \right) \\
= \frac{\operatorname{Vol}(\mathbb{S}^3)}{(2\pi)^4} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_0(0,s) \left( \int_{\mathbb{R}^4} V(x) \, \mathrm{d}x \right) \left( \int_0^\infty r^{4-1} (1 + (r^2 + \alpha)^2)^{-s} \, \mathrm{d}r \right) \right).$$

П

Applying Lemma 6.4.12 we have

$$\operatorname{Res}_{s=\frac{1}{2}} \left( Y_{0}(0,s) \int_{0}^{1} \operatorname{Tr}(V(\operatorname{Id} + H_{0}(\alpha)^{2})^{-s}) \, dt \right) \\
= \frac{\operatorname{Vol}(\mathbb{S}^{3})}{(2\pi)^{4}} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{0}(0,s) \left( \int_{\mathbb{R}^{4}} V(x) \, dx \right) \left( \int_{0}^{\infty} r^{4-1} (1 + (r^{2} + \alpha)^{2})^{-s} \, dr \right) \right) \\
= \frac{\operatorname{Vol}(\mathbb{S}^{3})}{(2\pi)^{4}} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{0}(0,s) \left( \int_{\mathbb{R}^{4}} V(x) \, dx \right) \left( -\alpha \frac{\Gamma\left(\frac{1}{2}\right) \Gamma\left(s - \frac{1}{2}\right)}{4\Gamma(s)} + \frac{\Gamma\left(1\right) \Gamma\left(s - 1\right)}{4\Gamma(s)} \right) \right) \\
= -\alpha \frac{\operatorname{Vol}(\mathbb{S}^{3})}{8(2\pi)^{4}} \int_{\mathbb{R}^{4}} V(x) \, dx.$$

We next consider the case  $\ell=1, k=0, p=2$ . We observe that  $Y_1(0,\frac{1}{2})=-\frac{1}{2}$  and find

$$\operatorname{Res}_{s=\frac{1}{2}} \left( Y_1(0,s) \int_0^1 2t \operatorname{Tr}(V^2 H_0(\operatorname{Id} + H_0(\alpha)^2)^{-s-1}) \, \mathrm{d}t \right) \\
= (2\pi)^{-4} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_1(0,s) \left( \int_{\mathbb{R}^4} V(x)^2 \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^4} |\xi|^2 (1 + (|\xi|^2 + \alpha)^2)^{-s-1} \, \mathrm{d}\xi \right) \right) \\
= \frac{\operatorname{Vol}(\mathbb{S}^3)}{(2\pi)^4} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_1(0,s) \left( \int_{\mathbb{R}^4} V(x)^2 \, \mathrm{d}x \right) \left( \int_0^\infty r^{6-1} (1 + (r^2 + \alpha)^2)^{-s-1} \, \mathrm{d}r \right) \right).$$

Applying Lemma 6.4.12 we have

$$\operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \int_{0}^{1} 2t \operatorname{Tr}(V^{2} H_{0}(\operatorname{Id} + H_{0}(\alpha)^{2})^{-s-1}) dt \right) \\
= \frac{\operatorname{Vol}(\mathbb{S}^{3})}{(2\pi)^{4}} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \left( \int_{\mathbb{R}^{4}} V(x)^{2} dx \right) \left( \int_{0}^{\infty} r^{6-1} (1 + (r^{2} + \alpha)^{2})^{-s-1} dr \right) \right) \\
= \frac{\operatorname{Vol}(\mathbb{S}^{3})}{(2\pi)^{4}} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \left( \int_{\mathbb{R}^{4}} V(x)^{2} dx \right) \left( \alpha^{2} \frac{\Gamma\left(\frac{1}{2}\right) \Gamma\left(s + \frac{1}{2}\right)}{4\Gamma(s+1)} - 2\alpha \frac{\Gamma\left(1\right) \Gamma\left(s\right)}{4\Gamma(s+1)} + \frac{\Gamma\left(\frac{3}{2}\right) \Gamma\left(s - \frac{1}{2}\right)}{4\Gamma(s+1)} \right) \right) \\
+ \frac{\Gamma\left(\frac{3}{2}\right) \Gamma\left(s - \frac{1}{2}\right)}{4\Gamma(s+1)} \right) \\
= -\frac{\operatorname{Vol}(\mathbb{S}^{3})}{8(2\pi)^{4}} \int_{\mathbb{R}^{4}} V(x)^{2} dx.$$

Adding the two contributions gives the statement.

Remark 6.4.18. In the notation of Theorem 2.5.34 and Definition 2.5.26, the statement of Lemma 6.4.17 is

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_0^1 \operatorname{Tr}(V(\operatorname{Id} + H_t(\alpha)^2)^{-s}) \, \mathrm{d}t \right) = \frac{1}{2} \beta_4(V) + \frac{\alpha}{4(2\pi i)} c_1(4, V).$$

In fact the n=2,4 calculations suggest the following observation.

**Proposition 6.4.19.** Suppose that n is even and V satisfies Assumption 6.2.1. Then

$$\lim_{\alpha \to 0^{+}} \operatorname{Res}_{s=\frac{1}{2}} \left( \int_{0}^{1} \operatorname{Tr}(V(\operatorname{Id} + H_{t}(\alpha)^{2})^{-s}) dt \right)$$

$$= \operatorname{Res}_{s=\frac{1}{2}} \left( \sum_{\ell=0}^{\frac{n}{2}} \sum_{(k,p)\in J_{\ell}(n)} Y_{\ell}(k,s) \int_{0}^{1} \operatorname{Tr}\left(VX_{p_{1}}^{(k_{1})} \cdots X_{p_{\ell}}^{(k_{\ell})}(\operatorname{Id} + H_{0})^{-s-\ell-|k|}\right) dt \right).$$

That is, the limit as  $\alpha \to 0^+$  commutes with the residue at  $s = \frac{1}{2}$ .

*Proof.* By the same arguments as the proof of Lemma 6.4.14 is suffices to prove for any  $q \in \mathbb{N}$ ,  $g \in C_c^{\infty}(\mathbb{R}^n)$  and multi-index  $\beta = (2b_1, \ldots, 2b_n)$  that

$$\lim_{\alpha \to 0} \operatorname{Res}_{s = \frac{1}{2}} \left( \operatorname{Tr} \left( V g \frac{\partial^{\beta}}{\partial x^{\beta}} (\operatorname{Id} + H_0(\alpha)^2)^{-s-q} \right) \right) = \operatorname{Res}_{s = \frac{1}{2}} \left( \operatorname{Tr} \left( V g \frac{\partial^{\beta}}{\partial x^{\beta}} (\operatorname{Id} + H_0^2)^{-s-q} \right) \right).$$
(6.12)

So we compute that

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \operatorname{Tr} \left( V g \frac{\partial^{\beta}}{\partial x^{\beta}} (\operatorname{Id} + H_{0}(\alpha)^{2})^{-s-q} \right) \right)$$

$$= i^{|\beta|} (2\pi)^{-n} \operatorname{Res}_{s=\frac{1}{2}} \left( \left( \int_{\mathbb{R}^{n}} V(x) g(x) \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^{n}} \xi_{1}^{\beta_{1}} \cdots \xi_{n}^{\beta_{n}} (1 + (|\xi|^{2} + \alpha)^{2})^{-s-q} \, \mathrm{d}\xi \right) \right)$$

$$= i^{|\beta|} \frac{2\Gamma \left( \frac{1+\beta_{1}}{2} \right) \cdots \Gamma \left( \frac{1+\beta_{n}}{2} \right)}{\Gamma \left( \frac{n+|\beta|}{2} \right) (2\pi)^{n}} \operatorname{Res}_{s=\frac{1}{2}} \left( \left( \int_{\mathbb{R}^{n}} V(x) g(x) \, \mathrm{d}x \right) \times \left( \int_{0}^{\infty} r^{n+|\beta|-1} (1 + (r^{2} + \alpha)^{2})^{-s-q} \, \mathrm{d}r \right) \right),$$

where we have changed to polar coordinates and used Lemma 2.5.20 to evaluate the integral over  $\mathbb{S}^{n-1}$ . Applying Lemma 6.4.12 we have

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\mathbb{R}^{n}} r^{n+|\beta|-1} (1 + (r^{2} + \alpha)^{2})^{-s-q} \, dr \right) \\
= \operatorname{Res}_{s=\frac{1}{2}} \left( \sum_{\ell=0}^{\frac{n+|\beta|}{2}-1} (-1)^{\frac{n+|\beta|}{2}-\ell-1} \alpha^{\frac{n+|\beta|}{2}-\ell-1} \frac{\Gamma\left(\frac{\ell+1}{2}\right) \Gamma\left(s+q-\frac{\ell+1}{2}\right)}{4\Gamma(s+q)} \right) \\
= \sum_{\ell=0}^{\frac{n+|\beta|}{2}-1} (-1)^{\frac{n+|\beta|}{2}-\ell-1} \alpha^{\frac{n+|\beta|}{2}-\ell-1} \operatorname{Res}_{s=\frac{1}{2}} \left( \frac{\Gamma\left(\frac{\ell+1}{2}\right) \Gamma\left(s+q-\frac{\ell+1}{2}\right)}{4\Gamma(s+q)} \right).$$

Taking the limit as  $\alpha \to 0^+$  we have

$$\lim_{\alpha \to 0^+} \operatorname{Res}_{s = \frac{1}{2}} \left( \int_{\mathbb{R}^n} r^{n+|\beta|-1} (1 + (r^2 + \alpha)^2)^{-s-q} \, \mathrm{d}r \right) = \operatorname{Res}_{s = \frac{1}{2}} \left( \frac{\Gamma\left(\frac{n+|\beta|}{4}\right) \Gamma\left(s + q - \frac{n+|\beta|}{4}\right)}{4\Gamma(s+q)} \right),$$

from which we see that

$$\lim_{\alpha \to 0^{+}} \operatorname{Res}_{s = \frac{1}{2}} \left( \operatorname{Tr} \left( V g \frac{\partial^{\beta}}{\partial x^{\beta}} (\operatorname{Id} + H_{0}(\alpha)^{2})^{-s-q} \right) \right)$$

$$= i^{|\beta|} \frac{2\Gamma \left( \frac{1+\beta_{1}}{2} \right) \cdots \Gamma \left( \frac{1+\beta_{n}}{2} \right)}{\Gamma \left( \frac{n+|\beta|}{2} \right) (2\pi)^{n}} \left( \int_{\mathbb{R}^{n}} V(x) g(x) \, \mathrm{d}x \right) \operatorname{Res}_{s = \frac{1}{2}} \left( \frac{\Gamma \left( \frac{n+|\beta|}{2} \right) \Gamma \left( s + q - \frac{n+|\beta|}{2} \right)}{4\Gamma (s+q)} \right).$$

For the right hand side of Equation (6.12) we can compute directly using Lemma 6.4.11 that

$$\begin{split} & \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( \operatorname{Tr} \left( V g \frac{\partial^{\beta}}{\partial x^{\beta}} (\operatorname{Id} + H_{0}^{2})^{-s-q} \right) \right) \\ &= i^{|\beta|} (2\pi)^{-n} \operatorname{Res}_{s=\frac{1}{2}} \left( \left( \int_{\mathbb{R}^{n}} V(x) g(x) \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^{n}} \xi_{1}^{\beta_{1}} \cdots \xi_{n}^{\beta_{n}} (1 + |\xi|^{4})^{-s-q} \, \mathrm{d}\xi \right) \right) \\ &= i^{|\beta|} \frac{2\Gamma \left( \frac{1+\beta_{1}}{2} \right) \cdots \Gamma \left( \frac{1+\beta_{n}}{2} \right)}{\Gamma \left( \frac{n+|\beta|}{2} \right) (2\pi)^{n}} \operatorname{Res}_{s=\frac{1}{2}} \left( \left( \int_{\mathbb{R}^{n}} V(x) g(x) \, \mathrm{d}x \right) \left( \int_{0}^{\infty} r^{n+|\beta|-1} (1 + r^{4})^{-s-q} \, \mathrm{d}r \right) \right) \\ &= i^{|\beta|} \frac{2\Gamma \left( \frac{1+\beta_{1}}{2} \right) \cdots \Gamma \left( \frac{1+\beta_{n}}{2} \right)}{\Gamma \left( \frac{n+|\beta|}{2} \right) (2\pi)^{n}} \operatorname{Res}_{s=\frac{1}{2}} \left( \left( \int_{\mathbb{R}^{n}} V(x) g(x) \, \mathrm{d}x \right) \left( \frac{\Gamma \left( \frac{n+|\beta|}{4} \right) \Gamma \left( s + q - \frac{n+|\beta|}{4} \right)}{4\Gamma (s+q)} \right) \right), \end{split}$$

which proves the claim.

In practice, it is much easier to compute the residue in Equation (6.11) after taking the limit as  $\alpha \to 0^+$ . The number of terms required increases significantly as the dimension increases, with 8 terms required for n=6 (after taking the limit as  $\alpha \to 0$  and using Proposition 6.4.19). Only some of these terms are non-zero, the result is the following.

**Lemma 6.4.20.** Let n = 6 and suppose V satisfies Assumption 6.2.1. Then

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{0}^{1} \operatorname{Tr}(V(\operatorname{Id} + H_{t}(0)^{2})^{-s}) \, dt \right)$$

$$= \frac{\operatorname{Vol}(\mathbb{S}^{5})}{12(2\pi)^{6}} \int_{\mathbb{R}^{6}} V(x)^{3} \, dx + \frac{\operatorname{Vol}(\mathbb{S}^{5})}{24(2\pi)^{6}} \int_{\mathbb{R}^{6}} |[\nabla V](x)|^{2} \, dx - \frac{\operatorname{Vol}(\mathbb{S}^{5})}{8(2\pi)^{6}} \int_{\mathbb{R}^{6}} V(x) \, dx.$$

The proof of Lemma 6.4.20 is similar to, but much more involved than, Lemma 6.4.17 and as such is provided in Appendix A.1.

Remark 6.4.21. In the notation of Theorem 2.5.34 and Definition 2.5.26, the statement

of Lemma 6.4.20 is

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_0^1 \operatorname{Tr}(V(\operatorname{Id} + H_t(0)^2)^{-s}) \, \mathrm{d}t \right) = \frac{1}{2} \beta_6(V) + \frac{c_1(6, V)}{4}.$$

We now return to more general statements. As a result of Lemma 6.4.13 we obtain the following formula for the spectral flow along the path  $H_t(\alpha)$ .

**Theorem 6.4.22.** Let n be odd and V satisfy Assumption 6.2.1. The spectral flow from  $H_0(\alpha)$  to  $H(\alpha)$  is given by

$$\operatorname{sf}(H(\alpha), H_0(\alpha)) = \frac{1}{4\pi i} \operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\alpha}^{\infty} C_s \eta_s(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)) \, \mathrm{d}\lambda \right) - \frac{1}{2} (N - 2N_0)$$

$$- \frac{1}{2} (\xi(0+) - \xi(0-) - N_0).$$
(6.13)

The number of bound states N of H can be computed via the formula

$$-N = \frac{1}{2\pi i} \operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\alpha}^{\infty} C_s \eta_s(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)) \, d\lambda \right) - (\xi(0+) - \xi(0-) - N_0). \tag{6.14}$$

*Proof.* A combination of Lemmas 6.4.2, 6.4.1 and 6.4.13 applied to Equation (6.3) gives

$$sf(H_t(\alpha)) = \frac{1}{4\pi i} \underset{s=\frac{1}{2}}{\text{Res}} \left( \int_{\alpha}^{\infty} C_s \eta_s(\lambda) \text{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)) \, d\lambda \right) - \frac{1}{2} (N - 2N_0)$$
$$- \frac{1}{2} (\xi(0+) - \xi(0-) - N_0),$$

where we have made the identifications  $\xi_{\alpha}(\lambda + \alpha) = \xi(\lambda)$ ,  $N(\alpha) = N$  and  $M_{\alpha}(\alpha) = N_0$ , the multiplicity of the zero eigenvalue for H. For the final statement note that the spectral flow is given by  $N_0 - N$ , independent of  $\alpha$ , and solve for N to obtain the result.

**Theorem 6.4.23.** Let n be even and V satisfy Assumption 6.2.1. The spectral flow from  $H_0(\alpha)$  to  $H(\alpha)$  is given by

$$sf(H(\alpha), H_0(\alpha))$$

$$= \frac{1}{4\pi i} \underset{s=\frac{1}{2}}{\text{Res}} \left( \int_{\alpha}^{\infty} C_s \eta_s(\lambda) \text{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)) d\lambda \right) - \frac{1}{2} (N - 2N_0) - \frac{1}{2} (\xi(0+) - \xi(0-) - N_0)$$

$$+ \underset{s=\frac{1}{2}}{\text{Res}} \left( \sum_{\ell=0}^{\frac{n}{2}} \sum_{(k,p)\in J_{\ell}(n)} \int_{0}^{1} \text{Tr}\left(V X_{p_1}^{(k_1)} \cdots X_{p_{\ell}}^{(k_{\ell})} (\text{Id} + H_0(\alpha)^2)^{-s-\ell-|k|}\right) dt \right).$$

$$(6.15)$$

The number of bound states N of H can be computed via the formula

$$-N = \frac{1}{2\pi i} \underset{s=\frac{1}{2}}{\text{Res}} \left( \int_{\alpha}^{\infty} \eta_{s}(\lambda) \text{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) \, d\lambda \right) - (\xi(0+) - \xi(0-) - N_{0})$$

$$+ 2 \underset{s=\frac{1}{2}}{\text{Res}} \left( \sum_{\ell=0}^{\frac{n}{2}} \sum_{(k,p)\in J_{\ell}(n)} \int_{0}^{1} \text{Tr}\left(V X_{p_{1}}^{(k_{1})} \cdots X_{p_{\ell}}^{(k_{\ell})} (\text{Id} + H_{0}(\alpha)^{2})^{-s-\ell-|k|}\right) \, dt \right).$$

$$(6.16)$$

*Proof.* A combination of Lemmas 6.4.2, 6.4.1 and 6.4.13 applied to Equation (6.3) gives

$$sf(H_t(\alpha)) = \frac{1}{4\pi i} \underset{s=\frac{1}{2}}{\text{Res}} \left( \int_{\alpha}^{\infty} C_s \eta_s(\lambda) \text{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)) \, d\lambda \right) - \frac{1}{2} (N - 2N_0) - \frac{1}{2} (\xi(0+) - \xi(0-)) + \underset{s=\frac{1}{2}}{\text{Res}} \left( \sum_{\ell=0}^{\frac{n}{2}} \sum_{(k,p)\in J_{\ell}(n)} \int_{0}^{1} \text{Tr}\left(V X_{p_1}^{(k_1)} \cdots X_{p_{\ell}}^{(k_{\ell})} (\text{Id} + H_0(\alpha)^2)^{-s-\ell-|k|}\right) \, dt \right).$$

We know that  $sf(H_t(\alpha)) = N_0 - N$ , independent of  $\alpha$ . Equating the left hand side to  $N_0 - N$  and solving for N gives the second statement.

We remark that there are far more terms than necessary, although carefully indexing the terms with non-zero residue is a tedious task which we do not complete here.

#### 6.5 Levinson's theorem

In this section we show how Theorems 6.4.22 and 6.4.23 imply that Levinson's theorem is a result of the spectral flow along the path  $H_t(\alpha)$ .

**Lemma 6.5.1** (Levinson's Theorem in dimension n=1). Let n=1 and suppose that V satisfies Assumption 2.2.14 for some  $\rho > \frac{5}{2}$ . Then

$$-N = \frac{1}{2\pi i} \int_{\mathbb{R}^+} \operatorname{Tr}(S(\lambda)^* S'(\lambda)) \, d\lambda + \frac{1}{2} (1 - M_R(0)),$$

where  $M_R(0) = 1$  if there exists a resonance and  $M_R(0) = 0$  otherwise.

*Proof.* By Theorem 6.4.22 we have

$$-N = \frac{1}{2\pi i} \operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\alpha}^{\infty} C_s \eta_s(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)) d\lambda \right) - (\xi(0+) - \xi(0-) - N_0).$$

By Proposition 6.4.5 we have

$$\frac{1}{2\pi i} \operatorname{Res}_{s=\frac{1}{3}} \left( \int_{\alpha}^{\infty} C_s \eta_s(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)) \, \mathrm{d}\lambda \right) = \frac{1}{2\pi i} \int_{0}^{\infty} \operatorname{Tr}(S(\lambda)^* S'(\lambda)) \, \mathrm{d}\lambda.$$

Corollary 5.3.18 then gives  $\xi(0+) - \xi(0-) - N_0 = -\frac{1}{2}(1 - M_R(0))$ , which completes the proof.

**Lemma 6.5.2** (Levinson's Theorem in dimension n=2). Let n=2 and suppose that V satisfies Assumption 2.2.14 for some  $\rho > 11$ . Then the number of bound states for  $H = H_0 + V$  is given by

$$-N = \frac{1}{2\pi i} \int_{\mathbb{R}^+} \operatorname{Tr}(S(\lambda)^* S'(\lambda)) \, d\lambda + M_p(0) + \frac{1}{4\pi} \int_{\mathbb{R}^2} V(x) \, dx,$$

where  $M_p(0) = -(\xi(0+) - \xi(0-) - N_0)$ ,  $\xi$  is the spectral shift function for the pair  $(H, H_0)$  and  $N_0$  is the number of zero eigenvalues for H.

*Proof.* By Theorem 6.4.23 we have

$$-N = \frac{1}{2\pi i} \underset{s=\frac{1}{2}}{\text{Res}} \left( \int_{\alpha}^{\infty} \eta_{s}(\lambda) \text{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) d\lambda \right) - (\xi(0+) - \xi(0-) - N_{0})$$

$$+ 2 \underset{s=\frac{1}{2}}{\text{Res}} \left( \sum_{\ell=0}^{1} \sum_{(k,p)\in J_{\ell}(n)} \int_{0}^{1} \text{Tr}\left(V X_{p_{1}}^{(k_{1})} \cdots X_{p_{\ell}}^{(k_{\ell})} (\text{Id} + H_{0}(\alpha)^{2})^{-s-\ell-|k|}\right) dt \right).$$

Lemma 6.4.15 shows that

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_0^1 \operatorname{Tr}(V(\operatorname{Id} + H_t(\alpha)^2)^{-s}) \, dt \right) = \frac{1}{8\pi} \int_{\mathbb{R}^2} V(x) \, dx.$$

By Proposition 6.4.5 we have

$$\frac{1}{2\pi i} \operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\alpha}^{\infty} C_s \eta_s(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)) \, \mathrm{d}\lambda \right) = \frac{1}{2\pi i} \int_{0}^{\infty} \operatorname{Tr}(S(\lambda)^* S'(\lambda)) \, \mathrm{d}\lambda.$$

Corollary 5.3.31 gives  $\xi(0+) - \xi(0-) - N_0 = 0$ , which completes the proof.

We note that in the case of no p-resonances, we have  $M_p(0) = (\xi(0+)-\xi(0-)-N_0) = 0$  by Corollary 5.3.31. Lemma 6.5.2 demonstrates that the value  $\xi(0+)$  of the spectral shift function at zero is an integer determined by the number of p resonances.

**Lemma 6.5.3** (Levinson's Theorem in higher odd dimensions). Let  $n \geq 3$  be odd and suppose that V satisfies Assumption 6.2.1 and define for  $1 \leq j \leq \lfloor \frac{n-1}{2} \rfloor$  constants  $c_j(n, V)$  as in Theorem 2.5.34. Let  $p_n(\lambda) = \sum_{j=1}^{\lfloor \frac{n-1}{2} \rfloor} c_j(n, V) \lambda^{\frac{n-2}{2}-j}$  be the high-energy polynomial of Definition 2.5.26. Then

$$-N = \frac{1}{2\pi i} \int_{\mathbb{R}^+} \left( \text{Tr}(S(\lambda)^* S'(\lambda)) - p_n(\lambda) \right) d\lambda + \frac{1}{2} M_R(0),$$

where  $M_R(0) = 1$  if there is a resonance in dimension n = 3 and  $M_R(0) = 0$  otherwise.

*Proof.* By Theorem 6.4.22 we have

$$-N = \frac{1}{2\pi i} \operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\alpha}^{\infty} C_s \eta_s(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)) d\lambda \right) - (\xi(0+) - \xi(0-) - N_0).$$

By Proposition 6.4.5 we have

$$\frac{1}{2\pi i} \operatorname{Res}_{s=\frac{1}{2}} \left( \int_{\alpha}^{\infty} C_s \eta_s(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)) \, \mathrm{d}\lambda \right) = \frac{1}{2\pi i} \int_{0}^{\infty} \left( \operatorname{Tr}(S(\lambda)^* S'(\lambda)) - p_n(\lambda) \right) \, \mathrm{d}\lambda.$$

Noting that  $-(\xi(0+)-\xi(0-)-N_0)=\frac{1}{2}M_R(0)$  by Corollaries 5.3.26 (for n=3) and 5.3.36 (for  $n \ge 5$ ) completes the proof.

The polynomial  $p_n$  is determined by Theorem 2.5.34. In dimension n=3 we have

$$p_3(\lambda) = -\frac{i\lambda^{-\frac{1}{2}}}{4\pi} \int_{\mathbb{R}^3} V(x) \, \mathrm{d}x.$$

In dimension n = 5 we have

$$p_5(\lambda) = -\frac{(2\pi i) \text{Vol}(\mathbb{S}^4) \lambda^{\frac{1}{2}}}{(2\pi)^5} \int_{\mathbb{R}^5} V(x) \, \mathrm{d}x + \frac{(2\pi i) \text{Vol}(\mathbb{S}^4) \lambda^{-\frac{1}{2}}}{2(2\pi)^5} \int_{\mathbb{R}^5} V(x)^2 \, \mathrm{d}x.$$

In dimension n = 7 we have

$$p_{7}(\lambda) = -\frac{(2\pi i)\operatorname{Vol}(\mathbb{S}^{6})\lambda^{\frac{3}{2}}}{(2\pi)^{7}} \int_{\mathbb{R}^{7}} V(x) \, dx + \frac{(2\pi i)\operatorname{Vol}(\mathbb{S}^{6})\lambda^{\frac{1}{2}}}{2(2\pi)^{7}} \int_{\mathbb{R}^{7}} V(x)^{2} \, dx - \frac{15(2\pi i)\operatorname{Vol}(\mathbb{S}^{6})\lambda^{-\frac{1}{2}}}{48(2\pi)^{7}} \int_{\mathbb{R}^{7}} \left(V(x)^{3} + \frac{1}{2}|[\nabla V](x)|^{2}\right) \, dx.$$

We do not compute  $p_n$  for higher n, since the computations become tedious quickly.

**Lemma 6.5.4** (Levinson's Theorem in dimension n = 4). Let n = 4 and suppose that V satisfies Assumption 6.2.1 and define

$$c_1(4, V) = -\frac{(2\pi i) \text{Vol}(\mathbb{S}^3)}{2(2\pi)^4} \int_{\mathbb{R}^4} V(x) \, dx.$$

as in Theorem 2.5.34. Then

$$-N = \frac{1}{2\pi i} \int_{\mathbb{R}^+} \left( \text{Tr}(S(\lambda)^* S'(\lambda)) - c_1(4, V) \right) d\lambda + M_R(0) + \frac{\text{Vol}(\mathbb{S}^3)}{4(2\pi)^4} \int_{\mathbb{R}^4} V(x)^2 dx,$$

where  $M_R(0) = -(\xi(0+) - \xi(0-) - N_0)$ .

*Proof.* By Theorem 6.4.23 we have

$$-N = \frac{1}{2\pi i} \underset{s=\frac{1}{2}}{\text{Res}} \left( \int_{\alpha}^{\infty} \eta_{s}(\lambda) \text{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) d\lambda \right) - (\xi(0+) - \xi(0-) - N_{0})$$

$$+ 2 \underset{s=\frac{1}{2}}{\text{Res}} \left( \sum_{\ell=0}^{2} \sum_{(k,p)\in J_{\ell}(n)} \int_{0}^{1} \text{Tr}\left(V X_{p_{1}}^{(k_{1})} \cdots X_{p_{\ell}}^{(k_{\ell})} (\text{Id} + H_{0}(\alpha)^{2})^{-s-\ell-|k|}\right) dt \right).$$

Corollary 6.4.8 shows that

$$\operatorname{Res}_{s=\frac{1}{2}} \frac{1}{2\pi i} \left( \int_{\alpha}^{\infty} C_s \eta_s(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^* S_{\alpha}'(\lambda)) \, \mathrm{d}\lambda \right) = \frac{1}{2\pi i} \int_{0}^{\infty} \left( \operatorname{Tr}(S(\lambda)^* S'(\lambda)) - c_1(4, V) \right) \, \mathrm{d}\lambda - \frac{c_1(4, V)}{2(2\pi i)} \alpha.$$

Lemma 6.4.17 gives

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_0^1 \operatorname{Tr}(V(\operatorname{Id} + H_t(\alpha)^2)^{-s}) \, \mathrm{d}t \right) = -\frac{\operatorname{Vol}(\mathbb{S}^3)}{8(2\pi)^4} \int_{\mathbb{R}^4} V(x)^2 \, \mathrm{d}x + \frac{c_1(4, V)}{4(2\pi i)} \alpha,$$

which completes the proof.

We note that by Corollary 5.3.36, if there are no resonances then  $M_R(0) = 0$ . Thus the value  $\xi(0+)$  of the spectral shift function at zero is an integer determined by the number of resonances.

**Lemma 6.5.5** (Levinson's theorem in dimension n = 6). Suppose that n = 6 and V satisfies Assumption 6.2.1. Then the number of bound states of  $H = H_0 + V$  is given by

$$-N = \frac{1}{2\pi i} \int_0^\infty (\text{Tr}(S(\lambda)^* S'(\lambda)) - c_1(6, V)\lambda - c_2(6, V)) \, d\lambda - \beta_6(V),$$

where the constants  $c_1(n, V), c_2(n, V), \beta_6(V)$  are defined by Theorem 2.5.34.

*Proof.* By Theorem 6.4.23 we have

$$-N = \frac{1}{2\pi i} \underset{s=\frac{1}{2}}{\text{Res}} \left( \int_{\alpha}^{\infty} \eta_{s}(\lambda) \text{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) d\lambda \right) - (\xi(0+) - \xi(0-) - N_{0})$$

$$+ 2 \underset{s=\frac{1}{2}}{\text{Res}} \left( \sum_{\ell=0}^{3} \sum_{(k,p) \in J_{\ell}(n)} \int_{0}^{1} \text{Tr} \left( V X_{p_{1}}^{(k_{1})} \cdots X_{p_{\ell}}^{(k_{\ell})} (\text{Id} + H_{0}(\alpha)^{2})^{-s-\ell-|k|} \right) dt \right).$$

Since the left hand side is independent of  $\alpha$ , we use Corollary 6.4.7 and Proposition 6.4.19

to take the limit as  $\alpha \to 0^+$  and find

$$-N = \frac{1}{2\pi i} \operatorname{Res}_{s=\frac{1}{2}} \left( \int_{0}^{\infty} \eta_{s}(\lambda) \operatorname{Tr}(S_{\alpha}(\lambda)^{*} S_{\alpha}'(\lambda)) d\lambda \right) - (\xi(0+) - \xi(0-) - N_{0})$$

$$+ 2 \operatorname{Res}_{s=\frac{1}{2}} \left( \sum_{\ell=0}^{3} \sum_{(k,p)\in J_{\ell}(n)} \int_{0}^{1} \operatorname{Tr}\left(V X_{p_{1}}^{(k_{1})} \cdots X_{p_{\ell}}^{(k_{\ell})} (\operatorname{Id} + H_{0}^{2})^{-s-\ell-|k|}\right) dt \right).$$

Proposition 6.4.5 then shows

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \frac{1}{2\pi i} \int_0^\infty C_s \eta_s(\lambda) \operatorname{Tr}(S(\lambda)^* S'(\lambda)) \, \mathrm{d}\lambda \right)$$

$$= \frac{1}{2\pi i} \int_0^\infty \left( \operatorname{Tr}(S(\lambda)^* S'(\lambda)) - c_1(6, V) \lambda - c_2(6, V) \right) \, \mathrm{d}\lambda + \frac{c_1(6, V)}{4(2\pi i)}.$$

We have the explicit expression  $\frac{c_1(6,V)}{4(2\pi i)} = -\frac{\text{Vol}(\mathbb{S}^5)}{8(2\pi)^6} \int_{\mathbb{R}^6} V(x) \, dx$ . Recall the constant  $\beta_6(V)$  from Theorem 2.5.34. From Lemma 6.4.20 we find

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \sum_{\ell=0}^{3} \sum_{(k,p)\in J_{\ell}(6)} Y_{\ell}(k) \int_{0}^{1} \operatorname{Tr} \left( V X_{p_{1}}^{(k_{1})} \cdots X_{p_{\ell}}^{(k_{\ell})} (\operatorname{Id} + H_{0}^{2})^{-s-\ell-|k|} \right) dt \right)$$

$$= -\frac{\operatorname{Vol}(\mathbb{S}^{5})}{8(2\pi)^{6}} \int_{\mathbb{R}^{4}} V(x) dx - \frac{\operatorname{Vol}(\mathbb{S}^{5})}{12(2\pi)^{6}} \int_{\mathbb{R}^{6}} \left( V(x)^{3} + \frac{1}{2} |[\nabla V](x)|^{2} \right) dx$$

$$= \frac{c_{1}(6, V)}{8(2\pi i)} + \frac{1}{2} \beta_{6}(V).$$

Note that by Corollary 5.3.36 we have  $(\xi(0+) - \xi(0-) - N_0) = 0$ . An application of Theorem 6.4.23 then gives

$$-N = \frac{1}{2\pi i} \underset{s=\frac{1}{2}}{\text{Res}} \left( \int_{0}^{\infty} \eta_{s}(\lambda) \text{Tr}(S(\lambda)^{*}S'(\lambda)) \, d\lambda \right) - (\xi(0+) - \xi(0-))$$

$$+ 2 \underset{s=\frac{1}{2}}{\text{Res}} \left( \sum_{\ell=0}^{\frac{n}{2}} \sum_{(k,p)\in J_{\ell}(6)} \int_{0}^{1} \text{Tr}\left(VX_{p_{1}}^{(k_{1})} \cdots X_{p_{\ell}}^{(k_{\ell})} (\text{Id} + H_{0}^{2})^{-s-\ell-|k|}\right) \, dt \right)$$

$$= \frac{1}{2\pi i} \int_{0}^{\infty} \left( \text{Tr}(S(\lambda)^{*}S'(\lambda)) - c_{3}(6, V)\lambda - c_{2}(6, V) \right) \, d\lambda + \frac{c_{1}(6, V)}{4(2\pi i)}$$

$$- 2\left( \frac{c_{1}(6, V)}{8(2\pi i)} + \frac{1}{2}\beta_{6}(V) \right),$$

which completes the proof.

Unfortunately due to the large number of terms involved, a general argument showing that in higher even dimensions the "integral of one form" contributions and the  $\eta$  contributions combine to give the correct statement of Levinson's theorem is not given.

## Appendix A

## Computations of coefficients

In this appendix we provide some detailed computations of the coefficients appearing in Levinson's theorem. In particular in Section A.1 we provide the proof of Lemma 6.4.20 for the terms appearing in the integral of a one form contribution to the spectral flow in dimension n = 6.

## A.1 Proof of Lemma 6.4.20

In this section we provide a proof of Lemma 6.4.20 for the contributions to the integral of a one form term to spectral flow in dimension n = 6.

**Lemma A.1.1.** Let n = 6 and suppose V satisfies Assumption 6.2.1. Then

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{0}^{1} \operatorname{Tr}(V(\operatorname{Id} + H_{t}(0)^{2})^{-s}) \, dt \right)$$

$$= \frac{\operatorname{Vol}(\mathbb{S}^{5})}{12(2\pi)^{6}} \int_{\mathbb{R}^{6}} V(x)^{3} \, dx + \frac{\operatorname{Vol}(\mathbb{S}^{5})}{24(2\pi)^{6}} \int_{\mathbb{R}^{6}} |[\nabla V](x)|^{2} \, dx - \frac{\operatorname{Vol}(\mathbb{S}^{5})}{8(2\pi)^{6}} \int_{\mathbb{R}^{6}} V(x) \, dx.$$

*Proof.* We have  $J_0(6) = \{(0,0)\}, J_1(6) = \{(0,0),(0,1),(1,1),(0,2),(1,2),(2,2)\}$  and  $J_2(6) = \{((0,0),(2,2))\}.$  For  $\ell \geq 3$  we have  $J_{\ell}(6) = \emptyset$ . So there are eight terms to consider.

We write  $X = X_0 + X_1 + X_2$ , where

$$X_0 = t^2 V^2 + t(\Delta V), \quad X_1 = -2t \sum_{j=1}^6 \frac{\partial V}{\partial x_j} \frac{\partial}{\partial x_j}$$
 and  $X_2 = 2tV H_0.$ 

We begin with the case  $\ell=0, k=0, p=0$ . Observe that  $\int_0^1 dt = 1$ ,  $Y_0(0, \frac{1}{2}) = 1$ ,

Res  $\left(\Gamma\left(s-\frac{3}{2}\right)\right)=-1$  and  $\Gamma\left(\frac{3}{2}\right)=\frac{1}{2}\Gamma\left(\frac{1}{2}\right)$ . So we find So we compute

$$\operatorname{Res}_{s=\frac{1}{2}} \left( Y_{0}(0,s) \int_{0}^{1} \operatorname{Tr}(V(\operatorname{Id} + H_{0}^{2})^{-s}) \, dt \right) \\
= \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{0}(0,s)(2\pi)^{-6} \left( \int_{\mathbb{R}^{6}} V(x) \, dx \right) \left( \int_{\mathbb{R}^{6}} (1 + |\xi|^{4})^{-s} \, d\xi \right) \right) \\
= \frac{\operatorname{Vol}(\mathbb{S}^{5})}{(2\pi)^{6}} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{0}(0,s) \left( \int_{\mathbb{R}^{6}} V(x) \, dx \right) \left( \int_{0}^{\infty} r^{6-1} (1 + r^{4})^{-s} \, dr \right) \right) \\
= \frac{\operatorname{Vol}(\mathbb{S}^{5})}{(2\pi)^{6}} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{0}(0,s) \left( \int_{\mathbb{R}^{6}} V(x) \, dx \right) \left( \frac{\Gamma\left(\frac{6}{4}\right) \Gamma\left(s - \frac{6}{4}\right)}{4\Gamma(s)} \right) \right) \\
= -\frac{\operatorname{Vol}(\mathbb{S}^{5})}{8(2\pi)^{6}} \int_{\mathbb{R}^{6}} V(x) \, dx.$$

The next case is  $\ell=1, k=0, p=0$ . Observe that  $\int_0^1 t^2 dt = \frac{1}{3}$ ,  $\int_0^1 t dt = \frac{1}{2}$ ,  $Y_1(0, \frac{1}{2}) = -\frac{1}{2}$  and  $\operatorname{Res}_{s=\frac{1}{2}}\left(\Gamma\left(s-\frac{1}{2}\right)\right)=1$ . For clarity, we introduce the notation  $(\Delta V)$  for the action of  $H_0$  on V. We compute

$$\begin{aligned} & \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \int_{0}^{1} \operatorname{Tr}(VX_{0}(\operatorname{Id} + H_{0}^{2})^{-s-1}) \, \mathrm{d}t \right) \\ & = \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \int_{0}^{1} \left( t^{2} \operatorname{Tr}(V^{3}(\operatorname{Id} + H_{0}^{2})^{-s-1}) + t \operatorname{Tr}(V(\Delta V)(\operatorname{Id} + H_{0}^{2})^{-s-1}) \right) \, \mathrm{d}t \right) \\ & = \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \left( \frac{1}{3} \operatorname{Tr}(V^{3}(\operatorname{Id} + H_{0}^{2})^{-s-1}) + \frac{1}{2} \operatorname{Tr}(V(\Delta V)(\operatorname{Id} + H_{0}^{2})^{-s-1}) \right) \right) \\ & = (2\pi)^{-6} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \left( \int_{\mathbb{R}^{6}} \left( \frac{1}{3} V(x)^{3} + \frac{1}{2} V(x)[\Delta V](x) \right) \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^{6}} \left( 1 + |\xi|^{4})^{-s-1} \, \mathrm{d}\xi \right) \right) \\ & = \frac{\operatorname{Vol}(\mathbb{S}^{5})}{(2\pi)^{6}} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \left( \int_{\mathbb{R}^{6}} \left( \frac{1}{3} V(x)^{3} + \frac{1}{2} V(x)[\Delta V](x) \right) \, \mathrm{d}x \right) \left( \frac{\Gamma\left(\frac{6}{4}\right) \Gamma\left(s+1-\frac{6}{4}\right)}{4\Gamma(s+1)} \right) \right) \\ & = -\frac{\operatorname{Vol}(\mathbb{S}^{5})}{8(2\pi)^{6}} \int_{\mathbb{R}^{6}} \left( \frac{1}{3} V(x)^{3} + \frac{1}{2} |[\nabla V](x)|^{2} \right) \, \mathrm{d}x. \end{aligned}$$

We move on to the case  $\ell = 1, k = 0, p = 1$ . Observe that  $\int_0^1 (-2t) dt = -1$ . Then we

find

$$\operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \int_{0}^{1} \operatorname{Tr}(V X_{1}(\operatorname{Id} + H_{0}^{2})^{-s-1}) dt \right) \\
= \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \int_{0}^{1} (-2t) \sum_{k=1}^{6} \operatorname{Tr}\left( V \frac{\partial V}{\partial x_{k}} \frac{\partial}{\partial x_{k}} (\operatorname{Id} + H_{0}^{2})^{-s-1} \right) dt \right) \\
= i(2\pi)^{-6} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \sum_{k=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial V}{\partial x_{k}} dx \right) \left( \int_{\mathbb{R}^{6}} \xi_{k} (1 + |\xi|^{4})^{-s} d\xi \right) \right) \\
= 0,$$

since the  $\xi$  integral contains the integral of an odd function over a symmetric region.

The next term is  $\ell=1, k=1, p=1$ . Observe that  $Y_1(1,\frac{1}{2})=\frac{3}{8}$ ,  $\int_0^1 (8t) \, \mathrm{d}t=4$  and  $\operatorname{Res}_{s=\frac{1}{2}} \left(\Gamma\left(s-\frac{1}{2}\right)\right)=1$ . We write  $X_1^{(1)}=[H_0^2,X_1]=H_0[H_0,X_1]+[H_0,X_1]H_0$ . Note that  $X_1^{(1)}$  is an order 4 operator and we can write  $X_1^{(1)}=\sum_{j=0}^4 G_j$ , with each  $G_j$  an order j operator. The operator  $VG_j(\operatorname{Id}+H_0^2)^{-s-2}$  has order j-4s-8 and so gives a contribution which is holomorphic at  $s=\frac{1}{2}$  when j<4. Thus only the operator  $G_4$  can contribute a non-zero residue. We compute the commutator

$$[H_0, X_1]$$

$$= -2t \left( -\sum_{k=1}^{6} \frac{\partial^2}{\partial x_k^2} \left( \sum_{j=1}^{6} \frac{\partial V}{\partial x_j} \frac{\partial}{\partial x_j} \right) - \sum_{j=1}^{6} \frac{\partial V}{\partial x_j} \frac{\partial}{\partial x_j} H_0 \right)$$

$$= -2t \left( -\sum_{j,k=1}^{6} \left( \frac{\partial^3 V}{\partial x_k^2 \partial x_j} \frac{\partial}{\partial x_j} + 2 \frac{\partial^2 V}{\partial x_k \partial x_j} \frac{\partial^2}{\partial x_k \partial x_j} + \frac{\partial V}{\partial x_j} \frac{\partial^3}{\partial x_k^2 \partial x_j} \right) - \sum_{j=1}^{6} \frac{\partial V}{\partial x_j} \frac{\partial}{\partial x_j} H_0 \right)$$

$$= 2t \sum_{j,k=1}^{6} \left( \frac{\partial^3 V}{\partial x_k^2 \partial x_j} \frac{\partial}{\partial x_j} + 2 \frac{\partial^2 V}{\partial x_k \partial x_j} \frac{\partial^2}{\partial x_k \partial x_j} \right).$$

From this we immediately obtain the highest order component  $G_4$  of  $X_1^{(1)}$  as

$$G_4 = 8t \sum_{j,k=1}^{6} \frac{\partial^2 V}{\partial x_k \partial x_j} \frac{\partial^2}{\partial x_k \partial x_j} H_0.$$

So we compute

$$\operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(1,s) \int_{0}^{1} \operatorname{Tr}(V X_{1}^{(1)} (\operatorname{Id} + H_{0}^{2})^{-s-2}) dt \right) \\
= \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(1,s) \int_{0}^{1} \operatorname{Tr}(V G_{4} (\operatorname{Id} + H_{0}^{2})^{-s-2}) dt \right) \\
= 4 \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(1,s) \sum_{j,k=1}^{6} \operatorname{Tr} \left( \frac{\partial^{2} V}{\partial x_{k} \partial x_{j}} \frac{\partial^{2}}{\partial x_{k} \partial x_{j}} H_{0} (\operatorname{Id} + H_{0}^{2})^{-s-2} \right) \right) \\
= i^{2} \frac{4}{(2\pi)^{6}} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(1,s) \sum_{j,k=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2} V}{\partial x_{k} \partial x_{j}} dx \right) \left( \int_{\mathbb{R}^{6}} \xi_{k} \xi_{j} |\xi|^{2} (1 + |\xi|^{4})^{-s-2} d\xi \right) \right).$$

The terms in the sum with  $j \neq k$  vanish, since the  $\xi$  integral contains the integral of an odd function over a symmetric interval. So we obtain

$$\begin{split} & \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(1,s) \int_{0}^{1} \operatorname{Tr}(VX_{1}^{(1)}(\operatorname{Id} + H_{0}^{2})^{-s-2}) \, \mathrm{d}t \right) \\ & = i^{2} \frac{4}{(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(1,s) \sum_{k=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2}V}{\partial x_{k}^{2}} \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^{6}} \xi_{k}^{2} |\xi|^{2} (1 + |\xi|^{4})^{-s-2} \, \mathrm{d}\xi \right) \right) \\ & = -\frac{4}{6(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(1,s) \sum_{k=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2}V}{\partial x_{k}^{2}} \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^{6}} |\xi|^{4} (1 + |\xi|^{4})^{-s-2} \, \mathrm{d}\xi \right) \right) \\ & = -\frac{4\operatorname{Vol}(\mathbb{S}^{5})}{6(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(1,s) \sum_{k=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2}V}{\partial x_{k}^{2}} \, \mathrm{d}x \right) \left( \int_{0}^{\infty} r^{10-1} (1 + r^{4})^{-s-2} \, \mathrm{d}r \right) \right) \\ & = -\frac{4\operatorname{Vol}(\mathbb{S}^{5})}{6(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(1,s) \sum_{k=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2}V}{\partial x_{k}^{2}} \, \mathrm{d}x \right) \left( \frac{\Gamma\left(\frac{10}{4}\right) \Gamma\left(s + 2 - \frac{10}{4}\right)}{4\Gamma(s + 2)} \right) \right) \\ & = -\frac{4\operatorname{Vol}(\mathbb{S}^{5})}{6(2\pi)^{6}} \left( \frac{3}{8} \right) \left( \frac{1}{4} \right) \sum_{k=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2}V}{\partial x_{k}^{2}} \, \mathrm{d}x \right) \\ & = \frac{\operatorname{Vol}(\mathbb{S}^{5})}{16(2\pi)^{6}} \int_{\mathbb{R}^{6}} V(x) [\Delta V](x) \, \mathrm{d}x \\ & = \frac{\operatorname{Vol}(\mathbb{S}^{5})}{16(2\pi)^{6}} \int_{\mathbb{R}^{6}} |[\nabla V](x)|^{2} \, \mathrm{d}x. \end{split}$$

Next we consider the term  $\ell=1, k=0, p=2$ . Observe that  $Y_1(0,\frac{1}{2})=-\frac{1}{2}$  and

 $\int_0^1 (2t) dt = 1$ . Then we find

$$\operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \int_{0}^{1} \operatorname{Tr}(V X_{2}(\operatorname{Id} + H_{0}^{2})^{-s-1}) dt \right) \\
= \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \int_{0}^{1} 2t \operatorname{Tr}(V^{2} H_{0}(\operatorname{Id} + H_{0}^{2})^{-s-1}) dt \right) \\
= (2\pi)^{-6} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \left( \int_{\mathbb{R}^{6}} V(x)^{2} dx \right) \left( \int_{\mathbb{R}^{6}} |\xi|^{2} (1 + |\xi|^{4})^{-s-1} d\xi \right) \right) \\
= \frac{\operatorname{Vol}(\mathbb{S}^{5})}{(2\pi)^{6}} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \left( \int_{\mathbb{R}^{6}} V(x)^{2} dx \right) \left( \int_{0}^{\infty} r^{8-1} (1 + r^{4})^{-s-1} dr \right) \right) \\
= \frac{\operatorname{Vol}(\mathbb{S}^{5})}{(2\pi)^{6}} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(0,s) \left( \int_{\mathbb{R}^{6}} V(x)^{2} dx \right) \left( \frac{\Gamma\left(\frac{8}{4}\right) \Gamma\left(s + 1 - \frac{8}{4}\right)}{4\Gamma(s + 1)} \right) \right) \\
= 0,$$

since the final expression is holomorphic at  $s = \frac{1}{2}$ .

We next move to the  $\ell=1, k=1, p=2$  term. Observe that  $Y_1(1, \frac{1}{2})=\frac{3}{8}$ ,  $\int_0^1 (-8t) dt = -4$ ,  $\int_0^1 (4t) dt = 2$ ,  $\int_0^1 (8t) dt = 4$  and  $\operatorname{Res}_{s=\frac{1}{2}} \left(\Gamma\left(s-\frac{1}{2}\right)\right) = 1$ . Note that the operator  $X_2^{(1)}$  is an order 5 operator and thus we can write  $X_2^{(1)} = \sum_{j=0}^5 G_j$ , where the operator  $G_j$  has order j. We determine that the operator  $VG_j(\operatorname{Id} + H_0^2)^{-s-2}$  has order j-4s-8 and so gives a contribution which is holomorphic at  $s=\frac{1}{2}$  if j<4. So only the operators  $G_4$  and  $G_5$  can give a non-zero residue. We compute that

$$[H_0^2, X_2] = 2t[H_0^2, VH_0] = 2t[H_0^2, V]H_0 = 2tH_0[H_0, V]H_0 + 2t[H_0, V]H_0^2.$$

We have also the relation

$$[H_0, V] = (\Delta V) - 2 \sum_{k=1}^{6} \frac{\partial V}{\partial x_k} \frac{\partial}{\partial x_k}.$$

The highest order term of  $X_2^{(1)}$  can be read off directly as

$$G_5 = -8t \sum_{k=1}^{6} \frac{\partial V}{\partial x_k} \frac{\partial}{\partial x_k} H_0^2.$$

A more involved calculation shows that

$$G_4 = 4t(\Delta V)H_0^2 + 8t\sum_{i,k=1}^{6} \frac{\partial^2 V}{\partial x_i \partial x_k} \frac{\partial^2}{\partial x_j \partial x_k} H_0.$$

For ease of exposition we write  $G_4 = G_{4,1} + G_{4,2}$  where where

$$G_{4,1} = 4t(\Delta V)H_0^2$$
, and  $G_{4,2} = 8t \sum_{j,k=1}^6 \frac{\partial^2 V}{\partial x_j \partial x_k} \frac{\partial^2}{\partial x_j \partial x_k}$ .

So we compute that

$$\operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(1,s) \int_{0}^{1} \operatorname{Tr}(VG_{5}(\operatorname{Id} + H_{0}^{2})^{-s-2}) dt \right) \\
= \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(1,s) \int_{0}^{1} (-8t) \sum_{k=1}^{6} \operatorname{Tr} \left( V \left( \frac{\partial V}{\partial x_{k}} \frac{\partial}{\partial x_{k}} \right) H_{0}^{2} (\operatorname{Id} + H_{0}^{2})^{-s-2} \right) dt \right) \\
= i \frac{4}{(2\pi)^{6}} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(1,s) \sum_{k=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial V}{\partial x_{k}} dx \right) \left( \int_{\mathbb{R}^{6}} \xi_{k} |\xi|^{4} (1 + |\xi|^{4})^{-s-2} d\xi \right) \right) \\
= 0,$$

since the  $\xi$  integral contains the integral of an odd function over a symmetric region. The  $G_{4,1}$  contribution can be determined as

$$\begin{aligned} & \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(1,s) \int_{0}^{1} \operatorname{Tr}(VG_{4,1}(\operatorname{Id} + H_{0}^{2})^{-s-2}) \, \mathrm{d}t \right) \\ & = \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(1,s) \int_{0}^{1} 4t \operatorname{Tr}\left( V(\Delta V) H_{0}^{2}(\operatorname{Id} + H_{0}^{2})^{-s-2} \right) \, \mathrm{d}t \right) \\ & = \frac{2}{(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(1,s) \left( \int_{\mathbb{R}^{6}} V(x) [\Delta V](x) \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^{6}} |\xi|^{4} (1 + |\xi|^{4})^{-s-2} \, \mathrm{d}\xi \right) \right) \\ & = \frac{2\operatorname{Vol}(\mathbb{S}^{5})}{(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(1,s) \left( \int_{\mathbb{R}^{6}} V(x) [\Delta V](x) \, \mathrm{d}x \right) \left( \int_{0}^{\infty} r^{10-1} (1 + r^{4})^{-s-2} \, \mathrm{d}r \right) \right) \\ & = \frac{2\operatorname{Vol}(\mathbb{S}^{5})}{(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(1,s) \left( \int_{\mathbb{R}^{6}} |[\nabla V](x)|^{2} \, \mathrm{d}x \right) \left( \frac{\Gamma\left(\frac{10}{4}\right) \Gamma\left(s + 2 - \frac{10}{4}\right)}{4\Gamma(s + 2)} \right) \right) \\ & = \frac{2\operatorname{Vol}(\mathbb{S}^{5})}{(2\pi)^{6}} \left( \frac{3}{8} \right) \left( \frac{1}{4} \right) \int_{\mathbb{R}^{6}} |[\nabla V](x)|^{2} \, \mathrm{d}x \\ & = \frac{3\operatorname{Vol}(\mathbb{S}^{5})}{16(2\pi)^{6}} \int_{\mathbb{R}^{6}} |[\nabla V](x)|^{2} \, \mathrm{d}x. \end{aligned}$$

Next we determine the  $G_{4,2}$  contribution as

$$\operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(1,s) \int_{0}^{1} \operatorname{Tr}(VG_{4,2}(\operatorname{Id} + H_{0}^{2})^{-s-2}) \, dt \right) \\
= \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(1,s) \int_{0}^{1} 8t \sum_{j,k=1}^{6} \operatorname{Tr}\left( V \frac{\partial^{2}V}{\partial x_{j}\partial x_{k}} \frac{\partial^{2}}{\partial x_{j}\partial x_{k}} H_{0}(\operatorname{Id} + H_{0}^{2})^{-s-2} \right) dt \right) \\
= i^{2} \frac{4}{(2\pi)^{6}} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(1,s) \sum_{j,k=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2}V}{\partial x_{j}\partial x_{k}} \, dx \right) \left( \int_{\mathbb{R}^{6}} \xi_{j} \xi_{k} |\xi|^{2} (1 + |\xi|^{4})^{-s-2} \, d\xi \right) \right).$$

For  $j \neq k$  the  $\xi$  integrals vanish since they are of an odd function. So we obtain

$$\begin{split} & \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(1,s) \int_{0}^{1} \operatorname{Tr}(VG_{4,2}(\operatorname{Id} + H_{0}^{2})^{-s-2}) \, \mathrm{d}t \right) \\ & = i^{2} \frac{4}{(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(1,s) \sum_{k=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2}V}{\partial x_{k}^{2}} \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^{6}} \xi_{k}^{2} |\xi|^{2} (1 + |\xi|^{4})^{-s-2} \, \mathrm{d}\xi \right) \right) \\ & = -\frac{4}{6(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(1,s) \sum_{k=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2}V}{\partial x_{k}^{2}} \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^{6}} |\xi|^{4} (1 + |\xi|^{4})^{-s-2} \, \mathrm{d}\xi \right) \right) \\ & = -\frac{4\operatorname{Vol}(\mathbb{S}^{5})}{6(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(1,s) \sum_{k=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2}V}{\partial x_{k}^{2}} \, \mathrm{d}x \right) \left( \int_{0}^{\infty} r^{10-1} (1 + r^{4})^{-s-2} \, \mathrm{d}r \right) \right) \\ & = -\frac{4\operatorname{Vol}(\mathbb{S}^{5})}{6(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(1,s) \sum_{k=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2}V}{\partial x_{k}^{2}} \, \mathrm{d}x \right) \left( \frac{\Gamma\left(\frac{10}{4}\right) \Gamma\left(s + 2 - \frac{10}{4}\right)}{4\Gamma(s + 2)} \right) \right) \\ & = \frac{4\operatorname{Vol}(\mathbb{S}^{5})}{6(2\pi)^{6}} \left( \frac{3}{8} \right) \left( \frac{1}{4} \right) \left( \int_{\mathbb{R}^{6}} V(x) [\Delta V](x) \, \mathrm{d}x \right) \\ & = \frac{\operatorname{Vol}(\mathbb{S}^{5})}{16(2\pi)^{6}} \int_{\mathbb{R}^{6}} |[\nabla V](x)|^{2} \, \mathrm{d}x. \end{split}$$

Combining the  $G_{4,1}$  and  $G_{4,2}$  contributions we find

$$\operatorname{Res}_{s=\frac{1}{2}} \left( \int_{0}^{1} \operatorname{Tr}(V X_{2}^{(1)} (\operatorname{Id} + H_{0}^{2})^{-s-2}) dt \right) 
= \frac{3 \operatorname{Vol}(\mathbb{S}^{5})}{16(2\pi)^{6}} \int_{\mathbb{R}^{6}} |[\nabla V](x)|^{2} dx + \frac{\operatorname{Vol}(\mathbb{S}^{5})}{16(2\pi)^{6}} \int_{\mathbb{R}^{6}} |[\nabla V](x)|^{2} dx 
= \frac{\operatorname{Vol}(\mathbb{S}^{5})}{4(2\pi)^{6}} \int_{\mathbb{R}^{6}} |[\nabla V](x)|^{2} dx.$$

We now move to the  $\ell=1, k=2, p=2$  term. Observe that  $Y_1(s,\frac{1}{2})=-\frac{5}{16}$  and  $\int_0^1 (32t) \, \mathrm{d}t = 16$ . The operator  $X_2^{(2)}$  is an order 8 operator and thus can be written as  $X_2^{(2)} = \sum_{j=0}^8 G_j$  where each operator  $G_j$  has order j. The operator  $VG_j(\mathrm{Id} + H_0^2)^{-s-3}$  has order j-4s-12 and so gives a contribution which is holomorphic at  $s=\frac{1}{2}$  if j<8.

Thus only the operator  $G_8$  will contribute a non-zero residue. By writing

$$X_2^{(2)} = 2t[H_0^2, [H_0^2, VH_0]]$$
  
=  $2tH_0^2[H_0, [H_0, V]]H_0 + 4tH_0[H_0, [H_0, V]]H_0^2 + 2t[H_0, [H_0, V]]H_0^3,$ 

we can compute directly the highest order component of  $X_2^{(2)}$  as

$$G_8 = 32t \sum_{j,k=1}^{6} \frac{\partial^2 V}{\partial x_j \partial x_k} \frac{\partial^2}{\partial x_j \partial x_k} H_0^3.$$

So we find

$$\operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(2,s) \int_{0}^{1} \operatorname{Tr}(V X_{2}^{(2)} (\operatorname{Id} + H_{0}^{2})^{-s-3}) dt \right) \\
= \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(2,s) \int_{0}^{1} 32t \sum_{j=1}^{6} \operatorname{Tr}\left( V \frac{\partial^{2} V}{\partial x_{j}^{2}} \frac{\partial^{2}}{\partial x_{j}} H_{0}^{3} (\operatorname{Id} + H_{0}^{2})^{-s-3} \right) dt \right) \\
= i^{2} \frac{16}{(2\pi)^{6}} \operatorname{Res}_{s=\frac{1}{2}} \left( Y_{1}(2,s) \sum_{j,k=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2} V}{\partial x_{j} \partial x_{k}} dx \right) \left( \int_{\mathbb{R}^{6}} \xi_{j} \xi_{k} |\xi|^{6} (1 + |\xi|^{4})^{-s-3} d\xi \right) \right).$$

For  $j \neq k$  the  $\xi$  integral vanishes since it contains the integral of an odd function over a symmetric region. We thus compute

$$\begin{split} & \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(2,s) \int_{0}^{1} \operatorname{Tr}(V X_{2}^{(2)} (\operatorname{Id} + H_{0}^{2})^{-s-3}) \, \mathrm{d}t \right) \\ & = i^{2} \frac{16}{(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(2,s) \sum_{j=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2} V}{\partial x_{j}^{2}} \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^{6}} \xi_{j}^{2} |\xi|^{6} (1 + |\xi|^{4})^{-s-3} \, \mathrm{d}\xi \right) \right) \\ & = -\frac{16}{6(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(2,s) \sum_{j=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2} V}{\partial x_{j}^{2}} \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^{6}} |\xi|^{8} (1 + |\xi|^{4})^{-s-3} \, \mathrm{d}\xi \right) \right) \\ & = -\frac{16 \operatorname{Vol}(\mathbb{S}^{5})}{6(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(2,s) \sum_{j=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2} V}{\partial x_{j}^{2}} \, \mathrm{d}x \right) \left( \int_{0}^{\infty} r^{14-1} (1 + r^{4})^{-s-3} \, \mathrm{d}r \right) \right) \\ & = -\frac{16 \operatorname{Vol}(\mathbb{S}^{5})}{6(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{1}(2,s) \sum_{j=1}^{6} \left( \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2} V}{\partial x_{j}^{2}} \, \mathrm{d}x \right) \left( \frac{\Gamma\left(\frac{14}{4}\right) \Gamma\left(s + 3 - \frac{14}{4}\right)}{4\Gamma(s + 3)} \right) \right) \\ & = -\frac{16 \operatorname{Vol}(\mathbb{S}^{5})}{6(2\pi)^{6}} \left( -\frac{5}{16} \right) \left( \frac{1}{4} \right) \sum_{j=1}^{6} \int_{\mathbb{R}^{6}} V(x) \frac{\partial^{2} V}{\partial x_{j}^{2}} \, \mathrm{d}x \right) \\ & = -\frac{5 \operatorname{Vol}(\mathbb{S}^{5})}{24(2\pi)^{6}} \int_{\mathbb{R}^{6}} V(x) [\Delta V](x) \, \mathrm{d}x \\ & = -\frac{5 \operatorname{Vol}(\mathbb{S}^{5})}{24(2\pi)^{6}} \int_{\mathbb{R}^{6}} |[\nabla V](x)|^{2} \, \mathrm{d}x. \end{split}$$

Finally, we consider the  $\ell=2, k=(0,0), p=(2,2)$  term. Observe that  $Y_2(0,\frac{1}{2})=\frac{3}{8}$  and  $\int_0^1 4t^2 dt = \frac{4}{3}$ . The operator  $X_2^2$  is an order 4 operator and thus can be written as  $X_2^2 = \sum_{j=0}^4 G_j$  with each operator  $G_j$  of order j. The operator  $VG_j(\mathrm{Id} + H_0^2)^{-s-2}$  has order j-4s-8 and so gives a contribution which is holomorphic at  $s=\frac{1}{2}$  if j<4. Thus the only term which contributes a non-zero residue is  $G_4$ , which can be computed easily as

$$G_4 = 4t^2 V^2 H_0^2$$

So we compute

$$\begin{aligned} & \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{2}(0,s) \int_{0}^{1} \operatorname{Tr}(V X_{2}^{2} (\operatorname{Id} + H_{0}^{2})^{-s-2}) \, \mathrm{d}t \right) \\ & = \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{2}(0,s) \int_{0}^{1} 4t^{2} \operatorname{Tr}(V^{3} H_{0}^{2} (\operatorname{Id} + H_{0}^{2})^{-s-2}) \, \mathrm{d}t \right) \\ & = \frac{4}{3(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{2}(0,s) \left( \int_{\mathbb{R}^{6}} V(x)^{3} \, \mathrm{d}x \right) \left( \int_{\mathbb{R}^{6}} |\xi|^{4} (1 + |\xi|^{4})^{-s-2} \, \mathrm{d}\xi \right) \right) \\ & = \frac{4 \operatorname{Vol}(\mathbb{S}^{5})}{3(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{2}(0,s) \left( \int_{\mathbb{R}^{6}} V(x)^{3} \, \mathrm{d}x \right) \left( \int_{0}^{\infty} r^{10-1} (1 + r^{4})^{-s-2} \, \mathrm{d}r \right) \right) \\ & = \frac{4 \operatorname{Vol}(\mathbb{S}^{5})}{3(2\pi)^{6}} \underset{s=\frac{1}{2}}{\operatorname{Res}} \left( Y_{2}(0,s) \left( \int_{\mathbb{R}^{6}} V(x)^{3} \, \mathrm{d}x \right) \left( \frac{\Gamma\left(\frac{10}{4}\right) \Gamma\left(s + 2 - \frac{10}{4}\right)}{4\Gamma(s + 2)} \right) \right) \\ & = \frac{4 \operatorname{Vol}(\mathbb{S}^{5})}{3(2\pi)^{6}} \left( \frac{3}{8} \right) \left( \frac{1}{4} \right) \int_{\mathbb{R}^{6}} V(x)^{3} \, \mathrm{d}x \\ & = \frac{\operatorname{Vol}(\mathbb{S}^{5})}{8(2\pi)^{6}} \int_{\mathbb{R}^{6}} V(x)^{3} \, \mathrm{d}x. \end{aligned}$$

## **Bibliography**

- [1] M. Abramowitz and I. Stegun, *Handbook of Mathematical Functions: with Formulas, Graphs, and Mathematical Tables* (Applied Mathematics Series 55), 10th ed. NBS, 1972.
- [2] R. A. Adams, *Sobolev spaces* (Pure and Applied Mathematics, Vol. 65). Academic Press, New York-London, 1975, pp. xviii+268.
- [3] S. Agmon, Spectral properties of Schrödinger operators and scattering theory. Accademia Nazionale dei Lincei; Scuola Normale Superiore di Pisa, Pisa, 1975, pp. 150–218.
- [4] S. Albeverio, F. Gesztesy, R. Høegh-Krohn, and H. Holden, *Solvable models in quantum mechanics* (Texts and Monographs in Physics). Springer-Verlag, New York, 1988, pp. xiv+452.
- [5] A. Alexander and A. Rennie, Levinson's theorem as an index pairing, 2023. eprint: arXiv:2304.04905.
- [6] P. Alsholm and G. Schmidt, "Spectral and scattering theory for Schrödinger operators," Arch. Rational Mech. Anal., vol. 40, pp. 281–311, 1971.
- W. Amrein, J. M. Jauch, and K. B. Sinha, Scattering theory in quantum mechanics.
   W. A. Benjamin, Inc., 1977, pp. xxiii+691, Physical principles and mathematical methods, Lecture Notes and Supplements in Physics, No. 16.
- [8] W. Amrein, A. Boutet de Monvel, and V. Georgescu, C<sub>0</sub>-Groups, Commutator Methods and Spectral Theory of N-Body Hamiltonians (Modern Birkhäuser Classics), 1st ed. Birkhäuser, 1996.
- [9] G. Andrews, R. Askey, and R. Roy, Special functions (Encyclopedia of mathematics and its applications 71). Cambridge University Press, 1999.
- [10] M. F. Atiyah, V. K. Patodi, and I. M. Singer, "Spectral asymmetry and Riemannian geometry. I," *Math. Proc. Cambridge Philos. Soc.*, vol. 77, pp. 43–69, 1975.
- [11] M. F. Atiyah, V. K. Patodi, and I. M. Singer, "Spectral asymmetry and Riemannian geometry. III," Math. Proc. Cambridge Philos. Soc., vol. 79, no. 1, pp. 71–99, 1976.

[12] N. A. Azamov, "Spectral flow inside essential spectrum," *Dissertationes Math.*, vol. 518, p. 156, 2016.

- [13] N. A. Azamov, A. L. Carey, P. G. Dodds, and F. A. Sukochev, "Operator integrals, spectral shift, and spectral flow," *Canad. J. Math.*, vol. 61, no. 2, pp. 241–263, 2009.
- [14] N. A. Azamov, A. L. Carey, and F. A. Sukochev, "The spectral shift function and spectral flow," *Comm. Math. Phys.*, vol. 276, no. 1, pp. 51–91, 2007.
- [15] R. Bañuelos and A. Sá Barreto, "On the heat trace of Schrödinger operators," Comm. Partial Differential Equations, vol. 20, no. 11-12, pp. 2153–2164, 1995.
- [16] S. Baaj and P. Julg, "Théorie bivariante de Kasparov et opérateurs non bornés dans les C\*-modules hilbertiens," C. R. Acad. Sci. Paris Sér. I Math., vol. 296, no. 21, pp. 875–878, 1983.
- [17] F. Bagemihl, "The Plemelj formulas with unrestricted approach, and the continuity of Cauchy-type integrals," *Proc. Nat. Acad. Sci. U.S.A.*, vol. 55, pp. 1046–1048, 1966.
- [18] H. Baumgärtel and M. Wollenberg, *Mathematical scattering theory* (Operator Theory: Advances and Applications). Birkhäuser Verlag, Basel, 1983, vol. 9.
- [19] B. Belchev, S. Neale, and M. Walton, "Flow of S-matrix poles for elementary quantum potentials," *Canadian Journal of Physics*, vol. 89, no. 11, pp. 1127–1140, 2011.
- [20] M.-T. Benameur, A. Carey, J. Phillips, A. Rennie, F. Sukochev, and K. Wojciechowski, "An analytic approach to spectral flow in von Neumann algebras," in Analysis, geometry and topology of elliptic operators, World Sci. Publ., Hackensack, NJ, 2006, pp. 297–352.
- [21] F. A. Berezin and M. A. Shubin, *The Schrödinger equation* (Mathematics and its Applications (Soviet Series)). Kluwer Academic Publishers Group, Dordrecht, 1991, vol. 66, pp. xviii+555, Translated from the 1983 Russian edition by Yu. Rajabov, D. A. Leĭtes and N. A. Sakharova and revised by Shubin, With contributions by G. L. Litvinov and Leĭtes.
- [22] E. Beth and G. E. Uhlenbeck, "The quantum theory of the non-ideal gas. ii. behaviour at low temperatures," *Physica*, vol. 4, no. 10, pp. 915–924, 1937.
- [23] M. Birman and M. Solomjak, Spectral Theory of Self-Adjoint Operators in Hilbert Space (Mathematics and its Applications), 1st ed. Springer, 1987.
- [24] M. v. Birman, "On the spectrum of singular boundary-value problems," Mat.~Sb. (N.S.), vol. 55(97), pp. 125–174, 1961.
- [25] M. v. Birman and M. G. Kreĭn, "On the theory of wave operators and scattering operators," *Dokl. Akad. Nauk SSSR*, vol. 144, pp. 475–478, 1962.

[26] D. Bollé, F. Gesztesy, and C. Danneels, "Threshold scattering in two dimensions," Ann. Inst. H. Poincaré Phys. Théor., vol. 48, no. 2, pp. 175–204, 1988.

- [27] D. Bollé, F. Gesztesy, C. Danneels, and S. F. J. Wilk, "Threshold behavior and Levinson's theorem for two-dimensional scattering systems: A surprise," *Phys. Rev. Lett.*, vol. 56, pp. 900–903, 9 1986.
- [28] D. Bollé, F. Gesztesy, H. Grosse, W. Schweiger, and B. Simon, "Witten index, axial anomaly, and Kreĭn's spectral shift function in supersymmetric quantum mechanics," *J. Math. Phys.*, vol. 28, no. 7, pp. 1512–1525, 1987.
- [29] D. Bollé, F. Gesztesy, and W. Schweiger, "Scattering theory for long-range systems at threshold," *J. Math. Phys.*, vol. 26, no. 7, pp. 1661–1674, 1985.
- [30] D. Bollé, F. Gesztesy, and S. F. J. Wilk, "A complete treatment of low-energy scattering in one dimension," *J. Operator Theory*, vol. 13, no. 1, pp. 3–31, 1985.
- [31] D. Bollé and T. A. Osborn, "An extended Levinson's theorem," *J. Mathematical Phys.*, vol. 18, no. 3, pp. 432–440, 1977.
- [32] N. V. Borisov, W. Müller, and R. Schrader, "Relative index theorems and supersymmetric scattering theory," *Comm. Math. Phys.*, vol. 114, no. 3, pp. 475–513, 1988.
- [33] T. Britz, A. Carey, F. Gesztesy, R. Nichols, F. Sukochev, and D. Zanin, "The product formula for regularized Fredholm determinants," *Proc. Amer. Math. Soc. Ser. B*, vol. 8, pp. 42–51, 2021.
- [34] E. Brüning and F. Gesztesy, "Continuity of wave and scattering operators with respect to interactions," *J. Math. Phys.*, vol. 24, no. 6, pp. 1516–1528, 1983.
- [35] P. Burke, Potential scattering in atomic physics. Plenum, 1977.
- [36] V. S. Buslaev, "Trace formulas for the Schrödinger operator in a three-dimensional space.," *Dokl. Akad. Nauk SSSR*, pp. 1067–1070, 1962.
- [37] A. Carey, F. Gesztesy, H. Grosse, G. Levitina, D. Potapov, F. Sukochev, and D. Zanin, "Trace formulas for a class of non-Fredholm operators: A review," Rev. Math. Phys., vol. 28, no. 10, pp. 1630002, 55, 2016.
- [38] A. Carey, F. Gesztesy, G. Levitina, R. Nichols, F. Sukochev, and D. Zanin, The limiting absorption principle for massless Dirac operators, properties of spectral shift functions, and an application to the Witten index of non-Fredholm operators (Memoirs of the European Mathematical Society). European Mathematical Society (EMS), Berlin, 2023, vol. 4, pp. viii+213.
- [39] A. Carey, F. Gesztesy, G. Levitina, D. Potapov, F. Sukochev, and D. Zanin, "On index theory for non-Fredholm operators: A (1+1)-dimensional example," Math. Nachr., vol. 289, no. 5-6, pp. 575–609, 2016.

[40] A. Carey and G. Levitina, *Index Theory Beyond the Fredholm Case* (Lecture Notes in Mathematics). Springer, 2022, vol. 2323, pp. xiv+175.

- [41] A. Carey and G. Levitina, On a conjectured property of the Witten index and an application to Levinson's theorem, 2023. eprint: arXiv:2308.06864.
- [42] A. Carey, G. Levitina, D. Potapov, and F. Sukochev, "The Witten index and the spectral shift function," *Rev. Math. Phys.*, vol. 34, no. 5, Paper No. 2250011, 44, 2022.
- [43] A. L. Carey, V. Gayral, A. Rennie, and F. A. Sukochev, "Index theory for locally compact noncommutative geometries," *Mem. Amer. Math. Soc.*, vol. 231, pp. vi+130, 2014.
- [44] A. L. Carey and J. Phillips, "Unbounded Fredholm modules and spectral flow," *Canad. J. Math.*, vol. 50, no. 4, pp. 673–718, 1998.
- [45] A. L. Carey, J. Phillips, and A. Rennie, "Spectral triples: Examples and index theory," in *Noncommutative geometry and physics: renormalisation, motives, index theory*, ser. ESI Lect. Math. Phys. Eur. Math. Soc., Zürich, 2011, pp. 175–265.
- [46] A. L. Carey, J. Phillips, A. Rennie, and F. A. Sukochev, "The local index formula in semifinite von Neumann algebras. I. Spectral flow," *Adv. Math.*, vol. 202, no. 2, pp. 451–516, 2006.
- [47] A. L. Carey, D. Potapov, and F. Sukochev, "Spectral flow is the integral of one forms on the Banach manifold of self adjoint Fredholm operators," *Adv. Math.*, vol. 222, no. 5, pp. 1809–1849, 2009.
- [48] C. A. A. de Carvalho and H. M. Nussenzveig, "Time delay," Phys. Rep., vol. 364, no. 2, pp. 83–174, 2002.
- [49] D. Çevik, M. Gadella, c. Kuru, and J. Negro, "Resonances and antibound states for the Pöschl-Teller potential: Ladder operators and SUSY partners," *Phys. Lett.* A, vol. 380, no. 18-19, pp. 1600–1609, 2016.
- [50] M. Cheney, "Two-dimensional scattering: The number of bound states from scattering data," J. Math. Phys., vol. 25, no. 5, pp. 1449–1455, 1984.
- [51] B. Cheng and M. Melgaard, "Poisson wave trace formula for Dirac resonances at spectrum edges and applications," *Asian J. Math.*, vol. 25, no. 2, pp. 243–276, 2021.
- [52] T. J. Christiansen, "Resonant rigidity for Schrödinger operators in even dimensions," *Ann. Henri Poincaré*, vol. 20, no. 5, pp. 1543–1582, 2019.
- [53] J. M. Cook, "Convergence to the Møller wave-matrix," J. Math. and Phys., vol. 36, pp. 82–87, 1957.

[54] E. B. Davies, "Energy dependence of the scattering operator," *Advances in applied mathematics*, vol. 1, no. 3, pp. 300–323, 1980.

- [55] J. Duoandikoetxea, Fourier analysis (Graduate Studies in Mathematics). American Mathematical Society, Providence, RI, 2001, vol. 29, pp. xviii+222.
- [56] S. Dyatlov and M. Zworski, *Mathematical theory of scattering resonances* (Graduate Studies in Mathematics). American Mathematical Society, Providence, RI, 2019, vol. 200, pp. xi+634.
- [57] L. Eisenbud and E. P. Wigner, "Higher angular momenta and long range interaction in resonance reactions," *Phys. Rev.*, vol. 72, pp. 29–41, 1947.
- [58] V. Enss, "Asymptotic completeness for quantum mechanical potential scattering.
   I. Short range potentials," Comm. Math. Phys., vol. 61, no. 3, pp. 285–291, 1978.
- [59] V. Enss, "Asymptotic completeness for quantum-mechanical potential scattering: II. singular and long-range potentials," *Annals of Physics*, vol. 119, 1979.
- [60] V. Enss, "Geometric methods in scattering theory," in New developments in mathematical physics (Schladming, 1981), ser. Acta Phys. Austriaca Suppl., XXIII, Springer, Vienna, 1981, pp. 29–63.
- [61] A. Erdélyi, W. Magnus, F. Oberhettinger, and F. G. Tricomi, *Higher transcendental functions. Vol. II.* Robert E. Krieger Publishing Co., 1981, pp. xviii+396.
- [62] M. B. Erdoğan, M. Goldberg, and W. R. Green, "Dispersive estimates for four dimensional Schrödinger and wave equations with obstructions at zero energy," *Comm. Partial Differential Equations*, vol. 39, no. 10, pp. 1936–1964, 2014.
- [63] M. B. Erdoğan and W. R. Green, "A weighted dispersive estimate for Schrödinger operators in dimension two," Comm. Math. Phys., vol. 319, no. 3, pp. 791–811, 2013.
- [64] M. B. Erdoğan and W. Schlag, "Dispersive estimates for Schrödinger operators in the presence of a resonance and/or an eigenvalue at zero energy in dimension three. I," *Dyn. Partial Differ. Equ.*, vol. 1, no. 4, pp. 359–379, 2004.
- [65] G. B. Folland, "How to integrate a polynomial over a sphere," *Amer. Math. Monthly*, vol. 108, no. 5, pp. 446–448, 2001.
- [66] K. O. Friedrichs, "On the perturbation of continuous spectra," Communications on Appl. Math., vol. 1, pp. 361–406, 1948.
- [67] K. Friedrichs, "Über die Spektralzerlegung eines Integraloperators," *Math. Ann.*, vol. 115, no. 1, pp. 249–272, 1938.
- [68] V. Georgescu and A. Iftimovici, "C\*-algebras of quantum Hamiltonians," in *Operator algebras and mathematical physics (Constanța, 2001)*, Theta, Bucharest, 2003, pp. 123–167.

[69] I. C. Gohberg and M. G. Kreĭn, "The basic propositions on defect numbers, root numbers and indices of linear operators," Amer. Math. Soc. Transl. (2), vol. 13, pp. 185–264, 1960.

- [70] I. C. Gohberg and M. G. Kreĭn, Introduction to the theory of linear nonselfadjoint operators (Translations of Mathematical Monographs, Vol. 18). American Mathematical Society, Providence, R.I., 1969, pp. xv+378, Translated from the Russian by A. Feinstein.
- [71] M. Goldberg and W. R. Green, "Dispersive estimates for higher dimensional Schrödinger operators with threshold eigenvalues I: The odd dimensional case," J. Funct. Anal., vol. 269, no. 3, pp. 633–682, 2015.
- [72] M. Goldberg and W. R. Green, "Dispersive estimates for higher dimensional Schrödinger operators with threshold eigenvalues II. The even dimensional case," J. Spectr. Theory, vol. 7, no. 1, pp. 33–86, 2017.
- [73] L. Guillopé, "Asymptotique de la phase de diffusion pour l'opérateur de Schrödinger avec potentiel," C. R. Acad. Sci. Paris Sér. I Math., vol. 293, no. 12, pp. 601–603, 1981.
- [74] B. Hall, Quantum Theory for Mathematicians (Graduate Texts in Mathematics 267), 1st ed. Springer-Verlag New York, 2013.
- [75] W. Heisenberg, "Die "beobachtbaren größen" in der theorie der elementarteilchen," Zeitschrift für Physik A Hadrons and Nuclei, vol. 120, 7-10 1943.
- [76] N. Higson and J. Roe, *Analytic K-homology* (Oxford Mathematical Monographs). Oxford University Press, Oxford, 2000, pp. xviii+405, Oxford Science Publications.
- [77] J. R. Holub, "Compactness in topological tensor products and operator spaces," *Proc. Amer. Math. Soc.*, vol. 36, 398–406 (1973), 1972.
- [78] L. Hörmander, "The existence of wave operators in scattering theory," *Math. Z.*, vol. 146, no. 1, pp. 69–91, 1976.
- [79] W. Ignatowski, "Reflexion elektromagnetischer Wellen an einem Draft," Annalen der Physik, vol. 18, 1905.
- [80] J. M. Jauch, "Theory of the scattering operator," *Helv. Phys. Acta*, vol. 31, pp. 127–158, 1958.
- [81] J. M. Jauch, "Theory of the scattering operator. II. Multichannel scattering," *Helv. Phys. Acta*, vol. 31, 1958.
- [82] J. M. Jauch and J. P. Marchand, "The delay time operator for simple scattering systems," *Helv. Phys. Acta*, vol. 40, pp. 217–229, 1967.
- [83] J. M. Jauch, K. B. Sinha, and B. N. Misra, "Time-delay in scattering processes," Helv. Phys. Acta, vol. 45, pp. 398–426, 1972/73.

[84] A. Jeffrey and H.-H. Dai, *Handbook of mathematical formulas and integrals*, Fourth. Elsevier/Academic Press, Amsterdam, 2008, pp. xlvi+541.

- [85] A. Jensen, "Spectral properties of Schrödinger operators and time-decay of the wave functions results in  $L^2(\mathbf{R}^m)$ ,  $m \geq 5$ ," Duke Math. J., vol. 47, no. 1, pp. 57–80, 1980.
- [86] A. Jensen, "Time-delay in potential scattering theory. Some "geometric" results," *Comm. Math. Phys.*, vol. 82, no. 3, pp. 435–456, 1981/82.
- [87] A. Jensen, "Spectral properties of Schrödinger operators and time-decay of the wave functions. Results in  $L^2(\mathbf{R}^4)$ ," J. Math. Anal. Appl., vol. 101, no. 2, pp. 397–422, 1984.
- [88] A. Jensen and T. Kato, "Spectral properties of Schrödinger operators and time-decay of the wave functions," *Duke Math. J.*, vol. 46, no. 3, pp. 583–611, 1979.
- [89] A. Jensen and G. Nenciu, "A unified approach to resolvent expansions at thresholds," *Rev. Math. Phys.*, vol. 13, no. 6, pp. 717–754, 2001.
- [90] X. Jia, F. Nicoleau, and X. Wang, "A new Levinson's theorem for potentials with critical decay," *Ann. Henri Poincaré*, vol. 13, no. 1, pp. 41–84, 2012.
- [91] R. Jost, "Über die falschen Nullstellen der Eigenwerte der S-Matrix," Helvetica Phys. Acta, vol. 20, pp. 256–266, 1947.
- [92] T. Kato, "On finite-dimensional perturbations of self-adjoint operators," *J. Math. Soc. Japan*, vol. 9, pp. 239–249, 1957.
- [93] T. Kato, "Growth properties of solutions of the reduced wave equation with a variable coefficient," Comm. Pure Appl. Math., vol. 12, pp. 403–425, 1959.
- [94] T. Kato, "Wave operators and similarity for some non-selfadjoint operators," *Math. Ann.*, vol. 162, pp. 258–279, 1965/66.
- [95] T. Kato, *Perturbation theory for linear operators* (Classics in Mathematics). Springer-Verlag, Berlin, 1995, pp. xxii+619, Reprint of the 1980 edition.
- [96] J. Kellendonk and S. Richard, "Levinson's theorem for Schrödinger operators with point interaction: A topological approach," J. Phys. A, vol. 39, no. 46, pp. 14397– 14403, 2006.
- [97] J. Kellendonk and S. Richard, "On the structure of the wave operators in one dimensional potential scattering," *Math. Phys. Electron. J.*, vol. 14, Paper 3, 21, 2008.
- [98] J. Kellendonk and S. Richard, "On the wave operators and Levinson's theorem for potential scattering in  $\mathbb{R}^3$ ," Asian-Eur. J. Math., vol. 5, no. 1, pp. 1250004, 22, 2012.

[99] N. Koutsonikos-Kouloumpis and M. Lesch, The product formula for regularized fredholm determinants: Two new proofs, 2022. eprint: arXiv:2202.12923.

- [100] M. G. Kreĭn, "On the trace formula in perturbation theory," Mat. Sbornik N.S., vol. 33(75), pp. 597–626, 1953.
- [101] M. G. Kreĭn, "On perturbation determinants and a trace formula for unitary and self-adjoint operators," *Dokl. Akad. Nauk SSSR*, vol. 144, pp. 268–271, 1962.
- [102] S. T. Kuroda, "Finite-dimensional perturbation and a representation of scattering operator.," *Pacific J. Math.*, vol. 13, no. 4, pp. 1305–1318, 1963.
- [103] S. T. Kuroda, "An abstract stationary approach to perturbation of continuous spectra and scattering theory," *J. Analyse Math.*, vol. 20, pp. 57–117, 1967.
- [104] S. T. Kuroda, "Scattering theory for differential operators. I. Operator theory," *J. Math. Soc. Japan*, vol. 25, pp. 75–104, 1973.
- [105] S. T. Kuroda, "Scattering theory for differential operators. II. Self-adjoint elliptic operators," J. Math. Soc. Japan, vol. 25, pp. 222–234, 1973.
- [106] S. T. Kuroda, An introduction to scattering theory (Lecture Notes Series). Aarhus Universitet, Matematisk Institut, Aarhus, 1978, vol. 51.
- [107] M. Lesch, "The uniqueness of the spectral flow on spaces of unbounded self-adjoint Fredholm operators," in *Spectral geometry of manifolds with boundary and decomposition of manifolds*, ser. Contemp. Math. Vol. 366, Amer. Math. Soc., Providence, RI, 2005, pp. 193–224.
- [108] N. Levinson, "On the uniqueness of the potential in a Schrödinger equation for a given asymptotic phase," *Danske Vid. Selsk. Mat.-Fys. Medd.*, vol. 25, no. 9, pp. 1–29, 1949.
- [109] I. Lifshits, "On a problem of the theory of perturbations connected with quantum statistics," *Uspekhi Mat. Nauk*, vol. 7, no. 1, pp. 171–180, 1952.
- [110] J.-L. Lions and E. Magenes, Problèmes aux limites non homogènes et applications. Vol. 1 (Travaux et Recherches Mathématiques, No. 17). Dunod, Paris, 1968, pp. xx+372.
- [111] B. A. Lippmann, "Variational principles for scattering processes. II. Scattering of slow neutrons by parahydrogen," *Phys. Rev.* (2), vol. 79, pp. 481–486, 1950.
- [112] B. A. Lippmann and J. Schwinger, "Variational principles for scattering processes. I," *Phys. Rev.* (2), vol. 79, pp. 469–480, 1950.
- [113] P. Macaulay-Owen, "Parseval's theorem for Hankel transforms," *Proc. London Math. Soc.*, vol. 45, pp. 458–474, 1939.

[114] P. A. Martin, "On the time-delay of simple scattering systems," Comm. Math. Phys., vol. 47, no. 3, pp. 221–227, 1976.

- [115] P. A. Martin, "Time delay of quantum scattering processes," in *New developments in mathematical physics (Schladming, 1981)*, ser. Acta Phys. Austriaca Suppl., XXIII, Springer, Vienna, 1981, pp. 157–208.
- [116] M. Melgaard, "Spectral properties in the low-energy limit of one-dimensional Schrödinger operators  $H = -d^2/dx^2 + V$ . The case  $\langle 1, V1 \rangle \neq 0$ ," Math. Nachr., vol. 238, pp. 113–143, 2002.
- [117] C. Møller, "General properties of the characteristic matrix in the theory of elementary particles. I," Danske Vid. Selsk. Mat.-Fys. Medd., vol. 23, no. 1, pp. 1–46, 1946.
- [118] W. Müller, Manifolds with cusps of rank one (Lecture Notes in Mathematics). Springer-Verlag, Berlin, 1987, vol. 1244, pp. xii+158, Spectral theory and  $L^2$ -index theorem.
- [119] W. Müller, "Relative zeta functions, relative determinants and scattering theory," *Comm. Math. Phys.*, vol. 192, no. 2, pp. 309–347, 1998.
- [120] M. Murata, "High energy resolvent estimates. I. First order operators," *J. Math. Soc. Japan*, vol. 35, no. 4, pp. 711–733, 1983.
- [121] N. I. Muskhelishvili, *Singular integral equations*, Russian. Dover Publications, Inc., New York, 1992, p. 447, Boundary problems of function theory and their application to mathematical physics, Corrected reprint of the 1953 English translation.
- [122] H. Narnhofer, "Another definition for time delay," Phys. Rev. D (3), vol. 22, no. 10, pp. 2387–2390, 1980.
- [123] R. G. Newton, "Analytic properties of radial wave functions," *J. Mathematical Phys.*, vol. 1, pp. 319–347, 1960.
- [124] R. G. Newton, "Noncentral potentials: The generalized Levinson theorem and the structure of the spectrum," J. Mathematical Phys., vol. 18, no. 7, pp. 1348–1357, 1977.
- [125] F. Oberhettinger, Tables of Bessel transforms. Springer-Verlag, New York-Heidelberg, 1972, pp. ix+289.
- [126] F. Oberhettinger, Tables of Mellin transforms. Springer-Verlag, New York-Heidelberg, 1974, pp. v+275.
- [127] D. B. Pearson, Quantum scattering and spectral theory (Techniques of Physics). Academic Press, Inc., 1988, vol. 9, pp. xii+519.
- [128] P. A. Perry, "Mellin transforms and scattering theory. I. Short range potentials," Duke Math. J., vol. 47, no. 1, pp. 187–193, 1980.

[129] J. Phillips, "Self-adjoint Fredholm operators and spectral flow," Canad. Math. Bull., vol. 39, no. 4, pp. 460–467, 1996.

- [130] J. Phillips, "Spectral flow in type I and II factors—a new approach," in *Cyclic cohomology and noncommutative geometry (Waterloo, ON, 1995)*, ser. Fields Inst. Commun. Vol. 17, Amer. Math. Soc., Providence, RI, 1997, pp. 137–153.
- [131] J. Plemelj, *Problems in the sense of Riemann and Klein* (Interscience Tracts in Pure and Applied Mathematics, No. 16). Interscience Publishers John Wiley & Sons, Inc., New York-London-Sydney, 1964, pp. vii+175, Edited and translated by J. R. M. Radok.
- [132] G. S. Popov, "Asymptotic behaviour of the scattering phase for the Schrödinger operator," C. R. Acad. Bulgare Sci., vol. 35, no. 7, pp. 885–888, 1982.
- [133] J. B. Prolla, Approximation of vector valued functions (North-Holland Mathematics Studies, Vol. 25). North-Holland Publishing Co., Amsterdam-New York-Oxford, 1977, pp. xiii+219.
- [134] M. Reed and B. Simon, Methods of modern mathematical physics. I. Functional analysis. Academic Press, New York-London, 1972, pp. xvii+325.
- [135] M. Reed and B. Simon, *Methods of modern mathematical physics. II. Fourier analysis, self-adjointness.* Academic Press [Harcourt Brace Jovanovich, Publishers], New York-London, 1975, pp. xv+361.
- [136] M. Reed and B. Simon, *Methods of modern mathematical physics. IV. Analysis of operators*. Academic Press [Harcourt Brace Jovanovich, Publishers], New York-London, 1978.
- [137] M. Reed and B. Simon, *Methods of Modern Mathematical Physics: Scattering the-ory* (Methods of Modern Mathematical Physics). Academic Press, 1979.
- [138] S. Richard, "Levinson's theorem: An index theorem in scattering theory," in *Spectral theory and mathematical physics*, ser. Oper. Theory Adv. Appl. Vol. 254, Birkhäuser/Springer, 2016, pp. 149–203.
- [139] S. Richard and R. Tiedra de Aldecoa, "Time delay is a common feature of quantum scattering theory," J. Math. Anal. Appl., vol. 387, no. 2, pp. 618–630, 2012.
- [140] S. Richard and R. Tiedra de Aldecoa, "Explicit formulas for the Schrödinger wave operators in  $\mathbb{R}^2$ ," C. R. Math. Acad. Sci. Paris, vol. 351, no. 5-6, pp. 209–214, 2013.
- [141] S. Richard and R. Tiedra de Aldecoa, "New expressions for the wave operators of Schrödinger operators in  $\mathbb{R}^3$ ," Lett. Math. Phys., vol. 103, no. 11, pp. 1207–1221, 2013.

[142] S. Richard, R. Tiedra de Aldecoa, and L. Zhang, "Scattering operator and wave operators for 2D Schrödinger operators with threshold obstructions," *Complex Anal. Oper. Theory*, vol. 15, no. 6, Paper No. 106, 25, 2021.

- [143] D. Robert, "Semi-classical and high energy asymptotics of the scattering phase for perturbations of elliptic operators," in *Mathematical physics*, X (Leipzig, 1991), Springer, Berlin, 1992, pp. 442–446.
- [144] D. Robert, "Relative time-delay for perturbations of elliptic operators and semiclassical asymptotics," *J. Funct. Anal.*, vol. 126, no. 1, pp. 36–82, 1994.
- [145] M. Rørdam, F. Larsen, and N. Laustsen, An introduction to K-theory for C\*-algebras (London Mathematical Society Student Texts). Cambridge University Press, Cambridge, 2000, vol. 49, pp. xii+242.
- [146] M. Schechter, Operator methods in quantum mechanics. North-Holland Publishing Co., New York-Amsterdam, 1981, pp. xx+324.
- [147] E. Schrödinger, "An undulatory theory of the mechanics of atoms and molecules," *Physical Review (Series I)*, vol. 28, 6 Dec. 1926.
- [148] J. Schwinger, "On the bound states of a given potential," *Proc. Nat. Acad. Sci. U.S.A.*, vol. 47, pp. 122–129, 1961.
- [149] B. Simon, Quantum mechanics for Hamiltonians defined as quadratic forms. Princeton University Press, Princeton, N. J., 1971, pp. xv+244, Princeton Series in Physics.
- [150] B. Simon, "Phase space analysis of simple scattering systems: Extensions of some work of Enss," *Duke Math. J.*, vol. 46, no. 1, pp. 119–168, 1979.
- [151] B. Simon, *Trace Ideals and Their Applications* (Mathematical surveys and monographs). American Mathematical Society, 2005.
- [152] F. T. Smith, "Lifetime matrix in collision theory," Phys. Rev., vol. 118, pp. 349–356, 1 1960.
- [153] E. M. Stein, Singular Integrals and Differentiability Properties of Functions (Princeton mathematical series 30). Princeton University Press, 1970.
- [154] E. M. Stein and G. Weiss, *Introduction to Fourier analysis on Euclidean spaces* (PMS-32). Princeton University Press, 1971.
- [155] A. Strohmaier and A. Waters, "Geometric and obstacle scattering at low energy," Communications in Partial Differential Equations, pp. 1–61, Jun. 2020.
- [156] A. G. Svešnikov, "The principle of radiation.," *Doklady Akad. Nauk SSSR (N.S.)*, pp. 917–920, 1950.

[157] A. G. Svešnikov, "The principle of limiting absorption for a wave-guide.," *Doklady Akad. Nauk SSSR (N.S.)*, pp. 345–347, 1951.

- [158] G. Teschl, Mathematical methods in quantum mechanics (Graduate Studies in Mathematics), Second. American Mathematical Society, Providence, RI, 2014, vol. 157, pp. xiv+358, With applications to Schrödinger operators.
- [159] B. R. Vaĭnberg, "Principles of radiation, limiting absorption and limiting amplitude in the general theory of partial differential equations," *Uspehi Mat. Nauk*, vol. 21, no. 3(129), pp. 115–194, 1966.
- [160] Y. Colin de Verdière, "Une formule de traces pour l'opérateur de Schrödinger dans  $\mathbb{R}^3$ ," Ann. Sci. École Norm. Sup. (4), vol. 14, no. 1, pp. 27–39, 1981.
- [161] R. L. Wheeden and A. Zygmund, *Measure and integral* (Pure and Applied Mathematics (Boca Raton)), Second. CRC Press, Boca Raton, FL, 2015, pp. xvii+514.
- [162] J. A. Wheeler, "On the mathematical description of light nuclei by the method of resonating group structure," *Physical Review (Series I)*, vol. 52, 1937.
- [163] E. P. Wigner, "On the behavior of cross sections near thresholds," *Phys. Rev.*, vol. 73, pp. 1002–1009, 9 1948.
- [164] D. Yafaev, Mathematical scattering theory: Analytic theory (Mathematical surveys and monographs). American Mathematical Society, 2010, vol. 158.
- [165] D. Yafaev, Mathematical Scattering Theory: General Theory. American Mathematical Soc., 1992.
- [166] K. Yajima, "The  $L^p$ -boundedness of wave operators for two dimensional Schrödinger operators with threshold singularities," J. Math. Soc. Japan, vol. 74, no. 4, pp. 1169–1217, 2022.